# UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

# **FORM 10-Q**

# ☑ QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the quarterly period ended September 30, 2012

OR

# □ TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the transition period from \_\_\_\_\_to\_

Commission File No. 000-50028

# WYNN RESORTS, LIMITED

(Exact name of registrant as specified in its charter)

NEVADA (State or other jurisdiction of

incorporation or organization)

46-0484987 (I.R.S. Employer Identification No.)

3131 Las Vegas Boulevard South—Las Vegas, Nevada 89109 (Address of principal executive offices) (Zip Code)

(702) 770-7555

(Registrant's telephone number, including area code)

N/A

(Former name, former address and former fiscal year, if changed since last report)

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days: Yes No

Indicate by check mark whether the registrant has submitted electronically and posted on its corporate Website, if any, every Interactive Data File required to be submitted and posted pursuant to Rule 405 of Regulation S-T (§232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit and post such files). Yes 🛛 No 🗆

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, or a smaller reporting company. See the definitions of "large accelerated filer," "accelerated filer," and "smaller reporting company" in Rule 12b-2 of the Exchange Act.

Large accelerated filer 🗵 Accelerated filer 🗆 Non-accelerated filer 🗆 Smaller reporting company 🗆

Indicate by check mark whether the registrant is a shell company (as defined in Rule12b-2 of the Exchange Act). Yes 🗌 No 🗵

Indicate the number of shares outstanding of each of the issuer's classes of common stock, as of the latest practicable date.

| Class                          | Outstanding at October 31, 2012 |
|--------------------------------|---------------------------------|
| Common stock, \$0.01 par value | 100,581,636                     |

# WYNN RESORTS, LIMITED AND SUBSIDIARIES INDEX

| Part I.          | Financial Information   |    |
|------------------|---|----|
| Item 1.          | Financial Statements  |    |
|                  | Condensed Consolidated Balance Sheets (unaudited) – September 30, 2012 and December 31, 2011                                    | 3  |
|                  | Condensed Consolidated Statements of Income (unaudited) – Three and nine months ended September 30, 2012 and 2011               | 4  |
|                  | Condensed Consolidated Statements of Comprehensive Income (unaudited) – Three and nine months ended September 30, 2012 and 2011 | 5  |
|                  | Condensed Consolidated Statements of Cash Flows (unaudited) – Nine months ended September 30, 2012 and 2011                     | 6  |
|                  | Condensed Consolidated Statement of Stockholders' Equity (unaudited) – Nine months ended September 30, 2012                     | 7  |
|                  | Notes to Condensed Consolidated Financial Statements (unaudited)  | 8  |
| Item 2.          | Management's Discussion and Analysis of Financial Condition and Results of Operations   | 29 |
| Item 3.          | Quantitative and Qualitative Disclosures About Market Risk  | 47 |
| Item 4.          | Controls and Procedures   | 49 |
| Part II.         | Other Information   |    |
| Item 1.          | Legal Proceedings   | 49 |
| Item 1A.         | Risk Factors  | 49 |
| Item 2.          | Unregistered Sales of Equity Securities and Use of Proceeds   | 50 |
| Item 6.          | Exhibits  | 51 |
| <u>Signature</u> |   | 52 |

# Part I – FINANCIAL INFORMATION Item 1. Financial Statements

# WYNN RESORTS, LIMITED AND SUBSIDIARIES

# CONDENSED CONSOLIDATED BALANCE SHEETS (amounts in thousands, except share data)

(unaudited)

|  | September 30,<br>2012 | December 31,<br>2011 |
|--|-----------------------|----------------------|
| ASSETS   |                       |                      |
| Current assets:  |                       |                      |
| Cash and cash equivalents  | \$ 2,330,937          | \$ 1,262,587         |
| Investment securities  | 198,897               | 122,066              |
| Receivables, net   | 213,322               | 238,490              |
| Inventories  | 65,531                | 72,061               |
| Prepaid expenses and other   | 32,597                | 31,248               |
| Total current assets   | 2,841,284             | 1,726,452            |
| Property and equipment, net  | 4,697,692             | 4,865,332            |
| Restricted cash and investment securities  | 193,206               | 91,501               |
| Intangibles, net   | 32,410                | 35,751               |
| Deferred financing costs, net  | 73,658                | 50,372               |
| Deposits and other assets  | 113,883               | 125,712              |
| Investment in unconsolidated affiliates  | 4,094                 | 4,376                |
| Total assets   | \$ 7,956,227          | \$ 6,899,496         |
| LIABILITIES AND STOCKHOLDERS' EQUITY   | <u> </u>              |                      |
| Current liabilities:   |                       |                      |
| Accounts and construction payables   | \$ 186.645            | \$ 171,608           |
| Current portion of long-term debt  | 1,400                 | 407,934              |
| Current portion of land concession obligation  | 27,248                | 13,425               |
| Customer deposits  | 595,243               | 576.011              |
| Gaming taxes payable   | 169,282               | 177,504              |
| Accuracy payable<br>Accuracy compensation and benefits   | 96,655                | 78,717               |
| Accured interest   | 71,783                | 49,989               |
| Other accrued liabilities  | 49,939                | 94,642               |
| Construction retention   | 2,346                 | 4,471                |
| Deferred income taxes, net   | 3,536                 | 3,575                |
| Income taxes payable   | 1,480                 | 2.017                |
| Total current liabilities  | 1,205,557             | 1,579,893            |
|  | 5,781,471             | 2,809,785            |
| Long-term debt<br>Land concession obligation   | 90,303                | 2,009,785            |
| Other long-term liabilities  | 122,878               | 105,054              |
| Deferred income taxes, net   | 36,718                | 54,294               |
|  |                       |                      |
| Total liabilities  | 7,236,927             | 4,676,042            |
| Commitments and contingencies (Note 15)<br>Stockholders' equity:   |                       |                      |
| Preferred stock, par value \$0.01; 40,000,000 shares authorized; zero shares issued and outstanding  | _                     | —                    |
| Common stock, par value \$0.01; 400,000,000 shares authorized; 113,400,866 and 137,937,088 shares issued; 100,537,136 and 125,080,998 shares outstanding | 1.134                 | 1,379                |
| Treasury stock, at cost; 12,863,730 and 12,856,090 shares  | (1,127,947)           | (1,127,036)          |
| Additional paid-in capital   | 1,257,958             | 3,177,471            |
| Accumulated other comprehensive income   | 3,625                 | 840                  |
| Retained earnings  | 276,565               | 36,368               |
| Total Wynn Resorts, Limited stockholders' equity   | 411,335               | 2,089,022            |
| Noncontrolling interest  | 307,965               | 134,432              |
| Total equity   | 719,300               | 2,223,454            |
|  |                       |                      |
| Total liabilities and stockholders' equity   | \$ 7,956,227          | \$ 6,899,496         |

The accompanying notes are an integral part of these condensed consolidated financial statements.

# WYNN RESORTS, LIMITED AND SUBSIDIARIES

# CONDENSED CONSOLIDATED STATEMENTS OF INCOME (amounts in thousands, except per share data)

(unaudited)

|  |                        | nths Ended<br>nber 30, | Nine Months Ended<br>September 30, |                    |  |  |
|--|------------------------|------------------------|------------------------------------|--------------------|--|--|
|  | 2012                   | 2011                   | 2012                               | 2011               |  |  |
| Operating revenues:<br>Casino                            | <u> </u>               | ¢ 1 020 205            | ¢ 2 015 510                        | ¢ 2 100 EE 2       |  |  |
|  | \$1,012,841<br>119.635 | \$1,020,205            | \$3,015,510                        | \$3,108,553        |  |  |
| Rooms  | 119,635                | 120,113<br>142,891     | 362,018<br>452,845                 | 355,492            |  |  |
| Food and beverage<br>Entertainment, retail and other     | 101,087                | 142,891                | 452,845<br>308,398                 | 419,542<br>306,900 |  |  |
|  |                        |                        |                                    |                    |  |  |
| Gross revenues   | 1,390,131              | 1,388,739              | 4,138,771                          | 4,190,487          |  |  |
| Less: promotional allowances                             | (91,636)               | (90,435)               | (273,571)                          | (264,558)          |  |  |
| Net revenues   | 1,298,495              | 1,298,304              | 3,865,200                          | 3,925,929          |  |  |
| Operating costs and expenses:                            |                        |                        |                                    |                    |  |  |
| Casino   | 653,863                | 679,479                | 1,974,207                          | 1,988,339          |  |  |
| Rooms  | 31,944                 | 31,135                 | 95,193                             | 93,594             |  |  |
| Food and beverage  | 80,652                 | 73,250                 | 235,570                            | 214,203            |  |  |
| Entertainment, retail and other                          | 46,881                 | 52,152                 | 144,647                            | 162,591            |  |  |
| General and administrative                               | 115,785                | 107,935                | 321,512                            | 287,508            |  |  |
| Provision for doubtful accounts                          | 5,283                  | 4,324                  | 6,068                              | 18,269             |  |  |
| Depreciation and amortization                            | 94,274                 | 100,522                | 280,142                            | 303,921            |  |  |
| Property charges and other                               | 22,721                 | 9,662                  | 36,547                             | 124,070            |  |  |
| Total operating costs and expenses                       | 1,051,403              | 1,058,459              | 3,093,886                          | 3,192,495          |  |  |
| Operating income   | 247,092                | 239,845                | 771,314                            | 733,434            |  |  |
| Other income (expense):                                  |                        |                        |                                    |                    |  |  |
| Interest income  | 3,759                  | 2,663                  | 7,807                              | 4,639              |  |  |
| Interest expense, net of capitalized interest            | (75,082)               | (57,462)               | (211,017)                          | (173,956)          |  |  |
| Increase in swap fair value                              |                        | 4,118                  | 4,930                              | 11,483             |  |  |
| Loss on extinguishment of debt                           | (19,663)               |                        | (24,491)                           |                    |  |  |
| Equity in income from unconsolidated affiliates          | 190                    | 376                    | 911                                | 1,242              |  |  |
| Other  | 1,249                  | (85)                   | 936                                | 1,616              |  |  |
| Other income (expense), net                              | (89,547)               | (50,390)               | (220,924)                          | (154,976)          |  |  |
| Income before income taxes                               | 157,545                | 189,455                | 550,390                            | 578,458            |  |  |
| Benefit (provision) for income taxes                     | 7,626                  | (4,270)                | 12,483                             | (11,607)           |  |  |
| Net income   | 165,171                | 185,185                | 562,873                            | 566,851            |  |  |
| Less: Net income attributable to noncontrolling interest | (53,136)               | (58,122)               | (172,210)                          | (143,953)          |  |  |
| Net income attributable to Wynn Resorts, Limited         | \$ 112,035             | \$ 127,063             | \$ 390,663                         | \$ 422,898         |  |  |
|  | ÷ 112,055              | \$ 127,005             | \$ 550,005                         | <b>4</b> 22,030    |  |  |
| Basic and diluted income per common share:               |                        |                        |                                    |                    |  |  |
| Net income attributable to Wynn Resorts, Limited:        | <b>4</b> 1 1 D         | ¢ 1.00                 | ¢ 0.55                             | ¢ 0.44             |  |  |
| Basic  | \$ 1.12                | \$ 1.02                | \$ 3.75                            | \$ 3.41            |  |  |
| Diluted  | \$ 1.11                | \$ 1.01                | \$ 3.71                            | \$ 3.37            |  |  |
| Weighted average common shares outstanding:              | 00.074                 | 104450                 | 10110:                             | 100.000            |  |  |
| Basic  | 99,871                 | 124,176                | 104,104                            | 123,969            |  |  |
| Diluted  | 100,892                | 125,860                | 105,291                            | 125,675            |  |  |
| Dividends declared per common share:                     | \$ 0.50                | \$ 0.50                | \$ 1.50                            | \$ 1.00            |  |  |

The accompanying notes are an integral part of these condensed consolidated financial statements.

# WYNN RESORTS, LIMITED AND SUBSIDIARIES

# CONDENSED CONSOLIDATED STATEMENTS OF COMPREHENSIVE INCOME

(amounts in thousands)

| (unaudited) |
|-------------|
|-------------|

|   | Three Mor<br>Septem |           | Nine Mon<br>Septem |            |
|---|---------------------|-----------|--------------------|------------|
|   | 2012                | 2011      | 2012               | 2011       |
| Net income  | \$165,171           | \$185,185 | \$ 562,873         | \$ 566,851 |
| Other comprehensive income:   |                     |           |                    |            |
| Foreign currency translation adjustments, net of tax                | 1,263               | (2,433)   | 2,006              | (2,389)    |
| Unrealized gain (loss) on available-for-sale securities, net of tax | 784                 | (2,619)   | 1,709              | (2,619)    |
| Total comprehensive income  | 167,218             | 180,133   | 566,588            | 561,843    |
| Less: Comprehensive income attributable to noncontrolling interest  | (53,654)            | (56,858)  | (173,140)          | (142,701)  |
| Comprehensive income attributable to Wynn Resorts, Limited          | \$113,564           | \$123,275 | \$ 393,448         | \$ 419,142 |

The accompanying notes are an integral part of these condensed consolidated financial statements.

# WYNN RESORTS, LIMITED AND SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (amounts in thousands) (unaudited)

|   |              | nths Ended<br>nber 30, |
|---|--------------|------------------------|
|   | 2012         | 2011                   |
| Cash flows from operating activities:   | ¢ = 00.050   | <b>*</b> = 00.0=1      |
| Net income  | \$ 562,873   | \$ 566,851             |
| Adjustments to reconcile net income to net cash provided by operating activities: |              | 000.001                |
| Depreciation and amortization   | 280,142      | 303,921                |
| Deferred income taxes   | (15,814)     | 10,081                 |
| Stock-based compensation  | 13,704       | 18,318                 |
| Excess tax benefits from stock-based compensation                                 | (1,638)      | (10,331                |
| Amortization and write-offs of deferred financing costs and other                 | 18,834       | 15,016                 |
| Loss on extinguishment of debt  | 13,116       | —                      |
| Provision for doubtful accounts   | 6,068        | 18,269                 |
| Property charges and other  | 35,049       | 97,150                 |
| Equity in income of unconsolidated affiliates, net of distributions               | 282          | 85                     |
| Increase in swap fair value   | (4,930)      | (11,483                |
| Increase (decrease) in cash from changes in:                                      |              |                        |
| Receivables, net  | 16,896       | (21,248                |
| Inventories and prepaid expenses and other  | 5,216        | 10,298                 |
| Accounts payable and accrued expenses   | 59,438       | 152,125                |
| Net cash provided by operating activities   | 989,236      | 1,149,052              |
| Cash flows from investing activities:   |              |                        |
| Capital expenditures, net of construction payables and retention                  | (168,315)    | (85,804                |
| Restricted cash and purchase of corporate debt securities                         | (297,781)    | (281,628               |
| Proceeds from sale or maturity of corporate debt securities                       | 118,168      | 37,712                 |
| Deposits and purchase of other assets   | (3,753)      | (34,848                |
| Proceeds from sale of equipment   | 551          | 310                    |
| Net cash used in investing activities   | (351,130)    | (364,258               |
| Cash flows from financing activities:   |              |                        |
| Proceeds from exercise of stock options   | 1,227        | 21,029                 |
| Excess tax benefits from stock-based compensation                                 | 1,638        | 10,331                 |
| Dividends paid  | (154,059)    | (127,668               |
| Proceeds from issuance of long-term debt  | 1,648,598    |                        |
| Purchase of treasury stock  | (911)        | (6,859                 |
| Principal payments on long-term debt  | (1,022,108)  | (163,910               |
| Interest rate swap settlement   | (2,368)      | _                      |
| Payments of financing costs   | (44,491)     | (58                    |
| Net cash provided by (used in) financing activities                               | 427,526      | (267,135               |
| Effect of exchange rate on cash   | 2,718        | (134                   |
| Cash and cash equivalents:  |              |                        |
| Increase in cash and cash equivalents   | 1,068,350    | 517,525                |
| Balance, beginning of period  | 1,262,587    | 1,258,499              |
| Balance, end of period  | \$ 2,330,937 | \$ 1,776,024           |
| שמומותכ, כווע טו אפווטת   | \$ 2,550,957 | φ <u>1</u> ,770,024    |

The accompanying notes are an integral part of these condensed consolidated financial statements.

# WYNN RESORTS, LIMITED AND SUBSIDIARIES

# CONDENSED CONSOLIDATED STATEMENT OF STOCKHOLDERS' EQUITY

(amounts in thousands, except share data)

| (unaudited) |  |
|-------------|--|
|             |  |

|  | Common s     | stock   |                      |                       |    |                            |            |    |   |     |             |     |                      |
|--|--------------|---------|----------------------|-----------------------|----|----------------------------|------------|----|---|-----|-------------|-----|----------------------|
|  | Shares       | Par     | Treasury             | Additional<br>paid-in | ot | nulated<br>her<br>chensive | Retained   | 5  | Total<br>nn Resorts,<br>Ltd.<br>ockholders' | Non | controlling | sto | Total<br>ockholders' |
|  | outstanding  | value   | stock                | capital               |    | ome                        | earnings   |    | equity                                      |     | interest    |     | equity               |
| Balances, January 1, 2012                            | 125,080,998  | \$1,379 | \$(1,127,036)        | \$ 3,177,471          | \$ | 840                        | \$ 36,368  | \$ | 2,089,022                                   | \$  | 134,432     | \$  | 2,223,454            |
| Stock redemption                                     | (24,549,222) | (245)   | _                    | (1,936,198)           |    | _                          | _          |    | (1,936,443)                                 |     | _           |     | (1,936,443)          |
| Net income   |              |         |                      |                       |    | —                          | 390,663    |    | 390,663                                     |     | 172,210     |     | 562,873              |
| Currency translation adjustment                      | —            | —       | —                    | —                     |    | 1,450                      | —          |    | 1,450                                       |     | 556         |     | 2,006                |
| Net unrealized gain on investments                   | _            | _       | _                    | _                     |    | 1,335                      | _          |    | 1,335                                       |     | 374         |     | 1,709                |
| Exercise of stock options                            | 42,000       | _       | _                    | 1,227                 |    | _                          | _          |    | 1,227                                       |     | _           |     | 1,227                |
| Cancellation of restricted stock                     | (29,000)     | _       | _                    | _                     |    | _                          | _          |    | _   |     | _           |     | _                    |
| Purchase of treasury stock                           | (7,640)      | _       | (911)                | _                     |    | _                          | _          |    | (911)                                       |     | _           |     | (911)                |
| Cash dividends                                       | _            | _       | _                    | 200                   |    | _                          | (150,466)  |    | (150,266)                                   |     | _           |     | (150,266)            |
| Excess tax benefits from stock-based<br>compensation | _            | _       | _                    | 1,801                 |    | _                          | _          |    | 1,801                                       |     | _           |     | 1,801                |
| Stock-based compensation                             | _            | _       | _                    | 13,457                |    | _                          | _          |    | 13,457                                      |     | 393         |     | 13,850               |
| Balances, September 30, 2012                         | 100,537,136  | \$1,134 | <u>\$(1,127,947)</u> | \$ 1,257,958          | \$ | 3,625                      | \$ 276,565 | \$ | 411,335                                     | \$  | 307,965     | \$  | 719,300              |

The accompanying notes are an integral part of these condensed consolidated financial statements.

# WYNN RESORTS, LIMITED AND SUBSIDIARIES NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (Unaudited)

#### 1. Organization and Basis of Presentation

#### Organization

Wynn Resorts, Limited, a Nevada corporation (together with its subsidiaries, "Wynn Resorts" or the "Company"), was formed in June 2002 and completed an initial public offering of its common stock on October 25, 2002.

In June 2002, the Company's indirect subsidiary, Wynn Resorts (Macau), S.A. ("Wynn Macau, S.A."), entered into an agreement with the government of the Macau Special Administrative Region of the People's Republic of China ("Macau"), granting Wynn Macau, S.A. the right to construct and operate one or more casino gaming properties in Macau. Wynn Macau, S.A.'s first casino resort in Macau is hereinafter referred to as "Wynn Macau."

The Company currently owns and operates casino hotel resort properties in Las Vegas, Nevada and Macau. In Las Vegas, Nevada, the Company owns Wynn Las Vegas, which opened on April 28, 2005 and was expanded with the opening of Encore at Wynn Las Vegas on December 22, 2008 ("Wynn Las Vegas" or the "Las Vegas Operations"). In Macau, the Company owns Wynn Macau, which opened on September 6, 2006 and was expanded with the opening of Encore at Wynn Macau on April 21, 2010 ("Wynn Macau" or the "Macau Operations").

In October 2009, Wynn Macau, Limited, an indirect wholly owned subsidiary of the Company and the developer, owner and operator of Wynn Macau, listed its ordinary shares of common stock on The Stock Exchange of Hong Kong Limited. Through an initial public offering, including the over allotment, Wynn Macau, Limited sold 1,437,500,000 shares (27.7%) of its common stock.

#### Basis of Presentation

The accompanying condensed consolidated financial statements include the accounts of the Company and its majority-owned subsidiaries. Investments in the 50%-owned joint ventures operating the Ferrari and Maserati automobile dealership and the Brioni mens' retail clothing store inside Wynn Las Vegas are accounted for under the equity method. All significant intercompany accounts and transactions have been eliminated. Certain amounts in the condensed consolidated financial statements for the previous periods have been reclassified to be consistent with the current period presentation. These reclassifications had no effect on the previously reported net income.

The accompanying condensed consolidated financial statements have been prepared by the Company pursuant to the rules and regulations of the Securities and Exchange Commission ("SEC"). Certain information and footnote disclosures normally included in financial statements prepared in accordance with accounting principles generally accepted in the United States of America have been condensed or omitted pursuant to such rules and regulations, although the Company believes that the disclosures herein are adequate to make the information presented not misleading. In the opinion of management, all adjustments (which include only normal recurring adjustments) necessary for a fair presentation of the results for the interim periods have been made. The results for the three and nine months ended September 30, 2012, are not necessarily indicative of results to be expected for the full fiscal year. These condensed consolidated financial statements should be read in conjunction with the consolidated financial statements and notes thereto in the Company's Annual Report on Form 10-K for the year ended December 31, 2011.

# 2. Summary of Significant Accounting Policies

#### Cash and Cash Equivalents

Cash and cash equivalents are comprised of highly liquid investments with purchase maturities of three months or less. Cash equivalents are carried at cost, which approximates fair value. Cash equivalents of \$1.7 billion and \$545 million at September 30, 2012 and December 31, 2011, respectively, were invested in time deposits, money market accounts and commercial paper. In addition, the Company held bank deposits and cash on hand of approximately \$638 million and \$717.5 million as of September 30, 2012 and December 31, 2011, respectively.

#### Restricted Cash and Investment Securities

Restricted cash consists primarily of certain proceeds of the Company's financing activities that are restricted by the agreements governing the Company's debt instruments for the payment of certain Cotai related construction and development costs. Restricted cash balances totaled approximately \$151.8 million at September 30, 2012, substantially all of which were invested in time deposits. There was no restricted cash at December 31, 2011.

Investment securities consist of short-term and long-term investments in domestic and foreign corporate debt securities and commercial paper. The Company's investment policy limits the amount of exposure to any one issuer with the objective of minimizing the potential risk of principal loss. Management determines the appropriate classification (held-to-maturity/available-for-sale) of its securities at the time of purchase and reevaluates such designation as of each balance sheet date. The Company's current investments are reported at fair value, with unrealized gains and losses, net of tax, reported in other comprehensive income. Adjustments are made for amortization of premiums and accretion of discounts to maturity computed under the effective interest method. Such amortization is included in interest income together with realized gains and losses and the stated interest on such securities.

#### Accounts Receivable and Credit Risk

Financial instruments that potentially subject the Company to concentrations of credit risk consist principally of casino accounts receivable. The Company issues credit in the form of "markers" to approved casino customers following investigations of creditworthiness. At September 30, 2012 and December 31, 2011, approximately 79% and 84%, respectively, of the Company's markers were due from customers residing outside the United States, primarily in Asia. Business or economic conditions or other significant events in these countries could affect the collectibility of such receivables.

Accounts receivable, including casino and hotel receivables, are typically non-interest bearing and are initially recorded at cost. Accounts are written off when management deems them to be uncollectible. Recoveries of accounts previously written off are recorded when received. An allowance for doubtful accounts is maintained to reduce the Company's receivables to their estimated carrying amount, which approximates fair value. The allowance is estimated based on specific review of customer accounts as well as management's experience with collection trends in the casino industry and current economic and business conditions. In June 2012, the Company recorded an adjustment to its reserve estimates for casino accounts receivable based on the results of historical collection patterns and current collection trends. For the nine months ended September 30, 2012, this adjustment benefitted operating income by \$30.9 million and net income attributable to Wynn Resorts, Limited by \$23.3 million (or \$0.22 per share on a fully diluted basis).

#### Inventories

Inventories consist of retail, food and beverage items, which are stated at the lower of cost or market value, and certain operating supplies. Cost is determined by the first-in, first-out, average and specific identification methods.

# Revenue Recognition and Promotional Allowances

Casino revenues are measured by the aggregate net difference between gaming wins and losses, with liabilities recognized for funds deposited by customers before gaming play occurs and for chips in the customers' possession. Cash discounts, other cash incentives related to casino play and commissions rebated through junkets to customers are recorded as a reduction to casino revenue. Hotel, food and beverage, entertainment and other operating revenues are recognized when services are performed. Entertainment, retail and other revenue includes rental income which is recognized on a time proportion basis over the lease term. Contingent rental income is recognized when the right to receive such rental income is established according to the lease agreements. Advance deposits on rooms and advance ticket sales are recorded as customer deposits until services are provided to the customer.

Revenues are recognized net of certain sales incentives which are required to be recorded as a reduction of revenue; consequently, the Company's casino revenues are reduced by discounts and commissions, and points earned in the player's club loyalty program.

The retail value of accommodations, food and beverage, and other services furnished to guests without charge is included in gross revenues and then deducted as promotional allowances. The estimated cost of providing such promotional allowances is primarily included in casino expenses as follows (amounts in thousands):

|                                 |          | Three Months Ended<br>September 30, |           | nths Ended<br>nber 30, |
|---------------------------------|----------|-------------------------------------|-----------|------------------------|
|                                 | 2012     | 2011                                | 2012      | 2011                   |
| Rooms                           | \$13,647 | \$13,080                            | \$ 39,845 | \$ 39,169              |
| Food and beverage               | 26,188   | 26,356                              | 80,271    | 77,551                 |
| Entertainment, retail and other | 4,803    | 4,188                               | 13,498    | 12,509                 |
|                                 | \$44,638 | \$43,624                            | \$133,614 | \$129,229              |

#### Gaming Taxes

The Company is subject to taxes based on gross gaming revenue in the jurisdictions in which it operates, subject to applicable jurisdictional adjustments. These gaming taxes are an assessment on the Company's gaming revenue and are recorded as an expense within the "Casino" line item in the accompanying Condensed Consolidated Statements of Income. These taxes totaled approximately \$448.6 million and \$472.9 million for the three months ended September 30, 2012 and 2011, respectively. These taxes totaled approximately \$1,356.2 million and \$1,383.7 million for the nine months ended September 30, 2012 and 2011, respectively.

## Advertising Costs

The Company expenses advertising costs the first time the advertising takes place and such costs are primarily included in general and administrative expenses. Advertising costs were \$5.5 million for both the three months ended September 30, 2012 and 2011. These costs totaled approximately \$16.8 million and \$12.9 million for the nine months ended September 30, 2012 and 2011, respectively.

#### Fair Value Measurements

The Company measures certain of its financial assets and liabilities, such as cash equivalents, available-for-sale securities and interest rate swaps, at fair value on a recurring basis pursuant to accounting standards for fair value measurements. Fair value is the price that would be received to sell an asset or paid to transfer a liability in an orderly transaction between market participants at the measurement date. These accounting standards establish a three-tier fair value hierarchy, which prioritizes the inputs used in measuring fair value. These tiers include: Level 1, defined as observable inputs such as quoted prices in active markets; Level 2,

defined as inputs other than quoted prices in active markets that are either directly or indirectly observable; and Level 3, defined as unobservable inputs in which little or no market data exists, therefore requiring an entity to develop its own assumptions.

The following table presents assets and liabilities carried at fair value (amounts in thousands):

|   |                            | Fair V  | alue Measurements Using:                   |                                     |  |  |
|---|----------------------------|---|--|-------------------------------------|--|--|
|   | Total<br>Carrying<br>Value | Quoted<br>Market<br>Prices in<br>Active<br>Markets<br>(Level 1) | Other<br>Observable<br>Inputs<br>(Level 2) | Unobservable<br>Inputs<br>(Level 3) |  |  |
| As of September 30, 2012                          |                            |   |  |                                     |  |  |
| Cash equivalents                                  | \$1,692,901                | \$908,046   | \$784,855                                  | \$ —                                |  |  |
| Restricted cash and available-for-sale securities | \$ 392,103                 | \$151,776   | \$240,327                                  | \$ —                                |  |  |
| As of December 31, 2011                           |                            |   |  |                                     |  |  |
| Cash equivalents                                  | \$ 545,045                 | \$363,104   | \$181,941                                  | \$ —                                |  |  |
| Interest rate swaps                               | \$ 7,298                   | \$ —  | \$ 7,298                                   | \$ —                                |  |  |
| Available-for-sale securities                     | \$ 213,567                 | \$ —  | \$213,567                                  | \$ —                                |  |  |

#### Recently Issued Accounting Standards

In July 2012, the Financial Accounting Standards Board ("FASB") issued an accounting standards update that is intended to simplify the guidance for testing the decline in the realizable value (impairment) of indefinite-lived intangible assets other than goodwill. The update allows for the consideration of qualitative factors in determining whether it is necessary to perform quantitative impairment tests. The effective date for this update is for the years and interim impairment tests performed for years beginning after September 15, 2012. This update is not expected to have a material impact on the Company's financial statements.

#### 3. Earnings Per Share

Basic earnings per share ("EPS") is computed by dividing net income attributable to Wynn Resorts by the weighted average number of shares outstanding during the period. Diluted EPS reflects the addition of potentially dilutive securities, which for the Company include stock options and nonvested stock. The weighted average common shares outstanding decreased for the three and nine months ended September 30, 2012, due to the redemption and cancellation of 24,549,222 common shares on February 18, 2012, from a former stockholder and related party as described in Notes 9 and 15.

The weighted average number of common and common equivalent shares used in the calculation of basic and diluted EPS consisted of the following (amounts in thousands):

|  | Three Mon<br>Septeml |         | Nine Months Ended<br>September 30, |         |  |
|--|----------------------|---------|------------------------------------|---------|--|
|  | 2012                 | 2011    | 2012                               | 2011    |  |
| Weighted average common shares outstanding (used in calculation of basic earnings per share)     | 99,871               | 124,176 | 104,104                            | 123,969 |  |
| Potential dilution from the assumed exercise of stock options and nonvested stock                | 1,021                | 1,684   | 1,187                              | 1,706   |  |
| Weighted average common and common equivalent shares outstanding (used in calculation of diluted |                      |         |                                    |         |  |
| earnings per share)  | 100,892              | 125,860 | 105,291                            | 125,675 |  |
| Anti-dilutive stock options excluded from the calculation of diluted earnings per share          | 724                  | 25      | 724                                | 610     |  |

#### 4. Comprehensive Income

The following table presents the changes by component in Accumulated Other Comprehensive Income of the Company (amounts in thousands):

|   | Foreign<br>currency<br>translation | Unrealized<br>gain/loss on<br>securities | Accumulated<br>other<br>comprehensive<br>income |
|---|------------------------------------|--|---|
| December 31, 2011                         | \$ 2,409                           | \$ (1,569)                               | \$ 840  |
| Current period other comprehensive income | 1,450                              | 1,335                                    | 2,785   |
| September 30, 2012                        | \$ 3,859                           | \$ (234)                                 | \$ 3,625  |

# 5. Supplemental Disclosure of Cash Flow Information

In February 2012, the Company redeemed and cancelled 24,549,222 shares of common stock from a former stockholder and related party with the issuance of a \$1.94 billion promissory note due in 2022. For details of this transaction see Notes 9 and 15.

Interest paid for the nine months ended September 30, 2012 and 2011, totaled approximately \$179.9 million and \$177.2 million, respectively. Capitalized interest was \$1 million for the nine months ended September 30, 2012. No interest was capitalized during the nine months ended September 30, 2011.

For the nine months ended September 30, 2012 and 2011, capital expenditures included an increase of \$6.9 million and \$16.8 million, respectively, in construction payables and retention.

#### 6. Investment Securities

Investment securities consisted of the following (amounts in thousands):

|                                      |                   | Available-for-               |                               |   |
|--------------------------------------|-------------------|------------------------------|-------------------------------|---|
|                                      | Amortized<br>cost | Gross<br>unrealized<br>gains | Gross<br>unrealized<br>losses | Fair value<br>(net<br>carrying<br>amount) |
| September 30, 2012                   |                   |                              |                               |   |
| Domestic and foreign corporate bonds | \$205,997         | \$ 140                       | \$ (496)                      | \$205,641                                 |
| Commercial paper                     | 34,688            | 8                            | (10)                          | 34,686                                    |
|                                      | \$240,685         | \$ 148                       | \$ (506)                      | \$240,327                                 |
| December 31, 2011                    |                   |                              |                               |   |
| Domestic and foreign corporate bonds | \$196,986         | \$ 20                        | \$ (2,070)                    | \$194,936                                 |
| Commercial paper                     | 18,651            | 1                            | (21)                          | 18,631                                    |
|                                      | \$215,637         | \$ 21                        | \$ (2,091)                    | \$213,567                                 |

For investments with unrealized losses as of September 30, 2012 and December 31, 2011, the Company has determined that (i) it does not have the intent to sell any of these investments, and (ii) it is not likely that the Company will be required to sell these investments prior to the recovery of the amortized cost. Accordingly, the Company has determined that no other-than-temporary impairments exist at the reporting date.

The Company obtains pricing information in determining the fair value of its available-for-sale securities from independent pricing vendors. Based on management's inquiries, the pricing vendors use various pricing models consistent with what other market participants would use. The assumptions and inputs used by the pricing

vendors are derived from market observable sources including: reported trades, broker/dealer quotes, issuer spreads, benchmark curves, bids, offers and other market-related data. Each quarter, the Company validates the fair value pricing methodology to determine the fair value consistent with applicable accounting guidance and to confirm that the securities are classified properly in the fair value hierarchy. The Company compares the pricing received from its vendors to independent sources for the same or similar securities and no adjustment to such prices have resulted.

The amortized cost and estimated fair value of these investment securities at September 30, 2012, by contractual maturity, are as follows (amounts in thousands):

|                                      | Amortized  |            |
|--------------------------------------|------------|------------|
|                                      | cost       | Fair value |
| Available-for-sale securities:       |            |            |
| Due in one year or less              | \$ 199,066 | \$ 198,897 |
| Due after one year through two years | 41,619     | 41,430     |
|                                      | \$ 240,685 | \$ 240,327 |

## 7. Receivables, net

Receivables, net consisted of the following (amounts in thousands):

|                                       | September 30,<br>2012 | December 31,<br>2011 |
|---------------------------------------|-----------------------|----------------------|
| Casino                                | \$ 245,774            | \$ 264,034           |
| Hotel                                 | 15,376                | 20,790               |
| Retail leases and other               | 48,482                | 45,520               |
|                                       | 309,632               | 330,344              |
| Less: allowance for doubtful accounts | (96,310)              | (91,854)             |
|                                       | \$ 213,322            | \$ 238,490           |

# 8. Property and Equipment, net

Property and equipment, net consisted of the following (amounts in thousands):

|                                   | September 30,<br>2012 | December 31,<br>2011 |
|-----------------------------------|-----------------------|----------------------|
| Land and improvements             | \$ 732,136            | \$ 730,335           |
| Buildings and improvements        | 3,796,697             | 3,777,612            |
| Airplanes                         | 77,436                | 77,436               |
| Furniture, fixtures and equipment | 1,638,682             | 1,655,655            |
| Leasehold interests in land       | 316,577               | 316,437              |
| Construction in progress          | 101,589               | 28,477               |
|                                   | 6,663,117             | 6,585,952            |
| Less: accumulated depreciation    | (1,965,425)           | (1,720,620)          |
|                                   | \$ 4,697,692          | \$ 4,865,332         |
|                                   |                       |                      |

#### 9. Long-Term Debt

Long-term debt consisted of the following (amounts in thousands):

|   | September 30,<br>2012 | December 31,<br>2011 |
|---|-----------------------|----------------------|
| 7 7/8% Wynn Las Vegas First Mortgage Notes, due November 1, 2017, net of original issue discount of \$7,691 at          |                       |                      |
| September 30, 2012 and \$8,578 at December 31, 2011   | \$ 492,309            | \$ 491,422           |
| 7 7/8% Wynn Las Vegas First Mortgage Notes, due May 1, 2020, net of original issue discount of \$1,673 at September 30, |                       |                      |
| 2012 and \$1,789 at December 31, 2011   | 350,337               | 350,221              |
| 7 3/4% Wynn Las Vegas First Mortgage Notes, due August 15, 2020   | 1,320,000             | 1,320,000            |
| 5 3/8% Wynn Las Vegas First Mortgage Notes, due March 15, 2022  | 900,000               | —                    |
| Wynn Las Vegas Revolving Credit Facility, due July 15, 2013; interest at LIBOR plus 3%                                  | —                     | —                    |
| Wynn Las Vegas Revolving Credit Facility, due July 17, 2015; interest at LIBOR plus 3%                                  | —                     | —                    |
| Wynn Las Vegas Term Loan Facility, due August 15, 2013; interest at LIBOR plus 1.875%                                   | —                     | 40,262               |
| Wynn Las Vegas Term Loan Facility, due August 17, 2015; interest at LIBOR plus 3%                                       | _                     | 330,605              |
| Wynn Macau Senior Term Loan Facilities (as amended July 2012), due July 31, 2017 and July 31,2018; interest at LIBOR    |                       |                      |
| or HIBOR plus 1.75%-2.50%, net of original issue discount of \$3,904 at September 30, 2012                              | 749,132               | —                    |
| Wynn Macau Senior Term Loan Facilities (as amended June 2007), due June 27, 2014; interest at LIBOR or HIBOR plus       |                       |                      |
| 1.25%-1.75%   | _                     | 477,251              |
| Wynn Macau Senior Revolving Credit Facilities, (as amended July 2012) due July 31, 2017; interest at LIBOR or HIBOR     |                       |                      |
| plus 1.75%-2.50%  | —                     | —                    |
| Wynn Macau Senior Revolving Credit Facility, due June 27, 2012; interest at LIBOR or HIBOR plus 1.25%                   | —                     | 150,400              |
| Redemption Price Promissory Note with former stockholder and related party, due February 18, 2022; interest at 2%       | 1,936,443             | —                    |
| \$42 million Note Payable, due April 1, 2017; interest at LIBOR plus 1.25%  | 34,650                | 35,350               |
| \$32.5 million Note Payable, due August 10, 2012; interest at LIBOR plus 1.15%  |                       | 22,208               |
|   | 5,782,871             | 3,217,719            |
| Current portion of long-term debt   | (1,400)               | (407,934)            |
|   | \$ 5,781,471          | \$2,809,785          |

#### 5<sup>3</sup>/<sub>8</sub>% Wynn Las Vegas First Mortgage Notes

On March 12, 2012, Wynn Las Vegas, LLC and Wynn Las Vegas Capital Corp. (together the "Issuers") issued, in a private offering, \$900 million aggregate principal amount of 5<sup>3</sup>/<sub>8</sub>% First Mortgage Notes due 2022 (the "2022 Notes") pursuant to an Indenture, dated as of March 12, 2012 (the "2022 Indenture"). A portion of the proceeds were used to repay all amounts outstanding under the Wynn Las Vegas term loan facilities. In October 2012, the Issuers commenced an offer to exchange all of the 2022 Notes for notes registered under the Securities Act of 1933, as amended. The exchange offer closed on November 6, 2012.

The 2022 Notes will mature on March 15, 2022 and bear interest at the rate of  $5 \frac{3}{8}\%$  per annum. The Issuers may redeem all or a portion of the 2022 Notes at any time on or after March 15, 2017, at a premium decreasing ratably to zero, plus accrued and unpaid interest. In addition, prior to March 15, 2015, the Issuers may redeem up to 35% of the aggregate principal amount of the 2022 Notes with the net proceeds of one or more qualified equity

contributions made to the Issuers by their parent, Wynn Resorts, Limited. If the Issuers undergo a change of control, they must offer to repurchase the 2022 Notes at 101% of the principal amount, plus accrued and unpaid interest. If the Issuers sell certain assets or suffer an event of loss, and the Issuers do not use the sale or insurance proceeds for specified purposes, they must offer to repurchase the 2022 Notes at 100% of the principal amount, plus accrued and unpaid interest. The 2022 Notes are also subject to mandatory redemption requirements imposed by gaming laws and regulations of gaming authorities in Nevada.

As described in Note 15 of the Condensed Consolidated Financial Statements, Elaine Wynn has submitted a cross claim against Steve Wynn and Kazuo Okada. The indentures for the Wynn Las Vegas, LLC 2022 Notes and Existing Notes (the "Indentures") provide that if Steve Wynn, together with certain related parties, in the aggregate beneficially owns a lesser percentage of the outstanding common stock of the Company than are beneficially owned by any other person, a change of control will have occurred. If Elaine Wynn prevails in her cross claim, Steve Wynn would not beneficially own or control Elaine Wynn's shares and a change in control may result under the Company's debt documents.

The 2022 Indenture contains covenants limiting the Issuers' and the Issuers' restricted subsidiaries' ability to: pay dividends or distributions or repurchase equity; incur additional debt; make investments; create liens on assets to secure debt; enter into transactions with affiliates; issue stock of, or member's interests in, subsidiaries; enter into sale-leaseback transactions; engage in other businesses; merge or consolidate with another company; transfer and sell assets; issue disqualified stock; create dividend and other payment restrictions affecting subsidiaries; and designate restricted and unrestricted subsidiaries. These covenants are subject to a number of important and significant limitations, qualifications and exceptions.

The 2022 Notes rank pari passu in right of payment with the Issuers' outstanding  $77/_8\%$  First Mortgage Notes due 2017 (the "2017 Notes"), the  $77/_8\%$  First Mortgage Notes due 2020 (" $77/_8\%$  2020 Notes") and the  $73/_4\%$  First Mortgage Notes due 2020 (the " $73/_4\%$  2020 Notes" and, together with the 2017 Notes and the  $77/_8\%$  2020 Notes, the "Existing Notes").

On September 17, 2012, the Wynn Las Vegas Credit Agreement was terminated as discussed below, and in accordance with the respective indentures, the liens on the assets of Wynn Las Vegas and its subsidiaries securing, and the subsidiary guarantees of, the 2022 Notes and the Existing Notes were released. The 2022 Notes and the Existing Notes are unsecured, except by a pledge of the equity interests of Wynn Las Vegas, granted by its parent, Wynn Resorts Holdings, LLC, and are not guaranteed by any of the Wynn Las Vegas subsidiaries.

#### Wynn Las Vegas Revolving Credit Facilities

On March 12, 2012, Wynn Las Vegas entered into an eighth amendment ("Amendment No. 8") to its Amended and Restated Credit Agreement, dated as of August 15, 2006 (as amended, the "Wynn Las Vegas Credit Agreement"). Amendment No. 8 amended the Wynn Las Vegas Credit Agreement to, among other things, permit the issuance of the 2022 Notes. Concurrently with the issuance of the 2022 Notes, Wynn Las Vegas, LLC prepaid all term loans under the Wynn Las Vegas Credit Agreement, terminated all of its revolving credit commitments that were due to expire in 2013, and terminated all but \$100 million of its revolving credit commitments expiring in 2015. In connection with this transaction, the Company expensed deferred financing fees of \$4.8 million, all related to the Wynn Las Vegas term loan and revolving credit facilities.

On September 17, 2012, Wynn Las Vegas terminated the Wynn Las Vegas Credit Agreement. No loans were outstanding under the Wynn Las Vegas Credit Agreement at the time of termination. Prior to such termination, certain letters of credit in which lenders had participated pursuant to the Wynn Las Vegas Credit Agreement were reallocated to a separate, unsecured letter of credit facility provided by Deutsche Bank, A.G. Wynn Las Vegas did not incur any early termination penalties related to the termination.

In connection with the termination, the Company expensed \$2 million of previously deferred financing costs and third party fees related to the Wynn Las Vegas Credit Agreement.



#### Wynn Macau Credit Facilities

During the nine months ended September 30, 2012, the Company repaid \$150.4 million of borrowings under the Wynn Macau Senior Revolving Credit Facility. On June 27, 2012, the Wynn Macau Senior Revolving Credit Facility matured with an outstanding balance of \$0.

On July 31, 2012, Wynn Macau, S.A., amended and restated its credit facilities, dated September 14, 2004 (as so amended and restated, the "Amended Wynn Macau Credit Facilities"), and appointed Bank of China Limited, Macau Branch as intercreditor agent, facilities agent and security agent. The Amended Wynn Macau Credit Facilities and related agreements took effect on July 31, 2012 and expand availability under Wynn Macau S.A.'s senior secured bank facility to US\$2.3 billion equivalent, consisting of a US\$750 million equivalent fully funded senior secured term loan facility and a US\$1.55 billion equivalent senior secured revolving credit facility. Wynn Macau, S.A. also has the ability to upsize the total senior secured facilities by an additional US\$200 million pursuant to the terms and provisions of the Amended Wynn Macau Credit Facilities. Borrowings under the Amended Wynn Macau Credit Facilities, which consist of both Hong Kong Dollar and United States Dollar tranches, were used to refinance Wynn Macau S.A.'s existing indebtedness, and will be used to fund the design, development, construction and pre-opening expenses of Wynn Cotai and for general corporate purposes.

The term loan facility matures in July 2018, and the revolving credit facility matures in July 2017. The principal amount of the term loan is required to be repaid in two equal installments in July 2017 and July 2018. The senior secured facilities bear interest for the first six months after closing at LIBOR or HIBOR plus a margin of 2.50% and thereafter will be subject to LIBOR or HIBOR plus a margin of between 1.75% to 2.50% based on Wynn Macau, S.A.'s leverage ratio.

Borrowings under the Amended Wynn Macau Credit Facilities are guaranteed by Palo Real Estate Company Limited ("Palo"), a subsidiary of Wynn Macau, S.A., and by certain subsidiaries of the Company that own equity interests in Wynn Macau, S.A., and are secured by substantially all of the assets of Wynn Macau, S.A., the equity interests in Wynn Macau, S.A. and substantially all of the assets of Palo.

In connection with amending the Wynn Macau Credit Facilities, the Company expensed \$17.7 million and capitalized \$32.9 million of financing costs.

#### Redemption Price Promissory Note

Based on the Board of Directors' finding of "unsuitability," on February 18, 2012, the Company redeemed and cancelled Aruze USA, Inc.'s 24,549,222 shares of Wynn Resorts' common stock. Following a finding of "unsuitability," Wynn Resorts' articles of incorporation authorize redemption of the shares held by unsuitable persons at a "fair value" redemption price. The Company engaged an independent financial advisor to assist in the fair value calculation and concluded that a discount to the then current trading price was appropriate because of, among other things, restrictions on most of the shares which are subject to the terms of an existing stockholder agreement. Pursuant to the articles of incorporation, the Company issued the Redemption Price Promissory Note to Aruze USA, Inc., a former stockholder and related party, in redemption of the shares. The Redemption Price Promissory Note (the "Redemption Note") has a principal amount of \$1.94 billion, matures on February 18, 2022 and bears interest at the rate of 2% per annum payable annually in arrears on each anniversary of the date of the Redemption Note. The Company may, in its sole and absolute discretion, at any time and from time to time, and without penalty or premium, prepay the whole or any portion of the principal or interest due under the Redemption Note. In no instance shall any payment obligation under the Redemption Note be accelerated except in the sole and absolute discretion of the Company or as specifically mandated by law. The indebtedness evidenced by the Redemption Note is and shall be subordinated in right of payment, to the extent and in the manner provided in the Redemption Note, to the prior payment in full of all existing and future obligations of Wynn Resorts and any of its affiliates in respect of indebtedness for borrowed money of any kind or nature.

The Company has recorded the fair value of the Redemption Note at its estimated present value of approximately \$1.94 billion in accordance with applicable accounting guidance. In determining this fair value,

the Company considered the stated maturity of the Redemption Note, its stated interest rate, and the uncertainty of the related cash flows of the Redemption Note as well as the potential effects of the following: uncertainties surrounding the potential outcome and timing of pending litigation with Aruze USA, Inc. (see Note 15); the outcome of on-going investigations by the Nevada Gaming Control Board; and other potential legal and regulatory actions. In addition, in the furtherance of various future business objectives, the Company considered its ability, at its sole option, to prepay the Redemption Note at any time in accordance with its terms without penalty. Accordingly, the Company reasonably determined that the estimated life of the Redemption Note could be less than the contractual life of the Redemption Note. When considering the appropriate rate of interest to be used to determine fair value for accounting purposes and in light of the uncertainty in the timing of the cash flows, the Company used observable inputs from a range of trading values of financial instruments with lives similar to the estimated life of the Redemption Note. As a result of this analysis, the Company concluded the Redemption Note's stated rate of 2% approximated a market rate.

#### Debt Covenant Compliance

As of September 30, 2012, management believes the Company was in compliance with all debt covenants.

#### Fair Value of Long-Term Debt

The net book value of the Company's outstanding first mortgage notes was approximately \$3.1 billion and \$2.2 billion at September 30, 2012 and December 31, 2011, respectively. The estimated fair value of the Company's outstanding first mortgage notes, based on recent trades (using level 2 inputs), was approximately \$3.3 billion and \$2.4 billion at September 30, 2012 and December 31, 2011, respectively. The net book value of the Company's other debt instruments, excluding the Redemption Note, was approximately \$783.8 million and \$1.1 billion at September 30, 2012 and December 31, 2011, respectively. The estimated fair value of the Company's other debt instruments was approximately \$750.1 million and \$1 billion at September 30, 2012 and December 31, 2011, respectively. The estimated fair value of the Redemption Note was approximately \$1.94 billion at September 30, 2012.

#### 10. Interest Rate Swaps

The Company has entered into floating-for-fixed interest rate swap arrangements in order to manage interest rate risk relating to certain of its debt facilities. These interest rate swap agreements modify the Company's exposure to interest rate risk by converting a portion of the Company's floating-rate debt to a fixed rate. These interest rate swaps essentially fixed the interest rate at the percentages noted below; however, changes in the fair value of the interest rate swaps for each reporting period have been recorded as an increase/decrease in swap fair value in the accompanying Condensed Consolidated Statements of Income, as the interest rate swaps do not qualify for hedge accounting.

The Company measured the fair value of its interest rate swaps on a recurring basis pursuant to accounting standards for fair value measurements. The Company utilized Level 2 inputs as described in Note 2 to determine fair value. The fair value approximates the amount the Company would pay if these contracts were settled at the respective valuation dates. Fair value is estimated based upon current, and predictions of future, interest rate levels along a yield curve, the remaining duration of the instruments and other market conditions, and therefore, is subject to significant estimation and a high degree of variability and fluctuation between periods. The fair value is adjusted to reflect the impact of credit ratings of the counterparties or the Company, as applicable. These adjustments resulted in a reduction in the fair values as compared to their settlement values. As of December 31, 2011, the interest rate swap liabilities of \$7.3 million were included in other current accrued liabilities.

#### Wynn Las Vegas Swap

In June 2012, the Company terminated its Wynn Las Vegas swap for a payment of \$2.4 million. As of December 31, 2011, the liability fair value of this interest rate swap was approximately \$4.6 million.

#### Wynn Macau Swaps

In June 2012, the Wynn Macau swap matured. As of December 31, 2011, the liability fair value of this interest rate swap was approximately \$2.7 million.

On September 28, 2012, the Company entered into two interest rate swap agreements intended to hedge a portion of the underlying interest rate risk on borrowings under the Amended Wynn Macau Credit Facilities. Under the two swap agreements, the Company pays a fixed interest rate of 0.73% (excluding the applicable interest margin) on notional amounts corresponding to borrowings of HK\$3.95 billion (approximately US\$509.3 million) incurred under the Amended Wynn Macau Credit Facilities in exchange for receipts on the same amount at a variable interest rate based on the applicable HIBOR at the time of payment. These interest rate swaps fix the all-in interest rate on such amounts at 2.48% to 3.23%. These interest rate swap agreements mature in July 2017.

#### **11. Related Party Transactions**

#### Related Party Share Redemption

Based on the Board of Directors' finding of "unsuitability," on February 18, 2012, the Company redeemed and cancelled Aruze USA, Inc.'s 24,549,222 shares of Wynn Resorts' common stock. Following a finding of "unsuitability," Wynn Resorts' articles of incorporation authorize redemption of the shares held by unsuitable persons at a "fair value" redemption price. The Company engaged an independent financial advisor to assist in the fair value calculation and concluded that a discount to the then current trading price was appropriate because of, among other things, restrictions on most of the shares which are subject to the terms of an existing stockholder agreement. Pursuant to the articles of incorporation, the Company issued the Redemption Note to Aruze USA, Inc., a former stockholder and related party, in redemption of the shares.

#### Amounts Due to Officers

The Company periodically provides services to Stephen A. Wynn, Chairman of the Board of Directors and Chief Executive Officer ("Mr. Wynn"), and certain other officers and directors of the Company, including household employees, construction work and other personal services. Mr. Wynn and the other officers and directors have deposits with the Company to prepay any such items, which are replenished on an ongoing basis as needed. As of September 30, 2012 and December 31, 2011, Mr. Wynn and the other officers and directors had a net deposit balance with the Company of approximately \$0.7 million and \$0.4 million, respectively.

#### Villa Suite Lease

On March 18, 2010, Mr. Wynn and Wynn Las Vegas entered into an Amended and Restated Agreement of Lease (the "SW Lease") for a villa suite to serve as Mr. Wynn's personal residence. The SW Lease amends and restates a prior lease. The SW Lease was approved by the Audit Committee of the Board of Directors of the Company. The term of the SW Lease commenced as of March 1, 2010 and runs concurrent with Mr. Wynn's employment agreement with the Company; provided that either party may terminate on 90 days notice. Pursuant to the SW Lease, the rental value of the villa suite is treated as imputed income to Mr. Wynn, and is equal to the fair market value of the accommodations provided. Effective March 1, 2010, and for the first two years of the term of the SW Lease, the rental value was \$503,831 per year. Effective March 1, 2012, the rental value is \$440,000 per year based on the current fair market value as established by the Audit Committee of the Company in reliance upon the opinion of an independent third-party appraisal. The rental value for the villa suite will be redetermined every two years during the term of the lease by the Audit Committee, with the assistance of an independent third-party appraisal. Certain services for, and maintenance of, the villa suite are included in the rental, as well as the use of minimal warehouse space at Wynn Las Vegas.

#### The "Wynn" Surname Rights Agreement

On August 6, 2004, the Company entered into agreements with Mr. Wynn that confirm and clarify the Company's rights to use the "Wynn" name and Mr. Wynn's persona in connection with its casino resorts. Under

the parties' Surname Rights Agreement, Mr. Wynn granted the Company an exclusive, fully paid-up, perpetual, worldwide license to use, and to own and register trademarks and service marks incorporating the "Wynn" name for casino resorts and related businesses, together with the right to sublicense the name and marks to its affiliates. Under the parties' Rights of Publicity License, Mr. Wynn granted the Company the exclusive, royalty-free, worldwide right to use his full name, persona and related rights of publicity for casino resorts and related businesses, together with the ability to sublicense the persona and publicity rights to its affiliates, until October 24, 2017.

#### 12. Property Charges and Other

Property charges and other consisted of the following (amounts in thousands):

|   |          | Three Months Ended<br>September 30, |          |           |  |  |
|---|----------|-------------------------------------|----------|-----------|--|--|
|   | 2012     | 2011                                | 2012     | 2011      |  |  |
| Donation to University of Macau                           | \$ 965   | \$ 1,033                            | \$ 3,105 | \$108,516 |  |  |
| Loss on show cancellation                                 | 6,056    | —                                   | 6,056    | 1,378     |  |  |
| Net loss on assets abandoned, retired for remodel or sold | 15,700   | 8,629                               | 27,386   | 14,176    |  |  |
|   | \$22,721 | \$ 9,662                            | \$36,547 | \$124,070 |  |  |

Property charges generally include costs related to the retirement of assets for remodels and asset abandonments. Property charges and other for the three and nine months ended September 30, 2012, included a remodel of a Las Vegas restaurant, charges associated with the termination of a Las Vegas show which will end its run in November 2012 and miscellaneous renovations and abandonments at our resorts. Property charges and other for the three and nine months ended September 30, 2011 included the present value of a charitable contribution made by Wynn Macau to the University of Macau Development Foundation. This contribution consists of a \$25 million payment made in May 2011, and a commitment for additional donations of \$10 million each year for the calendar years 2012 through 2022 inclusive, for a total of \$135 million. The amount reflected in the accompanying Condensed Consolidated Statements of Income has been discounted using the Company's then estimated borrowing rate over the time period of the remaining committed payments. In accordance with accounting standards for contributions, subsequent accretion of the discount is being recorded as additional donation expense and included in Property charges and other. Property charges and other for the nine months ended September 30, 2011 also include the write off of certain costs related to a show that ended its run in Las Vegas and miscellaneous renovations and abandonments at our resorts.

#### 13. Noncontrolling Interest

In October 2009, Wynn Macau, Limited, an indirect wholly owned subsidiary of the Company and the developer, owner and operator of Wynn Macau, listed its ordinary shares of common stock on The Stock Exchange of Hong Kong Limited. Through an initial public offering, including the over allotment, Wynn Macau, Limited sold 1,437,500,000 shares (27.7%) of its common stock. The shares of Wynn Macau, Limited were not and will not be registered under the Securities Act and may not be offered or sold in the United States absent a registration under the Securities Act as amended, or an applicable exception from such registration requirements. Net income attributable to noncontrolling interest was \$53.1 million and \$58.1 million for the three months ended September 30, 2012 and 2011, respectively. Net income attributable to noncontrolling interest was \$172.2 million and \$144 million for the nine months ended September 30, 2012 and 2011, respectively.

#### 14. Stock-Based Compensation

The total compensation cost relating both to stock options and nonvested stock is allocated as follows (amounts in thousands):

|  | Three Months Ended<br>September 30, |          | Nine Months Ended<br>September 30, |          |
|--|-------------------------------------|----------|------------------------------------|----------|
|  | 2012                                | 2011     | 2012                               | 2011     |
| Casino                                     | \$2,621                             | \$ 2,076 | \$ 3,571                           | \$ 6,804 |
| Rooms                                      | 34                                  | 90       | 234                                | 329      |
| Food and beverage                          | 27                                  | 99       | 82                                 | 371      |
| Entertainment, retail and other            |                                     | 3        | 14                                 | 19       |
| General and administrative                 | 3,168                               | 3,374    | 9,803                              | 10,795   |
| Total stock-based compensation expense     | 5,850                               | 5,642    | 13,704                             | 18,318   |
| Total stock-based compensation capitalized | 49                                  | 437      | 146                                | 827      |
|  | \$ 5,899                            | \$ 6,079 | \$13,850                           | \$19,145 |

For the nine months ended September 30, 2012, the Company reversed stock-based compensation expense allocated to casino operations related to stock options and restricted stock granted in 2008 with an approximate 8 year cliff vest provision that were forfeited during the first quarter of 2012.

#### 15. Commitments and Contingencies

#### Wynn Macau

<u>Cotai Development and Land Concession Contract.</u> In September 2011, Palo Real Estate Company Limited and Wynn Resorts (Macau) S.A., each an indirect subsidiary of Wynn Macau, Limited, formally accepted the terms and conditions of a draft land concession contract from the Macau government for approximately 51 acres of land in the Cotai area of Macau. On May 2, 2012, the land concession contract was gazetted by the government of Macau evidencing the final step in the granting of the land concession. The Company is constructing a full scale integrated resort containing a casino, luxury hotel, convention, retail, entertainment and food and beverage offerings on this land. The Company estimates the project budget to be in the range of \$3.5 billion to \$4.0 billion. The Company expects to establish a guaranteed maximum price for the project in the first half of 2013.

The initial term of the land concession contract is 25 years from May 2, 2012, and it may be renewed with government approval for successive periods. The total land premium payable, including interest as required by the land concession contract, is \$193.4 million. An initial payment of \$62.5 million was paid in December 2011, with eight additional semi-annual payments of approximately \$16.4 million each (including interest at 5%) due beginning November 2012. As of September 30, 2012, the Company has recorded this obligation and related asset with \$27.2 million included as a current liability and \$90.3 million included as a long-term liability. The Company will also be required to make annual lease payments of \$0.8 million during the resort construction period and annual payments of approximately \$1.1 million once the development is completed.

<u>Cotai Land Agreement.</u> On May 10, 2012, the Company made a \$50 million payment to an unrelated third party in consideration of that party's relinquishment of certain rights in and to any future development on the Cotai land noted above.

#### Litigation

In addition to the actions noted below, the Company's affiliates are involved in litigation arising in the normal course of business. In the opinion of management, such litigation will not have a material effect on the Company's financial condition, results of operations or cash flows.

#### Atlantic-Pacific Capital

On May 3, 2010, Atlantic-Pacific Capital, Inc. ("APC") filed an arbitration demand with Judicial Arbitration and Mediation Services regarding an agreement with the Company. The action concerns a claim for compensation of approximately \$32 million pursuant to an agreement entered into between APC and the Company on or about March 30, 2008 whereby APC was engaged to raise equity capital for an investment vehicle sponsored by the Company. APC is seeking compensation unrelated to the investment vehicle. The Company has denied APC's claims for compensation. The Company filed a Complaint for Damages and Declaratory Relief against APC in the Eighth Judicial District Court, Clark County, Nevada, on May 10, 2010. APC removed the action to the United States District Court, District of Nevada. In March 2011, the District Court denied APC's motion to compel arbitration, and dismissed the action. APC appealed the matter to the United States Court of Appeals for the Ninth Circuit. On October 15, 2012, the matter was argued and submitted. Management believes that APC's claims against the Company are without merit, and the Company intends to continue to defend this matter vigorously.

#### Determination of Unsuitability and Redemption of Aruze USA, Inc. and Affiliates and Related Matters

On February 18, 2012, Wynn Resorts' Gaming Compliance Committee concluded an investigation after receiving an independent report by Freeh, Sporkin & Sullivan, LLP (the "Freeh Report") detailing a pattern of misconduct by Aruze USA, Inc., at the time a stockholder of Wynn Resorts, Universal Entertainment Corporation, Aruze USA, Inc.'s parent company, and Kazuo Okada, the majority shareholder of Universal Entertainment Corporation, who is also a member of Wynn Resorts' Board of Directors and was at the time a director of Wynn Macau, Limited. The factual record presented in the Freeh Report included evidence that Aruze USA, Inc., Universal Entertainment Corporation and Mr. Okada had provided valuable items to certain foreign gaming officials who were responsible for regulating gaming in a jurisdiction in which entities controlled by Mr. Okada were developing a gaming resort. Mr. Okada has denied the impropriety of such conduct to members of the Board of Directors of Wynn Resorts and Mr. Okada has refused to acknowledge or abide by Wynn Resorts' antibribery policies and has refused to participate in the training all other directors have received concerning these policies.

Based on the Freeh Report, the Board of Directors of Wynn Resorts determined that Aruze USA, Inc., Universal Entertainment Corporation and Mr. Okada are "unsuitable persons" under Article VII of the Company's articles of incorporation. The Board of Directors was unanimous (other than Mr. Okada) in its determination. The Board of Directors also requested that Mr. Okada resign as a director of Wynn Resorts (under Nevada corporation law, a board of directors does not have the power to remove a director) and recommended that Mr. Okada be removed as a member of the board of directors of Wynn Macau, Limited. In addition, on February 18, 2012, Mr. Okada was removed from the board of directors of Wynn Las Vegas Capital Corp., an indirect wholly owned subsidiary of Wynn Resorts.

Based on the Board of Directors' finding of "unsuitability," on February 18, 2012, Wynn Resorts redeemed and cancelled Aruze USA, Inc.'s 24,549,222 shares of Wynn Resorts' common stock. Following a finding of "unsuitability," Article VII of Wynn Resorts' articles of incorporation authorize redemption at "fair value" of the shares held by unsuitable persons. The Company engaged an independent financial advisor to assist in the fair value calculation and concluded that a discount to the then current trading price was appropriate because of, among other things, restrictions on most of the shares held by Aruze USA, Inc. under the terms of the Stockholders Agreement (as defined below). Pursuant to the articles of incorporation, Wynn Resorts issued the Redemption Note to Aruze USA, Inc. in redemption of the shares. The Redemption Note has a principal amount of \$1.94 billion, matures on February 18, 2022 and bears interest at the rate of 2% per annum, payable annually in arrears on each anniversary of the date of the Redemption Note. The Company may, in its sole and absolute discretion, at any time and from time to time, and without penalty or premium, prepay the whole or any portion of the principal or interest due under the Redemption Note. In no instance shall any payment obligation under the Redemption Note is and shall be subordinated in right of

payment, to the extent and in the manner provided in the Redemption Note, to the prior payment in full of all existing and future obligations of Wynn Resorts or any of its affiliates in respect of indebtedness for borrowed money of any kind or nature. After authorizing the redemption of the Aruze USA, Inc. shares, the Board of Directors took certain actions to protect the Company and its operations from any influence of an unsuitable person, including placing limitations on the provision of certain operating information to unsuitable persons, evaluating whether to seek the removal of Mr. Okada from the Company's Board of Directors, and formation of an Executive Committee of the Board to manage the business and affairs of the Company during the period between each annual meeting. The Charter of the Executive Committee provides that "Unsuitable Persons" are not permitted to serve on the Committee. All members of the Board, other than Mr. Okada, were appointed to the Executive Committee on February 18, 2012.

On February 19, 2012, Wynn Resorts filed a complaint in the Eighth Judicial District Court, Clark County, Nevada against Mr. Okada, Aruze USA, Inc. and Universal Entertainment Corporation, companies controlled by Mr. Okada (the "Okada Parties"), alleging breaches of fiduciary duty and related claims. The Company is seeking compensatory and special damages as well as a declaration that it acted lawfully and in full compliance with its articles of incorporation, bylaws and other governing documents. On March 12, 2012, Aruze USA, Inc. and Universal Entertainment Corporation removed the action to the United States District Court for the District of Nevada. On that same date, Aruze USA, Inc. and Universal Entertainment Corporation filed an answer denying the claims and a counterclaim that purports to assert claims against the Company, each of the members of the Company's Board of Directors (other than Mr. Okada) and Wynn Resorts' General Counsel. Among other relief, the counterclaim seeks a declaration that the redemption of Aruze USA, Inc.'s share ownership, damages in an unspecified amount and rescission of the Stockholders Agreement. On March 29, 2012, the Company filed a motion to remand the action to state court and requested an extension to answer the denial of Wynn Resorts' claims and the Okada Parties' counterclaims. The federal district court granted the Company's motion to remand and awarded the Company its related attorneys' fees. This case is now pending in the state court, which has determined that this action will be coordinated with Mr. Okada's inspection action (discussed below). The Okada Parties filed a notice of intent to commence a separate federal securities action for the securities counterclaims previously asserted, but have not done so as of the date of this report.

On June 19, 2012, Elaine Wynn responded to the Okada Parties' Counterclaim and asserted a cross claim against Steve Wynn and Kazuo Okada seeking a declaration that (1) any and all of Elaine Wynn's duties under the January 2010 Stockholders Agreement (the "Stockholders Agreement") by and among Aruze USA, Inc., Steve Wynn, and Elaine Wynn be discharged; (2) the Stockholders Agreement is subject to rescission and is rescinded; (3) the Stockholders Agreement is an unreasonable restraint on alienation in violation of public policy; and/or (4) the restrictions on sale of shares shall be construed as inapplicable to Elaine Wynn. Mr. Wynn filed his answer to Elaine Wynn's cross claim on September 24, 2012. The indentures for the Wynn Las Vegas, LLC 2022 Notes and Existing Notes (the "Indentures") provide that if Steve Wynn, together with certain related parties, in the aggregate beneficially owns a lesser percentage of the outstanding common stock of the Company than are beneficially owned by any other person, a change of control will have occurred. If Elaine Wynn prevails in her cross claim, Steve Wynn would not beneficially own or control Elaine Wynn's shares and a change in control may result under the Company's debt documents. Under the Indentures, the occurrence of a change of control requires that the Company make an offer (unless the notes have been previously called for redemption) to each holder to repurchase all or any part of such holder's Notes at a purchase price equal to 101% of the aggregate principal amount thereof plus accrued and unpaid interest on the Notes purchased, if any, to the date of repurchase.

On February 24, 2012, the board of directors of Wynn Macau, Limited removed Mr. Kazuo Okada from the board.

The Company provided the Freeh Report to appropriate regulators and law enforcement agencies and is cooperating with related investigations that such regulators and agencies have undertaken. The conduct of the

Okada Parties and any resulting regulatory investigations could have adverse consequences to the Company and its subsidiaries. A finding by regulatory authorities that Mr. Okada violated anti-corruption statutes and/or other laws or regulations applicable to persons affiliated with a gaming licensee on Company property and/or otherwise involved the Company in criminal or civil violations could result in actions by regulatory authorities against the Company. Relatedly, as described below, the Salt Lake Regional Office of the U.S. Securities and Exchange Commission ("SEC") has commenced an informal inquiry into, and other regulators could pursue separate investigations into, the Company's compliance with applicable laws arising from the allegations in the matters described above and in response to litigation filed by Mr. Okada suggesting improprieties in connection with the Company's donation to the University of Macau. While the Company believes that it is in full compliance with all applicable laws, any such investigations could result in actions by regulators against the Company.

The Executive Committee continues to monitor such investigations and may decide at some future point to call a special meeting of stockholders for the purpose of removing Mr. Okada, if it considers such action to be desirable and in the best interests of the Company and its stockholders.

On August 28, 2012, Mr. Okada, Universal Entertainment Corporation and Okada Holdings filed a complaint in Tokyo District Court against the Company, all members of the Board (other than Mr. Okada) and the Company's General Counsel, alleging that the press release issued by the Company with respect to the redemption has damaged plaintiffs' social evaluation and credibility. The plaintiffs seek damages and legal fees from the defendants.

On August 31, 2012, the Company received a letter from Aruze USA, Inc. purportedly notifying the Company of its intent to nominate two individuals for election as directors pursuant to Section 2.13 of the Company's Fourth Amended and Restated Bylaws. Section 2.13 provides for nominations by stockholders. As a result of the Board's determination on February 18, 2012, that all of Mr. Okada, Aruze USA, Inc. and Universal are "unsuitable persons" as defined in the Articles of Incorporation of the Company, and the subsequent redemption of all shares previously owned by Aruze USA, Inc., the Company believes that Aruze USA, Inc. is not eligible to make such nominations. On October 19, 2012, Aruze USA, Inc. publicly announced that it is terminating its efforts to elect such nominees.

On August 31, 2012, Aruze USA, Inc. filed a motion for preliminary injunction with the state court in Nevada. The motion sought a preliminary injunction that would prohibit Wynn Resorts from barring or preventing Aruze USA, Inc. from exercising rights as a stockholder and an order that its purported nominees be presented to Wynn Resorts' stockholders and voted on (including by Aruze USA, Inc. as a stockholder) at the 2012 Annual Meeting of Stockholders (scheduled for November 2, 2012). At the conclusion of a hearing held on October 2, 2012, the Nevada state court denied Aruze USA, Inc.'s motion for preliminary injunction. On October 19, 2012, Aruze USA, Inc. filed a notice of appeal with the Nevada Supreme Court. The Company intends to vigorously defend against the appeal and to argue that the Nevada Supreme Court should affirm the state court's decision.

On September 7, 2012, Aruze USA, Inc. and Universal Entertainment Corporation filed a second amended counterclaim in the Nevada state court. Wynn Resorts and the other counter-defendants filed a motion to dismiss that pleading on September 26, 2012. A hearing on the motion is schedule for November 13, 2012.

In addition, on October 10, 2012, Mr. Okada filed a motion to dismiss the complaint that Wynn Resorts filed in the Nevada state court in February 2012. The Company filed an amended complaint on October 29, 2012.

#### Litigation Commenced by Kazuo Okada and Related Matters

On January 11, 2012, Mr. Okada, in his role as a Wynn Resorts' director, commenced a writ proceeding in the Eighth Judicial District Court, Clark County, Nevada, seeking to compel the Company to produce certain books and records relating to a donation to the University of Macau, among other things.

In May 2011, Wynn Macau, a majority owned subsidiary of the Company, made a commitment to the University of Macau Development Foundation in support of the new Asia-Pacific Academy of Economics and

Management. This contribution consists of a \$25 million payment made in May 2011 and a commitment for additional donations of \$10 million each year for the calendar years 2012 through 2022 inclusive. The pledge was consistent with the Company's long-standing practice of providing philanthropic support for deserving institutions in the markets in which it operates. The pledge was made following an extensive analysis which concluded that the gift was made in accordance with all applicable laws. The pledge was considered by the boards of directors of both the Company and Wynn Macau, Limited and approved by 15 of the 16 directors who serve on those boards. The sole dissenting vote was cast by Mr. Okada whose stated objection was to the length of time over which the donation would occur, not its propriety.

On February 8, 2012, following Mr. Okada's lawsuit, the Company received a letter from the Salt Lake Regional Office of the SEC requesting that, in connection with an informal inquiry by the SEC, the Company preserve information relating to the donation to the University of Macau, any donations by the Company to any other educational charitable institutions, including the University of Macau Development Foundation, and the Company's casino or concession gaming licenses or renewals in Macau. The Company is cooperating with the Salt Lake Regional Office staff.

At a hearing on February 9, 2012, the Nevada state court held that, as a director of the Company, Mr. Okada had the right to make a reasonable inspection of the Company's corporate books and records. Following the hearing, the Company released certain documents to Mr. Okada for his inspection. At a subsequent hearing on March 8, 2012, the court considered Mr. Okada's request that the Company's Board of Directors make additional documents available to him, and ruled that Mr. Okada was entitled to inspect two additional pages of documents. The Company promptly complied with the court's ruling.

On May 25, 2012, Mr. Okada amended his petition to request inspection of additional records. The Nevada state court ordered Mr. Okada to file a supplemental brief addressing how his requests relate to his duties as a director of the Company, and the Company was to respond by filing a supplemental brief on the reasonableness of Mr. Okada's requests. After Mr. Okada filed his supplemental brief, the Company moved to depose Mr. Okada prior to having to file its supplemental brief. At a hearing on June 28, 2012, the state court ordered Mr. Okada to appear for a deposition in Las Vegas, Nevada, which took place on September 18, 2012. Following Mr. Okada's deposition, the parties each submitted supplemental briefs. Following a hearing held on October 2, 2012, the court ruled that Mr. Okada is entitled to review certain additional Company documents from the 2000 to 2002 time period. The Company has complied with the court's ruling. On November 2, 2012, Mr. Okada filed a motion to compel the production of additional documents and to depose a Company representative. A hearing on this motion is scheduled for November 8, 2012.

#### Related litigation

Six derivative actions were commenced against the Company and all members of its Board of Directors: four in the United States District Court, District of Nevada, and two in the Eighth Judicial District Court of Clark County, Nevada.

The four federal actions brought by the following plaintiffs have been consolidated: (1) The Louisiana Municipal Police Employees' Retirement System, (2) Maryanne Solak, (3) Excavators Union Local 731 Welfare Fund, and (4) Boilermakers Lodge No. 154 Retirement Fund (collectively, the Federal Plaintiffs").

The Federal Plaintiffs filed a consolidated complaint on August 6, 2012, asserting claims for: (1) breach of fiduciary duty; (2) waste of corporate assets; (3) injunctive relief; and (4) unjust enrichment. The claims are against the Company all Company directors, including Mr. Okada, however, the plaintiffs voluntarily dismissed Mr. Okada as a defendant in this consolidated action on September 27, 2012. The Federal Plaintiffs claim that the individual defendants breached their fiduciary duties and wasted assets by: (a) failing to ensure the Company's officers and directors complied with federal and state laws and the Company's Code of Conduct; (b) voting to allow the Company's subsidiary to make the donation to the University of Macau; and (c) redeeming Aruze USA, Inc.'s stock such that the Company incurs the debt associated with the redemption. The Federal Plaintiffs seek unspecified compensatory damages, restitution in the form of disgorgement, reformation of corporate

governance procedures, an injunction against all future payments related to the donation/pledge, and all fees (attorneys, accountants, and experts) and costs. The directors responded to the consolidated complaint by filing a motion to dismiss on September 14, 2012. On October 15, 2012, the Federal Plaintiffs filed an opposition to the motion to dismiss, and the directors filed reply papers on November 5, 2012.

The two state court actions brought by the following plaintiffs have also been consolidated: (1) IBEW Local 98 Pension Fund and (2) Danny Hinson (collectively, the "State Plaintiffs"). Through a coordination of efforts by all parties, the directors and the Company (a nominal defendant) have been served in all of the actions.

The State Plaintiffs filed a consolidated complaint on July 20, 2012 asserting claims for (1) breach of fiduciary duty; (2) abuse of control; (3) gross mismanagement; and (4) unjust enrichment. The claims are against the Company and all Company directors, including Mr. Okada, as well as the Company's Chief Financial Officer, who signs financial disclosures filed with the SEC. The State Plaintiffs claim that the individual defendants failed to disclose to the Company's stockholders the investigation into, and the dispute with director Okada as well as the alleged potential violations of the FCPA related to, the University of Macau Development Foundation donation. The State Plaintiffs seek unspecified monetary damages (compensatory and punitive), disgorgement, reformation of corporate governance procedures, an order directing the Company to internally investigate the donation, as well as attorneys' fees and costs. The parties have entered into a stipulation providing for a stay of the state derivative action for 90 days, subject to the parties' obligation to monitor the progress of the pending litigation, discussed above, between Wynn Resorts (among others) and Mr. Okada (among others). Per the stipulation, Wynn Resorts and the individual defendants will not be required to respond to the consolidated complaint while the stay remains in effect. The state court entered the stipulation on October 25, 2012.

The consolidated actions are in a preliminary stage and management has determined that based on proceedings to date, it is currently unable to determine the probability of the outcome of this matter or the range of reasonably possible loss, if any.

#### 16. Income Taxes

For the three months ended September 30, 2012 and 2011, the Company recorded a tax benefit of \$7.6 million and tax expense of \$4.3 million, respectively. For the nine months ended September 30, 2012 and 2011, the Company recorded a tax benefit of \$12.5 million and tax expense of \$11.6 million, respectively. The Company's income tax benefit is primarily related to a decrease in deferred tax liabilities reduced by foreign taxes assessable on the dividends of Wynn Macau, S.A. and foreign tax provisions related to international marketing offices. Since June 30, 2010, the Company no longer considers its portion of the tax earnings and profits of Wynn Macau, Limited to be permanently invested. No additional U.S. tax provision has been made with respect to amounts not considered permanently invested as the Company anticipates that U.S. foreign tax credits should be sufficient to eliminate any U.S. tax provision relating to such repatriation. The Company has not provided deferred U.S. income taxes or foreign withholding taxes on temporary differences as these amounts are permanently reinvested. For the nine months ended September 30, 2012 and 2011, the Company recognized income tax benefits related to excess tax deductions associated with stock compensation costs of \$1.8 million and \$10.5 million, respectively.

Wynn Macau, S.A. has received a 5-year exemption from Macau's 12% Complementary Tax on casino gaming profits through December 31, 2015. Accordingly, the Company was exempted from the payment of \$20.3 million and \$22.8 million in such taxes during the three months ended September 30, 2012 and 2011, respectively. For the nine months ended September 30, 2012 and 2011, the Company was exempted from the payment of such taxes totaling \$66.9 million and \$57.3 million, respectively. The Company's non-gaming profits remain subject to the Macau Complementary Tax and casino winnings remain subject to the Macau Special Gaming tax and other levies together totaling 39% in accordance with its concession agreement.

In April 2012, the Company reached an agreement with the Appellate division of the Internal Revenue Service ("IRS") regarding issues raised during the examination of the 2006 through 2009 U.S. income tax

returns. The issues for consideration by the Appellate division were temporary differences related to the treatment of discounts extended to certain Las Vegas casino customers wagering on credit, the deduction of certain costs incurred during the development and construction of Encore at Wynn Las Vegas and the appropriate tax depreciation recovery periods applicable to certain assets. The settlement with the Appellate division does not impact the Company's unrecognized tax benefits. The settlement of the 2006 through 2009 examination issues resulted in a cash tax payment of \$1.3 million and the utilization of \$3.1 million and \$0.9 million in foreign tax credit and general business credit carryforwards, respectively.

In July 2012, the Macau Finance Bureau commenced an examination of the 2008 Macau income tax return of Wynn Macau, S.A. Since the examination is in its initial stages, the Company is unable to determine if it will conclude within the next 12 months. The Company believes that its liability for uncertain tax positions is adequate with respect to the 2008 examination year.

# **17. Segment Information**

The Company monitors its operations and evaluates earnings by reviewing the assets and operations of its Las Vegas Operations and its Macau Operations. The Company's total assets by segment are as follows (amounts in thousands):

|                      | September 30,<br>2012 | December 31,<br>2011 |
|----------------------|-----------------------|----------------------|
| Assets               |                       |                      |
| Las Vegas Operations | \$ 3,637,308          | \$4,035,398          |
| Macau Operations     | 2,899,741             | 2,202,683            |
| Corporate and other  | 1,419,178             | 661,415              |
|                      | \$ 7,956,227          | \$6,899,496          |

The Company's segment information for its results of operations are as follows (amounts in thousands):

|   | Septem      | Three Months Ended<br>September 30,<br>2012 2011 |             | ths Ended<br>ber 30, |
|---|-------------|--|-------------|----------------------|
| Net revenues                                    |             | 2011   | 2012        | 2011                 |
| Las Vegas Operations                            | \$ 388,044  | \$ 346,936                                       | \$1,096,405 | \$1,132,374          |
| Macau Operations                                | 910,451     | 951,368  | 2,768,795   | 2,793,555            |
| Total   | \$1,298,495 | \$1,298,304                                      | \$3,865,200 | \$3,925,929          |
| Adjusted Property EBITDA (1)                    |             |  |             |                      |
| Las Vegas Operations                            | \$ 110,390  | \$ 85,134  | \$ 293,193  | \$ 349,954           |
| Macau Operations                                | 292,161     | 295,960  | 884,144     | 883,139              |
| Total   | 402,551     | 381,094  | 1,177,337   | 1,233,093            |
| Other operating costs and expenses              |             |  |             |                      |
| Depreciation and amortization                   | 94,274      | 100,522  | 280,142     | 303,921              |
| Property charges and other                      | 22,721      | 9,662  | 36,547      | 124,070              |
| Corporate expenses and other                    | 38,274      | 30,689   | 88,423      | 70,426               |
| Equity in income from unconsolidated affiliates | 190         | 376  | 911         | 1,242                |
| Total   | 155,459     | 141,249  | 406,023     | 499,659              |
| Operating income                                | 247,092     | 239,845  | 771,314     | 733,434              |
| Non-operating costs and expenses                |             |  |             |                      |
| Interest income                                 | 3,759       | 2,663  | 7,807       | 4,639                |
| Interest expense, net of capitalized interest   | (75,082)    | (57,462)   | (211,017)   | (173,956)            |
| Increase in swap fair value                     | —           | 4,118  | 4,930       | 11,483               |
| Loss on extinguishment of debt                  | (19,663)    |  | (24,491)    | —                    |
| Equity in income from unconsolidated affiliates | 190         | 376  | 911         | 1,242                |
| Other   | 1,249       | (85)   | 936         | 1,616                |
| Total   | (89,547)    | (50,390)   | (220,924)   | (154,976)            |
| Income before income taxes                      | 157,545     | 189,455  | 550,390     | 578,458              |
| Benefit (provision) for income taxes            | 7,626       | (4,270)  | 12,483      | (11,607)             |
| Net income                                      | \$ 165,171  | \$ 185,185                                       | \$ 562,873  | \$ 566,851           |

(1) "Adjusted Property EBITDA" is earnings before interest, taxes, depreciation, amortization, pre-opening costs, property charges and other, corporate expenses, intercompany golf course and water rights leases, stock-based compensation, and other non-operating income and expenses and includes equity in income from unconsolidated affiliates. Adjusted Property EBITDA is presented exclusively as a supplemental disclosure because management believes that it is widely used to measure the performance, and as a basis for valuation, of gaming companies. Management uses Adjusted Property EBITDA as a measure of the operating performance of its segments and to compare the operating performance of its properties with those of its competitors. The Company also presents Adjusted Property EBITDA because it is used by some investors as a way to measure a company's ability to incur and service debt, make capital expenditures and meet working capital requirements. Gaming companies have historically reported EBITDA as a supplement to financial measures in accordance with U.S. generally accepted accounting principles ("GAAP"). In order to view the operations of their casinos on a more stand-alone basis, gaming companies, including Wynn Resorts, Limited, have historically excluded from their EBITDA calculations pre-opening expenses, property charges, corporate expenses and stock-based compensation, which do not relate to the management of specific casino properties. However, Adjusted Property EBITDA should not be considered as an alternative to operating income as an indicator of the Company's performance, as an alternative to cash flows from operating activities as a measure of liquidity, or as an alternative to any other measure

determined in accordance with GAAP. Unlike net income, Adjusted Property EBITDA does not include depreciation or interest expense and therefore does not reflect current or future capital expenditures or the cost of capital. The Company has significant uses of cash flows, including capital expenditures, interest payments, debt principal repayments, taxes and other non-recurring charges, which are not reflected in Adjusted Property EBITDA. Also, Wynn Resorts' calculation of Adjusted Property EBITDA may be different from the calculation methods used by other companies and, therefore, comparability may be limited.

#### **18. Subsequent Events**

On October 24, 2012 the Company announced a cash dividend of \$8.00 per share, payable on November 20, 2012 to stockholders of record as of November 7, 2012.

#### Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

The following discussion should be read in conjunction with, and is qualified in its entirety by, the condensed consolidated financial statements and the notes thereto included elsewhere in this Quarterly Report on Form 10-Q. Unless the context otherwise requires, all references herein to the "Company," "we," "us" or "our," or similar terms, refer to Wynn Resorts, Limited, a Nevada corporation, and its consolidated subsidiaries.

#### **Forward-Looking Statements**

The Private Securities Litigation Reform Act of 1995 provides a "safe harbor" for forward-looking statements. Certain information included in this Quarterly Report on Form 10-Q contains statements that are forward-looking, including, but not limited to, statements relating to our business strategy and development activities as well as other capital spending, financing sources, the effects of regulation (including gaming and tax regulations), expectations concerning future operations, margins, profitability and competition. Any statements contained in this report that are not statements of historical fact may be deemed to be forward-looking statements. Without limiting the generality of the foregoing, in some cases you can identify forward-looking statements by terminology such as "may," "will," "should," "could," "believe," "expect," "anticipate," "estimate," "intend," "plan," "continue" or the negative of these terms or other comparable terminology. Such forward-looking information involves important risks and uncertainties that could significantly affect anticipated results in the future and, accordingly, such results may differ from those expressed in any forward-looking statements made by us. These risks and uncertainties include, but are not limited to:

- adverse tourism trends reflecting current domestic and international economic conditions;
- volatility and weakness in world-wide credit and financial markets and from governmental intervention in the financial markets;
- general global macroeconomic conditions;
- decreases in levels of travel, leisure and consumer spending;
- results of probity investigations;
- regulatory or enforcement actions;
- pending or future legal proceedings;
- fluctuations in occupancy rates and average daily room rates;
- continued high unemployment;
- conditions precedent to funding under our credit facilities;
- continued compliance with all provisions in our credit agreements;
- competition in the casino/hotel and resort industries and actions taken by our competitors;
- doing business in foreign locations such as Macau (including the risks associated with developing gaming regulatory frameworks);
- · restrictions or conditions on visitation by citizens of mainland China to Macau;
- new development and construction activities of competitors;
- · our dependence on Stephen A. Wynn and existing management;
- our dependence on a limited number of resorts and locations for all of our cash flow;
- leverage and debt service (including sensitivity to fluctuations in interest rates);
- changes in federal or state tax laws or the administration of such laws;
- changes in gaming laws or regulations (including the legalization of gaming in certain jurisdictions);

- approvals under applicable jurisdictional laws and regulations (including gaming laws and regulations);
- cyber security risk including misappropriation of customer information or other breaches of information security;
- the impact that an outbreak of an infectious disease or the impact of a natural disaster may have on the travel and leisure industry; and
- the consequences of military conflicts in the Middle East and any future security alerts and/or terrorist attacks.

Further information on potential factors that could affect our financial condition, results of operations and business are included in this report and our other filings with the SEC. You should not place undue reliance on any forward-looking statements, which are based only on information currently available to us. We undertake no obligation to publicly release any revisions to such forward-looking statements to reflect events or circumstances after the date of this report.

#### Overview

We are a developer, owner and operator of destination casino resorts. We currently own and operate two casino resort complexes. In Las Vegas, Nevada, we own and operate Wynn Las Vegas, a destination casino resort which opened on April 28, 2005. In December 2008, we expanded Wynn Las Vegas with the opening of Encore at Wynn Las Vegas. We refer to the fully integrated Wynn Las Vegas and Encore at Wynn Las Vegas resort as our "Las Vegas Operations." In the Macau Special Administrative Region of the People's Republic of China ("Macau"), we own and operate Wynn Macau, which opened on September 6, 2006. On April 21, 2010, we opened Encore at Wynn Macau, a further expansion of Wynn Macau. We refer to the fully integrated Wynn Macau and Encore at Wynn Macau and Encore at Wynn Macau resort as our "Macau Operations."

#### Our Resorts

The following table sets forth information about our resorts as of October 2012:

|                      | Hotel Rooms &<br>Suites | Approximate Casino<br>Square Footage | Approximate<br>Number of Table<br>Games | Approximate<br>Number of Slots |
|----------------------|-------------------------|--------------------------------------|---|--------------------------------|
| Las Vegas Operations | 4,750                   | 186,000                              | 220                                     | 2,325                          |
| Macau Operations     | 1,008                   | 265,000                              | 480                                     | 950                            |

#### Las Vegas Operations

Wynn Las Vegas I Encore is located at the intersection of the Las Vegas Strip and Sands Avenue, and occupies approximately 215 acres of land fronting the Las Vegas Strip. In addition, we own approximately 18 acres across Sands Avenue, a portion of which is utilized for employee parking, and approximately 5 acres adjacent to the golf course on which an office building is located.

Our Las Vegas resort complex features:

- Approximately 186,000 square feet of casino space, offering 24-hour gaming and a full range of games, including private gaming salons, a sky casino, a poker room, and a race and sports book;
- Two luxury hotel towers with a total of 4,750 spacious hotel rooms, suites and villas;
- 32 food and beverage outlets featuring signature chefs;
- A Ferrari and Maserati automobile dealership;

- Approximately 94,000 square feet of high-end, brand-name retail shopping, including stores and boutiques by Alexander McQueen, Brioni, Cartier, Chanel, Dior, Graff, Hermes, Loro Piana, Louis Vuitton, Manolo Blahnik, Oscar de la Renta, Vertu and others;
- Recreation and leisure facilities, including an 18-hole golf course, swimming pools, private cabanas and two full service spas and salons;
- Two showrooms; and
- Three nightclubs and a beach club.

In response to our evaluation of our Las Vegas Operations and the reactions of our guests, we have made and expect to continue to make enhancements and refinements to this resort complex.

#### Macau Operations

We operate Wynn Macau I Encore under a 20-year casino concession agreement granted by the Macau government in June 2002.

Our Macau resort complex features:

- Approximately 265,000 square feet of casino space, offering 24-hour gaming and a full range of games, including private gaming salons, sky casinos
  and a poker pit;
- Two luxury hotel towers with a total of 1,008 spacious rooms and suites;
- Casual and fine dining in eight restaurants;
- Approximately 54,600 square feet of high-end, brand-name retail shopping, including stores and boutiques by Bvlgari, Cartier, Chanel, Dior, Dunhill, Ermenegildo Zegna, Ferrari, Giorgio Armani, Graff, Gucci, Hermes, Hugo Boss, Jaeger LeCoultre, Louis Vuitton, Miu Miu, Piaget, Prada, Rolex, Tiffany, Tudor, Vacheron Constantin, Van Cleef & Arpels, Versace, Vertu, and others;
- · Recreation and leisure facilities, including two health clubs and spas, a salon, a pool; and
- Lounges and meeting facilities.

In response to our evaluation of our Macau Operations and the reactions of our guests, we have made and expect to continue to make enhancements and refinements to this resort complex.

#### Future Development

Approximately 142 acres of land comprising our Las Vegas Operations is currently improved with a golf course. While we may develop this property in the future, we have no immediate plans to do so.

In September 2011, Palo Real Estate Company Limited and Wynn Resorts (Macau) S.A., each an indirect subsidiary of Wynn Macau, Limited, formally accepted the terms and conditions of a draft land concession contract from the Macau government for approximately 51 acres of land in the Cotai area of Macau. In December 2011, we made a \$62.5 million initial deposit under the draft land concession contract. On May 2, 2012, the land concession contract was gazetted by the government of Macau, evidencing the final step in the granting of the land concession. We are constructing a full scale integrated resort containing a casino, luxury hotel, convention, retail, entertainment and food and beverage offerings on this land. We estimate the project budget to be in the range of \$3.5 billion to \$4.0 billion. We expect to establish a guaranteed maximum price for the project in the first half of 2013.

## **Results of Operations**

The table below presents our net revenues (amounts in thousands):

|                      |             | Three Months Ended<br>September 30, |              |              |  |  |
|----------------------|-------------|-------------------------------------|--------------|--------------|--|--|
|                      | 2012        | 2012 2011                           |              | 2011         |  |  |
| Net revenues         |             |                                     |              |              |  |  |
| Macau Operations     | \$ 910,451  | \$ 951,368                          | \$2,768,795  | \$ 2,793,555 |  |  |
| Las Vegas Operations | 388,044     | 346,936                             | 1,096,405    | 1,132,374    |  |  |
|                      | \$1,298,495 | \$1,298,304                         | \$ 3,865,200 | \$ 3,925,929 |  |  |

Reliance on only two resort complexes (in two geographic regions) for our operating cash flow exposes us to certain risks that competitors, whose operations are more diversified, may be better able to control. In addition to the concentration of operations in two resort complexes, many of our customers are premium gaming customers who wager on credit, thus exposing us to credit risk. High-end gaming also increases the potential for variability in our results.

# **Operating Measures**

Certain key operating statistics specific to the gaming industry are included in our discussion of our operational performance for the periods for which a Condensed Consolidated Statement of Income is presented. There are two methods used to calculate win percentage in the casino industry. In Las Vegas and in the general casino in Macau, customers usually purchase cash chips at the gaming tables. The cash and net markers used to purchase the cash chips are deposited in the gaming table's drop box. This is the base of measurement that we use in the casino at our Las Vegas Operations and in the general casino at our Macau Operations for calculating win percentage.

In our VIP casino in Macau, customers primarily purchase non-negotiable chips, commonly referred to as rolling chips, from the casino cage and there is no deposit into a gaming table drop box from chips purchased from the cage. Non-negotiable chips can only be used to make wagers. Winning wagers are paid in cash chips. The loss of the non-negotiable chips in the VIP casino is recorded as turnover and provides a base for calculating VIP casino win percentage. Because of this difference in chip purchase activity, the measurement base used in the general casino is not the same that is used in the VIP casino. It is customary in Macau to measure VIP casino play using this rolling chip method. We expect our win as a percentage of turnover to be within the range of 2.7% to 3.0%.

The measurement method in Las Vegas and in the general casino in Macau tracks the initial purchase of chips at the table while the measurement method in our VIP casino in Macau tracks the sum of all losing wagers. Accordingly, the base measurement in the VIP casino is much larger than Las Vegas and the general casino in Macau. As a result, the expected win percentage with the same amount of gaming win is smaller in the VIP casino in Macau when compared to Las Vegas and the general casino in Macau.

Even though both use the same measurement method, we experience different table games win percentages in Las Vegas and the general casino in Macau. This difference is primarily due to the difference in the mix of table games and customer playing habits between the two casinos. Each type of table game has its own theoretical win percentage. Our expected table games win percentage in Las Vegas is 21% to 24%. Our expected table games win percentage in the general casino at Wynn Macau, which we have periodically revised based on our experience since the opening of the Encore at Wynn Macau expansion, is 28% to 30%.

#### Below are definitions of the statistics discussed:

Drop is the amount of cash and net markers issued that are deposited in a gaming table's drop box.

- Turnover is the sum of all losing rolling chip wagers within our Wynn Macau VIP program.
- Rolling chips are identifiable chips that are used to track VIP wagering volume (turnover) for purposes of calculating incentives.
- Table games win is the amount of drop or turnover that is retained and recorded as casino revenue.
- Slot win is the amount of handle (representing the total amount wagered) that is retained by us and is recorded as casino revenue.
- Average Daily Rate ("ADR") is calculated by dividing total room revenue including the retail value of promotional allowances (less service charges, if any) by total rooms occupied including complimentary rooms.
- Revenue per Available Room ("REVPAR") is calculated by dividing total room revenue including the retail value of promotional allowances (less service charges, if any) by total rooms available.
- Occupancy is calculated by dividing total occupied rooms including complimentary rooms by total rooms available.

#### Financial results for the three months ended September 30, 2012 compared to the three months ended September 30, 2011.

#### Revenues

Net revenues for the three months ended September 30, 2012, were comprised of \$1,012.8 million in casino revenues (78% of total net revenues) and \$285.7 million of net non-casino revenues (22% of total net revenues). Net revenues for the three months ended September 30, 2011, are comprised of \$1,020.2 million in casino revenues (78.6% of total net revenues) and \$278.1 million of net non-casino revenues (21.4% of total net revenues).

Casino revenues are comprised of the net win from our table games and slot machine operations. Casino revenues for the three months ended September 30, 2012, of \$1,012.8 million represents a \$7.4 million (0.7%) decrease from casino revenues of \$1,020.2 million for the three months ended September 30, 2011. For the three months ended September 30, 2012, our Las Vegas Operations experienced a \$28.7 million (22.6%) increase in casino revenues to \$155.6 million, compared to the prior year quarter due to a an increase in our table games win percentage (before discounts). Our Macau Operations experienced a \$36.1 million (4%) decrease in casino revenues to \$857.2 million for the three months ended September 30, 2012, compared to the prior year quarter due primarily to lower turnover in the VIP casino and lower slot machine handle associated with our premium customers. The table below sets forth key gaming statistics related to our Las Vegas and Macau operations.

|                            | Three Months Ended September 30, |           |                   |                  |            |            |         |
|----------------------------|----------------------------------|-----------|-------------------|------------------|------------|------------|---------|
|                            | 2012                             |           |                   | 2011             |            | Increase/  | Percent |
|                            |                                  |           | amounts in thousa |                  | (Decrease) |            | Change  |
| Las Vegas Operations:      |                                  |           |                   | (univants in the | sunds)     |            |         |
| Drop                       | \$                               | 682,349   | \$                | 603,485          | \$         | 78,864     | 13.1%   |
| Table games win %          |                                  | 21.9%     |                   | 18.3%            |            | 3.6pts     |         |
| Slot machine handle        | \$                               | 723,514   | \$                | 673,825          | \$         | 49,689     | 7.4%    |
| Slot machine win           | \$                               | 46,326    | \$                | 43,186           | \$         | 3,140      | 7.3%    |
| Macau Operations:          |                                  |           |                   |                  |            |            |         |
| VIP Casino                 |                                  |           |                   |                  |            |            |         |
| VIP turnover               | \$2                              | 7,622,665 | \$3               | 1,439,153        | \$         | 3,816,488) | (12.1)% |
| VIP win as a % of turnover |                                  | 3.08%     |                   | 2.95%            |            | 0.13pts    |         |
| General Casino             |                                  |           |                   |                  |            |            |         |
| Drop                       | \$                               | 686,122   | \$                | 704,274          | \$         | (18,152)   | (2.6)%  |
| Table games win %          |                                  | 30.8%     |                   | 27.7%            |            | 3.1pts     |         |
| Slot machine handle        | \$                               | 983,705   | \$                | 1,133,943        | \$         | (150,238)  | (13.2)% |
| Slot machine win           | \$                               | 54,412    | \$                | 64,494           | \$         | (10,082)   | (15.6)% |

For the three months ended September 30, 2012, room revenues were \$119.6 million, a decrease of \$0.5 million (0.4%) compared to prior year quarter room revenue of \$120.1 million. Room revenue at our Las Vegas Operations increased approximately \$1.3 million (1.4%) to \$91 million compared to the prior year quarter. ADR at our Las Vegas Operations has increased compared to the prior year quarter as we maintained our room rates in an effort to attract customers who would take advantage of all aspects of our resort. Room revenue at our Macau Operations decreased \$1.8 million (5.7%) to \$28.6 million compared to the prior year quarter due primarily to a decrease in average room rates.

The table below sets forth key operating measures related to room revenue.

|        | Three Months Ended<br>September 30,               |  |  |
|--------|---|--|--|
| 2012   | 2011  |  |  |
|        |   |  |  |
| \$ 244 | \$ 240  |  |  |
| 307    | 315   |  |  |
|        |   |  |  |
| 85.7%  | 88.3%   |  |  |
| 94.2%  | 93.7%   |  |  |
|        |   |  |  |
| \$ 209 | \$ 212  |  |  |
| 289    | 295   |  |  |
|        | 2012<br>\$ 244<br>307<br>85.7%<br>94.2%<br>\$ 209 |  |  |

Other non-casino revenues for the three months ended September 30, 2012, included food and beverage revenues of \$156.6 million, retail revenues of \$63.2 million, entertainment revenues of \$21.6 million, and other revenues from outlets, including the spa and salon, of \$16.3 million. Other non-casino revenues for the three months ended September 30, 2011, included food and beverage revenues of \$142.9 million, retail revenues of \$67.2 million, entertainment revenues of \$21.9 million, and other revenues from outlets such as the spa and salon, of \$16.4 million. Food and beverage revenues at our Las Vegas Operations increased \$13.3 million, while our Macau Operations increased \$0.4 million, as compared to the prior year quarter. The increase in Las Vegas is due primarily to strong business in our beach club and nightclubs. Retail revenues at our Macau Operations

decreased \$3.9 million due to lower sales in some of our watch stores, while retail revenues at our Las Vegas Operations declined only \$0.1 million despite the reduction in retail square footage from the reconfiguration of the Encore retail area.

#### Departmental, Administrative and Other Expenses

For the three months ended September 30, 2012, departmental expenses included casino expenses of \$653.9 million, room expenses of \$31.9 million, food and beverage expenses of \$80.7 million, and entertainment, retail and other expenses of \$46.9 million. Also included are general and administrative expenses of \$115.8 million and a charge of \$5.3 million for the provision for doubtful accounts receivable. For the three months ended September 30, 2011, departmental expenses included casino expenses of \$679.5 million, room expenses of \$31.1 million, food and beverage expenses of \$73.3 million, and entertainment, retail and other expenses of \$52.2 million. Also included for the three months ended September 30, 2011, are general and administrative expenses of \$107.9 million and \$4.3 million charged as a provision for doubtful accounts receivable. Casino expenses decreased for the three months ended September 30, 2012, from the prior year quarter due primarily to lower gaming taxes and junket commissions commensurate with the decrease in gaming revenues at our Macau Operations. Food and beverage expenses increased over the prior year quarter primarily due to additional nightclub promotional costs in Las Vegas. The decrease in entertainment, retail and other expenses was driven by the conversion of certain owned retail stores to leased outlets in Macau resulting in a lower cost of sales. General and administrative expense increased primarily due to legal and other costs incurred related to the share redemption and litigation with a former stockholder, as well as development costs.

#### Depreciation and amortization

Depreciation and amortization for the three months ended September 30, 2012, was \$94.3 million compared to \$100.5 million for the three months ended September 30, 2011. This decrease is primarily due to assets with a 5-year life at Wynn Macau being fully depreciated as of September 2011.

During the construction of our properties, costs incurred in the construction of the buildings, improvements to land and the purchases of assets for use in operations were capitalized. Once these properties opened, their assets were placed into service and we began recognizing the associated depreciation expense. Depreciation expenses will continue throughout the estimated useful lives of these assets. In addition, we continually evaluate the useful life of our property and equipment, intangibles and other assets and adjust them when warranted.

The maximum useful life of assets at Wynn Macau is the remaining life of the gaming concession or land concession, which currently expire in June 2022 and August 2029, respectively. Consequently, depreciation related to Wynn Macau is charged on an accelerated basis when compared to our Las Vegas Operations.

#### Property charges and other

Property charges and other for the three months ended September 30, 2012, were \$22.7 million compared to \$9.7 million for the three months ended September 30, 2011. For the three months ended September 30, 2012, property charges and other related primarily to a remodel of a Las Vegas restaurant, charges related to the termination of a Las Vegas show which will end its run in November 2012, and miscellaneous renovations and abandonments at our resorts. Property charges and other for the three months ended September 30, 2011 primarily include miscellaneous renovations and abandonments at our resorts.

In response to our evaluation of our resorts and the reactions of our guests, we continue to make enhancements and refinements at our resorts.

#### Other non-operating costs and expenses

Interest income was \$3.8 million for the three months ended September 30, 2012, compared to \$2.7 million for the three months ended September 30, 2011. During 2012 and 2011, our short-term investment strategy has

been to preserve capital while retaining sufficient liquidity. While the majority of our short-term investments were primarily in money market accounts and time deposits with a maturity of three months or less, beginning in April 2011, we have invested in certain corporate bond securities and commercial paper.

Interest expense was \$75.1 million, net of capitalized interest of \$0.5 million, for the three months ended September 30, 2012, compared to \$57.5 million for the three months ended September 30, 2011. No interest was capitalized during the three months ended September 30, 2011. Our interest expense increased compared to the prior year quarter primarily due to the issuance of the \$1.94 billion Redemption Note by Wynn Resorts, the Wynn Las Vegas \$900 million 5<sup>3</sup>/<sub>8</sub>% first mortgage notes in March 2012, and the increase in the Wynn Macau term loan offset by the reduction of \$370.9 million in Wynn Las Vegas term loan borrowings, all as described in Notes to Condensed Consolidated Financial Statements, Note 9 – "Long-Term Debt".

Changes in the fair value of our interest rate swaps are recorded as an increase (decrease) in swap fair value in each period. In June 2012, we terminated the Wynn Las Vegas interest rate swap for a payment of \$2.4 million and the Wynn Macau interest rate swap matured. For the three months ended September 30, 2011, we recorded a gain of \$4.1 million resulting from the increase in the fair value of interest rate swaps between June 30, 2011 and September 30, 2011. For further information on our interest rate swaps, see Item 3 – "Quantitative and Qualitative Disclosures about Market Risk."

As described in Note 9 of our Notes to Condensed Consolidated Financial Statements, we amended our Wynn Macau Credit Facilities in July 2012 and terminated the Wynn Las Vegas Credit Agreement in September 2012. In connection with amending the Wynn Macau Credit Facilities, we expensed \$17.7 million of deferred financing costs and third party fees. In connection with the termination of the Wynn Las Vegas credit agreement, we expensed \$2 million of deferred financing costs and third party fees.

#### Income Taxes

For the three months ended September 30, 2012 and 2011, we recorded a tax benefit of \$7.6 million and tax expense of \$4.3 million, respectively. Our income tax benefit is primarily related to a decrease in our deferred tax liabilities reduced by foreign taxes assessable on the dividends of Wynn Macau, S.A. and foreign tax provisions related to our international marketing offices. Since June 30, 2010, we have no longer considered our portion of the tax earnings and profits of Wynn Macau, Limited to be permanently invested. No additional U.S. tax provision has been made with respect to amounts not considered permanently invested as we anticipate that U.S. foreign tax credits should be sufficient to eliminate any U.S. tax provision relating to repatriation. We have not provided deferred U.S. income taxes or foreign withholding taxes on temporary differences as these amounts are permanently reinvested. For the three months ended September 30, 2012 and 2011, we recognized income tax benefits related to excess tax deductions associated with stock compensation costs of \$0.6 million and \$2.8 million, respectively.

Wynn Macau, S.A. has received an exemption from Macau's 12% Complementary Tax on casino gaming profits through December 31, 2015. Accordingly, we were exempt from the payment of \$20.3 million and \$22.8 million in such taxes during the three months ended September 30, 2012 and 2011, respectively. Our non-gaming profits remain subject to the Macau Complementary Tax and casino winnings remain subject to the Macau Special Gaming tax and other levies together totaling 39% in accordance with our concession agreement.

In April 2012, we reached an agreement with the Appellate division of the IRS regarding issues raised during the examination of our 2006 through 2009 U.S. income tax returns. The issues for consideration by the Appellate division were temporary differences related to the treatment of discounts extended to certain Las Vegas casino customers wagering on credit, the deduction of certain costs incurred during the development and construction of Encore at Wynn Las Vegas and the appropriate tax depreciation recovery periods applicable to certain assets. The settlement with the Appellate division does not impact our unrecognized tax benefits. The settlement of the 2006 through 2009 examination issues resulted in a cash tax payment of \$1.3 million and the utilization of \$3.1 million and \$0.9 million in foreign tax credit and general business credit carryforwards, respectively.

In July 2012, the Macau Finance Bureau commenced an examination of the 2008 Macau income tax return of Wynn Macau, S.A. Since the examination is in its initial stages, we are unable to determine if it will conclude within the next 12 months. We believe that our liability for uncertain tax positions is adequate with respect to the 2008 examination year.

#### Net income attributable to noncontrolling interests

In October 2009, Wynn Macau, Limited, our indirect wholly owned subsidiary and the developer, owner and operator of Wynn Macau, listed its ordinary shares of common stock on The Stock Exchange of Hong Kong Limited. Wynn Macau, Limited sold 1,437,500,000 shares (27.7%) of its common stock through an initial public offering. We recorded net income attributable to noncontrolling interests of \$53.1 million for the three months ended September 30, 2012, compared to \$58.1 million for the three months ended September 30, 2011. This represents the noncontrolling interests' share of net income from Wynn Macau, Limited during each quarter.

## Financial results for the nine months ended September 30, 2012 compared to the nine months ended September 30, 2011.

#### Revenues

Net revenues for the nine months ended September 30, 2012, were comprised of \$3,015.5 million in casino revenues (78% of total net revenues) and \$849.7 million of net non-casino revenues (22% of total net revenues). Net revenues for the nine months ended September 30, 2011, are comprised of \$3,108.6 million in casino revenues (79.2% of total net revenues) and \$817.4 million of net non-casino revenues (20.8% of total net revenues).

Casino revenues are comprised of the net win from our table games and slot machine operations. Casino revenues for the nine months ended September 30, 2012, of \$3,015.5 million represents a \$93.1 million (3%) decrease from casino revenues of \$3,108.6 million for the nine months ended September 30, 2011.

For the nine months ended September 30, 2012, our Las Vegas Operations experienced a \$67.6 million (14.1%) decrease in casino revenues to \$411.8 million, compared to the prior year period due to a decrease in our table games win percentage (before discounts) during the first half of 2012. Our Macau Operations experienced a \$25.5 million (1%) decrease in casino revenues to \$2,603.7 million for the nine months ended September 30, 2012, compared to the prior year period due primarily to lower turnover in the VIP casino and lower slot machine handle associated with our premium customers.

The table below sets forth key gaming statistics related to our Las Vegas and Macau operations.

|                            | Nine Months Ended September 30, |                   |                         |                   |
|----------------------------|---------------------------------|-------------------|-------------------------|-------------------|
|                            | 2012                            | 2011              | Increase/<br>(Decrease) | Percent<br>Change |
|                            |                                 | (amounts in thous |                         |                   |
| Las Vegas Operations:      |                                 |                   |                         |                   |
| Drop                       | \$ 1,912,430                    | \$ 1,772,201      | \$ 140,229              | 7.9%              |
| Table games win %          | 20.1%                           | 25.4%             | (5.3)pts                |                   |
| Slot machine handle        | \$ 2,150,263                    | \$ 2,078,129      | \$ 72,134               | 3.5%              |
| Slot machine win           | \$ 129,812                      | \$ 127,188        | \$ 2,624                | 2.1%              |
| Macau Operations:          |                                 |                   |                         |                   |
| VIP Casino                 |                                 |                   |                         |                   |
| VIP turnover               | \$91,512,158                    | \$93,386,297      | \$(1,874,139)           | (2.0)%            |
| VIP win as a % of turnover | 2.81%                           | 2.85%             | (0.04)pts               |                   |
| General Casino             |                                 |                   |                         |                   |
| Drop                       | \$ 2,065,323                    | \$ 2,077,029      | \$ (11,706)             | (0.6)%            |
| Table games win %          | 30.3%                           | 27.8%             | 2.5pts                  | _                 |
| Slot machine handle        | \$ 3,610,782                    | \$ 4,101,458      | \$ (490,676)            | (12.0)%           |
| Slot machine win           | \$ 191,294                      | \$ 212,465        | \$ (21,171)             | (10.0)%           |

For the nine months ended September 30, 2012, room revenues were \$362 million, an increase of \$6.5 million (1.8%) compared to prior year period room revenue of \$355.5 million. Room revenue at our Las Vegas Operations increased approximately \$5.8 million (2.2%) to \$274.5 million compared to the prior year period. ADR at our Las Vegas Operations has increased as we maintained our room rates in an effort to attract customers who would take advantage of all aspects of our resort. Room revenue at our Macau Operations increased \$0.7 million (0.8%) to \$87.5 million compared to the prior year period due to an increase in occupancy and in the average daily room rate during the first half of 2012.

The table below sets forth key operating measures related to room revenue.

|                    |        | Nine Months Ended<br>September 30, |  |
|--------------------|--------|------------------------------------|--|
|                    | 2012   | 2011                               |  |
| Average Daily Rate |        |                                    |  |
| Las Vegas          | \$ 251 | \$ 240                             |  |
| Macau              | 316    | 312                                |  |
| Occupancy          |        |                                    |  |
| Las Vegas          | 84.2%  | 88.4%                              |  |
| Macau              | 91.8%  | 91.0%                              |  |
| REVPAR             |        |                                    |  |
| Las Vegas          | \$ 211 | \$ 212                             |  |
| Macau              | 290    | 284                                |  |

Other non-casino revenues for the nine months ended September 30, 2012, included food and beverage revenues of \$452.8 million, retail revenues of \$192.8 million, entertainment revenues of \$60.8 million, and other revenues from outlets, including the spa and salon, of \$54.8 million. Other non-casino revenues for the nine months ended September 30, 2011, included food and beverage revenues of \$419.5 million, retail revenues of \$191.5 million, entertainment revenues of \$61.5 million, and other revenues from outlets such as the spa and salon, of \$53.9 million. Food and beverage revenues increased primarily due to the strength of our beach club and nightclub business at our Las Vegas Operations. Retail revenues at our Macau Operations increased \$3 million, while retail revenues at our Las Vegas Operations decreased \$1.7 million. The increase at Wynn Macau

is due to strong same store sales growth combined with new stores from the first half of 2012. Retail revenues at our Las Vegas operations decreased as we reconfigured the Encore retail area and are in the process of rebranding several retail outlets. Entertainment revenues decreased over the prior year period primarily due to a loss of revenues from the Sinatra "Dance with Me" show, which ended its run in Las Vegas on April 23, 2011.

#### Departmental, Administrative and Other Expenses

For the nine months ended September 30, 2012, departmental expenses included casino expenses of \$1,974.2 million, room expenses of \$95.2 million, food and beverage expenses of \$235.6 million, and entertainment, retail and other expenses of \$144.6 million. Also included are general and administrative expenses of \$321.5 million and \$6.1 million charged as a provision for doubtful accounts receivable. For the nine months ended September 30, 2011, departmental expenses included casino expenses of \$1,988.3 million, room expenses of \$93.6 million, food and beverage expenses of \$214.2 million, and entertainment, retail and other expenses of \$162.6 million. Also included for the nine months ended September 30, 2011, are general and administrative expenses of \$287.5 million and \$18.3 million charged as a provision for doubtful accounts receivable. Casino expenses have decreased for the nine months ended September 30, 2012, over the prior year period due primarily to lower gaming taxes and junket commissions commensurate with the decrease in gaming revenues. Food and beverage expenses increased over the prior year period primarily due to additional nightclub promotional costs in Las Vegas. The decrease in entertainment, retail and other expenses was driven by the conversion of certain owned retail stores to leased outlets in Macau resulting in lower cost of sales. General and administrative expense increased primarily due to legal and other costs incurred related to the share redemption and litigation with a former stockholder, higher advertising costs, and pay rate increases. The provision for doubtful accounts decreased during the nine months ended September 30, 2012 as we recorded an adjustment to our reserve estimates for casino accounts receivable based on the results of historical collection patterns and current collection trends.

#### Depreciation and amortization

Depreciation and amortization for the nine months ended September 30, 2012, was \$280.1 million compared to \$303.9 million for the nine months ended September 30, 2011. This decrease is primarily due to assets with a 5-year life at Wynn Macau being fully depreciated as of September 2011.

During the construction of our properties, costs incurred in the construction of the buildings, improvements to land and the purchases of assets for use in operations were capitalized. Once these properties opened, their assets were placed into service and we began recognizing the associated depreciation expense. Depreciation expenses will continue throughout the estimated useful lives of these assets. In addition, we continually evaluate the useful life of our property and equipment, intangibles and other assets and adjust them when warranted.

The maximum useful life of assets at Wynn Macau is the remaining life of the gaming concession or land concession, which currently expire in June 2022 and August 2029, respectively. Consequently, depreciation related to Wynn Macau is charged on an accelerated basis when compared to our Las Vegas Operations.

#### Property charges and other

Property charges and other for the nine months ended September 30, 2012, were \$36.5 million compared to \$124.1 million for the nine months ended September 30, 2011. For the nine months ended September 30, 2012, property charges and other related primarily to a remodel of a Las Vegas restaurant, charges related to the cancellation of a Las Vegas show which will end its run in November 2012, and miscellaneous renovations and abandonments at our resorts. Property charges and other for the nine months ended September 30, 2011 includes a charge of \$108.5 million reflecting the present value of a charitable contribution made by Wynn Macau to the University of Macau Development Foundation. This contribution consists of a \$25 million payment made in May 2011, and a commitment for additional donations of \$10 million each year for the calendar years 2012 through

2022 inclusive, for a total of \$135 million. The amount reflected in our accompanying Condensed Consolidated Statements of Income has been discounted using our then estimated borrowing rate over the time period of the remaining committed payments. Property charges and other for the nine months ended September 30, 2011 also include the write off of certain costs related to a show that ended its run in Las Vegas and miscellaneous renovations and abandonments at our resorts.

In response to our evaluation of our resorts and the reactions of our guests, we continue to make enhancements and refinements at our resorts.

#### Other non-operating costs and expenses

Interest income was \$7.8 million for the nine months ended September 30, 2012, compared to \$4.6 million for the nine months ended September 30, 2011. During 2012 and 2011, our short-term investment strategy has been to preserve capital while retaining sufficient liquidity. While the majority of our short-term investments were primarily in money market accounts and time deposits with a maturity of six months or less, beginning in April 2011, we have invested in certain corporate bond securities and commercial paper which contributed to the increase in interest income.

Interest expense was \$211 million, net of capitalized interest of \$1 million, for the nine months ended September 30, 2012, compared to \$174 million for the nine months ended September 30, 2011. No interest was capitalized during the nine months ended September 30, 2011. Our interest expense increased compared to the prior year period primarily due to the issuance of the \$1.94 billion Redemption Note by Wynn Resorts, and the Wynn Las Vegas \$900 million  $5_{3/8}\%$  first mortgage notes in March 2012, offset by the reduction of \$370.9 million in Wynn Las Vegas term loan borrowings, all as described in Note 9 of our Notes to Condensed Consolidated Financial Statements.

Changes in the fair value of our interest rate swaps and any applicable termination payments are recorded as an increase (decrease) in swap fair value in each period. We recorded a gain of \$4.9 million for the nine months ended September 30, 2012, resulting from the increase in the fair value of the interest rate swaps that existed during the nine months ended September 30, 2012. In June 2012, we terminated the Wynn Las Vegas interest rate swap for a payment of \$2.4 million and the Wynn Macau interest rate swap matured. For the nine months ended September 30, 2011, we recorded a gain of \$11.5 million resulting from the increase in the fair value of interest rate swaps between December 31, 2010 and September 30, 2011. For further information on our interest rate swaps, see Item 3 – "Quantitative and Qualitative Disclosures about Market Risk."

On March 12, 2012, Wynn Las Vegas entered into an eighth amendment to its Amended and Restated Credit Agreement (the "Wynn Las Vegas Credit Agreement"). In connection with this amendment Wynn Las Vegas prepaid all term loans under the Wynn Las Vegas Credit Agreement, terminated all of its revolving credit commitments that were due to expire in 2013, and terminated all but \$100 million of its revolving credit commitments expiring in 2015. In connection with this transaction, we expensed deferred financing fees of \$4.8 million.

As described in Note 9 of our Notes to Condensed Consolidated Financial Statements, we amended our Wynn Macau Credit Facilities in July 2012 and terminated the Wynn Las Vegas Credit Agreement in September 2012. In connection with amending the Wynn Macau Credit Facilities, we expensed \$17.7 million of deferred financing costs and third party fees. In connection with the termination of the Wynn Las Vegas credit agreement, we expensed \$2 million of deferred financing costs and third party fees.

#### Income Taxes

For the nine months ended September 30, 2012 and 2011, we recorded a tax benefit of \$12.5 million and tax expense of \$11.6 million, respectively. Our income tax benefit is primarily related to a decrease in our deferred

tax liabilities reduced by foreign taxes assessable on the dividends of Wynn Macau, S.A. and foreign tax provisions related to our international marketing offices. Since June 30, 2010, we have no longer considered our portion of the tax earnings and profits of Wynn Macau, Limited to be permanently invested. No additional U.S. tax provision has been made with respect to amounts not considered permanently invested as we anticipate that U.S. foreign tax credits should be sufficient to eliminate any U.S. tax provision relating to repatriation. We have not provided deferred U.S. income taxes or foreign withholding taxes on temporary differences as these amounts are permanently reinvested. For the nine months ended September 30, 2012 and 2011, we recognized income tax benefits related to excess tax deductions associated with stock compensation costs of \$1.8 million and \$10.5 million, respectively.

Wynn Macau, S.A. has received an exemption from Macau's 12% Complementary Tax on casino gaming profits through December 31, 2015. Accordingly, we were exempt from the payment of \$66.9 million and \$57.3 million in such taxes during the nine months ended September 30, 2012 and 2011, respectively. Our non-gaming profits remain subject to the Macau Complementary Tax and casino winnings remain subject to the Macau Special Gaming tax and other levies together totaling 39% in accordance with our concession agreement.

In April 2012, we reached an agreement with the Appellate division of the IRS regarding issues raised during the examination of our 2006 through 2009 U.S. income tax returns. The issues for consideration by the Appellate division were temporary differences related to the treatment of discounts extended to certain Las Vegas casino customers wagering on credit, the deduction of certain costs incurred during the development and construction of Encore at Wynn Las Vegas and the appropriate tax depreciation recovery periods applicable to certain assets. The settlement with the Appellate division does not impact our unrecognized tax benefits. The settlement of the 2006 through 2009 examination issues resulted in a cash tax payment of \$1.3 million and the utilization of \$3.1 million and \$0.9 million in foreign tax credit and general business credit carryforwards, respectively.

In July 2012, the Macau Finance Bureau commenced an examination of the 2008 Macau income tax return of Wynn Macau, S.A. Since the examination is in its initial stages, we are unable to determine if it will conclude within the next 12 months. We believe that our liability for uncertain tax positions is adequate with respect to the 2008 examination year.

#### Net income attributable to noncontrolling interests

In October 2009, Wynn Macau, Limited, our indirect wholly owned subsidiary and the developer, owner and operator of Wynn Macau, listed its ordinary shares of common stock on The Stock Exchange of Hong Kong Limited. Wynn Macau, Limited sold 1,437,500,000 shares (27.7%) of its common stock through an initial public offering. We recorded net income attributable to noncontrolling interests of \$172.2 million for the nine months ended September 30, 2011. This represents the noncontrolling interests' share of net income from Wynn Macau, Limited during each period.

#### **Adjusted Property EBITDA**

We use adjusted property EBITDA to manage the operating results of our segments. Adjusted property EBITDA is earnings before interest, taxes, depreciation, amortization, pre-opening costs, property charges and other, corporate expenses, intercompany golf course and water rights leases, stock-based compensation, and other non-operating income and expenses, and includes equity in income from unconsolidated affiliates. Adjusted property EBITDA is presented exclusively as a supplemental disclosure because we believe that it is widely used to measure the performance, and as a basis for valuation, of gaming companies. We use adjusted property EBITDA as a measure of the operating performance of our segments and to compare the operating performance of our properties with those of our competitors. We also present adjusted property EBITDA because it is used by some investors as a way to measure a company's ability to incur and service debt, make capital expenditures and meet working capital requirements. Gaming companies have historically reported EBITDA as a supplement to

financial measures in accordance with U.S. generally accepted accounting principles ("GAAP"). In order to view the operations of their casinos on a more standalone basis, gaming companies, including us, have historically excluded from their EBITDA calculations pre-opening expenses, property charges and corporate expenses that do not relate to the management of specific casino properties. However, adjusted property EBITDA should not be considered as an alternative to operating income as an indicator of our performance, as an alternative to cash flows from operating activities as a measure of liquidity, or as an alternative to any other measure determined in accordance with GAAP. Unlike net income, adjusted property EBITDA does not include depreciation or interest expense and therefore does not reflect current or future capital expenditures or the cost of capital. We have significant uses of cash flows, including capital expenditures, interest payments, debt principal repayments, taxes and other non-recurring charges, which are not reflected in adjusted property EBITDA. Also, our calculation of adjusted property EBITDA may be different from the calculation methods used by other companies and, therefore, comparability may be limited.

The following table summarizes adjusted property EBITDA (amounts in thousands) for our Las Vegas and Macau Operations as reviewed by management and summarized in Notes to Condensed Consolidated Financial Statements – Note 17 "Segment Information." That footnote also presents a reconciliation of adjusted property EBITDA to net income.

|                          |            | Three Months Ended<br>September 30, |              | Nine Months Ended<br>September 30, |  |
|--------------------------|------------|-------------------------------------|--------------|------------------------------------|--|
| Adjusted Property EBITDA | 2012       | 2011                                | 2012         | 2011                               |  |
| Macau Operations         | \$ 292,161 | \$ 295,960                          | \$ 884,144   | \$ 883,139                         |  |
| Las Vegas Operations     | 110,390    | 85,134                              | 293,193      | 349,954                            |  |
|                          | \$ 402,551 | \$ 381,094                          | \$ 1,177,337 | \$ 1,233,093                       |  |

For the three months ended September 30, 2012, our Las Vegas Operations benefitted from stronger operating results primarily in the casino and food and beverage departments, while our Macau Operations were negatively impacted by lower turnover in the VIP casino.

For the nine months ended September 30, 2012, our Las Vegas Operations were negatively impacted by a lower than normal table games win percentage as discussed above. Our Macau Operations were negatively impacted by lower turnover in the VIP casino. Results for the nine months in Las Vegas and Macau were positively impacted by a credit taken to the provision for doubtful accounts as discussed above. Refer to the discussions above regarding the specific details of our results of operations.

#### Liquidity and Capital Resources

#### Cash Flow from Operations

Our operating cash flows primarily consist of our operating income generated by our Las Vegas and Macau operations (excluding depreciation and other non-cash charges), interest paid, and changes in working capital accounts such as receivables, inventories, prepaid expenses, and payables. Our table games play both in Macau and Las Vegas is a mix of cash play and credit play, while our slot machine play is conducted primarily on a cash basis. A significant portion of our table games revenue is attributable to the play of a limited number of premium international customers who wager on credit. The ability to collect these gaming receivables may impact our operating cash flow for the period. Our rooms, food and beverage, and entertainment, retail, and other revenue is conducted primarily on a cash basis or as a trade receivable. Accordingly, operating cash flows will be impacted by changes in operating income and accounts receivables.

Net cash provided by operations for the nine months ended September 30, 2012, was \$989.2 million compared to \$1,149.1 million provided by operations for the nine months ended September 30, 2011. Cash flow from operations decreased due to lower casino department profitability and changes in ordinary working capital accounts such as accrued expenses.

#### Capital Resources

At September 30, 2012, we had approximately \$2.3 billion of cash and cash equivalents and \$240.3 million of available for sale investments in foreign and domestic debt securities with maturities of up to 2 years. Our cash is available for operations, debt service and retirement, development activities, general corporate purposes and enhancements to our resorts. In addition, we have \$151.8 million of restricted cash for Cotai related construction and development costs. Of these amounts, Wynn Macau, Limited and its subsidiaries held \$1,202 million and \$67.1 million in cash and available for sale investments, respectively, of which we own 72.3%. If our portion of this cash was repatriated to the U.S. on September 30, 2012, approximately one-third of this amount would be subject to U.S. tax in the year of repatriation. Wynn Resorts, Limited, which is not a guarantor of the debt of its subsidiaries, held \$1,038.4 million and \$173.2 million of cash and available for sale investments, respectively. Wynn Las Vegas, LLC held cash balances of \$90.5 million.

On September 17, 2012, Wynn Las Vegas terminated its Amended and Restated Credit Agreement. No loans were outstanding at the time of termination. On September 18, 2012, Wynn Las Vegas distributed to Wynn Resorts, Limited, the Wynn Las Vegas golf course land, the related water rights, and \$700 million in cash.

On July 31, 2012, Wynn Macau expanded its availability under the senior secured bank facility to US\$2.3 billion equivalent, consisting of a US\$750 million equivalent fully funded senior secured term loan facility and a US\$1.55 billion equivalent senior secured revolving credit facility. Wynn Macau also has the ability to upsize the total senior secured facilities by an additional US\$200 million pursuant to the terms and provisions of the Amended Wynn Macau Credit Facilities. These borrowings were used to refinance Wynn Macau's existing indebtedness, and will be used to fund the design, development, construction and preopening expenses of Wynn Cotai, and for general corporate purposes.

We believe that cash flow from operations, availability under our bank credit facility and our existing cash balances will be adequate to satisfy our anticipated uses of capital for the remainder of 2012. If any additional financing became necessary, we cannot provide assurance that future borrowings will be available.

Cash and cash equivalents include investments in money market funds, domestic and foreign bank time deposits and commercial paper, all with maturities of less than 90 days.

#### **Investing Activities**

Capital expenditures were approximately \$168.3 million for the nine months ended September 30, 2012, which included a one-time payment of \$50 million in consideration of an unrelated third party's relinquishment of certain rights in and to any future development on the Cotai land as well as approximately \$35 million of site preparation costs for our Cotai land and various renovations at our resorts. Capital expenditures for the nine months ended September 30, 2011, were approximately \$85.8 million and related primarily to the room and suite remodel at Wynn Las Vegas.

#### **Financing Activities**

#### Las Vegas Operations

On March 12, 2012, Wynn Las Vegas, LLC and Wynn Las Vegas Capital Corp. (together the "Issuers") issued, in a private offering, \$900 million aggregate principal amount of 5 3/8% First Mortgage Notes due 2022 (the "2022 Notes") pursuant to an Indenture, dated as of March 12, 2012 (the "2022 Indenture). A portion of the proceeds were used to repay all amounts outstanding under the Wynn Las Vegas term loan facilities. In October 2012, the Issuers commenced an offer to exchange all of the 2012 notes for notes registered under the Securities Act of 1933, as amended. The exchange offer closed on November 6, 2012.

The 2022 Notes will mature on March 15, 2022 and bear interest at the rate of 5  $\frac{3}{8}$ % per annum. The Issuers may redeem all or a portion of the 2022 Notes at any time on or after March 15, 2017, at a premium

decreasing ratably to zero, plus accrued and unpaid interest. In addition, prior to March 15, 2015, the Issuers may redeem up to 35% of the aggregate principal amount of the 2022 Notes with the net proceeds of one or more qualified equity contributions made to the Issuers by their parent, Wynn Resorts, Limited. The 2022 Notes are also subject to mandatory redemption requirements imposed by gaming laws and regulations of gaming authorities in Nevada.

The 2022 Indenture contains covenants limiting the Issuers' and the Issuers' restricted subsidiaries' ability to: pay dividends or distributions or repurchase equity; incur additional debt; make investments; create liens on assets to secure debt; enter into transactions with affiliates; issue stock of, or member's interests in, subsidiaries; enter into sale-leaseback transactions; engage in other businesses; merge or consolidate with another company; transfer and sell assets; issue disqualified stock; create dividend and other payment restrictions affecting subsidiaries; and designate restricted and unrestricted subsidiaries. These covenants are subject to a number of important and significant limitations, qualifications and exceptions.

The 2022 Notes rank pari passu in right of payment with the Issuers' outstanding  $77/_8\%$  First Mortgage Notes due 2017 (the "2017 Notes"), the  $77/_8\%$  First Mortgage Notes due 2020 (" $77/_8\%$  2020 Notes") and the  $73/_4\%$  First Mortgage Notes due 2020 (the " $73/_4\%$  2020 Notes" and, together with the 2017 Notes and the  $77/_8\%$  2020 Notes, the "Existing Notes").

On March 12, 2012, Wynn Las Vegas, LLC entered into an eighth amendment ("Amendment No. 8") to its Amended and Restated Credit Agreement (the "Wynn Las Vegas Credit Agreement"). Amendment No. 8 amends the Wynn Las Vegas Credit Agreement to, among other things, permit the issuance of the 2022 Notes. Concurrently with the issuance of the 2022 Notes, Wynn Las Vegas prepaid all term loans under the Wynn Las Vegas Credit Agreement, terminated all of its revolving credit commitments that were due to expire in 2013, and terminated all but \$100 million of its revolving credit commitments expiring in 2015. In connection with this transaction, Wynn Las Vegas expensed deferred financing costs of \$4.8 million.

On September 17, 2012, Wynn Las Vegas terminated the Wynn Las Vegas Credit Agreement. No loans were outstanding under the Wynn Las Vegas Credit Agreement at the time of termination. Prior to such termination, certain letters of credit in which lenders had participated pursuant to the Wynn Las Vegas Credit Agreement were reallocated to a separate, unsecured letter of credit facility provided by Deutsche Bank, A.G. Wynn Las Vegas, LLC did not incur any early termination penalties in connection with the termination.

In connection with the termination, we expensed \$2.0 million of previously deferred financing costs and third party fees related to the Wynn Las Vegas Credit Agreement.

#### Macau Operations

During the nine months ended September 30, 2012, we repaid \$150.4 million of borrowings under the Wynn Macau Revolver.

On July 31, 2012, Wynn Macau, amended and restated its credit facilities, dated September 14, 2004 (as so amended and restated, the "Amended Wynn Macau Credit Facilities"), and appointed Bank of China Limited, Macau Branch as intercreditor agent, facilities agent and security agent. The Amended Wynn Macau Credit Facilities took effect on July 31, 2012 and expand availability under Wynn Macau's senior secured bank facility to US\$2.3 billion equivalent, consisting of a US\$750 million equivalent fully funded senior secured term loan facility and a US\$1.55 billion equivalent senior secured revolving credit facility. Wynn Macau also has the ability to upsize the total senior secured facilities by an additional US\$200 million pursuant to the terms and provisions of the Amended Wynn Macau Credit Facilities. Borrowings under the Amended Wynn Macau Credit Facilities, which consist of both Hong Kong Dollar and United States Dollar tranches, were used to refinance Wynn Macau's existing indebtedness, and will be used to fund the design, development, construction and pre-opening expenses of Wynn Cotai, and for general corporate purposes.

The term loan facility matures in July 2018, and the revolving credit facility matures in July 2017. There are no amounts outstanding under the revolving credit facility. The principal amount of the term loan is required to be repaid in two equal installments in July 2017 and July 2018. The senior secured facilities will bear interest for the first six months after closing at LIBOR or HIBOR plus a margin of 2.50% and thereafter will be subject to LIBOR or HIBOR plus a margin of between 1.75% to 2.50% based on Wynn Macau's leverage ratio.

Borrowings under the Amended Wynn Macau Credit Facilities are guaranteed by Palo Real Estate Company Limited ("Palo"), a subsidiary of Wynn Macau, S.A., and by certain subsidiaries of the Company that own equity interests in Wynn Macau, S.A., and are secured by substantially all of the assets of Wynn Macau, S.A., the equity interests in Wynn Macau, S.A. and substantially all of the assets of Palo.

In connection with amending the Wynn Macau Credit Facilities, we expensed \$17.7 million and capitalized \$32.9 million of financing costs.

#### Wynn Resorts Redemption Price Promissory Note

Based on the Board of Directors' finding of "unsuitability," on February 18, 2012, we redeemed and cancelled Aruze USA, Inc.'s 24,549,222 shares of Wynn Resorts' common stock. Following a finding of "unsuitability," our articles of incorporation authorize redemption at "fair value" of the shares held by unsuitable persons. We engaged an independent financial advisor to assist in the fair value calculation and concluded that a discount to the then current trading price was appropriate because of, among other things, restrictions on most of the shares which are subject to the terms of an existing stockholder agreement. Pursuant to the articles of incorporation, we issued the Redemption Price Promissory Note (the "Redemption Note") to Aruze USA, Inc., a former stockholder and related party, in redemption of the shares. The Redemption Note has a principal amount of approximately \$1.94 billion, matures on February 18, 2022 and bears interest at the rate of 2% per annum, payable annually in arrears on each anniversary of the date of the Redemption Note. We may, in our sole and absolute discretion, at any time and from time to time, and without penalty or premium, prepay the whole or any portion of the principal or interest due under the Redemption Note. In no instance shall any payment obligation under the Redemption Note is and shall be subordinated in right of payment, to the extent and in the manner provided in the Redemption Note, to the prior payment in full of all existing and future obligations of Wynn Resorts and any of its affiliates in respect of indebtedness for borrowed money of any kind or nature.

We recorded the fair value of the Redemption Note at its estimated present value of approximately \$1.94 billion in accordance with applicable accounting guidance. In determining this fair value, we considered the stated maturity of the Redemption Note, its stated interest rate, and the uncertainty of the related cash flows of the Redemption Note as well as the potential effects of the following: uncertainties surrounding the potential outcome and timing of pending litigation with Aruze USA, Inc. (see Note 15 to our Notes to Condensed Consolidated Financial Statements); the outcome of on-going investigations by the Nevada Gaming Control Board; and other potential legal and regulatory actions. In addition, in the furtherance of various future business objectives, we considered our ability, at our sole option, to prepay the Redemption Note at any time in accordance with its terms without penalty. Accordingly, we reasonably determined that the estimated life of the Redemption Note could be less than the contractual life of the Redemption Note. When considering the appropriate rate of interest to be used to determine fair value for accounting purposes and in light of the uncertainty in the timing of the cash flows, we used observable inputs from a range of trading values of financial instruments with lives similar to the estimated life of the Redemption Note. As a result of this analysis, we concluded the Redemption Note's stated rate of 2% approximated a market rate.

#### **Off Balance Sheet Arrangements**

We have not entered into any transactions with special purpose entities nor do we engage in any derivatives except for floating-for-fixed interest rate swaps described under Item 3. Quantitative and Qualitative Disclosures About Market Risk. We do not have any retained or contingent interest in assets transferred to an unconsolidated entity. At September 30, 2012, we had unsecured outstanding letters of credit totaling \$15.8 million.

#### Contractual Obligations and Commitments

In March 2012, Wynn Las Vegas, LLC issued \$900 million in 5 3/<sub>8</sub>% first mortgage notes due 2022, repaid all amounts outstanding under the term loan facilities totaling \$370.9 million, reduced its revolving facilities to \$100 million, and in February 2012, Wynn Resorts issued a \$1.94 billion promissory note payable in 2022, all as described above. Additionally, in June 2012, the Wynn Macau Revolver matured with a zero outstanding balance, the Wynn Macau interest rate swap agreement matured, and the Wynn Las Vegas interest rate swap was terminated, all as described herein. In July 2012, Wynn Macau expanded its credit facilities to US\$2.3 billion equivalent, consisting of a US\$750 million equivalent fully funded senior secured term loan facility and a US\$1.55 billion equivalent senior secured revolving credit facility.

In September 2012, the Wynn Las Vegas Credit agreement was terminated and Wynn Macau entered into two interest rate swap agreements. Other than those transactions, there have been no material changes during the nine months ended September 30, 2012 to our contractual obligations or off balance sheet arrangements as disclosed in our Annual Report on Form 10-K for the year ended December 31, 2011.

## Other Liquidity Matters

Wynn Resorts is a holding company and, as a result, our ability to pay dividends is highly dependent on our ability to obtain funds and our subsidiaries' ability to provide funds to us. Restrictions imposed by our Wynn Las Vegas and Wynn Macau debt instruments significantly restrict our ability to pay dividends. Specifically, Wynn Las Vegas, LLC and certain of its subsidiaries are restricted under the indentures governing the Existing Notes and the 2022 Notes from making certain "restricted payments" as defined in the indentures. These restricted payments include the payment of dividends or distributions to any direct or indirect holders of equity interests of Wynn Las Vegas, LLC. These restricted payments may not be made unless certain financial and non-financial criteria have been satisfied. While the Amended Wynn Macau Credit Facilities contains similar restrictions, Wynn Macau is currently in compliance with all requirements, namely satisfaction of its leverage ratio, which must be met in order to pay dividends and is presently able to pay dividends in accordance with the Amended Wynn Macau Credit Facilities.

Wynn Las Vegas, LLC intends to fund its operations and capital requirements from cash on hand and operating cash flow. We cannot assure you however, that our Las Vegas Operations will generate sufficient cash flow from operations or the availability of additional indebtedness will be sufficient to enable us to service and repay Wynn Las Vegas, LLC's indebtedness and to fund its other liquidity needs. Similarly, we expect that Wynn Macau will fund Wynn Macau, S.A.'s debt service obligations with existing cash, operating cash flow and availability under the Amended Wynn Macau Credit Facilities. However, we cannot assure you that operating cash flows will be sufficient to do so. We may refinance all or a portion of our indebtedness on or before maturity. We cannot assure you that we will be able to refinance any of the indebtedness on acceptable terms or at all. Certain legal matters may also impact our liquidity. As described in our Notes to Condensed Consolidated Financial Statements, Note 15 – "Commitments and Contingencies", Elaine Wynn has submitted a cross claim against Steve Wynn and Kazuo Okada. The indentures for the Wynn Las Vegas, LLC 2022 Notes and Existing Notes (the "Indentures") provide that if Steve Wynn, together with certain related parties, in the aggregate beneficially owns a lesser percentage of the outstanding common stock of Wynn Resorts than are beneficially owned by any other person, a change of control will have occurred. If Elaine Wynn prevails in her cross claim, Steve Wynn would not beneficially own or control Elaine Wynn's shares and a change in control may result under our debt documents.

New business developments or other unforeseen events may occur, resulting in the need to raise additional funds. We continue to explore opportunities to develop additional gaming or related businesses in domestic and international markets. There can be no assurances regarding the business prospects with respect to any other opportunity. Any new development would require us to obtain additional financing. We may decide to conduct any such development through Wynn Resorts or through subsidiaries separate from the Las Vegas or Macau-related entities.

The Company's articles of incorporation provide that, to the extent required by the gaming authority making the determination of unsuitability or to the extent the Board of Directors determines, in its sole discretion, that a person is likely to jeopardize the Company's or any affiliate's application for, receipt of, approval for, right to the use of, or entitlement to, any gaming license, shares of Wynn Resorts' capital stock that are owned or controlled by an unsuitable person or its affiliates are subject to redemption by the Company. The redemption price may be paid in cash, by promissory note or both, as required by the applicable gaming authority and, if not, as we elect. Any promissory note that we issue to an unsuitable person or its affiliate in exchange for its shares could increase our debt to equity ratio and would increase our leverage ratio.

On February 18, 2012, the Board of Directors of Wynn Resorts determined that Aruze USA, Inc., Universal Entertainment Corporation and Mr. Kazuo Okada are "unsuitable" under the provision of our articles of incorporation and redeemed and cancelled all of Aruze USA, Inc's, 24,549,222 shares of Wynn Resorts' common stock. Pursuant to our articles of incorporation, we issued the Redemption Note to Aruze USA, Inc. in redemption of the shares. For additional information on the redemption and the Redemption Note, see Notes to Condensed Consolidated Financial Statements, Note 15 – "Commitments and Contingencies."

#### **Critical Accounting Policies**

A description of our critical accounting policies is included in Item 7 of our Annual Report on Form 10-K for the year ended December 31, 2011. There has been no material change to these policies for the nine months ended September 30, 2012.

#### **Recently Issued Accounting Standards**

In July 2012, the Financial Accounting Standards Board ("FASB") issued an accounting standards update that is intended to simplify the guidance for testing the decline in the realizable value (impairment) of indefinite-lived intangible assets other than goodwill. The update allows for the consideration of qualitative factors in determining whether it is necessary to perform quantitative impairment tests. The effective date for this update is for the years and interim impairment tests performed for years beginning after September 15, 2012. This update is not expected to have a material impact on the Company's financial statements.

#### Item 3. Quantitative and Qualitative Disclosures About Market Risk

Market risk is the risk of loss arising from adverse changes in market rates and prices, such as interest rates, foreign currency exchange rates and commodity prices.

#### Interest Rate Risks

One of our primary exposures to market risk is interest rate risk associated with our debt facilities that bear interest based on floating rates. We attempt to manage interest rate risk by managing the mix of long-term fixed rate borrowings and variable rate borrowings, and using hedging activities. We cannot assure you that these risk management strategies will have the desired effect, and interest rate fluctuations could have a negative impact on our results of operations. We do not use derivative financial instruments, other financial instruments or derivative commodity instruments for trading or speculative purposes.

#### Interest Rate Swap Information

In June 2012, the Wynn Macau swap matured and the Company terminated its Wynn Las Vegas swap for a payment of \$2.4 million.

On September 28, 2012, we entered into two interest rate swap agreements intended to hedge a portion of the underlying interest rate risk on borrowings under the Amended Wynn Macau Credit Facilities. Under the two swap agreements, we pay a fixed interest rate (excluding the applicable interest margin) of 0.73% on notional

amounts corresponding to borrowings of HK\$3.95 billion (approximately US\$509.3 million) incurred under the Amended Wynn Macau Credit Facilities in exchange for receipts on the same amount at a variable interest rate based on the applicable HIBOR at the time of payment. These interest rate swaps fix the all-in interest rate on such amounts at 2.48% to 3.23%. These interest rate swap agreements mature in July 2017.

We measure the fair value of our interest rate swaps on a recurring basis. Changes in the fair values of these interest rate swaps for each reporting period recorded are recognized as an increase (decrease) in swap fair value in our Condensed Consolidated Statements of Income as the swaps did not qualify for hedge accounting. The fair value approximates the amount the Company would pay if these contracts were settled at the respective valuation dates. Fair value is estimated based upon current, and predictions of future, interest rate levels along a yield curve, the remaining duration of the instruments and other market conditions, and therefore, is subject to significant estimation and a high degree of variability and fluctuation between periods. We adjust this amount by applying a non-performance valuation, considering our creditworthiness or the creditworthiness of our counterparties at each settlement date as applicable. As of December 31, 2011, the interest rate swap liabilities of \$7.3 million were included in other current accrued liabilities.

#### Interest Rate Sensitivity

As of September 30, 2012, approximately 95% of our long-term debt was based on fixed rates. Based on our borrowings as of September 30, 2012, an assumed 1% change in the variable rates would cause our annual interest cost to change by \$2.7 million.

#### Foreign Currency Risks

The currency delineated in Wynn Macau's concession agreement with the government of Macau is the Macau pataca. The Macau pataca, which is not a freely convertible currency, is linked to the Hong Kong dollar, and in many cases the two are used interchangeably in Macau. The Hong Kong dollar is linked to the U.S. dollar and the exchange rate between these two currencies has remained relatively stable over the past several years. However, the exchange linkages of the Hong Kong dollar and the Hong Kong dollar and the U.S. dollar, are subject to potential changes due to, among other things, changes in Chinese governmental policies and international economic and political developments.

If the Hong Kong dollar and the Macau pataca are not linked to the U.S. dollar in the future, severe fluctuations in the exchange rate for these currencies may result. We cannot assure you that the current rate of exchange fixed by the applicable monetary authorities for these currencies will remain at the same level.

Because many of Wynn Macau's payment and expenditure obligations are in Macau patacas, in the event of unfavorable Macau pataca or Hong Kong dollar rate changes, Wynn Macau's obligations, as denominated in U.S. dollars, would increase. In addition, because we expect that most of the revenues for any casino that Wynn Macau operates in Macau will be in Hong Kong dollars, we are subject to foreign exchange risk with respect to the exchange rate between the Hong Kong dollar and the U.S. dollar. Also, because our Macau-related entities incur U.S. dollar-denominated debt, fluctuations in the exchange rates of the Macau pataca or the Hong Kong dollar, in relation to the U.S. dollar, could have adverse effects on Wynn Macau's results of operations, financial condition and ability to service its debt.

As of September 30, 2012, in addition to Hong Kong dollars, Wynn Macau also holds other foreign currencies, primarily CNH (offshore renminbi).

#### **Item 4. Controls and Procedures**

(a) *Disclosure Controls and Procedures*. The Company's management, with the participation of the Company's Chief Executive Officer and Chief Financial Officer, has evaluated the effectiveness of the Company's disclosure controls and procedures (as such term is defined in Rules 13a-15(e) and 15d-15(e) under the Securities Exchange Act of 1934, as amended (the "Exchange Act")) as of the end of the period covered by this report. In designing and evaluating the disclosure controls and procedures, mo matter how well designed and operated, can only provide reasonable assurance of achieving the desired control objectives and management is required to apply its judgment in evaluating the cost-benefit relationship of possible controls and procedures. Based on such evaluation, the Company's Chief Executive Officer and Chief Financial Officer have concluded that, as of the end of such period, the Company's disclosure controls and procedures were effective, at the reasonable assurance level, in recording, processing, summarizing and reporting, on a timely basis, information required to be disclosed by the Company in the reports that it files or submits under the Exchange Act and were effective in ensuring that information required to be disclosed by the Company in the reports that it files or submits under the Exchange Act and communicated to the Company's management, including the Company's Chief Executive Officer and Chief Financial Officer, as appropriate to allow timely decisions regarding required disclosure.

(b) *Internal Control Over Financial Reporting*. There were no changes in our internal control over financial reporting (as such term is defined in Rules 13a-15(f) and 15d-15(f) under the Exchange Act) during the quarter to which this report relates that materially affected, or are reasonably likely to materially affect, our internal control over financial reporting.

#### Part II – OTHER INFORMATION

#### Item 1. Legal Proceedings

We are occasionally party to lawsuits. As with all litigation, no assurance can be provided as to the outcome of such matters and we note that litigation inherently involves significant costs. For information regarding the Company's legal matters see Note 15 to our Condensed Consolidated Financial Statements in this Quarterly Report on Form 10-Q.

#### Item 1A. Risk Factors

A description of our risk factors can be found in Item 1A of our Annual Report on Form 10-K for the year ended December 31, 2011. Except as noted below, there were no material changes to those risk factors during the nine months ended September 30, 2012.

# Our information technology and other systems are subject to cyber security risk including misappropriation of customer information or other breaches of information security.

We rely on information technology and other systems to maintain and transmit customer financial information, credit card settlements, credit card funds transmissions, mailing lists and reservations information. In addition, our financial and recordkeeping processes are run from one central location at a secured off site Network Operations Center. We have substantially completed the implementation of industry best practice systems that are designed to meet all requirements of the Payment Card Industry standards for data protection, however, our information and processes are exposed to the ever-changing threat of compromised security, in the form of a risk of potential breach, system failure, computer virus, or unauthorized or fraudulent use by customers, company employees, or employees of third party vendors. The steps we take to deter and mitigate these risks may not be successful, and any resulting compromise or loss of data or systems could adversely impact, operations or regulatory compliance and could result in remedial expenses, fines, litigation, and loss of reputation, potentially impacting our financial results.

#### Item 2. Unregistered Sales of Equity Securities and Use of Proceeds

#### Dividend Restrictions

In November 2009, our Board of Directors approved the commencement of a regular quarterly cash dividend program. On October 24, 2012 the Company announced a cash dividend of \$8.00 per share, payable on November 20, 2012 to stockholders of record as of November 7, 2012. On July 17, 2012, the Company announced a dividend of \$0.50 per share, payable on August 14, 2012 to stockholders of record as of July 31, 2012. On May 7, 2012, the Company announced a dividend of \$0.50 per share, payable on June 4, 2012 to stockholders of record as of May 21, 2012. Wynn Resorts is a holding company and, as a result, our ability to pay dividends is dependent on our ability to obtain funds and our subsidiaries' ability to provide funds to us. Restrictions imposed by our subsidiaries' debt instruments significantly restrict certain key subsidiaries holding a majority of our assets, including Wynn Las Vegas, LLC and Wynn Macau, S.A. from making dividends or distributions to Wynn Resorts. Specifically, Wynn Las Vegas, LLC and certain of its subsidiaries are restricted under the indentures governing the first mortgage notes from making certain "restricted payments" as defined in the indentures. These restricted payments include the payment of dividends or distributions to any direct or indirect holders of equity interests of Wynn Las Vegas, LLC. Restricted payments cannot be made unless certain financial and non-financial criteria have been satisfied. While the Amended Wynn Macau Credit Facility contains similar restrictions, Wynn Macau is currently in compliance with all requirements, namely satisfaction of its leverage ratio, which must be met in order to pay dividends and is presently able to pay dividends in accordance with the Wynn Macau Credit Facilities.

#### Issuer Purchases of Equity Securities

In July 2012, the Company repurchased a total of 662 shares at an average price of \$96.44 per share in satisfaction of tax withholding obligations on vested restricted stock. In addition, as previously disclosed, based on the Board of Directors' finding of "unsuitability," on February 18, 2012, we redeemed and cancelled Aruze USA, Inc.'s 24,549,222 shares for a Redemption Note of \$1.94 billion. None of the foregoing shares were purchased as part of a publicly announced repurchase plan or program.

## Item 6. Exhibits

(a) Exhibits

# EXHIBIT INDEX

| Exhibit<br>No. | Description  |
|----------------|--|
| 3.1            | Second Amended and Restated Articles of Incorporation of the Registrant. (1)   |
| 3.2            | Fourth Amended and Restated Bylaws of the Registrant, as amended. (2)  |
| *10.1          | Common Terms Agreement Fourth Amendment Agreement, dated as of July 31, 2012 between, among others, Wynn Resorts (Macau) S.A. as the company and Bank of China Limited Macau Branch as security agent.   |
| *10.2          | Revolving Credit Facility Agreement dated July 31, 2012 among Wynn Resorts (Macau), S.A., Bank of China, Limited Macau Branch, and certain financial institutions as Project Facility Lenders.   |
| *10.3          | Third Amendment Agreement to the Hotel Facility Agreement dated July 31, 2012 among Wynn Resorts (Macau), S.A., Bank of China, Limited Macau Branch, and certain financial institutions as Hotel Facility Lenders  |
| *10.4          | Sixth Amended and Restated Art Rental and Licensing Agreement, dated as of July 1, 2012 between Stephen A. Wynn, as lessor, Wynn Las Vegas, LLC, as lessee.  |
| *31.1          | Certification of Chief Executive Officer of Periodic Report Pursuant to Rule 13a – 14(a) and Rule 15d – 14(a).   |
| *31.2          | Certification of Chief Financial Officer of Periodic Report Pursuant to Rule 13a – 14(a) and Rule 15d – 14(a).   |
| *32.1          | Certification of CEO and CFO Pursuant to 18 U.S.C. Section 1350.   |
| *101           | The following financial information from the Company's Quarterly Report on Form 10-Q for the quarter ended September 30, 2012, filed with the SEC on November 9, 2012 formatted in Extensible Business Reporting Language (XBRL): (i) the Condensed Consolidated Statements of Income for the three and nine months ended September 30, 2012 and 2011, (ii) the Condensed Consolidated Statements of Comprehensive Income for the three and nine months ended September 30, 2012 and 2011, (iii) the Condensed Consolidated Balance Sheets at September 30, 2012 and December 31 2011, (iv) the Condensed Consolidated Statements of Cash Flows for the nine months ended September 30, 2012 and 2011, (v) the Condensed Consolidated Statement of Stockholders' Equity at September 30, 2012, and (vi) Notes to Condensed Consolidated Financial Statements.* |

<sup>\*</sup> Filed herewith.

<sup>(1)</sup> Previously filed with Amendment No. 4 to the Form S-1 filed by the Registrant on October 7, 2002 (File No. 333-90600) and incorporated herein by reference.

<sup>(2)</sup> Previously filed with the Quarterly Report on Form 10-Q filed by the Registrant on August 9, 2007 and incorporated herein by reference.

# SIGNATURE

Pursuant to the requirements of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

# WYNN RESORTS, LIMITED

Dated: November 9, 2012

By: /s/ Matt Maddox

Matt Maddox Chief Financial Officer and Treasurer (Principal Financial Officer)

Exhibit 10.1

CLIFFORD CHANCE 高偉紳律師行

#### **CONFORMED COPY**

#### **DATED 31 JULY 2012**

WYNN RESORTS (MACAU) S.A. the Company

**CERTAIN FINANCIAL INSTITUTIONS** 

as Hotel Facility Lenders, Project Facility Lenders, Revolving Credit Facility Lenders and Hedging Counterparties

THE FINANCIAL INSTITUTIONS NAMED HEREIN

as Outgoing Global Coordinating Lead Arrangers, Outgoing Project Facility Agent, Outgoing Hotel Facility Agent, Outgoing Revolving Credit Facility Agent, Outgoing Intercreditor Agent and Outgoing Security Agent

> THE FINANCIAL INSTITUTIONS NAMED HEREIN as Incoming Global Coordinating Lead Arrangers

BANK OF CHINA LIMITED, MACAU BRANCH as Incoming Term Facility Agent

BANK OF CHINA LIMITED, MACAU BRANCH as Incoming Revolving Credit Facility Agent

BANK OF CHINA LIMITED, MACAU BRANCH as Incoming Intercreditor Agent

**BANK OF CHINA LIMITED, MACAU BRANCH** as Incoming Security Agent and Incoming POA Agent

**BANCO NACIONAL ULTRAMARINO, S.A.** as Second Ranking Finance Party and Outgoing POA Agent

COMMON TERMS AGREEMENT FOURTH AMENDMENT AGREEMENT

# CLIFFORD

HANCE

| Clause     | e  | Page |
|------------|--|------|
| 1.         | Definitions and Interpretation                         | 2    |
| 2.         | Change of Agents                                       | 7    |
| 3.         | Global Coordinating Lead Arrangers                     | 8    |
| 4.         | Amendment  | 8    |
| 5.         | Term Facility Advance                                  | 8    |
| 6.         | Order of Events on the Fourth Amendment Effective Date | 10   |
| 7.         | Representations  | 10   |
| 8.         | Security   | 10   |
| 9.         | Continuity and Further Assurance                       | 11   |
| 10.        | Miscellaneous  | 12   |
| 11.        | Governing Law  | 12   |
| Sche       | edule 1 Conditions Precedent                           | 13   |
| Sche       | edule 2 Amended Common Terms Agreement                 | 18   |
| SIGNATURES |  | 299  |

THIS AGREEMENT is dated 31 July 2012 and made between:

- (1) WYNN RESORTS (MACAU) S.A. (the "Company");
- (2) THE FINANCIAL INSTITUTION named on the signing pages as the Hotel Facility Lender;
- (3) THE FINANCIAL INSTITUTION named on the signing pages as the Project Facility Lender;
- (4) THE FINANCIAL INSTITUTION named on the signing pages as the Revolving Credit Facility Lender;
- (5) **THE FINANCIAL INSTITUTIONS** named on the signing pages as the Hedging Counterparties;
- (6) **BANC OF AMERICA SECURITIES ASIA LIMITED, DEUTSCHE BANK AG, HONG KONG BRANCH** and **SG AMERICAS SECURITIES, LLC** in their capacities as the outgoing Global Coordinating Lead Arrangers (the "**Outgoing Global Coordinating Lead Arrangers**");
- BANCO NACIONAL ULTRAMARINO, S.A., BANK OF AMERICA, N.A., BANK OF CHINA LIMITED, MACAU BRANCH, DBS
   BANK LTD., DEUTSCHE BANK AG, SINGAPORE BRANCH, INDUSTRIAL AND COMMERCIAL BANK OF CHINA (MACAU)
   LIMITED, JPMORGAN CHASE BANK, N.A., HONG KONG BRANCH, STANDARD CHARTERED BANK (HONG KONG) LIMITED,
   THE BANK OF NOVA SCOTIA, THE ROYAL BANK OF SCOTLAND PLC, SINGAPORE BRANCH, UBS AG HONG KONG BRANCH
   and UNITED OVERSEAS BANK LIMITED in their capacities as the incoming Global Coordinating Lead Arrangers (the "Incoming Global Coordinating Lead Arrangers");
- (8) **SOCIÉTÉ GÉNÉRALE, HONG KONG BRANCH** in its capacity as the outgoing Hotel Facility Agent (the "**Outgoing Hotel Facility Agent**");
- (9) **SOCIÉTÉ GÉNÉRALE, HONG KONG BRANCH** in its capacity as the outgoing Project Facility Agent (the "**Outgoing Project Facility Agent**");
- (10) **SOCIÉTÉ GÉNÉRALE, HONG KONG BRANCH** in its capacity as the outgoing Revolving Credit Facility Agent (the "**Outgoing Revolving Credit Facility Agent**");
- (11) SOCIÉTÉ GÉNÉRALE, HONG KONG BRANCH in its capacity as the outgoing Intercreditor Agent (the "Outgoing Intercreditor Agent");
- (12) SOCIÉTÉ GÉNÉRALE, HONG KONG BRANCH in its capacity as the outgoing Security Agent (the "Outgoing Security Agent");
- (13) BANK OF CHINA LIMITED, MACAU BRANCH in its capacity as the incoming Term Facility Agent (the "Incoming Term Facility Agent");
- (14) **BANK OF CHINA LIMITED, MACAU BRANCH** in its capacity as the incoming Revolving Credit Facility Agent (the "**Incoming Revolving** Credit Facility Agent");

- 1 -

- (15) **BANK OF CHINA LIMITED, MACAU BRANCH** in its capacity as the incoming Intercreditor Agent (the "**Incoming Intercreditor Agent**");
- (16) BANK OF CHINA LIMITED, MACAU BRANCH in its capacity as the incoming Security Agent (the "Incoming Security Agent");
- (17) BANCO NACIONAL ULTRAMARINO, S.A. as Second Ranking Finance Party and outgoing POA Agent (the "Outgoing POA Agent"); and
- (18) **BANK OF CHINA LIMITED, MACAU BRANCH** in its capacity as the incoming POA Agent (the "**Incoming POA Agent**").

#### **RECITALS:**

- (A) The parties hereto have agreed to amend certain Senior Finance Documents and enter into additional Senior Finance Documents.
- (B) It has been agreed to further amend the Common Terms Agreement as set out below.

#### **IT IS AGREED** as follows:

## 1. **DEFINITIONS AND INTERPRETATION**

## 1.1 **Definitions and incorporation of defined terms**

(a) In this Agreement:

"Amended Common Terms Agreement" means the Original Common Terms Agreement, as amended by this Agreement, the terms of which are set out in Schedule 2 (*Amended Common Terms Agreement*);

"Assignment of Reinsurances Confirmation" means a confirmation in relation to an Assignment of Reinsurances in form and substance satisfactory to the Incoming Security Agent and made between the relevant Direct Insurer and the Incoming Security Agent;

"**Completion Memorandum**" means the completion memorandum relating to the matters contemplated in this Agreement, in the Agreed Form;

"Completion Request" means a request in the Agreed Form for:

- (i) an Advance denominated in USD (in an amount at least equal to all outstanding amounts denominated in USD under the Project Facility as at the Fourth Amendment Effective Date);
- (ii) an Advance denominated in HKD (in an amount at least equal to all outstanding amounts denominated in HKD under the Project Facility as at the Fourth Amendment Effective Date); and
- (iii) other Advances denominated in HKD and USD,

## - 2 -

under the Term Facility to be made on the Fourth Amendment Effective Date, in order that the Term Facility is fully drawn as at the Fourth Amendment Effective Date;

"**Deed of Appointment and Priority Third Deed of Amendment**" means the Deed of Appointment and Priority Third Deed of Amendment dated on or about the date hereof between, among others, the Original First Ranking Lenders, the Hedging Counterparties, the Second Ranking Finance Party, the Company, the Outgoing Security Agent, the Incoming Security Agent, the Incoming Intercreditor Agent and the Outgoing POA Agent;

"**English Security Confirmation**" means the document so entitled dated on or about the Fourth Amendment Effective Date between the Company and the Security Agent;

"**First Macau Security Confirmation**" means the document so entitled dated on or about the Fourth Amendment Effective Date between the Company and the Security Agent;

"Fourth Amendment Effective Date" has the meaning given to such term in Clause 4 (Amendment);

"Guarantee Third Deed of Amendment and Acknowledgement" means the Guarantee Third Deed of Amendment and Acknowledgement dated on or about the date hereof between Wynn Asia 2, Wynn International, Wynn Holdings, Wynn HK, Palo, the Outgoing Security Agent and the Incoming Security Agent;

"Hong Kong Security Confirmation" means the document so entitled dated on or about the Fourth Amendment Effective Date between the Company, Wynn Holdings and the Security Agent;

"Hotel Facility Agent" has the meaning given to such term in the Original Common Terms Agreement;

"Hotel Facility Lender" has the meaning given to such term in the Original Common Terms Agreement;

"Irish and New York Security Documents" means:

- the Security Assignment and Charge on Account (governed by the laws of the Republic of Ireland) dated 22 September 2008 and made between Wynn Resorts (Macau) S.A. as Chargor and Société Générale, Hong Kong Branch as Security Agent;
- the Account Control Agreement (governed by the laws of the State of New York) dated 3 October 2008 made between Wynn Resorts (Macau) S.A. as Pledgor, Deutsche Bank Trust Company Americas as Depository Bank and Société Générale, Hong Kong Branch as Security Agent; and
- (iii) the Account Security Agreement (governed by the laws of the State of New York) dated 3 October 2008 made between Wynn Resorts (Macau) S.A. as the Company and Société Générale, Hong Kong Branch as Security Agent;

- 3 -

"**Livrança Updated Covering Letter**" means the letter from the Company to the Security Agent dated on or about the Fourth Amendment Effective Date in relation to the Livranças and the Livrança Covering Letter, duly acknowledged by Palo;

"Macau Security Confirmations" means the First Macau Security Confirmation, the Second Macau Security Confirmation and the Wong Security Confirmation;

"Nevada Security Confirmation" means the document so entitled dated on or about the Fourth Amendment Effective Date between the Company and the Security Agent;

"Original Common Terms Agreement" means the Common Terms Agreement as amended from time to time prior to the date of this Agreement;

"**Palo**" means Palo Real Estate Company Limited a company with limited liability incorporated in the Macau SAR with registration number 27319 SO;

"**Palo Share Pledge**" means the Share Pledge to be entered into between the Company, Wynn International, Wynn HK, Palo and the Security Agent;

"POA Agent" has the meaning given to such term in the Deed of Appointment and Priority;

"**Post-Amendment Global Transfer Agreement**" means the agreement so entitled dated on or about the date of this Agreement between, among others, the Company and the Incoming Intercreditor Agent;

**"Pre-Amendment Global Transfer Agreement**" means the agreement so entitled dated on or about the date of this Agreement between, among others, the Company and the Outgoing Intercreditor Agent;

"Project Facility" has the meaning given to such term in the Original Common Terms Agreement;

"Project Facility Agent" has the meaning given to such term in the Original Common Terms Agreement;

"Project Facility Agreement" has the meaning given to such term in the Original Common Terms Agreement;

"Project Facility Lender" has the meaning given to such term in the Original Common Terms Agreement;

"**Required Filings**" means any filing, notification, recording, stamping and registration required in respect of any of the Senior Finance Documents

- 4 -

referred to in paragraph 2 of Schedule 1 (*Conditions Precedent*) to this Agreement at Companies House in England and Wales, the Companies Registry in Hong Kong, the Companies Registration Office in Ireland, the Financial Supervision Commission in the Isle of Man, the Gaming Inspection and Coordination Bureau in Macau and in the register of charges of Wynn Asia 2;

"**Revolving Credit Facility Agreement**" means the agreement so entitled dated on or about the date of this Agreement between the Company, the Incoming Revolving Credit Facility Agent and the Revolving Credit Facility Lenders;

"Second Macau Security Confirmation" means the document so entitled dated on or about the Fourth Amendment Effective Date between the Company, Wynn HK, Wynn International and the Security Agent;

"Security Confirmation Documents" means:

- (i) each Macau Security Confirmation;
- (ii) the Hong Kong Security Confirmation;
- (iii) the English Security Confirmation; and
- (iv) the Nevada Security Confirmation;

"**Subordination Deed Third Deed of Amendment and Acknowledgement of Security**" means the Subordination Deed Third Deed of Amendment and Acknowledgement of Security dated on or about the date hereof between, among others, the Company, Wynn Resorts, Wynn Resorts Holdings, LLC, Wynn Asia 2, Wynn International, Wynn Holdings, Wynn HK, Worldwide Wynn, LLC, Wynn Design & Development, LLC, Wynn International Marketing, Ltd., Palo, the Outgoing Security Agent and the Incoming Security Agent;

**"Term Facility Agreement Third Amendment Agreement**" means the agreement so entitled dated on or about the date of this Agreement between the Company, the Incoming Term Facility Agent and the Term Facility Lenders; and

"Wong Security Confirmation" means the document so entitled dated on or about the Fourth Amendment Effective Date between the Mr. Wong Chi Seng and the Security Agent.

- (b) Unless a contrary indication appears, a term defined in or by reference in Schedule 2 (*Amended Common Terms Agreement*) or, if not defined in or by reference in such Schedule, the Deed of Appointment and Priority, has the same meaning in this Agreement.
- (c) The principles of construction and rules of interpretation set out in Schedule 2 (*Amended Common Terms Agreement*) shall have effect as if set out in this Agreement.

- 5 -

#### 1.2 Clauses

In this Agreement any reference to a "Clause" or a "Schedule" is, unless the context otherwise requires, a reference to a Clause or a Schedule to this Agreement.

# 1.3 Designation

In accordance with the Common Terms Agreement, each of the Company and the Outgoing Intercreditor Agent designates:

- (a) the Term Facility Agreement Third Amendment Agreement as a Senior Finance Document;
- (b) the Revolving Credit Facility Agreement as a Senior Finance Document;
- (c) this Agreement as a Senior Finance Document;
- (d) the Subordination Deed Third Deed of Amendment and Acknowledgement of Security as a Security Document;
- (e) the Guarantee Third Deed of Amendment and Acknowledgement as a Security Document;
- (f) Deed of Appointment and Priority Third Deed of Amendment as a Security Document;
- (g) the Power of Attorney as a Security Document;
- (h) with effect from the date thereof, the Palo Share Pledge as a Security Document;
- (i) with effect from the date thereof, the Palo Assignment of Insurances as a Security Document;
- (j) with effect from the date thereof, the Palo Floating Charge as a Security Document;
- (k) the Palo Pledge over Onshore Accounts as a Security Document,
- (l) with effect from the date thereof, the Cotai Mortgage as a Security Document;
- (m) with effect from the date thereof, the Cotai Power of Attorney as a Security Document;
- (n) each of the Security Confirmation Documents as a Security Document;
- (o) the Livrança Updated Covering Letter as a Security Document; and
- (p) with effect from the date thereof, each Assignment of Reinsurances Confirmation as a Security Document.



## 2. CHANGE OF AGENTS

## 2.1 **Resignation and Appointment**

Société Générale, Hong Kong Branch hereby resigns as Hotel Facility Agent, Project Facility Agent, Revolving Credit Facility Agent and Intercreditor Agent and:

- (a) the Incoming Term Facility Agent is hereby appointed as Term Facility Agent (it being agreed by all parties that the renaming of the Hotel Facility Agent referred to in paragraph (d) of Clause 8.3 (*Naming*) shall occur immediately prior to such appointment);
  - the Incoming Revolving Credit Facility Agent is hereby appointed as Revolving Credit Facility Agent; and
- (c) the Incoming Intercreditor Agent is hereby appointed as Intercreditor Agent,

in each case, as successor to Société Générale, Hong Kong Branch and with effect on and from the Fourth Amendment Effective Date (but subject to Clause 6 (*Order of Events on the Fourth Amendment Effective Date*)).

## 2.2 Notice Period

(b)

The parties hereto agree that the resignations and appointments referred to in Clause 2.1 (*Resignation and Appointment*) shall be effective notwithstanding the notice period specified in sub-clause 23.12.1 of clause 23.12 (*Resignation*) of the Original Common Terms Agreement or any other provision of the Senior Finance Documents. The Company and each Lender hereby approves each of the appointments referred to in Clause 2.1 (*Resignation and Appointment*).

## 2.3 Accession

Each of the Incoming Term Facility Agent, the Incoming Revolving Credit Facility Agent and the Incoming Intercreditor Agent hereby agrees with each other person who is or who becomes a party to the Senior Finance Documents that with effect on and from the Fourth Amendment Effective Date such Incoming Term Facility Agent, Incoming Revolving Credit Facility Agent and Incoming Intercreditor Agent shall be bound by the Senior Finance Documents and be entitled to exercise rights and be subject to obligations thereunder as Term Facility Agent, Revolving Credit Facility Agent and the Intercreditor Agent respectively. The parties hereto agree that, for the purposes of sub-clause 23.12.6 of clause 23.12 (*Resignation*) of the Original Common Terms Agreement, this Agreement shall constitute an Agent's Deed of Accession duly completed, executed and delivered (notwithstanding that it is executed under hand, and not by way of deed) by each of the Incoming Term Facility Agent, the Incoming Revolving Credit Facility Agent and the Incoming Intercreditor Agent.

## 2.4 Notices

Each of the Incoming Term Facility Agent, the Incoming Revolving Credit Facility Agent and the Incoming Intercreditor Agent hereby confirms that its Facility Office, telephone number, fax number, address and person designated by it for the purposes of Clause 29 (*Notices*) of the Amended Common Terms Agreement are as set out next to its name on the signature pages hereto.

- 7 -

#### 2.5 Without prejudice

The provisions of this Clause 2 (*Change of Agents*) is without prejudice to the provisions of clause 23.12 (*Resignation*) of the Original Common Terms Agreement which shall apply, subject to the variations referred to in this Clause 2 (*Change of Agents*).

## 3. GLOBAL COORDINATING LEAD ARRANGERS

With effect on and from the Fourth Amendment Effective Date (subject to Clause 6 (*Order of Events on the Fourth Amendment Effective Date*)), each Outgoing Global Coordinating Lead Arranger, in its capacity as a Global Coordinating Lead Arranger, shall be discharged from any further obligations towards the other Parties under the Original Common Terms Agreement and their respective rights against one another shall be cancelled (except in each case for those rights which arose prior to the Fourth Amendment Effective Date) and, as from that date, each Incoming Global Coordinating Lead Arranger shall assume the same obligations, and become entitled to the same rights, as if it had been an original Global Coordinating Lead Arranger and party to the Original Common Terms Agreement.

## 4. **AMENDMENT**

With effect from the date upon which the Incoming Intercreditor Agent confirms to the Lenders and the Company that it has received (or the Incoming Intercreditor Agent has waived receipt of, as the case may be) each of the documents or evidence listed in Schedule 1 (*Conditions Precedent*) in a form and substance satisfactory to the Incoming Intercreditor Agent, (such date being the "**Fourth Amendment Effective Date**"), the Original Common Terms Agreement shall be amended so that it shall be read and construed for all purposes as set out in Schedule 2 (*Amended Common Terms Agreement*).

## 5. TERM FACILITY ADVANCE

## 5.1 Completion Request

Subject to the occurrence of the Fourth Amendment Effective Date and receipt by the Incoming Intercreditor Agent and the Incoming Term Facility Agent of the Completion Request, the requirements specified in Clause 3.1 (*Drawdown conditions*) of the Amended Common Terms Agreement (and clause 11.1 (*Advances*) of the Term Facility Agreement) shall, with effect from the Fourth Amendment Effective Date, be deemed to have been satisfied in respect of the Advances requested in the Completion Request as if such request comprised an Advance Request and had been delivered (and notified) in accordance with the provisions thereof and each of the other provisions of Clause 3 (*Drawdown of Advances*) of the Amended Common Terms Agreement and the other Senior Finance Documents shall apply accordingly. Each Advance under the Term Facility which is outstanding on the Fourth Amendment Effective Date shall have an initial Interest Period set out opposite that Advance in the Completion Memorandum.

- 8 -

#### 5.2 **Repayment of Project Facility**

The Incoming Term Facility Agent is hereby directed by the Company (with the agreement and acknowledgement of the parties hereto) to pay:

- (a) an amount of each Advance requested in the Completion Request equal to all amounts then outstanding and denominated in USD and HKD, as the case may be, under the Project Facility directly to the Project Facility Lender, to be applied in repayment in full of the Project Facility, whereupon the Project Facility shall be permanently cancelled and the Available Facility (as defined in the Project Facility Agreement) in respect thereof shall be zero; and
- (b) such amount of each such Advance remaining after the application referred to in paragraph (a) above, as the Company directs in the Completion Request.

#### 5.3 [Not used]

#### 5.4 **Project Facility Lender**

Upon the repayment in full of the Project Facility in accordance with paragraph (a) of Clause 5.2 (*Repayment of Project Facility*) the Project Facility Lender shall cease to be a Lender for the purpose of the Amended Common Terms Agreement and shall be released from the indemnity set out in clause 15.3 (*Indemnity to the Project Facility Agent*) of the Project Facility Agreement (as defined in the Original Common Terms Agreement) (other than to the extent of any claims arising thereunder prior to the Fourth Amendment Effective Date).

#### 5.5 Waiver of notice and pro-rata payments requirements

The parties hereto waive (a) the notice requirement specified in sub-clause 8.2.1 of clause 8.2 (*Voluntary Prepayment of the Term Loan Facilities*) of the Original Common Terms Agreement and (b) the requirement in sub-clause 8.2.2 of clause 8.2 (*Voluntary Prepayment of the Term Loan Facilities*) of the Original Common Terms Agreement for payments or prepayments to be made *pro rata* as between the Term Facility and the Project Facility, in each case, in respect of the repayment in full of the Project Facility referred to in Clause 5.2 (*Repayment of Project Facility*).

#### 5.6 Breakfunding and accrued interest

- (a) The Company shall pay to Bank of China Limited, Macau Branch (for its own account) the amounts (at the times) contemplated by a letter dated on or prior to the date hereof and made between Company and Bank of China Limited, Macau Branch in connection with the roles of Bank of China Limited, Macau Branch under the Pre-Amendment Global Transfer Agreement and the Post-Amendment Global Transfer Agreement (the "Letter").
- (b) Notwithstanding clause 26.1 (*Payments under the Senior Finance Documents*) of the Original Common Terms Agreement, payments of accrued interest as contemplated by the Letter shall be made to Bank of China Limited, Macau Branch by the time contemplated in the Completion Memorandum (in the applicable currency or currencies) and not to the Outgoing Hotel Facility Agent for the account of the Term Facility Lender or the Outgoing Project Facility Agent for the account of the Project Facility Lender, as the case may be.

- 9 -

#### 6. ORDER OF EVENTS ON THE FOURTH AMENDMENT EFFECTIVE DATE

The parties hereto agree in respect of the actions, events and other steps that are set out in the Pre-Amendment Global Transfer Agreement, Clauses 2 (*Change of Agents*), 3 (*Global Coordinating Lead Arrangers*), 4 (*Amendment*) and 5 (*Term Facility Advance*) and in the Post-Amendment Global Transfer Agreement which are stated to occur on the Fourth Amendment Effective Date, that such actions, events and other steps shall (where those actions, events and other steps have been carried out in accordance with the Completion Memorandum) occur on the Fourth Amendment Effective Date in the order set out in the Completion Memorandum.

#### 7. **REPRESENTATIONS**

## 7.1 **Prior to the Fourth Amendment Effective Date**

The representations and warranties set out in schedule 4 of the amended Common Terms Agreement set out as Schedule 2 (*Amended Common Terms Agreement*) to this Agreement are deemed to be made by the Company (by reference to the facts and circumstances then existing) on the date of this Agreement.

#### 7.2 On the Fourth Amendment Effective Date

The representations and warranties set out in schedule 4 of the amended Common Terms Agreement set out as Schedule 2 (*Amended Common Terms Agreement*) to this Agreement are deemed to be made by the Company (by reference to the facts and circumstances then existing) on the Fourth Amendment Effective Date, as if any reference therein to any Senior Finance Document in respect of which any amendment, acknowledgement, confirmation, consolidation, novation, restatement, replacement or supplement is expressed to be made by any of the documents referred to in Clause 1.3 (*Designation*) included, to the extent relevant, such document and the Senior Finance Document as so amended, acknowledged, confirmed, consolidated, novated, restated, replaced or supplemented.

#### 7.3 Palo Real Estate Company Limited

For the purposes of any applicable provision of the Senior Finance Documents (including, without limitation, this Clause 7 (*Representations*)), the Company (for the benefit of itself and each other Obligor) has disclosed to the Lenders that, as at the date hereof, Wynn International and Wynn HK each legally and beneficially own 0.1% of the total issued share capital of Palo.

## 8. SECURITY

## 8.1 Execution by Security Agent

The Incoming Security Agent shall execute and deliver each of the Subordination Deed Third Deed of Amendment and Acknowledgement of Security, the Guarantee Third Deed of Amendment and Acknowledgement, the Deed of Appointment and Priority Third Deed of Amendment, each of the Security Confirmation Documents

- 10 -

and each of the Assignment of Reinsurances Confirmations and is authorised and instructed by the Incoming Intercreditor Agent (acting on the instructions of the Term Facility Lender and the Project Facility Lender) to do so accordingly.

#### 8.2 Release of Security

Each of the Outgoing Intercreditor Agent (acting on the instructions of each of the Term Facility Lender and the Project Facility Lender) and the Second Ranking Finance Party hereby instructs the Outgoing Security Agent to release the Security created or purported to be created under each Irish and New York Security Document, such release to be effective on and from the Fourth Amendment Effective Date (immediately prior to the effectiveness of the Outgoing Security Agent's resignation as Security Agent pursuant to the resignation letter referred to in paragraph 5(a) of Schedule 1 (*Conditions Precedent*)).

## 8.3 Naming

The parties agree that with effect on and from Bank of China Limited, Macau Branch becoming the sole Hotel Facility Lender and the sole Project Facility Lender:

- (a) the Sponsors' Subordination Deed (as defined in the Original Common Terms Agreement) (as amended, consolidated, supplemented, novated or replaced from time to time, including pursuant to the Subordination Deed Third Deed of Amendment and Acknowledgement of Security) shall be renamed the "Subordination Deed";
- (b) the Wynn Pledgors' Guarantee (as defined in the Original Common Terms Agreement) (as amended, consolidated, supplemented, novated or replaced from time to time, including pursuant to the Guarantee Third Deed of Amendment and Acknowledgement) shall be renamed the "Guarantee";
- (c) the Hotel Facility Agreement (as defined in the Original Common Terms Agreement) (as amended, consolidated, supplemented, novated or replaced from time to time, including pursuant to the Term Facility Agreement Third Amendment Agreement) shall be renamed the "Term Facility Agreement"; and
- (d) the Hotel Facility Agent (as defined in the Original Common Terms Agreement) shall be renamed the Term Facility Agent and the Hotel Facility Lenders (as defined in the Original Common Terms Agreement) shall be renamed the Term Facility Lenders.

## 9. CONTINUITY AND FURTHER ASSURANCE

#### 9.1 **Continuing obligations**

The provisions of the Common Terms Agreement shall, save as amended by this Agreement, continue in full force and effect. In particular, nothing in this Agreement shall affect the rights of the Senior Secured Creditors in respect of the occurrence of any Default which is continuing or which arises on or after the date of this Agreement.

- 11 -

#### 9.2 Further assurance

The Company shall, upon the written request of the Incoming Intercreditor Agent and the Company's expense, do all such acts and things reasonably necessary to give effect to the amendments effected or to be effected pursuant to this Agreement.

#### 10. MISCELLANEOUS

## 10.1 Incorporation of terms

The provisions of Clauses 1.5.1(e), 1.5.2 and 1.5.3 (*Third Party Rights*), Clause 16.2 (*Transaction Expenses*), Clause 16.4 (*Enforcement costs*), Clause 28 (*Non-recourse Liability*), Clause 29.1 (*Communications in Writing*) to 29.5 (*Electronic communication*), Clause 31 (*Partial Invalidity*), Clause 32 (*Remedies and Waivers*) and Clause 38 (*Jurisdiction*) of Schedule 2 shall be incorporated into this Agreement as if set out in full herein and as if references in those clauses to "Agreement" are references to this Agreement and cross-references to specified clauses thereof are references to the equivalent clauses set out or incorporated herein.

## 10.2 **Counterparts**

This Agreement may be executed in any number of counterparts, and this has the same effect as if the signatures on the counterparts were on a single copy of this Agreement.

#### 10.3 Conditions Subsequent

The Company shall procure that:

- (a) as soon as reasonably practicable and in any case within 45 days of the Fourth Amendment Effective Date, each Direct Insurer that has entered into an Assignment of Reinsurances enters into an Assignment of Reinsurances Confirmation in respect of each such Assignment of Reinsurances; and
- (b) as soon as reasonably practicable and in any case within 21 days of the Fourth Amendment Effective Date, each share certificate relating to the shares in the Company pledged pursuant to the Company Share Pledge are endorsed by Wynn HK and Wynn International, as appropriate, in favour of the Incoming Security Agent and that such endorsement is reflected in the Company's share register book,

and that evidence of the same is (within the time period specified, in each case, above) provided to the Incoming Intercreditor Agent.

## 11. GOVERNING LAW

This Agreement and any non-contractual obligations arising out of or in connection with it are governed by English law.

## This Agreement has been entered into on the date stated at the beginning of this Agreement.

- 12 -

#### SCHEDULE 1 CONDITIONS PRECEDENT

## Due establishment, authority and certification

1.

In relation to each Obligor and each Wynn Non-Obligor Subordination Deed Party, receipt by the Incoming Intercreditor Agent of a certificate signed by a duly authorised signatory of that Person and which:

- (a) either (A) attaches a copy of that Person's Governing Documents or (B) certifies that the copy of that Person's Governing Documents (which was previously delivered to the Outgoing Intercreditor Agent on or about 14 September 2004 or subsequently) remains correct, complete and in full force and effect as at a date no earlier than the Fourth Amendment Effective Date;
- (b) attaches a copy of a board resolution or such other equivalent corporate authorisation approving the execution, delivery and performance of the Senior Finance Documents referred to in paragraph 2 below to which it is a party, the terms and conditions thereof and the transactions contemplated thereby, authorising a named person or persons to sign such Senior Finance Documents and any document to be delivered by that Person pursuant to such Senior Finance Documents and authorising the signatory of the relevant certificate to sign certificates in connection therewith;
- (c) (in the case of the Company only) certifies that each document listed in this Schedule 1 and delivered by an Obligor is correct, complete and in full force and effect and has not been amended or superseded as at a date no earlier than the Fourth Amendment Effective Date;
- (d) confirms that:
  - (i) (in the case of each Obligor only) borrowing, guaranteeing or securing as appropriate, the total commitments of all Lenders in respect of the Term Facility and the Revolving Credit Facility would not cause any borrowing, guarantee, security or similar limit binding on any such Person to be exceeded; and
  - (ii) (in the case of each Wynn Non-Obligor Subordination Deed Party other than the Licensor and the Corporate Services Provider) Liens granted by such person pursuant to the Security Documents to which it is a party would not cause any limit on the granting of security or similar restriction binding on it to be exceeded;
- (e) (in the case of each Obligor only) certifies that no Material Adverse Effect has occurred and is continuing nor could reasonably be expected to occur; and
- (f) (in the case of the Company only) certifies that no Default is continuing or would occur as a result of Palo becoming an Obligor.

- 13 -

#### Senior Finance Documents

2.

- (a) Receipt by the Incoming Intercreditor Agent of an original of each of the following Senior Finance Documents, in each case duly executed by the parties thereto:
  - (i) the Term Facility Agreement Third Amendment Agreement;
  - (ii) the Revolving Credit Facility Agreement;
  - (iii) this Agreement;
  - (iv) the Subordination Deed Third Deed of Amendment and Acknowledgement of Security;
  - (v) the Guarantee Third Deed of Amendment and Acknowledgement;
  - (vi) Deed of Appointment and Priority Third Deed of Amendment;
  - (vii) each of the Security Confirmation Documents;
  - (viii) a Fee Letter between the Incoming Term Facility Agent, the Incoming Revolving Credit Facility Agent, the Incoming Intercreditor Agent and the Incoming Security Agent and the Company;
  - (ix) a Fee Letter between the Incoming Intercreditor Agent and the Company relating to the payment by the Company of an upfront fee to each person that, immediately following the consummation of the transactions set out in the Post-Amendment Global Transfer Agreement, shall be a Lender;
  - (x) the Livranças, the Livrança Covering Letter (in each case duly endorsed by Palo) and the Livrança Updated Covering Letter (signed by the Company and endorsed and countersigned, as required, by Palo);
  - (xi) a Finance Party Accession Undertaking executed by each of the Incoming Term Facility Agent, the Incoming Revolving Credit Facility Agent and the Incoming Intercreditor Agent;
  - (xii) the Palo Pledge over Onshore Accounts; and
  - (xiii) any other document entered into which the Incoming Intercreditor Agent and the Company agree prior to the Fourth Amendment Signing Date to designate as a Senior Finance Document.
- (b) Each Senior Finance Document referred to in this paragraph 2 has been duly authorised, executed and delivered by such of the Obligors party thereto and (save in respect of any registration required in respect of the Subordination Deed Third Deed of Amendment and Acknowledgement of Security, the Palo Pledge over Onshore Accounts and each of the Security Confirmation Documents at the Companies House in England and Wales, the Hong Kong Companies Registry, the Conservatória dos Registos Comercial e de Bens

- 14 -

Móveis in Macau SAR, the Conservatória do Registo Predial in Macau SAR, the applicable Uniform Commercial Code filing office for local/county, state and federal Uniform Commercial Code filings and the Isle of Man Companies Registry, as applicable, based on the Senior Finance Document subject to the filing) duly filed, notified, recorded, stamped and registered as necessary.

 (c) All conditions precedent to the effectiveness thereof (other than any such conditions relating to the occurrence of the Fourth Amendment Effective Date) have been satisfied or waived in accordance with their respective terms and each such Senior Finance Document (save as provided in paragraph (b)) is in full force and effect accordingly.

## 3. Legal opinions

Receipt by the Incoming Intercreditor Agent of legal opinions (substantially in the form distributed to the Incoming Intercreditor Agent prior to the Fourth Amendment Signing Date) from:

- (a) Mr Henrique Saldanha, Macanese legal adviser to the Senior Secured Creditors;
- (b) Lionel Sawyer & Collins, Nevada legal adviser to the Senior Secured Creditors;
- (c) M&P Legal, Isle of Man legal adviser to the Senior Secured Creditors;
- (d) Walkers, Cayman legal adviser to the Senior Secured Creditors;
- (e) Clifford Chance, Hong Kong SAR legal advisers to the Senior Secured Creditors; and
- (f) Clifford Chance, English legal advisers to the Senior Secured Creditors.

#### Fees and expenses

4.

5.

Receipt by the Incoming Intercreditor Agent of evidence that:

- (a) all taxes, fees and other costs payable in connection with the execution, delivery, filing, recording, stamping and registering of the documents referred to in this Schedule 1; and
- (b) all fees, costs and expenses due to the Senior Secured Creditors and their advisers under the Senior Finance Documents on or before the Fourth Amendment Effective Date,

have been paid or shall be paid (to the extent that such amounts have been duly invoiced) by no later than the Fourth Amendment Effective Date.

# Outgoing Security Agent and Incoming Security Agent

Evidence that the Outgoing Security Agent has resigned and the Incoming Security Agent has been appointed as its successor as Security Agent, each in accordance with clause 17.1 (*Resignation of the Security Agent*) of the Deed of Appointment and Priority, such evidence to comprise (without limitation):

(a) a resignation letter from the Outgoing Security Agent addressed to the Secured Parties (with a copy to the Company);

- 15 -

- (b) evidence that the Incoming Security Agent has become a party to the Gaming Concession Consent Agreement and the Land Concession Consent Agreement (in place of the Outgoing Security Agent);
- (c) each assignment and transfer agreement (and each similar document as may be required) executed between the Outgoing Security Agent and the Incoming Security Agent whereby the Outgoing Security Agent assigns and transfers all of its rights, benefits and obligations (if any) under the Finance Documents to the Incoming Security Agent;
- (d) evidence that the Outgoing Security Agent has given prior notice of such assignment and transfer to the Government of the Macau SAR; and
- (e) evidence that all filings, registrations and notifications necessary to preserve the priority of the Security in connection with the resignation of the Outgoing Security Agent and the appointment of the Incoming Security Agent have been made.

## 6. Security

Receipt by the Incoming Intercreditor Agent of evidence that each Security Document has been duly filed, notified, recorded, stamped and (save as provided in paragraph 2(b) above) registered as necessary and all other actions necessary in the reasonable opinion of the Incoming Intercreditor Agent or the Incoming Security Agent to perfect the Security have been carried out.

## 7. **Process agents**

(c)

Where such appointment is required under any Senior Finance Document referred to in paragraph 2 above that Palo is a party to, a copy of process agent acceptance of its appointment by Palo for the acceptance of legal proceedings.

#### 8. Other documents and evidence

- (a) Evidence that all those things specified as being required to be done on or prior to the Fourth Amendment Effective Date in the Completion Memorandum have been done in accordance with the Completion Memorandum.
- (b) Receipt by the Incoming Intercreditor Agent and the Incoming Term Facility Agent of the Completion Request.
  - Receipt by the Incoming Intercreditor Agent of a copy of:
    - (i) the Pre-Amendment Global Transfer Agreement; and
    - (ii) the Post-Amendment Global Transfer Agreement,

- 16 -

in each case, duly executed by the parties thereto.

(d)

- Receipt by the Incoming Intercreditor Agent of evidence that:
  - (i) the transfers and acquisitions referred to in the Pre-Amendment Global Transfer Agreement have been completed; and
  - (ii) the transfers and acquisitions referred to in the Post-Amendment Global Transfer Agreement will be completed on the Fourth Amendment Effective Date.
- (e) A copy of any other authorisation or other document, opinion or assurance which the Incoming Intercreditor Agent considers to be necessary or desirable (if it has notified the Company accordingly prior to the Fourth Amendment Effective Date) in connection with the entry into and performance of the transactions contemplated by any Senior Finance Document or for the validity and enforceability of any Senior Finance Document.
- (f) A valuation report by Knight Frank Macau Limited relating to the Site Facilities and the Cotai Site Facilities and addressed to, and/or capable of being relied upon by, the Incoming Intercreditor Agent as Intercreditor Agent (for itself and for and on behalf of the other Senior Secured Creditors).
- (g) A copy, certified by an authorised signatory of the Company to be a true copy, of the Cotai Land Concession Contract.
- (h) Receipt by the Incoming Intercreditor Agent of evidence that the Macau SAR government has: (a) accepted and agreed unconditionally to the submission made by the Company to the Macau SAR government on 26 July 2012, (b) consented to the increase in the Company's financial indebtedness (contemplated by the Senior Finance Documents (including, without limitation, as the same may be entered into, amended, consolidated, supplemented, novated or replaced on or about the Fourth Amendment Effective Date) and (c) confirmed that the Gaming Concession Consent Agreement and the Land Concession Consent Agreement (as amended, consolidated, supplemented, novated or replaced from time to time prior to the Fourth Amendment Effective Date) continue to apply with respect to such financial indebtedness.

- 17 -

#### SCHEDULE 2 AMENDED COMMON TERMS AGREEMENT

DATED 14 SEPTEMBER 2004

WYNN RESORTS (MACAU) S.A. the Company

**CERTAIN FINANCIAL INSTITUTIONS** 

as Term Facility Lenders, Revolving Credit Facility Lenders and Hedging Counterparties

## BANK OF AMERICA, N.A., BANK OF CHINA LIMITED, MACAU BRANCH, BANCO NACIONAL ULTRAMARINO, S.A., DBS BANK LTD., DEUTSCHE BANK AG, SINGAPORE BRANCH, INDUSTRIAL AND COMMERCIAL BANK OF CHINA (MACAU) LIMITED, JPMORGAN CHASE BANK, N.A., HONG KONG BRANCH, STANDARD CHARTERED BANK (HONG KONG) LIMITED, THE BANK OF NOVA SCOTIA, THE ROYAL BANK OF SCOTLAND PLC, SINGAPORE BRANCH, UBS AG HONG KONG BRANCH and UNITED OVERSEAS BANK

# LIMITED

as Global Coordinating Lead Arrangers

#### BANK OF CHINA LIMITED, MACAU BRANCH as Term Facility Agent

BANK OF CHINA LIMITED, MACAU BRANCH as Revolving Credit Facility Agent

BANK OF CHINA LIMITED, MACAU BRANCH as Intercreditor Agent

BANK OF CHINA LIMITED, MACAU BRANCH as Security Agent

COMMON TERMS AGREEMENT

(As amended by the Common Terms Agreement Amendment Agreement dated 14 September 2005, the Common Terms Agreement Second Amendment Agreement dated 27 June 2007 the Common Terms Agreement Third Amendment Agreement dated 8 September 2009 and the Common Terms Agreement Fourth Amendment Agreement dated 31 July 2012)

- 18 -

# CONTENTS

| Claus | Se  | Page |
|-------|---|------|
| 1.    | Definitions And Interpretation                      | 22   |
| 2.    | Conditions Precedent                                | 79   |
| 3.    | Drawdown of Advances                                | 82   |
| 4.    | Availability Periods                                | 84   |
| 5.    | Purpose   | 85   |
| 6.    | Pro Rata Drawings                                   | 86   |
| 7.    | [NOT USED]  | 86   |
| 8.    | Repayments, Prepayments And Cancellation            | 86   |
| 9.    | Interest, Interest Periods And Default Interest     | 92   |
| 10.   | Changes To The Calculation Of Interest              | 94   |
| 11.   | Tax Gross Up And Indemnities                        | 96   |
| 12.   | Increased Costs                                     | 98   |
| 13.   | Currency And Other Indemnities                      | 99   |
| 14.   | Illegality  | 100  |
| 15.   | Mitigation By The Senior Secured Creditors          | 101  |
| 16.   | Fees, Costs And Expenses                            | 101  |
| 17.   | Representations And Warranties                      | 102  |
| 18.   | Covenants   | 103  |
| 19.   | Events Of Default                                   | 104  |
| 20.   | Application Of Enforcement Proceeds                 | 105  |
| 21.   | Changes To The Parties                              | 106  |
| 22.   | Hedging Counterparties                              | 110  |
| 23.   | Agents And Global Coordinating Lead Arrangers       | 111  |
| 24.   | Conduct Of Business By The Senior Secured Creditors | 119  |
| 25.   | Sharing Among The Senior Secured Creditors          | 119  |
| 26.   | Payment Mechanics                                   | 121  |
| 27.   | Set-Off   | 123  |
| 28.   | Non-Recourse Liability                              | 123  |
| 29.   | Notices   | 124  |
| 30.   | Calculations And Certificates                       | 127  |
| 31.   | Partial Invalidity                                  | 127  |
| 32.   | Remedies And Waivers                                | 128  |
| 33.   | Intercreditor Arrangements                          | 128  |
| 34.   | Amendments And Waivers                              | 131  |

- 19 -

| 35.  | Counterparts  | 132 |  |  |
|--|---|-----|--|--|
| 36.  | Language  | 132 |  |  |
| 37.  | Governing Law   | 132 |  |  |
| 38.  | Jurisdiction  | 132 |  |  |
| 39.  | Confidentiality   | 133 |  |  |
| 40.  | Gaming Authorities                                      | 134 |  |  |
| Sche   | Schedule 1 The Lenders and Hedging Counterparties       |     |  |  |
| Schedule 2 Conditions Precedent                        |   |     |  |  |
| Schedule 3 Form of Advance Request                     |   |     |  |  |
| Schedule 4 Representations and Warranties              |   | 162 |  |  |
| Schedule 5 Covenants                                   |   | 175 |  |  |
| Schedule 6 Accounts                                    |   | 210 |  |  |
| Schedule 7 Insurance                                   |   | 218 |  |  |
| Sche   | dule 8 Hedging Arrangements                             | 260 |  |  |
| Schedule 9 Mandatory Prepayment                        |   | 264 |  |  |
| Schedule 10 Events of Default                          |   | 267 |  |  |
| Schedule 11 Transfers and Accession                    |   | 276 |  |  |
| Schedule 12 Permits                                    |   | 285 |  |  |
| Schedule 13 [Not used]                                 |   | 287 |  |  |
| Schedule 14 Form of Additional Lender's Accession Deed |   | 288 |  |  |
| Schedule 15 Form of Compliance Certificate             |   | 289 |  |  |
| Schedule 16 [Not used]                                 |   |     |  |  |
| Schedule 17 [Not used]                                 |   |     |  |  |
| Sche   | Schedule 18 Monthly Construction Period Report          |     |  |  |
| Sche   | Schedule 19 Forms of Opening Conditions Certificates 29 |     |  |  |
|  |   |     |  |  |

- 20 -

#### THIS AGREEMENT is made on the 14th day of September 2004

## **BETWEEN:**

- (1) WYNN RESORTS (MACAU) S.A. (the "Company");
- (2) THE FINANCIAL INSTITUTIONS defined below as Term Facility Lenders;
- (3) THE FINANCIAL INSTITUTIONS defined below as Revolving Credit Facility Lenders;
- (4) **THE FINANCIAL INSTITUTIONS** defined below as Hedging Counterparties;
- (5) BANK OF AMERICA, N.A., BANK OF CHINA LIMITED, MACAU BRANCH, BANCO NACIONAL ULTRAMARINO, S.A., DBS BANK LTD., DEUTSCHE BANK AG, SINGAPORE BRANCH, INDUSTRIAL AND COMMERCIAL BANK OF CHINA (MACAU) LIMITED, JPMORGAN CHASE BANK, N.A., HONG KONG BRANCH, STANDARD CHARTERED BANK (HONG KONG) LIMITED, THE BANK OF NOVA SCOTIA, THE ROYAL BANK OF SCOTLAND PLC, SINGAPORE BRANCH, UBS AG HONG KONG BRANCH and UNITED OVERSEAS BANK LIMITED in their capacities as global coordinating lead arrangers of the Facilities (the "Global Coordinating Lead Arrangers" or "GCLAs");
- (6) **BANK OF CHINA LIMITED, MACAU BRANCH** in its capacity as Term Facility Agent;
- (7) BANK OF CHINA LIMITED, MACAU BRANCH in its capacity as Revolving Credit Facility Agent;
- (8) BANK OF CHINA LIMITED, MACAU BRANCH in its capacity as Intercreditor Agent; and
- (9) BANK OF CHINA LIMITED, MACAU BRANCH in its capacity as Security Agent.

## WHEREAS:

- (A) The Senior Secured Creditors have agreed, subject to the terms and conditions contained in the Senior Finance Documents, to make available to the Company certain loan facilities for the purpose of the Projects, the Cotai Project and for general corporate purposes and/or to enter into other agreements or arrangements associated therewith.
- (B) The parties have agreed to enter into this Agreement to set out certain terms and conditions which are common to all the Facility Agreements and to agree certain terms and conditions upon and subject to which the Senior Secured Creditors shall or may enjoy, exercise or enforce their rights, discretions and remedies under the Senior Finance Documents.

- 21 -

#### NOW IT IS HEREBY AGREED as follows:

## 1. DEFINITIONS AND INTERPRETATION

## 1.1 **Definitions**

In this Agreement, except as otherwise defined herein or to the extent the context otherwise requires, capitalised terms used shall have the following meanings:

"Acceptable Bank" means:

- (a) a bank notified by the Company to the Security Agent which:
  - (i) is licensed by the Hong Kong Monetary Authority, the Monetary Authority of Singapore, the Financial Supervisory
     Commission in Taiwan, the Office of the Superintendent of Financial Institutions in Canada, the Federal Reserve System in the United States of America and/or the Financial Services Agency in Japan; and
  - (ii) has, at all times, general obligations rated at least BBB-, Baa3 or BBB from one or more of S&P, Moody's or Fitch respectively;
- (b) any of Banco Comercial de Macau, S.A., Banco Espirito Santo Asia Limited, Banco Nacional Ultramarino, S.A., Bank of China Limited, Macau Branch, Industrial and Commercial Bank of China (Macau) Limited or Tai Fung Bank Limited; or
- (c) any bank which is confirmed by the Security Agent (acting reasonably) as acceptable (including any bank confirmed by the Security Agent or its predecessor as acceptable prior to the Fourth Amendment Effective Date).

"Account" means an account:

- (i) held in Macau, Hong Kong, the United States or any other jurisdiction, and on terms, reasonably acceptable to the Security Agent, by a member of the Restricted Group with an Acceptable Bank; and
- (ii) subject to Liens in favour of the Security Agent in form and substance satisfactory to the Security Agent.

"Account Bank" means, in relation to an Account, the bank with which the Account is maintained.

"Account Bank Notices and Acknowledgements" mean the notices and acknowledgements to be delivered to and executed by each Account Bank in respect of each Account in accordance with the Charges over Accounts and this Agreement.

"Acquisition Agreement" has the meaning given to that term in the Common Terms Agreement Third Amendment Agreement.

- 22 -

## "Additional Facility Agent" means:

(b)

- (a) each bank or financial institution appointed as a facility agent for Additional Lenders under an Additional Lender Facility Agreement and which has executed and delivered to the Intercreditor Agent:
  - (i) a duly completed Agent's Deed of Accession; and
  - a duly completed Finance Party Accession Undertaking executed by such party, the Intercreditor Agent and all other parties hereto acting through the Intercreditor Agent for this purpose (the authority for which is hereby conferred on the Intercreditor Agent and which the Intercreditor Agent shall promptly deliver to the Security Agent),

each of which the Intercreditor Agent shall promptly copy to the Company and the other Senior Secured Creditors; or

each successor to any such bank or financial institution appointed in accordance with this Agreement.

"Additional Lender Facility" means each term loan facility and each revolving credit facility provided by the Additional Lenders to the Company.

"Additional Lender Facility Agreement" means an agreement between the Additional Lenders, the Additional Facility Agent and the Company for the provision of the Additional Lender Facility.

"Additional Lender Facility Availability Period" means, in relation to the Additional Lender Facility, the period specified in respect thereof in Clause 4.4 (*Additional Lender Facility Availability Period*).

"Additional Lender's Accession Deed" means a deed of accession in substantially the form set out in Schedule 14 (Form of Additional Lender's Accession Deed).

"Additional Lenders" means the parties who have agreed to provide the Company with loan facilities permitted by paragraph 2.1(f) of Part B of Schedule 5 (*Covenants*) and who have each executed and delivered to the Intercreditor Agent:

- (a) a duly completed Additional Lender's Accession Deed; and
- (b) a duly completed Finance Party Accession Undertaking executed by such party, the Intercreditor Agent and all other parties hereto acting through the Intercreditor Agent for this purpose (the authority for which is hereby conferred on the Intercreditor Agent and which the Intercreditor Agent shall promptly deliver to the Security Agent),

each of which the Intercreditor Agent shall promptly copy to the Company and the other Senior Secured Creditors.

"Additional Lending Group" means, with respect to an Additional Lender Facility Agreement, the Additional Lenders party to such Additional Lender Facility Agreement, acting as a lending group in accordance with, and subject to the decision making rules under, such Additional Lender Facility Agreement.

- 23 -

"Advance" means an advance (as from time to time reduced by repayment or prepayment) made or to be made under a Facility.

"Advance Date" means the date on which an Advance is required to be made.

"Advance Request" means, in relation to an Advance under the Term Loan Facilities, a request for an Advance in substantially the form set out in Schedule 3 (*Form of Advance Request*) and, in relation to an Advance under the Revolving Credit Facilities, in substantially the form set out in schedule 2 to the Revolving Credit Facility Agreement or the equivalent schedule to the Additional Lender Facility Agreement setting out the form of advance request, as the case may be.

"Advisers" means the Technical Adviser, the Insurance Adviser and the Tax Adviser.

"Affiliate" as applied to any Person, means any other Person which, directly or indirectly, is in control of, is controlled by, or is under common control with, such Person. For purposes of this definition, "control" (including, with correlative meanings, the terms "controlling," "controlled by," and "under common control with") as applied to any Person means the power, directly or indirectly, to (a) vote 10% or more of the shares or other securities having ordinary voting power for the election of the Board of Directors (or persons performing similar functions) of such Person or (b) direct or cause the direction of the management and policies of such Person, whether by contract or otherwise (*provided* that Mr Wong Chi Seng shall not, by virtue of fulfilling either of these requirements alone as a result of the shares held by him in the Company as at the Signing Date or as a result of his role as executive director of the Company, be an Affiliate of the Company).

"Affiliate Agreement" means any agreement entered into by any Obligor with an Affiliate of that Obligor involving expenditures by any party thereto or any other flow of funds of not less than USD1,000,000 or its equivalent, but excluding any agreement entered into between the Company and a Subsidiary Obligor.

"Agent" means the Intercreditor Agent or a Facility Agent, as the case may be.

"Agent's Deed of Accession" means a deed of accession in substantially the form of Part A of Schedule 11 (Transfers and Accession).

"**Agreed Form**" means, in relation to any document, the form most recently initialled for the purposes of identification as such by the Company and the Intercreditor Agent with such changes as the Intercreditor Agent may agree with the Company.

"A. M. Best" means A.M. Best Company or its successor company.

"Ancillary Finance Documents" means:

- (a) the Fee Letters; and
- (b) the Underwriting Agreement.

- 24 -

"Anti-Bribery Laws" has the meaning given to that term in paragraph 38 of Schedule 4 (Representations and Warranties).

"Anti-Terrorism Law" means each of:

- (a) Executive Order No. 13224 of September 23, 2001 Blocking Property and Prohibiting Transactions With Persons Who Commit, Threaten To Commit, or Support Terrorism (the Executive Order);
- (b) the Uniting and Strengthening America by Providing Appropriate Tools Required to Intercept and Obstruct Terrorism Act of 2001, Public Law 107-56 (commonly known as the USA Patriot Act);
- (c) the Money Laundering Control Act of 1986, Public Law 99-570;
- (d) the International Emergency Economic Powers Act, 50 U.S.C. App. §§ 1701 et seq, the Trading with the Enemy Act, 50 U.S.C. App. §§ 1 et seq, any sanctions, prohibitions or requirements imposed by any U.S. executive order (an "**Executive Order**") or by any sanctions programme administered by the Office of Foreign Assets Control ("**OFAC**") of the U.S. Department of the Treasury; and
- (e) any similar law, sanction or measure enacted in the United States of America or resolutions adopted by the United Nations Security Council or by law or regulation in the jurisdictions in which any Obligor, any member of the Group or any Excluded Subsidiary conducts any business.

"Approved Corporate Administrative Fees" means, for any Fiscal Year, an amount, when added to any other Corporate Administrative Fees paid by or on behalf of the Company during such Fiscal Year, as does not exceed 50% of the corporate administrative overhead costs incurred by Wynn Resorts during such Fiscal Year in relation to its management of the Wynn Resorts Group.

"**Approved IP Fees**" means the IP Fees as set out in the IP Agreement but without regard to any amendment, variation or supplement, whether pursuant to the terms of the IP Agreement or otherwise, subsequent to the Third Amendment to Intellectual Property License Agreement referred to in the definition thereof in this Clause 1.1.

"Asset Sale" means any Disposition of Property other than:

- (a) the granting of any Lien permitted by paragraph 3 of Part B of Schedule 5 (*Covenants*);
- (b) any Disposition permitted by paragraph 5 of Part B of Schedule 5 (*Covenants*) (*provided* that, in the case of paragraph 5(a) of Part B of Schedule 5 (*Covenants*), Dispositions of Property thereunder shall be considered "Asset Sales" to the extent of any proceeds thereof not applied to the replacement of Property pursuant to paragraph 5(a)(ii) of Part B of Schedule 5 (*Covenants*)).

"Assignment of Rights" means the assignment so entitled dated on or about the date of this Agreement between the Company and the Security Agent.

- 25 -

"Assignment of Insurances" means the Assignment of Onshore Insurance Policies dated on or about the date of this Agreement between the Company and the Security Agent.

"Assignments of Reinsurances" means each assignment of Reinsurance so entitled between the relevant Direct Insurer and the Security Agent.

"Auditors" means Ernst & Young LLP or such other firm of independent accountants of international recognised standing as may be appointed by the Company.

"Availability Period" means, as the case may be, the Term Facility Availability Period, the Additional Lender Facility Availability Period or the Revolving Credit Facility Availability Period.

"Available Commitment", in relation to each Lender under each Facility Agreement, has the meaning given in that Facility Agreement.

"Board of Directors" means:

- (a) with respect to a corporation, the board of directors of the corporation;
- (b) with respect to a limited partnership, the board of directors of the general partner of the partnership; and
- (c) with respect to any other Person, the board or committee of such Person serving a similar function.

"Break Costs" means the amount (if any) by which:

(a) the additional interest which a Lender should have received for the period from the date of receipt by such Lender of all or any part of its participation in an Advance or Unpaid Sum to the last day of the current Interest Period in respect of that Advance or Unpaid Sum, had the principal amount or Unpaid Sum received been paid on the last day of that Interest Period;

exceeds:

(b) the amount which that Lender would be able to obtain by placing an amount equal to the principal amount or Unpaid Sum received by it on deposit with a leading bank in the London interbank market (or, in the case of any principal amount or Unpaid Sum denominated in HK dollars, the Hong Kong interbank market) for a period starting on the Business Day following receipt or recovery and ending on the last day of the current Interest Period,

provided that Break Costs shall not include any loss of margin.

"**Business Day**" means a day (other than a Saturday or Sunday) on which banks are open for general business in the Macau SAR, Hong Kong SAR and New York and, save for the purposes of Clause 3.1.1 (*Drawdown conditions*), London and Singapore.

"Capital Expenditure" means, in relation to any Person, for any period, the aggregate of all expenditures by such Person for the acquisition or leasing (pursuant

- 26 -

to a capital lease or a finance lease) of fixed or capital assets (including, without limitation, real property) or additions to equipment (including replacements, capitalized repairs and improvements during such period) which should be capitalized under applicable GAAP.

"**Capital Lease Obligations**" means, as to any Person, the obligations of such Person to pay rent or other amounts under any lease of (or other arrangement conveying the right to use) real or personal property, or a combination thereof, which obligations are required to be classified and accounted for as capital leases or finance leases under applicable GAAP, and, for the purposes of this Agreement, the amount of such obligations at any time shall be the capitalized amount thereof at such time determined in accordance with applicable GAAP.

"**Capital Stock**" means any and all shares, interests, participations or other equivalents (however designated) of capital stock of a corporation, any and all classes of membership interests in a limited liability company, any and all classes of partnership interests in a partnership, any and all equivalent ownership interests in a Person and any and all warrants, rights or options to purchase any of the foregoing.

"Cash Flow Available for Debt Service" or "CFADS" means, in relation to any period, EBITDA for such period *plus*, without duplication, the sum of:

- (a) decreases in Working Capital for such period; and
- (b) any other non-cash charges,

and minus, without duplication, the sum of:

- (c) increases in Working Capital for such period;
- (d) any other non-cash credits;
- (e) the aggregate amount actually paid by each member of the Restricted Group in cash during such period on account of Capital Expenditures;
- (f) the aggregate amount actually paid by each member of the Restricted Group in cash during such period on account of any accrued charges from any prior period; and
- (g) Tax paid by each member of the Restricted Group during such period,

determined on a consolidated basis which includes members of the Restricted Group only (and which, for the avoidance of doubt, does not take account of any amount to the extent it (i) would otherwise have the effect of increasing Cash Flow Available for Debt Service and (ii) is in any way derived from or attributable or otherwise related to or connected with an Excluded Project, an Excluded Subsidiary, a Resort Management Agreement, the grant of any Subconcession or any interest, right or claim in respect thereof).

"**Certificate of Substantial Completion**" means, in relation to the Projects, any "Original Project Certificate of Substantial Completion" or "Expansion Project Certificate of Substantial Completion", each as defined in the Construction Contract.

- 27 -

"Change of Control" means the occurrence of any of the following:

- (a) the direct or indirect sale, transfer, conveyance or other disposition (other than by way of merger or consolidation), in one or a series of related transactions, of all or substantially all of the properties or assets of any Wynn Obligor to any Person (except as may be permitted by this Agreement or any Security Document and excluding any disposition forming part of a Permitted Cotai Reorganisation);
- (b) the adoption of a plan relating to the liquidation or dissolution of any Wynn Obligor or any successor thereto, but excluding any voluntary liquidation, winding-up, dissolution of, or similar action with respect to, Palo after a Permitted Cotai Reorganisation; or
- (c) a Wynn Event.

"Charge over HK Accounts" means the charge so entitled between the Company and the Security Agent in the Agreed Form.

"**Charges over Accounts**" means the Pledge over Onshore Accounts, the Charge over HK Accounts, the US Operating Account Control Agreement, the Palo Pledge over Onshore Accounts and the documents granting the Liens referred to in the definition of "Account" in this Clause 1.1.

"Claim Proceeds" means the proceeds of a claim (a "Recovery Claim") against any party to a Project Document (or a Cotai Resort Management Agreement) or any of such party's Affiliates (or any employee, officer or adviser) in relation to the Project Documents (or a Cotai Resort Management Agreement) except for Excluded Claim Proceeds, and after deducting:

- (a) any reasonable expenses which are incurred by any member of the Restricted Group to persons who are not members of the Restricted Group; and
- (b) any Tax incurred and required to be paid by a member of the Restricted Group (as reasonably determined by the relevant member of the Restricted Group on the basis of existing rates and taking into account any available credit, deduction or allowance),

in each case in relation to that Recovery Claim.

"Code" means the Internal Revenue Code of 1986 of the United States of America, as amended from time to time.

"**Common Terms Agreement Amendment Agreement**" means the agreement so entitled dated 14 September 2005 between the Company, Banc of America Securities Asia Limited, Deutsche Bank AG, Hong Kong Branch, Société Générale Asia Limited, Société Générale Hong Kong Branch and certain other financial institutions.

"Common Terms Agreement Fourth Amendment Agreement" means the agreement so entitled dated 31 July 2012 between the parties thereto.

"Common Terms Agreement Second Amendment Agreement" means the agreement so entitled dated 27 June 2007 between the parties thereto.

"Common Terms Agreement Third Amendment Agreement" means the agreement so entitled dated 8 September 2009 between the parties thereto.

"**Commonly Controlled Entity**" means an entity, whether or not incorporated, which is under common control with any Obligor within the meaning of section 4001 of ERISA or is part of a group that includes such Person and that is treated as a single employer under section 414 of the Code.

"**Company Share Pledge**" means the pledge over shares in the Company dated on or about the date of this Agreement between Wynn HK, Wynn International, the Company and the Security Agent.

"Company's CP Satisfaction Date Certificate" has the meaning given in paragraph 6 of Part A of Schedule 2 (Conditions Precedent).

"Compensation Proceeds Account" means the account so designated in Schedule 6 (Accounts).

"Completion Memorandum" has the meaning given in the Common Terms Agreement Second Amendment Agreement.

"Completion Request" has the meaning given in the Common Terms Agreement Fourth Amendment Agreement.

"Compliance Certificate" means a certificate in substantially the form set out in Schedule 15 (Form of Compliance Certificate).

"**Concession Contract**" means the concession contract dated 24 June 2002 between the Macau SAR and the Company for the operation of games of chance and other games in casinos in the Macau SAR.

"Concession Contract Performance Bond" means the guarantee to be provided under article 61 of the Concession Contract.

"**Confidentiality Undertaking**" means a confidentiality undertaking in substantially the form set out in Part C of Schedule 11 (*Transfers and Accession*) or any other form agreed between the Company and the Intercreditor Agent.

"**Construction Contract**" means the amended and restated construction contract dated 14 September 2005 between the Prime Contractor and the Company for the construction of the Original Project and the Expansion.

"**Construction Contract Direct Agreement**" means the Amended and Restated Construction Contract Direct Agreement dated 14 September 2005 between the Prime Contractor, the Company and the Security Agent in the Agreed Form.

"Construction Disbursement Account" means the account so designated in Schedule 6 (Accounts).

- 29 -

"**Construction Period Insurances**" means the insurances identified as such in Appendix 1 (*Construction Period Insurances*) to Schedule 7 (*Insurance*) and effected in accordance with the terms of Schedule 7 (*Insurance*).

"**Contractors**" means any architects, consultants, designers, contractors, suppliers or any other Persons party to a Major Project Document and engaged by the Company or any other member of the Restricted Group in connection with the design, engineering, development, construction, installation, maintenance or operation of either Project (including the Prime Contractor).

"**Contractual Obligation**" means, as to any Person, any provision of any security issued by such Person or of any agreement, instrument or other undertaking to which such Person is a party or by which it or any of its Property is bound.

"**Corporate Administrative Fees**" means any fees payable by the Company to Wynn Resorts pursuant to the Corporate Administrative Fees Agreement in respect of any corporate administrative overhead costs incurred by Wynn Resorts in relation to its management of the Wynn Resorts Group.

"**Corporate Administrative Fees Agreement**" means the agreement dated as of 1 January 2007 between the Company and Wynn Resorts regarding, among other things, the payment of the Company's portion of any Corporate Administrative Fees.

"Corporate Services Provider" means Wynn Resorts in its capacity as a party to the Corporate Administrative Fees Agreement.

"**Cotai Land Concession Contract**" means the land concession contract agreed to by and between Palo, the Company and the Macau SAR dated 2 May 2012 pursuant to Dispatch number 16/2012, and includes any novation, assignment, transfer or other Disposition to the Company or replacement thereof in the name of the Company in connection with a Permitted Cotai Reorganisation.

"**Cotai Mortgage**" means the mortgage relating to the Cotai Site to be entered into between Palo (or, if Palo's interest in the Cotai Site is transferred to the Company pursuant to a Permitted Cotai Reorganisation, the Company) and the Security Agent.

"**Cotai Opening Date**" means the date upon which all Licenças de Ocupação required pursuant to applicable Legal Requirements in respect of the Cotai Project have been issued by the Macau SAR and the Cotai Project is fully open for business to the general public.

"**Cotai Power of Attorney**" means the irrevocable power of attorney to be granted by Palo (or, if Palo's interest in the Cotai Site is transferred to the Company pursuant to a Permitted Cotai Reorganisation, the Company) in favour of the Security Agent in connection with the Cotai Mortgage.

"Cotai Project" means a luxury hotel resort, retail and entertainment complex and casino facilities currently known as "Wynn Cotai" to be designed, developed, constructed, operated and maintained on land leased under the Cotai Land Concession Contract (excluding any Excluded Project located on a portion of the Cotai Site).

- 30 -

"**Cotai Resort Management Agreement**" means any agreement entered into by the Company and/or Palo (that does not conflict with the Cotai Land Concession Contract and all other applicable Legal Requirements) with an Excluded Subsidiary or other third party in connection with the use by such Excluded Subsidiary or other third party of a portion of the Cotai Site for the purposes of an Excluded Project.

"Cotai Site" means the land described in the Cotai Land Concession Contract.

"**Cotai Site Easements**" means the easements appurtenant, easements in gross, licence agreements and other rights running for the benefit of Palo (or, if Palo's interest in the Cotai Site is transferred to the Company pursuant to a Permitted Cotai Reorganisation, the Company) and/or appurtenant to the Cotai Site.

"Cotai Site Facilities" means

- (a) the Cotai Site; and
- (b) the Project Works (whether completed or uncompleted) in respect of the Cotai Project.

"**CP** Satisfaction Date" means the date on which all conditions precedent set out in Part A of Schedule 2 (*Conditions Precedent*) have been satisfied in accordance with sub-clause 2.1.2 of Clause 2.1 (*Conditions Precedent to the CP Satisfaction Date*).

"**Current Assets**" means, at any date, all amounts (other than cash) which would, in conformity with applicable GAAP, be set forth opposite the caption "total current assets" (or any like caption) on a balance sheet of the Restricted Group at such date prepared on a consolidated basis which includes members of the Restricted Group only (and which, for the avoidance of doubt, does not take account of any amount to the extent it (i) would otherwise have the effect of increasing Current Assets and (ii) is in any way derived from or attributable or otherwise connected with an Excluded Project, an Excluded Subsidiary, a Resort Management Agreement, the grant of any Subconcession or any interest, right or claim in respect thereof).

"**Current Liabilities**" means, at any date, all amounts that would, in conformity with applicable GAAP, be set forth opposite the caption "total current liabilities" (or any like caption) on a balance sheet of the Restricted Group at such date prepared on a consolidated basis which includes members of the Restricted Group only (and which, for the avoidance of doubt, does not take account of any amount to the extent it (i) would otherwise have the effect of reducing Current Liabilities and (ii) is in any way derived from or attributable or otherwise connected with an Excluded Project, an Excluded Subsidiary, a Resort Management Agreement, the grant of any Subconcession or any interest, right or claim in respect thereof), but excluding:

- (a) the current portion of any Funded Debt of any member of the Restricted Group; and
- (b) without duplication of paragraph (a) above, all Financial Indebtedness consisting of Revolving Credit Facilities Advances to the extent otherwise included therein.

- 31 -

"Debenture" means the debenture so entitled dated on or about the date of this Agreement between the Company and the Security Agent.

"**Decision**" means the giving of a consent, the making of an agreement or the exercise of any other right, power, discretion or determination in respect of any matter which, under this Agreement or any other Senior Finance Document, requires such consent, agreement or exercise to be given or made by more than one Senior Secured Creditor or by the Required Lenders.

"Decision Date" has the meaning given in Clause 33.1 (Notices of Required Decisions).

"**Deed of Appointment and Priority**" means the deed so entitled dated on or about the date of this Agreement between, among others, the Lenders, the Performance Bond Provider, the Company, the Agents and the Security Agent.

"**Default**" means an Event of Default or any event or circumstance specified in Schedule 10 (*Events of Default*) hereto which would become (with the expiry of a grace period, the giving of notice, the making of any determination as permitted under the Senior Finance Documents or any combination of any of the foregoing) an Event of Default.

"Derivatives Counterparty" has the meaning given in paragraph 6 of Part B of Schedule 5 (Covenants).

"Diamond Completion Date" means the date of completion of the Diamond Expansion as notified to the Intercreditor Agent by the Company.

"**Diamond Construction Contract**" means the contract for the design, engineering and construction of the Diamond Expansion to be entered into by the Company and the Prime Contractor (or any of its Affiliates).

"**Diamond Expansion**" means that part of the Projects which, as of the Second Amendment Signing Date, is being constructed on the site of the above-ground parking garage comprised in the Original Project and is contemplated to include the 40 floor Wynn Diamond suites, a parking facility, restaurants, retail space and gaming areas.

"**Diamond Opening Date**" means the date upon which all Licenças de Ocupação required pursuant to applicable Legal Requirements in respect of the Diamond Expansion have been issued by the Macau SAR and the Diamond Expansion is fully open for business to the general public.

"Direct Agreements" means each of the following documents:

- (a) the Gaming Concession Consent Agreement;
- (b) the Land Concession Consent Agreement;
- (c) the Construction Contract Direct Agreement;
- (d) the PASA Direct Agreement;

- 32 -

- (e) the Account Bank Notices and Acknowledgements; and
- (f) the Insurer Notices and Acknowledgements.

"**Direct Insurances**" means a contract or policy of insurance of any kind from time to time taken out or effected by, on behalf of or in favour of the Company or any other member of the Restricted Group (whether or not in conjunction with any other person) with one or more insurers in accordance with the terms of Schedule 7 (*Insurance*).

"Direct Insurer" means the insurer(s) with whom a Direct Insurance is placed from time to time in accordance with Schedule 7 (Insurance).

"**Disposition**" means, with respect to any Property, any sale, lease, sale and leaseback, assignment, conveyance, transfer or other disposition thereof (whether legal or equitable); and the terms "Dispose" and "Disposed of" shall have correlative meanings.

"**Disqualified Stock**" means any Capital Stock or other ownership or profit interest of any Obligor that any Obligor is or, upon the passage of time or the occurrence of any event, may become obligated to redeem, purchase, retire, defease or otherwise make any payment in respect thereof for consideration other than Capital Stock (other than Disqualified Stock).

"EBITDA" means, in relation to any period, the Net Income of the Restricted Group for such period *plus*, without duplication and to the extent reflected as a charge in the Company's statement of such Net Income for such period, the sum of:

- (a) income Tax expense (whether or not paid during such period) other than Tax on gross gaming revenue;
- (b) amortization or write-off of debt discount and debt issuance costs and interest, commissions, discounts and other fees and charges associated with Financial Indebtedness (including the Advances);
- (c) depreciation and amortization expense;
- (d) amortization of intangibles (including goodwill);
- (e) an amount equal to the aggregate net non-cash loss on the Disposition of Property during such period (other than sales of inventory in the ordinary course of business);
- (f) any extraordinary expenses or losses; and
- (g) any foreign currency translation losses,

and minus, without duplication and to the extent included in the statement of such Net Income for such period, the sum of:

(h) interest income;

- 33 -

- (i) an amount equal to the aggregate net non-cash gain on the Disposition of Property during such period (other than sales of inventory in the ordinary course of business);
- (j) any extraordinary income or gains;
- (k) any foreign currency translation gains; and
- (l) any upfront premium or similar income or gains derived from, or in connection with the grant of, any Subconcession,

all (including Net Income) as determined on a consolidated basis which includes members of the Restricted Group only (and which, for the avoidance of doubt, does not take account of any amount to the extent it (i) would otherwise have the effect of increasing EBITDA and (ii) is in any way derived from or attributable or otherwise related to or connected with an Excluded Project, an Excluded Subsidiary, a Resort Management Agreement, the grant of any Subconcession or any interest, right or claim in respect thereof) and otherwise in accordance with applicable GAAP.

"ECF Percentage" means, with respect to any period:

- (a) 50% of the Excess Cash Flow if the Leverage Ratio as of the last day of such period is greater than 4:1; or
- (b) zero if the Leverage Ratio as of the last day of such period is 4:1 or less.

"Effective Date" has the meaning given in the Common Terms Agreement Second Amendment Agreement.

"Eminent Domain Proceeds" means all amounts and proceeds (including monetary instruments) received in respect of any Event of Eminent Domain relating to any member of the Restricted Group or any of its assets, including either Project, less any costs or expenses incurred by any member of the Restricted Group or its agents in collecting such amounts and proceeds.

"Enforcement Notice" has the meaning given in the Deed of Appointment and Priority.

"Enforcement Proceeds" means all moneys received or recovered by the Security Agent after the Security has become enforceable in accordance with the terms of the Security Documents from the exercise or enforcement of the Security.

"Environment" means land, including any natural or man-made structures; water; and air.

"Environmental Claim" means any formal claim by any Person as a result of or in connection with any material violation of Environmental Law which claim could reasonably be expected to give rise to any remedy or penalty (whether interim or final) or liability for any member of the Restricted Group or any Senior Secured Creditor (in its capacity as such in the transactions contemplated by the Senior Finance Documents).

- 34 -

"Environmental Law" means any law or regulation of the Macau SAR with regard to:

- (a) harm to the health of humans; or
- (b) the pollution or protection of the Environment.

"Environmental Licence" means any material permit, licence, approval, registration, notification, exemption or any other authorisation required under any Environmental Law.

"Equator Principles" means the voluntary set of guidelines for determining, assessing and managing environmental and social risk in project financing promoted and published by the International Finance Corporation and first adopted by other financial institutions on or about 4 June 2003.

"Equity" means, at any time, the aggregate of the US dollar equivalents of:

- (a) the amounts paid up by the Shareholders by way of subscription for shares in the Company; and
- (b) the amounts advanced to the Company and outstanding at such time by way of Shareholder Loans.

#### "Equity Issuance" means:

- (a) any allotment or issuance (or the entering into by the Company or any other member of the Restricted Group of any agreement to allot or issue), or any grant to any Person of any right (whether conditional or unconditional) to call for or require the allotment or issuance of, any share or equity interest, or other securities (including without limitation bonds, notes, debentures, stock or similar instrument) which are convertible (whether at the option of the holder(s) thereof, the Company or otherwise) into shares or equity interests in the Company or other member of the Restricted Group, or any depositary receipt(s) in respect of any such share or equity interest; or
- (b) any grant of any option, warrant or other right of acquisition in respect of any such share, equity interest, other security or depositary receipt,

*provided* that for the avoidance of doubt, "Equity Issuance" shall not include (i) any secondary sales of any shares, equity interests or other securities of the Company or any other member of the Restricted Group by any or all of the holders of such shares, equity interests or other securities or (ii) any allotment or issuance or other grant to the Company or any other member of the Restricted Group of shares, equity interests or other securities in a Subsidiary Obligor.

"Equity Issuance Proceeds" means the amount of the proceeds (if not in cash, the monetary value thereof) of any Equity Issuance after deducting:

- (a) fees and expenses reasonably incurred in connection with such Equity Issuance by the Company or other member of the Restricted Group; and
- (b) any Taxes incurred or required to be paid by the Company or other member of the Restricted Group in connection with such Equity Issuance (as reasonably determined by the Company, on the basis of existing rates and taking account of any available credit, deduction or allowance).

- 35 -

"ERISA" means the Employee Retirement Income Security Act of 1974 of the United States of America, as amended from time to time.

"Event of Default" means any event or circumstance set out in Schedule 10 (Events of Default).

## "Event of Eminent Domain" means, with respect to any Property:

- (a) any compulsory transfer or taking by condemnation, seizure, eminent domain or exercise of a similar power, or transfer under threat of such compulsory transfer or taking or confiscation of such Property or the requisition of the use of such Property, by any agency, department, authority, commission, board, instrumentality or political subdivision of any Governmental Authority having jurisdiction; or
- (b) any settlement in lieu of paragraph (a) above.

"Event of Loss" means, with respect to any property or asset (tangible or intangible, real or personal), any of the following:

- (a) any loss, destruction or damage of such property or asset;
- (b) any actual condemnation, seizure or taking by exercise of the power of eminent domain or otherwise of such property or asset, or confiscation of such property or asset or the requisition of the use of such property or asset; or
- (c) any settlement in lieu of paragraph (b) above.

"Excess Cash Flow" means, in relation to any period, CFADS for such period plus, without duplication the US dollar equivalents of:

- (a) to the extent included in Net Income for such period, interest income received during such period,
- minus, without duplication, the US dollar equivalents of:
- (b) the aggregate amount of Financing Costs paid by the Company or any other member of the Restricted Group in cash during such period;
- (c) the aggregate amount of all prepayments of Advances under the Revolving Credit Facilities during such period to the extent accompanying permanent voluntary reductions of the commitments thereunder and all voluntary prepayments of Term Loan Facility Advances during such period;
- (d) the aggregate amount of all scheduled principal payments of the Company under the Facility Agreements made during such period (other than in respect of any Revolving Credit Facility to the extent there is not an equivalent

- 36 -

permanent reduction in commitments thereunder such that after giving effect to such commitment reduction the Company would not be able to reborrow all or any of the amount so prepaid); and

(e) the aggregate of all other scheduled payments of any Financial Indebtedness permitted to be incurred by a member of the Restricted Group pursuant to paragraphs 2.1(e) and 2.1(f) of Part B of Schedule 5 (*Covenants*) falling due and any voluntary prepayments thereof made during such period (other than in respect of any overdraft or revolving facility to the extent there is not an equivalent permanent reduction in commitments thereunder such that after giving effect to such commitment reduction the relevant member of the Restricted Group would not be able to reborrow all or any of the amount so prepaid),

determined on a consolidated basis which includes members of the Restricted Group only (and which, for the avoidance of doubt, does not take account of any amount to the extent it (i) would otherwise have the effect of increasing Excess Cash Flow and (ii) is in any way derived from or attributable or otherwise related to or connected with an Excluded Project, an Excluded Subsidiary, a Resort Management Agreement, the grant of any Subconcession or any interest, right or claim in respect thereof).

## "Excluded Claim Proceeds" means:

- (a) any proceeds of a Recovery Claim which the Company notifies the Intercreditor Agent are, or are to be, applied:
  - (i) to satisfy (or reimburse a member of the Restricted Group which has discharged) any liability, charge or claim upon a member of the Restricted Group by a person which is not a member of the Restricted Group; or
  - (ii) in the replacement, reinstatement and/or repair of assets of members of the Restricted Group which have been lost, destroyed or damaged,

in each case as a result of the events or circumstances giving rise to that Recovery Claim, if those proceeds are deposited into and retained in an Account pending such application and are so applied within 6 months, or such longer period as the Intercreditor Agent may agree, after receipt; and

(b) any proceeds of a Recovery Claim or Recovery Claims in respect of a Project Document or Project Documents (other than any Major Project Document and other than any proceeds referred to in paragraph (a) above) not exceeding an aggregate amount equal to USD25,000,000 (or its equivalent) during the period commencing on the Fourth Amendment Effective Date and ending on (and including) the Release Date.

"Excluded Project" means any gaming, hotel or resort related business, development or undertaking of any kind (including any business, development or undertaking referred to in clause (c) of the definition of Permitted Businesses) in the Macau SAR (other than the Projects or the Cotai Project) and, save as contemplated by any Resort Management Agreement therefor in the case of the Company or Palo, neither

- 37 -

involving nor permitting any claim, interest, liability, right of recourse of any kind in connection therewith against or in any member of the Restricted Group or its assets, including the Projects and the Cotai Project. For the avoidance of doubt, an Excluded Project may be located on a portion of the Cotai Site *provided that* to the extent any such Excluded Project located on a portion of the Cotai Site is funded by the proceeds of any Advance, such funding is applied solely towards the financing or refinancing of the development costs incurred in relation to such Excluded Project (it being agreed that such development costs include, but are not limited to, costs and expenses related to design, development, land acquisition, construction, site preparation, equipping, pre-opening expenses and capitalized interest), and *further provided that* the following additional conditions are met prior to the earlier of entry into of any contract for and the commencement of the construction of an Excluded Project on the Cotai Site:

- the entire Cotai Site shall remain part of the Project Security and subject to the Cotai Mortgage and there shall be no adverse effect on the validity or enforceability of, or the effectiveness or ranking of any Security of the Finance Parties as a result of the Excluded Project being located on the Cotai Site;
- (ii) the development, operation and maintenance of such Excluded Project:
  - (A) complies in all material respects with all applicable Legal Requirements (including, without limitation, Environmental Laws), the Cotai Land Concession Contract and the Concession Contract; and
  - (B) does not materially interfere with or materially obstruct (or otherwise have any material adverse effect on) the development, operation and maintenance of the Cotai Project; and
- (iii) any claim, interest or liability, or right of recourse of any kind of any counterparty to any Resort Management Agreement, in respect of such Excluded Project against or in the Company, Palo or any other member of the Restricted Group or any of their respective assets (including, without limitation, the Projects and the Cotai Project) is limited to an aggregate amount equal to all revenues derived by the Company (or, as the case may be, Palo) in respect of that Excluded Project and any other assets of the Company (or, as the case may be, Palo) comprised in that Excluded Project (and which do not form part of and are not necessary to ensure to the Restricted Group the full benefit of any Project or the Cotai Project).

"Excluded Subsidiary" means a Subsidiary of the Company exclusively engaged in the development, financing, ownership, leasing or operation of Excluded Projects on terms which, save as contemplated by any Resort Management Agreement to which such Subsidiary is party in the case of the Company, neither involve nor permit any claim, interest, liability, right of recourse of any kind in connection therewith against or in any member of the Restricted Group or its assets, including either of the Projects.

"Expansion" means that part of the Projects comprised in the "Expansion Project Casino" as defined in the Construction Contract as at the Third Amendment Signing Date, but excluding the Diamond Expansion.

- 38 -

"Expansion Opening Date" means the date upon which all Licenças de Ocupação required pursuant to applicable Legal Requirements in respect of the Expansion have been issued by the Macau SAR and the Expansion is fully open for business to the general public.

"Facility" means any of:

- (a) the Term Loan Facilities; or
- (b) the Revolving Credit Facilities.

"Facility Agents" means the Term Facility Agent, the Revolving Credit Facility Agent and each Additional Facility Agent.

"Facility Agreements" means:

- (a) the Term Facility Agreement;
- (b) the Revolving Credit Facility Agreement; and
- (c) each Additional Lender Facility Agreement.

"Facility Office" means the office or offices notified by a Senior Secured Creditor to the relevant Facility Agent under the Facilities and by the relevant Facility Agent to the Company and the Intercreditor Agent in writing on or before the date it becomes a Senior Secured Creditor (or, following that date, by not less than 10 Business Days' written notice) as the office or offices through which it shall perform its obligations under the relevant Facility.

"FCPA" has the meaning given to that term in paragraph 38 of Schedule 4 (Representations and Warranties).

"Fee Letters" means each of the fee letters entered into from time to time between the Company on the one hand and any of the Agents and the Security Agent on the other hand.

## "Final Repayment Date" means:

- (a) in relation to the Term Facility, the sixth anniversary of the Fourth Amendment Effective Date; and
- (b) in relation to an Additional Lender Facility providing for a term loan, the date set forth in the applicable Additional Lender Facility Agreement.

"Finance Document" has the meaning given in the Deed of Appointment and Priority.

"Finance Party Accession Undertaking" has the meaning given in the Deed of Appointment and Priority.

"Financial Indebtedness" means, in relation to any Person at any date, without duplication:

- (a) all indebtedness of such Person for borrowed money;
- (b) all obligations of such Person for the purchase price of Property or services to the extent the payment of such obligations is deferred for a period in excess of 90 days (other than trade payables incurred in the ordinary course of such Person's business);
- (c) all obligations of such Person evidenced by notes, bonds, debentures or other similar instruments;
- (d) all indebtedness created or arising under any conditional sale or other title retention agreement with respect to Property acquired by such Person (unless the rights and remedies of the seller or lender under such agreement in the event of default are limited to repossession or sale of such Property);
- (e) all Capital Lease Obligations (to the extent treated as finance or capital lease obligations in accordance with applicable GAAP) or Synthetic Lease Obligations of such Person;
- (f) any indebtedness of such Person for or in respect of receivables sold or discounted (other than any receivables to the extent they are sold on a non-recourse basis or on a basis where recourse is limited solely to warranty claims relating to title or objective characteristics of the relevant receivables);
- (g) any indebtedness of such Person in respect of any amount raised under any other transaction (including any forward sale or purchase agreement) having the commercial effect of a borrowing;
- (h) all indebtedness of such Person, contingent or otherwise, as an account party under acceptance, letter of credit, completion guaranties, performance bonds or similar facilities;
- (i) all obligations of such Person, contingent or otherwise, to purchase, redeem, retire or otherwise acquire for value any Capital Stock of such Person;
- (j) all obligations of such Person in respect of Swap Agreements or any other derivative transaction entered into in connection with protection against or benefit from fluctuation in any rate or price;
- (k) all Guarantee Obligations of such Person in respect of obligations (whether of such Person or another person) of the kind referred to in paragraphs (a) through (j) above;
- (l) all obligations of the kind referred to in paragraphs (a) through (k) above secured by (or for which the holder of such obligation has an existing right, contingent or otherwise, to be secured by) any Lien on Property (including, without limitation, accounts and contract rights) owned by such Person, whether or not such Person has assumed or become liable for the payment of such obligation; and
- (m) the liquidation value of any preferred Capital Stock of such Person or its Subsidiaries held by any Person other than such Person and its Wholly Owned Subsidiaries.

- 40 -

"Financial Model" means the computer model, stored on computer disc(s), and consisting of algorithms as set out on the print-out from such disc(s), each to be initialled at the Second Amendment Signing Date by the Intercreditor Agent and the Company solely for the purposes of identification.

"Financing Costs" means:

- (a) interest, fees, commissions, costs and expenses payable by the Company under the Senior Finance Documents;
- (b) interest, fees, commissions, costs and expenses payable by the Company under the Performance Bond Facility;
- (c) amounts payable by the Company under Clause 10 (*Changes to the Calculation of Interest*), Clause 11 (*Tax Gross Up and Indemnities*), Clause 12 (*Increased Costs*) and Clause 13 (*Currency and Other Indemnities*);
- (d) any other amounts of interest, fees, commissions, discounts, prepayment penalties or premiums and other finance payments payable in respect of Financial Indebtedness permitted to be incurred by a member of the Restricted Group pursuant to paragraphs 2.1(e) and 2.1(f) of Part B of Schedule 5 (*Covenants*);
- (e) any amounts of dividends, disbursements or other payments payable for the purpose of paying any amount with respect to Guaranteed WML Debt (pursuant to paragraph 6.1(v) of Part B of Schedule 5 (*Covenants*));
- (f) net amounts payable by the Company under any Hedging Agreement; and
- (g) any value added or other taxes payable by the Company or any other member of the Restricted Group in respect of paragraphs
   (a) through (f) above and, save to the extent already included in paragraph (c) above, any withholding tax on a party under a Senior
   Finance Document, the Performance Bond Facility or any other agreement relating to the provision of Financial Indebtedness referred to above in respect of which the Company or any other member of the Restricted Group has an obligation to gross up.

"FinCEN" means the Financial Crimes Enforcement Network of the U.S. Department of the Treasury.

"Fiscal Quarter" means any one of the four consecutive three calendar month periods comprised in a Fiscal Year.

"Fiscal Year" means the fiscal year of the Company, the Restricted Group and the Wynn Obligors ending on 31 December of each calendar year. "Fitch" means Fitch Ratings Ltd.

- 41 -

"Floating Charge" means the charge so entitled dated on or about the date of this Agreement between the Company and the Security Agent.

**"Fourth Amendment Effective Date"** has the meaning given in the Common Terms Agreement Fourth Amendment Agreement.

"Fourth Amendment Effective Date Fee Letters" means the Fee Letters referred to in Schedule 1 (*Conditions Precedent*) of the Common Terms Agreement Fourth Amendment Agreement.

"Fourth Amendment Signing Date" means the date of the Common Terms Agreement Fourth Amendment Agreement.

"Fundamental Term" means, in respect of a Senior Finance Document:

- (a) the definitions of Required Lenders and Fundamental Term in Clause 1.1;
- (b) the provisions setting out the date for, or the amount of, or the currency of, any payment of principal (other than Clause 8.2 (Voluntary Prepayment of the Term Loan Facilities) of this Agreement) or interest under a Senior Finance Document or any interest rate hedging payment to a Hedging Counterparty (but excluding, in each case, any amendment bringing forward the date of any such interest rate hedging payment or increasing the amount of any such interest rate hedging payment);
- (c) Clause 2.1 (*Conditions Precedent to the CP Satisfaction Date*) (save in relation to the identity of the opinion providers as set forth in paragraph 14 of Part A of Schedule 2 (*Conditions Precedent*) and paragraph 30 of Part A of Schedule 2 (*Conditions Precedent*));
- (d) the provisions setting out the amount of a Lender's Available Commitment under a Facility (otherwise than by a transfer in accordance with the terms of this Agreement) or the duration of its availability or any additional obligation on a Lender to lend money or provide any other form of credit;
- (e) a term which expressly requires the consent of each Lender or Senior Secured Creditor;
- (f) the provisions relating to the release of any Security (unless permitted under this Agreement or any other Senior Finance Document) or the priority or ranking of such Security or the Secured Obligations (and any other provisions which, if amended, would have the effect of changing the priority or ranking thereof);
- (g) the provisions dealing with the order of distribution on partial payment by the Company or the proceeds of Security;
- (h) paragraph 2.1(e) of Part B of Schedule 5 (*Covenants*) or paragraph 2.1(f) of Part B of Schedule 5 (*Covenants*);
- (i) Clause 25 (Sharing Among the Senior Secured Creditors); and
- (j) Clause 33 (Intercreditor Arrangements).

- 42 -

Notwithstanding the above, unanimity among the Lenders and Hedging Counterparties shall not be required with respect to any changes, additions, deletions, modifications or supplements (herein "changes") comprised in any amendment to the Deed of Appointment and Priority made in accordance with clause 24.1(c) (*Required Consents*) thereof with respect to subparagraphs (a), (e), (f), (g), (h) and (j) above and any Decision related to such changes shall be effected pursuant to subparagraph (a) of the definition of Required Lenders (and as if a Hedging Voting Right Event had occurred and was continuing in relation to each Hedging Counterparty) provided that, in each case, the Senior Secured Creditors' rights, benefits and interests in respect of the First Ranking Liabilities (as defined in the Deed of Appointment and Priority) and the Security, the enforcement thereof and the priority and ranking of their claims in respect thereof and the subordination thereto of all other claims, remain unaffected by any such changes.

"Funded Debt" means, in relation to any Person, all Financial Indebtedness of such Person of the types described in sub-clauses (a) through (g) of the definition of "Financial Indebtedness" in this Clause.

"**Funds**" means any funds that are unconditionally available and have been made available, raised, procured or obtained in a manner that does not breach the terms of this Agreement including such amount of Financial Indebtedness permitted to be created, incurred, assumed or suffered to exist pursuant to paragraph 2.1(e) of Part B of Schedule 5 as, when aggregated with all other amounts of Financial Indebtedness permitted to be created, incurred, assumed or suffered to exist pursuant to paragraph 2.1(e), does not exceed USD500,000,000 or its equivalent.

"GAAP" means, in respect of the Company and other members of the Restricted Group, IFRS and, in respect of any Wynn Obligor, generally accepted accounting principles in the United States of America as in effect from time to time.

"Gaming Concession Consent Agreement" means the Agreement Relating to Security (with the Exclusion of Land Concession and Immovable Property) dated on or about the date of this Agreement between the Government of the Macau SAR, the Company and the Security Agent and the Supplement in respect thereof dated 14 September 2005.

"Governing Documents" means, collectively, as to any Person, the certificate of incorporation, the memorandum and articles of association or bylaws, any shareholders agreement, certificate of formation, limited liability company agreement, partnership agreement or other formation or constituent documents applicable to such Person.

"Governmental Authority" means, as to any Person, the government of the Macau SAR, any other national, state, provincial or local government (whether domestic or foreign), any political subdivision thereof or any other governmental, quasi-governmental, judicial, public or statutory instrumentality, authority, body, agency, bureau or entity, any entity exercising executive, legislative, judicial, regulatory or administrative functions of or pertaining to government, in each case having jurisdiction over such Person, or any arbitrator with authority to bind such Person at law.

- 43 -

"Group" means the Company and each of the Company's Subsidiaries (other than any Excluded Subsidiary) for the time being (including, without limitation, Palo and any other Subsidiary of the Company which becomes an Obligor pursuant to paragraph 27 of Part A of Schedule 5 (*Covenants*)).

"**Guarantee**" means the guarantee formerly entitled the "Wynn Pledgors' Guarantee" dated on or about the date of this Agreement between Wynn Asia 2, Wynn Holdings, Wynn HK, Wynn International, Palo and the Security Agent, as amended, restated, supplemented and novated by, *inter alia*, the Guarantee Third Deed of Amendment and Acknowledgement.

"Guarantee Third Deed of Amendment and Acknowledgement" has the meaning given in the Common Terms Agreement Fourth Amendment Agreement.

"Guaranteed Date of Substantial Completion" means, in respect of the Original Project, the "Guaranteed Date of Original Project Substantial Completion" and, in respect of the Expansion, the "Guaranteed Date of Expansion Project Substantial Completion", each as defined in the Construction Contract.

"Guaranteed WML Debt" means WML Debt in respect of which the Company or any Obligor has incurred, assumed or suffered to exist any Guarantee Obligations in compliance with Clause 2.1(i) of Part B of Schedule 5 (*Covenants*) of this Agreement.

"Guarantee Obligation" means any guarantee, indemnity, letter of credit or other legally binding assurance against loss granted by one Person in respect of any Financial Indebtedness or other liability or obligation of another Person, or any agreement to assume any Financial Indebtedness of any other Person or to supply funds or to invest in any manner whatsoever in such other Person by reason of Financial Indebtedness of such Person; *provided, however*, that the term Guarantee Obligation shall not include endorsements of instruments for deposit or collection in the ordinary course of business. The amount of any Guarantee Obligation in respect of which such Guarantee Obligation is made and (2) the maximum amount for which such guaranteeing Person may be liable pursuant to the terms of the instrument embodying such Guarantee Obligation (unless such primary obligation and the maximum amount for which such guaranteeing Person may be liable are not stated or determinable, in which case the amount of such Guarantee Obligation shall be such guaranteeing Person's maximum reasonably anticipated liability in respect thereof as determined by the Company in good faith).

"Hazardous Substance" means radioactive materials, asbestos and other substances defined as "hazardous" or of a similar nature under any Environmental Law.

"Hedging Agreements" means any agreement entered into by the Company in accordance with the Hedging Arrangements.

- 44 -

"Hedging Arrangements" means the requirements concerning interest rate hedging set out in Schedule 8 (Hedging Arrangements).

"Hedging Counterparties" means a financial institution identified as such in Part D of Schedule 1 (*Hedging Counterparties*) and the parties, other than the Company, to the Hedging Agreements and who have executed a Hedging Counterparty's Deed of Accession.

"Hedging Counterparty's Deed of Accession" means a deed of accession in substantially the form set out in Appendix 1 to Schedule 8 (*Hedging Arrangements*).

"Hedging Voting Right Event" means, in relation to any Hedging Counterparty, the occurrence and continuation of both of the following events:

- (a) the serving of any notice given by the Intercreditor Agent pursuant to sub-clause 19.2.2 of Clause 19.2 (*Remedies following an Event of Default*); and
- (b) a Realised Hedge Loss is not paid when due under the Hedging Agreement to which such Hedging Counterparty is party.

"HIBOR" in relation to any Facility Agreement, has the meaning given in such Facility Agreement.

"HKD" or "HK dollars" denotes the lawful currency of the Hong Kong SAR.

"HKD Debt Service Account" means the account so designated in Schedule 6 (Accounts).

"HKD Debt Service Reserve Account" means the account so designated in Schedule 6 (Accounts).

"HKD Operating Account" means the account so designated in Schedule 6 (Accounts).

"Holding Company" in relation to a Person, means an entity of which that Person is a Subsidiary.

"Hong Kong SAR" means the Hong Kong Special Administrative Region.

"Hotel Project" means the design, development and construction in accordance with the Construction Contract and the Diamond Construction Contract of a luxury hotel resort, retail and entertainment complex on land leased to the Company under the Land Concession Contract and the ownership, operation and maintenance thereof by the Company but shall not include the design, development, construction, ownership, operation or maintenance by the Company of a casino pursuant to the Concession Contract nor the purchase of any associated gaming equipment or utensils.

"Hotel Project Costs" means such Project Costs as relate to the Hotel Project.

"IFRS" means the International Financial Reporting Standards issued by the International Accounting Standards Board or its successor.

- 45 -

"Increased Costs" has the meaning given in Clause 12 (Increased Costs).

"**Information Memorandums**" means the information memorandum dated June 2004, the information memorandum dated June 2005 and the information memorandum dated April 2007 prepared by the Company in relation to the Projects for the purposes of the financing of any or all of the Facilities.

"Initial Advance" means the first Advance made under each of the Facilities.

"Insolvency of a Multiemployer Plan" has the meaning given in section 4245 of ERISA.

"Insurance" means a Direct Insurance or a Reinsurance.

"Insurance Adviser" means, as the case may be:

- (a) JLT Risk Solutions Asia as the insurance adviser acting on behalf of all Senior Secured Creditors pursuant to the engagement letters dated 15 September 2003, 27 April 2005 and 10 May 2007;
- (b) the insurance adviser acting on behalf of all Senior Secured Creditors according to the scope of work and fees agreed by the Senior Secured Creditors and approved by the Company (such approval not to be unreasonably withheld or delayed) before the CP Satisfaction Date; or
- (c) the insurance adviser appointed by the Intercreditor Agent and, unless an Event of Default has occurred and is continuing, approved by the Company (such approval not to be unreasonably withheld or delayed) from time to time after the CP Satisfaction Date to act on behalf of the Senior Secured Creditors as and when required to advise the Senior Secured Creditors in respect of Projects.

"**Insurance Broker's Letter of Undertaking**" means a letter of undertaking in substantially the form set out in Appendix 5 to Schedule 7 (*Insurance*) or in such other form as may be approved by the Intercreditor Agent acting in consultation with the Insurance Adviser, such approval not to be unreasonably withheld.

"Insurance Proceeds" means all amounts and proceeds (including monetary instruments) paid under any insurance policy maintained by the Company or Palo (including, without limitation, any insurance policy required to be maintained by the Company or Palo under any Transaction Document but excluding any public liability, third party liability, workers compensation and legal liability insurances and also excluding any other insurance the proceeds of which are payable to the employees of the Company or Palo) less any costs or expenses incurred by the Company, Palo or their respective agents in collecting such amounts and proceeds.

"**Insurance Requirements**" means all material terms of any insurance policy required pursuant to the Senior Finance Documents (including Schedule 7 (*Insurance*)).

"Insurer" means a Direct Insurer or a Reinsurer.

- 46 -

"**Insurer Notices and Acknowledgements**" means the notices and acknowledgements to be delivered to and executed by each Insurer and Reinsurer in accordance with the Assignment of Insurances and the Assignments of Reinsurances, respectively (including those referred to in paragraphs 2.3.2 and 2.4.2 of Schedule 7 (*Insurance*)).

"**Intellectual Property**" means the collective reference to all rights, priorities and privileges relating to intellectual property, including copyrights, patents, trademarks, service-marks, technology, know-how and processes, formulas, trade secrets or licenses (under which the applicable Person is licensor or licensee) relating to any of the foregoing and all rights to sue at law or in equity for any infringement or other impairment thereof, including the right to receive all proceeds and damages therefrom.

"Intercreditor Agent" means Bank of China Limited, Macau Branch in its capacity as intercreditor agent for the Senior Secured Creditors or its successor appointed in accordance with this Agreement.

"Interest Coverage Ratio" means, in relation to any period, the ratio of EBITDA to Financing Costs for such period.

"Interest Payment Date" means each date on which an Interest Period ends.

"**Interest Period**" means, in relation to any Advance, each period for the calculation of interest in respect thereof ascertained in accordance with Clause 9 (*Interest, Interest Periods and Default Interest*).

"Investment Income" means any interest, dividends or other income arising from or in respect of a Permitted Investment.

"Investment Proceeds" means any net proceeds received upon any disposal, realisation or redemption of a Permitted Investment, but excluding any Investment Income.

"Investments" has the meaning given to it in paragraph 8 of Part B of Schedule 5 (Covenants).

"**IP Agreement**" means the Intellectual Property License Agreement dated 1 January 2003 between the Licensor and the Company, as amended by the First Amendment to Intellectual Property License Agreement dated 1 April 2004, the Second Amendment to Intellectual Property License Agreement dated 7 March 2005, the Third Amendment to Intellectual Property Licence Agreement dated on or about the date of the Common Terms Agreement Second Amendment Agreement and the Amended and Restated Fourth Amendment to Intellectual Property Licence Agreement dated 19 September 2009.

"IP Fees" means the "Licensing Fee" as defined in the IP Agreement.

"ISDA Master Agreement" has the meaning given in Schedule 8 (Hedging Arrangements).

"ISDA Schedule" means the schedule to the ISDA Master Agreement in form and substance reasonably satisfactory to the Intercreditor Agent.

- 47 -

"**Land Concession Contract**" means the land concession contract agreed to by the Company with the Macau SAR on 4 June 2004 which forms an integral part of Dispatch number 81/2004.

"Land Concession Consent Agreement" means the Agreement relating to Security under the Land Concession Contract dated on or about the date of this Agreement between the Government of the Macau SAR, the Company and the Security Agent and the Supplement in respect thereof dated 14 September 2005.

"Land Security Assignment" means the assignment so entitled dated on or about the date of this Agreement between the Company and the Security Agent.

"**Legal Requirements**" means all laws, statutes, orders, decrees, injunctions, licenses, permits, approvals, agreements and regulations of any Governmental Authority having jurisdiction over the matter in question.

"Lender" means a Term Facility Lender, a Revolving Credit Facility Lender or an Additional Lender.

"Lending Group" means the Term Lending Group, each Revolving Lending Group and each Additional Lending Group.

"Leverage Ratio" means, in relation to any period, the ratio of Total Debt on the last day of such period to EBITDA for such period.

"LIBOR", in relation to any Facility Agreement, has the meaning given in such Facility Agreement.

"Licensor" has the meaning given in the IP Agreement.

"Lien" means, with respect to any Property, any mortgage, lien, pledge, charge, security interest or encumbrance of any kind in respect of such Property, whether or not filed, recorded or otherwise perfected under applicable law (including any conditional sale or other title retention agreement, any option or other agreement to sell or give a security interest in and any filing of or agreement to give any financing statement under the UCC (or equivalent statutes of any jurisdiction)).

"Line Item" means each of the following line item categories:

(a) Hard Construction Costs

- (i) Construction and Building
- (ii) Interior Furnishings and Equipment
- (iii) Design Fees
- (iv) Contractor's Fees
- (v) Construction Contingency
- (vi) Owner's Contingency

- 48 -

| (b) | Pre-Opening Costs  |                           |  |
|-----|--|---------------------------|--|
|     | (i)  | Payroll                   |  |
|     | (ii)   | Direct Expenses           |  |
|     | (iii)  | Corporate Expenses        |  |
| (c) | Owner Furniture, Fittings and Equipmen   |                           |  |
|     | (i)  | Casino                    |  |
|     | (ii)   | Hotel                     |  |
|     | (iii)  | Food and Beverage         |  |
|     | (iv)   | Others                    |  |
| (d) | Land Cost<br>Pre-Opening Working Capital<br>Capitalised Interest and Commitment Fees |                           |  |
| (e) |  |                           |  |
| (f) |  |                           |  |
|     | (i)  | Revolving Credit Facility |  |
|     | (ii)   | Performance Bond Facility |  |
|     |  |                           |  |

- (g) Tax, Fees and Expenses
- (h) Contingency.

"Liquidated Damages" means any liquidated damages paid pursuant to any obligation, default or breach under any Project Document to which a member of the Restricted Group is party (other than any Termination Proceeds), in each case net of costs and expenses incurred by such member of the Restricted Group or its agent pursuant to arm's length transactions in connection with adjustment or settlement thereof and taxes paid with respect thereto.

"Livrança Covering Letter" means the letter from the Company to the Security Agent dated on or about the date of this Agreement in relation to the Livranças.

"Livrança Updated Covering Letter" means the letter from the Company to the Security Agent dated on or about the Fourth Amendment Effective Date in relation to the Livranças and the Livrança Covering Letter, duly acknowledged by Palo.

"Livranças" means the promissory notes dated on or about the date of this Agreement issued by the Company and endorsed and payable to the Security Agent.

"London Business Day" means a day (other than a Saturday or Sunday) on which banks are open for general business in London.

- 49 -

"Loss Proceeds" means all amounts and proceeds (including monetary instruments) in respect of any Event of Loss, including proceeds of any insurance policy required to be maintained by the Company or any other member of the Restricted Group under this Agreement, less any costs and expenses incurred by the Company or such member of the Restricted Group or its agents in collecting such amounts and proceeds.

"Macau Gaming Laws" means Law No. 16/2001 and Administrative Regulation No. 26/2001, as amended from time to time, and other laws promulgated by any Governmental Authority of the Macau SAR and applying to gaming operations in the Macau SAR.

"Macau SAR" means the Macau Special Administrative Region.

"Major Project Document" means any of:

- (a) the Concession Contract;
- (b) the Land Concession Contract;
- (c) the Construction Contract;
- (d) the Diamond Construction Contract;
- (e) the Prime Contractor's Completion Guarantee;
- (f) the Prime Contractor's Performance Bond;
- (g) the Project Administration Services Agreement;
- (h) the IP Agreement;
- (i) the Performance Bond Facility Agreement;
- (j) the Concession Contract Performance Bond;
- (k) any Resort Management Agreement;
- any other Project Document with respect to the Projects (but not the Cotai Project) with a total contract price payable (or expected aggregate amount to be paid in the case of "cost plus" contracts) by any member of the Restricted Group or which may otherwise involve liabilities, actual or contingent, of any member of the Restricted Group in each case in an amount in excess of USD25,000,000 or its equivalent; and
- (m) the Cotai Land Concession Contract.

"Major Project Participants" means:

- (a) each Obligor;
- (b) the Macau SAR;

- 50 -

- (c) the Prime Contractor (for so long as it has any actual or contingent liabilities under the Construction Contract);
- (d) Leighton Holdings Limited and China Overseas Holdings Limited (for so long as, in each case, it has any actual or contingent liabilities under the Prime Contractor's Completion Guarantee);
- (e) the PASA Agent (for so long as it has any actual or contingent liabilities under the PASA); and
- (f) each other Person who is party to a Major Project Document (other than any Resort Management Agreement).

"Market Disruption Event" has the meaning given in Clause 10.2 (Market disruption).

"Material Adverse Effect" means a material adverse condition or material adverse change in or affecting:

- (a) the business, assets, liabilities, property, condition (financial or otherwise), results of operations, prospects, value or management of the Company or the Company, the Restricted Group and the Wynn Obligors, taken as a whole, or that calls into question in any material respect the Projections or any of the material assumptions on which the Projections were prepared;
- (b) the Projects taken as a whole;
- (c) the ability of the Company to achieve Substantial Completion in respect of the Original Project on or prior to the Guaranteed Date of Substantial Completion thereof;
- (d) the validity or enforceability of any Senior Finance Document (other than any Palo Security Document *provided that* such material adverse condition or material adverse change in or affecting the validity or enforceability of that Palo Security Document arises as a result of the matters contemplated by Clause 18.3.4(a) (*Permitted Cotai Reorganisation; Release of Palo Security*) of this Agreement following a Permitted Cotai Reorganisation or the termination or recission of the Cotai Land Concession Contract, where such termination or recission could not reasonably be expected to have an adverse effect on the ability of the Company or any other Obligor to perform its payment obligations under the Senior Finance Documents);
- (e) the validity, enforceability or priority of any of the Liens purported to be created under any of the Security Documents (other than any of the Palo Security Documents, *provided that* such material adverse condition or material adverse change in or affecting the validity, enforceability or priority of any of the Liens purported to be created under that Palo Security Document arises as a result of the matters contemplated by Clause 18.3.4(a) (*Permitted Cotai Reorganisation; Release of Palo Security*) of this Agreement following a Permitted Cotai Reorganisation or the termination or recission of the Cotai Land Concession Contract where such termination or recission could not

- 51 -

reasonably be expected to have an adverse effect on the ability of the Company or any other Obligor to perform its payment obligations under the Senior Finance Documents);

- (f) the rights and remedies of any Secured Creditor under any Senior Finance Document (other than any Palo Security Document provided that such material adverse condition or material adverse change in or affecting the rights or remedies of any Secured Creditor under that Palo Security Document arises as a result of the matters contemplated by Clause 18.3.4(a) (*Permitted Cotai Reorganisation; Release of Palo Security*) of this Agreement following a Permitted Cotai Reorganisation or the termination or recission of the Cotai Land Concession Contract where such termination or recission could not reasonably be expected to have an adverse effect on the ability of the Company or any other Obligor to perform its payment obligations under the Senior Finance Documents); or
- (g) the Cotai Project (but only if such material adverse condition or material adverse change could reasonably be expected to result in a material adverse condition or material adverse change in or affecting any matter or thing set out in any of the preceding paragraphs).

"Money Laundering Laws" has the meaning given to that term in paragraph 39 of Schedule 4 (Representations and Warranties).

"Monthly Construction Period Report" has the meaning given in paragraph 2(b) of Part A of Schedule 5 (Covenants).

"**Monthly Construction Progress Report**" means each of the monthly reports prepared by the Prime Contractor under section 7.5 (*Notices under the Construction Cotnract*) of the Construction Contract.

"Moody's" means Moody's Investors Service, Inc or its successor.

"MOP Operating Account" means the account so designated in Schedule 6 (Accounts).

"Mortgage" means the mortgage so entitled dated on or about the date of this Agreement between the Company and the Security Agent.

"Multiemployer Plan" means a Plan that is a multiemployer plan as defined in section 3(37) or 4001(a)(3) of ERISA.

- "Net Cash Proceeds" means:
- (a) in connection with any Asset Sale, the proceeds thereof in the form of cash (including any such proceeds received by way of deferred payment of principal pursuant to a note or instalment receivable or purchase price adjustment receivable or otherwise, but only as and when received) of such Asset Sale, net of attorneys' fees, accountants' fees, investment banking fees, amounts required to be applied to the repayment of Financial Indebtedness secured by a Lien expressly permitted hereunder on any asset which is the subject of such Asset Sale (other than any Lien pursuant to a Security

Document) and other fees and expenses, in each case, to the extent actually incurred in connection with such Asset Sale and net of taxes paid or reasonably estimated to be payable as a result thereof (after taking into account any tax credits or deductions and any tax sharing arrangements, in each case reducing the amount of taxes so paid or estimated to be payable); and

(b) in connection with any issuance or sale of debt securities or instruments or the incurrence of loans, the cash proceeds received from such issuance or incurrence, net of attorneys' fees, investment banking fees, accountants' fees, underwriting discounts and commissions and other fees and expenses, in each case, to the extent actually incurred by the Company or any other member of the Restricted Group in connection therewith.

"**Net Income**" means, in relation to any period, the net income (or loss) of the Restricted Group for such period, determined on a consolidated basis which includes members of the Restricted Group only (and which, for the avoidance of doubt, does not take account of any amount to the extent it (i) would otherwise have the effect of increasing Net Income and (ii) is in any way derived from or attributable or otherwise related to or connected with an Excluded Project, an Excluded Subsidiary, a Resort Management Agreement, the grant of any Subconcession or any interest, right or claim in respect thereof) and otherwise in accordance with GAAP without taking account of any amount of cash or cash proceeds paid or received in respect of the grant or entry into any Subconcession and before any reduction in respect of preferred equity dividends.

"Notice to Proceed" has the meaning given to such term in the Construction Contract.

"Notional Amount", in relation to a Hedging Agreement, has the meaning referred to in paragraph 9 of Schedule 8 (Hedging Arrangements).

"Novation Certificate" means a novation certificate in substantially the form set out in Part B of Schedule 11 (Transfers and Accession).

- "Obligations" means:
- (a) all loans, advances, debts, liabilities and obligations howsoever arising, owed by the Company or any other Obligor under the Senior Finance Documents to any Senior Secured Creditors of every kind and description (whether or not evidenced by any note or instrument and whether or not for the payment of money), direct or indirect, absolute or contingent, due or to become due, now existing or hereafter arising, pursuant to the terms of the Senior Finance Documents or any of the other Transaction Documents, including all interest (including interest accruing after the maturity of any Advance and interest accruing after the filing of any petition in bankruptcy, or the commencement of any insolvency, reorganization or like proceeding, relating to any Obligor, whether or not a claim for post-filing or post-petition interest is allowed in such proceeding), fees, premiums, if any, and any charges, expenses, attorneys' fees and accountants' fees, in each case chargeable to any Obligor in connection with its dealings with such Obligor and payable by such Obligor thereunder;



- (b) any and all sums advanced by any Agent or any Lender in order to preserve the Project Security or preserve any Senior Secured Creditor's security interest in the Project Security as permitted by the Senior Finance Documents; and
- (c) in the event of any proceeding for the collection or enforcement of the Obligations after issuance of an Enforcement Notice, the expenses of retaking, holding, preparing for sale or lease, selling or otherwise disposing of or realising on the Project Security, or of any exercise by any Senior Secured Creditor of its rights under the Security Documents, together with attorneys' fees and court costs, in each case as permitted by the Senior Finance Documents.

"**Obligors**" means the Company, the Wynn Obligors, Palo, the other members of the Restricted Group (in each case, on and from the date on which the requirements of paragraph 27.1 of Part A of Schedule 5 have been satisfied in respect of such other member of the Restricted Group) and any party to a Senior Finance Document referred to in paragraph (f) of the definition of Senior Finance Document (other than, in each case, Wynn Asia, Wynn Resorts, Wynn Holdings, a Secured Party or a Person who is solely party to an acknowledgement of Security).

"Opening Conditions" means, collectively, the following in respect of the Original Project:

(a)

- the Intercreditor Agent shall have received from the Company a certificate, substantially in the form set out in Part A of Schedule 19 (*Forms of Opening Conditions Certificates*), pursuant to which the Company certifies that:
  - (i) furnishings, fixtures and equipment necessary to use and occupy the various portions of each Project comprised in the Original Project for their intended uses shall have been installed and shall be operational;
  - (ii) the Project Certificates of Occupancy for the Original Project shall have been issued, each area of the Original Project in which any operation of casino games of chance or other forms of gaming will be carried out shall have been classified as a casino or gaming zone in accordance with Article 9 of the Concession Contract and (other than any Permit made or issued by or with a Governmental Authority the failure of which to obtain could not reasonably be expected to affect the operations of the Original Project in any material respect) each other Permit made or issued by or with a Governmental Authority required under applicable Legal Requirements to be obtained prior to the Opening Date for the Original Project shall have been obtained;
  - (iii) the Original Project (other than the premises to be occupied by individual retail and restaurant tenants) shall be fully open for business to the general public and at least, notwithstanding the foregoing, in the case of the Original Project, 80% of each of the projected 380 slot machines and 200 gaming tables shall be operating, 80% of all rooms shall be ready for occupancy, at least 80% of restaurant outlets shall be open for business and at least 80% of the floor space comprised in the Original Project and designated in the Plans and Specifications for

- 54 -

retail tenants has been occupied (save for facilities which by their nature are not open to the general public in the ordinary course of business but are operating);

- (iv) any remaining work (including, in the case of the Original Project, any work on the Expansion) shall be such that it will not materially affect the operation of the Original Project;
- (v) the failure to complete the remaining work would not materially affect the operation of the Original Project; and
- (vi) the Company shall have available a fully trained staff to operate the Original Project; and
- (b) the Intercreditor Agent shall have received from the Technical Adviser a certificate, substantially in the form set out in Part B of Schedule 19 (*Forms of Opening Conditions Certificates*) in respect of the Original Project.

"**Opening Date**" means, in relation to the Original Project, the date on which all the Project Certificates of Occupancy required for the Original Project have been issued.

"**Operation Period Insurances**" means the insurances listed in Appendix 2 to Schedule 7 (*Insurance*) and effected in accordance with the terms of Schedule 7 (*Insurance*).

"**Operatives**" means a shareholder, officer, employee, servant, controlling Person, executive, director, agent, authorised representative or Affiliate of any of the Obligors.

"**Original Project**" means that part of the Projects the construction of which was contemplated by the original Construction Contract entered into by the Company and the Prime Contractor and dated 10 May 2004 as amended by the change order and amendment thereto dated 14 September 2004.

"PASA Agent" means Wynn Design & Development, LLC.

"PASA Direct Agreement" means the agreement so entitled between the PASA Agent, the Company and the Security Agent in the Agreed Form.

"**Palo**" means Palo Real Estate Company Limited a company with limited liability incorporated in the Macau SAR with registration number 27319 SO.

"Palo Assignment of Insurances" means the Palo Assignment of Onshore Insurance Policies to be entered into between between Palo and the Security Agent.

"Palo Floating Charge" means the charge so entitled to be entered into between between Palo and the Security Agent.

"**Palo Pledge over Onshore Accounts**" means the pledge so entitled dated on or about the Fourth Amendment Effective Date and made between Palo and the Security Agent.

- 55 -

"**Palo Share Pledge**" means the Share Pledge to be entered into between the Company, Wynn International, Wynn HK, Palo and the Security Agent. "**Palo Security Documents**" means (in each case, from the date thereof):

- (a) the Cotai Mortgage;
- (b) the Cotai Power of Attorney;
- (c) the Palo Assignment of Insurances;
- (d) the Palo Floating Charge;
- (e) the Palo Pledge over Onshore Accounts; and
- (f) the Palo Share Pledge.
- "Patacas" or "MOP" denotes the lawful currency of the Macau SAR.

"**Payment and Performance Bond**" means any payment and performance bond delivered under any Major Project Document in favour of the Company and supporting the Contractor's obligations under any such Major Project Document (including the Prime Contractor's Completion Guarantees and the Prime Contractor's Performance Bonds).

"PBGC" means the Pension Benefit Guaranty Corporation established pursuant to Subtitle A of Title IV of ERISA (or any successor).

"**Performance Bond Facility**" means a facility extended to the Company by the Performance Bond Provider in accordance with the terms of the Performance Bond Facility Agreement for the issuance of the Concession Contract Performance Bond and subordinated to amounts owed to the Senior Secured Creditors under the Senior Finance Documents in accordance with the Deed of Appointment and Priority.

"**Performance Bond Facility Agreement**" means the agreement dated as of 14 September 2004 between the Performance Bond Provider and the Company.

#### "Performance Bond Provider" means Banco Nacional Ultramarino, S.A. or such other Person as may be acceptable to the Intercreditor Agent.

"**Permits**" means all approvals, licences, consents, permits, authorisations, registrations and filings, necessary in connection with the execution, delivery or performance, admission into evidence or enforcement of the Transaction Documents and all material approvals, licences, consents, permits, authorisations, registrations and filings required for the development, construction, ownership or operation of the Projects as contemplated under the Transaction Documents, including those listed in Schedule 12 (*Permits*).

"Permitted Businesses" means the Projects and the Cotai Project, including:

(a) in the case of the Company only, the operation of casino games of chance or other forms of gaming in one or more locations in the Macau SAR in

- 56 -

connection with the Projects, the Cotai Project or any Excluded Project, in each case as permitted under the Concession Contract and, in the case of any Excluded Project, as contemplated by the Resort Management Agreement entered into by the Company in respect thereof;

- (b) the development, construction, financing, ownership and operation of hotel resorts and casinos as contemplated in the Concession Contract; and
- (c) food and beverage, spa, entertainment production, convention, retail, residential, foreign exchange, transportation and outsourcing of in-house facilities, businesses or other activities which are necessary for, incident to, arising out of, supportive of or connected to the development, construction, financing, ownership or operation of such hotel resorts and casinos,

and, in the case of the Company and other members of the Restricted Group, the holding of shares and other interests permitted hereunder in Subsidiary Obligors and Excluded Subsidiaries.

"**Permitted Cotai Reorganisation**" means any transaction or series of transactions in which Palo Disposes to the Company Palo's interest in the Cotai Site (whether by assignment, novation, transfer or replacement of the Cotai Land Concession Contract or otherwise) and Palo's ownership of, and interest in, all of Palo's other assets (including, without limitation, assets relating to the Cotai Site and/or the Cotai Project) *provided* that, in any such case:

- (i) such Disposal, does not conflict with the Cotai Land Concession Contract, the Concession Contract and applicable Legal Requirements;
- (ii) the Company gives prompt written notice to the Intercreditor Agent of the occurrence of such Disposal (and such written notice includes a confirmation that both the Company and Palo are then (and will, following the completion of such Disposal, be) Solvent);
- (iii) the entire Cotai Site shall remain at all times part of the Project Security and there shall be no adverse effect on the validity or enforceability of, or the effectiveness or ranking of any Security of the Finance Parties as a result of such Disposal; and
- (iv) (without prejudice to paragraph (iii)) the Company complies with paragraph 15 of Part A of Schedule 5 (*Covenants*) with respect to its acquisition of any Disposed Property.

"Permitted Financial Indebtedness" has the meaning given in paragraph 2.1 of Part B of Schedule 5 (Covenants).

### "**Permitted Investments**" means the following:

(a) securities issued, or directly and fully guaranteed or insured, by the United States government or any agency or instrumentality of the United States government (as long as the full faith and credit of the United States is pledged in support of those securities) having maturities of not more than nine months from the date of acquisition;

- 57 -

- (b) securities issued, or directly and fully guaranteed or insured, by the government of the Hong Kong SAR or any agency or instrumentality of the government of the Hong Kong SAR (as long as the full faith and credit of the Hong Kong SAR is pledged in support of those securities) having maturities of not more than nine months from the date of acquisition;
- (c) interest-bearing demand or time deposits (which may be represented by certificates of deposit) issued by banks having general obligations rated (on the date of acquisition thereof) at least "A" or the equivalent by S&P or Moody's or, if not so rated, secured at all times, in the manner and to the extent provided by law, by collateral security in paragraph (a) or (b) above, of a market value of no less than the amount of monies so invested;
- (d) repurchase obligations with a term of not more than seven days for underlying securities of the types described in paragraphs (a),
   (b) and (c) above entered into with any financial institution meeting the qualifications specified in paragraph (c) above;
- (e) commercial paper having a rating of A-1, P-1 or F1 from one or more of S&P, Moody's or Fitch respectively and in each case maturing within nine months after the date of acquisition;
- (f) corporate bonds having a rating of at least BBB-, Baa3 or BBB from two or more of S&P, Moody's or Fitch respectively, with an aggregate principal amount not exceeding (when aggregated with the principal amount of any Dim Sum Bonds then held by any Obligor pursuant to paragraph (h) below) USD200,000,000 or its equivalent at any time;
- (g) money market or mutual funds which are rated at least AAA by S&P or Aaa by Moody's or have an equivalent rating from another internationally recognised rating agency; and
- (h) corporate bonds denominated in Renminbi (commonly known as "offshore RMB" or "Dim Sum" bonds) which do not satisfy the ratings requirements set out in paragraph (f) above ("**Dim Sum Bonds**") with an aggregate principal amount not exceeding USD75,000,000 or its equivalent at any time.

"Permitted Liens" means the collective reference to:

- (a) in the case of any Property other than any Pledged Stock, Liens permitted by paragraph 3 of Part B of Schedule 5 (*Covenants*) hereto (but only of the priority and to the extent of coverage expressly set forth in paragraph 3 of Part B of Schedule 5 (*Covenants*) hereto); and
- (b) in the case of any Property consisting of Pledged Stock, non-consensual Liens permitted by paragraph 3 of Part B of Schedule 5 (*Covenants*) hereto to the extent arising by operation of law.

"**Permitted Loan Repurchase**" means any purchase of Advances by the Company made in accordance with Clause 21.10 (*Permitted Loan Repurchases*); *provided* that (i) the aggregate amount paid by the Company for such purchases (excluding

- 58 -

payments of accrued interest) during the period commencing on the Fourth Amendment Effective Date and ending on (and including) the Release Date shall not exceed the Permitted Loan Repurchase Amount and (ii) each such purchase is consummated pursuant to a written offer made to all Term Facility Lenders (if the Company proposes to purchase Advances under the Term Facility), all Revolving Credit Facility Lenders (other than Additional Lenders) (if the Company proposes to purchase Advances under a Revolving Credit Facility that is not an Additional Lender Facility), all Additional Lenders party to the applicable Additional Lender Facility (if the Company proposes to purchase Advances under a Revolving Credit for the applicable Lenders; *provided*, that the Company shall not be required to comply with clause (ii) of this definition with respect to purchases of Advances having an aggregate principal amount together with all other purchases under this proviso not to exceed USD50,000,000 that are consummated on or after the Fourth Amendment Effective Date and on or before the one year anniversary of the Fourth Amendment Effective Date, in each case, so long as such purchases are made at a discount to the aggregate principal amount of the Advances purchased.

#### "Permitted Loan Repurchase Amount" means USD150,000,000.

"**Person**" means any natural person, corporation, partnership, firm, association, Governmental Authority or any other entity whether acting in an individual, fiduciary or other capacity.

"**Plan**" means, at a particular time, any employee benefit plan that is subject to the requirements of section 412 of the Code or that is a Single Employer Plan and which any Obligor or any Commonly Controlled Entity maintains, administers, contributes to or is required to contribute to or under which any Obligor or any Commonly Controlled Entity could reasonably be expected to incur any liability.

"Plans and Specifications" means the plans, specifications, design documents, schematic drawings and related items for the design, architecture and construction of the Projects provided to the Technical Adviser in accordance with paragraph 28 of Part A of Schedule 2 (*Conditions Precedent*) as may be amended in accordance with any variation permitted pursuant to paragraph 15 of Part B of Schedule 5 (*Covenants*).

"Pledge over Gaming Equipment and Utensils" means the pledge so entitled dated on or about the date of this Agreement between the Company and the Security Agent.

"Pledge over Onshore Accounts" means the pledge so entitled dated on or about the date of this Agreement between the Company and the Security Agent.

"**Pledged Stock**" means any Property expressed to be subject to any Lien created or purported to be created under all and any of the Company Share Pledge, the Palo Share Pledge, the Wynn International Share Charge and the Wynn HK Share Charge.

"Post-Amendment Global Transfer Agreement" has the meaning given in the Common Terms Agreement Second Amendment Agreement.

#### - 59 -

"**Power of Attorney**" means the irrevocable power of attorney dated on or about the Fourth Amendment Effective Date granted by the Company in favour of the Security Agent in connection with the Mortgage.

"Pre-Amendment Global Transfer Agreement" has the meaning given in the Common Terms Agreement Second Amendment Agreement.

"**Prime Contractor**" means Leighton Contractors (Asia) Limited, China Construction Engineering (Macau) Company Limited and China State Construction Engineering (Hong Kong) Limited in their capacity as contractors under the Construction Contract.

## "Prime Contractor's Completion Guarantee" means:

- (a) the Amended and Restated Parent Completion Guarantee dated 14 September 2005 given by Leighton Holdings Limited and China Overseas Holdings Limited to the Company in support of the Prime Contractor's obligations under the Construction Contract; and
- (b) the completion guarantee, if any, given by Leighton Holdings Limited and China Overseas Holdings Limited to the Company in support of the Prime Contractor's obligations under the Diamond Construction Contract.

#### "Prime Contractor's Performance Bond" means:

- (a) the two Payment and Performance Bonds dated 14 September 2005 in an aggregate amount of not less than USD45,722,399 and delivered to the Company in support of the Prime Contractor's obligations under the Construction Contract; and
- (b) the Payment and Performance Bond, if any, delivered to the Company in support of the Prime Contractor's obligations under the Diamond Construction Contract.

"Proceedings" has the meaning given to it in paragraph 12(i) of Part A of Schedule 5 (*Covenants*) hereto.

"Project" means:

- (a) the Hotel Project; or
- (b) the design, development and construction in accordance with the Concession Contract, the Construction Contract and the Diamond Construction Contract of a casino on land leased to the Company under the Land Concession Contract, the ownership, operation and maintenance thereof by the Company and the purchase of associated gaming equipment and utensils,

and "Projects" means both of them.

"**Project Administration Services Agreement**" (or "**PASA**") means the Amended and Restated Project Administration Services Agreement between the PASA Agent and the Company dated 14 September 2005.

- 60 -

"**Project Certificates of Occupancy**" means the Licenças de Ocupação issued by the Macau SAR pursuant to applicable Legal Requirements for the Original Project.

"Project Costs" means all costs incurred, or to be incurred, in respect of the Projects, comprising, without double counting:

- (a) all costs incurred under the Construction Contract and the Diamond Construction Contract;
- (b) interest, commissions or other Financing Costs payable under the Senior Finance Documents prior to the Diamond Opening Date;
- (c) commitment commission payable under the Performance Bond Facility prior to the Diamond Opening Date;
- (d) guarantee fees, legal fees and expenses, financial advisory fees and expenses, technical fees and expenses (including fees and expenses of the Technical Adviser and the Insurance Advisor), commitment fees, management fees and corporate overhead agency fees (including fees and expenses of the Agents), interest, taxes (including value added tax) and other out-of-pocket expenses payable by the Company or any other member of the Restricted Group under any documents related to the financing and administration of the Projects prior to the Diamond Opening Date;
- (e) the costs of acquiring Permits for the Projects prior to the Diamond Opening Date;
- (f) costs incurred in settling insurance claims in connection with Events of Loss and collecting Loss Proceeds at any time prior to the Diamond Opening Date;
- (g) working capital costs incurred prior to the Diamond Opening Date; and
- (h) cash to collateralise commercial letters of credit to the extent that payment of any such cash amount to the vendor or materialman who is the beneficiary of such letter of credit would have constituted a "Project Cost".

#### "Project Documents" means:

- (a) the Concession Contract, the Land Concession Contract, the Cotai Land Concession Contract, the Construction Contract, the Prime Contractor's Completion Guarantee, the Prime Contractor's Performance Bond, the Project Administration Services Agreement, the IP Agreement and each Payment and Performance Bond issued to the Company or any other member of the Restricted Group; and
- (b) any other document or agreement entered into by the Company or any other member of the Restricted Group (other than the Senior Finance Documents) prior to the Fourth Amendment Signing Date, excluding any such document or agreement relating to the Cotai Project but including the Cotai Land Concession Contract, each as the same may be amended from time to time in accordance with the terms and conditions of this Agreement and thereof.

- 61 -

"Project Security" means any Property expressed to be subject to any Lien created or purported to be created under any of the Security Documents.

"**Project Schedule**" means, with respect to the Projects, the schedule referred to in paragraph 25 of Part A of Schedule 2 (*Conditions Precedent*). "**Project Works**" means:

- (a) the design, development and construction of the Projects and any other works contemplated by the Construction Contract; or
- (b) the design, development and construction of the Cotai Project,

as the case may be.

"Projections" has the meaning given in paragraph 2(c) of Part A of Schedule 5 (Covenants).

"**Property**" means any property or assets including without limitation any right or interest (whether legal or equitable) in or to property or assets of any kind whatsoever, whether real, personal or mixed and whether tangible or intangible, including, without limitation, Capital Stock.

"Quarterly Date" means:

- (a) with respect to the first Quarterly Date, the last day of the first full Fiscal Quarter falling after the Second Amendment Signing Date; and
- (b) with respect to each subsequent Quarterly Date, the last day of the next succeeding Fiscal Quarter.

"**Quotation Day**" means, in relation to any period for which an interest rate is to be determined, two London Business Days before the first day of that period or, in the case of any interest rate determined using HIBOR, the first day of that period.

"Realised Hedge Loss" has the meaning given in paragraph 6 of Schedule 8 (Hedging Arrangements).

"Recovering Senior Secured Creditor" has the meaning given in Clause 25.1 (Payments to Senior Secured Creditors).

"Reference Banks", in relation to any Facility Agreement, has the meaning given in such Facility Agreement.

"**Reinsurance**" means any contract or policy of reinsurance from time to time required by paragraph 1.2 or 1.4 of Schedule 7 (*Insurance*) to be taken out or effected in respect of any Direct Insurance.

- 62 -

"**Reinsurance Broker's Letter of Undertaking**" means a letter of undertaking in substantially the form set out in Appendix 6 to Schedule 7 (*Insurance*) or in such other form as may be approved by the Intercreditor Agent acting in consultation with the Insurance Adviser, such approval not to be unreasonably withheld.

"**Reinsurer**" means an international reinsurer of good standing and responsibility with whom a Reinsurance is placed from time to time in accordance with Schedule 7 (*Insurance*).

"Release Date" means the date on which the Intercreditor Agent notifies the Company that the following conditions have been satisfied:

- (a) receipt by the Intercreditor Agent of confirmation from each Agent and Hedging Counterparty that all liabilities to its Lending Group or, as the case may be, to it have been discharged in full under the Senior Finance Documents; and
- (b) unless the discharge in paragraph (a) above is effected using the proceeds of Financial Indebtedness incurred pursuant to paragraph 2.1(d) of Part B of Schedule 5 (*Covenants*), receipt by the Intercreditor Agent of a legal opinion from the Lenders' Macanese counsel in a form satisfactory to the Intercreditor Agent on the basis of which the Intercreditor Agent is able to determine that the risk of the discharge of the Financial Indebtedness owed by the Company to the Senior Secured Creditors in accordance with the Senior Finance Documents not being recognised or deemed to be discharged under the insolvency laws of the Macau SAR is acceptable to the Intercreditor Agent.

"**Relevant Party**" means a Person (other than a Substantial Shareholder or, in respect only of his direct shareholding in the Company as at the Signing Date, Mr Wong Chi Seng) who, legally or beneficially, directly or indirectly, owns or holds any of the outstanding Capital Stock of the Company other than directly or indirectly through Wynn International.

"**Relevant Party's Undertaking**" means the undertakings set out in paragraph 3 of schedule 4 of the term sheet referred to in the underwriting agreement between Wynn Resorts, the Company, Banc of America Securities Asia Limited, Deutsche Bank AG, Hong Kong Branch and Société Générale Asia Limited dated 23 June 2005.

"**Reorganization**" means, with respect to any Multiemployer Plan, the condition that such plan is in reorganization within the meaning of section 4241 of ERISA.

"Repair Plan" has the meaning given in paragraph 5 of Schedule 9 (Mandatory Prepayment).

"Repayment Date" means:

- (a) in relation to the Term Facility:
  - (i) the fifth anniversary of the Fourth Amendment Effective Date; and
  - (ii) the sixth anniversary of the Fourth Amendment Effective Date; and
- (b) in relation to an Additional Lender Facility providing for a term loan, the dates on which repayments of principal are scheduled to be made, as set forth in the applicable Additional Lender Facility Agreement.

- 63 -

"**Reportable Event**" means any of the events set forth in section 4043(c) of ERISA, other than those events as to which the thirty day notice period is waived under regulations under ERISA.

"Request" has the meaning given to that term in the Common Terms Agreement Fourth Amendment Agreement.

"Required Filings" has the meaning given to that term in the Common Terms Agreement Fourth Amendment Agreement.

"Required Lenders" means:

- (a) in relation to any Decision other than on a Fundamental Term, Lenders (and, after the occurrence and continuation of a Hedging Voting Right Event in relation to any Hedging Counterparty, that Hedging Counterparty) who:
  - (i) have notified the Intercreditor Agent of their vote in respect of such Decision within the time required by the Intercreditor Agent pursuant to this Agreement; and
  - hold, in aggregate, more than 50% of the Voting Entitlements of all such Senior Secured Creditors who have so notified their votes; and
- (b) in relation to a Decision on a Fundamental Term, all Lenders (and, after the occurrence of a Hedging Voting Rights Event in relation to a Hedging Counterparty that is continuing, that Hedging Counterparty).

"**Resort Management Agreement**" means (a) any agreement entered into by the Company (that does not conflict with the Concession Contract and all other applicable Legal Requirements) with an Excluded Subsidiary or other third party for the management or operation by the Company in compliance with the Concession Contract and all other applicable Legal Requirements of an Excluded Project or any part thereof (other than an Excluded Project or any part thereof located on the Cotai Site) and/or (b) any Cotai Resort Management Agreement.

"**Responsible Officer**" means, as to any Person in respect of any matter, the chief executive officer, president, managing director, chief financial officer, chief operating officer or treasurer of such Person duly authorised in respect of such matter, but in any event, with respect to financial matters, the chief financial officer or treasurer of such Person or the chief financial officer of WML or Wynn Resorts. Unless otherwise qualified, all references to a "Responsible Officer" shall refer to a Responsible Officer of the Company.

"Restricted Group" means the Group other than any Excluded Subsidiary.

- 64 -

### "**Restricted Party**" means any person:

- (a) listed in the annex to, or otherwise subject to the provisions of, any sanctions, prohibitions or requirements imposed by any U.S. executive order issued pursuant to an Anti-Terrorism Law;
- (b) named as a "Specially Designated National and Blocked Person" ("**SDN**") on the most current list (the "**SDN List**") published by OFAC at its official website or any replacement website or other replacement official publication of such list; or
- (c) in which an entity on the SDN List has 50% or greater ownership interest or that is otherwise controlled by an SDN.

"Restricted Payments" has the meaning given to it in paragraph 6 of Part B of Schedule 5 (Covenants) hereto.

"**Retainage Amounts**" means, at any given time, amounts which have accrued and are owing to a Contractor under the terms of a Project Document (other than the IP Agreement) for work or services already provided but which at such time (and in accordance with the terms of such Project Document) are being withheld from payment to the Contractor, until certain subsequent events (e.g. completion benchmarks) have been achieved under the Project Document.

"Revolving Credit Facilities" means each of the revolving loan facilities to be provided under the Revolving Credit Facility Agreements.

"**Revolving Credit Facility Agent**" means Bank of China Limited, Macau Branch as facility agent for the Revolving Credit Facility Lenders. "**Revolving Credit Facility Agreements**" means:

- (a) the agreement so entitled between the Company, the Revolving Credit Facility Agent and the Revolving Credit Facility Lenders; and
- (b) each Additional Lender Facility Agreement providing for a revolving credit facility.

"Revolving Credit Facility Availability Period" means the period specified in Clause 4.3 (Revolving Credit Facility Availability Period).

# "Revolving Credit Facility Lender" means:

- (a) a lender identified as such in Part C of Schedule 1 (*Revolving Credit Facility Lenders*) or in an Additional Lender Facility Agreement providing for a revolving credit facility; or
- (b) a Transferee in respect of a Revolving Credit Facility made available pursuant to a Revolving Credit Facility Agreement.

"**Revolving Credit Facility Termination Date**" means, in relation to the Revolving Credit Facilities, the fifth anniversary of the Fourth Amendment Effective Date.

- 65 -

"**Revolving Lending Group**" means the Revolving Credit Facility Lenders under a Revolving Credit Facility Agreement, acting as a lending group in accordance with, and subject to the decision making rules under, that Revolving Credit Facility Agreement.

"Second Amendment Signing Date" means the date of the Common Terms Agreement Second Amendment Agreement.

"Secured Obligations" has the meaning given in the Deed of Appointment and Priority.

"Secured Parties" has the meaning given in the Deed of Appointment and Priority.

"Security" means the Liens created or purported to be created under the Security Documents.

"Security Agent" means Bank of China Limited, Macau Branch in its capacity as agent and security trustee for the Secured Parties or its successor appointed in accordance with the Deed of Appointment and Priority.

"Security Confirmation Documents" has the meaning given to that term in the Common Terms Agreement Fourth Amendment Agreement.

## "Security Documents" means:

| (a) | the Mortgage; |
|-----|---------------|
| (d) | the Moltgage, |

- (b) (with effect from the date thereof) the Cotai Mortgage;
- (c) the Power of Attorney;
- (d) (with effect from the date thereof) the Cotai Power of Attorney;
- (e) the Land Security Assignment;
- (f) the Assignment of Rights;
- (g) the Pledge over Gaming Equipment and Utensils;
- (h) the Pledge over Onshore Accounts;
- (i) the Assignment of Insurances;
- (j) (with effect from the date thereof) the Palo Assignment of Insurances;
- (k) the Assignment(s) of Reinsurances;
- (l) the Floating Charge;
- (m) (with effect from the date thereof) the Palo Floating Charge;
- (n) the Livranças, the Livrança Covering Letter and the Livrança Updated Covering Letter;

- 66 -

- (o) the Debenture;
- (p) the US Operating Account Control Agreement;
- (q) the Palo Pledge over Onshore Accounts;
- (r) the Guarantee;
- (s) the Wong Share Pledge and the Wong Consent;
- (t) the Company Share Pledge;
- (u) (with effect from the date thereof) the Palo Share Pledge;
- (v) the Wynn International Share Charge;
- (w) the Wynn HK Share Charge;
- (x) the Subordination Deed;
- (y) the Deed of Appointment and Priority;
- (z) each Direct Agreement;
- (aa) any other document from time to time creating, evidencing or entered into as security for or guaranteeing the Obligations of the Company or any other Obligor or member of the Restricted Group (including, if and when entered into, the Charge over HK Accounts) and any documents entered into pursuant to any of the documents referred to in this definition, including any such document notifying or acknowledging the granting or creation of such security or creating or evidencing security over an Account; and
- (bb) any document entered into pursuant to any further assurance provisions set out in any of the documents referred to in this definition which the Intercreditor Agent and the Company (both acting reasonably) agree to designate as a Security Document and any other document which the Intercreditor Agent and the Company (both acting reasonably) agree to designate as a Security Document.

"Senior Debt" means, at any date, the sum of all outstanding Advances made under the Facilities as at such date.

"Senior Finance Documents" means:

- (a) each Facility Agreement;
- (b) this Agreement;
- (c) each Security Document;
- (d) the Ancillary Finance Documents;
- (e) each Hedging Agreement; and
- (f) any other document entered into which the Intercreditor Agent and the Company (both acting reasonably) agree to designate as a Senior Finance Document.

- 67 -

"Senior Secured Creditors" means the GCLAs, the Agents, the Security Agent, the Lenders and the Hedging Counterparties.

"Senior Secured Indebtedness" means all Financial Indebtedness (actual or contingent) of the Company to the Senior Secured Creditors under the Senior Finance Documents together with all other amounts payable by the Company to the Senior Secured Creditors (or any of them) under or arising out of the Senior Finance Documents.

"Shareholder Guarantees" means the Guarantee and the Wong Share Pledge.

"**Shareholder Loans**" means Financial Indebtedness advanced by one or more of the Shareholders, the Wynn Obligors or Affiliates of the Wynn Obligors (including, without limitation, Wynn Resorts, WML or any Affiliate of any of Wynn Resorts or WML) to the Company or any other member of the Group that is subordinated in accordance with the terms provided by the Subordination Deed.

"Shareholders" means Wynn HK, Wynn International and Mr Wong Chi Seng.

"Shareholders' Agreement" means the amended and restated shareholders' agreement entered into between the Shareholders and the Company dated 16 September 2004.

"Sharing Payment" has the meaning given in Clause 25.1 (Payments to Senior Secured Creditors).

"Signing Date" means the date of signing of this Agreement being 14 September 2004 (and any reference in this Agreement to the date of this Agreement shall be construed accordingly).

"Single Employer Plan" means any Plan that is covered by Title IV of ERISA, but which is not a Multiemployer Plan.

"Site" means the land described in the Land Concession Contract.

"Site Easements" the easements appurtenant, easements in gross, licence agreements and other rights running for the benefit of the Company and/or appurtenant to the Site.

"Site Facilities" means

(a) the Site; and

(b) the Project Works (whether completed or uncompleted) in respect of the Projects.

"Solvent" means, when used with respect to any Person, as of any date of determination, (a) the amount of the "present fair saleable value" of the assets of such Person will, as of such date, exceed the amount of all "liabilities of such Person,

- 68 -

contingent or otherwise", as of such date, as such quoted terms are determined in accordance with applicable federal and state laws governing determinations of the insolvency of debtors, (b) the present fair saleable value of the assets of such Person will, as of such date, be greater than the amount that will be required to pay the liability of such Person on its debts as such debts become absolute and matured, (c) such Person will not have, as of such date, an unreasonably small amount of capital with which to conduct its business, (d) such Person will be able to pay its debts as they mature, and (e) such Person is not insolvent within the meaning of any applicable Legal Requirements. For purposes of this definition, (i) "debt" means liability on a "claim", and (ii) "claim" means any (A) right to payment, whether or not such a right is reduced to judgment, liquidated, unliquidated, fixed, contingent, matured, unmatured, legal, equitable, secured or unsecured or (B) right to an equitable remedy for breach of performance if such breach gives rise to a right to payment, whether or not such right to an equitable remedy is reduced to judgment, fixed, contingent, matured, secured or unsecured.

"Special Gaming Tax Account" means the account so designated in Schedule 6 (Accounts).

"Standard & Poor's" or "S&P" means Standard & Poor's Ratings Services, a division of The McGraw-Hill Companies, Inc., or its successor

"**Subconcession**" means any subconcession for the operation of games of chance and other games in casinos in the Macau SAR granted by the Company under the Concession Contract with the approval of Macau SAR and in accordance with paragraph 17 of Part B of Schedule 5 (*Covenants*).

"Subcontract" means any subcontract or purchase order entered into with any Subcontractor.

"Subcontractor" means any direct or indirect subcontractor of any tier under any Project Document.

"Subordinated Debt" means Financial Indebtedness that is subordinated in accordance with the terms provided by the Subordination Deed.

"Subordination Deed" means the deed formerly entitled the "Sponsor's Subordination Deed" and dated on or about the Signing Date between the Wynn Obligors, the Company, Wynn Resorts, Wynn Holdings and the Security Agent as further amended, restated, supplemented and novated by, *inter alia*, the Subordination Deed Third Deed of Amendment and Acknowledgement of Security (as such term is defined in the Common Terms Agreement Fourth Amendment Agreement), the Sponsors' Subordination Deed Second Deed of Amendment and Acknowledgement of Security (as such term is defined in the Common Terms Agreement Third Amendment Agreement) and the Sponsors' Subordination Deed of Release, Amendment and Acknowledgement of Security (as such term is defined in the Common Terms Agreement).

"Subsidiary" means, as to any Person, a corporation, partnership, limited liability company or other entity of which shares of stock or other ownership interests having ordinary voting power (other than stock or such other ownership interests having such

- 69 -

power only by reason of the happening of a contingency) to elect a majority of the board of directors or other managers of such corporation, partnership or other entity are at the time owned, or the management of which is otherwise controlled, directly or indirectly through one or more intermediaries, or both, by such Person.

"Substantial Completion" means, in relation to the Original Project, "Original Project Substantial Completion" (as defined in the Construction Contract) in respect of the Original Project taken as a whole and, in relation to the Expansion, "Expansion Project Substantial Completion" (as defined in the Construction Contract) in respect of the Expansion taken as a whole.

"Subsidiary Obligor" has the meaning given to it in paragraph 2.1(g) of Part B of Schedule 5.

"**Substantial Shareholder**" means any Person (other than in respect only of his direct shareholding in the Company as at the Signing Date, Mr Wong Chi Seng) who, legally or beneficially, directly or indirectly, owns or holds 5% or more of the outstanding Capital Stock of the Company other than directly or indirectly through Wynn International.

"Substantial Shareholder's Undertaking" means the undertakings set out in paragraph 2 of schedule 4 of the term sheet referred to in the underwriting agreement between Wynn Resorts, the Company, Banc of America Securities Asia Limited, Deutsche Bank AG, Hong Kong Branch and Société Générale Asia Limited dated 23 June 2005 given by each Substantial Shareholder on or about 14 September 2005.

"Substitution" has the meaning given to that term in the Common Terms Agreement Third Amendment Agreement.

"Swap Agreements" means interest rate swaps, caps or collar agreements or similar arrangements providing for protection against fluctuations in interest rates or currency exchange rates or the exchange of nominal interest obligations, either generally or under specific contingencies.

"Synthetic Lease Obligations" means all monetary obligations of a Person under (a) a so-called synthetic, off-balance sheet or tax retention lease, or (b) an agreement for the use or possession of property creating obligations which do not appear on the balance sheet of such Person but which, upon the insolvency or bankruptcy of such Person, would be characterized as the Financial Indebtedness of such Person (without regard to accounting treatment).

"**Taking**" means a taking or voluntary conveyance of all or part of any of the Project Security, or any interest therein or right accruing thereto or use thereof, as the result of, or in settlement of, any condemnation or other eminent domain proceeding by any Governmental Authority affecting any of the Project Security or any portion thereof.

"**Tax**" means any tax (including, without limitation, value-added and income), levy, impost, duty or other charge or withholding of a similar nature (including any penalty or interest payable in connection with any failure to pay or any delay in paying any of the same).

- 70 -

"**Tax Adviser**" means Ernst & Young as the tax adviser acting on behalf of all Senior Secured Creditors pursuant to the engagement letter dated 10 June 2004.

"Tax Credit" means a credit against, relief or remission for, or repayment of any Tax.

"Tax Deduction" means any deduction or withholding for or on account of Tax.

"**Tax Payment**" means an increased payment made by the Company to a Senior Secured Creditor under Clause 11.2 (*Tax gross-up*) or a payment under Clause 11.3 (*Tax indemnity*).

"Technical Adviser" means with respect to the Projects (but not with respect to the Cotai Project), as the case may be:

- (a) Mott Connell Ltd. as the technical adviser acting on behalf of all Senior Secured Creditors pursuant to the engagement letters dated 3 February 2004, 27 April 2005, 30 April 2007 and any subsequent engagement letter to be entered into;
- (b) the technical adviser acting on behalf of all Senior Secured Creditors according to the scope of work and fees agreed by the Senior Secured Creditors and approved by the Company (such approval not to be unreasonably withheld or delayed) before the CP Satisfaction Date; or
- (c) the technical adviser appointed by the Intercreditor Agent and, unless an Event of Default has occurred and is continuing, approved by the Company (such approval not to be unreasonably withheld or delayed) from time to time after the CP Satisfaction Date to act on behalf of the Senior Secured Creditors as and when required to advise the Senior Secured Creditors in respect of the Projects.

"**Technical Adviser's Advance Certificate**" has the meaning given in paragraph 5 of sub-section I of Part B of Schedule 2 (*Conditions Precedent*) with respect to the Projects (but not with respect to the Cotai Project).

"**Technical Adviser's Monthly Report**" means with respect to the Projects (but not with respect to the Cotai Project) a monthly status report, in form and substance acceptable to the Intercreditor Agent, delivered to the Intercreditor Agent on or before the fifteenth day of each calendar month up to and including the calendar month immediately following the Expansion Opening Date and describing in reasonable detail the progress of the construction of the Projects, including reviews and assessments of the Project Schedule and the Monthly Construction Period Report and each of its attachments delivered during the preceding calendar month.

"Term Facility" means the term loan facilities provided pursuant to the Term Facility Agreement.

"**Term Facility Agent**" means Bank of China Limited, Macau Branch as facility agent for the Term Facility Lenders or its successor appointed in accordance with this Agreement.

- 71 -

"**Term Facility Agreement**" means the agreement formerly entitled the "Hotel Facility Agreement" between the Company, the Term Facility Agent and the Term Facility Lenders. as further amended, restated, supplemented and novated from time to time by, *inter alia*, the Term Facility Agreement Third Amendment Agreement.

**"Term Facility Agreement Third Amendment Agreement**" has the meaning given to such term in the Common Terms Agreement Fourth Amendment Agreement .

"Term Facility Availability Period" means the period specified in Clause 4.1 (Term Facility Availability Period).

"Term Facility HKD Disbursement Account" means the account so designated in Schedule 6 (Accounts).

"**Term Facility Lender**" means a lender identified as such in Part A of Schedule 1 (*Term Facility Lenders*) or a Transferee in respect of the Term Facility.

"Term Facility USD Disbursement Account" means the account so designated in Schedule 6 (Accounts).

"**Term Lending Group**" means the Term Facility Lenders, acting as a lending group in accordance with, and subject to the decision making rules under, the Term Facility Agreement.

"Term Loan Facilities" means each of the term loan facilities provided pursuant to the Term Loan Facilities Agreements.

"Term Loan Facilities Agreements" means:

(a) the Term Facility Agreement; and

(b) each Additional Lender Facility Agreement providing for a term loan facility.

"**Term Loan Facility Lender**" means a Term Facility Lender or an Additional Lender party to an Additional Lender Facility Agreement providing for a term loan facility.

"Termination Event" has the meaning given in paragraph 2 (Certificates; Other Information) of Part A of Schedule 5 (Covenants).

"**Termination Proceeds**" means compensation or other proceeds paid by the Macau SAR in relation to the termination or rescission of the Concession Contract.

"Third Amendment Effective Date" has the meaning given in the Common Terms Agreement Third Amendment Agreement.

"Third Amendment Signing Date" means the date of the Common Terms Agreement Third Amendment Agreement.

"**Total Debt**" means, in relation to the Restricted Group at any time, the aggregate principal amount of all Financial Indebtedness of each member of the Restricted Group at such time (without double counting) but:

(a) excluding Financial Indebtedness referred to in paragraph (j) of the definition thereof (save in relation to any Realised Hedge Loss);

- 72 -

- (b) excluding Financial Indebtedness arising in respect of the Performance Bond Facility (save in relation to any drawing under the Concession Contract Performance Bond);
- (c) excluding Financial Indebtedness arising in respect of any Shareholder Loans;
- (d) excluding such Financial Indebtedness to the extent it is owed to another member of the Restricted Group;
- (e) including the amount of any liability or obligation, whether or not contingent, assumed by the Company under any Resort
   Management Agreement and quantified in the same manner as though it were a Guarantee Obligation, in accordance with the deeming provision set out in the definition thereof in this Clause 1.1; and
- (f) deducting the aggregate amount of any balances standing to the credit of, amounts on deposit in and any Permitted Investments held, in each case, in any Account.

"Transaction Document" means a Senior Finance Document or a Major Project Document (other than any Resort Management Agreement).

"Transfer Date" means, in relation to a Transferee, the later of:

- (a) the proposed Transfer Date specified in the Novation Certificate; and
- (b) the date on which the Intercreditor Agent executes the Novation Certificate.

"**Transferee**" means a Person to whom the rights and obligations of a Lender under the Senior Finance Documents to which that Lender is a party are transferred in accordance with Clause 21.6 (*Transfers by Lenders*).

"UCC" means the Uniform Commercial Code of any State in the United States of America, as in effect from time to time.

"**Underwriting Agreement**" means the underwriting agreement between the Company and the parties referred to therein as "GCLAs" dated 30 April 2007.

"Unpaid Sum" means any sum due and payable by an Obligor but unpaid under the Senior Finance Documents.

"Upfront Premium Account" means the account so designated in Schedule 6 (Accounts).

"US Operating Account" means the account so designated in Schedule 6 (Accounts).

"US Operating Account Control Agreement" means the bank account control agreement so entitled dated 14 September 2005 between the Company, the Security Agent and Bank of America, N.A.

- 73 -

"USD" or "US dollars" denotes the lawful currency of the United States of America.

"USD Debt Service Account" means the account so designated in Schedule 6 (Accounts).

"USD Debt Service Reserve Account" means the account so designated in Schedule 6 (Accounts).

"USD Operating Account" means the account so designated in Schedule 6 (*Accounts*).

"Voting Entitlement" means, in respect of a Decision:

- (a) in relation to a Lender, the sum of the US dollar equivalent amounts, as at the Decision Date for such Decision, of its participations in the outstanding Advances and the aggregate undrawn Available Commitments of such Lender under the Facilities;
- (b) in relation to each Hedging Counterparty (after a Hedging Voting Right Event has occurred in relation to such Hedging Counterparty and is continuing), the US dollar equivalent value, as at the Decision Date for such Decision, of the Realised Hedge Loss due and payable but unpaid by the Company to such Hedging Counterparty under the Hedging Agreement to which such Hedging Counterparty is party.

"Voting Stock" means, with respect to any Person as of any date, the Capital Stock of such Person that is at the time entitled to vote in the election of the Board of Directors of such Person.

"WML" means Wynn Macau, Limited.

"WML Debt" means Financial Indebtedness of WML (or a Subsidiary of WML, other than an Obligor, any Excluded Subsidiary or any member of the Group).

"Wholly Owned Subsidiary" means, as to any Person, any other Person in which all (or, in the case of a *Sociedade Anonima* incorporated in Macau, not less than 99.8%) of the Capital Stock (other than directors' qualifying shares required by any Legal Requirement) of such other Person is beneficially owned by such first-mentioned Person directly and/or through other Wholly Owned Subsidiaries.

"**Wong Consent**" means the written consent dated 2 September 2004 given by Mrs Cheung Wai Hing in relation to the obligations of Mr Wong Chi Seng under and the Security granted pursuant to the Wong Share Pledge.

"Wong Option Agreement" means the option agreement between Mr Wong Chi Seng, Wynn International and the Company delivered on or about 14 September 2004.

"Wong Share Pledge" means the document so entitled dated on or about 14 September 2004 between Mr Wong Chi Seng and the Security Agent.

"Working Capital" means, at any date, an amount equal to Current Assets on such date minus Current Liabilities on such date.

- 74 -

"Wynn Asia" means Wynn Group Asia, Inc.

"Wynn Asia 2" means WM Cayman Holdings Limited II.

"Wynn Event" means:

- (a) the first day on which Wynn Resorts ceases to beneficially own, directly or indirectly, 51% of the outstanding Capital Stock of the Company (measured by both voting power and size of equity interests); or
- (b) the first day on which Wynn Resorts otherwise ceases to have, directly or indirectly, the ability or the right to direct or procure the direction of the management and policies of the Company.

"Wynn HK" means Wynn Resorts (Macau), Limited.

"Wynn HK Share Charge" means the share charge so entitled dated on or about the date of this Agreement between Wynn Holdings and the Security Agent.

"Wynn Holdings" means Wynn Resorts (Macau) Holdings, Ltd.

"Wynn International" means Wynn Resorts International, Ltd.

"Wynn International Share Charge" means the share charge so entitled dated on or about the date of this Agreement between Wynn Asia 2 and the Security Agent.

"Wynn Non-Obligor Subordination Deed Party" means each of Worldwide Wynn, LLC, Wynn Design & Development, LLC Wynn International Marketing, Ltd., the Corporate Services Provider and the Licensor.

"Wynn Obligor" means Wynn Holdings, Wynn Asia 2, Wynn International and Wynn HK.

"Wynn Resorts" means Wynn Resorts, Limited.

"Wynn Resorts Group" means Wynn Resorts and each of its Subsidiaries for the time being.

# 1.2 Principles of Construction

Any reference in this Agreement to:

"continuing", in relation to a Default or an Event of Default, shall be construed as a reference to a Default or an Event of Default which has not been remedied or waived;

the "**equivalent**" of one currency (the "**original currency**") in another currency (the "**conversion currency**") shall (unless otherwise specified) be determined by the Intercreditor Agent or such Person nominated by the Intercreditor Agent for that purpose by reference to its spot rate of exchange in Hong Kong for the purchase of the conversion currency with the original currency at or about 11:00 a.m. on the date of the determination or if no such spot rate of exchange exists on that date, by such other method as the Intercreditor Agent (in consultation with the Company) shall reasonably determine;

- 75 -

## "including" is without limitation;

a "**month**" is a reference to a period starting on one day in a calendar month and ending on the numerically corresponding day in the next succeeding calendar month save that:

- (a) if any such numerically corresponding day is not a Business Day, such period shall end on the immediately succeeding Business Day to occur in that next succeeding calendar month or, if none, it shall end on the immediately preceding Business Day; and
- (b) if there is no numerically corresponding day in that next succeeding calendar month, that period shall end on the last Business Day in that next succeeding calendar month,

(and references to "months" shall be construed accordingly);

"**repay**" (or any derivative form thereof) shall, subject to any contrary indication, be construed to include "**prepay**" (or, as the case may be, the corresponding derivative form thereof); and

a document being in "**substantially the Agreed Form**" or in substantially a specified form shall be construed as meaning such document being in the same form as the Agreed Form or the specified form save for the insertion of information left in blank or typographical errors.

## Rules of Interpretation

1.3

In this Agreement, unless the context otherwise requires:

- (a) words importing the singular include the plural and vice versa;
- (b) words importing a gender include every gender;
- (c) references to any document (including this Agreement) are references to that document as amended, consolidated, supplemented, confirmed, novated or replaced from time to time;
- (d) references to this Agreement are references to this Agreement and the Schedules;
- (e) references to clauses and Schedules are references to clauses of, and Schedules to, this Agreement;
- (f) headings are for convenience only and shall be ignored in construing this Agreement;
- (g) references to any party to this Agreement include references to its respective successors, permitted transferees and permitted assigns;

- (h) references to law shall be construed as references to any constitutional provision, treaty, decree, convention, statute, act, regulation, rule, ordinance, subordinate legislation, rule of common law and of equity and judgement;
- (i) references to any law are references to that law as amended, consolidated, supplemented or replaced from time to time;
- (j) references to any judgement include references to any order, injunction, decree, determination or award of any court or tribunal; and
- (k) a time of day is a reference to Hong Kong time unless otherwise stated.

Following the Fourth Amendment Effective Date, any reference to the Hotel Facility Agent or a Hotel Facility Lender in any Senior Finance Document shall be deemed to be a reference to the Term Facility Agent or a Term Facility Lender, as the case may be.

### 1.4 **Conflict with a Senior Finance Document**

In the case of any conflict between:

- 1.4.1 the terms of this Agreement and the terms of any other Senior Finance Document (save for the Facility Agreements and the Deed of Appointment and Priority), the terms of this Agreement shall prevail;
- 1.4.2 the terms of this Agreement and the terms of any Facility Agreement, the terms of that Facility Agreement shall prevail (save in the case of Clause 33 (*Intercreditor Arrangements*) which shall prevail over the terms of the Facility Agreement);
- 1.4.3 the terms of this Agreement and the terms of the Deed of Appointment and Priority, the terms of the Deed of Appointment and Priority shall prevail; or
- 1.4.4 the terms of the Deed of Appointment and Priority and the terms of any Facility Agreement, the terms of the Deed of Appointment and Priority shall prevail.

# 1.5 Third party rights

1.5.1 The Contracts (Rights of Third Parties) Act 1999 applies to:

- (a) sub-clause 3.2.5 of Clause 3.2 (*Completion of an Advance Request*) but only for the benefit of the relevant officer of the Company;
- (b) Clause 23.10 (*No Actions*) but only for the benefit of any director, officer or employee of any of the Agents or any of the GCLAs;
- (c) paragraph 5 of Part A of Schedule 2 (Conditions Precedent) but only for the benefit of the relevant Responsible Officer;
- (d) paragraph 1 of Part A of Schedule 5 (*Covenants*) but only for the benefit of the relevant Responsible Officers of the Company;

- 77 -

- (e) Clause 18.3 (*Permitted Cotai Reorganisation; Release of Palo Security*), paragraph 15.3 (*Additional Collateral, Discharge of Liens, etc.*) of Part A of Schedule 5 (*Covenants*), paragraph 6 (*Limitation on Restricted Payments*) of Part B of Schedule 5 (*Covenants*), paragraph 15.2.1 (*Permits and Other Contracts*) of Part B of Schedule 5 (*Covenants*) and paragraphs 1 and 3.7 of Schedule 6 (*Accounts*) but, in each case, only for the benefit of the Obligors and the Wynn Non-Obligor Subordination Deed Parties;
- (f) Clause 23.9 (*Exclusion of Liabilities*) but only for the benefit of any officer, employee or agent of an Agent or GCLA; and
- (g) Clause 28 (*Non-Recourse Liability*) but only for the benefit of the Operatives,
- subject always to the terms of Clause 37 (Governing Law) and Clause 38 (Jurisdiction).
- 1.5.2 Except as provided in sub-clause 1.5.1 above, a Person who is not a party to this Agreement has no right under the Contracts (Rights of Third Parties) Act 1999 to enforce or to enjoy the benefit of any term of this Agreement.
- 1.5.3 Notwithstanding any term of any Senior Finance Document, the consent of any Person who is not a party to this Agreement is not required to rescind or vary this Agreement.

## 1.6 Advisers

Where this Agreement contemplates the doing of any act or thing by an Adviser, the Intercreditor Agent shall use its reasonable endeavours to ensure that such act or thing is done by such Adviser in a timely manner.

## 1.7 [Not used]

### 1.8 Application of Agreement

- 1.8.1 The parties acknowledge that this Agreement contains references to certain obligations and contracts which have been performed and references to certain matters and circumstances that have occurred, in each case, prior to the Fourth Amendment Effective Date. The existence of such references shall not be construed so as to imply (in respect of such obligations, contracts, matters and circumstances) additional or continuing obligations beyond the express terms of this Agreement.
- 1.8.2 Without limiting the foregoing, the parties further acknowledge that certain provisions of this Agreement are expressed to apply (or be applied) to the Projects only, and are not intended to apply to the Cotai Project. This Agreement shall apply to the Cotai Project only to the extent expressly provided in this Agreement.

# 2. CONDITIONS PRECEDENT

# 2.1 Conditions Precedent to the CP Satisfaction Date

2.1.1 The right of the Company to submit an Advance Request for the Initial Advance is subject to the Company having satisfied the conditions precedent set out in Part A of Schedule 2 (*Conditions Precedent*) in form and substance acceptable to the Intercreditor Agent.

2.1.2 In relation to each Facility, the Company shall be deemed to have satisfied:

- (a) the conditions precedent set out in:
  - (i) paragraph 1(a) (in respect only of the Concession Contract, the Land Concession Contract, the Concession Contract Performance Bond and the IP Agreement);
  - (ii) paragraph 1(c) (in respect only of the Notice to Proceed);
  - (iii) paragraph 2(a) (in respect only of the Power of Attorney, the Livranças and the Livrança Covering Letter);
  - (iv) paragraph 2(b);
  - (v) paragraph 2(c) (in respect only of the Power of Attorney, the Livranças and the Livrança Covering Letter);
  - (vi) paragraph 3(a);
  - (vii) paragraph 3(c);
  - (viii) paragraph 3(d);
  - (ix) paragraph 4;
  - (x) paragraph 5(a)(i);
  - (xi) paragraph 5(b)(i) (in respect only of: China Overseas Holdings Limited, China State Construction Engineering (Hong Kong) Limited, Leighton Contractors (Asia) Limited, Leighton Holdings Limited, Wynn Asia, Wynn Resorts Holdings LLC, Wynn International, Wynn Resorts, Wynn HK, Wynn Holdings, Wynn Design & Development, LLC);
  - (xii) paragraph 8(b);
  - (xiii) paragraph 13;
  - (xiv) paragraph 14(o);
  - (xv) paragraph 14(p);
  - (xvi) paragraph 16;

- 79 -

| (xvii)       | paragraph 17(a)(i);        |
|--------------|----------------------------|
| (xviii)      | paragraph 17(a)(ii);       |
| (xix)        | paragraph 17(a)(iv);       |
| (xx)         | paragraph 17(a)(v);        |
| (xxi)        | paragraph 17(a)(vi);       |
| (xxii)       | paragraph 17(a)(vii);      |
| (xxiii)      | paragraph 17(b);           |
| (xxiv)       | paragraph 17(c);           |
| (xxv)        | paragraph 18;              |
| (xxvi)       | paragraph 19;              |
| (xxvii)      | paragraph 25;              |
| (xxviii)     | paragraph 26; and          |
| (xxix)       | paragraph 28,              |
| of Dowt A of | f Schodulo 2 (Conditions I |

of Part A of Schedule 2 (*Conditions Precedent*) following execution and delivery of this Agreement by each of the Persons expressed to be a party hereto; and

(b) the remaining conditions precedent in this Clause 2.1 if, prior to the date of the Advance Request for the Initial Advance, the Intercreditor Agent has notified the Company and the Facility Agents accordingly.

2.1.3 Notwithstanding anything in this Agreement to the contrary, the Company may, from time to time prior to the CP Satisfaction Date, but no more frequently than once every five Business Days, submit or resubmit to the Intercreditor Agent for approval (if such approval is required) additional documents, instruments, agreements, certificates and other items listed under conditions precedent in Part A of Schedule 2 (*Conditions Precedent*) that were not deemed satisfied under Clause 2.1.2(a). The Intercreditor Agent agrees to promptly review and, where it considers appropriate under the terms of the Senior Finance Documents, determine whether to approve any such submittals or resubmittals by the Company (or, if it considers it not appropriate to make such determination acting on its own, the Intercreditor Agent agrees promptly to forward to the relevant Senior Secured Creditors such submittals or resubmittals for approval). Upon approval (whether by the Intercreditor Agent or the Required Lenders, as the case may be), the Intercreditor Agent shall notify the Company and the Facility Agents accordingly, and the item(s) so submitted (or resubmitted) and notified approved shall be deemed to have been satisfied with the same effect as the conditions precedent enumerated in Clause 2.1.2(a) as of the date of such notification.

- 80 -

#### Conditions Precedent to each Advance

2.2

(a)

The obligation of each Lender to participate in each Advance under a Term Loan Facility and a Revolving Credit Facility is subject to:

- the Company having satisfied the conditions set out in Part B2 of Schedule 2 (*Conditions Precedent*) in respect of such Facility in form and substance acceptable to the Intercreditor Agent; and
- (b) in respect of each Advance under a Revolving Credit Facility, the Company having procured that each of the:
  - (i) Sponsor's Subordination Deed, Deed of Release, Amendment and Acknowledgement of Security;
  - (ii) Debenture Deed of Amendment and Release;
  - (iii) Wynn International Share Charge Deed of Amendment and Acknowledgement of Security;
  - (iv) Wynn HK Share Charge Deed of Amendment and Acknowledgement of Security;
  - (v) Charge over HK Accounts Deed of Acknowledgement;
  - (vi) US Operating Account Control Agreement Confirmation Agreement;
  - (vii) Amendment to Mortgage;
  - (viii) Amendment to Land Security Assignment;
  - (ix) Amendment to Assignment of Rights;
  - (x) Amendment to Pledge over Gaming Equipment and Utensils;
  - (xi) Amendment to Pledge over Onshore Accounts;
  - (xii) Amendment to Assignment of Insurances;
  - (xiii) Amendment to Floating Charge;
  - (xiv) Amendment to Wong Share Pledge; and
  - (xv) Amendment to Company Share Pledge,

have been registered as necessary at Companies House in England and Wales, the Hong Kong Companies Registry, the Financial Supervision Commission of the Isle of Man, the Conservatória dos Registos Comercial e de Bens Móveis in Macau SAR, the Conservatória do Registo Predial in Macau SAR and the applicable Uniform Commercial Code filing office for local/county, state and federal Uniform Commercial Code filings, as applicable, based on the Senior Finance Document subject to the filing by no later than the date falling 30 days from the Fourth Amendment Signing Date.

- 81 -

#### 2.3 Independent rights and obligations of Lenders

- 2.3.1 The obligations of each Senior Secured Creditor under the Senior Finance Documents are several. Failure by a Senior Secured Creditor to perform its obligations under the Senior Finance Documents does not affect the obligations of any other party under the Senior Finance Documents. No Senior Secured Creditor is responsible for the obligations of any other Senior Secured Creditor under the Senior Finance Documents.
- 2.3.2 The rights of each Senior Secured Creditor under or in connection with the Senior Finance Documents are separate and independent rights and any debt arising under the Senior Finance Documents to a Senior Secured Creditor from an Obligor shall be a separate and independent debt.
- 2.3.3 A Senior Secured Creditor may, except as otherwise stated in the Senior Finance Documents, separately enforce its rights under the Senior Finance Documents.
- 2.3.4 Notwithstanding any other provision of the Senior Finance Documents, no Lender under a Term Loan Facility shall be obliged to make or participate in an Advance on a proposed Advance Date if drawdowns under another Term Loan Facility have been withheld, suspended or cancelled in accordance with the Senior Finance Documents.

### 3. DRAWDOWN OF ADVANCES

# 3.1 Drawdown conditions

Subject to the terms of this Agreement and the Facility Agreements, the Company may request, and the relevant Lending Group shall make, Advances under a Facility if:

- 3.1.1 not later than 5:00 p.m. on the tenth Business Day before the proposed Advance Date (in the case of an Advance under the Term Loan Facilities and with any necessary amendments thereto made and received by the Intercreditor Agent and the relevant Facility Agent not later than 3:00 p.m. on the sixth Business Day before the proposed Advance Date) or the fourth day before the proposed Advance Date (in the case of an Advance under a Revolving Credit Facility), the Intercreditor Agent and the relevant Facility Agent have received a completed Advance Request;
- 3.1.2 in the case of the Term Loan Facilities, no other Advance Request has been served by the Company in respect of any Term Loan Facility in the same month; and
- 3.1.3 in the case of the Revolving Credit Facility, if the Advance to be requested under the Advance Request, when drawn on the proposed Advance Date, would result in the aggregate US dollar equivalent amount of all Advances outstanding under the Facilities to exceed USD1,550,000,000, the Intercreditor Agent has received (prior to the duly completed Advance Request in respect of such Advance being received) all documents and other evidence listed in paragraph Part B3 of Schedule 2 (*Conditions Precedent*), in form and substance satisfactory to the Intercreditor Agent.

- 82 -

#### 3.2 Completion of an Advance Request

Each Advance Request is irrevocable and shall not be regarded as having been completed unless:

- 3.2.1 it is signed by a Responsible Officer of the Company whose specimen signature has been delivered to the Intercreditor Agent and who is identified as being authorised to so sign on behalf of the Company by a resolution of its Board of Directors, a copy of which, together with a certification in relation thereto by a Responsible Officer in substantially the form set out in paragraph 3(ii) of Part C of Schedule 2 (*Conditions Precedent*), has also been delivered to the Intercreditor Agent;
- 3.2.2 the proposed Advance Date is a Business Day within the relevant Availability Period;

3.2.3 it specifies:

- (a) the amount and currency of the Advances to be made;
- (b) the Facility under which each such Advance shall be made;
- (c) the first Interest Period for such Advances (which shall be the same for each such Advance requested under a Term Loan Facility);
- (d) (in the case of Advances under the Term Facility) the purpose for which such Advances shall be applied; and
- (e) (in the case of Advances under the Revolving Credit Facility) the purpose for which such Advances shall be applied.
- 3.2.4 the amount requested under each Facility is not more than the aggregate for the time being of each Lender's Available Commitment under such Facility and, in the case of:
  - any Term Loan Facility, the amount requested is either a minimum amount of USD5,000,000 or (as the case may be) its HK dollar equivalent which is less than the aggregate for the time being of each Lender's Available Commitment under such Facility or, if less than such minimum amount, equal to this latter amount;
  - (b) [Not used]
  - (c) [Not used]
  - (d) the Revolving Credit Facilities, the aggregate amount requested is either a minimum amount of USD5,000,000 or (as the case may be) its HK dollar equivalent which is less than the aggregate for the time being of each Lender's Available Commitment under such Facility or, if less than such minimum amount, equal to this latter amount; and

- 83 -

- 3.2.5 it certifies, among other things and without any personal liability on the part of the officer of the Company signing such Advance Request, that:
  - (a) [Not used]
  - (b) [Not used]
  - (c) [Not used]
  - (d) no Default (or, in the case of any Rollover Advance (as defined in each Revolving Credit Facility Agreement), Event of Default) is continuing or would result from the proposed Advances; and
  - (e) the representations and warranties contained in Schedule 4 (*Representations and Warranties*) which are repeated by the Company at the Advance Date are true and correct in all material respects with reference to the facts and circumstances existing on the date of the Advance Request.

## 3.3 Funding of Development Account from Revolving Credit Facility

The Company shall ensure that a USD equivalent amount at least equal to the Relevant Amount is paid (immediately upon utilisation) from the first Advance made under the Revolving Credit Facility on or after the Fourth Amendment Effective Date into the applicable Development Account (as defined in the Term Facility Agreement); *provided* that if such first Advance is in a USD equivalent amount less than the Relevant Amount, the Company shall pay into the applicable Development Account (in each case, immediately upon utilisation) all of such first Advance and a USD equivalent amount from each subsequent Advance made under the Revolving Credit Facility until an aggregate USD equivalent amount at least equal to the Relevant Amount has been so paid.

For the purpose of this Clause 3.3, "**Relevant Amount**" means a USD equivalent amount equal to the aggregate USD equivalent amount required to pay any fees due and payable to any Senior Secured Creditor on the Fourth Amendment Effective Date under the Fourth Amendment Effective Date Fee Letters.

# 4. AVAILABILITY PERIODS

## 4.1 Term Facility Availability Period

Subject to other terms of the Senior Finance Documents, the Term Facility shall be made available from the Fourth Amendment Effective Date until the date falling 5 Business Days from the Fourth Amendment Effective Date.

- 84 -

# 4.2 [Not used]

# 4.3 **Revolving Credit Facility Availability Period**

Subject to other terms of the Senior Finance Documents, the Revolving Credit Facility shall be made available pursuant to the Revolving Credit Facility Agreement from the date of issue of the Project Certificate of Occupancy for the Original Project until the earliest of:

4.3.1 the Termination Date (as defined in the Revolving Credit Facility Agreement); and

4.3.2 the date upon which the Advances thereunder are declared to be immediately due and payable pursuant to Clause 19.2 (*Remedies following an Event of Default*).

# 4.4 Additional Lender Facility Availability Period

Subject to other terms of the Senior Finance Documents, each Additional Lender Facility shall be made available from the date of issue of the Project Certificate of Occupancy for the Original Project until the earliest of:

- (a) the termination date (as set out in the applicable Additional Lender Facility Agreement); and
- (b) the date upon which the Advances thereunder are declared to be immediately due and payable pursuant to Clause 19.2 (*Remedies following an Event of Default*).

### 5. PURPOSE

### 5.1 **Purpose - General**

The Company shall apply the proceeds of each Advance under a Facility in accordance with the relevant Facility Agreement and this Agreement.

# 5.2 No Obligation to be Concerned with Application

None of the Senior Secured Creditors shall be obliged to concern themselves with the application of proceeds of the Facilities.

# 5.3 Limitation on Advances

The proceeds of the Advances shall not be applied towards the acquisition (or maintenance or repair) of any equipment or utensils used in the operation of casino games of chance or other forms of gaming.

- 85 -

#### 6. **PRO RATA DRAWINGS**

The Company shall ensure that:

- (a) where an Advance is requested under a Term Loan Facility, an Advance, as a proportion of the Available Commitments under such Facility, in a US dollar equivalent amount *pro rata* with that requested has also been requested to be made on the same Advance Date under each of the other Term Loan Facilities; and
- (b) where an Advance is requested under a Revolving Credit Facility, an Advance, as a proportion of the Available Commitments under such Facility, in a US dollar equivalent amount *pro rata* with that requested has also been requested to be made on the same Advance Date under each of the other Revolving Credit Facilities.

#### 7. [NOT USED]

#### 8. REPAYMENTS, PREPAYMENTS AND CANCELLATION

# 8.1 **Repayments**

The Company may repay principal amounts falling due under any Facility Agreement only in accordance with that Facility Agreement and this Agreement.

#### 8.2 Voluntary Prepayment of the Term Loan Facilities

- 8.2.1 Subject to the other provisions of this Clause 8 and any applicable terms in the Facility Agreements, the Company may, on at least 30 days' prior written notice to the Intercreditor Agent (which notice shall, if not withdrawn prior thereto, become irrevocable on the tenth Business Day prior to the proposed prepayment date), make voluntary prepayments under the Term Loan Facilities on the last day of any Interest Period, *provided* that for each voluntary prepayment, the amount prepaid under the Term Loan Facilities must:
  - (a) if prepaid prior to the Expansion Opening Date (other than in the case of paragraph (b)(i) below), equal the balance of the principal amount owing to all Term Loan Facility Lenders (and each of the Term Loan Facilities (and any Available Commitments thereunder) shall be automatically cancelled); or
  - (b) if prepaid:
    - (i) using only the proceeds of Shareholder Loans not forming part of any other Equity required to be paid up or advanced in accordance with the terms of the Senior Finance Documents; or
    - (ii) following the Expansion Opening Date,

exceed an aggregate of USD10,000,000 or its equivalent or, if less, the balance of the principal amount owing to all Term Loan Facility Lenders.

- 8.2.2 Amounts prepaid under sub-clause 8.2.1 above shall be applied on the Interest Payment Date on which they are made *pro rata* between the Advances outstanding under the Term Loan Facilities and applied *pro rata* against the repayment instalments of those Advances.
- 8.2.3 For purposes of clarification, Permitted Loan Repurchases shall not constitute payments or prepayments of any amount under the Term Loan Facilities and shall not be subject to the provisions of this Clause 8.

- 86 -

### 8.3 Mandatory Prepayment

- 8.3.1 The Company shall prepay Advances and/or cancel Available Commitments under the Facilities on the dates and in the amounts specified in Schedule 9 (*Mandatory Prepayment*) and this Clause 8.3.
- 8.3.2 Any amount prepaid under this Clause 8.3 and Schedule 9 (*Mandatory Prepayment*) shall be applied in the following order:
  - (a) **first**, pro rata between the Advances outstanding under the Term Loan Facilities and then in inverse order of maturity against the remaining repayment instalments of those Advances;
  - (b) *second*, in cancellation of the Available Commitments under the Revolving Credit Facilities (and the Available Commitments of the Lenders under the Revolving Credit Facilities will be cancelled rateably); and
  - (c) *thirdly*, in prepayment pro rata of Advances outstanding under the Revolving Credit Facilities (and any Available Commitments of the Lenders under the Revolving Credit Facilities associated therewith shall be automatically cancelled).
- 8.3.3 Each of the Facilities (and any Available Commitments thereunder) shall be automatically cancelled upon the Company being required to make prepayment pursuant to paragraph 7 of Schedule 9 (*Mandatory Prepayment*).

#### 8.4 Cancellation

- (a) Save as provided in Clause 8.5 (*Prepayment and Cancellation of Individual Lenders*) and Clause 14 (*Illegality*), the Company may only cancel the whole or any part (being a minimum amount of USD25,000,000) of the Available Commitments under the Revolving Credit Facilities made available pursuant to the Revolving Credit Facility Agreements on not less than thirty days' prior irrevocable written notice to the Intercreditor Agent and the relevant Facility Agreement. Such cancellation shall apply *pro rata* across all Revolving Credit Facilities made available under the Revolving Credit Facility Agreements.
- (b) The Available Commitments of each Lender under each of the Term Loan Facilities will be automatically cancelled at the close of business in Hong Kong on the last day of the relevant Availability Period for that Term Loan Facility to the extent undrawn at that date.

#### Prepayment and Cancellation of Individual Lenders

8.5

If:

- (a) any sum payable to any Lender by the Company is required to be increased under Clause 11.2 (*Tax gross-up*); and/or
- (b) any Lender claims indemnification from the Company under Clause 11.3 (*Tax Indemnity*) or Clause 12 (*Increased costs*); and/or
- (c) a Market Disruption Event occurs in relation to any Advance for any Interest Period pursuant to Clause 10.2 (*Market disruption*); and/or
- (d) any Lender withholds its consent to the incurrence of any Financial Indebtedness by the Company such that the Intercreditor Agent is unable to approve the incurrence of additional Financial Indebtedness in accordance with paragraph 2.1(j) of Part B of Schedule 5 (*Covenants*) or the amendment or waiver of paragraph 2.1 of Part B of Schedule 5 (*Covenants*); and/or
- (e) any Lender becomes a Non-Consenting Lender (as defined in Clause 8.8.3 below) and that Non-Consenting Lender has not, within 10 Business Days of being supplied with the same, executed (and returned to the Intercreditor Agent) all agreements and other instruments (to which it is required to be a party) needed to effect the transfer contemplated by (and in accordance with) Clause 8.8 (*Replacement of Non-Consenting Lender*); and /or
- (f) any Lender becomes an Outgoing Lender (as defined in Clause 8.9 below) and that Outgoing Lender has not, within 10 Business Days of being supplied with the same, executed (and returned to the Intercreditor Agent) all agreements and other instruments (to which it is required to be a party) needed to effect the transfer contemplated by (and in accordance with) Clause 8.9 (*Anti-Terrorism and Restricted Party Events*),

then, the Company may, subject to the other provisions hereof and, in the case of paragraph (d) above, whilst the circumstances described therein continue, and on giving at least fifteen days' prior irrevocable written notice to the Intercreditor Agent:

- (i) prepay that Lender's participation in the Advances outstanding under the relevant Facility Agreement on the Interest Payment Date which immediately ends after the Company's notice; and/or
- (ii) cancel that Lender's undrawn and uncancelled Available Commitments under the relevant Facility Agreement.

# 8.6 **Restrictions**

8.6.1 Any notice of cancellation or prepayment given under this Clause 8 shall be irrevocable if not withdrawn in accordance with Clause 8.2.1 (*Voluntary Prepayment of the Term Loan Facilities*) and shall specify the date or dates upon which the relevant cancellation or prepayment is to be made and the amount of the cancellation or prepayment.

- 88 -

- 8.6.2 Any prepayment or cancellation pursuant to this Clause 8 shall be made together with accrued interest and fees on the amount prepaid or cancelled and without premium or penalty, save that the Company shall pay any Break Costs and any other fees specified in the relevant Facility Agreement.
- 8.6.3 The Company shall not repay or prepay all or any part of the Advances or cancel all or any part of the Available Commitments under any Facility Agreement except in accordance with that Facility Agreement and this Clause 8.
- 8.6.4 The Company may not reborrow any part of the Term Loan Facilities which is prepaid.
- 8.6.5 No amount of the Available Commitments cancelled under this Clause 8 may be subsequently reinstated.
- 8.6.6 If the Intercreditor Agent receives a notice under this Clause 8, it shall promptly forward a copy of that notice to either the Company or the affected Lender (or the Facility Agent acting for such Lender), as appropriate.
- 8.6.7 If, following any prepayment pursuant to this Clause 8, the US dollar equivalent of the aggregate of the Notional Amounts of the Hedging Agreements is more than 125% of the US dollar equivalent of the aggregate of the Advances outstanding under the Term Loan Facilities following such prepayment, the Company shall reduce each such Notional Amount *pro rata* so that their US dollar equivalent is, in aggregate, not less than 50% and not more than 125% of the US dollar equivalent of the aggregate of such Advances (and, if the Term Loan Facilities are prepaid in full, the Company shall, subject to Schedule 8 (*Hedging Arrangements*), unwind all remaining transactions under the Hedging Agreements).

#### 8.7 Replacement of Lender

If any Lender:

- (a) claims any amounts from the Company under Clauses 11.2 (*Tax gross-up*), 11.3 (*Tax indemnity*) or 12 (*Increased costs*) hereof;
- (b) fails to make its portion of any Advance to be made by it on the relevant Advance Date; or
- (c) withholds its consent in any of the circumstances contemplated in Clause 8.5(d),

(an "Affected Lender"), the Company may (after paying all amounts then due under Clauses 11.2 (*Tax gross-up*), 11.3 (*Tax indemnity*) or 12 (*Increased costs*) hereof to the Affected Lender and, in the case of paragraph (c) above, whilst the circumstances referred to therein continue) designate a non-Affected Lender, any commercial bank or any other financial institution or bank reasonably satisfactory to the Intercreditor Agent (the "**Replacement Lender**") to accept a transfer in accordance with Clause 21.6 (*Transfers by Lenders*) of the Affected Lender's rights, benefits and obligations hereunder, and, promptly following such designation, the Affected Lender shall be

obliged to execute the Novation Certificate required for such transfer in accordance with Clause 21.6 (*Transfers by Lenders*) and the non-Affected Lender may, but shall not be obliged to, execute such Novation Certificate and, if it does so, shall be obliged to accept such transfer in accordance with Clause 21.6 (*Transfers by Lenders*). Any such acceptance of transfer shall be for a purchase price in cash payable at the time of transfer equal to the outstanding principal amount of such Affected Lender's participation in the outstanding Advances and all accrued interest, Break Costs and other amounts payable in relation thereto under the Senior Finance Documents.

#### Replacement of Non-Consenting Lender

- 8.8.1 If at any time any Lender becomes a Non-Consenting Lender (as defined in Clause 8.8.3 below), then the Company may, on 5 Business Days' prior written notice to the Intercreditor Agent and such Lender, replace such Lender by requiring such Lender to (and such Lender shall) transfer pursuant to Clause 21 (*Changes to the Parties*) all (and not part only) of its rights and obligations under this Agreement to a Lender or other bank, financial institution, trust, fund or other entity (a "**Replacement Lender**") selected by the Company, and which is acceptable to the Intercreditor Agent (acting reasonably) which confirms its willingness to assume and does assume all the obligations of the transferring Lender (including the assumption of the transferring Lender's participations on the same basis as the transferring Lender) for a purchase price in cash payable at the time of transfer equal to the outstanding principal amount of such Lender's participation in the outstanding Advances and all accrued interest, Break Costs and other amounts payable in relation thereto under the Senior Finance Documents. Such Non-Consenting Lender shall provide all reasonable assistance to effect the foregoing transfer.
- 8.8.2 The replacement of a Lender pursuant to this Clause shall be subject to the following conditions:
  - (a) neither the Intercreditor Agent nor the Lender shall have any obligation to the Company to find a Replacement Lender;
    - (b) such replacement of a Non-Consenting Lender must take place no later than 60 days after the date the Lender was deemed a Non-Consenting Lender under Clause 8.8.3; and
    - (c) in no event shall the Lender replaced under this Clause 8.8.2 be required to pay or surrender to such Replacement Lender any of the fees previously received by such Lender pursuant to the Senior Finance Documents.
- 8.8.3 In the event that:
  - (a) the Intercreditor Agent (at the request of the Company) has pursuant to Clause 33.1.2 notified the relevant Senior Secured Creditors of a Decision required in respect of a waiver or amendment of any provisions of the Senior Finance Documents;

<sup>- 90 -</sup>

- (b) the waiver or amendment in question requires the consent of all Lenders and, after the occurrence of a Hedging Voting Rights Event in relation to a Hedging Counterparty that is continuing, that Hedging Counterparty; and
- (c) Lenders and, after the occurrence and continuation of a Hedging Voting Right Event in relation to any Hedging
   Counterparty, that Hedging Counterparty, who hold, in aggregate, more than 66 2/3% of the Voting Entitlements of all such
   Senior Secured Creditors have voted in favour of that Decision,

then any Lender who does not and continues not to vote in favour of such Decision shall be deemed a "Non-Consenting Lender".

# 8.9 Anti-Terrorism and Restricted Party Events

(b)

- (a) If any litigation, governmental, regulatory or other proceedings by OFAC, FinCEN, the Monetary Authority of Macau, the Hong Kong Monetary Authority or any other United States, European Communities, Macau or Hong Kong Governmental Authority (or any divisions of any of them or authority deriving power from any of them) is pending or adversely determined against a Lender (an "Outgoing Lender") as a direct result of that Outgoing Lender's (i) receipt of funds or other property from a Restricted Party, (ii) breach of any Anti-Terrorism Law, (iii) breach of any anti-money laundering law or (iv) breach of any anti-corruption law, the Company may on 5 Business Days' prior written notice to the Intercreditor Agent and such Outgoing Lender, replace such Outgoing Lender by requiring such Outgoing Lender to (and such Outgoing Lender shall) transfer pursuant to Clause 21.4 (Assignment and Transfer by Lenders) all (and not part only) of its rights and obligations under this Agreement to a Lender or other bank, financial institution, trust, fund or other entity (an "Incoming Lender") selected by the Company and which is acceptable to the Intercreditor Agent (acting reasonably), which confirms its willingness to assume and does assume all the obligations of the transferring Outgoing Lender for a purchase price in cash payable at the time of transfer equal to the outstanding principal amount of such Outgoing Lender's participation in the outstanding Advances and all accrued interest and/or Break Costs and other amounts payable in relation thereto under the Senior Finance Documents.
  - The replacement of an Outgoing Lender pursuant to this Clause shall be subject to the following conditions:
    - (i) neither the Intercreditor Agent nor the Outgoing Lender shall have any obligation to the Company to find an Incoming Lender;
    - (ii) in the event of a replacement of an Outgoing Lender such replacement must take place no later than 60 days after the date the Outgoing Lender notifies the Company and the Intercreditor Agent, or the Company notifies the Intercreditor Agent, as the case may be, of the occurrence of any event set out in paragraph (a) above; and
    - (iii) in no event shall the Outgoing Lender replaced under this paragraph (b) be required to pay or surrender to such Incoming Lender any of the fees previously received by such Outgoing Lender pursuant to the Senior Finance Documents.

- 91 -

## 9. INTEREST, INTEREST PERIODS AND DEFAULT INTEREST

## 9.1 Calculation of interest

The Company shall pay interest under each Facility Agreement at the rate specified in that Facility Agreement.

## 9.2 **Payment of interest**

- 9.2.1 Subject to Clause 9.2.2 below, interest on each Advance shall be due on each Interest Payment Date relating to that Advance.
- 9.2.2 If an Interest Period for a Revolving Credit Facility Advance is 6 months or longer, interest on that Advance shall be due on the dates falling on three month intervals after the first day of that Interest Period *provided that* the last such due date shall be brought forward or postponed (as the case may be) so as to coincide with the Interest Payment Date relating to that Advance.

## 9.3 Interest Periods

The duration of each Interest Period shall be determined as follows:

- 9.3.1 Each Interest Period for a Term Loan Facility Advance shall start on the Advance Date for such Advance or (if already made) on the Effective Date and, thereafter, on the last day of its preceding Interest Period. A Revolving Credit Facility Advance has one Interest Period only which shall start on the Advance Date for such Advance.
- 9.3.2 Subject to this Clause 9, the duration of each Interest Period for each Advance under:
  - (a) a Term Loan Facility shall be one, two, three or six months; and
  - (b) a Revolving Credit Facility shall be one, two, three or six months,

in each case as the Company may, by not less than five (or, in the case of any Term Loan Facility Advance made on or prior to the Effective Date, two) Business Days' prior notice to the Facility Agent for such Facility, select, provided that Term Loan Facility Advances with the same Advance Date shall have the same Interest Period and, save in the case of each Initial Advance made thereunder, the first Interest Period for each Advance made under a Term Loan Facility shall end on the same day as the end of the current Interest Period of any other outstanding Advance made under the same Facility.

9.3.3 If the Company fails to give such notice of its selection in relation to an Interest Period, the duration of such Interest Period shall, subject to this Clause 9, be 3 months.

#### - 92 -

- 9.3.4 Any Interest Period which would otherwise extend beyond:
  - (a) a Repayment Date (in the case of any Interest Period relating to an Advance under the Term Facility); or
  - (b) the Revolving Credit Facility Termination Date (in the case of any Interest Period relating to an Advance under a Revolving Credit Facility),

shall be of such duration that it shall end on such date.

- 9.3.5 If two or more Interest Periods relating to Advances in the same currency under the same Term Loan Facility end at the same time, then, on the last day of such Interest Periods, such Advances shall be consolidated into and treated as a single Advance.
- 9.3.6 The Company shall use reasonable efforts to at all times select the duration of Interest Periods so as to ensure that, in respect of such of the Advances outstanding under the Facilities as is from time to time equal to the Notional Amounts specified in the Hedging Agreements, the Interest Payment Dates for such Advances coincide with (and are no more frequent than) the selected dates for payment of amounts to the Company under the Hedging Agreements.
- 9.3.7 Any Interest Period which would end on a day which is not a Business Day shall be extended to the next Business Day in that calendar month (if there is one) or the preceding Business Day (if there is not).
- 9.3.8 Interest on an Advance shall accrue from and including the first day of an Interest Period relating to such Advance up to but excluding the last day of such Interest Period.

## 9.4 **Default interest**

- 9.4.1 If the Company fails to pay any amount payable by it under a Senior Finance Document on its due date, interest shall accrue on such Unpaid Sum from the due date up to the date of actual payment (both before and after judgment) at a rate which, subject to sub-clause 9.4.2 below, is 2% higher than the rate which would have been payable if the Unpaid Sum had, during the period of non-payment, constituted an Advance under the relevant Facility Agreement in the currency of the overdue amount (or, where there is no such relevant Facility Agreement, an Advance in the relevant currency under the Term Facility Agreement) for successive Interest Periods, each of a duration selected by the relevant Facility Agent (each acting reasonably). Any interest accruing under this sub-clause 9.4.1 shall be immediately payable by the Company on demand in writing by the Intercreditor Agent or the relevant Facility Agent.
- 9.4.2 If any Unpaid Sum consists of all or part of an Advance which became due on a day which was not the last day of an Interest Period relating to that Advance:
  - (a) the first Interest Period for that Unpaid Sum shall have a duration equal to the unexpired portion of the current Interest Period relating to that Advance; and
  - (b) the rate of interest applying to the Unpaid Sum during that first Interest Period shall be 2% per annum higher than the rate which would have applied if the overdue amount had not become due.
- 9.4.3 Default interest (if unpaid) arising on an Unpaid Sum shall be compounded with the Unpaid Sum at the end of each Interest Period applicable to that Unpaid Sum but shall remain immediately due and payable.

## 10. CHANGES TO THE CALCULATION OF INTEREST

### 10.1 Absence of quotations

Subject to Clause 10.2 (*Market disruption*), if LIBOR or HIBOR is to be determined by reference to the Reference Banks under any Facility Agreement but a Reference Bank does not supply a quotation on the Quotation Day under such Facility Agreement, the applicable LIBOR or HIBOR for the purpose of such Facility Agreement shall be determined on the basis of the quotations of the remaining Reference Banks under such Facility Agreement. Where LIBOR or HIBOR is to be determined by reference to the Reference Banks under any Facility Agreement and none or only one Reference Bank supplies a quotation, then LIBOR or, as the case may be, HIBOR shall be treated as incapable of being determined under such Facility Agreement.

## 10.2 Market disruption

- 10.2.1 If a Market Disruption Event occurs in relation to an Advance under a Facility for any Interest Period, the relevant Facility Agent shall promptly notify the Intercreditor Agent of the fact and that this Clause 10.2 is in operation and the Intercreditor Agent shall promptly notify the Company and the other Lenders.
- 10.2.2 For the purpose of this Clause 10.2, "**Market Disruption Event**" means:
  - (a) in the case of a US dollar Advance:
    - (i) it is not possible, in respect of the Facility under which such Advance is made, to determine LIBOR in accordance with the provisions of the relevant Facility Agreement and Clause 10.1 (*Absence of quotations*); or
    - (ii) before the close of business in London on the Quotation Day for the relevant Interest Period, the Intercreditor Agent has been notified by a Lender or Lenders (whose participations in such Advance exceed 50 per cent of the Advance) that the cost to it/them of obtaining matching deposits in the London interbank market would be in excess of LIBOR;
  - (b) in the case of a HK dollar Advance:
    - (i) it is not possible, in respect of the Facility under which such Advance is made, to determine HIBOR in accordance with the provisions of the relevant Facility Agreement and Clause 10.1 (*Absence of quotations*); or
    - before the close of business in Hong Kong on the Quotation Day for the relevant Interest Period, the
       Intercreditor Agent has been notified by a Lender or Lenders (whose participations in such Advance exceed
       per cent of the Advance) that the cost to it/them of obtaining matching deposits in the Hong Kong interbank
       market would be in excess of HIBOR.

- 94 -

- 10.2.3 Within five Business Days of the Intercreditor Agent notifying the Company in accordance with sub-clause 10.2.1 above, the Company and the Intercreditor Agent shall enter into good faith negotiations for a period of up to thirty days with a view to agreeing an alternative basis for determining the rate of interest applicable to the relevant Advances. Any alternative basis agreed shall be binding on all parties hereto until (subject to the terms of such agreement) the Market Disruption Event referred to in sub-clause 10.2.1 above is at an end and the Intercreditor Agent has notified the Facility Agents and the Company accordingly.
- 10.2.4 If no alternative basis is agreed pursuant to sub-clause 10.2.3 above by the earlier of (i) the thirty-day period provided in sub-clause 10.2.3 above and (ii) the Advance Date (where the notification under sub-clause 10.2.1 applies to any Advance which has not been made) or the last day of the Interest Period (where the notification under sub-clause 10.2.1 applies to an Advance which is outstanding), then each Lender participating in the relevant Advance shall, acting reasonably, certify an alternative basis for maintaining its participation in the relevant Advance which may include an alternative method of fixing the interest rate, alternative Interest Periods or alternative currencies but such alternative basis must reflect its cost of funding its participation in the relevant Advance from whatever sources it may in good faith select plus the applicable interest margin applicable to that Lender's participation in the relevant Advance. Each alternative basis so certified shall be binding on the Company and the certifying Lender and treated as part of this Agreement and the relevant Facility Agreement.

## 10.3 Break Costs

- 10.3.1 The Company shall, within three Business Days of demand by a Senior Secured Creditor, pay to that Senior Secured Creditor its Break Costs attributable to all or any part of an Advance or Unpaid Sum being paid by the Company on a day other than an Interest Payment Date for that Advance or Unpaid Sum.
- 10.3.2 Each Lender shall, as soon as reasonably practicable after a demand by the Intercreditor Agent or the Company, provide a certificate confirming the amount and providing reasonable supporting evidence of its Break Costs for any Interest Period in which they accrue.

- 95 -

## 11. TAX GROSS UP AND INDEMNITIES

#### 11.1 Construction

Unless a contrary indication appears, in this Clause 11 a reference to "determines" or "determined" means a determination made in the absolute discretion of the Person making the determination.

#### 11.2 Tax gross-up

- 11.2.1 The Company shall make all payments to be made by it under the Senior Finance Documents without any Tax Deduction, unless a Tax Deduction is required by law.
- 11.2.2 The Company or a Senior Secured Creditor shall promptly upon becoming aware that it must make a Tax Deduction (or that there is any change in the rate or the basis of a Tax Deduction) notify the Intercreditor Agent accordingly. Similarly, a Senior Secured Creditor shall notify the Intercreditor Agent on becoming so aware in respect of a payment payable to that Senior Secured Creditor. If the Intercreditor Agent receives such notification from a Senior Secured Creditor it shall promptly notify the Company.
- 11.2.3 If a Tax Deduction is required by law to be made by the Company, the amount of the payment due from the Company shall be increased to an amount which (after making any Tax Deduction) leaves an amount equal to the amount which would have been due if no Tax Deduction had been required.
- 11.2.4 If the Company is required to make a Tax Deduction, it shall make that Tax Deduction and any payment required in connection with that Tax Deduction within the time allowed and in the minimum amount required by law.
- 11.2.5 Within 30 days of making either a Tax Deduction or any payment required in connection with that Tax Deduction, the Company shall deliver to the Intercreditor Agent the payment evidence reasonably satisfactory to the relevant Senior Secured Creditor that the Tax Deduction has been made or (as applicable) any appropriate payment paid to the relevant taxing authority (*provided* that the Company shall not be obliged to provide any such evidence from a Governmental Authority to the extent that it is not provided by such Governmental Authority).

## 11.3 Tax indemnity

11.3.1 The Company shall (within fifteen days of demand by the Intercreditor Agent) pay to a Senior Secured Creditor an amount equal to the loss, liability or cost which that Senior Secured Creditor determines has been (directly or indirectly) suffered for or on account of Tax by that Senior Secured Creditor in respect of a Senior Finance Document including Tax arising on payment of any premia or other sums payable on an Ancillary Finance Document whether or not such payment is required to be made by such Senior Secured Creditor.

- 96 -

- 11.3.2 Sub-clause 11.3.1 above shall not apply:
  - (a) with respect to any Tax assessed on a Senior Secured Creditor:
    - (i) under the law of the jurisdiction in which that Senior Secured Creditor is incorporated or, if different, the jurisdiction (or jurisdictions) in which that Senior Secured Creditor is treated as resident for tax purposes; or
    - (ii) under the law of the jurisdiction in which that Senior Secured Creditor's Facility Office is located in respect of amounts received or receivable in that jurisdiction,

if that Tax is imposed on or calculated by reference to the net income received or receivable (but not any sum deemed to be received or receivable) by that Senior Secured Creditor; or

- (b) to the extent a loss, liability or cost is compensated for by an increased payment under Clause 11.2 (*Tax gross-up*).
- 11.3.3 A Senior Secured Creditor making, or intending to make a claim under sub-clause 11.3.1 above shall promptly notify the Intercreditor Agent of the event which shall give, or has given, rise to the claim, following which the Intercreditor Agent shall promptly notify the Company.
- 11.3.4 A Senior Secured Creditor shall, on receiving a payment from the Company under this Clause 11.3, notify the Intercreditor Agent.
- 11.3.5 Each Senior Secured Creditor shall, as soon as practicable after a demand by the Intercreditor Agent, provide a certificate confirming the amount of the loss, liability or cost referred to in sub-clause 11.3.1 above and the basis thereof.

### 11.4 Tax Credit

If the Company makes a Tax Payment and the relevant Senior Secured Creditor determines that:

- 11.4.1 a Tax Credit is attributable to that Tax Payment; and
- 11.4.2 that Senior Secured Creditor has obtained, utilised and retained that Tax Credit,

that Senior Secured Creditor shall pay an amount to the Company which that Senior Secured Creditor determines shall leave it (after that payment) in the same after-Tax position as it would have been in had the Tax Payment not been made by the Company.

## 11.5 Stamp taxes

The Company shall pay and, within fifteen days of demand, indemnify each Senior Secured Creditor against any cost, loss or liability that a Senior Secured Creditor incurs in relation to all stamp duty, registration and other similar Taxes payable in respect of any Senior Finance Document.

- 97 -

## 12. INCREASED COSTS

### 12.1 Increased costs

- 12.1.1 Subject to Clause 12.3 (*Exceptions*), the Company shall, within fifteen days of a demand by the Intercreditor Agent, pay for the account of a Lender the amount of any Increased Costs incurred by that Lender or any Affiliate of that Lender as a result of:
  - (a) the introduction of or change in (or in the interpretation, administration or application of) any law or regulation after the Signing Date;
  - (b) compliance with any request or requirement relating to the maintenance of capital or any other request from or requirement of any central bank or other fiscal, monetary, regulatory or other authority;
  - (c) the implementation or application of, or compliance with, Basel III or any law or regulation that implements or applies Basel III; or
  - (d) the implementation or application of, or compliance with, the Dodd-Frank Wall Street Reform and Consumer Protection Act of the United States of America (whether enacted, adopted or issued before, on or after the date of this Agreement) and/or all requests, rules, guidelines or directives in connection therewith (in each case whether enacted, adopted or issued before, on or after the date of this Agreement) and all amendments thereto from time to time.

### 12.1.2 In this Agreement:

### (a) "Increased Costs" means:

- (i) a reduction in the rate of return from a Facility or on a Lender's (or its Affiliate's) overall capital (including as a result of any reduction in the rate of return on capital brought about by more capital being required to be allocated by such Lender or Affiliate);
- (ii) an additional or increased cost; or
- (iii) a reduction of any amount due and payable under any Senior Finance Document,

which is incurred or suffered by a Lender or any of its Affiliates to the extent that it is attributable to that Lender's having entered into or maintaining its commitment or funding or performing its obligations under any Senior Finance Document; and

#### (b) "Basel III" means:

the agreements on capital requirements, a leverage ratio and liquidity standards contained in "Basel III: A global regulatory framework for more resilient banks and banking systems",

- 98 -

"Basel III: International framework for liquidity risk measurement, standards and monitoring" and "Guidance for national authorities operating the countercyclical capital buffer" published by the Basel Committee on Banking Supervision in December 2010, each as amended, supplemented or restated; and

(ii) any further guidance or standards published by the Basel Committee on Banking Supervision relating to "Basel III".

## 12.2 Increased cost claims

- 12.2.1 When a Senior Secured Creditor intends to make a claim pursuant to Clause 12.1 (*Increased costs*), it shall notify the Intercreditor Agent of the event giving rise to the claim, following which the Intercreditor Agent shall promptly notify the Company.
- 12.2.2 Each Senior Secured Creditor shall, as soon as practicable after a demand by the Intercreditor Agent, provide a certificate confirming the amount of its Increased Costs and the basis thereof.

## 12.3 Exceptions

Clause 12.1 (Increased costs) does not apply to the extent any Increased Cost is:

- 12.3.1 attributable to a Tax Deduction required by law to be made by the Company and compensated for by payment under Clause 11 (*Tax Gross Up and Indemnities*);
- 12.3.2 compensated for by Clause 11.3 (*Tax indemnity*) (or would have been compensated for under Clause 11.3 (*Tax indemnity*) but was not so compensated solely because one of the exclusions in sub-clause 11.3.2 of Clause 11.3 (*Tax indemnity*) applied); or
- 12.3.3 attributable to the wilful breach by the relevant Senior Secured Creditor or their Affiliates of any law or regulation.

## 13. CURRENCY AND OTHER INDEMNITIES

## 13.1 Currency Indemnity

If any Senior Secured Creditor receives an amount in respect of the Company's liability to that Senior Secured Creditor under any Senior Finance Document or if that liability is converted into a claim, proof, judgment or order in a currency other than the currency (the "**contractual currency**") in which the amount is expressed to be payable under that Senior Finance Document:

- 13.1.1 the Company shall indemnify that Senior Secured Creditor as an independent obligation against any costs, loss or liability arising out of or as a result of the conversion; and
- 13.1.2 if the amount received by that Senior Secured Creditor, when converted into the contractual currency at a market rate in the usual course of its business is less than the amount owed in the contractual currency and such is specified to the Company in reasonable detail, the Company shall, within 3 Business Days of its receipt of a written demand by such Senior Secured Creditor, pay to that Senior Secured Creditor an amount in the contractual currency equal to the deficit.

### - 99 -

## 13.2 Other Indemnities

The Company shall, within fifteen days of demand, indemnify each Senior Secured Creditor against any cost, loss or liability incurred by that Senior Secured Creditor as a result of:

13.2.1 the occurrence of any Event of Default;

- 13.2.2 funding, or making arrangements to fund, its participation in an Advance requested by the Company in an Advance Request but not made by reason of the operation of any one or more of the provisions of the Senior Finance Documents (other than by reason of default or negligence by that Senior Secured Creditor alone);
- 13.2.3 an Advance (or part of an Advance) not being prepaid in accordance with a notice of prepayment given by the Company; and
- 13.2.4 any claim concerning either Project and its participation therein to the extent that loss or liability is suffered or incurred by that Senior Secured Creditor (other than by reason of default or negligence by a Senior Secured Creditor),

provided that, prior to the delivery of an Enforcement Notice, any such cost, loss or liability shall be reasonable.

## 13.3 Indemnity to the Agents

The Company shall, within fifteen days of demand, indemnify each of the Agents against any cost, loss or liability incurred by such Agent (acting reasonably) as a result of:

13.3.1 investigating any event which it reasonably believes is a Default; or

acting or relying on any notice, request or instruction which it reasonably believes to be genuine, correct and appropriately authorised, *provided* that, prior to the delivery of an Enforcement Notice, any such cost, loss or liability shall be reasonable.

## 14. ILLEGALITY

If it becomes, or shall become, unlawful in any applicable jurisdiction for a Lender to perform any of its obligations as contemplated by the Senior Finance Documents to which it is a party or to fund or maintain its participation in any Advance:

14.1.1 that Lender shall promptly notify the Company through the Intercreditor Agent and the relevant Facility Agent upon becoming aware of that event; and

<sup>- 100 -</sup>

14.1.2 by the latest date necessary to ensure compliance with the relevant law or regulation:

- (a) if the relevant Facility Agent so requires, the Company shall prepay that Lender's participation in all the Advances (or such lesser amount if required to comply with the relevant law or regulation) together with all other relevant amounts payable by it to that Lender under the Senior Finance Documents to which it is a party; and
- (b) that Lender's undrawn Available Commitment (or such lesser amount if permitted by the relevant law or regulation) shall be cancelled.

## 15. MITIGATION BY THE SENIOR SECURED CREDITORS

#### 15.1 Mitigation

- 15.1.1 Each Lender shall, in consultation with the Company, take all reasonable steps to mitigate or remove any circumstances which arise and which would result in any amount becoming payable under or pursuant to, or cancelled pursuant to, any of Clause 10 (*Changes to the calculation of interest*), Clause 11 (*Tax Gross Up and Indemnities*), Clause 12 (*Increased Costs*) or Clause 14 (*Illegality*) including (but not limited to) transferring its rights and obligations under the Senior Finance Documents to another Affiliate or Facility Office.
- 15.1.2 Sub-clause 15.1.1 above does not in any way limit the obligations of any Obligor under the Senior Finance Documents.

#### 15.2 Indemnity by Company

- 15.2.1 The Company shall indemnify each Senior Secured Creditor for all costs and expenses reasonably incurred by that Senior Secured Creditor as a result of steps taken by it under Clause 15.1 (*Mitigation*).
- 15.2.2 A Senior Secured Creditor is not obliged to take any steps under Clause 15.1 (*Mitigation*) if, in the opinion of that Senior Secured Creditor (acting reasonably), to do so might be prejudicial to it.

#### 16. FEES, COSTS AND EXPENSES

### 16.1 Agency Fees

The Company shall pay to each Agent for its own account a fee in amounts and on dates separately agreed between that Agent and the Company in the relevant Fee Letter.

- 101 -

### 16.2 Transaction expenses

The Company shall, within fifteen days of receipt of a written demand, pay the Agents the amount of all reasonable costs and expenses (including legal fees) incurred by any of them in connection with the review, negotiation, preparation, printing and execution of:

- (a) this Agreement, the other Senior Finance Documents and any other documents referred to herein or therein; and
- (b) any other Senior Finance Documents executed after the Signing Date,

in accordance with, in the case of any fees, costs and expenses of the Technical Adviser, Insurance Adviser and legal advisers appointed on or prior to the Signing Date, the appointment or engagement letters (if any) executed by the Company on or prior to the Signing Date.

### 16.3 Amendment costs

If the Company or any other Obligor requests an amendment, waiver or consent under any Senior Finance Document, the Company shall, within thirty days of demand, reimburse the Agents for the amount of all costs and expenses (including legal fees) reasonably incurred by the Agents in responding to, evaluating, negotiating or complying with that request.

## 16.4 Enforcement costs

The Company shall, within fifteen days of written demand, pay to each Senior Secured Creditor the amount of all costs and expenses (including legal fees) incurred by that Senior Secured Creditor in connection with the enforcement of, or the preservation of, any rights under and in accordance with any Senior Finance Document *provided* that, prior to the delivery of an Enforcement Notice, such costs and expenses shall be reasonable.

## 17. **REPRESENTATIONS AND WARRANTIES**

## 17.1 Matters represented

The Company makes the representations and warranties set out in Schedule 4 (*Representations and Warranties*) to each Senior Secured Creditor as at each of the dates specified in Clause 17.2 (*Timing*).

## 17.2 Timing

- 17.2.1 Each of the representations and warranties set out in Schedule 4 (*Representations and Warranties*) are deemed to be made by the Company (by reference to the facts and circumstances then existing) on the Fourth Amendment Signing Date.
- 17.2.2 Unless otherwise stated to have been made as of a specific date, each of the representations and warranties set out in Schedule 4 (*Representations and Warranties*) is made by the Company (with reference to the facts and circumstances then existing) on the Fourth Amendment Effective Date and (other than the representations and warranties set out in paragraphs 10.5 (*Taxes*), 21.1 to 21.6 (inclusive) (*Subsidiaries and Beneficial Interest*), 36 (*Wynn Asia 2*), 37 (*Sanctions*), 38 (*Anti-bribery conduct*) and 39 (*Money-Laundering*)

- 102 -

*Laws*) of Schedule 4) is deemed to be repeated by the Company on each subsequent Advance Date (with reference to the facts and circumstances then existing), as if any reference therein to any Senior Finance Document in respect of which any amendment, acknowledgement, confirmation, consolidation, novation, restatement, replacement or supplement is expressed to be made by any of the documents referred to in Clause 1.3 (*Security Documents*) of the Common Terms Agreement Fourth Amendment Agreement or by the Common Terms Agreement Fourth Amendment Agreement included, to the extent relevant, such document and the Senior Finance Document as so amended, acknowledged, confirmed, consolidated, novated, restated, replaced or supplemented.

### 18. COVENANTS

## 18.1 Content

The Company undertakes to each of the Senior Secured Creditors that it shall comply with the covenants set out in Schedule 5 (Covenants).

## 18.2 Duration

The covenants in Schedule 5 (*Covenants*) shall remain in force from the Signing Date until the Senior Secured Indebtedness has been fully discharged.

## 18.3 Permitted Cotai Reorganisation; Release of Palo Security

- 18.3.1 Notwithstanding any other provision of this Agreement or the other Senior Finance Documents to the contrary, the Company shall be permitted to undertake a Permitted Cotai Reorganisation. If, for the purposes of carrying out such Permitted Cotai Reorganisation, where Palo has created a Lien over any of its assets or business under the Palo Security Documents (save for the Palo Share Pledge) in favour of the Security Agent and such Lien is required (in order to effect such Permitted Cotai Reorganisation) to be released, the Security Agent shall, at the cost and request of the Company, promptly release such Liens.
- 18.3.2 The Company may, in writing to the Intercreditor Agent, request that Palo be the subject of a voluntary liquidation, winding up or dissolution (or similar action) after a Permitted Cotai Reorganisation. Such written request from the Company shall also confirm that Palo has no (and shall not have any) assets, no Default is continuing or would result from such voluntary liquidation, winding up or dissolution (or similar action) and any payments or assets to be distributed as a result of such voluntary liquidation, winding up or dissolution (or similar action) shall be distributed to the Company.
- 18.3.3 Palo may, following such written request, be the subject of such voluntary liquidation, winding up or dissolution (or similar action) if the Intercreditor Agent confirms in writing that it is satisfied that the matters contemplated by paragraphs (i) to (iv) (inclusive) of the definition of "Permitted Cotai Reorganisation" in Clause 1.1 (*Definitions*) of this Agreement have occurred in a form and substance satisfactory to it (acting reasonably) and that no payment is due from Palo pursuant to the Guarantee or any Palo Security Document.

- 103 -

- 18.3.4 If Palo is or is proposed to be, pursuant to Clause 18.3.3 above, the subject of such voluntary liquidation, winding up or dissolution (or similar action) then:
  - (a) where Palo has created a Lien over any of its assets or business under the Security Documents in favour of the Security Agent, or any Lien in favour of the Security Agent was created over the shares (or equivalent) of Palo, the Security Agent shall, at the cost and request of the Company, promptly release such Liens and the Security Agent (and/or, as required, the Intercreditor Agent) shall promptly release Palo from its obligations under the Senior Finance Documents (including, without limitation, any obligations Palo has as an Obligor) and Palo shall, upon such release occurring, no longer be an Obligor for the purposes of the Senior Finance Documents;
  - (b) the releases referred to in paragraph (a) above shall not become effective until the date of such voluntary liquidation, winding up or dissolution (or similar action); and
  - (c) if such voluntary liquidation, winding up or dissolution (or similar action) does not occur in respect of Palo, the releases referred to in paragraph (a) above shall have no effect and the obligations of Palo under the Senior Finance Documents (including, without limitation, its obligations as an Obligor) and the Liens created or intended to be created by or over Palo shall continue in such force and effect (and Palo shall continue to be an Obligor for the purposes of the Senior Finance Documents) as if those releases had not been effected.

#### 19. EVENTS OF DEFAULT

## 19.1 Events of Default

Each of the events set out in Schedule 10 (Events of Default) is an Event of Default.

### 19.2 Remedies following an Event of Default

Upon the occurrence of an Event of Default and at any time thereafter whilst it is continuing or following issuance of an Enforcement Notice (as the case may be), the Intercreditor Agent shall, if so instructed by the Required Lenders, by written notice to the Company:

- 19.2.1 declare that the Available Commitments under any of the Facility Agreements be cancelled or suspended, whereupon they shall be cancelled or suspended;
- 19.2.2 declare that all or any part of the Advances, together with accrued interest, and all other amounts accrued or outstanding under the Senior Finance Documents be immediately due and payable, whereupon they shall become immediately due and payable;

- 104 -

- 19.2.3 declare that all or part of the Advances be payable on demand, whereupon they shall immediately become payable on demand by the Intercreditor Agent;
- 19.2.4 notify the Security Agent that an Event of Default has occurred and is continuing and instruct the Security Agent to issue an Enforcement Notice;
- 19.2.5 following the issue of an Enforcement Notice, require the Security Agent to take action to enforce all or any part of the Security or all or any of the Shareholder Guarantees (subject to the expiration of any cure periods contained therein), whereupon any such action shall be taken;
- 19.2.6 following the issue of an Enforcement Notice, instruct the Security Agent to require the perfection of the Liens granted pursuant to the Land Security Assignment and the Assignment of Rights;
- 19.2.7 following the issue of an Enforcement Notice, give (or require the Security Agent to give) notices regarding the payment of insurance proceeds in accordance with the terms of the Senior Finance Documents;
- 19.2.8 following the issue of an Enforcement Notice, give (or require the Security Agent to give) notice to any Account Bank in relation to the operation of the Accounts in accordance with paragraph 3.3 (*Default*) of Schedule 6 (*Accounts*); and/or
- 19.2.9 exercise any or all other remedies available at law not inconsistent with the foregoing,

provided that the foregoing shall not in any way affect the Intercreditor Agent's or the Security Agent's right to separately enforce its rights under the Senior Finance Documents.

## 19.3 **Remedies following a Wong Event**

Without prejudice to Clause 19.2 (*Remedies following an Event of Default*) or in any way affecting the Intercreditor Agent's or the Security Agent's right to separately enforce its rights under the Senior Finance Documents, upon the occurrence of a Wong Event (as defined in the Wong Share Pledge) and at any time thereafter whilst it is continuing or following issuance of an Enforcement Notice (as the case may be), the Intercreditor Agent shall, if so instructed by the Required Lenders, require the Security Agent to take action to enforce all or any part of the Security granted pursuant to the Wong Share Pledge.

## 20. APPLICATION OF ENFORCEMENT PROCEEDS

After delivery of an Enforcement Notice and notwithstanding the provisions of Schedule 6 (*Accounts*), all Enforcement Proceeds shall be applied in accordance with the Deed of Appointment and Priority and Clause 33.6 (*Application of Enforcement Proceeds*).

- 105 -

### 21. CHANGES TO THE PARTIES

#### 21.1 Binding Agreement

This Agreement shall be binding upon and enure to the benefit of each party hereto and its or any subsequent successors and transferees.

#### 21.2 Assignment and Transfer by the Company or the GCLAs

- 21.2.1 The Company may not assign, transfer, novate or dispose of any of its rights or obligations under this Agreement or the other Senior Finance Documents.
- 21.2.2 The GCLAs may not assign or transfer its rights and/or obligations under this Agreement without the prior written consent of the Company.

## 21.3 Assignment and Transfer by Agents

Each Agent may assign or transfer any of its rights and obligations under any Senior Finance Document to which it is party only in accordance with its voluntary or requested resignation under and subject to the relevant Senior Finance Document and this Agreement and then only if it first procures that its assignee or transferee executes a duly completed Agent's Deed of Accession and Finance Party Accession Undertaking (also executed, in the case of the latter, by such Agent, the Intercreditor Agent and all other parties hereto acting through the Intercreditor Agent for this purpose (the authority for which is hereby conferred on the Intercreditor Agent) and promptly delivered by the Intercreditor Agent to the Security Agent) and enters into such other acknowledgements as may be necessary or desirable to protect the Security.

## 21.4 Assignment and Transfer by Lenders

- 21.4.1 Subject to the provisions of the Facility Agreement to which it is a party and execution and delivery by the assignee or Transferee of a Finance Party Accession Undertaking, any Lender may, at any time, assign in accordance with Clause 21.5 (*Assignments by Lenders*) all or any of its rights and benefits under the Senior Finance Documents or transfer in accordance with Clause 21.6 (*Transfers by Lenders*) all or any of its rights, benefits and obligations under the Senior Finance Documents to:
  - (a) another Lender or an Affiliate of a Lender;
  - (b) any commercial bank;
  - (c) any other bank or financial institution or trust, fund or other entity which is regularly engaged in or established for the purpose of making, purchasing or investing in loans, securities or other financial assets;
  - (d) in the case of a Permitted Loan Repurchase, the Company; or
  - (e) any other entity with the consent of the Company (such consent not to be unreasonably withheld or delayed and which consent shall not be required in respect of any assignment or transfer after the occurrence of an Event of Default which is continuing),

- 106 -

*provided* that all transactional costs (including any stamp duties, transfer taxes and any costs attributable to any transfer of Security) of such assignment or transfer shall be borne by the relevant Lender or assignee or Transferee except for:

- (i) any transfer in connection with the syndication of the Facilities, all such costs of which (including those set forth in Clause 21.7 (*Assignment and Transfer Fees*)) shall be borne by the Company; and
- (ii) any transfer contemplated by the Pre-Amendment Global Transfer Agreement or the Post-Amendment Global Transfer Agreement, all such costs of which shall be borne by the Company.
- 21.4.2 Any assignment or transfer of a Lender's participations in Advances outstanding or, as the case may be, Available Commitments under:
  - (a) the Term Facility shall be in a minimum amount of USD1,000,000 or its equivalent or, if less, equal to the aggregate of such Lender's participations or Available Commitments under such Facility; or
  - (b) a Revolving Credit Facility shall be in a minimum amount of USD1,000,000 or its equivalent or, if less, equal to the aggregate of such Lender's participations or Available Commitments under such Facility.

#### 21.5 Assignments by Lenders

If any Lender assigns all or any of its rights and benefits under the Senior Finance Documents in accordance with Clause 21.4 (Assignment and Transfer by Lenders), then, unless and until the assignee has delivered:

- (a) a notice to the Intercreditor Agent (which the Intercreditor Agent shall promptly copy to the Company and the other Senior Secured Creditors) confirming in favour of the Senior Secured Creditors that it shall be under the same obligations towards each of them as it would have been under if it had been an original party hereto as a Lender and to the relevant Facility Agreement as a Term Facility Lender, Additional Lender or Revolving Credit Facility Lender (as the case may be); and
- (b) a duly completed Finance Party Accession Undertaking executed by such Lender, such assignee, the Intercreditor Agent and all other parties hereto acting through the Intercreditor Agent for this purpose (the authority for which is hereby conferred on the Intercreditor Agent and which the Intercreditor Agent shall promptly deliver to the Security Agent),

(whereupon such assignee shall become a party hereto as a "Lender" or thereto as a "Term Facility Lender", "Additional Lender" or "Revolving Credit Facility Lender"), the Company and the Senior Secured Creditors shall not be obliged to recognise such assignee as having the rights against each of them which it would have had if it had been such a party hereto or thereto. This Clause 21.5 shall not apply to the Company's acquisition of an Advance pursuant to a Permitted Loan Repurchase.

- 107 -

### 21.6 Transfers by Lenders

Except in the case of a Permitted Loan Repurchase, if any Lender wishes to transfer all or any of its rights, benefits and/or obligations under this Agreement and the corresponding rights, benefits and/or obligations under the other Senior Finance Documents as contemplated in Clause 21.4 (*Assignment and Transfer by Lenders*), then such transfer shall only be effective if the procedure set out in this Clause 21.6 is complied with. Such transfer shall be effected by the delivery to the Intercreditor Agent (which the Intercreditor Agent shall promptly copy to the Company and the other Senior Secured Creditors) of:

- (a) a duly completed Novation Certificate executed by such Lender, the relevant Transferee, the Intercreditor Agent and all other parties hereto acting through the Intercreditor Agent for this purpose (the authority for which is hereby conferred on the Intercreditor Agent); and
- (b) a duly completed Finance Party Accession Undertaking executed by such Lender, the relevant Transferee, the Intercreditor Agent and all other parties hereto acting through the Intercreditor Agent for this purpose (the authority for which is herby conferred on the Intercreditor Agent and which the Intercreditor Agent shall promptly deliver to the Security Agent),

in which event, on the later of the Transfer Date specified in such Novation Certificate and the fifth Business Day after (or such earlier Business Day endorsed by the Intercreditor Agent on such Novation Certificate falling on or after) the date of delivery of such Novation Certificate and Finance Party Accession Undertaking to the Intercreditor Agent:

- 21.6.1 to the extent that in such Novation Certificate the Lender party thereto seeks to transfer by novation its rights, benefits and obligations under this Agreement and the corresponding rights, benefits and obligations under the other Senior Finance Documents, the Company and such Lender shall be released from further obligations towards one another under this Agreement and the corresponding rights, benefits and/or obligations under the other Senior Finance Documents and their respective rights against one another shall be cancelled (such rights and obligations being referred to in this Clause 21.6 as "discharged rights and obligations");
- 21.6.2 each of the Company and the Transferee shall assume obligations towards one another and/or acquire rights against one another which differ from such discharged rights and obligations only insofar the Company and such Transferee have assumed and/or acquired the same in place of such other party and such Lender;
- 21.6.3 the Agents, the GCLAs, such Transferee and the other Lenders shall acquire the same rights and benefits and assume the same obligations between themselves as they would have acquired and assumed had such Transferee been an original party to this Agreement and the other relevant Senior Finance



Documents as a Lender with the rights, benefits and/or obligations acquired or assumed by it as a result of such transfer and to that extent the Agents, the GCLAs and the relevant Lender shall each be released from further obligations to each other under this Agreement and the other relevant Senior Finance Documents; and

21.6.4 such Transferee shall become a party hereto as a "Lender" and to the relevant Facility Agreement as a "Term Facility Lender", "Additional Lender" or "Revolving Credit Facility Lender" (as the case may be).

### 21.7 Assignment and Transfer Fees

On the date upon which an assignment takes effect pursuant to Clause 21.5 (*Assignments by Lenders*) or a transfer takes effect pursuant to Clause 21.6 (*Transfers by Lenders*), the relevant assignee or Transferee shall pay to the Intercreditor Agent for its own account a fee of USD2,000.

### 21.8 Disclosure of Information

Any Senior Secured Creditor may disclose to any of its Affiliates and any other Person:

- 21.8.1 to (or through) whom such Senior Secured Creditor assigns or transfers (or may potentially assign or transfer) all or any of its rights, benefits and obligations in accordance with the Senior Finance Documents;
- 21.8.2 in the case of a Lender, with (or through) whom such Lender enters into (or may potentially enter into) any sub-participation in relation to the Senior Finance Documents or any Obligor; or
- 21.8.3 to whom information may be required to be disclosed by any applicable law or pursuant to any regulatory or stock exchange requirement;

such information about any Obligor, the Projects, the Cotai Project and the Senior Finance Documents as such Senior Secured Creditor may consider appropriate, *provided* that the Person to whom such information is provided under sub-clause 21.8.1 or 21.8.2 first enters into a Confidentiality Undertaking (or, in the case of the Security Agent, the confidentiality undertaking referred to in clause 18.9 (*Disclosure of Information by Security Agent*) of the Deed of Appointment and Priority) and that the Company has received a copy of such signed undertaking.

# 21.9 Change of Facility Office

Any Lender may change its Facility Office *provided* that the Company shall have no liability (or no increase in liability) under Clause 11 (*Tax gross-up and Indemnities*) or Clause 12 (*Increased costs*) which would not exist as at the date of such change but for such change, unless such change was requested by the Company pursuant to Clause 15 (*Mitigation by the Senior Secured Creditors*).

- 109 -

### 21.10 Permitted Loan Repurchases

Notwithstanding anything to the contrary contained in this Agreement or the other Senior Finance Documents, the Company shall be permitted to acquire Advances pursuant to a Permitted Loan Repurchase so long as any Advances so acquired are cancelled and retired immediately upon such Permitted Loan Repurchase becoming effective. For all purposes under this Agreement and the other Senior Finance Documents, upon a Permitted Loan Repurchase becoming effective, any Advances acquired by the Company pursuant to such Permitted Loan Repurchase (i) shall be deemed not to be outstanding and to have no principal amount (or any other amount owing in respect thereof) and (ii) shall be deemed to be automatically cancelled and retired without any further action by the Company, the Intercreditor Agent, the Lenders or any other Person; *provided*, however, that the Company shall take such actions and execute such documents and agreements as may be reasonably requested by the Intercreditor Agent to further evidence such cancellation and retirement. For purposes of clarification, Permitted Loan Repurchases shall not constitute payments (or prepayments) of Advances for any purpose hereunder.

### 22. HEDGING COUNTERPARTIES

## 22.1 Accession

Each Hedging Counterparty shall execute and deliver to the Intercreditor Agent a Hedging Counterparty's Deed of Accession and shall execute and deliver to the Security Agent in accordance with the Deed of Appointment and Priority a Finance Party Accession Undertaking. A Hedging Counterparty may, at any time, assign all or any of its rights and benefits or transfer all or any of its rights, benefits and obligations under and in accordance with the Senior Finance Documents subject to delivery to the Intercreditor Agent of a duly completed:

- (a) Hedging Counterparty's Deed of Accession executed by the assignee or transferee; and
- (b) Finance Party Accession Undertaking executed by the assignee or transferee, the Hedging Counterparty, the Intercreditor Agent and all other parties hereto acting through the Intercreditor Agent for this purpose (the authority for which is hereby conferred on the Intercreditor Agent and which the Intercreditor Agent shall promptly deliver to the Security Agent).

### 22.2 Interest in the Security

The obligations of the Company owed to each Hedging Counterparty shall be secured by the Security and each Hedging Counterparty shall be entitled to share in the Enforcement Proceeds in accordance with the Deed of Appointment and Priority and Clause 33.6 (*Application of Enforcement Proceeds*).

- 110 -

## 22.3 Voting rights

Nothing in this Clause 22 nor any other provisions of any Senior Finance Document shall be deemed to entitle any Hedging Counterparty in its capacity as such under any Hedging Agreement to exercise any voting, consent, approval or similar right under the Senior Finance Documents (other than the Hedging Agreements) including any right to participate in any Decision *provided* that:

- 22.3.1 each Hedging Counterparty shall have the right to participate in all Decisions after the occurrence of a Hedging Voting Right Event in relation to such Hedging Counterparty that is continuing; and
- 22.3.2 the consent of all Hedging Counterparties shall be required for any change to the matters referred to in paragraphs (a), (b), (f), (g), (h), (i) and (j) in the definition of "Fundamental Term" in Clause 1.1 (*Definitions*) and for any amendment to Clause 33.6 (*Application of Enforcement Proceeds*) and this Clause 22.

## 22.4 Restrictions on Amendment

Each Hedging Counterparty agrees that, except with the prior written consent of the Intercreditor Agent, no amendment may be made to a Hedging Agreement to an extent which would result in:

- 22.4.1 any payment under that Hedging Agreement being required to be made by the Company on any date other than the dates originally provided for in that Hedging Agreement; or
- 22.4.2 the Company becoming liable to make an additional payment under any Hedging Agreement which liability does not arise from the original provisions of that Hedging Agreement; or
- 22.4.3 the Company becoming liable to make any payment under that Hedging Agreement in any currency other than in the currency provided for under the original provisions of that Hedging Agreement.

### 22.5 Restrictions on Termination

No Hedging Counterparty may terminate a hedging facility or close out any hedging transaction under a Hedging Agreement prior to its stated maturity except in accordance with the terms of the ISDA Master Agreement and the ISDA Schedule (each as may be amended pursuant to paragraph 4 of Schedule 8 (*Hedging Arrangements*).

## 22.6 Termination at request of Intercreditor Agent

After a notice has been given by the Intercreditor Agent pursuant to sub-clause 19.2.2 of Clause 19.2 (*Remedies following an Event of Default*), a Hedging Counterparty shall, at the written request of the Intercreditor Agent, terminate the hedging facility or close out any hedging transaction under the Hedging Agreement to which it is party in accordance with the terms of such Hedging Agreement.

## 23. AGENTS AND GLOBAL COORDINATING LEAD ARRANGERS

## 23.1 Appointment and duties of the Agents

- 23.1.1 Each of:
  - (a) the Senior Secured Creditors appoints the Intercreditor Agent;

- 111 -

- (b) the Term Facility Lenders appoints the Term Facility Agent;
- (c) the Revolving Credit Facility Lenders appoints the Revolving Credit Facility Agent; and
- (d) the Additional Lenders under each Additional Lender Facility Agreement appoint the applicable Additional Facility Agent as facility agent for those Additional Lenders (or its successor approved in accordance with this Agreement),

to act as its agent under and in connection with the Senior Finance Documents and irrevocably authorises it on its behalf to perform the duties and to exercise the rights, powers and discretions that are specifically delegated to it under or in connection with the Senior Finance Documents, together with any other incidental rights, powers and discretions.

- 23.1.2 None of the Agents may begin any legal action or proceeding in the name of a Senior Secured Creditor (other than itself) without that Senior Secured Creditor's consent.
- 23.1.3 Each Agent has only those duties which are expressly specified in the Senior Finance Documents, and those duties are solely of a mechanical and administrative nature.

## 23.2 Relationship

- 23.2.1 The relationship between each Agent and the relevant Senior Secured Creditors is that of principal and agent only. Nothing in this Agreement constitutes any Agent as trustee or fiduciary for any other Person and no Agent need hold in trust any moneys paid to it for a Person or be liable to account for interest on those moneys except to the extent expressly stated in a Senior Finance Document.
- 23.2.2 No Agent shall in any respect be the agent of the Company by virtue of this Agreement.
- 23.2.3 No Agent shall be liable to the Company for any breach by any other Senior Secured Creditor of any Senior Finance Document or be liable to any other Secured Creditor for any breach by the Company of the Senior Finance Documents.

### 23.3 Role of the GCLAs

Except as specifically provided in the Senior Finance Documents, none of the GCLAs has any obligations of any kind to any other Party under or in connection with any Senior Finance Document.

## 23.4 Delegation

Each Agent may act through its personnel and agents.

- 112 -

## 23.5 Instructions

- 23.5.1 Unless otherwise expressly provided in the Senior Finance Documents, the Intercreditor Agent shall act (and shall be fully protected if it so acts) in accordance with the instructions of the Required Lenders in connection with the exercise of any right, power or discretion under or in connection with the Senior Finance Documents.
- 23.5.2 Each Facility Agent shall be fully protected if it acts in accordance with the instructions of its Lending Group in connection with the exercise of any right, power or discretion under or in connection with any matter not expressly provided for in the Senior Finance Documents.
- 23.5.3 In the absence of such instructions each Agent may act, subject to the terms of the Senior Finance Documents, as that Agent, in its sole discretion, considers to be in the best interests of all the Senior Secured Creditors or, in the case of each Facility Agent, its Lending Group.

## 23.6 Discretions

Notwithstanding any provision of the Senior Finance Documents, each Agent may:

- 23.6.1 assume, unless it has, in its capacity as Agent, received written notice to the contrary from any other Party, that (a) any representation made or deemed to be made by an Obligor in connection with the Senior Finance Documents is true, (b) no Default has occurred, (c) no Obligor is in breach of or default under its obligations under the Senior Finance Documents and (d) any right, power, authority or discretion vested in the Senior Finance Documents upon the Required Lenders, a Lending Group, the Lenders or any other Person or group of Persons has not been exercised;
- 23.6.2 assume that (a) the Facility Office of each Lender is that notified to it by such Lender in writing and (b) the information provided by each Lender pursuant to Clause 29 (*Notices*) is true and correct in all respects until it has received from such Lender notice of a change to the Facility Office or any such information and act upon any such notice until the same is superseded by a further notice;
- 23.6.3 engage and pay for the advice or services of any lawyers, accountants, surveyors or other experts whose advice or services may to it seem necessary, expedient or desirable and rely upon any advice so obtained;
- 23.6.4 rely as to any matters of fact which might reasonably be expected to be within the knowledge of an Obligor upon a certificate signed by or on behalf of such Obligor;
- 23.6.5 rely upon any communication or document believed by it to be genuine;
- 23.6.6 refrain from exercising any right, power or discretion vested in it as Agent under the Senior Finance Documents unless and until instructed as described in Clause 23.5 (*Instructions*) as to whether or not such right, power or discretion is to be exercised and, if it is to be exercised, as to the manner in which it should be exercised;

- 113 -

- 23.6.7 refrain from acting in accordance with any instructions to begin any action or proceeding arising out of or in connection with the Senior Finance Documents until it shall have received such security as it may require (whether by way of payment in advance or otherwise) for all costs, claims, losses, expenses (including legal fees) and liabilities which it shall or may expend or incur in complying with such instructions;
- 23.6.8 refrain from acting where to do so would put it in breach of an applicable Legal Requirement;
- 23.6.9 treat each Facility Agent as the duly appointed and authorised agent of the relevant Lenders until it receives written notice to the contrary from the relevant Lenders; and
- 23.6.10 (in the case of the Intercreditor Agent) in applying any moneys received by it under any Security Document under Clause 33.6 (*Application of Enforcement Proceeds*), rely on any certificate made by the relevant Facility Agent or Hedging Counterparty as to the identity of, and the amounts owing to, any of the Senior Secured Creditors and shall be protected in so relying.

## 23.7 Agents' Obligations

23.7.1 Each Agent shall:

- (a) promptly inform each Senior Secured Creditor (in the case of the Intercreditor Agent) or the Intercreditor Agent and the Lenders in its Lending Group (in the case of each Facility Agent) of the contents of any notice or document received by it pursuant to the terms of any Senior Finance Document in its capacity as Agent from the Security Agent or an Obligor under the Senior Finance Documents; and
- (b) promptly notify each Senior Secured Creditor (in the case of the Intercreditor Agent) or the Intercreditor Agent and the Lenders in its Lending Group (in the case of each Facility Agent) of the occurrence of any Event of Default or any default by an Obligor in the due performance of or compliance with its obligations under the Senior Finance Documents of which such Agent has notice from any other party.
- 23.7.2 The Intercreditor Agent shall promptly inform the Security Agent of the occurrence of the Release Date.

### 23.8 Excluded Obligations

Notwithstanding anything to the contrary expressed or implied herein, none of the Agents nor any of the GCLAs shall:

23.8.1 be bound to enquire as to (a) whether or not any representation made or deemed to be made by an Obligor in connection with the Senior Finance Documents is true, (b) the occurrence or otherwise of any Default, (c) the performance by an Obligor of its obligations under the Senior Finance Documents or (d) any breach of or default by an Obligor of or under its obligations under the Senior Finance Documents;

- 114 -

23.8.2 be bound to account to any Senior Secured Creditor for any sum or the profit element of any sum received by it for its own account;

- 23.8.3 be bound to disclose to any other Person any information relating to any Obligor, any party to a Project Document or any of their respective related entities if (a) such Person, on providing such information, expressly stated to such Agent or, as the case may be, such GCLA, that such information was confidential or (b) such disclosure would or might in its opinion constitute a breach of any law or be otherwise actionable at the suit of any Person; or
- 23.8.4 be under any obligations other than those for which express provision is made herein or in any other Senior Finance Document to which such Agent or GCLA is a party.

## 23.9 Exclusion of Liabilities

None of the Agents and the GCLAs accepts any responsibility:

- 23.9.1 for the adequacy, accuracy and/or completeness of the Information Memorandums or any other information supplied by the Agents or the GCLAs, by an Obligor or by any other Person in connection with the Senior Finance Documents, the transactions therein contemplated or any other agreement, arrangement or document entered into, made or executed in anticipation of, pursuant to or in connection with the Senior Finance Documents;
- 23.9.2 for the legality, validity, effectiveness, adequacy or enforceability of the Senior Finance Documents or any other agreement, arrangement or document entered into, made or executed in anticipation of, pursuant to or in connection with the Senior Finance Documents; or
- 23.9.3 for the exercise of, or the failure to exercise, any judgement, discretion or power given to any of them by or in connection with the Senior Finance Documents or any other agreement, arrangement or document entered into, made or executed in anticipation of, pursuant to or in connection with the Senior Finance Documents.

Accordingly, none of the Agents and the GCLAs shall be under any liability (whether in negligence or otherwise) in respect of such matters.

No party hereto (other than the applicable Agent or GCLA) may take any proceedings against any officer, employee or agent of an Agent or a GCLA in respect of any claim it might have against such Agent or such GCLA, as the case may be, or in respect of any act or omission of any kind by that officer, employee or agent (other than by reason of the fraud, gross negligence or wilful misconduct of such officer, employee or agent) in relation to any Senior Finance Document and any officer, employee or agent of such Agent or GCLA may rely on this Clause subject to Clause 1.5 (*Third party rights*) and the provisions of the Third Parties Act.

- 115 -

## 23.10 No Actions

Each of the Lenders and the Hedging Counterparties agrees that it shall not assert or seek to assert against any director, officer or employee of any of the Agents or any of the GCLAs any claim it might have against any of them.

## 23.11 Business with the Obligors

Each Agent and GCLA may accept deposits from, lend money to and generally engage in any kind of banking or other business with any of the Obligors or their Affiliates.

## 23.12 Resignation

- 23.12.1 An Agent may resign its appointment hereunder at any time without assigning any reason therefor by giving not less than 25 Business Days' prior notice to that effect to the Senior Secured Creditors and the Company, in which case:
  - (a) the Required Lenders (in the case of the Intercreditor Agent) may appoint a successor Agent and, unless an Event of Default has occurred and is continuing, such appointment shall be subject to the prior agreement of the Required Lenders and the Company or, failing such agreement within 15 Business Days after such notice of resignation, the Required Lenders may appoint a successor Agent on the same terms and conditions as previously applied to the outgoing Agent; and
  - (b) the relevant Lending Group under a Facility (in the case of a Facility Agent for that Facility) may appoint a successor Agent and, unless an Event of Default has occurred and is continuing, such appointment shall be subject to the prior agreement of such Lending Group and the Company or, failing such agreement within 15 Business Days after such notice of resignation, the Lending Group may appoint a successor Agent on the same terms and conditions as previously applied to the outgoing Agent.
- 23.12.2 If the Required Lenders have not, within 15 Business Days after notice of resignation, appointed a successor Intercreditor Agent which accepts the appointment, the outgoing Agent may appoint a successor Agent and, unless an Event of Default has occurred and is continuing, such appointment shall be subject to the prior agreement of the Intercreditor Agent and the Company or, failing such agreement within 25 Business Days after notice of resignation, the Intercreditor Agent may appoint a successor Agent on the same terms and conditions as previously applied to it.
- 23.12.3 If a Lending Group has not, within 15 Business Days after notice of resignation appointed a successor Facility Agent which accepts the appointment, the outgoing Facility Agent may appoint a successor Facility Agent and, unless an Event of Default has occurred and is continuing, such appointment shall be subject to the prior agreement of the outgoing Facility Agent and the Company or, failing such agreement within 25 Business Days after notice of resignation, the outgoing Facility Agent may appoint a successor Facility Agent on the same terms and conditions as previously applied to it.

- 116 -

- 23.12.4 If, at the time of expiry of the period specified in sub-clause 23.12.2 or, as the case may be, sub-clause 23.12.3 above, the outgoing Agent cannot find a successor owing to the unwillingness of any proposed successor to accept the terms and conditions which apply to the outgoing Agent, the Company shall offer to any proposed successor such terms and conditions as are consistent with the role to be performed, taking into account the current market for the performance of such duties and the then existing circumstances of the Projects.
- 23.12.5 If the Agent has not been paid an amount due to it under the Senior Finance Documents and gives notice thereof as its reason for resigning together with its notice pursuant to Clause 23.12.1, it shall not be obliged to appoint a successor. If, at the time of expiry of the period specified in clause 23.12.1, the Required Lenders or, as the case may be, the relevant Lending Group, cannot find a successor owing to the unwillingness of any proposed successor to accept the terms and conditions which apply to the outgoing Agent, the Company shall offer to any proposed successor such terms and conditions as are consistent with the role to be performed, taking into account the current market for the performance of such duties and the then existing circumstances of the Projects.
- 23.12.6 The resignation of an Agent and the appointment of any successor Agent shall both become effective only upon the successor Agent executing an Agent's Deed of Accession *provided* that, where the Agent has notified the reason for its resignation pursuant to Clause 23.12.5, its resignation shall become effective upon the expiry of the period notified by it pursuant to Clause 23.12.1. Upon the execution of an Agent's Deed of Accession, the successor Agent shall succeed to the position of the retiring Agent (as the case may be) under the Senior Finance Documents and the term "Agent" shall mean the successor Agent.
- 23.12.7 The Intercreditor Agent agrees that it shall, if so requested in writing by the Required Lenders, tender its resignation in accordance with this Clause 23.12.
- 23.12.8 Each Facility Agent agrees that it shall, if so requested in writing by its Lending Group, tender its resignation in accordance with this Clause 23.12.
- 23.12.9 Upon the appointment of a successor (or, as the case may be, its resignation becoming effective), the retiring Agent shall be discharged from any future (but not accrued) obligations in respect of the Senior Finance Documents but shall remain entitled to the benefit of Clause 13.2 (*Other Indemnities*) and sub-clauses 23.1, 23.2, 23.5.2, 23.5.3, 23.6.10, 23.8, 23.9, 23.10 and 23.15 of this Clause 23.

## 23.13 Own Responsibility

It is understood and agreed by each Senior Secured Creditor that at all times it has itself been, and shall continue to be, solely responsible for making its own independent appraisal of and investigation into all risks arising under or in connection with the Senior Finance Documents including, but not limited to:

23.13.1 the financial condition, creditworthiness, condition, affairs, status and nature of the Projects, the Cotai Project and each Obligor;

- 117 -

- 23.13.2 the legality, validity, effectiveness, adequacy and enforceability of the Senior Finance Documents and any other agreement, arrangement or document entered into, made or executed in anticipation of, pursuant to or in connection with the Senior Finance Documents:
- 23.13.3 whether such Senior Secured Creditor has recourse, and the nature and extent of that recourse, against an Obligor or any other Person or any of their respective assets under or in connection with the Senior Finance Documents, the transactions therein contemplated or any other agreement, arrangement or document entered into, made or executed in anticipation of, pursuant to or in connection with the Senior Finance Documents; and
- 23.13.4 the adequacy, accuracy and/or completeness of the Information Memorandums and any other information provided by the Agents or the GCLAs, an Obligor, or by any other Person in connection with the Senior Finance Documents, the transactions contemplated therein or any other agreement, arrangement or document entered into, made or executed in anticipation of, pursuant to or in connection with the Senior Finance Documents.

Accordingly, each Senior Secured Creditor acknowledges to the Agents and the GCLAs that it has not relied on and shall not hereafter rely on the Agents and the GCLAs or any of them in respect of any of these matters.

#### 23.14 Agency Division Separate

In acting as Agent under the Senior Finance Documents, each of the Agents shall be regarded as acting through its agency division which shall be treated as a separate entity from any other of its divisions or departments and, notwithstanding the foregoing provisions of this Clause 23, any information received by some other division or department of such Agent may be treated as confidential and shall not be regarded as having been given to such Agent's agency division.

### 23.15 Indemnity to Intercreditor Agent

23.15.1 Each Senior Secured Creditor shall rateably in accordance with the proportion that the US dollar equivalent of the sum of its Available Commitments and its participations in any outstanding Advances bear to the US dollar equivalent of the aggregate of the Available Commitments and such participations of all the Senior Secured Creditors (or, if all such amounts have been reduced to zero, such proportion determined immediately prior to such reduction) for the time being, indemnify the Intercreditor Agent, within fifteen days of demand (accompanied by reasonable written certification), against cost, loss or liability incurred by the Intercreditor Agent (other than by reason of fraud, negligence or wilful misconduct of the Intercreditor Agent) in acting as Intercreditor

- 118 -

Agent in accordance with the terms of the Senior Finance Documents (unless the Intercreditor Agent has been reimbursed by, or indemnified to its satisfaction by, an Obligor pursuant to a Senior Finance Document or otherwise in writing). For the purposes of this Clause 23.15.1, each Hedging Counterparty shall, in respect of each Hedging Agreement entered into by it, be deemed to have made an Advance to the Company in an amount equal to the Realised Hedge Loss (if any) under the Hedging Agreement to which such Hedging Counterparty is party.

- 23.15.2 Clause 23.15.1 shall not apply to the extent that the Intercreditor Agent is otherwise actually indemnified or reimbursed by any Party under any other provision of the Senior Finance Documents.
- 23.15.3 Provided that the Company is required to reimburse or indemnify the Intercreditor Agent for such cost, loss or liability in accordance with the terms of the Senior Finance Documents, the Company shall, within fifteen days of demand in writing by any Senior Secured Creditor, indemnify such Senior Secured Creditor in relation to any payment actually made by such Senior Secured Creditor pursuant to Clause 23.15.1 above.

## 24. CONDUCT OF BUSINESS BY THE SENIOR SECURED CREDITORS

No provision of the Senior Finance Documents shall:

- 24.1.1 interfere with the right of any Senior Secured Creditor to arrange its affairs (tax or otherwise) in whatever manner it thinks fit;
- 24.1.2 subject to Clause 15 (*Mitigation by Senior Secured Creditors*), oblige any Senior Secured Creditor to investigate or claim any credit, relief, remission or repayment available to it or the extent, order and manner of any claim; or
- 24.1.3 oblige any Senior Secured Creditor to disclose any information relating to its affairs (tax or otherwise) or any computations in respect of Tax.

### 25. SHARING AMONG THE SENIOR SECURED CREDITORS

## 25.1 Payments to Senior Secured Creditors

If a Senior Secured Creditor (a **"Recovering Senior Secured Creditor**") receives or recovers any amount from an Obligor other than in accordance with the provisions of the Senior Finance Documents (excluding any such provision which permits the setting off of obligations owed by such Obligor against obligations owed to it by such Recovering Senior Secured Creditor but allowing, for the avoidance of doubt, any such provision in any Hedging Agreement permitting netting off between transactions under such Hedging Agreement) and applies that amount to a payment due under the Senior Finance Documents then:

- 25.1.1 the Recovering Senior Secured Creditor shall, within 5 Business Days, notify details of the receipt or recovery, to the Intercreditor Agent;
- 25.1.2 the Intercreditor Agent shall determine whether the receipt or recovery is in excess of the amount the Recovering Senior Secured Creditor would have

- 119 -

been paid had the receipt or recovery been received and distributed in accordance with this Agreement, without taking account of any Tax which would be imposed on that Agent in relation to the receipt, recovery or distribution; and

25.1.3 the Recovering Senior Secured Creditor shall, within 10 Business Days of demand by the Intercreditor Agent, pay to the Intercreditor Agent an amount (the **"Sharing Payment**") equal to such receipt or recovery less any amount which the Intercreditor Agent determines may be retained by the Recovering Senior Secured Creditor as its share of any payment to be made, in accordance with this Agreement.

For purposes of clarification, Permitted Loan Repurchases shall not constitute payments (or prepayments) of Loans or the recovery of any amount from an Obligor for any purpose hereunder.

## 25.2 Redistribution of payments

The Intercreditor Agent shall treat the Sharing Payment as if it had been paid by the Company and distribute it between the Senior Secured Creditors (other than the Recovering Senior Secured Creditor) in accordance with this Agreement.

## 25.3 Recovering Senior Secured Creditor's Rights

- 25.3.1 On a distribution by the Intercreditor Agent under 25.2 (*Redistribution of payments*), the Recovering Senior Secured Creditor shall be subrogated to the rights of the Senior Secured Creditors which have shared in the redistribution.
- 25.3.2 If and to the extent that the Recovering Senior Secured Creditor is not able to rely on its rights under sub-clause 25.3.1 above, the Company shall be liable to the Recovering Senior Secured Creditor for a debt equal to the Sharing Payment which is immediately due and payable.

### 25.4 Reversal of redistribution

If any part of the Sharing Payment received or recovered by a Recovering Senior Secured Creditor becomes repayable and is repaid by such Recovering Senior Secured Creditor, then:

- 25.4.1 each Senior Secured Creditor which has received a share of such Sharing Payment pursuant to Clause 25.2 (*Redistribution of payments*) shall, upon request of the Intercreditor Agent, pay to the Intercreditor Agent for account of that Recovering Senior Secured Creditor an amount equal to the appropriate part of its share of the Sharing Payment (together with an amount as is necessary to reimburse that Recovering Senior Secured Creditor for its proportion of any interest on the Sharing Payment which that Recovering Senior Secured Creditor is required to pay); and
- 25.4.2 that Recovering Senior Secured Creditor's rights of subrogation in respect of any reimbursement shall be cancelled and the Company shall be liable to the reimbursing Senior Secured Creditor for the amount so reimbursed.

- 120 -

## 25.5 Exceptions

This Clause 25 shall not apply to the extent that the Recovering Senior Secured Creditor would not, after making any payment pursuant to this Clause, have a valid and enforceable claim against the relevant Obligor.

## 25.6 Benefit

The provisions of this Clause 25 are for the sole benefit of the Senior Secured Creditors and may be waived or amended by the Required Lenders without the consent of the Company provided there is no increase in the liability of the Company as a result.

### 26. PAYMENT MECHANICS

#### 26.1 Payments under the Senior Finance Documents

#### 26.1.1 **Prior to an Event of Default**

Unless and until an Event of Default has occurred and is continuing and except with respect to any payments pursuant to a Permitted Loan Repurchase:

- (a) all payments to be made by the Company to or for the account of any Lender under the Facility Agreement to which that Lender is a party shall be made to the relevant Facility Agent under that Facility Agreement for the account of that Lender, in the manner stipulated in the relevant Facility Agreement; and
- (b) all payments to be made by a Lender under a Facility Agreement shall be made to the relevant Facility Agent, not later than the time (if any) specified in the relevant Facility Agreement, to its account at such office or bank as it may notify to that Lender from time to time for this purpose.

#### 26.1.2 After the occurrence of an Event of Default

Subject to the Deed of Appointment and Priority, after the occurrence of an Event of Default that is continuing and unless the Intercreditor Agent agrees in writing that payment should continue to be made in accordance with sub-clause 26.1.1 (*Prior to an Event of Default*) and except with respect to any payments pursuant to a Permitted Loan Repurchase:

- (a) all payments to be made by the Company to or for the account of any Senior Secured Creditor or under any Senior Finance Document shall be made to the Intercreditor Agent (other than any such payments to be made to or for the account of the Security Agent which shall continue to be made to the Security Agent);
- (b) all payments to be made by any Lender under any Senior Finance Document (whether pursuant to Clause 25 (*Sharing Among the Senior Secured Creditors*) or otherwise) shall be paid to the Intercreditor Agent; and
- (c) all payments received by the Intercreditor Agent under this sub-clause 26.1.2 shall be distributed in accordance with Clause 33.6 (*Application of Enforcement Proceeds*) to the Person(s) specified therein or, where any such Person is a Lender, to the relevant Facility Agent for the account of that Lender.

- 121 -

## 26.2 Payments by an Agent

- 26.2.1 Save as otherwise provided herein, each payment received by an Agent as agent for or otherwise for the benefit of another Person shall, subject to Clause 26.3 (*Distributions to an Obligor*) and Clause 26.4 (*Clawback*), be made available by that Agent to the Person entitled to receive such payment for value the same day by transfer to such account of such Person with such bank in the principal financial centre of the country of the relevant currency as such Person shall have previously notified to that Agent.
- 26.2.2 A payment shall be deemed to have been made by an Agent on the date on which it is required to be made under the Senior Finance Documents if such Agent has, on or before that date, taken steps to make that payment in accordance with the regulations or operating procedures of the clearing or settlement system used by such Agent in order to make the payment.

### 26.3 Distributions to an Obligor

Each Agent may (with the consent of the relevant Obligor or in accordance with Clause 27 (*Set-off*)) apply any amount received by it for that Obligor in or towards payment (on the date and in the currency and funds of receipt) of any amount due from that Obligor under the Senior Finance Documents or in or towards purchase of any amount of any currency to be so applied at a market rate of exchange in its usual course of business.

## 26.4 Clawback

- 26.4.1 Where a sum is to be paid to an Agent under the Senior Finance Documents for another Person, that Agent is not obliged to pay that sum to that Person until it has been able to establish to its satisfaction that it has actually received that sum.
- 26.4.2 If an Agent pays an amount to another Person and it proves to be the case that that Agent had not actually received that amount, then the Person to whom that amount was paid by that Agent shall on demand refund the same to that Agent together with interest on that amount from the date of payment to the date of receipt by that Agent, calculated by that Agent to reflect its cost of funds.

### 26.5 No Set-off by Obligors

All payments to be made by an Obligor under the Senior Finance Documents shall be calculated and be made without (and free and clear of any deduction for) set-off or counterclaim.

- 122 -

| 26.6 | Business Days       |   |
|------|---------------------|---|
|      | 26.6.1              | Any payment which is due to be made under any Senior Finance Document on a day that is not a Business Day shall be made on the next Business Day in the calendar month (if there is one) or the preceding Business Day (if there is not). |
|      | 26.6.2              | During any extension of the due date for payment of any principal pursuant to sub-clause 26.6.1 above, interest is payable on that principal at the rate payable on the original due date.  |
| 26.7 | Currency of account |   |
|      | 26.7.1              | A repayment of an Advance or Unpaid Sum or a part of an Advance or Unpaid Sum shall be made in the currency in which the Advance or Unpaid Sum is denominated on its due date.  |
|      | 26.7.2              | Each payment of interest shall be made in the currency in which the sum in respect of which the interest is payable was denominated when that interest accrued.   |
|      | 26.7.3              | Each payment in respect of costs, expenses or Taxes under the Senior Finance Documents shall be made in the currency in which the costs, expenses or Taxes are incurred.  |
|      | 26.7.4              | Any other amount payable under any of the Senior Finance Documents is, except as otherwise provided elsewhere in the Senior Finance Documents, payable in US dollars.   |
| 27.  | SET-OFF             |   |

Without prejudice to the provisions of Schedule 6 (*Accounts*) and subject to the terms of Clause 25 (*Sharing Among the Senior Secured Creditors*) and Clause 33 (*Intercreditor Arrangements*), a Senior Secured Creditor may, upon the occurrence of an Event of Default and for so long as it is continuing, set off any matured obligations owed by the Company under the Senior Finance Documents (to the extent beneficially owned by that Senior Secured Creditor) against any obligation (which, for the purpose of this provision only, shall be treated as due and payable, save for unmatured obligations under the Hedging Agreements) owed by that Senior Secured Creditor to the Company, regardless of the place of payment, booking branch or currency of either obligation. If the obligations are in different currencies, the Senior Secured Creditor may convert either obligation at a market rate of exchange in its usual course of business for the purpose of the set-off.

## 28. NON-RECOURSE LIABILITY

Notwithstanding any provision in the Senior Finance Documents to the contrary no Operative shall be personally liable for payments due hereunder or under any of the Senior Finance Documents or for the performance of any obligation hereunder or thereunder, save, in relation to any Operative, pursuant to any Senior Finance Document to which such Operative is party. The sole recourse of the Senior Secured Creditors for satisfaction of any of the obligations of any of the Obligors hereunder and under the other Senior Finance Documents shall be against the Obligors, and not

- 123 -

against any assets or property of any Operative save to the extent such Operative is party to a Senior Finance Document and is expressed to be liable for such obligation thereunder. In the case of Mr Wong Chi Seng, his liability shall be limited to his shares in the Company.

#### 29. NOTICES

### 29.1 Communications in Writing

Any notice, demand or other communication (each, for the purposes of this Clause 29, a "**communication**") to be made under or in connection with the Senior Finance Documents shall be made in writing but, unless otherwise stated, may be made by fax or letter.

### 29.2 Addresses

The address and fax number (and the department or officer, if any, for whose attention the communication is to be made) of each party for any communication or document to be made or delivered under or in connection with the Senior Finance Documents is:

- 29.2.1 in the case of the Company, each of the GCLAs, each of the Agents, each of the Lenders and each of the Hedging Counterparties party to the Common Terms Agreement Second Amendment Agreement, identified with its name on the signing pages thereto; and
- 29.2.2 in the case of each other Lender, each other Hedging Counterparty and each other Obligor, that notified in writing to the Intercreditor Agent prior to the date it becomes a party to the Senior Finance Documents,

or any substitute address, fax number or department or officer as the party may notify to the Intercreditor Agent (or the Intercreditor Agent may notify to the other parties, if a change is made by the Intercreditor Agent) by not less than 10 Business Days' notice.

# 29.3 Delivery

- 29.3.1 Any communication or document made or delivered by one Person to another under or in connection with the Senior Finance Documents shall only be effective:
  - (a) if delivered personally or by overnight courier, when left at the relevant address;
  - (b) if by way of fax, when received in legible form; or
  - (c) if by way of letter, when it has been left at the relevant address or 10 Business Days after being deposited in the post postage prepaid in an envelope addressed to it at that address,

and, if a particular department or officer is specified as part of its address details provided under Clause 29.2 (*Addresses*), if addressed to that department or officer.

- 124 -

29.3.2 Any communication or document to be made or delivered to an Agent shall be effective only when actually received by that Agent and then only if it is expressly marked for the attention of the department or officer specified as part of its address details provided under Clause 29.2 (*Addresses*) (or any substitute department or officer as that Agent shall specify for this purpose).

29.3.3 All notices to an Obligor shall be sent through a Facility Agent or the Intercreditor Agent (but always with a copy to the Intercreditor Agent). All notices from an Obligor under the Senior Finance Documents shall be sent to the Intercreditor Agent who shall distribute them to the Senior Secured Creditors.

## 29.4 Notification of address and fax number

Promptly upon receipt of notification of an address and fax number or change of address or fax number pursuant to Clause 29.2 (*Addresses*) or changing its own address or fax number, the Intercreditor Agent shall notify the other parties.

#### 29.5 Electronic communication

- 29.5.1 Any communication to be made between an Agent and a Lender or between an Agent and another Agent under or in connection with the Senior Finance Documents may be made by electronic mail or other electronic means, if that Agent and the relevant Lender or Agent:
  - (a) agree that, unless and until notified to the contrary, this is to be an accepted form of communication;
  - (b) notify each other in writing of their electronic mail address and/or any other information required to enable the sending and receipt of information by that means; and
  - (c) notify each other of any change to their address or any other such information supplied by them.
- 29.5.2 Any electronic communication made between an Agent and a Lender or another Agent shall be effective only when actually received in readable form and in the case of any electronic communication made by a Lender to an Agent or by an Agent to another Agent only if it is addressed in such a manner as the relevant Agent shall specify for this purpose.

#### 29.6 Electronic supply of materials

(a) The Company shall (and shall ensure that each other Obligor shall), unless otherwise requested by the Intercreditor Agent, provide to the Intercreditor Agent all information, documents and other materials that such Obligor is obligated to furnish to the Intercreditor Agent pursuant to the Senior Finance Documents, including, without limitation, all notices, requests, financial statements, financial and other reports, certificates and other information materials, but excluding any such communication that (i) relates to a request for a new, or a conversion of an existing, Advance or other extension of credit (including any election of an interest rate or Interest Period relating thereto),

- 125 -

(ii) relates to the payment of any principal or other amount due under any Senior Finance Document prior to the scheduled date therefor, (iii) provides notice of any Default under any Senior Finance Document, (iv) is required to be delivered to satisfy any condition precedent to the effectiveness of any Senior Finance Document and/or any Advance or other extension of credit hereunder or (v) initiates or responds to legal process (all such non-excluded information being referred to herein collectively as the "**Communications**") by transmitting the Communications in an electronic/soft medium (provided such Communications contain any required signatures) in a format acceptable to the Intercreditor Agent to each of Loan Agency@bocmacau.com, wong iaokun@bocmacau.com and cheong hoikam@bocmacau.com (or such other e-mail address or addresses designated by the Intercreditor Agent from time to time).

- (b) Each party hereto agrees that the Intercreditor Agent may make the Communications available to the any Senior Secured Creditor by posting the Communications on IntraLinks or another relevant website, if any, to which such Senior Secured Creditor has access (whether a commercial, third-party website or whether sponsored by the Intercreditor Agent) (the "**Platform**"). Nothing in this Clause 29.6 shall prejudice the right of the Intercreditor Agent to make the Communications available to any Senior Secured Creditor in any other manner specified in this Agreement or any other Senior Finance Documents.
- (c) Each Senior Secured Creditor agrees that e-mail notice to it (at the address provided pursuant to the next sentence and deemed delivered as provided in the next paragraph) specifying that Communications have been posted to the Platform shall constitute effective delivery of such Communications to such Senior Secured Creditor for purposes of this Agreement and the other Senior Finance Documents. Each Senior Secured Creditor agrees (i) to notify the Intercreditor Agent in writing (including by electronic communication) from time to time to ensure that the Intercreditor Agent has on record an effective e-mail address for such Senior Secured Creditor to which the foregoing notice may be sent by electronic transmission and (ii) that the foregoing notice may be sent to such e-mail address.
- (d) Notwithstanding Clause 29.6(e) below, each party hereto agrees that any electronic communication referred to in this Clause 29.6 shall be deemed delivered upon the posting of a record of such communication (properly addressed to such party at the e-mail address provided to the Intercreditor Agent) as "sent" in the e-mail system of the sending party or, in the case of any such communication to the Intercreditor Agent, upon the posting of a record of such communication as "received" in the e-mail system of the Intercreditor Agent; provided that if such communication is not so received by the Intercreditor Agent during the normal business hours of the Intercreditor Agent, such communication shall be deemed delivered at the opening of business on the next Business Day for the Intercreditor Agent.
- (e) Each party hereto acknowledges that (i) the distribution of material through an electronic medium is not necessarily secure and that there are confidentiality and other risks associated with such distribution, (ii) the Communications and

- 126 -

the Platform are provided "as is" and "as available", (iii) none of the Intercreditor Agent, its affiliates nor any of their respective officers, directors, employees, agents, advisors or representatives (collectively, the "**Intercreditor Parties**") warrants the adequacy, accuracy or completeness of the Communications or the Platform, and each Intercreditor Party expressly disclaims liability for errors or omissions in any Communications or the Platform and (iv) no representation or warranty of any kind, express, implied or statutory, including any representation or warranty of merchantability, fitness for a particular purpose, non-infringement of third party rights or freedom from viruses or other code defects, is made by any Intercreditor Party in connection with any Communications or the Platform.

## 29.7 English language

- 29.7.1 Any notice given under or in connection with any Senior Finance Document must be in English.
- 29.7.2 All other documents provided under or in connection with any Senior Finance Document must be in English or, if not in English, and if so required by the relevant Agent, accompanied by a certified English translation and, in this case, the English translation shall prevail unless the document is a constitutional, statutory or other official document.

### 30. CALCULATIONS AND CERTIFICATES

#### 30.1 Accounts

In any litigation or arbitration proceedings arising out of or in connection with a Senior Finance Document, the entries made in the accounts maintained by a Senior Secured Creditor are *prima facie* evidence of the matters to which they relate.

## 30.2 Certificates and Determination

Any certification or determination by a Senior Secured Creditor of a rate or amount under any Senior Finance Document is, in the absence of manifest error, conclusive evidence of the matters to which it relates.

#### 30.3 Day count convention

Any interest, commission or fee accruing under a Senior Finance Document shall accrue from day to day and is calculated on the basis of the actual number of days elapsed and a year of 360 days (where due in US dollars) and 365 days (where due in HK dollars).

## 31. PARTIAL INVALIDITY

If, at any time, any provision of the Senior Finance Documents is or becomes illegal, invalid, or unenforceable in any respect under the law of any jurisdiction, neither the legality, validity or enforceability of the remaining provisions nor the legality, validity or enforceability of such provision under the law of any other jurisdiction shall in any way be affected or impaired.

- 127 -

### 32. **REMEDIES AND WAIVERS**

No failure to exercise, nor any delay in exercising, on the part of any Senior Secured Creditor, any right or remedy under the Senior Finance Documents shall operate as a waiver, nor shall any single or partial exercise of any right or remedy prevent any further or other exercise or the exercise of any other right or remedy. The rights and remedies provided in the Senior Finance Documents are cumulative and not exclusive of any rights or remedies provided by law.

# 33. INTERCREDITOR ARRANGEMENTS

## 33.1 Notices of Required Decisions

- 33.1.1 If, at any time, a matter requiring a Decision comes to the notice of a Senior Secured Creditor, that Senior Secured Creditor shall promptly inform the Intercreditor Agent and, where relevant, its Facility Agent in writing.
- 33.1.2 If, at any time, a matter requiring a Decision comes to the notice of the Intercreditor Agent, the Intercreditor Agent shall promptly notify in writing each Facility Agent (and, after the occurrence of a Hedging Voting Right Event in relation to any Hedging Counterparty that is continuing, that Hedging Counterparty) of that matter specifying:
  - (i) whether the matter concerns a Fundamental Term and, if not, which Senior Secured Creditors may vote in respect of the Decision and the aggregate Voting Entitlement required for the Decision to be made;
  - (ii) the date and time by which the Intercreditor Agent requires receipt of all votes in respect of the Decision (the "**Decision Date**"); and
  - (iii) following a Hedging Voting Right Event, the identity of the relevant Hedging Counterparty.

## 33.2 Notice of Votes

Each Senior Secured Creditor shall copy notice of its vote to the Intercreditor Agent, to each Facility Agent and each Hedging Counterparty notified by the Intercreditor Agent pursuant to Clause 33.1.2(iii) (*Notices of Required Decisions*).

## 33.3 Decisions under the Senior Finance Documents

Subject to the other provisions of this Agreement, the exercise of any right, power, discretion or determination which has been delegated to the Intercreditor Agent under the Senior Finance Documents (save for any such right, power, discretion or determination to be exercised by any such party for its own account) shall require the consent or agreement of the Required Lenders *provided* that the Intercreditor Agent may exercise any such right, power, discretion or determination (including giving instructions to the Security Agent) without requiring any Decision which the Intercreditor Agent, acting reasonably, considers is a minor, administrative or technical matter which does not adversely affect the rights of the Senior Secured Creditors under the Senior Finance Documents.

- 128 -

#### 33.4 Restrictions On Remedies

Subject to this Clause 33 (Intercreditor Arrangements), no Senior Secured Creditor may, at any time:

- 33.4.1 set off, or purport to set off, at any time, any amount owing to it under the Senior Finance Documents against any amount payable by it to an Obligor (except that any Hedging Counterparty may net off between transactions under a single Hedging Agreement);
- 33.4.2 take any action or commence any legal proceedings of whatsoever nature against an Obligor under or in respect of a Senior Finance Document to which that Obligor is a party including taking any steps or legal proceedings for the winding-up, dissolution or administration of any of the Obligors or for the appointment of a receiver, administrator, administrative receiver, conservator, custodian, trustee or similar officer of any of the Obligors or of any or all of its assets or revenues; or
- 33.4.3 foreclose on, or enforce or seek an order of the court to enforce all or any of the Security.

### 33.5 Notice of Default

- 33.5.1 If any Lender has actual knowledge of a Default which has occurred, it shall so advise the Intercreditor Agent and, where relevant, its Facility Agent in writing.
- 33.5.2 If any Facility Agent has actual knowledge, or has received notice, of a Default which has occurred, it shall so advise the Intercreditor Agent and, where relevant, each Lender in its Lending Group in writing.
- 33.5.3 If the Intercreditor Agent has actual knowledge, or has received notice, of a Default which has occurred, it shall notify each Facility Agent and each Hedging Counterparty in writing and, in the case of an Event of Default, it shall issue a notice under Clause 33.1 (*Notices of Required Decisions*) in respect of that Event of Default.

### 33.6 Application of Enforcement Proceeds

Following the delivery of an Enforcement Notice, all Enforcement Proceeds paid to the Intercreditor Agent in accordance with the Deed of Appointment and Priority shall be applied by it (together with any other payments received by it pursuant to Clause 26.1.2 (*Payments under the Senior Finance Documents*)) in the following order:

- (i) first, in payment of all costs and expenses incurred by or on behalf of the Intercreditor Agent in connection with such enforcement or recovery and which have been certified, in writing, as having been incurred by the Intercreditor Agent;
- second, in payment *pro rata* of all costs and expenses incurred by or on behalf of the Facility Agents in connection with such enforcement or recovery and which have been certified, in writing, as having been incurred by the Facility Agent seeking recovery;

- 129 -

- (iii) third, in payment pro rata of all amounts paid by the Senior Secured Creditors under Clause 23.15 (Indemnity to Intercreditor Agent) of the Common Terms Agreement but which have not been reimbursed by the Company;
- (iv) fourth, in payment pro rata of all amounts paid by the Term Facility Lenders under clause 15.3 (Indemnity to Term Facility Agent) of the Term Facility Agreement or the Revolving Credit Facility Lenders under clause 15.3 (Indemnity to Revolving Credit Facility Agent) of the Revolving Credit Facility Agreement or the Additional Lenders under the equivalent provisions of each Additional Lender Facility Agreement in respect of indemnities to each Additional Facility Agent but which, in each case, have not been reimbursed by the Company;
- (v) fifth, in payment *pro rata* of all costs and expenses incurred by or on behalf of each Senior Secured Creditor in accordance with the Senior Finance Documents in connection with such enforcement and which have been certified, in writing, as having been incurred by the Senior Secured Creditor seeking recovery;
- (vi) sixth, in payment *pro rata* of all accrued and unpaid fees owing to the Agents under the Senior Finance Documents;
- (vii) seventh, in payment *pro rata* of all accrued and unpaid fees and commissions due to the Lenders under the Senior Finance Documents;
- (viii) eighth, in payment *pro rata* of all accrued but unpaid interest (including default interest) due under the Facility Agreements and all sums due under the Hedging Agreements;
- (ix) ninth, in payment *pro rata* of all principal instalments due under the Facility Agreements;
- (x) tenth, in payment *pro rata* of all other amounts owing to the Senior Secured Creditors due and payable under the Senior Finance Documents; and
- (xi) eleventh, in payment of the surplus (if any) to the Security Agent in accordance with the Deed of Appointment and Priority or to its order,

*provided* that, following the giving of any notice by the Intercreditor Agent pursuant to sub-clause 19.2.2 of Clause 19.2 (*Remedies following an Event of Default*), the amounts referred to in paragraphs (viii) and (ix) above shall rank *pari passu*.

- 130 -

#### 33.7 **Representations and Warranties**

On the Fourth Amendment Signing Date and on the Effective Date, each Senior Secured Creditor party hereto represents and warrants to the other Senior Secured Creditors Party hereto that:

- 33.7.1 it is duly organised and validly existing under the laws of the jurisdiction in which it is incorporated;
- 33.7.2 it has power to enter into and has duly authorised the execution, delivery and performance of this Agreement;
- 33.7.3 the obligations expressed to be assumed by it hereunder are legal and valid obligations binding on it and enforceable against it in accordance with the terms hereof; and
- 33.7.4 it is not the beneficiary of any Liens in respect of any Financial Indebtedness owed to it by the Company other than under the Senior Finance Documents.
- 33.8 The provisions of this Clause 33 (*Intercreditor Arrangements*) are for the sole benefit of the Senior Secured Creditors and may be waived or amended without the consent or agreement of the Company provided there is no increase in the liability of the Company as a result.

#### 34. AMENDMENTS AND WAIVERS

## 34.1 Amendment and waiver of common terms

Subject to Clause 25.6 (*Benefit*) and to Clause 34.2 to Clause 34.4 below, any term of, or matter dealt with under, this Agreement and any other Senior Finance Document may be amended, waived or supplemented with the agreement of the Company and/or the other Obligors which are a party to that Senior Finance Document and/or, as the case may be, the Required Lenders.

#### 34.2 Amendment and waiver of Facility Agreements

Subject to Clause 34.3 and Clause 34.4 below, any term of, or matter dealt with under, a Facility Agreement may be amended, waived or supplemented with the agreement of the Company and/or, as the case may be, the required Senior Secured Creditors as specified in that Facility Agreement.

#### 34.3 Amendment and waiver of Fundamental Terms

A Fundamental Term may only be amended or waived by agreement between the Obligors which are a party to the Senior Finance Document which contains that Fundamental Term and each Lender (and, in the case of the provisions referred to in paragraphs (a), (b), (f), (g), (h), (i) and (j) of the definition of Fundamental Term, each Hedging Counterparty).

- 131 -

### 34.4 Amendment and waiver affecting Agents

An amendment or waiver of any term of the Senior Finance Documents which relates to the rights and/or obligations of any Agent may not be effected without the prior written consent of that Agent.

## 35. COUNTERPARTS

This Agreement may be executed in any number of counterparts, all of which taken together shall constitute one and the same instrument.

#### 36. LANGUAGE

The English language shall be the only official and recognised language of this Agreement. If for any reason a translation of this Agreement is required, such translation shall in the event of any dispute be secondary to the original English version which shall take precedence.

### 37. GOVERNING LAW

This Agreement and any non-contractual obligations arising out of or in connection with it are governed by English law.

#### 38. JURISDICTION

#### 38.1 Jurisdiction of English courts

- 38.1.1 The courts of England have exclusive jurisdiction to settle any dispute arising out of or in connection with this Agreement (including a dispute regarding the existence, validity or termination of this Agreement or the consequences of its nullity) (a "**Dispute**").
- 38.1.2 The parties agree that the courts of England are the most appropriate and convenient courts to settle Disputes and accordingly they shall not argue to the contrary.
- 38.1.3 This Clause 38.1 is for the benefit of the Senior Secured Creditors only. As a result, no Senior Secured Creditor shall be prevented from taking proceedings relating to a Dispute in any other courts with jurisdiction. To the extent allowed by law and the Senior Finance Documents, the Senior Secured Creditors may take concurrent proceedings in any number of jurisdictions.

# 38.2 Service of process

Without prejudice to any other mode of service allowed under any relevant law, the Company:

- 38.2.1 irrevocably appoints Law Debenture Corporate Services Limited as its agent for service of process in relation to any proceedings before the English courts in connection with this Agreement; and
- 38.2.2 agrees that failure by a process agent to notify the Company of the process shall not invalidate the proceedings concerned.

- 132 -

## 39. CONFIDENTIALITY

Subject to Clause 40 (*Gaming Authorities*), each of the Senior Secured Creditors agrees to keep confidential all non-public information of a proprietary or confidential nature provided to it by any Obligor or any Wynn Non-Obligor Subordination Deed Party pursuant to this Agreement *provided* that nothing herein shall prevent any Senior Secured Creditor from disclosing any such information:

- (a) to any other Senior Secured Creditor, or any Affiliate thereof that is bound by confidentiality obligations;
- (b) to any other Person pursuant to Clause 21.8 (*Disclosure of Information*) or clause 18.9 (*Disclosure of Information by Security Agent*) of the Deed of Appointment and Priority;
- (c) to any of its or its Affiliates' employees, directors, agents, auditors, attorneys, accountants and other professional advisors who or that is bound by confidentiality obligations;
- (d) upon the request or demand of any Governmental Authority having jurisdiction over it;
- (e) in response to any order of any court or other Governmental Authority or as may otherwise be required pursuant to any Legal Requirement or the rules of any stock exchange on which the shares or other securities of such Senior Secured Creditor or any Affiliate thereof are listed or by any other competent supervisory or regulatory body;
- (f) if required to do so in connection with any litigation or similar proceeding;
- (g) that has been publicly disclosed other than in breach of this Clause;
- (h) in connection with the exercise of any remedy hereunder or under any other Senior Finance Document;
- to any insurer, insurance broker or other service providers of such Senior Secured Creditor or any of its Affiliates who are under a duty of confidentiality to such Senior Secured Creditor or its Affiliate, as the case may be; or
- (j) to any rating agency or direct or indirect provider of credit protection to such Senior Secured Creditor or any of its Affiliates.

The obligations in this Clause 39 (*Confidentiality*) are continuing and, in particular, shall survive and remain binding on each Senior Secured Creditor until the earlier of:

(A) the date that is 12 months after the Final Repayment Date in respect of the Term Facility (or if such Senior Secured Creditor is an Additional Lender, the later of (i) the date that is 12 months after the Final Repayment Date in respect

- 133 -

of the Term Facility and (ii) the date that is 12 months after the Final Repayment Date in respect of the Additional Lender Facility in respect of which it is an Additional Lender); and

) the date that is 12 months after the date on which such Senior Secured Creditor ceases to be a Senior Secured Creditor (unless such date falls after the date referred to in paragraph (A) above in respect of such Senior Secured Creditor, in which case, for the avoidance of doubt, the obligations in this Clause 39 (*Confidentiality*) shall cease to remain binding on the date referred to in paragraph (A) above in respect of such Senior Secured Creditor).

# 40. GAMING AUTHORITIES

Each of the Senior Secured Creditors agrees to cooperate, having regard to its internal procedures and policies, with the reasonable requests of any and all gaming authorities in connection with the administration of their regulatory jurisdiction over the Company, any other Obligor or any Wynn Non-Obligor Subordination Deed Party, to the extent not inconsistent with any applicable legal, regulatory or contractual restrictions (including any duties of confidentiality) or the terms of the Senior Finance Documents, *provided* that the Senior Secured Creditors are indemnified for any cost, loss or liabilities incurred in connection with such cooperation.

- 134 -

(B)

# SCHEDULE 1 THE LENDERS AND HEDGING COUNTERPARTIES

Part A Term Facility Lender

Bank of China Limited, Macau Branch

Part B Not Used

Part C Revolving Credit Facility Lender

Bank of China Limited, Macau Branch

Part D Hedging Counterparties

Deutsche Bank AG

Merrill Lynch International

- 135 -

# SCHEDULE 2 CONDITIONS PRECEDENT

#### Part A

#### **Conditions Precedent to the CP Satisfaction Date**

THE CONDITIONS PRECEDENT SET OUT IN PART A OF THIS SCHEDULE 2 (CONDITIONS PRECEDENT) HAVE BEEN RECEIVED IN FORM AND SUBSTANCE ACCEPTABLE TO THE INTERCREDITOR AGENT OR OTHERWISE WAIVED.

#### 1. **Project Documents**

- (a) Receipt by the Intercreditor Agent and the Technical Adviser of:
  - (i) a list setting out the title, date, parties and subject matter of each Project Document entered into as of the CP Satisfaction Date with a total contract price payable by the Company (or expected aggregate amount to be paid by the Company in the case of "cost plus" contracts) or which may otherwise involve liabilities, actual or contingent, in each case in an amount of less than USD5,000,000 or its equivalent; and
  - a copy of any other Project Document and each Affiliate Agreement entered into as of the CP satisfaction Date duly executed by all parties thereto,

each certified by a Responsible Officer of the Company to be true, complete and up-to-date, the Project Documents and Affiliate Agreements to be in full force and effect, the Project Documents are consistent with the Project Budget, the Project Schedule and the Plans and Specifications and, in the case of each Affiliate Agreement, the transactions contemplated therein comply with the requirements of paragraph 10 of Part B of Schedule 5 (*Covenants*).

- (b) (i) Each Major Project Document has been duly authorised, executed and delivered by the parties thereto and duly filed, recorded, stamped and registered as necessary; and
  - all conditions precedent to the effectiveness thereof (other than any such conditions relating to the occurrence of the CP Satisfaction Date) have been satisfied or waived in accordance with their respective terms and each such Major Project Document is in full force and effect accordingly.
- (c) The Notice to Proceed and the Prime Contractor's Performance Bonds (in an aggregate amount of not less than USD45,722,399) have been duly issued in accordance with the terms of the Construction Contract.

- 136 -

### Senior Finance Documents

(a)

(c)

- Receipt by the Intercreditor Agent of an original of each of the following Senior Finance Documents duly executed by the parties thereto:
  - (i) each Facility Agreement;
  - (ii) the Common Terms Agreement;
  - (iii) each Security Document (in the case of any Operating Account referred to in paragraphs (b), (c) or (d) of the definition thereof or any Reinsurance, required to be entered into prior to the CP Satisfaction Date and, in the case of the Wong Consent, the Intercreditor Agent shall accept a notarised copy of such document in lieu of the original in satisfaction of the requirements in respect thereof under this paragraph 2(a)(iii));
  - (iv) any other Senior Finance Documents (other than any Ancillary Finance Document) entered into prior to the CP Satisfaction Date; and
  - (v) any other document entered into which the Intercreditor Agent and the Company agree prior to the CP Satisfaction Date to designate as a Senior Finance Document.
- (b) Each of the Ancillary Finance Documents has been duly executed by the parties thereto.
  - Save in respect of the authorisation by the Macau SAR required to be given in accordance with the Land Concession Consent Agreement in relation to the Land Security Assignment and the notice required to be given to the Macau SAR in relation to the Assignment of Rights, each Senior Finance Document referred to in this paragraph 2 has been duly authorised, executed and delivered by such of the Obligors, the Performance Bond Provider and the other Major Project Participants as parties thereto and duly filed, notified, recorded, stamped and registered as necessary;
    - (ii) all conditions precedent to the effectiveness thereof (other than any such conditions relating to the occurrence of the CP Satisfaction Date) have been satisfied or waived in accordance with their respective terms and each such Senior Finance Document (save as provided in this sub-paragraph (c)) is in full force and effect accordingly; and
    - (iii) none of such of the Obligors, the Performance Bond Provider or the other Major Project Participants as is party to any such Senior Finance Document is or, but for the passage of time and/or giving of notice will be, in breach of any obligation thereunder.

### Advisers' Reports and Certificates

Receipt by the Intercreditor Agent of the following reports and other documents and, in the case of each report, confirmed by the relevant adviser to be the final report prior to the CP Satisfaction Date:

(a) the revised Technical Adviser's report in relation to the Projects (including confirmation that the Projects constitute Category C projects under the Equator Principles);

3.

- (b) the revised Insurance Adviser's report in relation to the Insurances;
- (c) the revised Market Adviser's report in relation to the Projects; and
- (d) the revised Financial Model in relation to the Projects.

### 4. Approved List

(a)

5.

Receipt by the Intercreditor Agent of a copy of the agreement between the GCLAs and the Company approving the Approved List.

#### Due establishment, authority and certification

- In relation to the Company, receipt by the Intercreditor Agent of a certificate in substantially the form set out in Part C (*Form of Company's Due Establishment CP Satisfaction Date Certificate*) of this Schedule 2 signed by a Responsible Officer of the Company and which:
  - (i) attaches a copy of the Company's Governing Documents; and
  - (ii) attaches a board resolution approving the execution, delivery and performance of the Transaction Documents to which it is a party and the terms and conditions thereof and authorising a named person or persons to sign such Transaction Documents and any document to be delivered by the Company pursuant to any Transaction Documents.
- (b) In relation to each of the Wynn Obligors and the other Major Project Participants (other than the Macau SAR and the Persons referred to in sub-paragraph 5(c) below), receipt by the Intercreditor Agent of a certificate signed by a Responsible Officer of each such Person in substantially the form set out in Part D (*Form of Major Project Participant's CP Satisfaction Date Certificate*) of this Schedule 2 attaching copies of:
  - (i) its Governing Documents; and
  - (ii) a board resolution approving the execution, delivery and performance of the Transaction Documents to which it is a party and the terms and conditions thereof and authorising a named person or persons to sign such Transaction Documents and any document to be delivered by it pursuant thereto.
- (c) In relation to the issuer of the Prime Contractor's Performance Bond and Banco Nacional Ultramarino, S.A. in its capacity as the Performance Bond Provider, party to the Performance Bond Facility Agreement and Second Ranking Finance Party (as defined in the Deed of Appointment and Priority), receipt by the Intercreditor Agent of evidence (which may comprise a copy of the relevant portion of such Person's signature book or equivalent) confirming such Person's approval of the execution, delivery and performance of the Transaction Documents to which it is party and authorisation of the Person or Persons who have signed such Transaction Documents on its behalf to do so.

- 138 -

#### 6. **Project Documents and Permits**

Receipt by the Intercreditor Agent of a certificate, in substantially the form set out in Part G (*Form of Company's CP Satisfaction Date Certificate*) of this Schedule 2 (the "**Company's CP Satisfaction Date Certificate**") signed by a Responsible Officer of the Company and which, amongst other things:

- (a) certifies that copies of documents delivered in satisfaction of other conditions precedent in this Part A remain, save to the extent of any subsequent amendments made in accordance with the Senior Finance Documents, true, complete and up-to-date copies in full force and effect as at the CP Satisfaction Date;
- (b) attaches copies of any such amendments referred to in sub-paragraph (a) above and certifies that such copies are true, complete and up-to-date and in full force and effect as at the CP Satisfaction Date; and
- (c) confirms that:
  - (i) all Permits described in Part A of Schedule 12 (*Permits*) shall have been issued and be in full force and effect and not subject to current legal proceedings or to any unsatisfied conditions (that are required to be satisfied by the CP Satisfaction Date) that could reasonably be expected to materially adversely modify any Permit, to revoke any Permit, to restrain or prevent the construction or operation of the Projects or otherwise impose adverse conditions on either Project or the financing contemplated under the Senior Finance Documents and all applicable appeal periods with respect thereto shall have expired;
  - (ii) with respect to any of the Permits described in Part B of Schedule 12 (*Permits*), (i) each such Permit is of a type that is routinely granted on application and compliance with the conditions for issuance and (ii) the Company is not aware of any facts or circumstances which indicate that any such Permit will not be timely obtainable without undue expense or delay by the Company or the applicable Person, respectively, prior to the time that it becomes required; and
  - (iii) all other Permits have been obtained or effected to the extent they are required as of the date of the certificate and, as far as the Company is aware, no reasonable basis exists for it to believe that the Permits which are not required at such date will not be obtained when they are required,

and, in the case of the Permits described in Part A of Schedule 12 (*Permits*), attaching a copy of each such Permit (other than the Permit referred to in paragraph 8(iv) thereof, which it is acknowledged is comprised in the consent acknowledged by the Macau SAR Government in paragraph 3 of the Gaming Concession Consent Agreement).

7. [Not used]

- 139 -

### 8. Security

Receipt by the Intercreditor Agent of evidence that:

- (a) save in respect of the authorisation by the Macau SAR required to be given in accordance with the Land Concession Consent Agreement in relation to the Land Security Assignment and the notice required to be given to the Macau SAR in relation to the Assignment of Rights, each Security Document has been duly filed, notified, recorded, stamped and registered as necessary and all other actions necessary in the reasonable opinion of the Intercreditor Agent or the Security Agent to perfect the Security have been carried out; and
- (b) the Securities Account has been established in accordance with the Wynn Resorts Support Agreement and is fully funded as provided therein.

#### 9. Insurance

- (a) Receipt by the Intercreditor Agent of a certificate addressed to it from the Insurance Adviser in substantially the form set out in Part E (Form of Insurance Adviser's CP Satisfaction Date Certificate) of this Schedule 2 certifying that all Construction Period Insurances required by the Senior Finance Documents to be in effect at the CP Satisfaction Date are in full force and effect and all premia (if any) due and payable as at the CP Satisfaction Date have been paid and the Secured Creditors have been named as co-insureds under those insurances in accordance with Schedule 7 (Insurance).
- (b) Receipt by the Intercreditor Agent of certified copies of all insurance policies, updates, cover notes or insurance slips then available in respect of Insurances then required to be maintained by the Company.
- (c) Receipt by the Intercreditor Agent of a Reinsurance Broker's Letter of Undertaking from each of the brokers referred to in paragraph 2.2 of Schedule 7 (*Insurance*) through whom Reinsurances have been effected.
- (d) Receipt by the Intercreditor Agent of confirmation from the Licensor that:
  - (i) the Insurances set out in Schedule 7 (*Insurance*) satisfy the insurance requirements set out in Section 8.01 of the IP Agreement; and
  - (ii) the insurers with whom such Insurances are effected are acceptable to the Licensor.

## 10. Accounts

Each of the Accounts specified in paragraph 1.1 of Schedule 6 (Accounts) has been established.

- 140 -

# 11. Financials

Receipt by the Intercreditor Agent of:

- (i) pro forma statement of the Company's assets and liabilities as at a date not more than one month prior to the CP Satisfaction Date; and
- (ii) the most recent quarterly and annual financial statements of the type and in respect of the Persons specified in paragraph 1 of Part A of Schedule 5 (*Covenants*), together with certificates from a Responsible Officer of each such Person certifying such financial statements and stating that no material adverse change in the assets, liabilities, operations or financial condition of each such Person has occurred since the dates of the respective financial statements, except as otherwise provided in such certificate.

## 12. **Process agents**

Where such appointment is required under any Senior Finance Document, process agent acceptance of its appointment by the Company and each of such of the other Obligors, the Performance Bond Provider and the other Major Project Participants as is party to such Senior Finance Document for the acceptance of legal proceedings.

## 13. Specimen signatures

Receipt by the Intercreditor Agent of specimen signatures of all Responsible Officers of the Company who shall be issuing Advance Requests.

# 14. Legal opinions

Receipt by the Intercreditor Agent of legal opinions from:

- (a) Mr Henrique Saldanha, Macanese legal adviser to the Senior Secured Creditors;
- (b) Mr Alexandre Correia da Silva, Macanese legal adviser to the Company;
- (c) Lovells, English legal advisers to the Company in relation to the Major Project Documents governed by English law and referred to in sub-paragraphs (c) and (d) of the definition thereof;
- (d) Lionel Sawyer & Collins, Nevada legal adviser to the Senior Secured Creditors;
- (e) Schreck Brignone, Nevada legal adviser to Wynn Resorts;
- (f) Mann and Partners, Isle of Man legal adviser to the Senior Secured Creditors;
- (g) Clifford Chance, Hong Kong SAR legal advisers to the Senior Secured Creditors;
- (h) Clifford Chance, English legal advisers to the Senior Secured Creditors;
- (i) Mallesons Stephen Jaques, Hong Kong SAR legal advisers to Leighton Contractors (Asia) Limited;

- 141 -

- (j) Mallesons Stephen Jaques, Hong Kong SAR legal advisers to China State Construction Engineering (Hong Kong) Limited;
- (k) Ms Elisa Costa, Macanese legal adviser to China Construction Engineering (Macau) Company Limited;
- (l) Mallesons Stephen Jaques, Australian legal advisers to Leighton Holdings Limited;
- (m) Mallesons Stephen Jaques, Hong Kong SAR legal advisers to China Overseas Holdings Limited;
- (n) Skadden, Arps, Slate, Meagher & Flom LLP, New York legal advisers to the Company;
- (o) Mr Leonel A. Alves, Macanese legal adviser to Banco Nacional Ultramarino, S.A.; and
- (p) Mr Leonel A. Alves, Macanese legal adviser to Companhia de Seguros de Macau, S.A.,

or, save in the case of sub-paragraphs (a), (b), (g), (h), (n), (o) and (p), such other lawyers or law firms as may be reasonably acceptable to the Intercreditor Agent.

# 15. Fees and expenses

Receipt by the Intercreditor Agent of evidence that:

- (a) all taxes, fees and other costs payable in connection with the execution, delivery, filing, recording, stamping and registering of the documents referred to in this Part A; and
- (b) all fees, costs and expenses due to the Secured Creditors (including all amounts payable pursuant to any Fee Letter) and their advisers under the Senior Finance Documents on or before the CP Satisfaction Date,

have been paid or shall be paid out of the proceeds of the Initial Advance (to the extent that such amounts have been duly invoiced).

## 16. **Performance Bond**

Receipt by the Intercreditor Agent of evidence that the Concession Contract Performance Bond has been issued under the Performance Bond Facility as required by Article 61 of the Concession Contract.

# 17. Concession Contract

- (a) Receipt by the Intercreditor Agent of evidence that:
  - (i) the capital requirement under Article 15 of the Concession Contract has been complied with;

- 142 -

- (ii) the procedural requirements under Article 16 of the Concession Contract relevant to the creation of security over shares have been complied with;
- the Company has notified the Macau SAR regarding the Facilities and the Security granted in favour of the Secured Creditors and produced copies of the Senior Finance Documents to the Macau SAR pursuant to Articles 34(2) and (3) of the Concession Contract;
- (iv) expenditure of the total Project Costs referred to in the Project Budget which will be included in the calculation of the investment amount of MOP 4 billion will cause the Company to satisfy the requirement under the Concession Contract of expending MOP 4 billion within seven years after the date of the Concession Contract;
- legislation has been enacted by the Macau SAR which provides for casino operators such as the Company to be the grantors of credit to casino patrons;
- (vi) the letter agreements dated 1 May 2002 between the Company and Banco Nacional Ultramarino and the Company and the Macau SAR in relation to the termination of the Concession Contract are of no force or effect; and
- (vii) each area in which any operation of casino games of chance or other forms of gaming will be carried out following the opening of the Original Project has been classified by the Macau SAR as a casino or gaming zone in accordance with article 9 of the Concession Contract.
- (b) Receipt by the Intercreditor Agent of copies of all documents submitted to the Macau SAR as required under Article 21 of the Concession Contract and evidence reasonably satisfactory to the Intercreditor Agent that approval of the Company's delegation of management authority (including the appointment of the executive director, the scope of power of the executive director and term of authorisation) has been granted by the Macau SAR pursuant to Article 21 of the Concession Contract.
- (c) Receipt by the Intercreditor Agent of a letter from Macau SAR in relation to the granting of extension of time for completion of the Projects from 31 December 2006.

# 18. Land and Land Concession Contract

Receipt by the Intercreditor Agent of evidence that:

- (a) the Land Concession Contract is registered with the Macau Real Estate Agency;
- (b) the Land Concession Contract has been published in the Official Bulletin; and
- (c) the amount payable to Sociedade de Empreendimentos Nam Van S.A.R.L. for the surrendering by Sociedade de Investimento Imobiliario Hang Keng Van,

- 143 -

S.A.R.L., Sociedade de Investimento Imobiliario Lok Keng Van, S.A.R.L. and Sociedade de Investimento Imobiliario Hei Keng Van, S.A.R.L. of their respective interests in the land registered with the Macau Real Estate Registry with registration numbers 22322, 22324 and 22325 does not exceed MOP 140,000,000.

### 19. Auditors

Receipt by the Intercreditor Agent of evidence that the Company has appointed the Auditors and the Auditors have accepted such appointment.

#### 20. Continued Appointment of Advisers

The Insurance Adviser and the Technical Adviser have been appointed to act after the Signing Date in accordance with a scope of work as agreed by the Company and the Intercreditor Agent.

#### 21. No Material Adverse Effect

Receipt by the Intercreditor Agent of a certificate, in substantially the form of the Company's CP Satisfaction Date Certificate, from the Company confirming that, as at the CP Satisfaction Date, no Material Adverse Effect has occurred and is continuing nor could reasonably be expected to occur.

#### 22. Projections

Receipt by the Intercreditor Agent of Projections for the Company for the period from 1 July 2007 to 31 December 2008 and which:

- (a) demonstrate compliance with paragraph 1 of Part B of Schedule 5 (*Covenants*);
- (b) otherwise comply with the requirements of paragraph 2(c) of Part A of Schedule 5 (*Covenants*).

#### 23. Litigation

Receipt by the Intercreditor Agent of a certificate, in substantially the form of the Company's CP Satisfaction Date Certificate, from the Company confirming that no action, suit, proceeding or investigation of any kind shall have been instituted or, to the Company's knowledge, threatened, including actions or proceedings of or before any Governmental Authority, to which the Company, either Project or, to the knowledge of the Company, any other Obligor, the Performance Bond Provider or other Major Project Participant, is a party or is subject, or by which any of them or any of their properties or either Project are bound, in each case, that could reasonably be expected to have a Material Adverse Effect, nor is the Company aware of any reasonable basis for any such action, suit, proceeding or investigation and no injunction or other restraining order shall have been issued and no hearing to cause an injunction or other restraining order to be issued shall be pending or noticed with respect to any action, suit or proceeding if the same reasonably could be expected to have a Material Adverse Effect.

- 144 -

# 24. **Project Budget**

Receipt by the Intercreditor Agent at or prior to the Expansion Signing Date of a budget in the form of Schedule 13 (Form of Project Budget) which:

- (a) includes such other information and supporting data as any of the Senior Secured Creditors or the Technical Adviser may reasonably require; and
- (b) demonstrates, without taking account of any amount of Net Operating Cashflow in excess of the Base Net Operating Cashflow Amount, any amount of Contingent Equity, any amount of Contingent Subordinated Funding or any amount of the Contingent Debt Facilities, no Forecast Funding Shortfall.

### 25. **Project Schedule**

Receipt by the Intercreditor Agent at or prior to the Expansion Signing Date of a schedule for construction and completion of the Projects in the Agreed Form which demonstrates that Substantial Completion will occur and the Opening Conditions specified in paragraph (a) of the definition thereof will have been satisfied on or before the following dates in respect of the Original Project, the Expansion and each phase of the Expansion specified:

| Original Project | 31 December 2006 |
|------------------|------------------|
| Area 4           | 18 August 2007   |
| Area 5           | 10 November 2007 |
| Expansion        | 10 January 2008  |

#### 26. Schedule of Values

Receipt by the Intercreditor Agent of the "Expansion Project Schedule of Values" submitted and approved in accordance with section 5.1 of the Construction Contract and the "Original Project Schedule of Approved Values" (each as defined in the Construction Contract).

### 27. Representations and Warranties

Receipt by the Intercreditor Agent of a certificate, in substantially the form of the Company's CP Satisfaction Date Certificate, from the Company confirming that the representations and warranties of:

- (a) the Company, each Wynn Obligor and the Licensor set forth in Schedule 4 (*Representations and Warranties*) or in any of the other Transaction Documents is true and correct in all material respects as if made on such date (except that any representation and warranty that relates expressly to an earlier date shall be deemed made only as of such earlier date); and
- (b) to the Company's knowledge, the Performance Bond Provider, any Obligor and each other Major Project Participant (in each case other than the Company, any Wynn Obligor or the Licensor) set forth in any of the Transaction

- 145 -

Documents is true and correct in all material respects as if made on such date (except that any representation and warranty that relates expressly to an earlier date shall be deemed made only as of such earlier date), unless the failure of any such representation and warranty referred to in this sub-paragraph (b) to be true and correct could not reasonably be expected to have a Material Adverse Effect.

## 28. Plans and Specifications

The Company shall have delivered to the Technical Adviser or (on such terms as the Technical Adviser may reasonably require) made available to it in Hong Kong or Macau all plans, specifications, design documents, schematic drawings and related items for the design, architecture and construction of the Projects.

## 29. PASA and IP Agreement

- (a) The PASA and PASA Direct Agreement have been entered into with Wynn Design & Development, LLC and the Intercreditor Agent has received such legal opinions, in form and substance reasonably satisfactory to it, concerning the entry by Wynn Design & Development, LLC into the PASA and PASA Direct Agreement, the efficacy thereof and such other matters as it may reasonably require.
- (b) The IP Agreement and IP Direct Agreement have been entered into with Wynn Resorts and Wynn Resorts Holdings, LLC as Licensor and the Intercreditor Agent has received such legal opinions, in form and substance reasonably satisfactory to it, concerning the entry by Wynn Resorts and Wynn Resorts Holdings, LLC into the IP Agreement and the IP Direct Agreement, the efficacy thereof and such other matters as it may reasonably require.
- (c) The Licensor has acceded to the Subordination Deed in accordance with the terms thereof.

## 30. Shareholder Reorganisation

- (a) Receipt by the Intercreditor Agent of a Substantial Shareholder's Undertaking in form and substance satisfactory to it from each Substantial Shareholder (if any) as at the CP Satisfaction Date.
- (b) Receipt by the Intercreditor Agent of a Relevant Party's Undertaking in form and substance satisfactory to it from each Relevant Party (if any) as at the CP Satisfaction Date.
- (c) Receipt by the Intercreditor Agent of evidence that, as at the CP Satisfaction Date, all shares in Wynn Holdings not legally and beneficially owned by Wynn International as at the Signing Date are either legally and beneficially owned by Wynn International or their original owners (direct and indirect) as at the Signing Date.

- 146 -

Part B1

### (I) Conditions Precedent to each Advance under the Term Loan Facilities prior to the Fourth Amendment Effective Date

## 1. No Default

No Default shall have occurred and be continuing.

#### 2. **Representations and Warranties**

The representations and warranties contained in Schedule 4 (*Representations and Warranties*) which are repeated by the Company pursuant to Clause 17.2 (*Timing*) are true and correct in all material respects with reference to the facts and circumstances existing on the Advance Date.

### 3. Sufficiency of Funds; Project Schedule Prior to the Expansion Opening Date

Solely with respect to the period prior to the Expansion Opening Date, certification by a Responsible Officer of the Company that it has sufficient Funds to achieve Substantial Completion of the Expansion.

## 4. Advance Request and Certificate

Receipt by the Intercreditor Agent and the relevant Facility Agent of an Advance Request containing all attachments, exhibits and certificates required thereby, all appropriately completed and duly executed by a Responsible Officer of the Company, including:

- (a) solely with respect to the period prior to the Expansion Opening Date, a statement of the specific purposes to which each Advance shall be applied, including:
  - whether payment (or refinancing of payment) for Hotel Project Costs or other Project Costs, for general corporate purposes, for the Diamond Expansion, whether for variations to Project Works or otherwise permitted pursuant to paragraph 15 of Part B of Schedule 5 (*Covenants*) and, where for refinancing of payment the original source of funds for such payment; and
  - (ii) a break down by Line Item of each such category of Project Costs supported:
    - (A) in the case of any payment exceeding USD1,000,000 or its equivalent, to the reasonable satisfaction of the Intercreditor Agent, by invoices, receipts and other documentary evidence attached to the Advance Request; and
    - (B) in the case of any progress payment claimed pursuant to the Construction Contract, to the reasonable satisfaction of the Intercreditor Agent, by the relevant approved "Application for Progress Payment" and "Contractor's Certificate" (each as

- 147 -

defined in the Construction Contract) and supporting documentary evidence delivered by the Prime Contractor in accordance with article V of the Construction Contract attached to the Advance Request;

- (b) solely with respect to the period prior to the Expansion Opening Date, insofar as any previous Advance was requested for the purpose of payments which, at the time of its Advance Date, were not yet payable, a statement of such payments made since the last Advance Request supported, in the case of any payment exceeding USD1,000,000 or its equivalent, to the reasonable satisfaction of the Intercreditor Agent, by invoices, receipts and other documentary evidence attached to the Advance Request; and
- (c) solely with respect to the period prior to the Expansion Opening Date, certification that each Advance is required for the purpose specified.

# Adviser's Certificates

5.

- (a) Solely during the period prior to the Expansion Opening Date, receipt by the Intercreditor Agent by not later than 3:00 p.m. on the fifth Business Day prior to the proposed Advance Date of a certificate from the Technical Adviser in substantially the form set out in Part F (*Form of Technical Advisor's Advance Certificate*) of this Schedule 2 (the "**Technical Adviser's Advance Certificate**") certifying in connection with the Advance Request and any amendment thereto made pursuant to Clause 3.1 (*Drawdown Conditions*) that the Technical Adviser has no reason to believe that the current Project Schedule (being the Project Schedule delivered by the Company pursuant to paragraph 25 of Part A of this Schedule 2) is not accurate in all material respects;
- (b) Solely during the period prior to the Expansion Opening Date and with respect to any Advances to be utilised for the purpose of funding any Project Costs (excluding Project Costs related to the Diamond Expansion), receipt by the Intercreditor Agent by not later than 3:00 p.m. on the fifth Business Day prior to the proposed Advance Date of a certificate from the Technical Adviser in substantially the form of the Technical Adviser's Advance Certificate certifying in connection with the Advance Request and any amendment thereto made pursuant to Clause 3.1 (*Drawdown Conditions*) that, to the best of its knowledge:
  - (i) the Advances requested by the Company under the Term Loan Facilities are required prior to the date falling 30 days after the proposed Advance Date to make payments falling due or to refinance such payments previously made; and
  - (ii) certifications and statements made by the Company in the Advance Request (as referred to in Part F (*Form of Technical Adviser's Advance Certificate*) of this Schedule 2) and all attachments thereto are correct in all material respects.

- 148 -

### 6. Recent Reports, Financial Statements and Other Information

- (a) Receipt by the Intercreditor Agent of each of the reports, financial statements and other information due pursuant to paragraphs 1 and 2 of Part A of Schedule 5 (*Covenants*) on or before the date of the Advance Request.
- (b) Solely during the period prior to the Expansion Opening Date, receipt by the Intercreditor Agent of the Technical Adviser's Monthly Report due on or before the date of the Advance Request.

# 7. Hedging

- (a) Receipt by the Intercreditor Agent of a copy (certified as being true and correct by a Responsible Officer) of each Hedging Agreement (and each confirmation thereunder) required to be entered into in accordance with the Hedging Arrangements.
- (b) The Company shall have delivered notice in respect of each Hedging Agreement to the counter-party thereto and the Security Agent shall have received an acknowledgement from each such counter-party, each such notice and acknowledgement being in the form required by clause 4 of the Debenture.

## 8. Fees and Expenses

The Company shall have paid or arranged for payment out of the requested Advance of all fees, expenses and other charges then due and payable by it under the Senior Finance Documents or under any agreements between the Company and any of the Advisers (to the extent that such amounts have been duly invoiced).

### 9. Major Project Documents

The Company shall have delivered to the Intercreditor Agent and the Technical Adviser promptly after mutual execution and delivery thereof:

- (a) a list setting out the title, date, parties and subject matter of each Major Project Document; and
- (b) a copy of any Major Project Document (other than any Resort Management Agreement) listed pursuant to subparagraph (a) above and requested in writing by the Intercreditor Agent or the Technical Adviser.

# (II) Conditions Precedent to each Advance under the Revolving Credit Facility prior to the Fourth Amendment Effective Date

As set out in the Revolving Credit Facility Agreement.

# (III) Conditions Precedent to each Advance under each Additional Lender Facility prior to the Fourth Amendment Effective Date

As set out in the applicable Additional Lender Facility Agreement.

- 149 -

#### Part B2

# Conditions Precedent to each Advance under the Term Loan Facilities on and from the Fourth Amendment Effective Date

In respect of the Advances to be made under the Term Loan Facility on the Fourth Amendment Effective Date (as detailed in the Completion Memorandum), as set out in Schedule 1 to the Common Terms Agreement Fourth Amendment Agreement and the Term Facility Agreement.

(II) Conditions Precedent to each Advance under each Additional Lender Facility on and from the Fourth Amendment Effective Date

As set out in the applicable Additional Lender Facility Agreement.

### Part B3

Conditions Precedent to certain Advances under the Revolving Credit Facility Agreement on and from the Fourth Amendment Effective Date

#### 1. Security Documents

(I)

Each of:

- (a) the Cotai Mortgage;
- (b) the Cotai Power of Attorney;
- (c) the Palo Share Pledge;
- (d) the Palo Assignment of Insurances;
- (e) the Palo Floating Charge; and
- (f) such other security document over other assets of Palo, as may reasonably be required by the Intercreditor Agent,

has been duly authorised, executed and delivered by such of the Obligors party thereto and duly filed, notified, recorded, stamped and registered as necessary and all other actions necessary in the reasonable opinion of the Intercreditor Agent to perfect the Security have been carried out.

### 2. Legal Opinion

Receipt by the Intercreditor Agent of legal opinions from the Macanese legal adviser to the Senior Secured Creditors in relation to the Security Documents listed in paragraph 1 and other legal opinions required in relation to any other governing law of the security documents or in relation to the Obligors granting security under the Security Documents from local legal advisers to the Senior Secured Creditors.

- 150 -

# 3. **Process agents**

Where such appointment is required under any Security Document referred to in paragraph 1 above that Palo is a party to, a copy of process agent acceptance of its appointment by Palo for the acceptance of legal proceedings.

## 4. Other

A copy of any other authorisation (including corporate authorisations of the relevant Obligors entering into the Security Documents) or other document, opinion or assurance which the Intercreditor Agent considers to be necessary or desirable in connection with the entry into and performance or validity and enforceability of the Security Documents listed in paragraph 1.

#### Part C

#### Form of Company's Due Establishment CP Satisfaction Date Certificate

# NOT APPLICABLE AFTER THE FOURTH AMENDMENT EFFECTIVE DATE

To: [ ] as Intercreditor Agent

]

Date: [

Dear Sirs,

- 1. We refer to the common terms agreement dated 14 September 2004 between Wynn Resorts (Macau) S.A. and the financial institutions referred to therein as Senior Secured Creditors (the "**Common Terms Agreement**"). Terms defined in the Common Terms Agreement shall have the same meaning herein and the principles of construction and rules of interpretation set out therein shall also apply.
- 2. This certificate is provided for the purposes of paragraph 5(a) (*Due establishment and authority*) and paragraph 13 (*Specimen signatures*) of Part A of Schedule 2 (*Conditions Precedent*) of the Common Terms Agreement.
- 3. We hereby certify, as at the date of this certificate, that:
  - (i) attached hereto as Annex 1 is a true and complete copy of the Governing Documents of the Company together with all amendments thereto;
  - (ii) attached hereto as Annex 2 is a true and complete copy of resolutions duly adopted by the Board of Directors of the Company at a meeting on [*date*], at which a quorum was present and acting throughout, which resolutions have duly approved the execution, delivery and performance of the Transaction Documents to which the Company is a party and the terms and conditions thereof, duly authorised [*names*] to sign such Transaction Documents and any document to be delivered by the Company pursuant to any Transaction Documents (including Advance Requests), and such resolutions have not been revoked, modified, amended or rescinded and are in full force and effect. Except as attached hereto as Annex 2, no resolutions have been adopted by the Board of Directors of the Company in relation to the execution, delivery or performance of any of the Transaction Documents to which the Company is a party; and
  - (iii) attached hereto as Annex 3 are the specimen signatures of all Responsible Officers of the Company who shall be issuing Advance Requests.

Yours faithfully,

Name: Responsible Officer for and on behalf of **Wynn Resorts (Macau) S.A.** 

- 152 -

# Attachments:

- Annex 1 Governing Documents of the Company
- Annex 2 Board resolutions of the Company
- Annex 3 Specimen signatures of Responsible Officers

- 153 -

#### Part D

### Form of Major Project Participant's CP Satisfaction Date Certificate

# NOT APPLICABLE AFTER THE FOURTH AMENDMENT EFFECTIVE DATE

To: [ ] as Intercreditor Agent

]

Date: [

Dear Sirs,

- 1. We refer to the common terms agreement dated 14 September 2004 between Wynn Resorts (Macau) S.A. and the financial institutions referred to therein as Senior Secured Creditors (the "**Common Terms Agreement**"). Terms defined in the Common Terms Agreement shall have the same meaning herein and the principles of construction and rules of interpretation set out therein shall also apply.
- 2. We hereby certify that:
  - (i) attached hereto as Annex 1 is a true and complete copy of the Governing Documents of [*Major Project Participant*] together with all amendments thereto as of the date of this certificate; and
  - (ii) attached hereto as Annex 2 is a true and complete copy of the resolutions duly adopted by our Board of Directors at a meeting on [*date*], at which a quorum was present and acting throughout, which resolutions have duly approved the execution, delivery and performance of the Transaction Documents to which we are a party and the terms and conditions thereof, duly authorised [*names*] to sign such Transaction Documents and have not been revoked, modified, amended or rescinded and are in full force and effect.

Yours faithfully,

Name: for and on behalf of [Major Project Participant]

- 154 -

#### Part E

### Form of Insurance Adviser's CP Satisfaction Date Certificate

# NOT APPLICABLE AFTER THE FOURTH AMENDMENT EFFECTIVE DATE

To: [ ] as Intercreditor Agent

]

Date: [

- 1. We refer to the common terms agreement dated 14 September 2004 between Wynn Resorts (Macau) S.A. and the financial institutions referred to therein as Senior Secured Creditors (the "**Common Terms Agreement**"). Terms defined in the Common Terms Agreement shall have the same meaning herein and the principles of construction and rules of interpretation set out therein shall also apply.
- 2. This certificate is provided for the purpose of paragraph 9(a) (*Insurance*) of Part A of Schedule 2 (*Conditions Precedent*) of the Common Terms Agreement.
- 3. We hereby certify that, as at the date of this certificate, all Construction Period Insurances required by the Senior Finance Documents to be in effect are in full force and effect and all premia (if any) due and payable as at such date have been paid and the Secured Creditors have been named as co-insureds under those insurances in accordance with Schedule 7 (*Insurance*) of the Common Terms Agreement.

Yours faithfully,

Name: for and on behalf of [Insurance Adviser]

- 155 -

#### Part F

#### Form of Technical Adviser's Advance Certificate

# NOT APPLICABLE AFTER THE FOURTH AMENDMENT EFFECTIVE DATE

To: [ ] as Intercreditor Agent

]

Date: [

Dear Sirs,

3.

- 1. We refer to the common terms agreement dated 14 September 2004 between Wynn Resorts (Macau) S.A. and the financial institutions referred to therein as Senior Secured Creditors (the "**Common Terms Agreement**"). Terms defined in the Common Terms Agreement shall have the same meaning herein and the principles of construction and rules of interpretation set out therein shall also apply.
- 2. This certificate is provided for the purposes of paragraph 5 (*Adviser's Certificates*) and paragraph 9 (*Major Project Documents*) of sub-section I of Part B of Schedule 2 (*Conditions Precedent*) of the Common Terms Agreement and relates to the Advance Request [and the amendment thereto, each]<sup>1</sup> dated on or about the date of this certificate.
  - We hereby certify, as at the date of this certificate, that:
    - (i) we have no reason to believe that the current Project Schedule is not accurate in all material respects;
    - (ii) to the best of our knowledge, the Advances requested by the Company are required prior to the date falling 30 days after the proposed Advance Date to make payments in respect of Project Costs falling due in such period or to refinance such payments previously made;
    - (iii) to the best of our knowledge, all certifications and statements made by the Company in paragraphs 4(ii), 4(iv), 4(viii), 4(ix), 5 and 6 of the Advance Request to which this certificate relates, and all attachments relevant thereto, are correct in all material respects; [and]
    - (iv) [Substantial Completion has been achieved in respect of the Original Project and the Expansion and we agree with the estimate of Final Completion Costs provided to the Intercreditor Agent.]<sup>2</sup>

Yours faithfully,

Name: for and on behalf of [*Technical Adviser*]

<sup>1</sup> Delete if no amendment made.

- 156 -

<sup>&</sup>lt;sup>2</sup> In the case of Advances in respect of Final Completion Costs.

#### Part G

#### Form of Company's CP Satisfaction Date Certificate

## NOT APPLICABLE AFTER THE FOURTH AMENDMENT EFFECTIVE DATE

To: [ ] as Intercreditor Agent

]

Date: [

Dear Sirs,

- 1. We refer to the common terms agreement dated 14 September 2004 between Wynn Resorts (Macau) S.A. and the financial institutions referred to therein as Senior Secured Creditors (the "**Common Terms Agreement**"). Terms defined in the Common Terms Agreement shall have the same meaning herein and the principles of construction and rules of interpretation set out therein shall also apply.
- 2. This certificate is provided for the purposes of paragraph 6 (*Permits*), paragraph 21 (*No Material Adverse Effect*), paragraph 23 (*Litigation*) and paragraph 27 (*Representations and Warranties*) of Part A of Schedule 2 (*Conditions Precedent*) of the Common Terms Agreement.
- 3. We hereby certify, as at the date of this certificate, that:
  - copies of all documents referred to in Part A of Schedule 2 (*Conditions Precedent*) of the Common Terms Agreement and delivered by us to the Intercreditor Agent on or prior to the date of this certificate are and remain, save (in the case of any such documents delivered prior to the date of this certificate) to the extent of any amendments made subsequent to such delivery in accordance with the Senior Finance Documents, true, complete and up-to-date copies of such documents in full force and effect as at the date of this certificate;
  - (ii) attached hereto as Annex 1 are true, complete and up-to-date copies of the amendments referred to in sub-paragraph (i) above, each of which is and remains in full force and effect;
  - (iii) attached hereto as Annex 2 are all Permits described in Part A of Schedule 12 (*Permits*) of the Common Terms Agreement which have been issued and are in full force and effect and not subject to current legal proceedings or to any unsatisfied conditions (that are required to be satisfied by the CP Satisfaction Date) that could reasonably be expected to materially adversely modify any Permit, to revoke any Permit, to restrain or prevent the construction or operation of the Projects or otherwise impose adverse conditions on either Project or the financing contemplated under the Senior Finance Documents and all applicable appeal periods with respect thereto have expired;
  - (iv) with respect to any of the Permits described in Part B of Schedule 12 (*Permits*), (i) each such Permit is of a type that is routinely granted on application and compliance with the conditions for issuance and (ii) no facts or circumstances exist which indicate that any such Permit will not be timely obtainable without undue expense or delay by the Company or the applicable Person, respectively, prior to the time that it becomes required;

- 157 -

- (v) all other Permits have been obtained or effected to the extent they are required as of the date hereof and, as far as the Company is aware, no reasonable basis exists for it to believe that the Permits which are not required as of the date of this certificate will not be obtained when they are required;
- (vi) no Material Adverse Effect has occurred and is continuing nor could reasonably be expected to occur;
- (vii) no action, suit, proceeding or investigation of any kind has been instituted or, to the Company's knowledge, threatened, including actions or proceedings of or before any Governmental Authority, to which the Company, either Project or, to the knowledge of the Company, any other Obligor or Major Project Participant, is a party or is subject, or by which any of them or any of their properties or either Project are bound, in each case, that could reasonably be expected to have a Material Adverse Effect, nor is the Company aware of any reasonable basis for any such action, suit, proceeding or investigation and no injunction or other restraining order has been issued and no hearing to cause an injunction or other restraining order to be issued is pending nor been notified with respect to any action, suit or proceeding to the extent the same reasonably could be expected to have a Material Adverse Effect;
- (viii) the representations and warranties of the Company, each Wynn Obligor and the Licensor set forth in Schedule 4 (*Representations and Warranties*) of the Common Terms Agreement or in any of the other Transaction Documents are true and correct in all material respects as if made on such date (except that any representation and warranty that relates expressly to an earlier date shall be deemed made only as of such earlier date); and
- to the Company's knowledge, the representations and warranties of the Performance Bond Provider and each other Major Project Participant (in each case other than the Company, any Wynn Obligor or the Licensor) set forth in any of the Transaction Documents are true and correct in all material respects as if made on such date (except that any representation and warranty that relates expressly to an earlier date shall be deemed made only as of such earlier date) unless the failure of any such representation and warranty to be true and correct could not reasonably be expected to have a Material Adverse Effect.

Yours faithfully,

Name: Responsible Officer for and on behalf of **Wynn Resorts (Macau) S.A.** 

- 158 -

# Attachments:

- Annex 1 Amendments to Project Documents
- Annex 2 Permits described in Part A of Schedule 12 (Permits)

- 159 -

### SCHEDULE 3 FORM OF ADVANCE REQUEST

| To: | [ | ] as Intercreditor Ager |
|-----|---|-------------------------|
|     |   |                         |

[ ] as Term Facility Agent

Date: [ ]

Dear Sirs,

# Advance Request No. [ ]

- 1. We refer to the common terms agreement (the "**Common Terms Agreement**") dated 14 September 2004 between Wynn Resorts (Macau) S.A. and the financial institutions referred to therein as Senior Secured Creditors. Terms defined in the Common Terms Agreement shall have the same meaning herein and the principles of construction and rules of interpretation set out therein shall also apply.
- 2. This is an Advance Request given pursuant to Clause 3 (*Drawdown of Advances*) of the Common Terms Agreement.
- 3. We hereby give you notice that, upon the terms and subject to the conditions contained in the Common Terms Agreement and the Facility Agreements, we wish to borrow the following Advances under the following Facilities on [proposed Advance Date] (the "**Proposed Advance Date**") to be applied towards the following purposes:

[USD/HKD] [amount] under [the [Tranche A/Tranche B]<sup>3</sup> Facility of the Term Facility to be applied towards a permitted use under the Term Facility Agreement.

- 4. We confirm that:
  - the above purposes and Advances comply with the permitted use of the Facilities under the Facility Agreements and Clause 5
     (*Purpose*) of the Common Terms Agreement and that no part of the above Advances shall be applied otherwise than as mentioned in paragraph 3 above;
  - (ii) each Advance is required for the purpose specified;
  - (iii) each condition specified in Clause 2.2 (*Conditions Precedent to each Advance*) of the Common Terms Agreement is satisfied on the date of this Advance Request; and
  - (iv) since the CP Satisfaction Date, no Material Adverse Effect has occurred and is continuing nor could reasonably be expected to occur.
- 5. We attach signed but undated receipts for the Advances requested above and hereby authorise the Intercreditor Agent to date such receipts on the date such Advances are made.<sup>4</sup>

<sup>3</sup> Delete as appropriate.

<sup>4</sup> Each receipt must be numbered in series, the number corresponding to the number in the heading of the relevant Advance Request.

- 160 -

- 6. The above Advances shall have a [first] Interest Period ending on [*date*].
  - The [proceeds/specified amounts] of the above Advances should be credited to, respectively, the following Accounts:
  - [specify relevant Account and amount]
  - We further confirm, without any personal liability on the part of our Responsible Officer signing this Advance Request, that:
    - (i) no Default is continuing; and
    - (ii) the representations and warranties contained in Schedule 4 (*Representations and warranties*) of the Common Terms Agreement which are repeated by the Company pursuant to Clause 17.2 (*Timing*) of the Common Terms Agreement are true and correct in all material respects with reference to the facts and circumstances existing on the date of this Advance Request.

Yours faithfully,

7.

8.

Name: <sup>5</sup>Responsible Officer for and on behalf of **Wynn Resorts (Macau) S.A.** 

Attachments: [list]

- 161 -

<sup>&</sup>lt;sup>5</sup> Each Advance Request must be signed by a Responsible Officer of the Company whose specimen signature has been delivered to the Intercreditor Agent pursuant to Clause 3.2.1 (*Completion of an Advance Request*).

## SCHEDULE 4 REPRESENTATIONS AND WARRANTIES

# 1. Organization

Each of the Obligors is duly organized, incorporated, validly existing and (if applicable) in good standing under the laws of the jurisdiction of its organization and has all requisite corporate or limited liability company power and authority to:

- (a) carry on its business as now conducted;
- (b) own or hold under lease and operate the Properties it purports to own or hold under lease;
- (c) carry on its business as now being conducted and as now proposed to be conducted in respect of the Projects and the Cotai Project;
- (d) incur the Financial Indebtedness contemplated hereunder; and
- (e) execute, deliver and perform under each of the Transaction Documents to which it is a party and create any Lien on its Property contemplated thereunder.

# 2. Authorization; No Conflict

- 2.1 Each of the Obligors has taken all necessary corporate or limited liability company action, as the case may be, to authorize the execution, delivery and performance of the Transaction Documents to which it is a party. Neither the execution, delivery or performance of each Transaction Document to which it is a party nor the consummation of the transactions contemplated thereby:
  - (a) by each Obligor does or will contravene the formation or constitutional documents or any other material Legal Requirement then applicable to or binding on each such Obligor; or
  - (b) does or will contravene or result in any breach or constitute any default under, or result in or require the creation or imposition of any Lien upon any of the Properties of any Obligor or under any security or agreement or instrument to which any Obligor is a party or by which it or any of its respective properties may be bound, except for Permitted Liens or as could not, in the aggregate, reasonably be expected to have a Material Adverse Effect.
- 2.2 No consent or authorization or filing with, notice to or other act by or in respect of, any Governmental Authority or any Person is required in connection with the borrowings under the Senior Finance Documents or with the execution, delivery, performance, validity or enforceability of any of the Transaction Documents, except consents, authorisations, recordings, stampings, filings, registrations and notices described in the definition of 'Required Filings' in Clause 1.1 of this Agreement or Schedule 2 (*Conditions Precedent*) and Schedule 12 (*Permits*) (in the case of Schedule 2 (*Conditions Precedent*) and Schedule 12 (*Permits*) which consents, authorisations, registrations, filings and notices have, unless otherwise indicated on such schedule, been obtained or made and are in full force and effect) or, in the case of any Transaction Documents entered into after the date of the Initial Advance under

- 162 -

the Term Loan Facilities, as have been obtained or made and are in full force and effect at the time this representation is deemed to be made (except consents, authorisations, recordings, stampings, filings, registrations and notices described in the definition of 'Required Filings' in Clause 1.1 of this Agreement or Schedule 2 (*Conditions Precedent*) and Schedule 12 (*Permits*) (in the case of Schedule 2 (*Conditions Precedent*) and Schedule 12 (*Permits*) which consents, authorisations, registrations, filings and notices have, unless otherwise indicated on such schedule, been obtained or made and are in full force and effect).

# 3. Legality, Validity and Enforceability

- 3.1 Each of the Transaction Documents to which any of the Obligors is a party is a legal, valid and binding obligation of such Obligor, enforceable against such Obligor in accordance with its terms, subject only to bankruptcy and similar laws and principles of equity.
- 3.2 None of the Transaction Documents to which any of the Obligors is a party has been amended or modified except in accordance with this Agreement.
- 3.3 No Obligor has entered into any additional contracts in contravention of the terms of any Senior Finance Document.

#### 4. Compliance with Law and Permits

Each Obligor is in compliance in all material respects with all material Legal Requirements (including all material Environmental Laws) and Permits and no notices of any material violation of any Permit made or issued by or with a Governmental Authority relating to any of the Projects have been issued, entered or received by any such Obligor (and which violation is continuing).

#### 5. Permits

- 5.1 As of the Third Amendment Signing Date, there are no Permits of which the Company is aware or any material Permits, in each case, made or issued by or with a Governmental Authority that are required or will become required under existing Legal Requirements for the ownership, development, construction or financing of either Project, other than the Permits described in Schedule 12 (*Permits*).
- 5.2 All Permits have been obtained or effected to the extent they are required at the time this representation is deemed to be made and, as far as the Company is aware, no reasonable basis exists for it to believe that the Permits which are not required at such time shall not be obtained when they are required.

#### 6. Litigation

There are no pending or, to any Obligor's knowledge, threatened actions, suits, proceedings or investigations of any kind, including actions or proceedings of or before any Governmental Authority, to which any Obligor is a party or is subject, or by which any of them or any of their Properties (including, without limitation, revenue) or either of the Projects or the Cotai Project is bound that, individually or collectively, could reasonably be expected to have a Material Adverse Effect nor is any Obligor aware of any reasonable basis for any such action, suit, proceeding or investigation.

- 163 -

### 7. Financial Statements

The financial statements of the Obligors, delivered to the Intercreditor Agent pursuant to Clause 2.1 (*Conditions Precedent to the CP Satisfaction Date*, were, and, in the case of financial statements to be delivered after the CP Satisfaction Date pursuant to paragraph 1 of Part A of Schedule 5 (*Covenants*) hereto, will be prepared in conformity with applicable GAAP and fairly present in all material respects the financial position of the entities described in such financial statements as of the respective dates thereof and the results of operations and cash flows of the entities described therein for each of the periods then ended subject, in the case of any such unaudited financial statements, to changes resulting from audit and normal year-end adjustments. No such financial statement fails to disclose any material Guarantee Obligations, contingent liabilities or liabilities for taxes, or any long-term leases or unusual forward or long-term commitments, including, without limitation, any interest rate or foreign currency swap or exchange transaction or other obligation in respect of derivatives, required to be reflected therein.

#### 8. Security Interests

8.1

- As of the Fourth Amendment Effective Date, save in respect of any Excluded Subsidiary or Excluded Project (in respect of which (save for any Excluded Project located on the Cotai Site and assets comprised therein or connected thereto) it is acknowledged that no Senior Secured Creditors have any Lien) and except for the obtaining of any consents or approvals, recording, filing, registration, giving of notice or other similar action as described in the definition of 'Required Filings' in Clause 1.1 of this Agreement or Schedule 2 (*Conditions Precedent*), paragraph 15 of Part A of Schedule 5 and Schedule 12 (*Permits*) (in the case of Schedule 2 (*Conditions Precedent*) and Schedule 12 (*Permits*) which consents, authorisations, registrations, filings and notices have, unless otherwise indicated on such schedule, been obtained or made and are in full force and effect):
  - (a) the security interests granted or purported to be granted to the Senior Secured Creditors pursuant to the Security Documents in the Project Security (i) (notwithstanding, without limitation, the Substitution) constitute as to Properties included in the Project Security existing on the date on which this representation is made or deemed to be made or repeated and, with respect to subsequently acquired Properties included in the Project Security, will constitute, a perfected security interest under all applicable law and/or the UCC and (ii) have, and, with respect to such subsequently acquired Properties, will have been perfected under all applicable law and/or the UCC, and grant the Senior Secured Creditors superior priority and rights in respect of the full amount of the Obligations over the rights of any third Persons now existing or hereafter arising whether by way of mortgage, lien, security interests, encumbrance, assignment or otherwise, subject to the rights and priorities of Permitted Liens;
  - (b) all such action as is necessary has been taken to establish, perfect and maintain the Senior Secured Creditors' rights in and to the Project Security, including any obtaining of consents or approvals, recording, filing, registration, giving of notice or other similar action; and
  - (c) each of the Security Documents is effective to create a legal, valid, binding and enforceable security interest in the Project Security described therein and products thereof.

- 164 -

Each of the Obligors has properly delivered or caused to be delivered to the Security Agent all Project Security that requires perfection of the Lien and security interest described above by possession.

- 8.2 As of the Fourth Amendment Effective Date, no authorization, approval or other action by, and no notice to or filing with, any Governmental Authority is required for either (a) the pledge or grant by the Obligors of the Liens purported to be created in favour of the Secured Parties pursuant to any of the Security Documents, or (b) the exercise by the Security Agent, or any of the other Secured Parties of any rights or remedies in respect of any Project Security (whether specifically granted or created pursuant to any of the Security Documents or created for by applicable law), except for registrations, filings, giving of notices or recordings contemplated by paragraph 8.1 of this Schedule 4 or as described in definition of 'Required Filings' in Clause 1.1 of this Agreement or Schedule 2 (*Conditions Precedent*) and Schedule 12 (*Permits*).
- 8.3 As of the Fourth Amendment Effective Date, except such as may have been filed in favour of the Security Agent as contemplated by paragraph 8.1 of this Schedule 4 or as set forth in Schedule 2 (*Conditions Precedent*) and Schedule 12 (*Permits*), no effective UCC financing statement, fixture filing or other instrument similar in effect covering all or any part of the Project Security is on file in any filing or recording office in the United States of America or elsewhere.

# 9. No Existing Defaults

- 9.1 No Event of Default has occurred which is continuing.
- 9.2 None of the Obligors, or, to the Company's knowledge, the Performance Bond Provider or any other Major Project Participant is in default under or with respect to any of its material Contractual Obligations under any of the Transaction Documents to which it is a party, which default (in the case only of a Major Project Participant other than an Obligor) could reasonably be expected to have a Material Adverse Effect.

# 10. **Taxes**

10.1 Each of the Obligors has filed, or caused to be filed, all tax and informational returns that are required to have been filed by it in any jurisdiction and all such tax and informational returns are correct and complete in all material respects. Each of the Obligors has paid all taxes shown to be due and payable on such returns and all other taxes and assessments payable by it, to the extent the same have become due and payable (other than (a) those taxes that it is contesting in good faith and by appropriate proceedings and (b) those taxes not yet due, *provided* that with respect to each of sub-paragraph (a) and (b) of this paragraph 10.1, it has established reserves therefor in amounts that are adequate for the payment thereof and are required by applicable GAAP.

- 165 -

- 10.2 None of the Obligors has incurred any material tax liability in connection with any of the Projects or the other transactions contemplated by the Transaction Documents which has not been disclosed in writing to the Intercreditor Agent, including as disclosed in the financial statements delivered to the Intercreditor Agent under this Agreement.
- 10.3 There are no Liens for Taxes on any of the Properties of any of the Obligors other than Liens permitted pursuant to paragraph 3(a) of Part B of Schedule 5 (*Covenants*) hereto.
- 10.4 The Company is resident only in its jurisdiction of incorporation for Tax purposes.
- 10.5 It is not required to make any Tax Deduction from any payment it may make under any Senior Finance Document to a Senior Secured Creditor.

# 11. Business, Debt, Etc.

The Obligors have not conducted any business other than a Permitted Business and the Substitution. The Obligors have no place of business outside the Macau SAR except (in the case of the Company) as otherwise permitted under Part B of Schedule 5 (*Covenants*) or, in the case of any other Obligor, outside its jurisdiction of incorporation. No Obligor has any outstanding Financial Indebtedness other than (in the case of the Company) Permitted Financial Indebtedness or (in the case of any other Obligor) as permitted by the Senior Finance Documents to which it is a party.

# 12. Environmental Laws

- 12.1 Each Obligor is in compliance with all applicable material Environmental Laws in all material respects and, so far as it is aware, there are no circumstances that could at any time be reasonably expected to prevent or interfere with such compliance.
- 12.2 No material Environmental Claim has been made which has not been fully discharged, released, satisfied or withdrawn.
- 12.3 So far as each Obligor is aware:
  - (a) neither the Site nor the Cotai Site contains any Hazardous Substances whose presence on the Site (or, as the case may be, Cotai Site) could reasonably be expected to affect the Company (or any other Obligor) or the Projects or the Cotai Project in a material and adverse manner; and
  - (b) there are no antiquities or obstructions on the Site or the Cotai Site which could reasonably be considered likely to prevent or impede the carrying out of either Project or the Cotai Project in any material respect.

<sup>- 166 -</sup>

## 13. Utilities

All material utility services (including, without limitation, gas, water and electrical interconnection) necessary for the Projects are or will be available at the Site as and when required.

### 14. [Not used]

### 15. Sufficiency of Funds

As of the first day of each Fiscal Quarter of the Company and as of each Advance Date, the Company together with each other Obligor has sufficient Funds to operate their respective businesses as they are then conducted and pay their respective debts when due.

#### 16. Sufficiency of Interests and Project Documents

- 16.1 The Company (or, prior to a Permitted Cotai Reorganisation and in the case of the Cotai Site, Palo) is the sole legal and beneficial owner of, and has good title to, or has a valid leasehold interest in, the land comprised in the Site and the Cotai Site, and each Obligor has good title to, or a valid license or leasehold interest in, all its Property, and, other than as provided by the Transaction Documents, none of its Property is subject to any claims, liabilities, obligations, charges or restrictions of any kind, nature or description (other than claims, liabilities, obligations, charges or restrictions that individually or in the aggregate could not reasonably be expected to materially interfere with the business or Property of the Company), or to any Lien except for Permitted Liens. None of the Pledged Stock is subject to any Lien except for Permitted Liens.
- 16.2 The Intercreditor Agent has received a true, complete and correct copy of each of the Major Project Documents (excluding any Resort Management Agreements) in effect or required to be in effect as of the date this representation is made or deemed to be made (including all exhibits, schedules, disclosure letters, modifications and amendments referred to therein or delivered or made pursuant thereto, if any). Each Major Project Document (excluding any Resort Management Agreements) is in full force and effect, enforceable against the Persons party thereto in accordance with its terms, subject only to bankruptcy and similar laws and principles of equity and except for any such Major Project Document that has expired by its terms.
- 16.3 All conditions precedent to the obligations of the respective parties (other than the Company or any other Obligor) under the Major Project Documents (excluding any Resort Management Agreements) have been satisfied or waived, except for such conditions precedent which by their terms cannot be met until a later stage in the construction or operation of the Projects or the Cotai Project, and the Obligors have no reason to believe that any such condition precedent which could reasonably be expected to have a Material Adverse Effect cannot be satisfied on or prior to the appropriate stage in the development, construction or operation of the Projects or the Cotai Project, as the case may be.

- 167 -

### 17. Intellectual Property

Each Obligor owns, or is licensed to use, all Intellectual Property necessary for the conduct of its business as currently conducted. To each Obligor's knowledge, no claim has been asserted or is pending by any Person challenging or questioning the use of any such Intellectual Property or the validity or effectiveness of any such Intellectual Property, nor does any Obligor know of any valid basis for any such claim, except with respect to Intellectual Property (other than any Intellectual Property related to or otherwise associated with the Company's use of the "Wynn" name), as could not, individually or collectively, reasonably be expected to have a Material Adverse Effect. The use by the Company of the Intellectual Property related to or otherwise associated with the Company's use of the "Wynn" name, does not infringe on the rights of any Person. The use by the Company's use of the "Wynn" name, does not infringe on the rights of any Person, which infringement, individually or collectively, could reasonably be expected to have a Material Adverse Effect.

### 18. **Projections**

As of the Third Amendment Signing Date, the Projections:

- (a) are based on assumptions the Company believes to be reasonable as to all legal and factual matters material to the estimates set forth therein;
- (b) are consistent with the provisions of the Transaction Documents in all material respects (including paragraph 1 of Part B and paragraph 2(c) of Part A of Schedule 5 (*Covenants*));
- (c) set forth all material costs and expenses anticipated to be incurred; and
- (d) represent the Company's views as to costs and expenses anticipated to be incurred in the manner contemplated by the Transaction Documents.

### 19. Fees and Enforcement

Other than amounts that have been paid in full or will have been paid in full by the Fourth Amendment Effective Date, no fees or taxes, including stamp, transaction, registration or similar taxes, are required to be paid for the legality, validity or enforceability of any of the Transaction Documents then in effect.

### 20. ERISA

Either (a) there are no Plans or Multiemployer Plans for the any Obligor or any Commonly Controlled Entity or (b) (i) each Obligor and each Commonly Controlled Entity have fulfilled their obligations (if any) under the minimum funding standards of ERISA and the Code for each Plan and for contributions to any Multiemployer Plan; (ii) each Plan is in compliance in all material respects with the currently applicable provisions of ERISA and the Code; (iii) neither any Obligor nor any Commonly Controlled Entity has incurred any liability to the PBGC or a Plan under Title IV of ERISA (other than liability or contributions for premiums due in the ordinary course). Assuming that the credit extended hereunder does not involve the

- 168 -

assets of any employee benefit plan subject to Title I of ERISA sponsored, monitored or contributed by for any Obligor or any Commonly Controlled Entity, neither the execution of the Transaction Documents nor the consummation of the transactions contemplated thereby will involve a "prohibited transaction" with respect to any Plans within the meaning of section 406 of ERISA or section 4975 of the Code which is not exempt under section 408 of ERISA or under section 4975(d) of the Code.

### 21. Subsidiaries and Beneficial Interest

- 21.1 The Company
  - (a) As of the Fourth Amendment Effective Date, Mr Wong Chi Seng legally and beneficially owns 20,010 Class A Shares (as defined in the Governing Documents of the Company), representing 10% of the total issued share capital and 10% of the Voting Stock of the Company;
  - (b) As of the Fourth Amendment Effective Date, Wynn HK, a company incorporated in the Hong Kong SAR, legally and beneficially owns 102,000 Class B Shares (as defined in the Governing Documents of the Company), representing 51% of the total issued share capital and 51% of the Voting Stock of the Company; and
  - (c) As of the Fourth Amendment Effective Date, Wynn International, a company incorporated in the Isle of Man, legally and beneficially owns 78,000 Class C Shares (as defined in the Governing Documents of the Company), representing 39% of the total issued share capital and 39% of the Voting Stock of the Company.

### 21.2 Wynn HK

As of the Fourth Amendment Effective Date, Wynn Holdings, a company incorporated in the Isle of Man, legally and beneficially owns 99% and beneficially owns 1%, and Wynn International, as nominee of Wynn Holdings, legally owns 1%, of the total issued share capital of Wynn HK.

21.3 Wynn Holdings

As of the Fourth Amendment Effective Date, Wynn International legally and beneficially owns 100% of Wynn Holdings.

21.4 Wynn International

As of the Fourth Amendment Effective Date, Wynn Asia 2, a company incorporated in the Cayman Islands, legally and beneficially owns 100% of the total issued share capital of Wynn International.

21.5 Palo

As of the Fourth Amendment Effective Date, the Company legally and beneficially owns 99.8%, and Wynn International and Wynn HK each legally and beneficially own 0.1%, of the total issued share capital of Palo.

- 169 -

21.6 Save as provided by the Security Documents or the Wong Option Agreement or as otherwise permitted by the Senior Finance Documents, there are no outstanding subscriptions, options, warrants, calls, rights or other agreements or commitments (other than stock options granted to employees or directors and directors' qualifying shares) of any nature relating to any Capital Stock issued by any Obligor (other than any Capital Stock in Wynn Asia 2). No Obligor has issued, or authorized the issuance of, any Disqualified Stock.

### 22. Labour Disputes and Acts of God

- 22.1 Neither the business nor the Properties of any Obligor, nor, to the knowledge of each Obligor, any other Major Project Participant, is affected by any fire, explosion, accident, drought, storm, hail, earthquake, embargo, act of God or of the public enemy, or other casualty or other event of force majeure, that could reasonably be expected to have a Material Adverse Effect.
- 22.2 There are no strikes, lockouts, stoppages, slowdowns or other labour disputes against any Obligor pending or, to the knowledge of each Obligor, threatened that (individually or in the aggregate) could reasonably be expected to have a Material Adverse Effect. Hours worked by and payment made to employees of each Obligor have not been in violation of any applicable Legal Requirement dealing with such matters that (individually or in the aggregate) could reasonably be expected to have a Material Adverse Effect. All payments due from each Obligor on account of employee health and welfare insurance that (individually or in the aggregate) could reasonably be expected to have a Material Adverse Effect if not paid have been paid or accrued as a liability on the books of such Obligor.

### 23. Liens

Except for Permitted Liens, none of the Obligors have secured or agreed to secure any Financial Indebtedness by any Lien upon any of their present or future revenues or other Properties or Capital Stock. None of the Obligors have outstanding any Lien or obligation to create Liens on or with respect to any of their Properties (including, without limitation, revenues), other than Permitted Liens and as provided in the Security Documents.

### 24. **Title**

Save, in the case of any such Property which, pursuant to the Security Documents, is expressed to be subject only to the floating charge granted pursuant to the Floating Charge, to the extent such floating charge has not been consolidated and the absence of such title could not reasonably be expected to have a Material Adverse Effect, each of the Obligors owns and has good, legal and beneficial title to the Property upon which it purports to grant Liens pursuant to the Security Documents, free and clear of all Liens, except Permitted Liens.

# 25. **Project Schedule**

Solely during the period prior to the Expansion Opening Date, the Project Schedule:

(a) accurately specifies in summary form the work that the Company proposes to complete in each calendar quarter from the CP Satisfaction Date through Final Completion of the Projects, all of which the Company expects to be achieved;

# - 170 -

- (b) is, to the Company's knowledge, based on assumptions the Company believes to be reasonable as to all legal and factual matters material to the estimates set forth therein;
- (c) is consistent with the provisions of the Transaction Documents in all material respects;
- (d) has been and will (in the case of any update) be prepared in good faith and with due care; and
- (e) fairly represents the Company's expectation as to the matters covered thereby as of its date or the date of its most recent update.

# 26. Location of Accounts and Records

The Company's (and each of its Subsidiaries') books of accounts and records are located at the Company's principal place of business in the Macau SAR.

# 27. Solvency

Each Obligor is, and after giving effect to:

- (a) the incurrence of all Financial Indebtedness;
- (b) the use of the proceeds of such Financial Indebtedness (including, in the case of the Company, the use of proceeds of Advances made under the Senior Finance Documents); and
- (c) obligations being incurred in connection with the Transaction Documents,

will be and will continue to be Solvent.

# 28. Plans and Specifications

The Plans and Specifications:

- (a) are, to the Company's knowledge as of the Third Amendment Effective Date, based on assumptions that the Company believes to be reasonable as to all legal and factual matters material thereto;
- (b) are, and except to the extent permitted under paragraph 15 of Part B of Schedule 5 (*Covenants*) will be from time to time, consistent with the provisions of the Transaction Documents;
- (c) have been prepared in good faith with due care; and
- (d) are accurate in all material respects and fairly represent the Company's expectations as to the matters covered thereby.

- 171 -

# 29. No subsidiaries

Save as permitted hereunder, other than Palo, the Company has no subsidiaries and does not legally or beneficially own any Capital Stock in any Person.

### 30. Pari Passu

The payment obligations under the Senior Finance Documents of each of the Obligors rank at least *pari passu* with all its other present and future unsecured and unsubordinated obligations, except for obligations mandatorily preferred by law applying to companies generally.

### 31. Insurance

The Company and each other Obligor is insured by insurers of recognized financial responsibility against such losses and risks and in such amounts as are prudent and customary in the businesses in which it is engaged and in any event in accordance with Schedule 7 (*Insurance*); and no Obligor has reason to believe that it will not be able to renew its existing insurance coverage as and when such coverage expires or to obtain similar coverage from similar insurers at a cost that could not reasonably be expected to have a Material Adverse Effect (other than as a result of general market conditions).

### 32. Fiscal Year

The fiscal year of each of the Obligors ends on 31 December of each calendar year.

### 33. Accuracy of Information, etc.

As of the Fourth Amendment Signing Date, or, in the case of the Information Memorandums, the respective dates thereof, no statement or information contained in each Information Memorandum or any other document, certificate or written statement furnished to any Secured Creditor, by or on behalf of any Obligor or the PASA Agent for use in connection with the transactions contemplated by any of the Senior Finance Documents, contained (when, in the case of any such document, certificate or written statement (other than the Information Memorandums), read as a whole with all such documents, certificates and written statements furnished on or prior to the Fourth Amendment Signing Date to such Senior Secured Creditor) any untrue statement of a material fact or omitted to state a material fact necessary in order to make the statements contained herein or therein not misleading. As of the Fourth Amendment Signing Date, or, in the case of the Information Memorandums, the dates thereof, the projections and pro forma financial information contained in the materials referenced above (including the Projections) are based upon good faith estimates and assumptions believed by management of the Obligors to be reasonable at the time made, it being recognized by the Senior Secured Creditors that such financial information may differ from the projected results set forth therein by a material amount. As of the Fourth Amendment Signing Date, there are no facts known to any Obligor that could, individually or collectively, reasonably be expected to have a Material Adverse Effect that has not been expressly disclosed in the Senior Finance Documents, in the

- 172 -

Information Memorandums or in any other documents, certificates and written statements furnished to the Senior Secured Creditors for use in connection with the transactions contemplated by the Senior Finance Documents.

#### 34. Site and Site Easements

- 34.1 The Site, the material Site Easements and the current use thereof comply in all material respects with all applicable Legal Requirements and with all Insurance Requirements.
- 34.2 No Taking has been commenced or, to the Company's knowledge, is contemplated with respect to all or any portion of the Site or the Site Easements or for the relocation of roadways providing access thereto except, in each case, as could not, individually or collectively, reasonably be expected to have a Material Adverse Effect.
- 34.3 There are no current, pending or, to the knowledge of the Company, proposed special or other assessments for public improvements or otherwise affecting the Site or the Site Easements, nor are there any contemplated improvements thereto that may result in such special or other assessments, in any case that could reasonably be expected to result in a material liability to the Company.
- 34.4 There are no outstanding options to purchase or rights of first refusal or restrictions on transferability affecting the Site or the material Site Easements (other than those set forth in the Senior Finance Documents and the Land Concession Contract or arising by mandatory operation of law).
- 34.5 Except as could not, individually or collectively, reasonably be expected to have a Material Adverse Effect, no Project building or structure or any appurtenance thereto or equipment thereon, or the use, operation or maintenance thereof, violates any restrictive covenant or encroaches on any easement or on any property owned by others.

### 35. Affiliate Agreements

Each Affiliate Agreement in effect as of the date this representation is made or deemed to be made has been entered into on arm's length terms and for full market value, in compliance with the Concession Contract and all other applicable Legal Requirements and otherwise in compliance with the terms hereof.

# 36. Wynn Asia 2

As of the Fourth Amendment Effective Date, Wynn Asia 2 has not carried on any business other than holding shares in Wynn International, entering into the Senior Finance Documents as a Wynn Obligor, entering into all agreements that are necessary or desirable for the purpose of (i) maintaining its corporate status (including without limitation agreements for the appointment of lawyers and auditors), (ii) the Substitution and (iii) complying with its obligations under the Finance Documents (as defined in the Deed of Appointment and Priority) (including without limitation the appointment of process agents).

- 173 -

### 37. Sanctions

No Obligor, member of the Group or Excluded Subsidiary or, to any Obligor's knowledge, any director or officer or any broker or other agent in acting on behalf of any of them, has violated or is currently in violation of any Anti-Terrorism Law or is a Restricted Party.

#### 38. Anti-bribery conduct

No Obligor, member of the Group or Excluded Subsidiary or, to any Obligor's knowledge, any director or officer or any broker or other agent in acting on behalf of any of them, has violated or is in violation of any provision of the U.S. Foreign Corrupt Practices Act of 1977 ("FCPA") or any similar Legal Requirements in respect of bribery, undue influence or similar matters (the FCPA and such similar Legal Requirements being "Anti-Bribery Laws") applicable to any Obligor, the Group, any Excluded Subsidiary or any of the individuals or entities specified above.

#### 39. Money-Laundering Laws

The operations of each Obligor, each member of the Group and each Excluded Subsidiary are and have been conducted in compliance with applicable money laundering statutes and the rules and regulations thereunder and any related or similar rules, regulations or guidelines (including any financial record keeping and reporting requirements), which in each case are issued, administered or enforced by any Governmental Authority having jurisdiction over it or to which it is subject (collectively, "**Money Laundering Laws**") and no action, suit or proceeding by or before any court or Governmental Authority involving an Obligor, any member of the Group or any Excluded Subsidiary with respect to Money Laundering Laws is pending and, to each Obligor's knowledge, no such actions, suits or proceedings are threatened or contemplated.

# 40. Development Account

As of the date of each withdrawal from a Development Account (as defined in the Term Facility Agreement), the Company (and/or Palo, as the case may be) has incurred costs or expended funds on the development of the Cotai Project (or will immediately upon the making of such withdrawal, have expended funds on the development of the Cotai Project) in an amount at least equal to the aggregate amount of all withdrawals from the Development Accounts (other than a withdrawal of funds for the immediate transfer of those funds to another Development Account). For the purposes of this paragraph 40, costs incurred or funds expended on the development of the Cotai Project include, but are not limited to, costs and expenses related to design, development, land acquisition, construction, site preparation, equipping, pre-opening expenses and capitalized interest.

- 174 -

## SCHEDULE 5 COVENANTS

### Part A Affirmative Covenants

The Company shall:

2.

(a)

- 1. **Financial Statements** Furnish to the Intercreditor Agent:
  - (a) as soon as available, but in any event not later than 90 days after the end of each Fiscal Year, a copy of the audited balance sheets (on a consolidated basis) of the Company prepared in accordance with IFRS, as at the end of such Fiscal Year and the related audited statements of income and of cash flows for such Fiscal Year, setting forth in each case in comparative form the figures for the previous Fiscal Year, reported on without a "going concern" or like qualification or exception, or qualification arising out of the scope of the audit, by the Auditors and prepared (in the case of the Company) for the Restricted Group only and without taking account of any contribution from any amounts derived from or under (i) the grant of any Subconcession, (ii) any Resort Management Agreement or (iii) any Excluded Project or Excluded Subsidiary; and
  - (b) as soon as available, but in any event not later than 45 days after the end of each Fiscal Quarter of each Fiscal Year, the unaudited (on a consolidated basis) balance sheets of the Company prepared in accordance with IFRS, as at the end of such quarter and the related unaudited statements of income and of cash flows for such quarter and the portion of the Fiscal Year through the end of such quarter, setting forth in each case in comparative form the figures for the previous year and prepared (in the case of the Company) for the Restricted Group only and without taking account of any contribution from any amounts derived from or under (i) the grant of any Subconcession, (ii) any Resort Management Agreement or (iii) any Excluded Project or Excluded Subsidiary.

All such financial statements shall be fair in all material respects (in the case of financial statements delivered pursuant to sub-paragraph (b) of this paragraph 1, subject to normal year-end audit adjustments) and were prepared in accordance with applicable GAAP applied consistently throughout the periods reflected therein and with prior periods (except as approved by such Auditors or Responsible Officer, as the case may be, and disclosed therein).

# Certificates; Other Information - Furnish to the Intercreditor Agent:

- concurrently with the delivery of any financial statements pursuant to paragraph 1 of this Part A:
  - (i) a certificate of a Responsible Officer of the relevant Obligor certifying that the statement is fair in all material respects (subject, where relevant, to normal year-end audit adjustments);

### - 175 -

- (ii) a certificate of a Responsible Officer of the relevant Obligor stating that, to the best of each such Responsible Officer's knowledge, the Obligor during such period has observed or performed all of its material covenants and other agreements, and satisfied every material condition contained in the Senior Finance Documents to which it is a party to be observed, performed or satisfied by it, and that such Responsible Officer has obtained no actual knowledge of any Default except as specified in such certificate or in the equivalent certificate for such period issued pursuant to this sub-paragraph (a)(ii) by a Responsible Officer of another Obligor;
- (iii) when applicable, in the case of quarterly or annual financial statements, a Compliance Certificate containing all information and calculations necessary for determining compliance by the Obligors with the provisions of this Agreement as of the last day of the applicable Fiscal Quarter or Fiscal Year, as the case may be; and
- (iv) in the case of the Company:
  - (1) a list of each of its accounts;
  - (2) upon the reasonable request of the Intercreditor Agent, a summary of the balances for each such account as at the end of the preceding Fiscal Quarter; and
  - (3) a certificate of a Responsible Officer of the Company setting out the amount(s) and details of any Permitted Financial Indebtedness, Subordinated Debt or Shareholder Loans (including, in each case, all terms and conditions thereof) made available to the Company during the preceding Fiscal Quarter;
- (b) for each calendar month during the period up to and including the first calendar month by which the Cotai Opening Date has occurred, deliver to the Facility Agents and the Intercreditor Agent, within 45 days following the end of the relevant calendar month, a status report for the Cotai Project (the "**Monthly Construction Period Report**") in reasonable detail;
- (c) no later than January 31 of each Fiscal Year, projections (each, together with the projections provided pursuant to paragraph 22 of Part A of Schedule 2 (*Conditions Precedent*), the "**Projections**") of the Company and the Restricted Group for such Fiscal Year (or portion thereof from the Original Project Opening Date through the end of such Fiscal Year), including a projected balance sheet of the Company as of the end of such Fiscal Year and the related statements of projected cash flow and profit and loss and the detailed assumptions supporting such Projections and prepared on a consolidated basis for the Restricted Group and without taking account of any contribution from any amounts derived from or under (i) the grant of any Subconcession, (ii) any Resort Management Agreement or (iii) any Excluded Project or Excluded Subsidiary and, no less frequently than quarterly, significant revisions, if any, of such Projections with respect to such Fiscal Year, which Projections shall in each case be accompanied by a certificate of a Responsible Officer of the

- 176 -

Company stating that such Projections are based on reasonable estimates, information and assumptions and that such Responsible Officer has no reason to believe that such Projections are incorrect or misleading in any material respect and which such Projections shall demonstrate compliance with paragraph 1 of Part B of this Schedule;

- (d) within 45 days after the end of each Fiscal Quarter after the Original Project Opening Date and prior to the Fourth Amendment Effective Date, a narrative discussion and analysis (the "Quarterly Operating Period Report") in form and substance substantially similar to the narrative discussions and analyses delivered by the Company to the Intercreditor Agent prior to the Second Amendment Signing Date of the financial condition and results of operations of the Company and the Restricted Group for such Fiscal Quarter and for the period from the beginning of the then current Fiscal Year (or if the then current Fiscal Year is the Fiscal Year in which the Original Project Opening Date has occurred, from the Original Project Opening Date) to the end of such Fiscal Quarter, as compared to the portion of the Projections covering such periods and to the comparable periods of the previous Fiscal Year;
- (e) promptly, and in any event within ten Business Days after any Major Project Document or Affiliate Agreement is terminated (save upon expiration in accordance with its terms) or amended (save in respect of amendments of a purely administrative or mechanical nature which do not, in each case, adversely affect the interests of any of the Finance Parties) or any new Major Project Document or Affiliate Agreement is entered into (other than, in each case, any Resort Management Agreement), or upon becoming aware of any material default by any Person or the occurrence of any event under a Major Project Document or Affiliate Agreement (other than, in each case, any Resort Management Agreement), or upon becoming aware of any determination provided thereunder, or any combination of the foregoing, would give rise to a right to terminate (a "Termination Event"), a written statement describing such event with copies of such amendments or new Major Project Document or Affiliate Agreement (including, in the case of the latter, a certificate from a Responsible Officer confirming the transactions contemplated therein comply with the requirements of paragraph 10 of Part B of Schedule 5 (*Covenants*)) and, with respect to any such terminations or material defaults, an explanation of any actions being taken by the Company with respect thereto;
- (f) promptly, and in any event within 30 days of the end of each Fiscal Year, deliver to the Intercreditor Agent a certificate certifying that the insurance requirements of Schedule 7 (*Insurance*) have been implemented and are being complied with;
- (g) [Not used];
- a copy of each written notice which is given under or pursuant to the Concession Contract, the Land Concession Contract or the Cotai Land Concession Contract by the Macau SAR to the Company (other than any administrative or routine notice which could not reasonably be expected to be prejudicial to the interests of the Finance Parties) promptly upon receipt of such notice;

- 177 -

- (i) at the same time as the giving of any written notice under or pursuant to the Concession Contract, the Land Concession Contract or the Cotai Land Concession Contract by the Company to the Macau SAR (other than any administrative or routine notice which could not reasonably be expected to be prejudicial to the interests of the Finance Parties), a copy of such notice; and
- (j) promptly, such additional financial and other information as the Intercreditor Agent may from time to time reasonably request.
- **Permits** Deliver to the Intercreditor Agent and (in the case of Permits made or issued by or with a Governmental Authority) the Technical Adviser (for such time as the Technical Adviser is appointed) promptly, but in no event later than 20 days after the receipt thereof by the Company, copies of:
- (a) all material Permits (including those specified in Part B of Schedule 12 (*Permits*)) in respect of the Projects (but not the Cotai Project) that are obtained or entered into by the Company or any other Obligor after the Signing Date; and
- (b) any material amendment, supplement or other modification to any such Permit received by the Company or any other Obligor after the Signing Date.
- Plans and Specifications Provide to the Technical Adviser copies of, and maintain at the Site, a complete set of the Plans and Specifications.

# 5. Technical Adviser

3.

4.

5.1

- Solely with respect to the Original Project and the Expansion, cooperate and cause the Prime Contractor and the PASA Agent to cooperate with the Technical Adviser in the performance of the Technical Adviser's duties. Without limiting the generality of the foregoing, the Company shall and shall cause the Prime Contractor to:
  - (a) communicate with and promptly provide all invoices, documents, plans and other information reasonably requested by the Technical Adviser relating to the work;
  - (b) provide the Technical Adviser with access to the Site and, subject to required safety precautions and reasonable site management restrictions, the construction areas; and
  - (c) solely in the case of the PASA Agent, provide the Technical Adviser with reasonable working space and access to telephone, copying and telecopying equipment at the Site (or such other location in reasonable proximity to the Site as the Projects near completion),

and the Company shall take reasonable measures to otherwise facilitate the Technical Adviser's review of the construction of the Projects and preparation of the certificates and reports required hereunder. The Company shall also maintain in Hong Kong or Macau a complete set and, promptly upon written request, provide the Technical

- 178 -

Adviser with reasonable access to and copies of, each first tier Subcontract entered into by the Prime Contractor (or any other Contractor who is party to a Project Document entered into with the Company that is not subject to a fixed price) with a contract price (or expected aggregate amount to be paid in the case of "cost plus" contracts) in excess of USD500,000 or its equivalent.

- 5.2 Solely with respect to the Original Project and the Expansion, in addition to any other consultation required under this Agreement, following the end of each quarter, upon the request of the Intercreditor Agent, consult with any such Person regarding any adverse event or condition identified in any report prepared by the Technical Adviser.
- 6. **Management Letters** Deliver to the Intercreditor Agent a copy of any "management letter" or other similar communication received by the Company or any other Obligor from the Auditors in relation to the Company's or any other Obligor's financial, accounting and other systems, management or accounts.
- 7. **Governmental and Environmental Reports -** Deliver (a) to the Intercreditor Agent and the Technical Adviser copies of all material reports required to be filed by the Company or any other Obligor with any Governmental Authority in connection with the construction of the Projects and (b) to the Intercreditor Agent (promptly upon the Intercreditor Agent's written request) the same, in respect of the Cotai Project.
- 8. **Payment of Obligations** To the extent not otherwise subject to valid subordination, standstill, intercreditor or similar arrangements, pay, discharge or otherwise satisfy (and ensure each other Obligor shall pay, discharge or otherwise satisfy) at or before maturity or before they become delinquent, as the case may be, all its material obligations of whatever nature, except where the amount or validity thereof is currently being contested in good faith by appropriate (under the circumstances) proceedings and reserves in conformity with applicable GAAP with respect thereto have been provided on the books of the Company (and, where relevant, such other Obligor).

# 9. Conduct of Business and Maintenance of Existence, etc.

- (a) Preserve, renew and keep in full force and effect (and ensure each other Obligor shall preserve, renew and keep in full force and effect) its corporate or limited liability company existence and remain a Subsidiary of Wynn Resorts;
- (b) take all reasonable action to maintain all rights, privileges, franchises, Permits and licenses necessary (and ensure each other Obligor shall take all reasonable action to maintain all rights, privileges, franchises, Permits and licenses necessary) in the normal conduct of its business, except to the extent that failure to do so could not (individually or in the aggregate) reasonably be expected to have a Material Adverse Effect; and
- (c) engage only in the businesses which are Permitted Businesses.

- 179 -

### 10. Maintenance of Property; Leases; Insurance

- 10.1 Keep (and ensure each other Obligor shall keep) all material Property and systems useful and necessary in its business in good working order and condition, ordinary wear and tear excepted.
- 10.2 Maintain (and ensure each other Obligor shall maintain) all material rights of way, easements, grants, privileges, licenses, certificates, and Permits necessary for the intended use of the Site and the Site Easements, except any such item the loss of which, individually or in the aggregate, could not reasonably be expected to materially and adversely affect or interfere with the Permitted Businesses or Property of the Company (or, as the case may be, any other Obligor).
- 10.3 Comply (and ensure each other Obligor shall comply) with the terms of each lease or other grant of rights in respect of land, including easement grants, so as to not permit any material uncured default on its part to exist thereunder, except, in each case, where non-compliance therewith could not reasonably be expected to materially and adversely affect or interfere with the Permitted Businesses or Property of the Company (or, as the case may be, any other Obligor).
- 10.4 At all times maintain in full force and effect:
  - (a) the insurance policies listed in Schedule 7 (*Insurance*); and
  - (b) such other insurances on and in relation to its business and assets against those risks and to the extent as is usual for companies carrying on the same or substantially similar business, and

otherwise comply with Schedule 7 (Insurance).

- 10.5 Ensure that each other Obligor maintains insurances on and in relation to its business and assets and against those risks and to an extent (i) as is usual for companies carrying on the same or substantially similar business (ii) which is, where relevant, at least comparable to that required to be maintained by the Company and (iii) in respect of such insurances, is maintained on substantially the same terms and conditions (including in respect of any reinsurance or any assignment or grant of other Liens in favour of the Security Agent over or in respect thereof) as those set out in Schedule 7 (*Insurance*).
- 10.6 Preserve and protect (and ensure each other Obligor preserves and protects) the Liens created pursuant to the Security Documents and, if any Lien (other than Permitted Liens) is asserted against any of the Project Security, promptly give (and ensure each other Obligor promptly gives) the Intercreditor Agent a written notice of such Lien and pay the underlying claim in full or take such other action so as to cause it to be released or bonded over in a manner reasonably satisfactory to the Intercreditor Agent.

#### 11. Inspection of Property; Books and Records; Discussions

(a) Keep (and ensure each other Obligor shall keep) proper books of records and account in which full, true and correct entries in conformity with applicable GAAP and all Legal Requirements.

- 180 -

- (b) Subject to any Macau Gaming Laws, data protection laws or other applicable Legal Requirements restricting such actions and, where no Event of Default has occurred which is continuing, prior reasonable request and notice, procure (and ensure each other Obligor procures) that each of the Agents or their respective nominees and the Technical Adviser (for such time as the Technical Adviser is appointed) (each a "**Relevant Person**") be allowed reasonable rights of inspection and access during normal business hours to the Site Facilities, the Cotai Site Facilities, the Projects, the Cotai Project, the Auditors and other Company and Obligor officers, the Company's and each other Obligor's accounting books and records and any other documents relating to the Projects or the Cotai Project as they may reasonably require, and so as not unreasonably to interfere with the operations of the Company or any Major Project Participant and to take copies of any documents inspected. Any information and documents made available for inspection by a Relevant Person pursuant to this paragraph shall be made available subject to customary confidentiality undertakings being executed by such Relevant Person.
- (c) For all expenditures with respect to which Advances under the Term Loan Facilities are made, the Company shall retain, until at least three years after delivery of the last report specified in paragraph 2(b) of this Part A for the Fiscal Year in which the last Advance was made under the Term Loan Facilities, all records and other documents evidencing such expenditures as are required hereunder to be attached to an Advance Request made in respect of any Term Loan Facility.
- Notices Promptly give notice to the Intercreditor Agent of:
  - (a) the occurrence of any Default;

12.

- (b) unless already notified pursuant to paragraph 2(e) of this Part A, any notice of termination (other than expiration in accordance with the terms thereof), default or any Termination Event under any Major Project Document;
- (c) (i) any fact, circumstance, condition or occurrence at, on, or arising from, any of the Site or the Site Easements that results in non-compliance with any Environmental Law that has resulted or could reasonably be expected to result in a Material Adverse Effect, and (ii) any Environmental Claim pending or, to the Company's or any other Obligor's knowledge, threatened against the Company or any other Obligor or, to the Company's or any other Obligor's knowledge, pending or threatened against any Contractor or any Subcontractor arising in connection with its occupying or conducting operations on or in respect of the Projects, the Site or the Site Easements which could reasonably be expected to have a Material Adverse Effect;
- (d) any change in the Responsible Officers of the Company or any other Obligor, and such notice shall include a certified specimen signature of any new officer so appointed and, if requested by any Facility Agent or the Intercreditor Agent, evidence of the authority of such new Responsible Officer;
- (e) any proposed material change in (i) the nature or scope of either Project or (ii) the business or operations of the Company or any other Obligor;

- 181 -

- (f) any notice of any material schedule delay delivered under the Construction Contract and all remedial plans by the Prime Contractor and updates thereof;
- (g) any "Substantial Completion" certificates or notices thereof delivered under any Major Project Documents (including any Certificate of Substantial Completion or any notices of Substantial Completion);
- (h) any (i) default or event of default (or alleged default) under any Contractual Obligation of the Company or any other Obligor or
   (ii) litigation, investigation or proceeding which may exist at any time between the Company or any other Obligor and any
   Governmental Authority, that in either case, could reasonably be expected to have a Material Adverse Effect;
- (i)
- within twenty days after the Company receives actual notice of a Proceeding or Proceedings instigated or threatened in writing involving an alleged liability of, or claims against or affecting, the Company, Wynn HK, Wynn International, Wynn Asia 2 or any other Obligor equal to or greater than USD25,000,000; or
- (ii) upon any officer of the Company or any other Obligor obtaining knowledge thereof, the instigation or written threat of any action, suit, proceeding (whether administrative, judicial or otherwise), governmental investigation or arbitration against or affecting the Company or any other Obligor, or any Property of the Company or any other Obligor (collectively, "Proceedings") not previously disclosed in writing by the Company to the Lenders (including pursuant to paragraph (i)(i) above) that, in any case (i) is reasonably likely to give rise to a Material Adverse Effect or (ii) seeks to enjoin or otherwise prevent the consummation of, or to recover any damages or obtain relief as a result of, the transactions under the Transaction Documents, or any material development in any such Proceeding,

in each case together with a schedule describing such Proceeding or Proceedings and such other information as the Intercreditor Agent may reasonably require and promptly after request by the Intercreditor Agent, such other information as may be reasonably requested by the Intercreditor Agent to enable the Intercreditor Agent and its counsel to evaluate any such Proceeding or Proceedings;

(j) the following events, as soon as possible and in any event within 30 days after the Company or any other Obligor knows or has reason to know thereof: (i) the occurrence of any Reportable Event with respect to any Plan, a material failure to make any required contribution to a Plan, the creation of any Lien in favour of the PBGC or a Plan or any withdrawal from, or the termination, Reorganization or Insolvency of, any Multiemployer Plan or (ii) the institution of proceedings or the taking of any other action by the PBGC, the Company, any other Obligor or any Commonly Controlled Entity or any Multiemployer Plan with respect to the withdrawal from, or the termination, Reorganization or Insolvency of, any Plan;

- 182 -

- (k) solely during the period prior to the Expansion Opening Date, any event, occurrence or circumstance which reasonably could be expected to render the Company incapable of or prevent the Company from meeting any material obligation of the Company under the Construction Contract or the other Major Project Documents as and when required thereunder; and
- (l) any development or event that has had or could reasonably be expected to have a Material Adverse Effect.

Each notice pursuant to this paragraph shall be accompanied by a statement of a Responsible Officer of the Company setting forth details of the occurrence referred to therein and stating what action the Company and/or the relevant Obligor proposes to take with respect thereto.

# 13. Environmental Laws; Permits

- 13.1 Inform the Intercreditor Agent promptly on becoming aware of any, or any anticipated, release, emission, discharge or disposal of any matter that might reasonably be expected to form the basis for any Environmental Claim involving an alleged liability of, or claims against or affecting, the Company or any other Obligor equal to or greater than an aggregate amount of 100,000,000 Patacas or its equivalent.
- 13.2 Comply (and ensure each other Obligor complies) with all applicable Environmental Laws and Environmental Licences (save for any failure to comply that could not reasonably be expected to have a Material Adverse Effect).
- 13.3 Provide copies of any notices from any Governmental Agency of non-compliance with any material Environmental Law or Environmental Licence and any notices of any Environmental Claims to the Intercreditor Agent.
- 13.4 Deliver to the Intercreditor Agent:
  - (a) any notice that any Governmental Authority may condition approval of, or any application for, any Permit held by the Company or any other Obligor on terms and conditions that are materially burdensome to the Company or any other Obligor, or to the operation of any of its businesses or any Property owned, leased or otherwise operated by the Company or any other Obligor in each case in a manner not previously contemplated; and
  - (b) with reasonable promptness, such other documents and information as from time to time may be reasonably requested by the Intercreditor Agent in relation to any matters disclosed pursuant to this paragraph 13.

# 14. Hedging

Comply with the Hedging Arrangements and not enter (and ensure no other Obligor enters into) into any other Swap Agreement or derivative transaction unless:

- (a) the Company is in compliance with the Hedging Arrangements; and
- (b) unless such arrangement is with a Lender which has executed a duly completed Hedging Counterparty's Deed of Accession and Finance Party

Accession Undertaking (also executed, in the case of the latter, by the Intercreditor Agent and all other parties hereto acting through the Intercreditor Agent for this purpose (the authority for which is hereby conferred on the Intercreditor Agent) and promptly delivered by the Intercreditor Agent to the Security Agent) and is on the same terms as those comprised in the Hedging Arrangements, the counter-party thereto has no Lien or any right to share in any Lien over any Property of the Company.

#### 15. Additional Collateral, Discharge of Liens, etc.

- 15.1 With respect to any Property acquired by the Company or any other Obligor after the Signing Date as to which the Senior Secured Creditors do not have a perfected security interest, subject to compliance with applicable Macau Gaming Laws and restrictions on the grant of Liens permitted pursuant to paragraph 3 of Part B of this Schedule, promptly:
  - (a) execute and deliver (and ensure each other Obligor executes and delivers) to the Intercreditor Agent such amendments to the Security Documents or execute and deliver such other documents as the Intercreditor Agent, acting reasonably, deems necessary or advisable to grant to the Senior Secured Creditors a security interest in such Property; and
  - (b) take all actions necessary or advisable to grant to the Senior Secured Creditors a perfected first priority security interest in such Property (subject to Permitted Liens).

In addition to the foregoing, in the event any such Property acquired after the Signing Date consists of land or other Property with respect to which a recording or registration in the real property or other records of an appropriate jurisdiction is required or advisable in order to perfect a security interest therein, promptly (and, in any event, within 30 days following the date of such acquisition):

- (i) execute and deliver (and ensure each other Obligor executes and delivers) a mortgage, substantially in the form of the Mortgage (with such modifications, if any, as are necessary to comply with Legal Requirements that the Security Agent may reasonably request), such mortgage to be recorded in the real property records of the appropriate jurisdiction, or execute and deliver to the Security Agent for recording a supplement to the Mortgage, in either case pursuant to which the Company or other Obligor grants to the Senior Secured Creditors a Lien on such Property subject only to Permitted Liens; and
- (ii) execute and/or deliver (and ensure each other Obligor executes and/or delivers) such other documents or provide such other information in furtherance thereof as the Security Agent may reasonably request, including delivering documents and taking such other actions which would have been required pursuant to Clause 2 (*Conditions Precedent*) if such Property were part of the Project Security at the CP Satisfaction Date.

- 184 -

15.2 Notwithstanding anything to the contrary in this paragraph 15, sub-paragraph 15.1 shall not apply to:

- any Property created or acquired after the Signing Date, as applicable, as to which the Intercreditor Agent has reasonably determined that the collateral value thereof is insufficient to justify the difficulty, time and/or expense of obtaining or maintaining a perfected security interest therein;
- (ii) any Subconcession proceeds or any Property purchased with Subconcession proceeds (which is neither comprised in any Project or the Cotai Project nor necessary or desirable to ensure the full benefit of any Project or the Cotai Project to the Company);
- (iii) any works of art, antiquities, precious stones, precious metals or other similar assets (which are not of a type that will become affixed to the Site or the other Site Facilities such that the removal thereof could reasonably be expected to materially interfere with the ongoing ordinary course of operations of either Project);
- (iv) any Property comprised in or derived from Resort Management Agreements (which is neither comprised in any Project or the Cotai Project nor necessary or desirable to ensure the full benefit of any Project or the Cotai Project to the Company save for amounts received by the Company or another Obligor in respect of any Excluded Project located on the Cotai Site and the entry into or performance of obligations under, any Cotai Resort Management Agreement); or
- (v) stock or other ownership interests in Excluded Subsidiaries and Excluded Projects.
- 15.3 Notwithstanding anything to the contrary in this paragraph 15 or the Senior Finance Documents, (unless otherwise agreed by the Company and the Intercreditor Agent or the Security Agent) neither the Company nor Palo shall at any time be required to grant Security over its rights under any contract for the design, development, construction, operation or maintenance of the Cotai Project entered into between a contractor (or an equipment supplier or a service provider) and the Company or Palo, as the case may be.

### 16. Use of Proceeds and Revenues

- 16.1 Use the proceeds of each of the Facilities only for the purposes specified or allowed in this Agreement and the Facility Agreement relating to such Facility.
- 16.2 Ensure that all of its funds and those of any other Obligor and all other amounts received by it or any other Obligor (other than any Subconcession proceeds or amounts received by the Company or another Obligor in respect of any Excluded Subsidiaries or Excluded Projects or its entry into, or performance of its obligations under, any Resort Management Agreement (but including amounts received by the Company or another Obligor in respect of any Excluded Project located on the Cotai Site and the entry into, or performance of obligations under, any Cotai Resort Management Agreement)) are deposited into any Account or combination of Accounts and otherwise in accordance with the provisions of this Agreement and that it and each other Obligor otherwise complies with Schedule 6 (*Accounts*).

- 185 -

### 17. Compliance with Laws, Project Documents, etc.; Permits

- 17.1 Comply (and ensure that each Obligor, each member of the Group and each Excluded Subsidiary complies) in all material respects with all material Legal Requirements (including, without limitation, all Anti-Terrorism Laws, Money Laundering Laws and Anti-Bribery Laws), and its Governing Documents.
- 17.2 Comply (and ensure each other Obligor complies), duly and promptly, in all material respects with its material obligations and enforce all of its material rights under all Project Documents, except:
  - (a) in the case of Project Documents (other than the Major Project Documents) where the failure to comply could not reasonably be expected to have a Material Adverse Effect; and
  - (b) in the case of any Resort Management Agreement, where the failure to comply could not reasonably be expected to threaten the Concession Contract, the Cotai Land Concession Contract or the Land Concession Contract.
- 17.3 From time to time obtain, maintain, retain, observe, keep in full force and effect and comply (and ensure each other Obligor from time to time obtains, maintains, retains, observes, keeps in full force and effect and complies) in all material respects with the terms, conditions and provisions of all Permits made or issued by or with a Governmental Authority as shall now or hereafter be necessary under applicable laws.
- 17.4 Comply (and ensure each of its Affiliates and, where relevant, each Major Project Participant complies) with all Legal Requirements related to the Land Concession Contract, the Cotai Land Concession Contract and the Concession Contract.
- 18. **Pari Passu Ranking** Procure that the obligations of the Company and each other Obligor under the Senior Finance Documents do and shall rank at least *pari passu* with all its other present and future unsecured and unsubordinated obligations, except for obligations mandatorily preferred by law applying to companies generally.
- 19. **Further Assurances** From time to time execute and deliver, or cause to be executed and delivered, such additional instruments, certificates or documents, and take all such actions, as any of the Intercreditor Agent or the Security Agent may reasonably request, for the purposes of implementing or effectuating the provisions of the Senior Finance Documents, or of more fully perfecting or renewing the rights of the Senior Secured Creditors with respect to the Project Security (or with respect to any additions thereto or replacements or proceeds or products thereof or with respect to any other Property acquired after the date of the Senior Finance Documents by the Company or any other Obligor which may be deemed to be part of the Project Security) pursuant to the Senior Finance Documents. Upon the exercise by the Intercreditor Agent, the Security Agent or any other Senior Secured Creditor of any power, right, privilege or remedy pursuant to any of the Senior Finance Documents which requires any consent, approval, notification, registration or authorisation of any Governmental Authority, the Company shall execute and deliver, or will cause the

- 186 -

execution and delivery of, all applications, certifications, instruments and other documents and papers that the Intercreditor Agent, the Security Agent or such Senior Secured Creditor may reasonably be required to obtain from the Company or any other Obligor for such governmental consent, approval, notification, registration or authorisation.

- 20. **Diligent Construction of the Project** The Company shall take or cause to be taken all action, make or cause to be made all contracts and do or cause to be done all things necessary to construct the Projects diligently and in accordance with the relevant construction contracts, and the Plans and Specifications, the Project Schedule and the other Transaction Documents. The Company shall construct the Cotai Project in accordance with the Cotai Land Concession Contract.
- 21. **Retainage Amounts** Withhold or cause to be withheld from each Contractor party to a Major Project Document such retainage from any payment to be made to such Contractor as is permitted by such Major Project Document.
- 22. **Preserving Project Security** Undertake and cause the other Obligors to undertake, all actions which are necessary or appropriate in the reasonable judgment of the Intercreditor Agent to:
  - (a) maintain the Senior Secured Creditors' respective security interests under the Security Documents in the Project Security in full force and effect at all times (including the priority thereof); and
  - (b) preserve and protect the Project Security and protect and enforce the Company's or, as the case may be, other Obligor's rights and title and the respective rights of the Senior Secured Creditor to the Project Security,

including the making or delivery of all filings and registrations, the payments of fees and other charges, the issuance of supplemental documentation, the discharge of all claims or other liens other than Permitted Liens adversely affecting the respective rights of the Senior Secured Creditors to and under the Project Security and the publication or other delivery of notice to third parties.

### 23. [Not used]

- 24. **Termination of Concession Contract** Notify the Intercreditor Agent promptly upon receiving:
  - (a) notice of any consultations with the Macau SAR as contemplated by paragraph B1(c) of the Gaming Concession Consent Agreement in relation to any termination of the Concession Contract;
    - (b) notice of any consultations with the Macau SAR (i) as contemplated by paragraph C1 of the Land Concession Consent Agreement in relation to any termination or rescission of the Land Concession Contract or (ii) in relation to any termination or rescission of the Cotai Land Concession Contract;
  - (c) notice of any negotiations with the Macau SAR pursuant to article 83 of the Concession Contract;

- 187 -

- (d) any notice from the Macau SAR pursuant to clause 3 of article 80 of the Concession Contract; or
- (e) any notice from the Macau SAR pursuant to clause 4 of article 80 of the Concession Contract,
- and keep the Intercreditor Agent fully appraised thereof.

# Capital Expenditures

25.

Solely with respect to the Projects (and not with respect to the Cotai Project or any Excluded Project) in respect of any:

- (a) contract to be entered into in respect of Capital Expenditure by any Obligor with a total contract price payable (or expected aggregate amount to be paid in the case of "cost plus" contracts) or which may otherwise involve liabilities, actual or contingent, in each case in an amount in excess of USD75,000,000 or its equivalent; or
- (b) contracts (whether related or not) entered into, or to be entered into, in respect of Capital Expenditure by any Obligor in any Fiscal Year which when taken together have a total contract price payable (or expected aggregate amount to be paid in the case of "cost plus" contracts) or which may otherwise involve liabilities, actual or contingent, in an aggregate amount in excess of USD75,000,000 or its equivalent,

deliver to the Intercreditor Agent:

- (1) a certificate of a Responsible Officer of the Company certifying that it has sufficient Funds to achieve construction completion of the Projects and to operate its business as it is then conducted and pay its debts when due in respect of the relevant contract to be entered into or otherwise; and
- (2) such other substantiating information and evidence in respect of the relevant contract or contracts that the Intercreditor Agent (acting on the instructions of any Lender) may reasonably require.

#### 26. Sufficiency of Funds

Ensure that it together with each other Obligor has, at all times, sufficient Funds to operate its business as it is then conducted and pay its debts when

# due. 27. Additional Obligors

- 27.1 Notwithstanding any other provision to the contrary in this Agreement or the other Senior Finance Documents, promptly (and in any case within 30 days of such formation, creation or acquisition) shall ensure that any newly formed, created or acquired (directly or indirectly) Subsidiary (other than any Excluded Subsidiary) of the Company or any other Obligor, agrees to be subject to the obligations under the terms of the Senior Finance Documents and to grant such Liens in respect of its Property as may be required by the terms of the Senior Finance Documents by:
  - (a) giving not less than 5 Business Days prior written notice to the Intercreditor Agent (which shall promptly notify the Lenders) of such formation, creation or acquisition;

- 188 -

- (b) the Company confirming that no Default is continuing or would occur as a result of the formation, creation or acquisition of such proposed Subsidiary or as a result of such proposed Subsidiary becoming an Obligor; and
- (c) delivering to the Intercreditor Agent all of the documents and other evidence required by it in relation to the relevant Subsidiary (in form and substance reasonably satisfactory to the Intercreditor Agent), including (without limitation):
  - (i) accession by the relevant Subsidiary to the Subordination Deed and the Guarantee; and
  - (ii) any other documentation and evidence reasonably required by the Intercreditor Agent to ensure that the relevant Subsidiary makes such representations and warranties and is subject to such obligations under the terms of the Senior Finance Documents and grants such Liens in respect of its Property,

in each case on such terms and by such instrument or combination of instruments as the Intercreditor Agent reasonably deems necessary or desirable; and provided that the Intercreditor Agent shall have informed the Company of all such documents and evidence to be required by the Intercreditor Agent within a reasonable time after receiving the notice set forth in paragraph 27.1(a).

- 27.2 The Intercreditor Agent shall notify the Company and the Lenders promptly upon being satisfied that it has received (in form and substance reasonably satisfactory to it) all the documents and other evidence referred to in paragraph 27.1(c) above.
- 27.3 Following the giving of any notice pursuant to paragraph 27.2 above, if the accession of such additional Obligor obliges the Intercreditor Agent or any Lender to comply with "know your customer" or similar identification procedures in circumstances where the necessary information is not already available to it, promptly upon the request of the Intercreditor Agent or any Lender supply, or procure the supply of, such documentation and other evidence as is reasonably requested by the Intercreditor Agent or such Lender or any prospective new Lender) in order for the Intercreditor Agent or such Lender or any prospective new Lender) in order for the Intercreditor Agent or such Lender or any prospective new Lender to carry out and be satisfied it has complied with all necessary "know your customer" or other similar checks under all applicable laws and regulations pursuant to the accession of such Subsidiary to this Agreement as an additional Obligor.

# 28. Cotai Site Excluded Project

28.1 Ensure that to the extent any Excluded Project located on a portion of the Cotai Site is funded by the proceeds of any Advance, such funding is applied solely towards the financing or refinancing of the development costs incurred in relation to such Excluded Project (it being agreed that such development costs include, but are not

- 189 -

limited to, costs and expenses related to design, development, land acquisition, construction, site preparation, equipping, pre-opening expenses and capitalized interest).

28.2 Ensure that prior to the earlier of the entry into of any contract for and the commencement of the construction of an Excluded Project on a portion of the Cotai Site, the conditions set out in paragraphs (i) to (iii) (inclusive) of the definition of "Excluded Project" in Clause 1.1 (*Definitions*) are met and that such conditions continue to be met for the duration of the period during which such Excluded Project remains located on a portion of the Cotai Site.

## 29. Sanctions laws and regulations

Ensure that none of the funds or assets of any Obligor, the Group or any Excluded Subsidiary that are used to pay any amount due under the Finance Documents shall constitute funds or assets knowingly obtained from transactions with or relating to Restricted Parties or countries which are the subject of any Anti-Terrorism Laws.

#### **Negative Covenants**

The Company shall not directly or indirectly:

# 1. Financial Condition Covenants

(a)

*Leverage Ratio* - Permit the Leverage Ratio as at the last day of any period of four full consecutive Fiscal Quarters ending on any Quarterly Date in any Fiscal Year set forth below to exceed the ratio set forth below opposite such Fiscal Year:

| Fiscal Year | Leverage<br>Ratio |
|-------------|-------------------|
| 2012        | 3.75:1            |
| 2013        | 4.00:1            |
| 2014        | 4.50:1            |
| 2015        | 4.75:1            |
| 2016        | 5.00:1            |
| 2017        | 4.75:1            |
| 2018        | 4.50:1            |

*provided* that, for purposes of calculating EBITDA pursuant to this paragraph 1(a) for any period which is less than four full Fiscal Quarters, EBITDA shall be calculated on an annualised basis.

(b) *Interest Coverage Ratio* - Permit the Interest Coverage Ratio for each period of four full consecutive Fiscal Quarters ending on each Quarterly Date to be less than 2.0:1 at any time.

# 2. Limitation on Financial Indebtedness

2.1 Create, incur, assume or suffer to exist (or permit any other Obligor to create, incur, assume or suffer to exist) any Financial Indebtedness, except:

(a) Financial Indebtedness of the Company or Palo created under any Senior Finance Document (other than an Additional Lender Facility Agreement), the Performance Bond Facility, any Shareholder Loan, any Subordinated Debt or any Guarantee Obligations represented by the guarantee required to be issued pursuant to clause 7 of the Land Concession Contract or clause 7 of the Cotai Land Concession Contract;

# - 191 -

- (b) Financial Indebtedness of the Company or Palo (including Capital Lease Obligations) secured by Liens permitted by paragraph 3 (l) of this Part B provided that any recourse in respect of such Financial Indebtedness is limited solely to the Property secured by such Liens;
- (c) Financial Indebtedness of the Company or Palo in an aggregate principal amount not to exceed USD50,000,000 or its equivalent at any time *provided* that the provider (or providers) of such Financial Indebtedness do not (in respect of such Financial Indebtedness) have the benefit of any Lien over or in respect of the Project Security (or any part thereof) (save for a Lien referred to in paragraph 3(i) of Part B of Schedule 5 (*Covenants*) to the extent that such attachment or judgment Lien arises in respect of that Financial Indebtedness above or a Lien referred to in paragraph 3(m) of Part B of Schedule 5 (*Covenants*));
- (d) Financial Indebtedness of the Company or Palo incurred for the purpose of repaying the balance of the principal amount owing to all Lenders *provided* the Intercreditor Agent is satisfied that upon the incurrence of such Financial Indebtedness or immediately thereafter the Release Date (without reference to paragraph (b) of the definition of Release Date) will occur;
- (e) Financial Indebtedness of the Company or Palo created, incurred, assumed or suffered to exist during the period on and following the Expansion Opening Date *provided* that the Leverage Ratio as at the most recent Quarterly Date, if determined on a *pro forma* basis after giving effect to the creation, assumption or sufferance to exist of such Financial Indebtedness (when taken together with all such other Financial Indebtedness of the Company permitted pursuant to this paragraph 2.1(e)), would not exceed the ratio set forth opposite that Quarterly Date in paragraph 1(a) and *further provided* that the provider (or providers) of such Financial Indebtedness do not (in respect of such Financial Indebtedness) have the benefit of any Lien over or in respect of the Project Security (or any part thereof) (save for a Lien referred to in paragraph 3(i) of Part B of Schedule 5 (*Covenants*) to the extent that such attachment or judgment Lien arises in respect of that Financial Indebtedness above or a Lien referred to in paragraph 3(m) of Part B of Schedule 5 (*Covenants*));
- (f) Financial Indebtedness of the Company, pursuant to one or more Additional Lender Facility Agreements:
  - (i) in an aggregate principal amount not to exceed USD200,000,000 or its equivalent;
  - (ii) on terms no more favourable to any creditor to which such Financial Indebtedness is owed than those to which (i) the Revolving Credit Facility made available pursuant to the Revolving Credit Facility Agreement referred to in paragraph (a) of the definition of "Revolving Credit Facility Agreement" in Clause 1.1 (*Definitions*) of this Agreement are subject (in respect of any Additional Lender Facility that is a revolving credit facility) or (ii) the Term Facility is subject (in respect of any Additional Lender Facility that is a term loan facility) (*provided that* such Financial Indebtedness will be tranched as agreed

- 192 -

between the Company and all such creditors (A) between US dollars and Hong Kong dollars and (B) further tranched (if required) to address any requirements of any creditor for its participations thereunder to be applied solely to finance costs or other amounts unconnected with the operation of casino games of chance or other forms of gaming);

(iii) provided (x) each such creditor has executed and delivered to the Intercreditor Agent a duly completed Additional Lender's Accession Deed and Finance Party Accession Undertaking (also executed, in the case of the latter, by the Intercreditor Agent and all parties hereto acting through the Intercreditor Agent for this purpose (the authority for which is hereby conferred on the Intercreditor Agent) and promptly delivered by the Intercreditor Agent to the Security Agent), whereupon, provided the other conditions of this paragraph 2.1(f) have also been satisfied, such creditor shall become a party hereto as an Additional Lender and that creditor and each of the other parties hereto shall assume obligations towards, and acquire rights against, one another accordingly or (y) the Financial Indebtedness is otherwise on terms satisfactory to the Intercreditor Agent concerning the relationship between each creditor and the Senior Secured Creditors, including the exercise of rights against the Company and any other Obligor and the sharing and enforcement of the Security and any other liens over Property of the Company or any other Obligor; and

*further provided* that it shall be a condition to the making of any Advance under an Additional Lender Facility Agreement that the Company shall, prior to the making of such Advance, procure that evidence that the Macau SAR government has (a) consented to the increase in the Company's financial indebtedness contemplated by the proposed Additional Lender Facility Agreement and (b) confirmed that the Gaming Concession Consent Agreement and the Land Concession Consent Agreement (as amended, consolidated, supplemented, novated or replaced from time to time prior to the proposed date of such Additional Lender Facility Agreement) continue to apply with respect to such financial indebtedness) together with a legal opinion of the Macanese legal adviser to the Senior Secured Creditors, in respect thereof and in respect of the validity, enforceability, effectiveness and ranking of the Mortgage (as well as confirming the Secured Obligations thereunder also cover such Additional Lender Facility) is provided to the Intercreditor Agent (in each case in form and substance satisfactory to the Intercreditor Agent, acting reasonably);

- (g) Financial Indebtedness of a Wholly Owned Subsidiary of the Company which is an Obligor (a "**Subsidiary Obligor**") to any other Subsidiary Obligor or to the Company;
- (h) Financial Indebtedness of the Company to any Subsidiary Obligor;
- (i) Guarantee Obligations of the Company or any Obligor incurred, assumed or suffered to exist with respect to any WML Debt *provided that* such WML

- 193 -

Debt (i) is not secured by any Lien granted by any Obligor or any member of the Group, (ii) does not have any scheduled amortisation prior to the Final Repayment Date of any Facility (as determined: (A) in respect of the Term Facility, as of the Fourth Amendment Effective Date; and (B) in respect of any Additional Lender Facility which provides for a Term Loan Facility, as of the date of that Additional Lender Facility Agreement) and (iii) has a stated maturity that extends beyond the later of (A) the Revolving Credit Facility Termination Date (as determined as of the Fourth Amendment Effective Date), (B) the Final Repayment Date of the Term Facility (as determined as of the Fourth Amendment Effective Date); and (C) the Final Repayment Date of any Additional Lender Facility which provides for a Term Loan Facility (as determined as of the date of the date of the Additional Lender Facility Agreement); and

(j) other Financial Indebtedness approved in writing by the Intercreditor Agent,

#### (together, "Permitted Financial Indebtedness").

#### Limitation on Liens

3.

Create, incur, assume or suffer to exist (or permit any other Obligor to create, incur, assume or suffer to exist) any Lien upon any of its Property, whether now owned or hereafter acquired, except for:

- (a) Liens for taxes not yet due or which are being contested in good faith by appropriate proceedings, *provided* that adequate reserves with respect thereto are maintained on the books of the Company and the relevant Obligor in conformity with applicable GAAP;
- (b) carriers', warehousemen's, mechanics', materialmen's, repairmen's or other like Liens arising in the ordinary course of business for amounts which are not overdue for a period of more than 30 days or that are being contested in good faith by appropriate proceeding (such contest proceedings conclusively operating to stay the sale of any portion of the Project Security on account of such Lien);
- (c) pledges or deposits in connection with workers' compensation, unemployment insurance and other social security legislation *provided* that if such pledges are being contested, appropriate reserves (determined in accordance with the applicable GAAP are maintained on the books of the Company and the relevant Obligor;
- (d) deposits by or on behalf of the Company (or the Obligor) to secure the performance of bids, trade contracts (other than for borrowed money), leases, statutory obligations, appeal bonds and other obligations of a like nature incurred in the ordinary course of business;
- (e) easements, rights-of-way, restrictions, encroachments and other similar encumbrances and other minor defects and irregularities in title, in each case incurred in the ordinary course of business that, in the aggregate, do not materially detract from the value of the Property subject thereto or materially interfere with the ordinary conduct of the business of the Company (or the Obligor);

- 194 -

- (f) Liens created pursuant to paragraph 15.2 of Part A of this Schedule 5;
- (g) Liens created pursuant to the Security Documents;
- (h) licenses of patents, trademarks and other intellectual property rights granted by the Company (or the Obligor) in the ordinary course of business and not interfering in any material respect with the ordinary conduct of the business of the Company (or the Obligor);
- (i) any attachment or judgment Lien not constituting an Event of Default;
- Liens in favour of customs and revenue authorities arising as a matter of law to secure payment of customs duties in connection with the importation of goods;
- (k) any zoning or similar law or right reserved to or vested in any Governmental Authority to control or regulate the use of any Site and Site Easements or the Cotai Site and Cotai Site Easements;
- (l) Liens securing Financial Indebtedness of the Company or Palo incurred pursuant to paragraph 2.1(b) of this Part B to finance the acquisition of fixed or capital assets *provided* that:
  - (i) such Liens shall be created substantially simultaneously with the acquisition of such fixed or capital assets (or the refinancing of such Financial Indebtedness as otherwise permitted hereunder);
  - (ii) such Liens do not at any time encumber any Property other than the Property (and proceeds of the sale or other disposition thereof) financed by such Financial Indebtedness; and
  - (iii) the Property financed by such Financial Indebtedness is not of a type that will become affixed to the Site or the other Site Facilities or the Cotai Site or the other Cotai Site Facilities such that the removal thereof could reasonably be expected to materially interfere with the ongoing ordinary course of operations of either Project or the Cotai Project;
- (m) Liens on cash deposited with, or held for the account of, the Company or Palo securing reimbursement obligations owing by the Company or Palo and permitted to be incurred by it pursuant to paragraph 2.1(c) above under performance bonds, guaranties, commercial or standby letters of credit, bankers' acceptances or similar instruments or the guarantee required to be issued pursuant to clause 7 of the Land Concession Contract or clause 7 of the Cotai Land Concession Contracts, granted in favour of the issuers of such performance bonds, guaranties, commercial letters of credit or bankers' acceptances, so long as (i) any cash used as security for such reimbursement obligations is invested (if at all) in Permitted Investments only (to the extent the Company or Palo has the right to direct the investment thereof) and is

- 195 -

segregated from the Company's or Palo's general cash accounts so that such Liens attach only to such cash and Permitted Investments and (ii) the amount of cash and/or Permitted Investments secured by such Liens does not exceed 110% of the amount of the Financial Indebtedness secured thereby (ignoring any interest earned or paid on such cash and any dividends or distributions declared or paid in respect of such Permitted Investments);

- (n) without prejudice to Schedule 6 (*Accounts*), Liens arising by reason of any netting or set-off arrangements entered into by the Company or any other Obligor in the normal course of its banking arrangements and the standard account operating procedures of the bank for the purpose of netting debit and credit balances;
- (o) Liens arising under title transfer or retention of title arrangements entered into by the Company or any other Obligor in the normal course of its trading activities on the counterparty's standard or usual terms *provided* that such arrangements shall be limited to Property of an aggregate value not exceeding USD1,000,000 or its equivalent; and
- (p) other Liens approved in writing by the Intercreditor Agent.

### Limitation on Fundamental Changes

4.

5.

Enter (and ensure no other Obligor shall enter) into any merger, consolidation or amalgamation, or liquidate, wind up or dissolve itself (or suffer any liquidation or dissolution), or Dispose of all or substantially all of its Property or business, except:

- (a) that each of the Company and Palo may dispose of any of its Property in accordance with paragraph 5 of this Part B; and
- (b) for any voluntary liquidation, winding up or dissolution of, or similar action with respect to, Palo after a Permitted Cotai Reorganisation.

# Limitation on Disposition of Property

Dispose (and ensure no other Obligor shall dispose) of any of its Property (including, without limitation, receivables and leasehold interests), whether now owned or hereafter acquired, or issue or sell any Capital Stock to any Person, except:

- (a) the Disposition for fair market value on arm's length commercial terms in the ordinary course of business of any Property or obsolete or worn out Property or Property no longer used or useful in the business of the Company or the Obligor *provided* that such Disposition could not reasonably be expected to materially adversely affect either Project or any of the Project Security and either:
  - the Net Cash Proceeds from the disposal of all such Property (excluding shares in any Obligor) do not exceed in aggregate an amount equal to USD100,000,000 or its equivalent in any Fiscal Year when taken together with any other Net Cash Proceeds received by the Company or any other Obligor in respect of any Asset Sale in that Fiscal Year; or
  - (ii) where such Net Cash Proceeds referred to in sub-paragraph (i) above exceed in aggregate an amount equal to USD100,000,000 or its equivalent in any Fiscal Year (when taken together with any other Net Cash Proceeds received by the Company or any other Obligor in respect of any Asset Sale in that Fiscal Year), such Net Cash Proceeds are reinvested within 12 months of receipt in assets used by the Company or any other Obligor in the conduct of its Permitted Business (and pending such reinvestment are deposited and retained in an Account) and, if such Net Cash Proceeds are not so reinvested within 12 months of receipt, are applied in mandatory prepayment of the Advances pursuant to paragraph 2 of Schedule 9 (*Mandatory Prepayment*);

- 196 -

- (b) the Disposition of cash (in each case in transactions otherwise permitted under this Agreement), Investments permitted pursuant to paragraph 8 of this Part B, inventory (in the ordinary course of business), receivables (in connection with the collection thereof and otherwise as customary in business activities of the type conducted by the Company) and cash or non-cash prizes and other complimentary items for customers customary in business activities of the type conducted by the Company;
- (c) the sale or issuance of the Company's or any other Obligor's Capital Stock (other than Disqualified Stock) to its direct Shareholders *provided* that such Capital Stock is fully paid upon such issuance (or, as the case may be, sale) and is subject to the Liens created under the Company Share Pledge or such other Liens created under the Senior Finance Documents with regard to such other Obligor's Capital Stock;
- (d) the entry into of any leases or licences with respect to any space on or within the Site Facilities or the Cotai Site Facilities where the entry into of such leases or licences is not otherwise prohibited by the terms of the Senior Finance Documents;
- (e) any Property purchased using the Company's Subconcession proceeds (which is neither comprised in any Project or the Cotai Project or nor necessary or desirable to ensure the full benefit of any Project or the Cotai Project to the Company);
- (f) any Property associated with an Excluded Subsidiary or Excluded Project (which is neither comprised in any Project or the Cotai Project (other than, in the case of an Excluded Project on the Cotai Site, the grant of any right or interest to use the Cotai Site pursuant to a Cotai Resort Management Agreement) nor necessary or desirable to ensure the full benefit of any Project or the Cotai Project to the Company);
- (g) any Property associated with Resort Management Agreements (which is neither comprised in any Project or the Cotai Project (other than, in the case of an Excluded Project on the Cotai Site, the grant of any right or interest to use the Cotai Site pursuant to a Cotai Resort Management Agreement) nor necessary or desirable to ensure the full benefit of any Project or the Cotai Project to the Company);

- 197 -

- (h) the incurrence of Liens permitted under paragraph 3 of this Part B;
- (i) any Event of Eminent Domain *provided* that the requirements of Schedule 9 (*Mandatory Prepayment*) are complied with in connection therewith;
- (j) subject to compliance with paragraph 17 of this Part B, the Company may enter into Subconcessions;
- (k) any Disposition by the Company to a Subsidiary Obligor or by a Subsidiary Obligor to the Company *provided* that the requirements of paragraph 15 of Part A of this Schedule 5 are satisfied with respect to the acquisition by the transferee of the Disposed Property;
- (l) the Disposition by Palo to the Company of Palo's interest in the Cotai Site (whether by assignment, novation, transfer or replacement of the Cotai Land Concession Contract or otherwise) and/or Palo's ownership of, or interest in, other assets relating to the Cotai Site and/or the Cotai Project pursuant to a Permitted Cotai Reorganisation *provided* that the conditions in the proviso in the definition of Permitted Cotai Reorganisation are satisfied with respect to the same; or
- (m) any Disposition not falling within any of the preceding sub-paragraphs of this paragraph 5 made with the prior written consent of the Intercreditor Agent (such consent not to be unreasonably withheld).

#### Limitation on Restricted Payments

6.

6.1

- (a) Declare or pay any dividend on, or make any payment on account of, or set apart assets for a sinking or other analogous fund (and ensure each other Obligor which is a Subsidiary of the Company shall not declare or pay any dividend on, or make any payment on account of, or set apart assets for a sinking or other analogous fund) for, the purchase, redemption, defeasance, retirement or other acquisition of, any Capital Stock of the Company (including any Equity) or such other Obligor, whether now or hereafter outstanding, or make any other distribution in respect thereof, either directly or indirectly, whether in cash or property or in obligations of the Company or such other Obligor;
  - (b) enter (and ensure each other Obligor which is a Subsidiary of the Company shall not enter) into any derivatives or other transaction with any financial institution, commodities or stock exchange or clearinghouse (a "Derivatives Counterparty") obligating the Company or such other Obligor to make payments to such Derivatives Counterparty as a result of any change in market value of any such Capital Stock;
  - (c) make any repayment of, or pay any interest on or other amount in respect of (and ensure each other Obligor shall not make any repayment of, or pay any interest on or other amount in respect of) the Performance Bond Facility or any other Subordinated Debt (including any Equity) or any WML Debt;

- 198 -

- (d) make any payment of IP Fees;
- (e) make any payment of Corporate Administrative Fees; or
- (f) take any action or make any payment permitted pursuant to, or in connection with any transaction permitted pursuant to, paragraph 10 of this Part B,

(collectively, "**Restricted Payments**") except (i) to the extent constituting payment of any Restricted Payment by any Subsidiary Obligor to the Company or (ii) the Company and each Obligor may:

- (i) pay dividends, distributions or other payments if (A) permitted under paragraphs 6.2 or 17(d) of this Part B or (B) made using any amounts derived from or under (i) any Resort Management Agreement or (ii) any Excluded Project or Excluded Subsidiary;
- (ii) make payments permitted in accordance with the Deed of Appointment and Priority;
- (iii) pay Approved IP Fees and Approved Corporate Administrative Fees *provided* that, in each case, no Event of Default has occurred and is continuing or would result from such payment;
- (iv) pay to any of its Affiliates the direct cost incurred by that Affiliate (excluding any administration charges, handling fees, mark-ups or any other fees, costs, charges or impositions of a similar nature levied, imposed or charged by that Affiliate) for the acquisition of any assets, or the procurement or any services, to be used by the Company or such Obligor in its Permitted Business; and
- make dividends, distributions or other payments (including loans or advances), directly or indirectly, for the purpose of paying any amount with respect to Guaranteed WML Debt *provided* that (i) such payment would not cause the aggregate amount of such payments made pursuant to this paragraph (v) to exceed USD75,000,000 in any Fiscal Year and (ii) had the Guaranteed WML Debt (and any Guarantee Obligations of the Company or Obligors in connection therewith) been incurred immediately prior to the last Quarterly Date to occur before the incurrence of such Guaranteed WML Debt, the Company would, as at that Quarterly Date, still have been in compliance with paragraph 1 of Part B of this Schedule 5.

6.2

- At any time after the first Quarterly Date, the Company and each Subsidiary Obligor may declare and/or pay at the times set out in paragraph 6.3 of this Part B by way of dividend or other distribution or make other Restricted Payments, subject to compliance with applicable Legal Requirements and *provided* that all of the following conditions have been satisfied:
  - (i) no Event of Default has occurred and is continuing or might reasonably be expected to occur as a result of the declaration and/or payment of such dividend or other distribution or Restricted Payment; and
  - (ii) had such declaration, payment, distribution or other Restricted Payment been made immediately prior to the most recent Quarterly Date, the Company would, as at that Quarterly Date, still have been in compliance with paragraph 1 of Part B of this Schedule 5.

- 6.3 Any payment or making of dividends or other distribution or Restricted Payment under paragraph 6.2 of this Part B may be made once in each Fiscal Quarter (or such other date or with such other frequency as may be agreed by the Intercreditor Agent) *provided that* where such payment or making of dividends or other distributions or Restricted Payments is proposed to be made in the Fiscal Quarter immediately following the end of the previous Fiscal Year, that payment or making of dividends or other distributions or Restricted Payments or Restricted Payments shall be made after the date of prepayment of Advances under the Facilities in respect of Excess Cash Flow pursuant to paragraph 4 of Schedule 9 (*Mandatory Prepayment*).
- 6.4 The financial ratios set out in paragraph 6.2 of this Part B shall be tested by reference to the financial statements of the Company delivered pursuant to paragraph 1(a) of Part A of this Schedule 5 and/or each Compliance Certificate delivered pursuant to paragraph 2(a) of Part A of this Schedule 5.

# 7. [Not used]

8.

# Limitation on Investments

Make (and ensure that no other Obligor shall make) any advance, loan, extension of credit (by way of guarantee or otherwise) or capital contribution to, or purchase any Capital Stock, bonds, notes, debentures or other debt securities of, or any assets constituting an ongoing business from, or make any other investment in, any other Person (all of the foregoing, "**Investments**"), except:

- (a) Investments by the Company in any of its Wholly Owned Subsidiaries (other than any Excluded Subsidiary) which is an Obligor;
- (b) extensions of trade credit in the ordinary course of business (including, without limitation, advances to patrons of the Project or the Cotai Project casino and gaming operations consistent with ordinary course gaming operations) *provided* that such extensions are in compliance with all Legal Requirements;
- (c) Permitted Investments;
- (d) Investments made using any amounts derived from or under (i) the grant of any Subconcession, (ii) any Resort Management Agreement or (iii) any Excluded Project or Excluded Subsidiary;
- (e) Investments in any Excluded Subsidiary or Excluded Project or any third party whose primary business is the development, construction, ownership and operation of hotel resorts and casinos, provided that such Investments are made using amounts which are not required to be applied for any other purpose under or in connection with the Senior Finance Documents;

- 200 -

- (f) loans and advances to employees of the Company or Palo in the ordinary course of business (including, without limitation, for travel, entertainment and relocation expenses) in an aggregate amount not to exceed USD5,000,000 or its equivalent at any one time outstanding;
- (g) Investments consisting of securities received in settlement of debt created in the ordinary course of business and owing to the Company or in satisfaction of judgments;
- (h) acquisition of Advances in connection with a Permitted Loan Repurchase;
- (i) loans and advances permitted under paragraph 6.1(v) of this Part B; and
- (j) in addition to Investments otherwise expressly permitted by this paragraph 8, so long as no Default shall have occurred and be continuing or would result therefrom and no Material Adverse Effect shall have occurred and be continuing or would result therefrom, Investments by the Company or any other Obligor in an aggregate amount (valued at cost) not to exceed USD50,000,000 or its equivalent at any one time outstanding.

## 9. Limitation on Optional Payments and Modifications of Governing Documents

- (a) Make or offer to make (and ensure no other Obligor makes or offers to make) any optional or voluntary payment, prepayment, repurchase or redemption of, or otherwise voluntarily or optionally defease, any Financial Indebtedness, or segregate funds for any such payment, prepayment, repurchase, redemption or defeasance, or enter into any derivative or other transaction with any Derivatives Counterparty obligating the Company to make payments to such Derivatives Counterparty as a result of any change in market value of such Financial Indebtedness, other than (i) the prepayment of (A) Financial Indebtedness incurred under the Senior Finance Documents in accordance with the terms of the Senior Finance Documents or (B) Permitted Financial Indebtedness (excluding any Subordinated Debt) or (ii) Permitted Loan Repurchases; or
- (b) amend or modify, or permit the amendment or modification of (and ensure no other Subsidiary Obligor amends, modifies or permits the amendment or modification of) its Governing Documents in any manner adverse to any of the Secured Creditors (other than where such amendment or modification could not reasonably be expected to have a Material Adverse Effect).

#### 10. Limitation on Transactions with Affiliates

Enter (and ensure that no other Obligor shall enter) into any single transaction, or series of transactions, whether related or not (including, without limitation, any purchase, sale, lease or exchange of Property, the rendering of any service or the payment of any management, advisory or similar fees, the entry into of any contract or the amendment, novation, supplementation, extension or restatement of any existing contract) with any Affiliate of the Company unless:

(i) all amounts payable by any Obligor thereunder or in respect thereof have been subordinated in accordance with the Subordination Deed and that Obligor and the relevant Affiliate counterparty to such transaction have acceded to the Subordination Deed (in each case, on such terms and by such instrument or combination of instruments, as the Intercreditor Agent deems necessary or desirable and such Obligor and the relevant Affiliate counterparty have delivered to the Intercreditor Agent a legal opinion, in form and substance satisfactory to the Intercreditor Agent (acting reasonably), issued by legal counsel acceptable to the Intercreditor Agent (acting reasonably) or (in respect of a Macau SAR or Nevada incorporated Obligor or Affiliate counterparty) issued by (if that General Counsel or other in-house counsel is admitted, in accordance with all applicable Legal Requirements, to practise law in Macau SAR or Nevada, as the case may be) the General Counsel or other in-house counsel of Wynn Resorts (for such Nevada Obligor or Affiliate counterparty) or the General Counsel or other in-house counsel of the Company (for such Macau SAR Obligor or Affiliate counterparty), in each case in respect of the enforceability of such Person's obligations under the Subordination Deed;

- 201 -

the relevant transaction is a genuine commercial transaction on terms that are not less favourable to the Company or (b) (i) applicable Obligor than those that might be obtained at the time in a comparable arm's length transaction with Persons who are not Affiliates of the Company; and

- (ii) in respect of:
  - any transaction which has (or as a result of any amendment, novation, supplementation, extension or (1)restatement of any existing contract would have) a total amount or contract price payable (or expected aggregate amount to be paid in the case of "cost plus" contracts) or which otherwise involves liabilities, actual or contingent, in an aggregate amount in excess of USD50,000,000 or its equivalent in any Fiscal Year; or
  - (2) any series of transactions (whether related or not) which when taken together have (or as a result of any amendment, novation, supplementation, extension or restatement of any existing contract would have) a total amount or contract price payable (or expected aggregate amount to be paid in the case of "cost plus" contracts) or which otherwise involves liabilities, actual or contingent, in an aggregate amount in excess of USD50,000,000 or its equivalent in any Fiscal Year,

deliver to the Intercreditor Agent:

- a certificate of a Responsible Officer of the Company certifying that it has sufficient Funds to operate (a) its business as it is then conducted and pay its debts when due in respect of the relevant transaction or transactions to be entered into or otherwise; and
- (b) such other substantiating information and evidence in respect of the relevant transaction or transactions that the Intercreditor Agent (acting on the instructions of any Lender) may reasonably require; or
- the relevant transaction is between the Company and its Wholly Owned Subsidiary (other than any Excluded Subsidiary) which is an Obligor.

- 202 -

(c)

### 11. Limitation on Sales and Leasebacks

Save as permitted pursuant to paragraph 5 of this Part B, enter (and ensure that no other Obligor shall enter) into any arrangement with any Person providing for the leasing by the Company or such other Obligor as lessee of Property which has been or is to be sold or transferred by the Company or such other Obligor to such Person or to any other Person to whom funds have been or are to be advanced by such Person on the security of such Property or rental obligations of the Company or such other Obligor.

#### 12. Limitation on Changes in Fiscal Periods

Permit the Fiscal Year of the Company or any other Obligor to end on a day other than December 31 or change the Company's or any other Obligor's method of determining Fiscal Quarters.

### 13. Limitation on Negative Pledge Clauses

Enter into or suffer to exist or become effective any agreement that prohibits or limits the ability (and ensure that no other Obligor enters into or suffers to exist or become effective any agreement that prohibits or limits the ability) of the Company or any other Obligor to create, incur, assume or suffer to exist any Lien upon any of its Property or revenues, whether now owned or hereafter acquired, to secure the Obligations other than:

- (a) the Senior Finance Documents;
- (b) the Concession Contract;
- (c) any agreements governing any purchase money Liens or Capital Lease Obligations otherwise permitted hereby (in which case, any prohibition or limitation shall only be effective against the assets financed thereby and proceeds thereof);
- (d) in connection with WML Debt (provided that the terms of such WML Debt shall not prohibit the creation, incurrence, assumption or suffering to exist of any Lien in favour of the Secured Parties existing or contemplated under the Senior Finance Documents as such documents are in effect at the time such WML Debt is incurred); and
- (e) as required by applicable Legal Requirements.

- 203 -

#### 14. Limitation on Lines of Business

- 14.1 Enter (and ensure that no other Obligor enters) into any business activities, whether directly or indirectly, other than Permitted Businesses.
- 14.2 Establish (and ensure that no other Obligor establishes) any representative office other than (in the case of the Company) marketing offices or other place of business in a jurisdiction outside the Macau SAR or its place of incorporation unless (in the case of the Company) such office or place of business has been notified to the Intercreditor Agent, the requirements of paragraph 15 of Part A of this Schedule 5 have been satisfied in respect of any Property of the Company which may at any time be located in such jurisdiction within 30 days of such establishment and the Company has taken all such steps as may be required in such jurisdiction to perfect, maintain and protect the Security.

#### 15. **Restrictions on Changes**

#### 15.1 Concession Contract, Land Concession Contract and Cotai Land Concession Contract

Notwithstanding any other provision of this paragraph 15, agree to any amendment to or termination of, or assign, transfer, cancel or waive any of its rights under the Concession Contract, the Land Concession Contract or the Cotai Land Concession Contract without obtaining the prior written consent of the Intercreditor Agent (other than: (i) in relation to the Cotai Land Concession Contract, to increase the gross construction area and/or the contract premium payable thereunder in connection with an Excluded Project located or to be located on the Cotai Site *provided* that the entire Cotai Site shall remain part of the Project Security and subject to the Cotai Mortgage and there shall be no adverse effect on the validity or enforceability of, or the effectiveness or ranking of any Security of the Finance Parties as a result of the Excluded Project being located on the Cotai Site and *further provided* that such increase is in accordance with the applicable Macau SAR Legal Requirements and could not reasonably be expected to be prejudicial to the interests of any of the Finance Parties, (ii) any mechanical or administrative amendments or any amendments required by any Macau SAR Governmental Authority of which reasonable prior notice has been given and which could not reasonably be expected to be prejudicial to the interests of any of the Finance Parties, and (iii) in relation to the Cotai Land Concession Contract, in connection with a Permitted Cotai Reorganisation).

# 15.2 **Permits and Other Contracts**

- 15.2.1 Directly or indirectly enter into, amend, modify, terminate, supplement or waive a right or permit or consent to the amendment, modification, termination (except expiration in accordance with its terms), supplement or waiver of any of the provisions of, or give any consent or exercise any other discretion under (and ensure that no other Obligor directly or indirectly enters into, amends, modifies, terminates, supplements or waives a right or permits or consents to the amendment, modification, termination (except expiration in accordance with its terms), supplement or waiver of any of the provisions of, or gives any consent or exercises any other discretion under in accordance with its terms), supplement or waiver of any of the provisions of, or gives any consent or exercises any other discretion under):
  - (a) any Permit, the effect of which could reasonably be expected to have a Material Adverse Effect;

- 204 -

- (b) the Prime Contractor's Completion Guarantee or the Prime Contractor's Performance Bond; or
- any Project Document, Major Project Document or other contract (other than the Concession Contract, the Land Concession (c) Contract, the Cotai Land Concession Contract or any contract entered into with respect to the Guaranteed WML Debt or incidental thereto (including any underwriting agreements, dealer manager agreements, solicitation agency agreements and other contracts entered into in connection with any issuance of, amendment to, tender offer for, other repurchase of or negotiation or refinancing of the Guaranteed WML Debt)) unless in the case of the Company or any of the Company's Wholly Owned Subsidiaries which are members of the Restricted Group (1) it could not reasonably be expected to have a Material Adverse Effect and (2) in respect of any: (A) contract with respect to any Project or Permitted Cotai Reorganisation (but not with respect to the Cotai Project or any Excluded Project) to be entered into with, or any existing contract which as a result of amendment, novation, supplementation, extension or restatement has, a total contract price payable (or expected aggregate amount to be paid in the case of "cost plus" contracts) or which has or may otherwise involve liabilities, actual or contingent, in each case in an amount in excess of USD50,000,000 or its equivalent in any Fiscal Year or (B) series of contracts (whether related or not) to be entered into, in such case, with respect to any Project or Permitted Cotai Reorganisation (but not with respect to the Cotai Project or any Excluded Project), or any series of contracts (whether related or not) which has been entered into which as a result of amendment, novation, supplementation, extension or restatement would have, when taken together a total contract price payable (or expected aggregate amount to be paid in the case of "cost plus" contracts) or which has or may otherwise involve liabilities, actual or contingent, in each case in an amount in excess of USD50,000,000 or its equivalent in any Fiscal Year, the Company delivers to the Intercreditor Agent: (X) a certificate of a Responsible Officer of the Company certifying that it has sufficient Funds to operate its business as it is then conducted and pay its debts when due in respect of the relevant contract to be entered into or otherwise and (Y) such other substantiating information and evidence in respect of the relevant contract that the Intercreditor Agent (acting on the instructions of any Lender) may reasonably require,

without in each case giving written notice to the Intercreditor Agent. Notwithstanding any of the foregoing, the Company may only take or, as the case may be, permit or consent to the taking of, any such action under or in respect of, or otherwise agree to any variation to the Project Works or the Plans and Specifications or any other variation to the Construction Contract without the prior written consent of the Intercreditor Agent where:

- (1) the actions or variations do not involve any extension of the Guaranteed Dates of Substantial Completion for the Original Project (beyond that effective prior to the variations); and
- (2) no Material Adverse Effect could reasonably be expected to result therefrom,

- 205 -

and the Intercreditor Agent has received a certificate from the Company signed by a Responsible Officer certifying that the conditions set out in subparagraphs (1) and (2) above have been satisfied. Notwithstanding anything in this Clause 15.2 to the contrary, the Company and the Obligors shall be permitted to enter into, amend, modify, terminate, supplement or waive any contract with respect to the Guaranteed WML Debt or incidental thereto (including any underwriting agreements, dealer manager agreements, solicitation agency agreements and other contracts entered into in connection with any issuance of, amendment to, tender offer for, other repurchase of or negotiation or refinancing of the Guaranteed WML Debt) and incur obligations and liabilities in connection therewith, provided that (i) no Default has occurred and is continuing or could reasonably be expected to occur as a result of such entry into, amendment, modification, termination, supplement, or waiver and (ii) such entry into, amendment, modification, termination, supplement or waiver (1) would not result in there being any scheduled amortisation of the WML Debt prior to the Final Repayment Date of any Facility (as determined: (A) in respect of the Term Facility, as of the Fourth Amendment Effective Date; and (B) in respect of any Additional Lender Facility which provides for a Term Loan Facility, as of the date of that Additional Lender Facility Agreement), or cause the stated maturity of the WML Debt to precede the later of (x) the Revolving Credit Facility Termination Date (as determined as of the Fourth Amendment Effective Date), (y) the Final Repayment Date of the Term Facility (as determined as of the Fourth Amendment Effective Date); and (z) the Final Repayment Date of any Additional Lender Facility which provides for a Term Loan Facility (as determined as of the date of the Additional Lender Facility Agreement) or (2) could otherwise reasonably be expected to have an adverse effect on the ability of the Company or any other Obligor to perform its payment obligations under the Finance Documents, and (iii) any event of default (howsoever described) in respect of the WML Debt which references or concerns in any way the First Ranking Liabilities (as defined in the Deed of Appointment and Priority), or any part thereof, will only occur on (in respect of non-payment of any of the First Ranking Liabilities) such non-payment event of default occurring as contemplated by Schedule 10 (Events of Default) of this Agreement and (in respect of any other events or circumstances described in Schedule 10 (Events of Default) of this Agreement) will only occur on the Intercreditor Agent issuing written notice to the Company pursuant to (or taking steps contemplated by) Clause 19.2 (Remedies following an Event of Default) of this Agreement.

15.2.2 Accept, agree or determine the achievement of or waive any requirement in respect of (or be deemed to have done any of the foregoing) or issue, accept, or be deemed to have confirmed any certificate or notice of "Original Project Substantial Completion" in respect of all or any part of the Original Project under article 12 of the Construction Contract without the written approval of the Intercreditor Agent (such approval not to be unreasonably withheld or, without prejudice to any other provision of this Agreement, delayed).

- 206 -

- 15.2.3 Reduce the level of Retainage Amounts withheld pursuant to section 5.6 of the Construction Contract or paragraph 21 of Part A of this Schedule 5 except in accordance with the Construction Contract.
- 15.2.4 Fail to withhold a sum equal to 100% of the costs reasonably estimated by the Company (and confirmed by the Technical Adviser) as necessary to complete "Original Project Punch List Items" or, as the case may be, "Expansion Project Punch List Items" (as defined in the Construction Contract) as Retainage Amounts pursuant to section 5.7 of the Construction Contract unless such retention is not permitted under applicable laws.
- 15.2.5 Accept any non-conforming "Work" (as defined in the Construction Contract) of a material nature unless the Company shall have complied with the requirements of paragraph 15.2.1 of this Part B.
- 15.2.6 Accept or agree to any increase in the "Original Project Contractor's Fee" or the "Expansion Project Contractor's Fee" (as defined in the Construction Contract) for any reason, except to the extent required pursuant to section 18.5 of the Construction Contract following compliance with paragraph 15.2.1 of this Part B or such increase is paid from the proceeds of additional Equity (other than any Equity required to be paid or advanced in accordance with the terms hereof).

#### 16. Limitation on Formation and Acquisition of Subsidiaries

Without the prior written consent of the Intercreditor Agent (such consent not to be unreasonably withheld), form, create or acquire (and ensure that no other Obligor forms, creates or acquires) any direct or indirect Subsidiary other than any Obligor or (in the case of the Company and the Company's Subsidiaries) any Excluded Subsidiary.

# 17. Limitation on Subconcessions

Grant or enter into any Subconcession unless:

- (a) no breach of the Concession Contract shall exist and be continuing at the time such Subconcession is granted or entered into or would occur after or as a result of granting or entering into such Subconcession;
- (b) no Event of Default shall exist and be continuing at the time such Subconcession is granted or entered into or would occur after or as a result of granting or entering into such Subconcession (or immediately after any renewal or extension thereof at the option of the Company);
- such Subconcession and any business or other activities carried out pursuant thereto shall be self-contained and shall not adversely affect the operation of the Projects (excluding the ability of the Company to grant further Subconcessions), no breach or termination thereof shall result in a breach or an entitlement to terminate the Concession Contract, the Company shall have no responsibility nor any liability, actual or contingent, for the performance by the subconcessionaire of its obligations under or in respect of the Subconcession and, subject to receipt by the Intercreditor Agent of evidence

- 207 -

reasonably satisfactory to it, neither the Subconcession nor any business or other activities carried out pursuant thereto could otherwise reasonably be expected to materially interfere with, impair or detract from the operation of the Projects or otherwise have a Material Adverse Effect; and

(d) if any cash or cash proceeds are paid or received in respect of the grant or entry into such Subconcession, it shall first be deposited into an Account and, after deduction and payment therefrom of all Taxes, costs and expenses incurred in connection with such payment, receipt, grant or entry, may thereafter be used, applied, dividended or otherwise distributed by the Company and applied in its discretion (including, subject to compliance with applicable Legal Requirements, to make Restricted Payments) *provided* that no Default has occurred and is continuing. Notwithstanding any provisions contained in the Senior Finance Documents to the contrary, upon receipt of any funds distributed by the Company in accordance with this paragraph 17 by any Wynn Obligor, such Wynn Obligor shall also be entitled to use, apply, dividend or otherwise distribute such funds in its discretion (subject to compliance with applicable Legal Requirements).

#### 18. Limitation on Sale or Discount of Receivables

Except as permitted pursuant to paragraph 5(b) of this Part B, directly or indirectly, sell with recourse, or discount or otherwise sell for less than the face value thereof (and ensure that no other Obligor, directly or indirectly, sells with recourse, or discounts or otherwise sells for less than the face value thereof) any of its notes or accounts receivable other than an assignment for purposes of collection in the ordinary course of business.

### 19. Compliance

Use or permit the use of the Projects or the Cotai Project in any manner that could result in such use becoming a non-conforming use under any applicable land use law, rule or regulation that is a Legal Requirement.

#### 20. [Not used]

#### 21. Amendment to Transaction Documents

Enter (and ensure that no other Obligor enters) into any agreement (other than the Senior Finance Documents) restricting its ability to amend any of the Transaction Documents where such entry could reasonably be expected to have an adverse effect on the ability of the Company or any other Obligor to perform its payment obligations under the Finance Documents

## 22. No Other Powers of Attorney

Execute or deliver (and ensure that no other Obligor executes or delivers) any agreement creating any powers of attorney (other than powers of attorney for signatories of documents permitted or contemplated by the Transaction Documents), or similar documents, instruments or agreements, except to the extent such documents, instruments or agreements comprise part of the Security Documents or are required to be executed or delivered in the Company's or such other Obligor's ordinary course of business.

- 208 -

# 23. [Not used]

# 24. [Not used]

#### 25. Casino and Gaming Zone Areas

Solely with respect to the Projects and the Expansion (and not the Cotai Project or any Excluded Project), designate any area (other than the horizontal property identified as comprising the casino in the Plans and Specifications as at the Signing Date) as a casino or gaming zone unless such designation would not cause the aggregate area which is classified as casino or gaming zones by the Macau SAR to exceed 40,000m2 in respect of the Original Project and the Expansion and the Intercreditor Agent has received evidence that, in the event of the reversion of such area to the Macau SAR upon termination of the Concession Contract, such reversion would not materially affect the ongoing operation of the Hotel Project.

# 26. Junkets

Enter into or permit to subsist any arrangement with any gaming junket-tour promoters, directors or cooperators unless such Persons and any such arrangement are in compliance with the requirements of the Concession Contract and all other applicable Legal Requirements and the Company shall monitor the activities of such Persons in regard to such arrangements and shall take all necessary or appropriate reasonable measures to ensure such compliance.

# 27. Sanctions laws and regulations

Knowingly use (and shall ensure that no Obligor, member of the Group or any Excluded Subsidiary shall knowingly use) the proceeds of any Facility, or lend, contribute or otherwise make available such proceeds to any subsidiary, joint venture partner or other person or entity, (i) to fund any activities or business of or with any Restricted Party or (ii) for the purpose of financing the activities of any person currently subject to, or located in a country or territory that is at the time of such funding the subject of, any Anti-Terrorism Law.

- 209 -

#### SCHEDULE 6 ACCOUNTS

| 1. | Accounts |
|----|----------|
|    |          |

1.1

Accounts

The members of the Restricted Group may (but shall not be required to) maintain in accordance with the requirements of this Schedule 6 and the Senior Finance Documents, operating, disbursement and other accounts (which as of the Second Amendment Signing Date comprise of the following bank accounts maintained by the Company):

- (i) an account denominated in US dollars opened in Macau and designated "Term Facility USD Disbursement Account";
- (ii) an account denominated in HK dollars opened in Macau and designated "Term Facility HKD Disbursement Account";
- (iii) an account denominated in US dollars opened in Macau and designated "Project Facility USD Disbursement Account";
- (iv) an account denominated in HK dollars opened in Macau and designated "Project Facility HKD Disbursement Account";
- (v) an account denominated in US dollars opened in Nevada and designated "USD Operating Account";
- (vi) an account denominated in US dollars opened in Macau and designated "USD Operating Account";
- (vii) an account denominated in HK dollars opened in Macau and designated "HKD Operating Account";
- (viii) an account denominated in Patacas opened in Macau and designated "MOP Operating Account";
- (ix) an account denominated in US dollars opened in Macau and designated "USD Debt Service Account";
- (x) an account denominated in HK dollars opened in Macau and designated "HKD Debt Service Account";
- (xi) an account denominated in US dollars opened in Macau and designated "USD Debt Service Reserve Account";
- (xii) an account denominated in HK dollars opened in Macau and designated "HKD Debt Service Reserve Account";
- (xiii) an account denominated in Patacas opened in Macau and designated "Special Gaming Tax Account";
  - 210 -

- (xiv) an account denominated in Patacas opened in Macau (being the account referred to in paragraph B6 of the Gaming Concession Consent Agreement) and an account denominated in US dollars opened in Macau, together designated "Compensation Proceeds Account";
- (xv) an account opened in Macau and designated "Upfront Premium Account"; and
- (xvi) an account denominated in HK dollars and an account denominated in Patacas each opened in Macau and designated "Construction Disbursement Account"),

in each case, subject to Security under the relevant Senior Finance Documents and each, for the purposes of the Senior Finance Documents, also an Account.

- (b) Notwithstanding the provisions of paragraph (a) above, each Obligor and each member of the Restricted Group shall comply with the provisions of Clauses 1.1(b) and (c), and Clauses 1.2 to 3.7 (inclusive) of this Schedule 6 (*Accounts*) and the provisions of the Term Facility Agreement in respect of the operation of each Development Account (as defined in the Term Facility Agreement), including (without limitation) any withdrawal therefrom.
- (c) Any member of the Restricted Group may open one or more additional operating, disbursement or other accounts *provided* that such accounts satisfy the requirements of the definition of "Account" in Clause 1.1.
- 1.2 Maintenance of Accounts

The Accounts shall, save as otherwise provided by the Charges over Accounts, the Account Bank Notices and Acknowledgements or herein, be maintained by the relevant member of the Restricted Group with the relevant Account Bank in accordance with the Account Bank's usual practice and may from time to time be sub-divided into such sub-accounts as that member of the Restricted Group may reasonably request.

1.3 Restrictions

Each member of the Restricted Group shall maintain each of its Accounts (and shall procure that each other Account is maintained) as a separate account with the relevant Account Bank and:

- (a) none of the restrictions contained in this Schedule on the withdrawal of funds from Accounts shall affect the obligations of any Obligor to make any payments of any nature required to be made to the Senior Secured Creditors on the due date for payment thereof in accordance with any of the Senior Finance Documents; and
- (b) no withdrawal shall be made from any Account if it would cause such account to become overdrawn.

- 211 -

### 1.4 Credits to Accounts

Save as otherwise provided in any of the Security Documents after enforcement thereof, each member of the Restricted Group shall credit, and shall procure that there is credited, to the Accounts all such amounts as are provided for in this Agreement and ensure that such other credits are made thereto as are required to be made pursuant to any other provision of any other Senior Finance Document.

# 1.5 Interest

Each amount from time to time standing to the credit of each Account (for the avoidance of doubt excluding amounts for the time being applied in acquiring Permitted Investments) shall bear interest at such rate as may from time to time be agreed between the relevant member of the Restricted Group and the relevant Account Bank, and the relevant member of the Restricted Group shall ensure that such interest is credited to such account at such time or times as may be agreed from time to time between that member of the Restricted Group and the Account Bank or, failing agreement, in arrears on 31 December.

## 1.6 Payments

Save as otherwise provided in this Agreement or pursuant to the Charges over Accounts, the Account Bank Notices and Acknowledgements or any other relevant Security Document, no party shall be entitled to require any Account Bank to make any payment out of the amount standing to the credit of any Account maintained with it.

## 1.7 Other Accounts

No member of the Restricted Group will open or maintain any accounts other than:

- (i) the Accounts; and
- (ii) any account opened by the Company for the deposit of any amounts derived from or under (i) subject to paragraph 17 of Part B of Schedule 5, the grant of any Subconcession, (ii) any Resort Management Agreement or (iii) any Excluded Project or Excluded Subsidiary,

except with the prior approval of the Intercreditor Agent.

# 2. **Permitted Investments**

# 2.1 Power of Investment

The Company may require, subject as provided in this Agreement, that such part of the amounts outstanding to the credit of any Account as it considers prudent shall be invested from time to time in Permitted Investments in accordance with this paragraph 2.

- 212 -

# 2.2 Procedure for Investment

- 2.2.1 Unless held for the account of the Company or other member of the Restricted Group and secured by first ranking fixed charge in favour of the Security Agent pursuant to a Charge over Accounts, the Company shall ensure that all Permitted Investments are made in the name of the Company and secured by a first ranking fixed lien in favour of the Security Agent in such form and on such terms as the Intercreditor Agent may reasonably require.
  - 2.2.2 The Company will at all times seek to match the maturities of the Permitted Investments made out of moneys standing to the credit of an Account having regard to the availability of Permitted Investments which are readily marketable, and shall liquidate (or procure that there are liquidated) Permitted Investments to the extent necessary for the purposes of payment of any amount due under the Senior Finance Documents.
- 2.2.3 The Company shall ensure that all documents of title or other documentary evidence of ownership with respect to Permitted Investments made out of any Account are held in the possession of or for the benefit of the Security Agent and, if any such document or other evidence comes into the possession or control of the Company or any other Obligor, it shall procure that the same is delivered immediately to the Security Agent.

## 2.3 Realisation

- 2.3.1 The Company shall ensure that, whenever any Investment Proceeds or Investment Income is received in respect of a Permitted Investment made from amounts standing to the credit of an Account the Investment Proceeds and the Investment Income are:
  - (a) reinvested in further Permitted Investments; or
  - (b) paid into the relevant Account from which the Permitted Investment derives
- 2.3.2 The Company shall give (and shall ensure that each other Obligor gives) directions to the relevant Account Bank under paragraph 2.3.1 of this Schedule 6 and otherwise exercise its rights hereunder in such manner as will ensure compliance with the applicable provisions of the Senior Finance Documents with respect to Accounts, Permitted Investments, Investment Proceeds and Investment Income.

# 2.4 Non-qualifying criteria

If any Permitted Investment ceases to be a Permitted Investment, the Company will upon becoming aware thereof procure that the relevant investment is replaced by a Permitted Investment or by cash.

- 2.5 Accounts include Permitted Investments
  - 2.5.1 Subject to sub-paragraph 2.5.2 of this Schedule 6, any reference herein to the balance standing to the credit of one of the Accounts will be deemed to include a reference to the Permitted Investments in which all or part of such balance is for the time being invested. In the event of any dispute as to the value of the credit of an Account pursuant to this paragraph 2.5, that value

- 213 -

shall be determined in good faith by the Intercreditor Agent. If the Company so requests, the Intercreditor Agent will give the Company details of the basis and method of that determination.

2.5.2 If the amount standing to the credit of any Account (excluding for this purpose any amount deemed to be included pursuant to subparagraph 2.5.1 of this Schedule 6) is insufficient to make a payment under the Senior Finance Documents when due out of such Account, the Security Agent is authorised, in its discretion and without any liability for loss or damage thereby incurred by the Company or any other Obligor, to require the relevant Account Bank or, as the case may be, the Company to sell or otherwise realise, or to enter into any exchange transaction with respect to, (or, as the case may be, to require the Company to ensure any other Obligor so sells or otherwise releases or enters into any exchange transaction) any Permitted Investment concerned with that Account to the extent that the same is, in the opinion of the Intercreditor Agent, necessary for the payment of any amount due under the Senior Finance Documents which could not otherwise be paid out of the cash balance standing to the credit of the relevant Account.

# 2.6 Information

Commencing with the quarter in which a Permitted Investment is first made on behalf of the Company or any other Obligor, the Company shall, together with any other statement to be provided under this Schedule, ensure the delivery to the Security Agent of a schedule of the investments made, realised or liquidated during the quarter in respect of each Account, in such detail as the Intercreditor Agent may reasonably require.

## 2.7 No Responsibility

No Senior Secured Creditor will be responsible for any loss, cost or expense suffered by any Obligor in respect of any of its actions or those of any Account Bank in relation to the acquisition, disposal, deposit or delivery of Permitted Investments pursuant to this Agreement save for any such loss, cost or expense directly caused by its gross negligence or wilful misconduct. The Account Banks shall be acting solely for and on behalf of the Company (or the relevant other Obligor) in acquiring, holding or disposing of any Permitted Investment.

# 3. General Account Provisions

# 3.1 Transfers/Withdrawals

Save as otherwise agreed in writing with the Intercreditor Agent, the Company shall ensure that where this Schedule expressly provides for the making of payments to, or withdrawals or transfers from any Account, no other payments to, or, as the case may be, other withdrawals or transfers from, such Account shall be made except as expressly permitted under this Schedule or under the Security Documents.

# 3.2 Application of Amounts

The Company shall ensure that all amounts withdrawn or transferred from any Account for application in or towards making a specific payment or meeting a specific liability shall be applied in or towards making that payment or meeting that liability, and for no other purpose.

- 214 -

3.3

Default

- 3.3.1 Notwithstanding any other provisions of this Schedule, at any time following an Enforcement Notice, the Intercreditor Agent may request the Security Agent to give notice to any Account Bank and the relevant member of the Restricted Group in whose name the Account has been opened instructing the Account Bank not to act on the instructions or requests of that member of the Restricted Group in relation to any sums at any such time standing to the credit of any of the Accounts and the Company and that member of the Restricted Group shall procure that the Account Bank shall, in accordance with the Charges over Accounts and the Account Bank Notices and Acknowledgements, not so act and the Company or, as the case may be, any other member of the Restricted Group shall not be entitled (and the Company shall ensure such other member of the Restricted Group is not entitled) to give or make any further such instructions or requests.
- 3.3.2 Notwithstanding the other provisions of this Agreement, at any time following an Enforcement Notice, the Intercreditor Agent may request the Security Agent to:
  - (i) give written notice to any Account Bank (with a copy to the relevant member of the Restricted Group in whose name the Account has been opened) that the Security Agent shall be the sole signatory in relation to the Accounts;
  - (ii) apply the credit balances in the Accounts in or towards repayment of the Facilities and such other liabilities of the Obligors as the Intercreditor Agent may elect; and
  - (iii) generally use amounts standing to the credit of the Accounts at its discretion in order to discharge the Obligors' obligations under the Transaction Documents,

and, pursuant to the Charges over Accounts and the Account Bank Notices and Acknowledgements, the Company and the relevant member of the Restricted Group in whose name the Account has been opened shall procure that the Account Bank so acts and makes such payments accordingly.

3.4 Review of Accounts

The Company and each other member of the Restricted Group irrevocably grants, solely for the purposes of its role as agent of the Senior Secured Creditors hereunder, (and the Company shall ensure each such other member of the Restricted Group shall so grant) the Security Agent or any of its appointed representatives access to review the books and records of the Accounts (and shall irrevocably authorise (and the Company shall ensure each other member of the Restricted Group authorises) each Account Bank to disclose the same to the Security Agent and its appointed representatives) and irrevocably waives (and the Company shall ensure each other member of the Restricted Group so waives) any right of confidentiality which may

- 215 -

exist in respect of such books and records solely to the extent necessary to allow disclosure of such books and records to any Senior Secured Creditor and its advisers *provided* that, to the extent any such right of confidentiality does exist and the information comprised in such books and records is not otherwise in the public domain or required to be notified by the Company or any other member of the Restricted Group or the Account Bank to any of the Senior Secured Creditors and subject to any requirement to disclose any such information pursuant to any Legal Requirement or any other regulatory or stock exchange requirement, any Senior Secured Creditor or adviser to whom such disclosure is made shall undertake to the Company to keep the information comprised therein confidential.

# 3.5 Statements

The Company and each other member of the Restricted Group shall arrange (and the Company shall ensure each such member of the Restricted Group so arranges) for each Account Bank to provide to the Security Agent, at the latter's request:

- (a) a list of all Accounts maintained with it;
- (b) upon the reasonable request of the Security Agent, in respect of each calendar month, a statement of the balance of and each payment into and from each of the Accounts and the global amount of interest earned on each such Account during the preceding three month period or, if less, since the opening of the relevant Account; and
  - such other information concerning the Accounts as the Security Agent may require.

# (c) su 3.6 Waiver of Rights

#### 3.6.1 Waiver of rights by the Company

Save as provided in this Agreement, the Company and each other member of the Restricted Group agrees not to (and the Company shall ensure each other member of Restricted Group does not) exercise any right which it (or such other member of the Restricted Group) may have under any applicable law to direct the transfer of any amount standing to the credit of an Account to the Company or any other member of the Restricted Group or its order or to direct the transfer of any Permitted Investment to the Company or any other member of the Restricted Group or to its order.

3.6.2 Waiver of rights by Account Banks

The Company and each other member of the Restricted Group shall procure (and the Company shall ensure each such other member of the Restricted Group procures) that each Account Bank acknowledges and agrees that each Account and Permitted Investment is the subject of a Lien in favour of the Senior Secured Creditors collectively and acknowledges and agrees that (save, in the case of the Performance Bond Provider, as permitted pursuant to (i) section 2.6 of the Performance Bond Facility Agreement prior to the issuance of an Enforcement Notice) it is not entitled to, and shall undertake not to, claim or exercise any lien, right of set-off, combination of accounts or other right, remedy or security with respect to:

- (a) moneys standing to the credit of such Account or in the course of being credited to it or any earnings; or
- (b) any Permitted Investment.

# - 216 -

# 3.7 Closing of Accounts

The Company and each other member of the Restricted Group may close any Account and instruct each Account Bank to transfer any credit balances on any Account maintained with it *provided that* all balances standing to the credit of any such Account are transferred before the closure of such Account to another account or combination of accounts *provided* that each such account satisfies the requirements of the definition of "Account" in Clause 1.1.

### SCHEDULE 7 INSURANCE

References in this Schedule 7 to Clauses and Appendices refer to the Clauses and Appendices of this Schedule 7, unless the context otherwise requires.

# 1. INSURANCES TO BE EFFECTED

# 1.1 Direct Insurances

The Company shall effect:

- 1.1.1 the Construction Period Insurances as set out in Appendix 1 (*Construction Period Insurances*) on or before the issuance of the full Notice to Proceed (as defined in the Construction Contract) and shall maintain such Direct Insurances (including such additional Direct Insurances as may be required pursuant to paragraphs 1.2 and 4 of Appendix 1 (*Construction Period Insurances*)) from the issuance of such notice until the Diamond Opening Date has been achieved (or such later date as may be specified in Appendix 1 (*Construction Period Insurances*));
- 1.1.2 the Operation Period Insurances as set out in Appendix 2 (*Operation Period Insurances*) and the Company shall maintain such Direct Insurances until the Release Date; and
- 1.1.3 all other Direct Insurances that may be required to be effected by the Company from time to time by any applicable law or under any contract to which it is a party and shall maintain such Direct Insurances until the Release Date,

in each case, in a form reasonably satisfactory to the Intercreditor Agent (after consultation with the Insurance Adviser).

### 1.2 Reinsurance

The Company shall, if required under Clause 2.1.1 (*Policies*), procure that facultative reinsurance of each Direct Insurance is purchased and maintained in full force and effect throughout the period that such Direct Insurance is required by this Schedule 7 to be maintained.

# 1.3 Additional Insurances

- 1.3.1 The Intercreditor Agent may at any time, having consulted with the Insurance Adviser and acting reasonably and taking into account the availability in the international market place of the following relevant item on reasonable commercial terms, require the Company to:
  - (a) procure the amendment of any or all Insurances to cover increased risks and/or liabilities; and/or
  - (b) effect additional Insurances to cover risks and/or liabilities other than those specified in the scope of the Construction Period Insurances, the

- 218 -

Operation Period Insurances and the other Direct Insurances as would from time to time be insured in accordance with standard industry practice by a luxury resort and casino owner and operator in Macau carrying on the Permitted Businesses which does not self-insure (except in respect of deductibles required by insurers generally) and which is financed on a limited recourse basis,

in such amounts and, in the case of additional Insurances, with such deductibles, in each case as the Intercreditor Agent may reasonably require, taking into account, among other things, the basis on which the Projects are financed and the interests of the Senior Secured Creditors under the Senior Finance Documents.

- 1.3.2 In the event that the Company fails to effect any Insurance required to be effected pursuant to Clause 1.3.1 above, the Intercreditor Agent may effect such Insurance and the Company shall indemnify the Intercreditor Agent for the direct costs and expenses incurred by it as a result of effecting such Insurance.
- 1.3.3 The Company may effect additional Insurances other than those required by Clause 1.1 (*Direct Insurances*), Clause 1.2 (*Reinsurance*) or the other sub-clauses of this Clause 1.3 *provided* that such Insurances do not prejudice the interests of the Company or the Senior Secured Creditors under or in respect of any Insurance effected pursuant to such clauses.

#### Insurance Requirements with Respect to the Cotai Project

1.4

Notwithstanding any other provision of this Agreement or the other Senior Finance Documents to the contrary, with effect from the Fourth Amendment Effective Date:

- (a) except as set forth in paragraphs (b) and (c), and except for Clause 2.2 (save for Clause 2.2.4 and Clause 2.2.19 thereof) and Clause 5 below, this Schedule 7 (including its Appendices) shall not apply to the Cotai Project;
- (b) the Company shall effect and maintain, or cause Palo to effect and maintain, insurances and reinsurance on and in relation to the Cotai Project, its business constituted thereby and its assets necessary therefor against those risks and to the extent as is usual for companies carrying on the same or substantially similar business as the Cotai Project, in each case, with reputable insurers and reinsurers; and
- (c) the Company shall (i) assign, and procure that Palo assigns, such insurances and reinsurances to the Security Agent on behalf of the Secured Parties in accordance with paragraphs 2.3 and 2.4 of this Schedule 7 and (ii) ensure that the Senior Secured Creditors are named as co-insureds, as applicable, under such insurances.

- 219 -

## 2. INSURANCE UNDERTAKINGS

#### 2.1 Policies

The Company shall procure that:

- 2.1.1 each of the Direct Insurances is placed and maintained with one or more insurers authorised to operate in the Macau SAR. The coverage provided by such Direct Insurance shall be provided by insurers rated at least A- by Standard & Poor's Corporation or at least A by AM Best. Where this is not complied with the Company shall procure that the Direct Insurance is reinsured (minimum 95% of 100%) by insurers rated at least A- by Standard & Poor's Corporation or at least A by AM Best for their long term unsecured and unsubordinated debt or, in each case, in respect of:
  - (a) the Professional Indemnity Insurance referred to in paragraph 3 of Appendix 1 (*Construction Period Insurances*), rated at least BBB+ by Standard & Poor's Corporation or B by AM Best; and
  - (b) the Workers Compensation Insurance referred to in paragraph 4 of Appendix 1 (*Construction Period Insurances*), rated at least BBB by Standard & Poor's Corporation or B- by AM Best);
- 2.1.2 each of the Insurances is in a form and on terms acceptable to the Intercreditor Agent (including, without limitation, the level or period of any deductibles) consistent with the obligations of the Company under this Schedule 7;
- 2.1.3 each Construction Period Insurance and each Operation Period Insurance has endorsements in substantially the form set out in Part A of Appendix 3 (*Form of Endorsements for Direct Insurances*) (or, in the case of the Professional Indemnity Insurance referred to in paragraph 3 of Appendix 1 (*Construction Period Insurances*), the form set out in Part B of Appendix 3 (*Form of Endorsements for Direct Insurances*)) or in such other forms as the Intercreditor Agent (after consultation with the Insurance Adviser) reasonably approves in writing; and
- 2.1.4 each Reinsurance of each Construction Period Insurance and Operation Period Insurance has endorsements in substantially the form set out in Appendix 4 (*Form of Endorsements for Reinsurances*) or in such other forms as the Intercreditor Agent (after consultation with the Insurance Adviser) reasonably approves in writing.

#### 2.2 General Undertakings

The Company shall:

- 2.2.1 pay or procure the payment of all premiums payable under each of the Insurances promptly as required under the Insurances and if requested by the Intercreditor Agent, promptly produce to the Intercreditor Agent copies of receipts or other evidence of payment satisfactory to the Intercreditor Agent;
- 2.2.2 indemnify the Intercreditor Agent and any Senior Secured Creditor against any premium or premiums paid by that Senior Secured Creditor for any of the Insurances;
- 2.2.3 promptly on receipt by the Company, deliver an original cover note and an original policy for each of the Insurances to the Intercreditor Agent;

- 220 -

- 2.2.4 at least fifteen days prior to the earlier of the Substantial Completion or physical acceptance, use or occupancy by the Company of any part of the Projects or the expiry of the Construction Period Insurances, provide evidence reasonably satisfactory to the Intercreditor Agent (after consultation with the Insurance Adviser) that the Operation Period Insurances shall be in effect on and from the expiry of the Construction Period Insurances;
- 2.2.5 at least ten days prior to the expiry of any Insurance (and provided such Insurance is being renewed), provide to the Intercreditor Agent a certificate from the Company's insurance brokers (or, if the Company has no broker, Insurers) confirming the renewal of the policy relating to such Insurance, the renewal period, the amounts insured and any changes in terms or conditions;
- 2.2.6 take all action within its power to procure that nothing is at any time done or suffered to be done whereby any Insurance may be rendered void or voidable or may be suspended, impaired or defeated or any claim becomes uncollectable in full or in part, including, without limitation:
  - (a) complying with all of the requirements imposed on it under the Insurances;
  - (b) taking all action within its power to procure that at all times all parties to the Insurances (other than the Company and the Senior Secured Creditors) comply with all of the requirements under the Insurances; and
  - (c) taking all action necessary to maintain the Insurances as valid and up-to-date insurances;
- 2.2.7 not make any misrepresentation of any material facts or fail to disclose any material facts in respect of the Insurances which may have an adverse impact on the Insurances;
- 2.2.8 comply with each Direct Insurer's, and, where reinsurance is placed by the Company's insurance brokers, Reinsurer's risk management requirements set out in the policy documents for each Insurance;
- 2.2.9 promptly make and diligently pursue claims under the Insurances;
- 2.2.10 notify the Intercreditor Agent promptly upon becoming aware of any claim made under any of the Insurances where the actual or estimated totality of the amount of that claim exceeds USD500,000 (or its equivalent in other currencies) and of any occurrence which the Company considers could reasonably be expected to entitle it to submit a claim under any of the Insurances where the actual or estimated totality of the amount of that claim exceeds USD500,000 or its equivalent in other currencies;
- 2.2.11 in the event of any claim made under any of the Insurances where the actual or estimated totality of the amount of that claim exceeds USD500,000 (or its equivalent in other currencies) (not taking into account any relevant deductible for this purpose), provide a report to the Intercreditor Agent (and, if possible, procure a report from the Company's insurance broker to the Intercreditor Agent) which shall include a description of the loss;

- 221 -

- 2.2.12 notify the Intercreditor Agent immediately upon receipt of any proceeds in relation to any claims in excess of USD500,000 under the Direct Insurances;
- 2.2.13 ensure so far as reasonably possible that no Insurance can be terminated by the Direct Insurer, and, where reinsurance is placed by the Company's insurance brokers, Reinsurer for any reason (including failure to pay the premium or any other amount) unless the Intercreditor Agent and the Company receive at least thirty days' written notice (or such lesser period, if any, as may be specified from time to time by Direct Insurers, and, where reinsurance is placed by the Company's insurance brokers, Reinsurers in the case of war risks and kindred perils);
- 2.2.14 without prejudice to sub-clause 2.2.13 above, notify the Intercreditor Agent if any Direct Insurer, and, where reinsurance is placed by the Company's insurance brokers, Reinsurer cancels or gives notice of cancellation of any of the Insurances promptly on receipt of such notice;
- 2.2.15 notify the Intercreditor Agent of any act or omission or of any event which would reasonably be foreseen as invalidating or rendering unenforceable in whole or in part any of the Insurances;
- 2.2.16 notify the Intercreditor Agent promptly on becoming aware of any written proposal to make any material variation to any terms of any of the Insurances by any party to it;
- 2.2.17 not rescind, terminate or cancel any of the Insurances (unless replaced by a policy with the same coverage and otherwise meeting the requirements of this Schedule 7) nor agree to any variation to any of the material terms of the Insurances unless it obtains the prior written agreement of the Intercreditor Agent, which permission shall not be unreasonably withheld;
- 2.2.18 give the Intercreditor Agent and the Insurance Adviser such information about the Insurances (or as to any matter relevant to the Insurances) as the Intercreditor Agent reasonably requests from time to time; and
- 2.2.19 procure the delivery to the Intercreditor Agent by each of the insurance brokers (acceptable to the Intercreditor Agent (after consultation with the Insurance Adviser)) through whom (if any) at any time any of the (i) Direct Insurances are effected, prior to the Fourth Amendment Effective Date, of an Insurance Broker's Letter of Undertaking and (ii) Reinsurances are effected, prior to the Fourth Amendment Effective Date, of a Reinsurance Broker's Letter of Undertaking.

# 2.3 Assignment of Insurances

2.3.1 The Company (or in respect of any insurance required pursuant to Clause 1.4 (*Insurance Requirements with Respect to the Cotai Project*) the Company or Palo) shall, in accordance with the Assignment of Insurances (or the Palo

- 222 -

Assignment of Insurances, as the case may be), grant assignments in favour of the Security Agent on behalf of the Secured Parties over all of its rights, title and interest in the Direct Insurances (other than any public liability, third party liability, workers compensation or legal liability insurance or any other insurances the proceeds of which are payable to employees of the Company) held by it from time to time.

2.3.2 The Company shall give notice to each of the Direct Insurers (other than with respect to any public liability, third party liability, workers compensation, legal liability or any other insurances the proceeds of which are payable to employees of the Company) who are Macau Direct Insurers (as defined in the Assignment of Insurances (or the Palo Assignment of Insurances, as the case may be)) in the form of the Form of Notice of Assignment of Onshore Insurance Policy set out in the schedule to the Assignment of Insurances (or the Palo Assignment of Insurances, as the case may be) and shall procure an acknowledgement from each such Direct Insurer in the form of the Form of Acknowledgement from Macau Direct Insurer set out in the schedule to the Assignment of Insurances (or the Palo Assignment of Insurances, as the case may be) or such other form reasonably acceptable to the Security Agent.

# 2.4 Assignment of Reinsurances

- 2.4.1 The Company (or in respect of any insurance required pursuant to Clause 1.4 (*Insurance Requirements with Respect to the Cotai Project*) the Company or Palo) shall procure that each of the Direct Insurers grants an assignment in favour of the Security Agent on behalf of the Secured Parties, over all of its rights, title and interest in any Reinsurance held from time to time under Clause 1.2 (*Reinsurance*) or Clause 1.4 (*Insurance Requirements with Respect to the Cotai Project*) and/or the Reinsurance proceeds (other than relating to any public liability, third party liability or legal liability insurance or any other insurances the proceeds of which are payable to employees of the Company or Palo). Each assignment shall at all times be in the form of the Assignment of Reinsurances dated on or about the date of this Agreement unless otherwise agreed by the Security Agent (acting on the instructions of the Intercreditor Agent).
- 2.4.2 The Company shall procure that each such Direct Insurer gives notice to each Reinsurer with whom it has effected such Reinsurance in the form set out in Part 1 of Schedule 2 to the Assignment of Reinsurances and shall procure an acknowledgement from each such Reinsurer in the form set out in Part 2 of Schedule 2 to the Assignment of Reinsurances or such other form reasonably acceptable to the Security Agent.

## 3. FAILURE TO COMPLY WITH PROVISIONS OF INSURANCES

# 3.1 Notice of Non-Compliance

The Company shall notify the Intercreditor Agent as promptly as practicable if the Company has at any time failed to comply with this Schedule 7, explaining in reasonable detail the failure, whether the Company reasonably believes it can be remedied and, if so, how and by when.

- 223 -

#### 3.2 Annual Compliance Certificate

The Company shall, at the same time as delivery of its annual audited accounts pursuant to paragraph 1 of Part A of Schedule 5 (*Covenants*) to the Common Terms Agreement but, in any event, not less frequently than once every 12 months after the Signing Date, deliver to the Intercreditor Agent a certificate confirming compliance with this Schedule 7 or, if there is any non-compliance with this Schedule 7, explaining, in reasonable detail, the non-compliance, whether the Company reasonably believes it can be remedied and, if so, how and by when.

### 3.3 Action by Intercreditor Agent

If at any time and for any reason any Insurance required hereunder is not in full force and effect or if the Company fails to comply with any other provision of this Schedule 7, then, without prejudice to the rights of any of the Senior Secured Creditors under any Senior Finance Document, the Intercreditor Agent may (after consultation with the Insurance Adviser) thereupon on behalf of itself and the other Senior Secured Creditors that Insurance at the Company's expense is maintained such that full compliance with this Schedule 7 is restored. If that Insurance cannot be procured by the Intercreditor Agent, the Company shall (without prejudice to any other obligations of the Company under this Schedule 7 or any of the Senior Finance Documents) take or procure the taking of all reasonable steps to eliminate or minimise uninsured hazards as required by the Intercreditor Agent in writing (after consultation with the Insurance Adviser).

# 3.4 Continuing Obligations

Any notification by the Company of its failure to comply with this Schedule 7 shall not prejudice the rights of the Senior Secured Creditors under the Senior Finance Documents.

# 4. MARKET AVAILABILITY

Notwithstanding the foregoing provisions of this Schedule 7, the Company shall not be in breach of its obligations under this Schedule 7 if any Operation Period Insurances (other than insurances which the Company is required to maintain under applicable Legal Requirements) required to be entered into or maintained by the Company in the terms required under this Schedule 7 are not available to the Company or, as the case may be, the Prime Contractor in the international insurance market place on reasonable commercial terms. Without prejudice to any other terms that may be commercially reasonable, for the purposes of this paragraph 4, such commercial terms shall be reasonable if the premium payable in respect of the sum required to be insured under such Operation Period Insurance pursuant to Appendix 2 to this Schedule 7 (or, where no such sum is referred to, the limit of indemnity or limit of liability specified therein) is not more than 125% of the amount derived by multiplying the rate set out opposite such Insurance in Appendix 7 (*Operation Period Insurance Costs*) by such sum or in respect of all Operation Period Insurances (including such Operation Period Insurance) is not more than 125% of the total of such amounts derived for each Operation Period Insurance.

- 224 -

# 5. INSURANCE PROCEEDS

# 5.1 **Conduct of Claims - Company**

Subject to Clause 5.3 (*Conduct of Claims - Default*) below, the Company, or as the case may be, Palo has the sole conduct of all claims under the Insurances arising from any one loss (for which purpose, two or more claims made in respect of the same, or reasonably related, circumstances are taken to relate to one loss) where the actual or estimated totality of that loss is less than or equal to USD5,000,000 (or its equivalent in other currencies). For any loss where the actual or estimated totality of claims arising is more than USD5,000,000 (or its equivalent in other currencies), the Company, or as the case may be, Palo shall not negotiate, compromise or settle any claim without the prior consent of the Intercreditor Agent (after consultation with the Insurance Adviser) (not to be unreasonably withheld).

## 5.2 Application of Proceeds

The Company, or as the case may be, Palo shall ensure that:

- 5.2.1 subject to sub-clause 5.2.4 below and prior to the delivery of an Enforcement Notice, all proceeds of any claim under any Insurance relating to physical damage or loss:
  - (a) to the extent not required for the purpose of mandatory prepayment pursuant to Schedule 9 (*Mandatory Prepayment*), shall be applied towards the repair or restoration of the damaged Project Site Facilities or Cotai Site Facilities, as applicable, in accordance with Schedule 6 (*Accounts*); or
  - (b) to the extent required for the purpose of mandatory prepayment pursuant to Schedule 9 (*Mandatory Prepayment*), shall be applied towards mandatory prepayment in accordance with Clause 8.3 (*Mandatory Prepayment*);
- 5.2.2 subject to sub-clause 5.2.4 below and prior to the delivery of an Enforcement Notice, all proceeds of any claim under any Insurance relating to loss of revenue or business interruption shall be applied in accordance with Schedule 9 (*Mandatory Prepayment*);
- 5.2.3 subject to sub-clause 5.2.4 below and following the delivery of an Enforcement Notice, all proceeds of any claim under any Insurance shall be applied as directed by the Security Agent; and
- 5.2.4 all proceeds of any public liability, third party liability, workers compensation or legal liability insurance, or directors and officers insurance or any other insurances the proceeds of which are payable to employees of the Company, or as the case may be, Palo shall be applied to its intended purpose.

- 225 -

#### 5.3 **Conduct of Claims - Default**

Notwithstanding any other provisions of this Clause 5, if an Enforcement Notice has been delivered, then the Security Agent in consultation with the Insurance Adviser shall have sole conduct of all claims under the Insurances.

# 5.4 Insolvency of Direct Insurers

For the purpose of conduct of claims and application of proceeds under any Reinsurance taken out by a Direct Insurer, references to delivery of an Enforcement Notice in this Clause 5 shall include delivery of the notice referred to in Clause 2.2.1 (*Enforcement of Assignment*) of the Assignment of Reinsurances to such Direct Insurer following an "Insolvency Event" (as defined in the Assignment of Reinsurances).

### **APPENDIX 1**

#### INSURANCES

# CONSTRUCTION PERIOD INSURANCES

#### 1. Construction All Risks Insurance

#### 1.1 Insured

The Company and the following Co-Insureds:

- (1) the Prime Contractor and/or all other Contractors and Subcontractors of any tier and/or suppliers and/or others engaged to provide goods and services in connection with the Projects;
- (2) in respect of activities on or about the Site only, architects, surveyors, engineers and other professional consultants (including the Advisers); and
- (3) the Secured Parties,
- each for their respective rights and interests.

# 1.2 Insured Property

The permanent and temporary works, equipment, services, materials, plant (other than constructional plant, tools and equipment belonging to or the responsibility of the Contractors or Subcontractors of any tier), machinery, stocks, spares, temporary buildings (if any) and all other property used or for use in connection with the Projects which the Company owns or for which it is responsible (excluding such other property comprising operation period furniture, fittings and equipment in respect of which separate Insurances, on substantially the same terms as the Construction All Risks Insurance required hereunder and/or otherwise reasonably acceptable to the Intercreditor Agent (after consultation with the Insurance Adviser), have been placed prior to the Company acquiring ownership thereof or being responsible therefor).

#### 1.3 Coverage

All risks of physical loss or damage which are normally insurable.

#### 1.4 Sum Insured

An amount equivalent to the full reinstatement or replacement value of the Insured Property plus Removal of Debris plus Professional Fees.

#### 1.5 Territorial Limits

Anywhere within Macau SAR.

## 1.6 **Period of Insurance**

For the full period up to the Diamond Opening Date plus the first 12 months of the Defects Liability Period.

- 227 -

## 1.7 **Permitted Exclusions**

•

•

To include:

- War, Civil War etc.
- Nuclear Risks
- Wear and Tear
- Unexplained shortage
- Consequential financial losses
- Terrorism
- DE3 type defective design, workmanship and materials exclusion
- Mold

# **Required Extensions and Conditions**

- Professional Fees Clause
- Debris Removal Clause
- 72 Hour Clause
- Free Issue Materials Clause
- Automatic Increase Clause (110 per cent.)
- Extra Charges (20 per cent.)
- Strikes, Riot and Civil Commotion
- Local/Public Authorities and Clause
- Munitions of War Clause
- Extended Maintenance
- Automatic Reinstatement of Sum Insured
- Plans and Documents
- Inland Transit/Offsite Storage and Offsite Fabrication
- Advance Payments Clause
- Marine 50/50 Clause

- 228 -

1.8

- General Waiver of Subrogation
- Multiple Insureds Clause (LEG form) Senior Secured Creditors Special Conditions (Loss Payee and Notices)
- Primary Insurance Clause
- Assignment of Insurance

# 1.9 Maximum Deductible

Not to exceed:

- USD250,000 each and every loss in respect of damage due to Acts of God/Theft/Fire/Maintenance Period/Consequences of Defective Design, Material and Workmanship (DE3)
- 20% of the loss amount each and every loss subject to a minimum deductible of USD100,000 in respect of Water Damage
- USD250,000 each and every loss in respect of Defective Design, Material and Workmanship (DE5)
- 50% of the loss amount each and every loss subject to a minimum of USD35,000 in respect of Scaffolding
- USD35,000 each and every loss for Others

#### 2. Third Party Liability Insurance

#### Insured

2.1

2.2

The Company and the following Co-insureds:

- (1) the Prime Contractor and/or all other Contractors and Subcontractors of any tier and/or suppliers and/or others engaged to provide goods and services in connection with the Projects;
- (2) in respect of activities on or about the Site only, architects, surveyors, engineers and other professional consultants (including the Advisers); and
- (3) the Secured Parties,
- each for their respective rights and interests.

### Interest

To indemnify the Insured in respect of all sums that it may become legally liable to pay consequent upon death, personal injury and disease to persons, loss or damage to property, obstruction, loss of amenities, stoppage of traffic happening or arising out of or in connection with the Projects.

- 229 -

# 2.3 Limit of Indemnity

Not less than USD50,000,000 in respect of any one occurrence, the number of occurrences being unlimited.

# 2.4 Territorial Limits/Jurisdiction

# Worldwide.

# 2.5 **Period of Insurance**

As per the Construction All Risks Insurance.

# 2.6 **Permitted Exclusions**

To include:

- Liability for death, illness, disease or bodily injury sustained by employees of the Insured
- Liability for loss or damage to property which is reasonably foreseeable as being inevitable having regard for the nature of work undertaken
- Liability arising out of the use of mechanically propelled vehicles whilst required to be compulsorily insured by legislation in respect of such vehicles
- Liability in respect of predetermined penalties or liquidated damages imposed under any contract entered into by the Insured
- Liability in respect of loss or damage to property in the care, custody and control of the Insured or to the permanent or temporary works
- Liability arising from the ownership, possession or use of any aircraft or waterborne vessel
- Liability arising from seepage and pollution unless caused by a sudden, unintended and unexpected occurrence
- War, civil war etc.
- Nuclear risks
- Terrorism

# 2.7 **Required Extensions and Conditions**

- Cross Liability Clause
- Contractual Liability
- Underground Services

- 230 -

- Vibration Removal or Weakening of Support
- Munitions of War Clause
- General Waiver of Subrogation
- Costs and Expenses in addition to the Limit of Indemnity (other than North America)
- Worldwide jurisdiction
- Multiple Insureds Clause (LEG form) Senior Secured Creditors Special Conditions (Loss Payee and Notices)
- Primary Insurance Clause

## 2.8 Maximum Excess

Not to exceed:

- 20% of the loss amount each and every loss subject to a maximum of USD200,000 and minimum of USD10,000 in respect of claims on Vibration, Removal and Weakening of Support
- 10% of the loss amount each and every loss subject to a maximum of USD100,000 and minimum of USD10,000 in respect of Underground Services/Water Damage
- 40% of the loss amount each and every loss subject to a maximum of USD100,000 and minimum of USD10,000 in respect of Oil Filled Cables and Fibre Optic Cables
- USD200,000 each and every occurrence in respect of damage to property owned by or in the care custody and control of the Company
- USD25,000 each and every occurrence in respect of Third Party Property Damage and Interference with right of way/property/enjoyment of use by obstruction, trespass, loss of amenities or nuisance

# 3. Professional Indemnity Insurance (Prime Contractor)

#### 3.1 Insured

The Prime Contractor.

#### 3.2 Interest

To indemnify the Insured in respect of their legal liability for claims made against them during the period of insurance for any negligent act, error or omission in the conduct and execution of their professional activities and duties.

- 231 -

# 3.3 Limit of Indemnity

USD20,000,000 in respect of any one claim and in the aggregate during the period of insurance.

3.4 **Territorial Limits** 

Macau SAR.

# 3.5 **Period of Insurance**

To be effected and maintained on an annually renewable basis for the period commencing no later than the date of this Agreement until 6 years from the later of the Expansion Opening Date and the Diamond Opening Date.

### 3.6 Excess

Not to exceed USD2,500,000 in respect of each occurrence or such higher amount as may be agreed by the Intercreditor Agent.

# 4. Compulsory Insurance

Insurances required to comply with all statutory requirements including Workers Compensation and Motor Liability Insurances. The Compulsory Insurance effected by the Company shall contain an indemnity clause in favour of the Senior Secured Creditors.

### **APPENDIX 2**

# **OPERATION PERIOD INSURANCES**

# 1. **Property All Risks Insurance**

### 1.1 Insured

The Company and the following Co-insureds:

(1) The Secured Parties,

each for their respective rights and interests.

### 1.2 Insured Property

Property and interests of every description used for or in connection with the ownership and/or maintenance and operation of the facilities unless more specifically insured under the Construction All Risks Insurance (Item 1 of Appendix 1) - this shall include mechanical and electrical equipment if applicable.

#### 1.3 Coverage

All risks of physical loss or damage which are normally insurable.

#### 1.4 Sum Insured

A first loss limit of USD850,000,000 any one occurrence or any other first loss limit to be agreed by the Intercreditor Agent after review of the estimated maximum loss calculation

#### 1.5 Territorial Limits

Anywhere in the Macau SAR.

# 1.6 **Period of Insurance**

From the earlier of the Substantial Completion or physical acceptance, use or occupancy by the Company of any part of the Projects or termination of the Construction All Risks Insurance until the Release Date (or such longer period of insurance as may be agreed by the Intercreditor Agent and the Company).

# 1.7 Permitted Exclusions

To include:

- War, civil war etc.
- Nuclear Risks
- Wear, tear and general deterioration
- Unexplained shortages

- 233 -

- Terrorism
- Consequential financial losses
- Any part of the Insured Property which is in itself defective in design, workmanship and materials but this exclusion shall not apply to other parts of the Insured Property damaged in consequence of such a defect.

#### 1.8 **Required Extensions and Conditions**

- 72 hour clause
- Temporary removal
- Munitions of War Clause
- Strikes, Riot and Civil Commotion
- Minimisation of Loss
- Advance Payment Clause
- Temporary Repairs
- Automatic Reinstatement of Sum Insured
- Including pollution and contamination to the Insured Property arising from an event which itself is not otherwise excluded
- Contract works including works and temporary works erected or in the course of erection including materials and other things for incorporation in the Project Works up to a sum of USD1,000,000
- Capital Additions
- Debris Removal
- Professional Fees
- Local/Public Authorities Clause
- Mechanical or electrical breakdown of the Insured Property
- Replacement of computer records
- General Waiver of Subrogation (to include Expansion construction participants)
- Multiple Insureds Clause (LEG form) Senior Secured Creditors Special Conditions (Loss Payee and Notices)
- Primary Insurance Clause
- Assignment of Insurance

- 234 -

#### 1.9 Maximum Deductible

Not to exceed USD500,000 in respect of each occurrence or such lesser amount as may be agreed between the Company and the Intercreditor Agent if available on commercially reasonably terms.

# 2. Business Interruption Insurance

## 2.1 Insured

The Company and the following Co-insureds:

(1) the Secured Parties,

each for their respective rights and interests.

# 2.2 Interest

To indemnify the Insured for fixed costs following loss or damage which is indemnifiable or would be indemnifiable but for the application of the excess under the Property All Risks Insurance.

# 2.3 Sum Insured

A sum sufficient to cover the sums the subject of the Indemnity for the Indemnity Period.

# 2.4 Indemnity Period

The period commencing from the date of the loss or damage and ending when the results of the insured business cease to be affected in consequence of the loss or damage. Not exceeding the Indemnity Period Limit.

The Indemnity Period Limit shall not be less than 12 months.

# 2.5 Territorial Limits

As for the Property All Risks Insurance.

# 2.6 **Period of Insurance**

As for the Property All Risks Insurance.

# 2.7 Permitted Exclusions

As for the Property All Risks Insurance.

# 2.8 Required Extensions

Suppliers Extension

- 235 -

- Prevention of Access
- Public Utilities
- Payments on Account
- Automatic Reinstatement of Sum Insured
- Professional Accountants Clause
- General Waiver of Subrogation (including the Trusts and to include Expansion construction participants)
- Multiple Insureds Clause (LEG form) Senior Secured Creditors Special Conditions (loss payee and notices)
- Primary Insurance Clause
- Assignment of Insurance
- Interruption or interference arising out of an event insured under the Defects Liability Period covers for the Original Project and the Expansion under the Construction All Risks Insurance

#### 2.9 Maximum Excess

Not to exceed 30 days in the aggregate for each and every loss.

#### 3. Third Party and Products Liability Insurance

#### 3.1 Insured

3.2

The Company and the following Co-insureds:

(1) The Secured Parties,

each for their respective rights and interests.

# Interest

To indemnify the Insured in respect of all sums that it may become legally liable to pay consequent upon death, personal injury and disease to persons, loss or damage to property, obstruction, loss of amenities, stoppage of traffic happening or arising from or in connection with the operation and maintenance of the Projects and the services to be provided thereby.

# 3.3 Limit of Indemnity

Not less than USD50,000,000 in respect of any one occurrence, the number of occurrences being unlimited but in the aggregate in respect of sudden and accidental pollution and products liability.

- 236 -

# 3.4 Territorial Limits/Jurisdiction

Worldwide excluding USA and Canada in respect of Territorial Limit

Worldwide in respect of Jurisdiction

# 3.5 **Period of Insurance**

As for Wynn Resorts Umbrella and Excess Liability Insurances until the Release Date (or such longer period of insurance as may be agreed by the Intercreditor Agent and the Company)

# 3.6 **Permitted Exclusions**

To include:

- Liability for death, illness, disease or bodily injury sustained by employees of the Insured
- Liability for loss or damage to property which is reasonably foreseeable as being inevitable having regard for the nature of work undertaken
- Liability arising out of the use of mechanically propelled vehicles whilst required to be compulsorily insured by legislation in respect of such vehicles
- Liability in respect of predetermined penalties or liquidated damages imposed under any contract entered into by the Insured
- Liability in respect of loss or damage to property in the care, custody and control of the Insured but this exclusion not to apply to employees' or visitors' property including vehicles and their contents
- Liability arising out of technical or professional advice other than in respect of death or bodily injury to persons
- Liability arising from the ownership, possession or use of any aircraft or waterborne vessel
- Liability arising from seepage and pollution unless caused by a sudden, unintended and unexpected occurrence
- War, civil war etc.
- Nuclear risks

3.7

# **Required Extension and Conditions**

- Cross Liability Clause
- Contractual Liability

- 237 -

- Costs and Expenses in addition to the Limit of Indemnity (other than North America)
- Advertising Liability (relating to physical damage from billboards, signs, etc.)
- General Waiver of Subrogation
- Multiple Insureds Clause (LEG form) Senior Secured Creditors Special Conditions (loss payee and notices)
- Primary Insurance Clause
- Assignment of Insurance

# 3.8 Maximum Excess

Not to exceed USD100,000 each occurrence or such higher amount as may be agreed by the Intercreditor Agent.

#### 4. Terrorism Insurance (Including Business Interruption)

# 4.1 Insured

To follow the terms and conditions of the Property All Risks Insurance (1) and Business Interruption Insurance (2), with the following exceptions: Maximum Deductible - Not to exceed USD1,000,000 (or such lesser amount as may be agreed between the Company and the Intercreditor Agent if available at commercially reasonable terms) each and every loss for Property Damage and 60 days for Business Interruption. Maximum Sum Insured - USD400,000,000.

#### 5. Fidelity Guarantee/Crime Insurance

# 5.1 Insured

The Company and the Secured Parties

each for their respective rights and interests.

# 5.2 Coverage

Direct pecuniary loss of money, negotiable instruments or goods caused by acts of fraud or dishonesty by any employee or any other person.

# 5.3 Limits of Liability

Not less than USD25,000,000 in respect of any one occurrence or such higher amount as may be required to fully cover the amount of money on site at any one time.

- 238 -

# 5.4 Territorial Limits

Worldwide

# 5.5 **Period of Insurance**

From the earlier of the Substantial Completion or physical acceptance, use or occupancy by the Company of any part of the Projects or termination of the Construction All Risks Insurance until the Release Date (or such longer period of insurance as may be agreed by the Intercreditor Agent and the Company).

# 5.6 Maximum Deductible

Not to exceed USD150,000 in respect of each occurrence or such higher amounts as may be agreed by the Intercreditor Agent.

#### 6. **Money Insurance**

#### 6.1 Insured

The Company and the Secured Parties

each for their respective rights and interests.

# 6.2 Coverage

Loss, destruction or damage of money in transit, money at the business premises of the Insured during office hours and money in locked safe/drawer in the business premises of the Insured after office hours.

# 6.3 Limits of Liability

Not less than USD30,000,000 in respect of any one occurrence or such higher amount as may be required to fully cover the amount of money on site at any one time.

# 6.4 Territorial Limits

Worldwide

# 6.5 **Period of Insurance**

From the earlier of the Substantial Completion or physical acceptance, use or occupancy by the Company of any part of the Projects or termination of the Construction All Risks Insurance until the Release Date (or such longer period of insurance as may be agreed by the Intercreditor Agent and the Company).

# 6.6 Maximum Deductible

Not to exceed USD50,000 in respect of each occurrence or such higher amounts as may be agreed by the Intercreditor Agent.

- 239 -

#### APPENDIX 3

# FORM OF ENDORSEMENTS FOR DIRECT INSURANCES

# PART A

| INSURED PARTIES                       | [ | ] (the " <b>Insureds</b> ")                                 |
|---------------------------------------|---|---|
| [POLICY/COVER NOTE] REFERENCE NUMBER: | [ | ] (together with these endorsements, the " <b>Policy</b> ") |
| EFFECTIVE DATE:                       | [ | ]   |

It is understood and agreed between the Insurer and the Insureds that, notwithstanding any other provision of this Policy, the following endorsement shall apply:

#### **Section I: Definitions**

#### 1. In this endorsement:

"**Common Terms Agreement**" means the agreement dated 14 September 2004 between, amongst others, the Company, the Intercreditor Agent and the Security Agent as amended, consolidated, supplemented, novated or replaced from time to time;

"**Company**" means Wynn Resorts (Macau) S.A.;

"Intercreditor Agent" means [ — ] in its capacity as Intercreditor Agent for the Senior Secured Creditors and includes its successors in that capacity;

"Insured" means those parties so described in the Policy Schedule;

"Insurers" means the insurer or insurers underwriting this insurance policy;

"Projects" has the meaning given to it in the Common Terms Agreement;

"Secured Parties" has the meaning referred to in the Common Terms Agreement;

"Security Agent" means [ — ] in its capacity as agent and security trustee for the Secured Parties and includes its successors in that capacity;

"Senior Secured Creditors" has the meaning given in the Common Terms Agreement.

# Section II: Policy formation / basis

Separate Policy

2.

All the provisions of this Policy (except for those relating to limits of liability) shall operate as if there were a separate policy covering each Insured. Accordingly, the

- 240 -

liability of the Insurers under this Policy to any one of the Insured shall not be conditional upon the due observance and fulfilment of any other Insured of the terms of this Policy and of any duties imposed upon it relating thereto and shall not be affected by any failure in such observance or fulfilment of any such other Insured.

#### 3. Interest of the Secured Parties

The Insurers acknowledge that the Secured Parties and (in respect of third party liabilities) their respective officers, directors, employees, secondees and assigns are each additional co-insureds under this Policy and that the premium specified in this Policy provides consideration for their being co-insured parties.

# 4. Liability for premium

Neither the Intercreditor Agent, the Security Agent nor the Secured Parties shall be liable for the payment of any premium under this Policy although they may choose to pay the premium. This shall not relieve the Company from its obligations to pay any premium under this Policy.

- 5. Disclosure
- 5.1 The Insurers acknowledge to the Secured Parties alone that (i) they have received adequate information in order to evaluate the risk of insuring the Company in respect of the risks hereby insured on the assumption that such information is not materially misleading, (ii) there is no information which has been relied on or is required by the Insurers in respect of their decision to co-insure the Secured Parties or their directors, officers, employees or agents, and (iii) in agreeing to enter into this Policy, they have not relied upon or taken into account any information supplied to them by any Secured Party. The acknowledgements provided by the Insurers in this clause 5.1 shall have no effect on any rights that the Insurers might have had under or in relation to the Policy against any party (including the Company) other than the Secured Parties in the absence of such acknowledgements.
- 5.2 Non-disclosure or misrepresentation by one Insured shall not be attributable to any other Insured who did not actively participate in that nondisclosure or misrepresentation. Without prejudice to the protections afforded to the Insured by this endorsement, no one Insured represents or warrants the adequacy or accuracy of any information provided or representation made by or on behalf of any other Insured.

# Section III: Rights to avoid / cancel or change Policy terms

- 6. Non-vitiation
- 6.1 Paragraph (i) It is noted and agreed that if the insured described in the schedule comprises more than one insured party each operating as a separate and distinct entity then (save as provided in this multiple insured's clause) cover hereunder shall apply in the same manner and to the same extent as if the individual polices had been issued to each such insured party provided that the total liability of the insurers to all of the insured parties collectively shall not exceed the sums insured and limits of indemnity including and inner limits set by memorandum or endorsement stated in the policy.

- 6.2 Paragraph (ii) It is understood and agreed that any payment or payments by Insurers to any one or more such insured parties shall reduce to the extent of that payment insurers liability to all such parties arising from any one event giving rise to a claim under this policy and (if applicable) in the aggregate.
- 6.3 Paragraph (iii) It is further understood that the insured parties will at all times preserve the various contractual rights and agreements entered into by the insured parties and the contractual remedies of such parties in the event of loss or damage.
- 6.4 Paragraph (iv) It is further understood and agreed that insurers shall be entitled to avoid liability to or (as maybe appropriate) claim damages from any of the insured parties in circumstances of fraud, material misrepresentation, material non-disclosure or breach of any warranty or condition of this policy each referred to in this clause as a vitiating act.
- 6.5 Paragraph (v) It is however agreed that (save as provided in this multiple insured's clause) a vitiating act committed by one insured party shall not prejudice the right to indemnity of any other insured party who has an insurable interest and who has not committed a vitiating act.
- 6.6 Paragraph (vi) Insurers hereby agree to waive all rights of subrogation which they may have or acquire against any insured party except where the rights of subrogation or recourse are acquired in consequence of or otherwise following a vitiating act in which circumstances insurers may enforce such rights notwithstanding the continuing or former status of the vitiation party as insured.
- 7. Cancellation
- 7.1 The Insurers agree that they shall not seek to cancel or suspend this insurance except: (i) for non payment of premium; or (ii) where an insured party consistently fails to comply with Insurers' requirements relating to survey or loss control action points.
- 7.2 The Insurers shall promptly notify the Intercreditor Agent in writing:
  - (a) in the event of any suspension, cancellation or termination of this Policy by the Insurers or the Insured; and
  - (b) of any act or omission or any event of which any Insurer has knowledge and which might invalidate or render unenforceable, in whole or in part, the Policy.

The cover provided by this Policy shall continue in force and unaltered for at least 30 days after written notice of such suspension, cancellation, termination or non-renewal is given to the Intercreditor Agent. Nothing in this clause shall give the Insurers any right to suspend, cancel or terminate this Policy which the Insurers do not otherwise have under this Policy.

7.3 The Insurers shall promptly notify the Intercreditor Agent in writing of any default in the payment of premium and shall give the Intercreditor Agent at least 30 days notice in writing before voiding this Policy for non-payment of premium, in order to give an opportunity for that premium to be paid within the notice period.

- 242 -

## 8. Changes in cover

The Insurers shall give the Intercreditor Agent at least 30 days notice in writing before any reduction in cover or increase in excess or deductible under this Policy takes effect and shall promptly notify the Intercreditor Agent of any fact, change of circumstance or occurrence of which any Insurer is aware which is material to the risks insured against under the Policy or which would result in any reduction in limits or alteration in coverage or increase in deductions or exclusions. Nothing in this clause shall give the Insurers any right which they do not otherwise have to reduce cover or increase any excess or deductible under this Policy.

#### 9. Amendments to Endorsement

During the term of this Policy, the provisions of this endorsement may only be amended by written agreement between the Company, the Insurers and the Intercreditor Agent, such amendment to be endorsed on the Policy.

#### Section IV: Claims

10. Notice of claims

Notice of claim by the Secured Parties or any other party entitled to indemnity under the Policy shall, in the absence of manifest error, be accepted by Insurers as a valid notification of claim on behalf of all other Insureds subject to the full terms of the Policy.

#### 11. Claim Payments / Loss Payee

Payments made in accordance with this clause 11 shall, to the extent of the payment, discharge the Insurers' liability to pay Company or any other Insured.

11.1 In respect of the insurance under this Policy of material damage risks only

All claim payments or return premium shall be paid into such account as the Security Agent as loss payee may specify in writing.

11.2 In respect of the insurance under this Policy of public liability risks only

All claim payments in respect of a third party liability shall be paid to person(s) whose claim(s) constitute the risk or liability insured against except in the case where the Insured has properly discharged its liability to such person(s), in which case the claim payment shall be paid to such account as the Security Agent directs in writing.

Any return premiums shall be paid to such account as the Security Agent directs in writing.

11.3 In respect of the insurance under this Policy of loss of revenue risks only

All claim payments or return premiums shall be paid to such account as the Security Agent directs in writing.

- 243 -

# 11.4 Set-off

The Insurers may, at their discretion, deduct overdue unpaid premium from claims settlements but shall not set off or deduct premium that is not overdue or any other amounts payable by Company under or in relation to the Policy.

# 12. Waiver of subrogation

The Insurers waive all rights of subrogation howsoever arising which they may have or acquire against any Insured described within the appropriate Schedules arising out of any Occurrence in respect of which any claim is admitted and is insured hereunder for the benefit of such Insured except against any:

- (i) such Insured (or officer, director, employee, agent or assign) who has caused or contributed to such an occurrence or claim by fraud, deliberate misrepresentation, deliberate non-disclosure or deliberate breach of policy condition; or
- (ii) consultant or equivalent professional party to the extent that their professional errors, omissions or activities not covered by this Policy have caused or contributed to a loss covered under this Policy; or
- (iii) supplier or manufacturer to the extent that their errors, omissions or activities not covered by this Policy have caused or contributed to a loss covered under this Policy; or
- (iv) such Insured to the extent that they are entitled to recover in respect of a loss under cover falling within sub-clause 13(a)–(e) below (or would be so insured if cover in the terms set out in this Policy had not been taken out);

# 13. Primary insurance

The Insurers agree that this insurance provides the primary cover for risks insured under this Policy. In the event that any risk insured under this Policy is also insured under any other policy of insurance effected by any Insured, the Insurers agree to indemnify the Insured as if such other policy of insurance did not exist except in respect of:

- (a) excess layers of third party cover effected specifically for the Projects;
- (b) any public liability claim against the Insured which exceeds the applicable limit of indemnity under this Policy, in which case the liability of the Insurers for additional legal costs and expenses shall be limited to the proportion that the applicable limit of indemnity bears to the total claim against the Insured;
- (c) any claim under this Policy to which a Marine 50/50 Clause applies;
- (d) any claim made under a Contingent Motor Liability extension to this Policy; or

- 244 -

(e) any claim relating to a loss which is insured against (or would be insured but for a double insurance provision or similar or the application of a deductible) under:

- (i) any other policy specifically effected for the construction or operational phase(s) of the Projects; or
- (ii) a latent or inherent defects policy or engineering or mechanical breakdown policy specifically effected for the Projects;
- or a related business interruption insurance policy.

#### Section V: Miscellaneous

14. Notice of Security Interest

14.1 The Insurers acknowledge that by an assignment contained in the Assignment of Insurances between the Company and the Security Agent dated [ ] (as amended, consolidated, supplemented, novated or replaced from time to time) (the "Assignment"), the Company assigned by way of security to the Secured Parties all benefits and rights in respect of this insurance and all claims and returns of premiums in respect thereof to which the Insured is or may at a future time become entitled. The Insurers confirm that they have not been notified of any other assignment of or security interest in the Company's interest in this insurance.

- 14.2 At the reasonable request of the Intercreditor Agent and at its expense and subject to any legal, contractual **or** regulatory restrictions, the Insurers shall make those documents contained within their placing and claims files to which the Company would be entitled to have access available to the Intercreditor Agent or the Insurance Adviser and provide copies of such documents to them.
- 15. Notice
- 15.1 All notices or other communications under or in connection with the Policy will be given by fax and post. Any such notice given by Insurers will be deemed to be given on the earlier of:
  - (i) if by fax, when transmitted but only if the sender's fax machine confirms successful transmission; and
    - (ii) if by post, within 2 business days of release from the relevant Insurer's office.
- 15.2 The address and fax number of the Intercreditor Agent and the Security Agent for all notices under or in connection with the Policy are those notified from time to time by the Intercreditor Agent or the Security Agent for this purpose to the insurance broker at the relevant time. The initial address and fax number of the Intercreditor Agent and the Security Agent are as follows:

Intercreditor Agent: [ — ]

Address: [ — ]

- 245 -

Tel: [ — ] Fax: [ — ] Attention: [ — ] Security Agent: [ — ] Address: [ — ] Tel: [ — ] Fax: [ — ] Attention: [ — ]

16. Governing law & Jurisdiction

The Policy shall be governed and interpreted in accordance with English law.

This endorsement overrides any conflicting provision in this Policy.

#### PART B

At the request of the Insured, it is noted that the Company will give written notice to [ — ] (herein referred to as "the Lender") at its last known address notified to the Company in writing in the following events:

(a) issuance and/or receipt of cancellation notice pursuant to the cancellation clause of the policy;

(b) event(s) leading to suspension or termination of this insurance other than natural expiry of this insurance;

(c) default in payment of premium by the insured; and

(d) request for reduction in the Limit of Indemnity and/or increase in Excess by the Insured or the Company.

Subject to other terms and conditions of this insurance, the Company agrees not to cancel, suspend or terminate this insurance (other than natural expiry); to reduce the Limit of Indemnity and/or increase the Excess before the expiry of the 30th day from the issuance of such notice to the Lender.

- 246 -

#### **APPENDIX 4**

Γ

ſ

# FORM OF ENDORSEMENTS FOR REINSURANCES

# REINSURANCE [CONTRACT/COVER NOTE] REFERENCE NUMBER:

**EFFECTIVE DATE:** 

] (together with these endorsements, the "Reinsurance Contract") ]

It is understood and agreed between the Reinsurer, the Insurers and the Insureds that, notwithstanding any other provision of this Policy, the following endorsement shall apply:

# Section I: Definitions

1.

In this endorsement:

"Common Terms Agreement" means the agreement dated 14 September 2004 between, amongst others, the Company, the Intercreditor Agent and the Security Agent as amended, consolidated, supplemented, novated or replaced from time to time;

"Company" means Wynn Resorts (Macau) S.A.;

"Intercreditor Agent" means [ — ] in its capacity as Intercreditor Agent for the Senior Secured Creditors and includes its successors in that capacity;

"Insured" means those parties so described in the Original Policy;

"Original Policy" means the direct insurance policy reinsured by this reinsurance contract;

"Projects" has the meaning given to it in the Common Terms Agreement;

"Reinsured" means the Insurer or Insurers of the Original Policy;

"Reinsurers" means the insurer or insurers underwriting this insurance policy;

"Secured Parties" has the meaning referred to in the Common Terms Agreement;

"Security Agent" means [ — ] in its capacity as agent and security trustee for the Secured Parties and includes its successors in that capacity;

"Senior Secured Creditors" has the meaning given in the Common Terms Agreement.

- 247 -

#### Section II: Policy formation / basis

#### 2. Separate Policy

All the provisions of this Policy (except for those relating to limits of liability) shall operate as if there were a separate Original Policy covering each Insured. Accordingly, the liability of the Reinsurers under this Policy to any one of the Re-insured shall not be conditional upon the due observance and fulfilment of any other Insured of the terms of the Original Policy and of any duties imposed upon it relating thereto and shall not be affected by any failure in such observance or fulfilment of any such other Insured.

3. Interest of the Secured Parties

The Reinsurers acknowledge that the Secured Parties and (in respect of third party liabilities) their respective officers, directors, employees, secondees and assigns are each additional co-insureds under the Original Policy and that the premium specified in this Policy provides consideration for their being co-insured parties.

4. Liability for premium

Neither the Intercreditor Agent, the Security Agent nor the Secured Parties shall be liable for the payment of any premium under this Policy although they may choose to pay the premium. This shall not relieve the Reinsured from its obligations to pay any premium under this Policy.

#### 5. Disclosure

- 5.1 The Reinsurers acknowledge to the Secured Parties alone that (i) they have received adequate information in order to evaluate the risk of insuring the Reinsured in respect of the risks hereby insured on the assumption that such information is not materially misleading, (ii) there is no information which has been relied on or is required by the Reinsurers in respect of their decision to insure the Secured Parties or their directors, officers, employees or agents, and (iii) in agreeing to enter into this Policy, they have not relied upon or taken into account any information supplied to them by any Secured Party. The acknowledgements provided by the Reinsurers in this clause 5.1 shall have no effect on any rights that the Reinsurers might have had under or in relation to the Policy against any party (including the Company) other than the Secured Parties and the Intercreditor Agent in the absence of such acknowledgements.
- 5.2 The Reinsurers acknowledge that under the Original Policy non-disclosure or misrepresentation by one Insured shall not be attributable to any other Insured who did not actively participate in that non-disclosure or misrepresentation. Without prejudice to the protections afforded to the Insured by this endorsement, no one Insured represents or warrants the adequacy or accuracy of any information provided or representation made by or on behalf of any other Insured.

- 248 -

#### Section III: Rights to avoid / cancel or change Policy terms

#### 6. Non-vitiation

- 6.1 Paragraph (i) It is noted and agreed that if the insured described in the schedule comprises more than one insured party each operating as a separate and distinct entity then (save as provided in this multiple insured's clause) cover hereunder shall apply in the same manner and to the same extent as if the individual polices had been issued to each such insured party provided that the total liability of the insures to all of the insured parties collectively shall not exceed the sums insured and limits of indemnity including and inner limits set by memorandum or endorsement stated in the policy.
- 6.2 Paragraph (ii) It is understood and agreed that any payment or payments by Insurers to any one or more such insured parties shall reduce to the extent of that payment insurers liability to all such parties arising from any one event giving rise to a claim under this policy and (if applicable) in the aggregate.
- 6.3 Paragraph (iii) It is further understood that the insured parties will at all times preserve the various contractual rights and agreements entered into by the insured parties and the contractual remedies of such parties in the event of loss or damage.
- 6.4 Paragraph (iv) It is further understood and agreed that insurers shall be entitled to avoid liability to or (as maybe appropriate) claim damages from any of the insured parties in circumstances of fraud, material misrepresentation, material non-disclosure or breach of any warranty or condition of this policy each referred to in this clause as a vitiating act.
- 6.5 Paragraph (v) It is however agreed that (save as provided in this multiple insured's clause) a vitiating act committed by one insured party shall not prejudice the right to indemnity of any other insured party who has an insurable interest and who has not committed a vitiating act.
- 6.6 Paragraph (vi) Insurers hereby agree to waive all rights of subrogation which they may have or acquire against any insured party except where the rights of subrogation or recourse are acquired in consequence of or otherwise following a vitiating act in which circumstances insurers may enforce such rights notwithstanding the continuing or former status of the vitiation party as insured.
- 7. Cancellation
- 7.1 The Reinsurers agree that they shall not seek to cancel or suspend the this insurance except: (i) for non payment of premium; or (ii) where an Insured consistently fails to comply with Reinsurers' requirements relating to survey or loss control action points.
- 7.2 The Reinsurers shall promptly notify the Intercreditor Agent in writing in the event of any suspension, cancellation, termination of this Policy by the Reinsurers or by the Reinsured or Insured. The cover provided by this Policy shall continue in force and unaltered for at least 30 days after written notice of such suspension, cancellation, termination or non-renewal is given to the Intercreditor Agent. Nothing in this clause shall give the Reinsurers any right to suspend, cancel or terminate this Policy which the Reinsurers do not otherwise have under this Policy.

- 249 -

- 7.3 The Reinsurers shall promptly notify the Intercreditor Agent in writing of any default in the payment of premium and shall give the Intercreditor Agent at least 30 days notice in writing before voiding this Policy for non-payment of premium, in order to give an opportunity for that premium to be paid within the notice period.
- 8. Changes in cover

The Reinsurers shall give the Intercreditor Agent at least 30 days notice in writing before any reduction in cover or increase in excess or deductible under this Policy takes effect. Nothing in this clause shall give the Reinsurers any right which they do not otherwise have to reduce cover or increase any excess or deductible under this Policy.

9. Amendments to Endorsement

During the term of this Policy, the provisions of this endorsement may only be amended by written agreement between the Company, the Reinsurers and the Intercreditor Agent, such amendment to be endorsed on the Policy.

# Section IV: Claims

10. Notice of claims

Notice of claim by the Secured Parties or any other party entitled to indemnity under the Original Policy shall, in the absence of manifest error, be accepted by Reinsurers as a valid notification of claim on behalf of all other Reinsureds and Insureds subject to the full terms of the Policy.

11. Claim Payments / Loss Payee

Payments made in accordance with this Clause 11 shall, to the extent of the payment, discharge the Reinsurers' liability to pay the Reinsured or any other Insured.

11.1 In respect of the insurance under this Policy of material damage risks only

All claim payments or return premium shall be paid to such account as the Security Agent as loss payee may specify in writing.

11.2 In respect of the insurance under this Policy of public liability risks only

All claim payments in respect of a third party liability shall be paid to person(s) whose claim(s) constitute the risk or liability insured against except in the case where the Reinsured or Insured has properly discharged its liability to such person(s), in which case the claim payment shall be paid to such account as the Security Agent directs in writing.

Any return premiums shall be paid to such account as the Security Agent directs in writing.

- 250 -

11.3 In respect of the insurance under this Policy of loss of revenue risks only

All claim payments or return premiums shall be paid to such account as the Security Agent directs in writing.

11.4 Set-off

The Reinsurers may, at their discretion, deduct overdue unpaid premium from claims settlements but shall not set off or deduct premium that is not overdue or any other amounts payable by the Reinsured under or in relation to the Policy.

#### 12. Waiver of subrogation

The Reinsurers waive all rights of subrogation howsoever arising which they may have or acquire against any Reinsured or Insured described within the appropriate Schedules arising out of any Occurrence in respect of which any claim is admitted and is insured hereunder for the benefit of such Reinsured or Insured except against any:

- (i) such Reinsured or Insured (or officer, director, employee, agent or assign) who has caused or contributed to such an occurrence or claim by fraud, deliberate misrepresentation, deliberate non-disclosure or deliberate breach of policy condition; or
- (ii) consultant or equivalent professional party to the extent that their professional errors, omissions or activities not covered by this Policy have caused or contributed to a loss covered under this Policy; or
- (iii) supplier or manufacturer to the extent that their errors, omissions or activities not covered by this Policy have caused or contributed to a loss covered under this Policy; or
- (iv) such Insured to the extent that they are entitled to recover in respect of a loss under cover falling within sub-clause 13(a)–(e) below (or would be so insured if cover in the terms set out in this Policy had not been taken out);

#### 13. Primary insurance

The Reinsurers agree that the Original Policy provides the primary cover for risks insured under the Original Policy. In the event that any risk insured under the Original Policy is also insured under any other policy of insurance effected by any Reinsured or Insured, the Reinsurers agree to indemnify the Reinsured or Insured as if such other policy of insurance did not exist except in respect of:

- (a) excess layers of third party cover effected specifically for the Projects;
- (b) any public liability claim against the Insured which exceeds the applicable limit of indemnity under this Policy, in which case the liability of the Reinsurers for additional legal costs and expenses shall be limited to the proportion that the applicable limit of indemnity bears to the total claim against the Reinsured or Insured;
- (c) any claim under this Policy to which a Marine 50/50 Clause applies;

- 251 -

(d) any claim made under a Contingent Motor Liability extension to this Policy; or

- any claim relating to a loss which is insured against (or would be insured but for a double insurance provision or similar or the application of a deductible) under:
  - (i) any other policy specifically effected for the construction or operational phase(s) of the Projects; or
  - (ii) a latent or inherent defects policy or engineering or mechanical breakdown policy specifically effected for the Projects;

or a related business interruption insurance policy.

#### Section V: Miscellaneous

(e)

14. Notice of Security Interest

The Reinsurers acknowledge that by an assignment contained in the Assignment of Reinsurances dated [ ] between the Reinsured and the Security Agent (as amended, consolidated, supplemented, novated or replaced from time to time) (the "Assignment"), Reinsured assigned by way of security to the Secured Parties all benefits and rights in respect of this insurance and all claims and returns of premiums in respect thereof to which the Reinsured or Insured is or may at a future time become entitled. The Reinsurers confirm that they have not been notified of any other assignment of or security interest in the Reinsured's interest in this insurance.

#### 15. Insurer access to Due Diligence Reports

The Senior Secured Creditors shall have no duty of disclosure to the Reinsurers in relation to any Reinsurance. Nevertheless, on the written reasonable request of the Reinsurers, the Senior Secured Creditors shall, subject to the Company obtaining any other consents required for such access (which consents the Company agrees it shall use its reasonable endeavours to obtain), provide the Reinsurers with reasonable access to any relevant due diligence report(s) commissioned by the Senior Secured Creditors relating to the Projects. The Company consents to such access and the Reinsurers shall keep such report(s) confidential and shall accept such information without rights of recourse against any of the Senior Secured Creditors, the Company or the party/parties that prepared the said reports.

# 16. Notice

16.1 All notices or other communications under or in connection with the Policy will be given via the reinsurance broker by fax and post. Any such notice given will be deemed to be given on the earlier of:

- (a) if by fax, when transmitted but only if the sender's fax machine confirms successful transmission; and
- (b) if by post, within 2 business days of release from the relevant Insurer's office.

- 252 -

16.2 The address and fax number of the Intercreditor Agent and the Security Agent for all notices under or in connection with the Policy are those notified from time to time by the Intercreditor Agent or the Security Agent for this purpose to the insurance broker at the relevant time. The initial address and fax number of the Intercreditor Agent and the Security Agent are as follows:

Intercreditor Agent: [—]

Address: [—] Tel: [—] Fax: [—] Attention: [—] Security Agent: [—] Address: [—] Tel: [—] Fax: [—] Attention: [—] Governing law & Jurisdiction

16.3

The Policy shall be governed and interpreted in accordance with English law.

This endorsement overrides any conflicting provision in this Policy.

- 253 -

#### **APPENDIX 5**

#### FORM OF INSURANCE BROKER'S LETTER OF UNDERTAKING

# NOT APPLICABLE AFTER THE FOURTH AMENDMENT EFFECTIVE DATE

[Date]

To: [ ] as Intercreditor Agent

Dear Sirs,

1.

- We refer to Schedule 7 (*Insurance*) (the "**Insurance Schedule**") to the Common Terms Agreement dated 14 September 2004 between Wynn Resorts (Macau) S.A. (the "**Company**") and the financial institutions referred to therein as Senior Secured Creditors, as amended, consolidated, supplemented, novated or replaced from time to time. Terms used herein shall bear the same meaning as in the Insurance Schedule. Any reference herein to a document being in substantially a specified form shall be construed as meaning such document being in the same form as the specified form save for the insertion of information left in blank or typographical errors.
- 2. We, in our capacity as insurance brokers to the Company, confirm that each Direct Insurance as required pursuant to Clause 1.1 of the Insurance Schedule is in full force and effect as of the date of this letter on and in respect of the risks set out in the Direct Insurances evidenced in the policy cover notes attached hereto as Annex A and that all premiums which are required to have been paid at the date hereof in respect of such Direct Insurances have been paid in full.
- 3. We confirm that each Direct Insurance referred to in paragraph 2 above contains endorsements in substantially the form set out in Appendix 3 (*Form of Endorsements for Direct Insurances*) to the Insurance Schedule.
- 4. Pursuant to instructions received from the Company, we hereby undertake in respect of the Direct Insurances referred to in the attached cover notes:
  - (a) in the case of any such Direct Insurance, as and when the same is renewed, extended or replaced, and subject to market conditions current at the time of application for such renewal, extension or replacement, to use our best efforts to ensure that it complies with the requirements of the Insurance Schedule or such other requirements as you may approve in writing and that it contains endorsements in substantially the form set out in Appendix 3 (*Form of Endorsements for Direct Insurances*) to the Insurance Schedule or in such other form as you may approve in writing;
  - (b) to pay to the accounts specified in the relevant loss payable clauses in the relevant policy documents without any set-off or deduction of any kind for any reason any and all proceeds from or other payments made pursuant to the Direct Insurances (including refunds of premiums) received by us from the insurers except as might otherwise be permitted in the relevant policy endorsement provisions or required by law or court order or as you may otherwise agree in writing;

- 254 -

to advise you:

(i) as soon as practical upon our becoming aware of any actual or proposed:

- (A) cancellation or suspension of cover under any Direct Insurance;
- (B) reduction in limits or coverage or any increase in deductibles; and
- (C) termination prior to the original expiry date of any Direct Insurance;
- (ii) as soon as practical of any default in the payment of any premium for any of the Direct Insurances;
- (iii) at least 45 days prior to the expiry of any Direct Insurance if we have not received instructions to negotiate the renewal thereof from the Company and/or any jointly insured parties or the agents of any such party and at least 5 Business Days prior to the expiry thereof if such Direct Insurance has not been renewed;
- (iv) if we receive instructions to negotiate the renewal of any Direct Insurance, the details of such instructions and, upon the renewal of such Direct Insurance, the terms of such renewal; and
- (v) of any act or omission of any event of which we have knowledge and which might invalidate or render unenforceable in whole or in part any of the Direct Insurances promptly upon our becoming aware of the same;
- (d) to disclose to you any fact, change of circumstance or occurrence material to the risks insured against under the Direct Insurances or which would result in any reduction in limits or alteration in coverage or increase in deductions or exclusions promptly upon our becoming aware of the same;
- (e) to notify you promptly following our becoming aware that we shall cease to act as insurance broker to the Company; and
- (f) on your reasonable request and at your expense and subject to any legal, contractual or regulatory restrictions, to make those documents contained within our placing and claims files to which the Company would be entitled to have access available to you or your Insurance Adviser at reasonable times and places, and to provide you with copies of any such documents.

# The above undertakings are given subject to our continuing appointment as insurance brokers to the Company and in relation to the Direct Insurances and the handling of claims in relation to them.

- The contents of this letter may not be disclosed to any other party other than to any person:
  - (a) to (or through) whom a Secured Party assigns or transfers (or may potentially assign or transfer) all or any of its rights, benefits and/or obligations under the Senior Finance Documents;

- 255 -

5.

6.

- (b) to (or through) whom a Secured Party enters into (or may potentially enter into) any sub-participation in relation to, or any other transaction under which payments are to be made by reference to, any of the Senior Finance Documents or any Obligor; or
- (c) to whom all or any contents of this letter may be required to be disclosed by any applicable law or pursuant to any regulatory or stock exchange requirement

(a "**Third Party**") and, in the event that it is disclosed to a Third Party, any and all liability howsoever arising to such Third Party is hereby expressly excluded. No person except the Senior Secured Creditors has any rights arising out of this letter under the Contracts (Rights of Third Parties) Act 1999.

- 7. Our aggregate liability to you for any and all matters arising from this letter and the contents thereof shall in any and all events be limited to the sum of USD1,000,000 and confined to direct losses in contract. Any and all other liability including but not limited to liability in tort and liability for consequential losses is hereby expressly excluded. Notwithstanding the foregoing, nothing in this letter shall serve to limit our liability for death or personal injury caused by our negligence.
- 8. This letter shall be governed and construed in accordance with the laws of England and any disputes arising in connection with this letter shall be finally settled under the Rules of Arbitration of the International Chamber of Commerce by one arbitrator appointed in accordance with the said Rules. The arbitration shall take place in Hong Kong and shall be conducted in English. The arbitration award shall be binding upon both parties.

Yours faithfully,

[name of insurance broker]

- 256 -

#### **APPENDIX 6**

#### FORM OF REINSURANCE BROKER'S LETTER OF UNDERTAKING

# NOT APPLICABLE AFTER THE FOURTH AMENDMENT EFFECTIVE DATE

| To: | [ | ] as Intercreditor Agent |
|-----|---|--------------------------|
|     | [ | ] as Security Agent      |

[Date]

#### Dear Sirs,

4.

- 1. We refer to Schedule 7 (*Insurance*) (the "**Insurance Schedule**") to the Common Terms Agreement dated 14 September 2004 between Wynn Resorts (Macau) S.A. (the "**Company**") and the financial institutions referred to therein as Senior Secured Creditors, as amended, consolidated, supplemented, novated or replaced from time to time. Terms used herein shall bear the same meaning as in the Insurance Schedule. Any reference herein to a document being in substantially a specified form shall be construed as meaning such document being in the same form as the specified form save for the insertion of information left in blank or typographical errors.
- 2. We, in our capacity as reinsurance brokers to [ ] (the "**Direct Insurer**"), confirm that facultative reinsurance of each Direct Insurance (other than those indicated by you) as required pursuant to Clause 1.2 of the Insurance Schedule are in full force and effect as of the date of this letter on and in respect of the risks set out in the Reinsurances evidenced in the policy cover notes attached hereto as Annex A and that all premiums which are required to have been paid at the date hereof in respect of such Reinsurances have been paid in full and such Reinsurances are placed with reinsurers and underwriters whose identities we have disclosed to you and whom we in good faith believe to be reputable and financially sound.
- 3. We confirm that each Reinsurance referred to in paragraph 2 above contains endorsements in substantially the form set out in Appendix 4 (*Form of Endorsements for Reinsurances*) to the Insurance Schedule.
  - Pursuant to instructions received from the Direct Insurer, we hereby undertake in respect of the Reinsurances referred to in the attached cover notes:
    - (a) in the case of any such Reinsurance, as and when the same is renewed, extended or replaced, and subject to market conditions current at the time of application for such renewal, extension or replacement, to use our best efforts to ensure that it complies with the requirements of the Insurance Schedule or such other requirements as you may approve in writing and that it contains endorsements in substantially the form set out in Appendix 4 (*Form of Endorsements for Reinsurances*) to the Insurance Schedule or in such other form as you may approve in writing;
    - (b) to pay to the accounts specified in the relevant loss payable clauses in the relevant policy documents without any set-off or deduction of any kind for any reason any and all proceeds from or other payments made pursuant to the

- 257 -

Reinsurances (including refunds of premiums) received by us from the reinsurers except as might otherwise be permitted in the relevant policy endorsement provisions or required by law or court order or as you may otherwise agree in writing;

(c)

5.

to advise you:

- (i) as soon as practical upon our becoming aware of any actual or proposed:
  - (A) cancellation or suspension of cover under any Reinsurance;
  - (B) reduction in limits or coverage or any increase in deductibles; and
  - (C) termination prior to the original expiry date of any Reinsurance;
- (ii) as soon as practical of any default in the payment of any premium for any of the Reinsurances;
- (iii) at least 45 days prior to the expiry of any Reinsurance if we have not received instructions to negotiate the renewal thereof from the Direct Insurer and/or any jointly insured parties or the agents of any such party and at least 5 Business Days prior to the expiry thereof if such Reinsurance has not been renewed;
- (iv) if we receive instructions to negotiate the renewal of any Reinsurance, the details of such instructions and, upon the renewal of such Reinsurance, the terms of such renewal; and
- (v) of any act or omission or any event of which we have knowledge and which might invalidate or render unenforceable in whole or in part any of the Reinsurances promptly upon our becoming aware of the same;
- (d) to disclose to you any fact, change of circumstance or occurrence material to the risks insured against under the Reinsurances or which would result in any reduction in limits or alteration in coverage or increase in deductions or exclusions promptly upon our becoming aware of the same;
- (e) to notify you promptly following our becoming aware that we shall cease to act as reinsurance broker to the Direct Insurer; and
- (f) on your reasonable request and at your expense and subject to any legal, contractual or regulatory restrictions, to make those documents contained within our placing and claims files to which the Direct Insurer would be entitled to have access available to you or your Insurance Adviser at reasonable times and places, and to provide you with copies of any such documents.
- The above undertakings are given subject to our continuing appointment as reinsurance brokers to the Direct Insurer and in relation to the Reinsurances and the handling of claims in relation to them.

The contents of this letter may not be disclosed to any other party other than to any person:

- (a) to (or through) whom a Secured Party assigns or transfers (or may potentially assign or transfer) all or any of its rights, benefits and/or obligations under the Senior Finance Documents;
- (b) to (or through) whom a Secured Party enters into (or may potentially enter into) any sub-participation in relation to, or any other transaction under which payments are to be made by reference to, the Senior Finance Documents or any Obligor; or
- (c) to whom all or any contents of this letter may be required to be disclosed by any applicable law or pursuant to any regulatory or stock exchange requirement

(a "**Third Party**") and, in the event that it is disclosed to a Third Party, any and all liability howsoever arising to such Third Party is hereby expressly excluded. No person except the Senior Secured Creditors has any rights arising out of this letter under the Contracts (Rights of Third Parties) Act 1999.

- 7. Our aggregate liability to you for any and all matters arising from this letter and the contents thereof shall in any and all events be limited to the sum of USD1,000,000 and confined to direct losses in contract. Any and all other liability including but not limited to liability in tort and liability for consequential losses is hereby expressly excluded. Notwithstanding the foregoing, nothing in this letter shall serve to limit our liability for death or personal injury caused by our negligence.
- 8. This letter shall be governed and construed in accordance with the laws of England and any disputes arising in connection with this letter shall be finally settled under the Rules of Arbitration of the International Chamber of Commerce by one arbitrator appointed in accordance with the said Rules. The arbitration shall take place in Hong Kong and shall be conducted in English. The arbitration award shall be binding upon both parties.

Yours faithfully,

6.

[name of reinsurance broker]

- 259 -

#### SCHEDULE 8 HEDGING ARRANGEMENTS

- The Company shall, no later than 60 days after the Fourth Amendment Effective Date, enter into agreements to the extent necessary to ensure that at least 50% of the aggregate amount drawn under the Term Loan Facilities (including the amount of such Advance) is subject, through interest rate swaps, caps, collars or other products agreed with the Intercreditor Agent, to either a fixed interest rate or interest rate protection for such period as reflects the repayment schedule for such Facility and with a final maturity date of not less than three years from the date upon which such Hedging Arrangement was first entered into in respect of such amount.
- 2. The purchase price of any such products may be paid for:

1.

- (a) by one or other of the other Obligors for the benefit of the Company (*provided* that such payment shall either constitute Equity (other than any Equity required to be paid up or advanced in accordance with the terms of the Senior Finance Documents) or otherwise create no recourse to the Company); or
- (b) out of the amount provided in the "Interest" Line Item (*provided* that the sum of the purchase prices, interest and any other amounts payable by the Company in respect of all such products does not exceed such amount).
- 3. Only a Lender or an Affiliate of a Lender may act as a Hedging Counterparty in respect of the Hedging Arrangements required by paragraph 1 above.
- 4. The Hedging Agreements are to be on the terms of the 1992 standard International Swap & Dealers Association, Inc. Master Agreement (the "ISDA Master Agreement") and the ISDA Schedule, together with such amendments as are acceptable to the Intercreditor Agent. All Hedging Agreements for swap transactions will provide for full two way payments (with the Company being a Fixed Rate Payer (as defined in the 2000 ISDA Definitions as published by the International Swaps and Derivatives Association, Inc. (the "2000 Definitions")) and the Hedging Counterparty being a Floating Rate Payer (as defined in the 2000 Definitions)) and the payment measure and payment method for such swap transactions in the event of early termination, whether upon a "Termination Event" or an "Event of Default", shall be "Second Method" and "Market Quotation" respectively. Terms in quotations in this paragraph 4 shall have the meaning ascribed in the ISDA Master Agreement.
- 5. The Hedging Counterparties shall have equal security over the assets of the Company with the Senior Secured Creditors in accordance with the terms of this Agreement and the Deed of Appointment and Priority.
- 6. Any payments due from the Company under the Hedging Agreements (except for the purchase price or other amounts in respect of any interest rate caps, collars or other products paid by another Obligor) prior to the Expansion Opening Date, including any Realised Hedge Loss plus any accrued default interest in accordance with paragraph 10 below, shall be a permitted Project Cost.

In this paragraph and paragraphs 7 and 10 below, "**Realised Hedge Loss**" means, in relation to a Hedging Counterparty at any time, the amount (if any) payable (but

- 260 -

unpaid) by the Company to such Hedging Counterparty under the Hedging Agreement to which such Hedging Counterparty is a party (but excluding any default interest) upon an early termination of any transaction or transactions thereunder which has been terminated in accordance with paragraph 9 below. The amount is to be calculated on a net basis across the transactions under such Hedging Agreement in accordance with the terms of the applicable Hedging Agreement.

- Payments due from the Company under the Hedging Agreements following the Expansion Opening Date, including any Realised Hedge Loss plus any accrued default interest in accordance with paragraph 10 below, shall (save for any such amounts paid by another Obligor) be a Financing Cost.
- 8. Except with the prior consent of the Intercreditor Agent, no amendments may be made to a Hedging Agreement to an extent that might reasonably be expected to result in:
  - (a) any payment under the Hedging Agreement being required to be made by the Company on any date other than the dates originally provided for in the Hedging Agreement;
  - (b) the Company becoming liable to make an additional payment under any Hedging Agreement which liability does not arise from the original provisions of the Hedging Agreement; or
  - (c) the Company becoming liable to make any payment under the Hedging Agreement in any currency other than in the currency provided for under the original provisions of the Hedging Agreement.

9.

- (a) The Company may terminate a transaction under a Hedging Agreement prior to its stated maturity only in circumstances provided for in such Hedging Agreement and with the approval of the Intercreditor Agent *provided* that the approval of the Intercreditor Agent shall not be required in the case of any termination by reason of illegality when the requirements of paragraph 1 above are met following such termination.
  - (b) A Hedging Counterparty may terminate a transaction under a Hedging Agreement prior to its stated maturity only in circumstances provided for in such Hedging Agreement.
  - (c) Unless a Hedging Counterparty has already exercised such rights in accordance with sub-paragraph (b) above, the Intercreditor Agent may require a Hedging Counterparty to terminate transactions under a Hedging Agreement where a declaration has been made by the Intercreditor Agent pursuant to Clause 19.2.2 (*Remedies following an Event of Default*).
  - (d) If a voluntary or mandatory prepayment is to be made in accordance with Clause 8 (*Repayments, Prepayments and Cancellation*) and following such prepayment the aggregate amount of the "Notional Amounts" (as defined in the 2000 Definitions) of all Hedging Agreements at such time would be greater than 125% of the principal amounts outstanding under the Term Loan Facilities, the Company shall unwind, in order of maturity immediately

- 261 -

following such prepayment (unless otherwise agreed by the Intercreditor Agent), sufficient transactions under the Hedging Agreements (and pay associated breakage costs) on the first Payment Date (as defined in the 2000 Definitions) (or, where such prepayment falls within 5 Business Days (as defined in the relevant Hedging Agreement) prior to such first Payment Date, the second Payment Date) in respect of such transaction immediately succeeding such prepayment such that the Intercreditor Agent is satisfied that, following such terminations, the aggregate Notional Amounts of all transactions under all Hedging Agreements is not less than 50% and not more than 125% of the principal amounts outstanding under the Facilities.

10. In the event that a Hedging Agreement is terminated and the Company fails to pay any Realised Hedge Loss, such Realised Hedge Loss shall comprise an Unpaid Sum and interest shall accrue in respect thereof accordingly.

#### **APPENDIX 1**

# FORM OF HEDGING COUNTERPARTY'S DEED OF ACCESSION

THIS DEED dated [ ] is supplemental to (i) a common terms agreement (the "**Common Terms Agreement**") dated 14 September 2004 between Wynn Resorts (Macau) S.A. as Company and the Senior Secured Creditors (as defined therein) and (ii) each of the Security Documents as defined in the Common Terms Agreement to which the Senior Secured Creditors are expressed to be party (the "**Security Documents**").

Words and expressions defined in the Common Terms Agreement have the same meaning when used in this Deed and the principles of construction and rules of interpretation set out therein shall also apply.

[*name of new Hedging Counterparty*] (the "**New Hedging Counterparty**") of [*address*] hereby agrees with each other person who is or who becomes a party to the Common Terms Agreement that with effect on and from the date of this Deed it shall be bound by the Common Terms Agreement and be entitled to exercise rights and be subject to obligations thereunder as a Hedging Counterparty.

The New Hedging Counterparty hereto agrees with each other person who is or who becomes a party to the Security Documents that with effect on and from the date of this Deed it shall be bound by each of the Security Documents and be entitled to exercise rights and be subject to obligations thereunder as a Senior Secured Creditor.

The initial telephone number, fax number, address and person designated by the New Hedging Counterparty for the purposes of Clause 29 (*Notices*) of the Common Terms Agreement are:

[

1

This Deed is governed by and shall be construed in accordance with English law.

| Executed as a deed by       | ) |
|-----------------------------|---|
| [insert name of New Hedging | ) |
| Counterparty and execution  | ) |
| clause appropriate thereto  | ) |
| and to manner of execution] | ) |
|                             |   |

- 263 -

#### SCHEDULE 9 MANDATORY PREPAYMENT

- 1. To the extent that any Equity Issuance Proceeds are received by (or paid to the order of) the Company or any other Obligor, the Company shall ensure that an amount equal to the amount of such Equity Issuance Proceeds (excluding any such Equity Issuance Proceeds derived from any Equity to the extent such Equity is required or permitted under this Agreement to fund any Project Costs or the design, development, construction, ownership, operation or maintenance of the Cotai Project) shall be applied on the next Interest Payment Date after such receipt (or payment) towards prepayment and cancellation of the Facilities in accordance with Clause 8.3 and, pending such application, shall be deposited and retained in an Account.
- 2. With respect to the Net Cash Proceeds from any Asset Sale (where such Net Cash Proceeds exceed in aggregate an amount equal to USD100,000,000 or its equivalent in any Fiscal Year when taken together with any other Net Cash Proceeds received by the Company or any other Obligor in respect of any Asset Sale in that Fiscal Year) made by the Company or any other Obligor as to which the Company or any other Obligor has not re-invested such Net Cash Proceeds within 12 months of receipt in assets used by the Company or any other Obligor in the conduct of its Permitted Business, such Net Cash Proceeds shall be applied on the next Interest Payment Date after the date falling 12 months from the date of receipt towards prepayment and cancellation of the Facilities in accordance with Clause 8.3 and, pending such application, shall be deposited and retained in an Account.
- 3. On the next Interest Payment Date following the date on which the Company or any other Obligor receives any Termination Proceeds or Eminent Domain Proceeds, the Company shall ensure that such proceeds are applied towards prepayment and cancellation of the Facilities in accordance with Clause 8.3 and, pending such application, shall be deposited and retained in an Account.
  - If, for any Fiscal Year commencing with the Fiscal Year in which the Original Project Opening Date occurs, there shall be Excess Cash Flow, the Company shall apply the ECF Percentage of such Excess Cash Flow towards prepayment and cancellation of the Facilities in accordance with Clause 8.3. Each such prepayment shall be made on the next Interest Payment Date following the earlier of:
    - (a) the date on which the financial statements of the Company referred to in paragraph 1 of Part A of Schedule 5 (*Covenants*), for the Fiscal Year with respect to which such prepayment is made, are required to be delivered to the Lenders; and
    - (b) the date such financial statements are actually delivered,

4.

5.

and, pending such application, shall be deposited and retained in an Account.

An amount equal to any Insurance Proceeds (other than those received by the Company or any other Obligor for any single loss or series of related losses not in excess of USD50,000,000 or its equivalent) shall be applied to the prepayment and cancellation of the Facilities in accordance with Clause 8.3 on the next Interest Payment Date falling not less than 30 days after the Company's or Obligor's receipt of such Insurance Proceeds, unless each of the following conditions are satisfied or waived by the Intercreditor Agent within such 30 day period, in which event such amounts shall be applied to the repair or restoration of the Projects or the Cotai Project, as appropriate:

(a) the damage or destruction does not constitute the destruction of all or substantially all of the Projects or the Cotai Project, as the case may be;

- (b) a Default has not occurred and is continuing (other than a Default resulting solely from such damage or destruction) and after giving effect to any proposed repair and restoration, no Default will result from such damage or destruction or proposed repair and restoration;
- (c) the Company certifies that repair or restoration of the Projects or the Cotai Project, as the case may be, to a condition substantially similar to their condition immediately prior to the event or events to which the relevant Insurance Proceeds relate, is technically and economically feasible within a 9 month period (where the damage or destruction occurs prior to the Diamond Completion Date) or a 12 month period (where it occurs thereafter) and that a sufficient amount of funds is or will be available to the Company to make such repairs and restorations;
- (d) the Company delivers to the Intercreditor Agent a plan (the "**Repair Plan**") describing in reasonable detail the nature of the repairs or restoration to be effected and the anticipated costs and schedule associated therewith, in form and substance reasonably satisfactory to the Intercreditor Agent;
- (e) the Company certifies that (A) the Company has sufficient Funds to achieve construction completion of the Projects or the Cotai Project, as the case may be, to operate its business as it is then conducted and pay its debts when due and (B) a sufficient amount of funds is or will be available to the Company to make all payments on Financial Indebtedness which will become due during and following the repair period and, in any event, to maintain compliance with the covenants set forth in paragraph 1 of Part B of Schedule 5 (*Covenants*) during such repair period;
- (f) no Permit is necessary to proceed with the repair and restoration of the Projects and no material amendment to the Project Documents, or, except with the consent of the Senior Secured Creditors, any of the Senior Finance Documents, and no other instrument is necessary for the purpose of effecting the repairs or restoration of the Projects or subjecting the repairs or restoration to the Liens of the applicable Security Documents and maintaining the priority of such Liens or, if any of the above is necessary, the Company will be able to obtain the same as and when required; and
- (g) the Intercreditor Agent shall receive such certificates, opinions or other matters as it may reasonably request as necessary or appropriate in connection with such repairs or restoration of the Projects or the Cotai Project, as the case may be, or to preserve or protect the Senior Secured Creditors' interests hereunder and in the applicable Project Security,

and, pending such application, such amount shall be deposited and retained in an Account.

- 265 -

After Insurance Proceeds have been applied to the repair or restoration of the Projects or the Cotai Project, as the case may be, as provided in this Agreement, any excess Insurance Proceeds shall be applied on the next Interest Payment Date thereafter to the prepayment and cancellation of the Facilities in accordance with Clause 8.3.

6. [Not used.]

If all or substantially all of either Project is lost, damaged or destroyed or declared by any relevant Insurer to be a constructive total loss, the
 Company shall prepay and cancel the Facilities in accordance with Clause 8.3 and prepay all other amounts outstanding under the Senior Finance
 Documents within 90 days or, if earlier, upon receipt of Insurance Proceeds in respect of such loss, damage, destruction or declaration.

8. On the next Interest Payment Date following the date on which the Company or any other Obligor receives any Claim Proceeds in relation to a Project Document (other than a Resort Management Agreement that is not a Cotai Resort Management Agreement), the Company shall apply an amount equal to such proceeds towards prepayment and cancellation of the Facilities in accordance with Clause 8.3 and, pending such application, shall deposit and retain such proceeds in an Account.

# SCHEDULE 10 EVENTS OF DEFAULT

- (i) The Company shall have failed to pay any principal of any Advance when due in accordance with the terms of the relevant Facility Agreement; or
  - (ii) the Company shall have failed to pay any interest on any Advance within 5 days after any such interest becomes due in accordance with the terms of the relevant Senior Finance Document; or
  - (iii) any other Obligor or a Wynn Non-Obligor Subordination Deed Party shall have failed to pay any other amount payable under any Senior Finance Document within 10 days after any such other amount becomes due in accordance with the terms of the relevant Senior Finance Document or in the event that any such other amount becomes due without a notice being given to the relevant Obligor, 10 days after notice to the relevant Obligor of the non-payment of such amount.
- (b) Any representation or warranty made or deemed made by any Obligor or a Wynn Non-Obligor Subordination Deed Party in any Senior Finance Document to which it is a party or that is contained in any certificate, document or financial or other statement furnished by it at any time under or in connection with any Senior Finance Document shall prove to have been incorrect in any material respect (or, in the case of the representations and warranties set out in paragraphs 21.1, 21.2 and 21.3 of Schedule 4 (*Representations and Warranties*), in any respect) on or as of the date made or deemed made and, in the case of an Obligor (other than the Company or a Wynn Obligor) or a Wynn Non-Obligor Subordination Deed Party, such event, could reasonably be expected to give rise to a Material Adverse Effect.
- (c) The Company shall default in the observance or performance of:

(a)

- (i) paragraph 12(a) of Part A of Schedule 5 (*Covenants*) and, other than in respect of a Default arising by reason of paragraphs (v), (w), (x), (y), (z) or (aa) of Schedule 10 (*Events of Default*), such default shall continue unremedied for a period of 30 days;
- (ii) paragraph 12(b) of Part A of Schedule 5 (*Covenants*) in so far as it relates to the Concession Contract or the Land Concession Contract;
- (iii) paragraph 24 of Part A of Schedule 5 (*Covenants*);
- (iv) paragraph 1 of Part B of Schedule 5 (Covenants) and such default shall continue unremedied for a period of 5 days; or
- (v) clause 10.3 (Conditions Subsequent) of the Common Terms Agreement Fourth Amendment Agreement.
- (d) Any Obligor or the Performance Bond Provider (or a Wynn Non-Obligor Subordination Deed Party) shall default in the observance or performance of any other covenant or agreement contained in any Senior Finance Document to which it is a party (other than as provided in paragraphs (a) through (c) of this Schedule), and such default shall continue unremedied for a period of 30 days or, *provided* the relevant Obligor or the Performance Bond Provider (or Wynn Non-Obligor Subordination Deed Party) is diligently pursuing action to remedy the default and it is of a nature that it is capable of being remedied, 60 days after the earlier of:
  - (i) the Company or such Obligor (or such Wynn Non-Obligor Subordination Deed Party) becoming aware of such default; and
  - (ii) receipt by the Company of notice from the Intercreditor Agent or any Lender of such default.

- 267 -

- (e) Save where paragraph (a) of this Schedule applies, the Company or any other Obligor (or, in relation to any Financial Indebtedness that is Guaranteed WML Debt, WML or any Obligor that has incurred, assumed or suffered to exist any Guarantee Obligation with respect to such Guaranteed WML Debt) shall:
  - default in making any payment of any principal of any Financial Indebtedness (including, without limitation, any Guarantee Obligation, but excluding the Advances) on the scheduled due date with respect thereto;
  - (ii) default in making any payment of any interest on any such Financial Indebtedness beyond the period of grace, if any, provided in the instrument or agreement under which such Financial Indebtedness was created; or
  - default in the observance or performance of any other agreement or condition relating to any such Financial Indebtedness or contained in any instrument or agreement evidencing, securing or relating thereto, or any other event shall occur or condition exist, the effect of which default or other event or condition (or any declaration by the holder of such Financial Indebtedness by reason thereof) is to cause immediately such Financial Indebtedness to become due prior to its stated maturity or (in the case of any such Financial Indebtedness constituting a Guarantee Obligation) to become payable,

provided that a default event or condition described in sub-paragraphs (i), (ii) or (iii) of this paragraph (e) shall not at any time constitute a Default or Event of Default unless, at such time, the aggregate amount of the default in the principal payment in the case of sub-paragraph (i), the default in the interest payment in the case of sub-paragraph (ii) and the amount accelerated in the case of sub-paragraph (iii) of this paragraph (e) exceeds USD25,000,000 or its equivalent in the case of the Company and the other Obligors taken as a whole (or, where such default event or condition relates to Financial Indebtedness that is Guaranteed WML Debt, WML and any Obligors that have incurred, assumed or suffered to exist any Guarantee Obligation with respect to such Guaranteed WML Debt taken as a whole).

Any Obligor or the Performance Bond Provider shall commence any case, proceeding or other action (A) under any existing or future law of any jurisdiction, domestic or foreign, relating to bankruptcy, insolvency, reorganization or relief of debtors, (x) seeking to have an order for relief entered with respect to it, or seeking to adjudicate it a bankrupt or insolvent, or (y) seeking reorganization, administration, arrangement, adjustment, winding-up, liquidation, dissolution, composition or other relief with respect to it or its debts, or (B) seeking appointment of a receiver, trustee, custodian, conservator

(f)

or other similar official for it or for all or any substantial part of its assets, or any Obligor or the Performance Bond Provider shall make a general assignment for the benefit of its creditors, but excluding in the case of paragraph (A)(y) any voluntary liquidation, winding-up or dissolution of, or similar action with respect to, Palo after a Permitted Cotai Reorganisation;

- there shall be commenced against any Obligor or the Performance Bond Provider any case, proceeding or other action of a nature referred to in sub-paragraph (i) above that (A) results in the entry of an order for relief or any such adjudication or appointment or (B) remains undismissed, undischarged or unbonded for a period of 60 days;
- there shall be commenced against or any Obligor or the Performance Bond Provider any case, proceeding or other action seeking issuance of a warrant of attachment, execution, distraint or similar process against all or any substantial part of its assets that results in the entry of an order for any such relief that shall not have been vacated, discharged, or stayed or bonded pending appeal within 60 days from the entry thereof;
- (iv) any Obligor or the Performance Bond Provider shall consent to, approve, or acquiesce in, any of the acts set forth in sub-paragraphs (i), (ii) or (iii) above; or
- (v) any Obligor or the Performance Bond Provider shall generally not, or shall be unable to, or shall admit in writing its inability to, pay its debts as they become due,

provided that no Event of Default shall be deemed to have occurred under this paragraph (f) as a result of any such action, event of condition by, against or concerning the Performance Bond Provider if:

- (vi) immediately upon (and, in any event, no more than three Business Days after) becoming aware or receiving notice thereof, the Company gives notice to the Intercreditor Agent of its intention to replace the Performance Bond Provider; and
- (vii) within 60 days (or such shorter period as the Intercreditor Agent may determine is required pursuant to the Concession Contract) after such action, event or condition has occurred:
  - (A) the Performance Bond Provider has been replaced by a Person (the "**Replacement Performance Bond Provider**") acceptable to the Intercreditor Agent;
  - (B) the Replacement Performance Bond Provider has provided a replacement Concession Contract Performance Bond and acceded to the terms of the Deed of Appointment and Priority, in each case on terms acceptable to the Intercreditor Agent; and
  - (C) the Intercreditor Agent is satisfied that no breach of the Concession Contract or any applicable Legal Requirement has occurred or will

- 269 -

result from such replacement, and that the replacement complies, and has been authorised by the Macau SAR in accordance with, the Concession Contract and all other applicable Legal Requirements.

(i) Any Person shall engage in any "prohibited transaction" (as defined in section 406 of ERISA or section 4975 of the Code) involving any Plan;

(g)

- Any failure by any Plan to satisfy the minimum funding standard (within the meaning of Section 412 of the Code or Section 312 of ERISA) applicable to such Plan, whether or not waived, or any Lien in favour of the PBGC or a Plan shall arise on the assets of any Obligor or any Commonly Controlled Entity;
- a Reportable Event shall occur with respect to, or proceedings shall commence to have a trustee appointed, or a trustee shall be appointed, to administer or to terminate, any Single Employer Plan, which Reportable Event or commencement of proceedings or appointment of a trustee is, in the reasonable opinion of the Intercreditor Agent, likely to result in the termination of such Plan for purposes of Title IV of ERISA;
- (iv) any Single Employer Plan shall terminate for purposes of Title IV of ERISA other than in a standard termination under section 4041(b) of ERISA;
- (v) any Obligor or any Commonly Controlled Entity shall, or in the reasonable opinion of the Intercreditor Agent is likely to, incur any liability in connection with a withdrawal from, or the Insolvency or Reorganization of, a Multiemployer Plan, or a determination that a Multiemployer Plan is, or is expected, to be "endangered" or "critical status" within the meaning of Section 305 of ERISA;
- (vi) any Obligor, or any of their Subsidiaries or any Commonly Controlled Entity shall be required to make during any Fiscal Year payments pursuant to any employee welfare benefit plan (as defined in section 3(1) of ERISA) that provides benefits to retired employees (or their dependents), other than as required by sections 601 and following of ERISA, section 4980B of the Code, or the corresponding provisions of applicable state law; or
- (vii) a determination that any Plan is, or is expected to be, in "at risk" status (as defined under Section 303(i)(4) or ERISA or Section 430(i)
   (4) of the Code)),

and in each case in sub-paragraphs (i) through (vii) above, such event or condition, together with all other such events or conditions, if any, could reasonably be expected to have a Material Adverse Effect.

(h) One or more judgments or decrees shall be entered against the Company or any other Obligor involving for the Company and the other Obligors taken as a whole a liability (not paid or covered by insurance as to which the relevant insurance company has acknowledged coverage) of USD50,000,000 or its equivalent or more, and all such judgments or decrees, in either case, shall not have been vacated, discharged, stayed or bonded pending appeal within 30 days from the entry thereof.

- 270 -

- (i) Any of the Senior Finance Documents shall cease, for any reason (other than pursuant to the terms thereof or, in respect of any Palo Security Document, as a result of the matters contemplated by Clause 18.3.4(a) (*Permitted Cotai Reorganisation; Release of Palo Security*) of this Agreement following a Permitted Cotai Reorganisation), to be in full force and effect, or any Obligor (or a Wynn Non-Obligor Subordination Deed Party, in respect of the Subordination Deed only) shall in writing to any Senior Secured Creditor in the event that any Senior Secured Creditor is seeking to exercise its rights or in any Proceedings so assert, or any Lien created or acknowledged by any of the Security Documents shall cease to be enforceable and of the same effect and priority purported to be created or acknowledged thereby.
  - The Company or any other Obligor shall breach or default under in any material respect any material term, condition, provision, covenant, representation or warranty contained in any Major Project Document (other than as set forth in the proviso at the end of this paragraph (j)) and such breach or default shall continue unremedied for 30 or, save in the case of any payment default, provided the Company or such Obligor is diligently pursuing action to remedy the default and it is of a nature that is capable of being remedied, 60 days after the earlier of:
    - (i) the Company or such Obligor becoming aware of such breach or default; and
    - (ii) receipt by the Company of notice from the Intercreditor Agent or any Lender of such breach or default,

provided that:

(j)

- (A) in respect of a Resort Management Agreement, this paragraph (j) shall only apply to any such breach or default under such Resort Management Agreement where such breach or default has not been remedied as set forth above and such breach of default could reasonably be expected to prejudice:
  - (i) the Concession Contract (or any rights, benefits or interests arising thereunder);
  - (ii) the Land Concession Contract (or any rights, benefits or interests arising thereunder); or
  - (iii) the Cotai Land Concession Contract (or any rights, benefits or interests arising thereunder) and such prejudice could reasonably be expected to have an adverse effect on the ability of the Company or any other Obligor to perform its payment obligations under the Senior Finance Documents; and
- (B) in respect of the Cotai Land Concession Contract, this paragraph (j) shall only apply to such breach or default under the Cotai Land Concession Contract where such breach or default has not been remedied as set forth above and such breach of default could reasonably be expected to have an adverse effect on the ability of the Company or any other Obligor to perform its payment obligations under the Senior Finance Documents).

- 271 -

- (k) Any party (other than the Company) shall breach or default under any term, condition, provision, covenant, representation or warranty contained in any Major Project Document (other than as set forth in the proviso at the end of this paragraph (k)) and such breach or default shall continue unremedied for 90 days after the earlier of:
  - (i) the Company or any other Obligor becoming aware of such breach or default; and
  - (ii) receipt by the Company of notice from the Intercreditor Agent or any Lender of such breach or default,

and such breach or default could reasonably be expected to have a Material Adverse Effect,

provided that:

(l)

- (A) in respect of a Resort Management Agreement, this paragraph (k) shall only apply to any such breach or default under such Resort Management Agreement where such breach or default has not been remedied as set forth above and such breach of default could reasonably be expected to prejudice:
  - (i) the Concession Contract (or any rights, benefits or interests arising thereunder);
  - (ii) the Land Concession Contract (or any rights, benefits or interests arising thereunder); or
  - (iii) the Cotai Land Concession Contract (or any rights, benefits or interests arising thereunder) and such prejudice could reasonably be expected to have an adverse effect on the ability of the Company or any other Obligor to perform its payment obligations under the Senior Finance Documents; and
- (B) in respect of the Cotai Land Concession Contract, this paragraph (k) shall only apply to such breach or default under the Cotai Land Concession Contract where such breach or default has not been remedied as set forth above and such breach of default could reasonably be expected to have an adverse effect on the ability of the Company or any other Obligor to perform its payment obligations under the Senior Finance Documents).
- Any of the Transaction Documents shall terminate or be terminated or cancelled, become invalid or illegal or otherwise cease to be in full force and effect prior to its stated expiration date (other than any such termination, cancellation, invalidity, illegality or other ceasing to be in full force and effect prior to its stated expiration date in respect of any Palo Security Document as a result of the matters contemplated by Clause 18.3.4(a) (*Permitted Cotai Reorganisation; Release of Palo Security*) of this Agreement following a Permitted Cotai Reorganisation) *provided* that the occurrence of any of the foregoing events with respect to the Cotai Land Concession Contract shall constitute an Event of Default under this paragraph (l) only if the same could reasonably be expected to have an adverse effect on the ability of the Company or any other Obligor to perform its payment obligations under the Senior Finance

Documents and *provided* further that the occurrence of any of the foregoing events with respect to any Major Project Document (other than the Concession Contract, the Land Concession Contract, the Cotai Land Concession Contract, the Construction Contract or the IP Agreement) shall constitute an Event of Default under this paragraph (l) only if the same could reasonably be expected to result in a Material Adverse Effect and the same shall continue unremedied for 90 days after the earlier of:

- (i) the Company or any other Obligor becoming aware of such occurrence; and
- (ii) receipt by the Company of notice from the Intercreditor Agent or any Lender of such occurrence,

provided that in the case of any such Major Project Document (other than the Concession Contract, the Land Concession Contract, the Cotai Land Concession Contract, the Construction Contract or the IP Agreement), if the occurrence is not the result of the breach or default by an Obligor in any material respect of any material term, condition, provision, covenant, representation or warranty, then no Event of Default shall be deemed to have occurred as a result thereof under this paragraph (l) if the Company provides written notice to the Intercreditor Agent immediately upon (but in no event more than 10 Business Days after) the Company or such other Obligor becoming aware of such occurrence that it intends to replace such Project Document and:

- (A) the Company obtains a replacement obligor or obligors for the affected party;
- (B) the Company or such other Obligor enters into a replacement Major Project Document on terms no less beneficial to the Company or such other Obligor and the Senior Secured Creditors in any material respect than the Major Project Document being replaced within 60 days of such occurrence, *provided* that the replacement Major Project Document may require the Company or such other Obligor to pay amounts under the replacement Major Project Document in excess of those that would have been payable under the replaced Major Project Document; and
- (C) in the reasonable opinion of the Intercreditor Agent, such occurrence, after considering any replacement obligor and replacement Major Project Document and the time required to implement such replacement, has not had and could not reasonably be expected to have a Material Adverse Effect.
- (m) A Change of Control shall occur.
- (n) Any Subordinated Debt or the Performance Bond Facility Agreement shall cease, for any reason, to be validly subordinated to the Obligations of the Obligors as provided in the Senior Finance Documents and the documentation, instruments or other agreements related to the Subordinated Debt, as the case may be.
- (o) The Company or any other Obligor shall fail to observe, satisfy or perform, or there shall be a violation or breach of, any of the terms, provisions, agreements, covenants or conditions attaching to or under the issuance to such Person of any Permit or any such Permit or any provision thereof shall be suspended, revoked, cancelled, terminated or materially and adversely modified or fail to be in full force and effect or any Governmental Authority shall challenge or seek to revoke any such Permit if such failure to perform, violation, breach, suspension, revocation, cancellation, termination or modification could reasonably be expected to have a Material Adverse Effect.

- 273 -

(p) [Not used]

- (q) [Not used]
- (i) The Company shall abandon either of the Projects (other than the Diamond Expansion) or otherwise cease to pursue them or (ii) Palo (or, following a Permitted Cotai Reorganisation, the Company) shall abandon the Cotai Project or otherwise cease to pursue it (development of the Cotai Project having ceased for a period of at least ninety days) where such abandonment or cessation could reasonably be expected to have an adverse effect on the ability of the Company or any other Obligor to perform its payment obligations under the Finance Documents.
- (s) Any call or drawing made by the Macau SAR under the Concession Contract Performance Bond unless the Concession Contract Performance Bond is fully reinstated within 30 days thereof in accordance with the Concession Contract, no other Event of Default has occurred or will result from such reinstatement.
- (t) The Company or any other Obligor shall fail to at all times maintain in full force and effect the insurance policies required by Schedule 7 (*Insurance*) and paragraph 10.4 of Part A of Schedule 5; *provided* that if any insurer or reinsurer shall be subject to any bankruptcy, insolvency or similar proceedings that causes any such insurance policy to no longer be in full force and effect or if the ratings requirement set forth in paragraph 2.1.1 of Schedule 7 (*Insurance*) ceases to be satisfied with respect to any insurer or reinsurer, no Event of Default shall be deemed to have occurred as a result thereof under this paragraph (t) if the Company provides prompt written notice to the Intercreditor Agent of such occurrence and the Company or such other Obligor, as applicable, obtains replacement insurance complying with the terms of this Agreement within 30 days of such event occurring.
- (u) Any Governmental Authority takes any action or there is a change in (or in the interpretation, administration or application of) or the introduction of any Legal Requirement:
  - which deprives the Company or any other Obligor of the use of all or any material part of its Property (including nationalisation, expropriation, modification, suspension or extinguishment of any material rights benefiting or the imposition of any restrictions materially and adversely affecting any Project by such Governmental Authority);
  - (ii) which prevents the Company or any other Obligor from conducting its business or operations, or a material part thereof, in a similar manner as contemplated at the Fourth Amendment Effective Date; or
  - (iii) which otherwise could reasonably be expected to have a Material Adverse Effect,

and, in each case, such action, change or introduction or the effects thereof, are not removed or stayed within 60 days of the occurrence of such action, change or

- 274 -

introduction, *provided* further that the occurrence of any of the foregoing events with respect to the Cotai Project shall constitute an Event of Default under this paragraph (u) only if the same could reasonably be expected to have an adverse effect on the ability of the Company or any other Obligor to perform its payment obligations under the Senior Finance Documents.

- (v) Any temporary administrative intervention is made by the Macau SAR pursuant to article 79 of the Concession Contract.
- (w) The Macau SAR takes any formal measure seeking the unilateral dissolution of the Concession Contract pursuant to article 80 thereof or the Macau SAR gives notice pursuant to article 80(3) of the Concession Contract and the Company fails to comply with the terms thereof within the grace period specified therein.
- (x) The Intercreditor Agent considers the subject matter of any negotiations required to be notified to it pursuant to paragraph 24(c) of Part A of Schedule 5 (*Covenants*) is such as could reasonably give rise to an entitlement of the Macau SAR to unilaterally dissolve the Concession Contract pursuant to article 80 thereof.
- (y) Any consultations are commenced as contemplated by paragraph B1(c) of the Gaming Concession Consent Agreement and (save for the purpose of giving notice to the Macau SAR that the Security Agent intends to take action to enforce all or any of the pledges constituted by the Company Share Pledge, the Wynn International Share Charge and/or the Wynn HK Share Charge which may be given by the Security Agent immediately upon such consultations being commenced) the Intercreditor Agent considers the subject matter of such consultations is reasonably likely to give rise to (a) the taking of any action to terminate the Concession Contract or (b) an agreement to terminate the Concession Contract.
- (z) The Land Concession Contract or the Cotai Land Concession Contract is terminated or rescinded or the Macau SAR takes any formal measure seeking any termination of (i) the Land Concession Contract pursuant to Clause 15 thereof or any rescission pursuant to Clause 16 thereof or (ii) the Cotai Land Concession Contract pursuant to Clause 14 thereof or any rescission pursuant to Clause 15 thereof, *provided* that, the occurrence of the foregoing events with respect to the Cotai Land Concession Contract shall constitute an Event of Default under this paragraph (z), only if the same could reasonably be expected to have an adverse effect on the ability of the Company or any other Obligor to perform its payment obligations under the Senior Finance Documents .
- (aa) The Macau SAR gives any notice pursuant to paragraph C7 of the Land Concession Consent Agreement.
- (bb) A Material Adverse Effect has occurred which is continuing.

- 275 -

## SCHEDULE 11 TRANSFERS AND ACCESSION

#### Part A

#### Form of Agent's Deed of Accession

THIS DEED dated [ ] is supplemental to each of the Senior Finance Documents as defined in a common terms agreement (the "**Common Terms Agreement**") dated 14 September 2004 between Wynn Resorts (Macau) S.A. and the Senior Secured Creditors (as defined therein) to which [*name of existing Agent*] is expressed to be a party (the "**Senior Finance Documents**").

Words and expressions defined in the Common Terms Agreement have the same meaning when used in this Deed and the principles of construction and rules of interpretation set out therein shall also apply.

[*name of new Agent*] (the "**New Agent**") of [*address*] hereby agrees with each other person who is or who becomes a party to the Senior Finance Documents that with effect on and from the date of this Deed it shall be bound by the Senior Finance Documents and be entitled to exercise rights and be subject to obligations thereunder as [*specify Agent*].

The Facility Office of the New Agent is located at [

The initial telephone number, fax number, address and person designated by the New Agent for the purposes of Clause 29 (*Notices*) of the Common Terms Agreement are:

].

[

]

This Deed is governed by and shall be construed in accordance with English law.

| Executed as a deed by         | ) |
|-------------------------------|---|
| [insert name of new Agent and | ) |
| execution clause appropriate  | ) |
| thereto and to manner of      | ) |
| execution]                    | ) |
|                               |   |

- 276 -

#### Part B

#### Form of Novation Certificate

To: [ ] as Intercreditor Agent

#### NOVATION CERTIFICATE

relating to [description of the relevant Facility Agreement] (the "Facility Agreement") dated 14 September 2004 between Wynn Resorts (Macau) S.A. (the "Company") and [list other parties] and the common terms agreement (the "Common Terms Agreement") dated 14 September 2004 between the Company and the Senior Secured Creditors (as defined therein).

- 1. Terms defined in the Common Terms Agreement shall, subject to any contrary indication, have the same meanings herein and the principles of construction and rules of interpretation set out therein shall also apply. The terms Lender, Transferee, Proposed Transfer Date, Lender's Participation and Amount Transferred are defined in the schedule hereto.
- 2. The Lender confirms that the Lender's Participation is an accurate summary of its participation in the Facility Agreement and requests the Transferee to accept and procure the transfer by novation to the Transferee of a percentage of the Lender's Participation (equal to the percentage that the Amount Transferred is of the aggregate of the component amounts (as set out in the schedule hereto) of the Lender's Participation the "Transferred Percentage") by counter-signing and delivering this Novation Certificate to the Intercreditor Agent at its address for the service of notices specified in the Common Terms Agreement, in accordance with Clause 21.6 (*Transfers by Lenders*) of the Common Terms Agreement. The Lender assigns, subject to the aforementioned acceptance by the Transferee, a proportion of the rights and benefits held by the Lender (in its capacity as such) under or in connection with the Senior Finance Documents which proportion shall be equal to the Transferred Percentage.
- 3. The Transferee hereby requests the Intercreditor Agent to accept this Novation Certificate as being delivered to the Intercreditor Agent pursuant to and for the purposes of Clause 21.6 (*Transfers by Lenders*) of the Common Terms Agreement so as to take effect in accordance with the terms thereof on the Proposed Transfer Date or on such later date as may be determined in accordance with the terms thereof.
- 4. The Transferee confirms that it has received a copy of each of the Senior Finance Documents together with such other information as it has required in connection with this transaction and that it has not relied and shall not hereafter rely on the Lender to check or enquire on its behalf into the legality, validity, effectiveness, adequacy, accuracy or completeness of any such information and further agrees that it has not relied and shall not rely on the Lender to assess or keep under review on its behalf the financial condition, creditworthiness, condition, affairs, status or nature of the Obligors, the Projects or the Cotai Project.
- 5. The Transferee hereby undertakes with the Lender and each of the other parties to the Facility Agreement that it shall perform in accordance with their terms all those obligations which by the terms of the Facility Agreement shall be assumed by it after

- 277 -

delivery of this Novation Certificate to the Intercreditor Agent and satisfaction of the conditions (if any) subject to which this Novation Certificate is expressed to take effect.

- The Transferee also agrees that, with effect from the Proposed Transfer Date or such later date as may be determined in accordance with Clause 21.6 (*Transfers by Lenders*) of the Common Terms Agreement, it shall be bound by the terms of:
  - (a) the Common Terms Agreement as if it had been a party to such agreement in the capacity of a [specify Lender]; and
  - (b) each of the Security Documents to which the Lenders are party as if it had been a party to those documents in the capacity of a Lender thereunder.
- 8. The Lender hereby gives notice that nothing herein or in the Senior Finance Documents (or any document relating thereto) shall oblige the Lender to (a) accept a re-transfer from the Transferee of the whole or any part of its rights, benefits and/or obligations under the Senior Finance Documents transferred pursuant hereto or (b) support any losses directly or indirectly sustained or incurred by the Transferee for any reason whatsoever including the non-performance by any of the Obligors or any other party to the Senior Finance Documents (or any document relating thereto) of its obligations under any such document. The Transferee hereby acknowledges the absence of any such obligation as is referred to in (a) or (b).
- 9. This Novation Certificate and the rights, benefits and obligations of the parties hereunder shall be governed by and construed in accordance with English law.

#### THE SCHEDULE

1. Lender:

6.

- 2. Transferee:
- 3. Proposed Transfer Date:
- 4. Lender's Participation:

Lender's undrawn Available Commitment\*

Lender's Portion of each Advance

- 278 -

| 5. 4            | Amount Transferred:    |               |
|-----------------|------------------------|---------------|
| [Lender]        |                        | [Transferee]  |
| By:             |                        | By:           |
| Date:           |                        | Date:         |
| [               | ]                      |               |
| as Intercredite | or Agent               |               |
| By:             |                        |               |
| Date:           |                        |               |
|                 | Administrative Details | of Transferee |
| Address:        |                        |               |
| Contact Name    | e:                     |               |
| Account for I   | Payments:              |               |
| Standing Pay    | ment Instruction:      |               |
| Fax:            |                        |               |
| Telex:          |                        |               |
| Telephone:      |                        |               |

\* Details of the Lender's undrawn Available Commitment should not be completed after the last day of the Availability Period.

## Part C

#### Form of Confidentiality Undertaking

[Date]

To: [Insert name of potential Transferee/participant]

Dear Sirs,

We understand that you are considering [acquiring an interest (the "Acquisition") in/accepting an appointment as facility agent under [description of the relevant Facility Agreement] (the "Facility Agreement")/accepting an appointment as intercreditor agent under the Senior Finance Documents (the "Appointment")] in relation to the design, development, construction, ownership, operation and maintenance of the Wynn Macau hotel, retail and destination gaming resort project (the "Project") and the Wynn Cotai hotel, retail and destination gaming resort project (the "Cotai Project"). In consideration of us agreeing to make available to you certain information, by your signature of a copy of this letter you agree as follows:

1. *Confidentiality Undertaking* You undertake (a) to keep the Confidential Information confidential and not to disclose it to anyone except as provided for by paragraph 2 below and to ensure that the Confidential Information is protected with security measures and a degree of care that would apply to your own confidential information, (b) to use the Confidential Information only for the Permitted Purpose, (c) to use all reasonable endeavours to ensure that any person to whom you pass any Confidential Information (unless disclosed under paragraph 2(c) below) acknowledges and complies with the provisions of this letter as if that person were also a party to it, and (d) not to make enquiries of any of the Obligors or any of their officers, directors, employees or professional advisers relating directly or indirectly to the [Acquisition/Appointment].

- 279 -

*Permitted Disclosure* We agree that you may disclose Confidential Information:

- (a) to members of the [Purchaser/Appointee] Group and their officers, directors, employees and professional advisers to the extent necessary for the Permitted Purpose and to any auditors of members of the [Purchaser/Appointee] Group;
- (b) [subject to the requirements of the Senior Finance Documents, to any person to (or through) whom you assign or transfer (or may potentially assign or transfer) all or any of the rights, benefits and obligations which you may acquire under the Facility Agreement or with (or through) whom you enter into (or may potentially enter into) any sub-participation in relation to, or any other transaction under which payments are to be made by reference to, the Senior Finance Documents or any of the Obligors so long as that person has delivered a letter to you in equivalent form to this letter;] and
- (c) (i) where requested or required by any court of competent jurisdiction or any competent judicial, governmental, supervisory or regulatory body, (ii) where required by the rules of any stock exchange on which the shares or other securities of any member of the [Purchaser/Appointee] Group are listed or (iii) where required by the laws or regulations of any country with jurisdiction over the affairs of any member of the [Purchaser/Appointee] Group.
- 3. *Notification of Required or Unauthorised Disclosure* You agree (to the extent permitted by law) to inform us of the full circumstances of any disclosure under paragraph 2[(c)/(b)<sup>6</sup>] or upon becoming aware that Confidential Information has been disclosed in breach of this letter.
- 4. *Return of Copies* If we so request in writing, you shall return all Confidential Information supplied to you by us and destroy or permanently erase all copies of Confidential Information made by you and use all reasonable endeavours to ensure that anyone to whom you have supplied any Confidential Information destroys or permanently erases such Confidential Information and any copies made by them, in each case save to the extent that you or the recipients are required to retain any such Confidential Information by any applicable law, rule or regulation or by any competent judicial, governmental, supervisory or regulatory body or in accordance with internal policy, or where the Confidential Information has been disclosed under paragraph 2[(c)/(b)<sup>7</sup>] above.

2.

If paragraph 2(b) deleted.

- 280 -

<sup>&</sup>lt;sup>6</sup> If paragraph 2(b) deleted.

- 5. *Continuing Obligations* The obligations in this letter are continuing and, in particular, shall survive the termination of any discussions or negotiations between you and us. Notwithstanding the previous sentence, the obligations in this letter shall cease (a) if you become a party to [or otherwise acquire (by assignment or sub-participation) an interest, direct or indirect, in] the [Facility Agreement/Senior Finance Documents] or (b) twelve months after you have returned all Confidential Information supplied to you by us and destroyed or permanently erased all copies of Confidential Information made by you (other than any such Confidential Information or copies which have been disclosed under paragraph 2 above (other than sub-paragraph 2(a)) or which, pursuant to paragraph 4 above, are not required to be returned or destroyed).
  - No Representation; Consequences of Breach, etc You acknowledge and agree that:

6.

- (a) neither we nor any Obligor nor any of our or their respective officers, employees or advisers (each a "Relevant Person") (i) make any representation or warranty, express or implied, as to, or assume any responsibility for, the accuracy, reliability or completeness of any of the Confidential Information or any other information supplied by us or the assumptions on which it is based or (ii) shall be under any obligation to update or correct any inaccuracy in the Confidential Information or any other information in respect to the Confidential Information; and
- (b) we or any Obligor may be irreparably harmed by the breach of the terms hereof and damages may not be an adequate remedy; each Relevant Person may be granted an injunction or specific performance for any threatened or actual breach of the provisions of this letter by you.
- 7. *No Waiver; Amendments, etc* This letter sets out the full extent of your obligations of confidentiality owed to us in relation to the information the subject of this letter. No failure or delay in exercising any right, power or privilege hereunder will operate as a waiver thereof nor will any single or partial exercise of any right, power or privilege preclude any further exercise thereof or the exercise of any other right, power or privileges hereunder. The terms of this letter and your obligations hereunder may only be amended or modified by written agreement between us.
- 8. *Inside Information* You acknowledge that some or all of the Confidential Information is or may be price-sensitive information and that the use of such information may be regulated or prohibited by applicable legislation relating to insider dealing and you undertake not to use any Confidential Information for any unlawful purpose.
- 9. *Nature of Undertakings* The undertakings given by you under this letter are given to us and (without implying any fiduciary obligations on our part) are also given for the benefit of each Obligor.

- 281 -

## 10. Third party rights

- (a) Subject to this paragraph 10 and to paragraphs 6 and 9, a person who is not a party to this letter has no right under the Contracts (Rights of Third Parties) Act 1999 (the "**Third Parties Act**") to enforce or to enjoy the benefit of any term of this letter.
- (b) The Relevant Persons may enjoy the benefit of the terms of paragraphs 6 and 9 subject to and in accordance with this paragraph 10 and the provisions of the Third Parties Act.
- (c) The parties to this letter do not require the consent of the Relevant Persons to rescind or vary this letter at any time.
- 11. *Governing Law and Jurisdiction* This letter (including the agreement constituted by your acknowledgement of its terms) shall be governed by and construed in accordance with the laws of England and the parties submit to the non-exclusive jurisdiction of the English courts.
- 12. *Definitions* In this letter (including the acknowledgement set out below) terms defined in or by reference to the Facility Agreement shall, unless the context otherwise requires, have the same meaning, the principles of construction and rules of interpretation referred to therein shall also apply and:

"**Confidential Information**" means any information relating to the Project, the Cotai Project, any Obligor, the Transaction Documents, any agreement relating to the [Facility Agreement/Senior Finance Documents] and/or the [Acquisition/Appointment] provided to you by us or any of our affiliates or advisers, in whatever form, and includes information given orally and any document, electronic file or any other way of representing or recording information which contains or is derived or copied from such information but excludes information that (a) is or becomes public knowledge other than as a direct or indirect result of any breach of this letter or (b) is known by you before the date the information is disclosed to you by us or any of our affiliates or advisers or is lawfully obtained by you thereafter, other than from a source which is connected with the Obligors and which, in either case, as far as you are aware, has not been obtained in violation of, and is not otherwise subject to, any obligation of confidentiality.

"**Obligors**" means Wynn Resorts (Macau) S.A. and certain other persons, as defined in the Senior Finance Documents, who have a direct or indirect interest in its share capital.

"Permitted Purpose" means considering and evaluating whether to [enter into/accept] the [Acquisition/Appointment].

"Project Documents" means the documents entered into by Wynn Resorts (Macau) S.A. and its contractors or subcontractors in connection with the Project.

"[Purchaser/Appointee] Group" means you, each of your holding companies and subsidiaries and each subsidiary of each of your holding companies (as each such term is defined in the Companies Act 1985).

"Senior Finance Documents" means the Facility Agreement, the Common Terms Agreement dated 14 September 2004 between Wynn Resorts (Macau) S.A. and certain financial institutions and other Senior Finance Documents as defined in such Common Terms Agreement.

- 282 -

"Transaction Documents" means the Senior Finance Documents and the Project Documents.

Please acknowledge your agreement to the above by signing and returning the enclosed copy.

Yours faithfully

For and on behalf of

[Insert name of Lender/Agent]

- 283 -

# To: [Lender/Agent] The Obligors

We acknowledge and agree to the above:

For and on behalf of

# [Potential Transferee/participant/appointee]

Date:

- 284 -

#### SCHEDULE 12 PERMITS

THE PERMITS DESCRIBED IN PART A AND PART B OF THIS SCHEDULE 12 (PERMITS) HAVE BEEN ISSUED AND ARE IN FULL FORCE AND EFFECT.

## Part A

## Permits required on or before the CP Satisfaction Date

- 1. Construction licence.
- 2. Enactment of legislation by the Macau SAR providing for casino operators to be grantors of credit to patrons.
- 3. Macau SAR approval of the location of the horizontal property comprised in the Casino Project.
- 4. Publication of the Land Concession Contract in the Official Bulletin.
- 5. Provisional registration of the rights of the Company to the land which is the subject of the Land Concession Contract (for the purposes of this Schedule, the "Land").
- 6. Macau SAR confirmation that the scope of insurances set out in Schedule 7 satisfy the requirements of Article 40 of the Concession Contract.
- 7. Agreement by the Gambling Inspection and Coordination Bureau and the Financial Bureau of the Macau SAR as to the identity of the Auditor in accordance with Article 57 of the Concession Contract.
- 8. Macau SAR authorisation: (i) pursuant to article 16(1) of the Concession Contract to pledge the Company's shares; (ii) pursuant to article 16(5) of the Concession Contract to charge the shares of the Company's shareholders; (iii) pursuant to article 21(3) of the Concession Contract to execute a power of attorney in relation to the Land; (iv) pursuant to article 42(1) of the Concession Contract to pledge the portion of the Land, including any constructions thereon, referred to therein; and (v) pursuant to article 42(1) of the Concession Contract to pledge the gaming equipment and utensils of the Company.
- 9. To the extent applicable, disclosure to Macau SAR of any serious alteration to the economic or financial conditions of the Company, the Company's shareholders or certain of the Company's Affiliates pursuant to article 23 of the Concession Contract.

- 285 -

#### Part B

#### Permits required after the CP Satisfaction Date

- 1. Project Certificates of Occupancy.
- 2. Definitive registration of the rights of the Company to the land which is the subject of the Land Concession Contract and definitive registration of the horizontal property comprised in the Projects.
- 3. Permits required under Article 84 of the Concession Contract.
- 4. Amended construction licence for the Projects (including the Expansion).
- 5. Written confirmation from Macau SAR Department of Public Works that, upon approval of the amended construction licence for the Projects (including the Expansion), the Land Concession Contract will be amended as soon as practicable to the extent required to reflect the Projects as actually built pursuant to the original construction licence and the amended construction licence.

- 286 -

## SCHEDULE 13 [NOT USED]

- 287 -

## SCHEDULE 14 FORM OF ADDITIONAL LENDER'S ACCESSION DEED

THIS DEED dated [ ] is supplemental to a common terms agreement (the "**Common Terms Agreement**") dated 14 September 2004 between Wynn Resorts (Macau) S.A. and the Senior Secured Creditors (as defined therein).

Words and expressions defined in the Common Terms Agreement have the same meaning when used in this Deed and the principles of construction and rules of interpretation set out therein shall also apply.

[*name of Additional Lender*] (the "**New Additional Lender**") of [*address*] hereby agrees with each other person who is or who becomes a party to the Common Terms Agreement that with effect on and from the date of this Deed it shall be bound by the Common Terms Agreement and be entitled to exercise rights and be subject to obligations thereunder as an Additional Lender.

The initial telephone number, fax number, address and person designated by the New Additional Lender for the purposes of Clause 29 (*Notices*) of the Common Terms Agreement are:

[

]

This Deed is governed by and shall be construed in accordance with English law.

| Executed as a deed by       | ) |  |
|-----------------------------|---|--|
| [insert name of Additional  | ) |  |
| Lender and execution clause | ) |  |
| appropriate thereto and to  | ) |  |
| manner of execution]        | ) |  |
|                             |   |  |

- 288 -

#### SCHEDULE 15 FORM OF COMPLIANCE CERTIFICATE

To: [ ] as Intercreditor Agent

|            |  |  | Date: [ | ] |
|------------|--|--|---------|---|
| Dear Sirs, |  |  |         |   |

]

We refer to an agreement (the "**Common Terms Agreement**") dated 14 September 2004 between Wynn Resorts (Macau) S.A. and the financial institutions named therein as Senior Secured Creditors. Terms defined in the Common Terms Agreement shall bear the same meaning herein and the principles of construction and rules of interpretation set out therein shall also apply.

We confirm on [insert date of relevant financial statements] the following:

|                            | Actu | ıal  | Required | 1 |
|----------------------------|------|------|----------|---|
| 1. Leverage Ratio          | [    | ]    | [        | ] |
| 2. Interest Coverage Ratio | [    | ]    | [        | ] |
|                            |      | Actu | ıal      |   |
| 3. Excess Cash Flow*       |      |      |          |   |

[\* Excess Cash Flow calculation to be included only if the Leverage Ratio for the relevant period is greater than 4:1]

We attach the information and calculations necessary for determining the above ratios and amounts.

We hereby confirm that no Default has occurred and is continuing.

OR

We hereby give you notice of the occurrence of the following Default which is continuing:

```
[].
```

].

We set out below the steps being taken to remedy such Default:

[

- 289 -

Yours faithfully,

Name: Responsible Officer for and on behalf of **Wynn Resorts (Macau) S.A.**  SCHEDULE 16 [NOT USED]

- 291 -

SCHEDULE 17 [NOT USED]

- 292 -

## SCHEDULE 18 MONTHLY CONSTRUCTION PERIOD REPORT

#### List of Minimum Information to be Included

## APPLICABLE TO THE PROJECTS ONLY (AND NOT TO THE COTAI PROJECT)

## A Summary

## B Project Schedule

- 1. Describe (in respect of both work under the Construction Contract and Company scope including FF&E and pre-opening activities):
  - 1.1 Overall progress of work broken down by major area
  - 1.2 Major activities that have taken place in the period since the last report
  - 1.3 Major activities scheduled to take place in the period until the next report
  - 1.4 The Company's estimate of:
    - (a) The date of Substantial Completion for the Original Project and the Expansion
    - (b) The Opening Date (and satisfaction of the Opening Conditions specified in paragraph (a) of the definition thereof) for the Original Project and the Expansion Opening Date
    - (c) The date of Expansion Completion
  - With reference to the attached Project Schedule:
    - 2.1 Highlight changes in the Project Schedule from the last report
    - 2.2 Highlight major milestones achieved in the period since the last report
    - 2.3 Describe remedial activities being taken to accelerate the works (if any)

## C Manpower

2.

- 1. Indicate current staffing level vs. projected for the Company, the PASA Agent and the Construction Contractor
- 2. Indicate any fatalities or injuries associated with the Projects incurred by the Company, the PASA Agent, the Prime Contractor, any Subcontractor or any other person in the period since the previous report with detail as to the nature of injuries incurred and cumulative figures since the [Date of Commencement] under and as defined in the Construction Contract
- 3. Highlight any major executive positions filled or vacated in the period since the last report

- 293 -

| D | Other                |  |
|---|----------------------|--|
|   | 1.                   | List material Permits issued or made by or with a Governmental Authority obtained by the Company in the period since the last report   |
|   | 2.                   | List any requests for change orders or variations under the Construction Contract received, requested, agreed or approved in the period since the last report                          |
|   | 3.                   | Hedging Arrangements entered into in the period since the last report  |
| Е | Lease and            | Subconcession Agreements   |
|   | 1.                   | List total space leased vs. vacant for each of the following categories:   |
|   |                      | 1.1 Restaurants  |
|   |                      | 1.2 Retail   |
|   |                      | 1.3 Other facilities   |
|   | 2.                   | List tenants secured and target date of opening for each space indicated as leased   |
|   | 3.                   | Describe any agreements, whether in-principle or otherwise, entered into by the Company in the period since the last report in respect of subconcessions under the Concession Contract |
| F | Operating            | Results of Projects  |
|   | Following month, inc | the Opening Date of the Original Project until the Expansion Opening Date, provide the operating results of the Company in the preceding luding:                                       |
|   | (a)                  | Operating revenues, broken down by major source category   |
|   | (b)                  | Operating costs, broken down by major line item  |
|   | (c)                  | Working capital and other adjustments for non-cash items   |
|   | (d)                  | Other cash payments or reserves made from operating cashflow   |
|   | (e)                  | Net cashflow from operations and the sum transferred to the Construction Disbursement Accounts   |
| G | Schedules            |  |

# Schedules

Photographs

#### н Attachments

- 1. Project Schedule 2. Actual vs. projected expenditure "S" curve 3.
  - Monthly Construction Progress Report

- 294 -

## SCHEDULE 19 FORMS OF OPENING CONDITIONS CERTIFICATES

## PARTS A AND B APPLICABLE TO THE PROJECTS ONLY (AND NOT TO THE COTAI PROJECT)

#### Part A

## Form of Company's Opening Conditions Certificate

Date: [ ]

Dear Sirs,

- 1. We refer to the common terms agreement (the "**Common Terms Agreement**") dated 14 September 2004 between Wynn Resorts (Macau) S.A. and the financial institutions referred to therein as Senior Secured Creditors. Terms defined in the Common Terms Agreement shall have the same meaning herein and the principles of construction and rules of interpretation set out therein shall also apply.
- 2. This certificate is provided for the purposes of the definition of "Opening Conditions" in Clause 1.1 (*Definitions*) of the Common Terms Agreement in respect of the Original Project.

3. We hereby certify, as at the date of this certificate, that:

- (i) furnishings, fixtures and equipment necessary to use and occupy the various portions of the Original Project for their intended uses have been installed and are operational;
- (ii) all the Project Certificates of Occupancy have been issued, each area in which any operation of casino games of chance or other forms of gaming will be carried out has been classified as a casino or gaming zone in accordance with Article 9 of the Concession Contract and (other than any Permit issued or made by or with a Governmental Authority the failure of which to obtain or make could not reasonably be expected to affect the operations of the Projects in any material respect) each other Permit issued or made by or with a Governmental Authority required under applicable Legal Requirements to be obtained or made prior to the Opening Date has been obtained or made;
- (iii) the Original Project (other than the premises to be occupied by individual retail and restaurant tenants in the Projects) is fully open for business to the general public and at least, in the case of the Original Project, 80% of each of the projected 380 slot machines and 200 gaming tables are in operation, 80% of restaurant outlets are open for business and at least 80% of the floor space comprised in the Original Project and designated in the Plans and Specifications for retail tenants has been occupied (save for facilities which by their nature are not open to the general public in the ordinary course of business but are operating);

- 295 -

- (iv) the remaining work on the Projects will not materially affect the operation of the Original Project;
- (v) the failure to complete the remaining work will not materially affect the operation of the Original Project; and
- (vi) the Company has available a fully trained staff to operate the Original Project including the hotel and casino.

Yours faithfully,

- 296 -

## Part B Form of Technical Adviser's Opening Conditions Certificate

To: ſ ] as Intercreditor Agent ]

ſ Date:

Dear Sirs,

- 1. We refer to the common terms agreement (the "Common Terms Agreement") dated 14 September 2004 between Wynn Resorts (Macau) S.A. and the financial institutions referred to therein as Senior Secured Creditors. Terms defined in the Common Terms Agreement shall have the same meaning herein and the principles of construction and rules of interpretation set out therein shall also apply.
- This certificate is provided for the purposes of the definition of "Opening Conditions" in Clause 1.1 (Definitions) of the Common Terms Agreement 2. in respect of the Original Project.
- We hereby certify, as at the date of this certificate, that: 3.
  - furnishings, fixtures and equipment necessary to use and occupy the various portions of the Original Project for their intended uses (i) have been installed and are operational;
  - (ii) all the Project Certificates of Occupancy have been issued and (other than any Permit made or issued by or with a Governmental Authority the failure of which to make or obtain could not reasonably be expected to affect the operations of the Projects in any material respect) each other Permit made or issued by or with a Governmental Authority required under applicable Legal Requirements to be made or obtained prior to the Opening Date have been obtained;
  - (iii) the Original Project (other than the premises to be occupied by individual retail and restaurant tenants in the Projects) is fully open for business to the general public and at least, in the case of the Original Project, 80% of each of the projected 380 slot machines and 200 gaming tables are in operation, 80% of restaurant outlets are open for business and at least 80% of the floor space comprised in the Original Project and designated in the Plans and Specifications for retail tenants has been occupied (save for facilities which by their nature are not open to the general public in the ordinary course of business but are operating);
  - the remaining work on the Projects will not materially affect the operation of the Original Project; and (iv)
  - the failure to complete the remaining work will not materially affect the operation of the Original Project. (v)

- 297 -

Yours faithfully,

Name: For and on behalf of [*Technical Adviser*]

# The Company

| S.A. |
|------|
|      |

| By:        | /s/ Jason M. Schall  |
|------------|--|
| Address:   | Wynn Macau<br>Rua Cidade de Sintra<br>NAPE<br>Macau                              |
| Tel:       | +853 2888 9966   |
| Fax:       | +853 2832 9966   |
| Attention: | Chief Financial Officer  |
| Copy to:   | Wynn Resorts (Macau) S.A.<br>Wynn Macau<br>Rua Cidade de Sintra<br>NAPE<br>Macau |
| Tel:       | +853 2888 9966   |
| Fax:       | +853 2832 9966   |
| Attention  | Logal Department of Warn Desorts (Macau) S.A.                                    |

Attention: Legal Department of Wynn Resorts (Macau) S.A.

- 299 -

## The Term Facility Lender

# BANK OF CHINA LIMITED, MACAU BRANCH

| By:        | /s/ Leong Sio Hong  |
|------------|---|
| Address:   | 13/F Bank of China Building<br>Avenida Doutor Mario Soares, Macau |
| Tel:       | +853 8792 1639 / 8792 1682  |
| Fax:       | +853 8792 1659 / 8792 0308  |
| Attention: | Mr. James Wong / Ms. Amy Cheong                                   |
| Copy to:   |   |
| Address:   | 17/F Bank of China Building<br>Avenida Doutor Mario Soares, Macau |
| Tel:       | +853 8792 1623 / 8792 1706  |
| Fax:       | +853 8792 1677  |

Attention: Ms. Wendy Sun / Mr. Jerry Chan

- 300 -

## The Project Facility Lender

# BANK OF CHINA LIMITED, MACAU BRANCH

| By:        | /s/ Leong Sio Hong  |
|------------|---|
| Address:   | 13/F Bank of China Building<br>Avenida Doutor Mario Soares, Macau |
| Tel:       | +853 8792 1639 / 8792 1682  |
| Fax:       | +853 8792 1659 / 8792 0308  |
| Attention: | Mr. James Wong / Ms. Amy Cheong                                   |
| Copy to:   |   |
| Address:   | 17/F Bank of China Building<br>Avenida Doutor Mario Soares, Macau |
| Tel:       | +853 8792 1623 / 8792 1706  |
| Fax:       | +853 8792 1677  |

Attention: Ms. Wendy Sun / Mr. Jerry Chan

- 301 -

## The Revolving Credit Facility Lender

# BANK OF CHINA LIMITED, MACAU BRANCH

| By:        | /s/ Leong Sio Hong  |
|------------|---|
| Address:   | 13/F Bank of China Building<br>Avenida Doutor Mario Soares, Macau |
| Tel:       | +853 8792 1639 / 8792 1682  |
| Fax:       | +853 8792 1659 / 8792 0308  |
| Attention: | Mr. James Wong / Ms. Amy Cheong                                   |
| Copy to:   |   |
| Address:   | 17/F Bank of China Building<br>Avenida Doutor Mario Soares, Macau |
| Tel:       | +853 8792 1623 / 8792 1706  |
| Fax:       | +853 8792 1677  |

Attention: Ms. Wendy Sun / Mr. Jerry Chan

- 302 -

## The Hedging Counterparties

# BANC OF AMERICA SECURITIES ASIA LIMITED

| By:             | /s/ Chin Voon Fat Frederick   |  |  |  |
|-----------------|---|--|--|--|
| Address:        | 42/F Two International Finance Centre<br>8 Finance Street<br>Central<br>Hong Kong |  |  |  |
| Tel:            | +852 2847 5266  |  |  |  |
| Fax:            | +852 2847 5886  |  |  |  |
| Attention:      | Brian Ho  |  |  |  |
| With a copy to: |   |  |  |  |
| Tel:            | +852 2847 6778 / +852 2847 6760   |  |  |  |
| Fax:            | +852 2810 0821 / +852 2847 6767   |  |  |  |
| Attention:      | Kevin Salerno / Jackson Lum   |  |  |  |

- 303 -

# DEUTSCHE BANK AG

| By:        | /s/ Sameen Farooqui   | Sameen Farooqui<br>Managing Director | /s/ Anand Joshi | Anand Joshi<br>Director |
|------------|---|--------------------------------------|-----------------|-------------------------|
| Address:   | Deutsche Bank AG, Singapore Branc<br>One Raffles Quay, #14-00<br>Singapore 048593 | h                                    |                 |                         |
| Tel:       | +65 6423 6862 / +65 6423 6865   |                                      |                 |                         |
| Fax:       | +65 6883 0835   |                                      |                 |                         |
| Attention: | Amelia Tan / Vincent Ho   |                                      |                 |                         |
| Email:     | amelia.tan@db.com / vincent.ho@db   | .com                                 |                 |                         |
|            |   | - 304 -                              |                 |                         |

## MERRILL LYNCH INTERNATIONAL

| By:                     | /s/ Anne Barraclough  |  |
|-------------------------|---|--|
| Address:                | Bank of America Merrill Lynch Financial Centre<br>2 King Edward Street<br>London EC1A 1HQ<br>United Kingdom |  |
| Tel:                    | +(44-20) 7995-5648  |  |
| Fax:                    | +(44-20) 7996-2995  |  |
| Attention:              | Client Integration and Documentation Group  |  |
| with a copy to:         |   |  |
| with a copy             | to:   |  |
| with a copy<br>Address: | to:<br>Merrill Lynch International<br>26 Elmfield Road<br>Bromley<br>Kent BR1 1WA<br>United Kingdom         |  |
| 10                      | Merrill Lynch International<br>26 Elmfield Road<br>Bromley<br>Kent BR1 1WA                                  |  |
| Address:                | Merrill Lynch International<br>26 Elmfield Road<br>Bromley<br>Kent BR1 1WA<br>United Kingdom                |  |

Attention: Swaps Operations

- 305 -

# SOCIÉTÉ GÉNÉRALE

| By:        | /s/ Patricia Wright   |
|------------|---|
| Address:   | 1221 Avenue of the Americas, 10th Floor<br>New York<br>NY 10020 |
| Tel:       | +1 212 278 6335   |
| Fax:       | +1 212 278 7614   |
| Attention: | Patricia Wright   |

- 306 -

### The Outgoing Global Coordinating Lead Arrangers

# BANC OF AMERICA SECURITIES ASIA LIMITED

By: /s/ Chin Voon Fat Frederick

# DEUTSCHE BANK AG, HONG KONG BRANCH

By: /s/ Elizabeth Morgan

Elizabeth Morgan Director /s/ Michael Lam

Michael Lam Director

- 308 -

# SG AMERICAS SECURITIES, LLC

By: /s/ Meaghan Black

Meaghan Black

# The Incoming Global Coordinating Lead Arrangers

# BANCO NACIONAL ULTRAMARINO, S.A.

| By:        | /s/ Pedro Manuel de Oliveira Cardoso<br>Pedro Manuel de Oliveira Cardoso | /s/ Chiu Ha Evonne<br>Chiu Ha Evonne |
|------------|--|--------------------------------------|
| Address:   | No. 22, Avenida de Almeida Ribeiro,<br>Macau                             |                                      |
| Fax:       | +853 2835 5653 / +853 2835 6867  |                                      |
| Attention: | Ms. Monica Wong / Ms. Violet Choi  |                                      |
|            |  | 210                                  |

- 310 -

# BANK OF AMERICA, N.A.

| By:      | /s/ Chin Voon Fat Frederick                                  |
|----------|--|
| Address: | 10/F Devon House<br>979 King's Road<br>Quarry Bay, Hong Kong |
| Fax:     | +852 2847 3914   |

Attention: Danny Chu

- 311 -

# BANK OF CHINA LIMITED, MACAU BRANCH

| Bu  | /s/ Leong Sie Hong |
|-----|--------------------|
| By: | /s/ Leong Sio Hong |

| Bank of China Building      |  |
|-----------------------------|--|
| Avenida Doutor Mário Soares |  |
| Macau                       |  |
|                             |  |

Fax: +853 8792 1659 / +853 8792 0308

Attention: Mr. James Wong / Ms. Amy Cheong

- 312 -

## DBS BANK LTD.

| By:      | /s/ Wong Hon Keung, Benjamin<br>Wong Hon Keung, Benjamin            |
|----------|---|
| Address: | 18/F., The Center,<br>99 Queen's Road Central<br>Central, Hong Kong |

Fax: +852 2596 0577

Attention: Mr. Colum Ting / Andy Chan

# DEUTSCHE BANK AG, SINGAPORE BRANCH

| By:      | /s/ Elizabeth Morgan                         | /s/ Deepak Dangayach         |
|----------|--|------------------------------|
|          | Elizabeth Morgan<br>Director                 | Deepak Dangayach<br>Director |
| Address: | One Raffles Quay, #14-00<br>Singapore 048593 |                              |
| Fax:     | +65 65361328 / +65 62212306                  |                              |

Attention Vvonne Choo / Hee-Jin Kim / Kenneth Ting

- 314 -

### INDUSTRIAL AND COMMERCIAL BANK OF CHINA (MACAU) LIMITED

By: /s/ Mr. Cheng Wing Fai

/s/ Mr. Chan Iat Meng

- Address: 18/F., ICBC Tower Macau Landmark 555 Avenida da Amizade, Macau
- Fax: +853 2858 4496

Attention: Ms. Linda Chan / Mr. David Chan

- 315 -

# JPMORGAN CHASE BANK, N.A., HONG KONG BRANCH

| By:      | /s/ William Frank Agee   | William Frank Agee<br>Managing Director |
|----------|--|---|
| Address: | 8 Connaught Road, Floor 20<br>Central, Hong Kong S.A.R<br>People's Republic of China |   |

Fax: +91 22 6646 6865

Attention: Sneha Prasad / Neha M Kumar

- 316 -

# STANDARD CHARTERED BANK (HONG KONG) LIMITED

| By:      | /s/ Eric Chan  |
|----------|--|
| Address: | 12/F, Two International Finance Centre<br>8 Finance Street<br>Central, Hong Kong |
| Fax:     | + 852 3983 8604  |

Attention: Eric Chan

## THE BANK OF NOVA SCOTIA

| By:             | /s/ Andy Poon   |  |
|-----------------|---|--|
| Address:        | 25 <sup>th</sup> Floor United Centre<br>95 Queensway<br>Hong Kong |  |
| Fax:            | +852 2527 2527  |  |
| Attention:      | Ms Phyllis Leung  |  |
| With a copy to: |   |  |
| Address:        | The Bank of Nova Scotia, Singapo                                  |  |

Address: The Bank of Nova Scotia, Singapore Branch One Raffles Quay, #20-01 North Tower One Raffles Quay Singapore 048583

Fax: +65 65347817

Attention: Loo Jee Ken

- 318 -

# THE ROYAL BANK OF SCOTLAND PLC, SINGAPORE BRANCH

| By:      | /s/ Raghu Narain  | Managing Director |
|----------|---|-------------------|
| Address: | 30 Floor, AIA Central<br>1 Connaught Road<br>Central, Hong Kong |                   |

Fax: +65 6517 5037

Attention: Ms Kate Chua

- 319 -

### UBS AG HONG KONG BRANCH

| By:      | /s/ Eric Sim   | /s/ Mohamed Atmani |
|----------|--|--------------------|
| Address: | 50/F, Two International Finance Centre<br>8 Finance Street<br>Central, Hong Kong |                    |
| Fax:     | +852 3712 4979   |                    |

Attention: Mohamed Atmani

- 320 -

### UNITED OVERSEAS BANK LIMITED

| By:      | /s/ Mr. Chow Yew Hon  |
|----------|---|
| Address: | 25/F., Gloucester Tower<br>The Landmark<br>15 Queen's Road Central<br>Hong Kong |

Fax: +852 2501 5738

Attention: Wanna So / Stephen Cheung / Simmy Law / Terry Wong

- 321 -

### The Outgoing Hotel Facility Agent

# SOCIÉTÉ GÉNÉRALE, HONG KONG BRANCH

| By:        | /s/ Kenneth Choi  | Kenneth Choi                    |
|------------|---|---------------------------------|
| Address:   | Level 34, Three Pacific Place<br>1 Queen's Road East<br>Hong Kong                         |                                 |
| Tel:       | (852) 2166 5671 / (852) 2166 5665   |                                 |
| Fax:       | (852) 2166 4631   |                                 |
| Attention: | Sunny Lui ( <u>sunny.lui@sgcib.com</u> ) / Michael Poon ( <u>Michael.poon@sgcib.com</u> ) |                                 |
| Copy to:   | Société Générale, Hong Kong Branch  |                                 |
|            | c/o Société Générale Global Solution  | n Centre                        |
| Tel:       | (852) 2166 5182   |                                 |
| Fax:       | (852) 2868 1874   |                                 |
| Attention: | Mr. Parag Gautam / Mr. Sandeep Ra   | vi - Asia Loan Operation Centre |

Attention: Mr. Parag Gautam / Mr. Sandeep Ravi - Asia Loan Operation Centre (<u>hkg-loan-ops@sgcib.com</u>)

- 322 -

### The Outgoing Project Facility Agent

# SOCIÉTÉ GÉNÉRALE, HONG KONG BRANCH

| By:        | /s/ Kenneth Choi  | Kenneth Choi                    |
|------------|---|---------------------------------|
| Address:   | Level 34, Three Pacific Place<br>1 Queen's Road East<br>Hong Kong                         |                                 |
| Tel:       | (852) 2166 5671 / (852) 2166 5665   |                                 |
| Fax:       | (852) 2166 4631   |                                 |
| Attention: | Sunny Lui ( <u>sunny.lui@sgcib.com</u> ) / Michael Poon ( <u>Michael.poon@sgcib.com</u> ) |                                 |
| Copy to:   | Société Générale, Hong Kong Branch  |                                 |
|            | c/o Société Générale Global Solution  | n Centre                        |
| Tel:       | (852) 2166 5182   |                                 |
| Fax:       | (852) 2868 1874   |                                 |
| Attention  | Mr. Parag Gautam / Mr. Sandeep Ra   | vi - Asia Loan Operation Centre |

Attention: Mr. Parag Gautam / Mr. Sandeep Ravi - Asia Loan Operation Centre (<u>hkg-loan-ops@sgcib.com</u>)

- 323 -

### The Outgoing Revolving Credit Facility Agent

# SOCIÉTÉ GÉNÉRALE, HONG KONG BRANCH

| By:        | /s/ Kenneth Choi  | Kenneth Choi                  |
|------------|---|-------------------------------|
| Address:   | Level 34, Three Pacific Place<br>1 Queen's Road East<br>Hong Kong                         |                               |
| Tel:       | (852) 2166 5671 / (852) 2166 5665   |                               |
| Fax:       | (852) 2166 4631   |                               |
| Attention: | Sunny Lui ( <u>sunny.lui@sgcib.com</u> ) / Michael Poon ( <u>Michael.poon@sgcib.com</u> ) |                               |
| Copy to:   | Société Générale, Hong Kong Branch  |                               |
|            | c/o Société Générale Global Solution  | n Centre                      |
| Tel:       | (852) 2166 5182   |                               |
| Fax:       | (852) 2868 1874   |                               |
| Attention  | Mr. Darag Cautam / Mr. Sandoon Da   | vi Asia Loan Operation Contro |

Attention: Mr. Parag Gautam / Mr. Sandeep Ravi - Asia Loan Operation Centre (<u>hkg-loan-ops@sgcib.com</u>)

- 324 -

### The Outgoing Intercreditor Agent

# SOCIÉTÉ GÉNÉRALE, HONG KONG BRANCH

| By:        | /s/ Kenneth Choi  | Kenneth Choi                    |
|------------|---|---------------------------------|
| Address:   | Level 34, Three Pacific Place<br>1 Queen's Road East<br>Hong Kong                         |                                 |
| Tel:       | (852) 2166 5671 / (852) 2166 5665   |                                 |
| Fax:       | (852) 2166 4631   |                                 |
| Attention: | Sunny Lui ( <u>sunny.lui@sgcib.com</u> ) / Michael Poon ( <u>Michael.poon@sgcib.com</u> ) |                                 |
| Copy to:   | Société Générale, Hong Kong Branch  |                                 |
|            | c/o Société Générale Global Solution  | n Centre                        |
| Tel:       | (852) 2166 5182   |                                 |
| Fax:       | (852) 2868 1874   |                                 |
| Attention: | Mr. Parag Gautam / Mr. Sandeep Ra   | vi - Asia Loan Operation Centre |

Attention: Mr. Parag Gautam / Mr. Sandeep Ravi - Asia Loan Operation Centre (<u>hkg-loan-ops@sgcib.com</u>)

- 325 -

### The Outgoing Security Agent

# SOCIÉTÉ GÉNÉRALE, HONG KONG BRANCH

| By:        | /s/ Kenneth Choi  | Kenneth Choi                    |
|------------|---|---------------------------------|
| Address:   | Level 34, Three Pacific Place<br>1 Queen's Road East<br>Hong Kong                         |                                 |
| Tel:       | (852) 2166 5671 / (852) 2166 5665   |                                 |
| Fax:       | (852) 2166 4631   |                                 |
| Attention: | Sunny Lui ( <u>sunny.lui@sgcib.com</u> ) / Michael Poon ( <u>Michael.poon@sgcib.com</u> ) |                                 |
| Copy to:   | Société Générale, Hong Kong Branch  |                                 |
|            | c/o Société Générale Global Solutio   | n Centre                        |
| Tel:       | (852) 2166 5182   |                                 |
| Fax:       | (852) 2868 1874   |                                 |
| Attention: | Mr. Parag Gautam / Mr. Sandeep Ra   | vi - Asia Loan Operation Centre |

Attention: Mr. Parag Gautam / Mr. Sandeep Ravi - Asia Loan Operation Centre (<u>hkg-loan-ops@sgcib.com</u>)

- 326 -

### The Incoming Term Facility Agent

# BANK OF CHINA LIMITED, MACAU BRANCH

| By:        | /s/ Leong Sio Hong  |
|------------|---|
| Address:   | 13/F Bank of China Building<br>Avenida Doutor Mario Soares, Macau |
| Tel:       | +853 8792 1639 / 8792 1682  |
| Fax:       | +853 8792 1659 / 8792 0308  |
| Attention: | Mr. James Wong / Ms. Amy Cheong                                   |
| Copy to:   |   |
| Address:   | 17/F Bank of China Building<br>Avenida Doutor Mario Soares, Macau |
| Tel:       | +853 8792 1623 / 8792 1706  |
| Fax:       | +853 8792 1677  |

Attention: Ms. Wendy Sun / Mr. Jerry Chan

- 327 -

### The Incoming Revolving Credit Facility Agent

# BANK OF CHINA LIMITED, MACAU BRANCH

| By:        | /s/ Leong Sio Hong  |
|------------|---|
| Address:   | 13/F Bank of China Building<br>Avenida Doutor Mario Soares, Macau |
| Tel:       | +853 8792 1639 / 8792 1682  |
| Fax:       | +853 8792 1659 / 8792 0308  |
| Attention: | Mr. James Wong / Ms. Amy Cheong                                   |
| Copy to:   |   |
| Address:   | 17/F Bank of China Building<br>Avenida Doutor Mario Soares, Macau |
| Tel:       | +853 8792 1623 / 8792 1706  |
| Fax:       | +853 8792 1677  |

Attention: Ms. Wendy Sun / Mr. Jerry Chan

- 328 -

### The Incoming Intercreditor Agent

# BANK OF CHINA LIMITED, MACAU BRANCH

| By:        | /s/ Leong Sio Hong  |
|------------|---|
| Address:   | 13/F Bank of China Building<br>Avenida Doutor Mario Soares, Macau |
| Tel:       | +853 8792 1639 / 8792 1682  |
| Fax:       | +853 8792 1659 / 8792 0308  |
| Attention: | Mr. James Wong / Ms. Amy Cheong                                   |
| Copy to:   |   |
| Address:   | 17/F Bank of China Building<br>Avenida Doutor Mario Soares, Macau |
| Tel:       | +853 8792 1623 / 8792 1706  |
| Fax:       | +853 8792 1677  |

Attention: Ms. Wendy Sun / Mr. Jerry Chan

- 329 -

### The Incoming Security Agent

# BANK OF CHINA LIMITED, MACAU BRANCH

| By:        | /s/ Leong Sio Hong  |
|------------|---|
| Address:   | 13/F Bank of China Building<br>Avenida Doutor Mario Soares, Macau |
| Tel:       | +853 8792 1639 / 8792 1682  |
| Fax:       | +853 8792 1659 / 8792 0308  |
| Attention: | Mr. James Wong / Ms. Amy Cheong                                   |
| Copy to:   |   |
| Address:   | 17/F Bank of China Building<br>Avenida Doutor Mario Soares, Macau |
| Tel:       | +853 8792 1623 / 8792 1706  |
| Fax:       | +853 8792 1677  |

Attention: Ms. Wendy Sun / Mr. Jerry Chan

- 330 -

#### The Second Ranking Finance Party and Outgoing POA Agent

Executed as a deed by BANCO NACIONAL ULTRAMARINO, S.A. acting by Pedro Manuel de Oliveira Cardoso and Chiu Ha Evonne in the presence of: Violet Chui Po Lin

Name of Witness: Violet Chui Po Lin Occupation of Witness: Bank staff Address of Witness: Av. Almeida Ribeiro, 22, Macau

Address: Avenida de Almeida Ribeiro, No. 22 Macau

## Tel: +(853) 83989184 / (853) 83989106

Fax: +(853) 28331206/ (853) 28356867

Attention: Ms. Evonne Chiu / Ms. Violet Choi

) ) /s/ Pedro Manuel de Oliveira Cardoso ) /s/ Chiu Ha Evonne ) /s/ Violet Chui Po Lin

)

- 331 -

### The Incoming POA Agent

# BANK OF CHINA LIMITED, MACAU BRANCH

| By:        | /s/ Leong Sio Hong  |
|------------|---|
| Address:   | 13/F Bank of China Building<br>Avenida Doutor Mario Soares, Macau |
| Tel:       | +853 8792 1639 / 8792 1682  |
| Fax:       | +853 8792 1659 / 8792 0308  |
| Attention: | Mr. James Wong / Ms. Amy Cheong                                   |
| Copy to:   |   |
| Address:   | 17/F Bank of China Building<br>Avenida Doutor Mario Soares, Macau |
| Tel:       | +853 8792 1623 / 8792 1706  |
| Fax:       | +853 8792 1677  |

Attention: Ms. Wendy Sun / Mr. Jerry Chan

- 332 -

Exhibit 10.2

CLIFFORD CHANCE 高偉紳律師行

EXECUTION VERSION

DATED July 31, 2012

WYNN RESORTS (MACAU) S.A. as Company

BANK OF CHINA LIMITED, MACAU BRANCH as Revolving Credit Facility Agent

and

THE REVOLVING CREDIT FACILITY LENDERS referred to herein

REVOLVING CREDIT FACILITY AGREEMENT

LIFFORD С

CHANCE

- 1 -

Page

| Claus  | Clause   |  |  |  |
|--|--|--|--|--|
| 1.   | Definitions And Interpretation   |  |  |  |
| 2.   | Common Terms Agreement   |  |  |  |
| 3.   | The Revolving Credit Facilities  |  |  |  |
| 4.   | Purpose  |  |  |  |
| 5.   | Conditions Of Utilisation  |  |  |  |
| 6.   | Availability Of The Revolving Credit Facilities                            |  |  |  |
| 7.   | Repayment  |  |  |  |
| 8.   | Cancellation   |  |  |  |
| 9.   | Interest   |  |  |  |
| 10.  | Interest Periods   |  |  |  |
| 11.  | Notification   |  |  |  |
| 12.  | Commitment Fee   |  |  |  |
| 13.  | Changes To The Parties   |  |  |  |
| 14.  | Payments   |  |  |  |
| 15.  | Decision Making Amongst Revolving Credit Facility Lenders                  |  |  |  |
| 16.  | Counterparts   |  |  |  |
| 17.  | Governing Law  |  |  |  |
| 18.  | Jurisdiction   |  |  |  |
| Schedule 1 The Revolving Credit Facility Lenders |  |  |  |  |
| Sche   | Schedule 2 Form Of Advance Request For A Revolving Credit Facility Advance |  |  |  |

#### BETWEEN:

(1) WYNN RESORTS (MACAU) S.A. (the "Company");

(2) BANK OF CHINA LIMITED, MACAU BRANCH (the "Revolving Credit Facility Agent"); and

(3) THE FINANCIAL INSTITUTION listed in Schedule 1 (*The Revolving Credit Facility Lenders*) as lender (the "Revolving Credit Facility Lender").

#### WHEREAS:

The Revolving Credit Facility Lender has agreed to make certain loan facilities available to the Company in connection with the Projects and the Cotai Project (including the design, development and construction of the Cotai Project), for the refinancing of the Company's existing indebtedness and for general corporate purposes of the Group (including investment in Excluded Subsidiaries, Excluded Projects or Resort Management Agreements), in each case, upon the terms and subject to the conditions set out in this Agreement and the Common Terms Agreement.

#### IT IS AGREED as follows:

### 1. DEFINITIONS AND INTERPRETATION

#### 1.1 **Definitions**

In this Agreement, unless otherwise defined herein, all terms defined or referred in the Common Terms Agreement shall have the same meaning herein and in addition:

"Advance Request" means a request for a Revolving Credit Facility Advance in substantially the form set out in Schedule 2 (*Form of Advance Request for a Revolving Credit Facility Advance*).

"Available Commitment" means, in relation to a Revolving Credit Facility Lender at any time and save as otherwise provided herein, the aggregate US dollar equivalent amount of Available Tranche A Commitment and Available Tranche B Commitment of such Revolving Credit Facility Lender.

"Available Facility" means, at any time, the aggregate US dollar equivalent amount of the Available Tranche A Facility and the Available Tranche B Facility.

"Available Tranche A Commitment" means, in relation to a Revolving Credit Facility Lender at any time, its Tranche A Facility Commitment less:

- (a) the aggregate amount of its participation in any outstanding Tranche A Advances (other than, in relation to any proposed Tranche A Advance, such Revolving Credit Facility Lender's participation in any Tranche A Advances that are due to be repaid or prepaid on or before the proposed Advance Date); and
- (b) in relation to any proposed Tranche A Advance, the aggregate amount of its participation in any Tranche A Advances that are due to be made on or before the proposed Advance Date.

- 1 -

"Available Tranche B Commitment" means, in relation to a Revolving Credit Facility Lender at any time, its Tranche B Facility Commitment less:

- (a) the aggregate amount of its participation in any outstanding Tranche B Advances (other than, in relation to any proposed Tranche B Advance, such Revolving Credit Facility Lender's participation in any Tranche B Advances that are due to be repaid or prepaid on or before the proposed Advance Date); and
- (b) in relation to any proposed Tranche B Advance, the aggregate amount of its participation in any Tranche B Advances that are due to be made on or before the proposed Advance Date.

"Available Tranche A Facility" means, at any time, the aggregate amount of the Available Tranche A Commitments of all the Revolving Credit Facility Lenders at such time.

"Available Tranche B Facility" means, at any time, the aggregate amount of the Available Tranche B Commitments of all the Revolving Credit Facility Lenders at such time.

"Common Terms Agreement" means the common terms agreement dated 14 September 2004 and made between, among others, the Company, the financial institutions defined therein as Term Facility Lenders and Revolving Credit Facility Lenders, the Term Facility Agent, the Revolving Credit Facility Agent, the Intercreditor Agent and the Security Agent, as amended and restated by the Common Terms Agreement Amendment Agreement, the Common Terms Agreement Second Amendment Agreement, the Common Terms Agreement Fourth Amendment Agreement.

"HIBOR" means, in relation to any Tranche B Advance:

- (a) the applicable Screen Rate; or
- (b) (if no Screen Rate is available for HK dollars or the Interest Period of that Revolving Credit Facility Advance) the arithmetic mean of the rates (rounded upwards to four decimal places) as supplied to the Revolving Credit Facility Agent at its request quoted by the Reference Banks to leading banks in the Hong Kong interbank market,

at or about 11.00 am (Hong Kong time) on the Quotation Day for the offering of deposits in HK dollars for a period comparable to the Interest Period for that Revolving Credit Facility Advance (or, if such Screen Rate or arithmetic mean of rates, as the case may be, is less than zero per annum, zero per annum).

"Interest Period" means, in relation to a Revolving Credit Facility Advance, each period determined in accordance with Clause 10 (Interest Periods).

"LIBOR" means, in relation to any Tranche A Advance:

- (a) the applicable Screen Rate; or
- (b) (if no Screen Rate is available for US dollars or for the Interest Period for that Revolving Credit Facility Advance) the arithmetic mean of the rates (rounded upwards to four decimal places) as supplied to the Revolving Credit Facility Agent at its request quoted by the Reference Banks to leading banks in the London interbank market,

- 2 -

at or about 11:00 a.m. (London time) on the Quotation Day for the offering of deposits in US dollars and for a period comparable to the Interest Period for that Revolving Credit Facility Advance (or, if such Screen Rate or arithmetic mean of rates, as the case may be, is less than zero per annum, zero per annum).

"**Majority Revolving Credit Facility Lenders**" means a Revolving Credit Facility Lender or Revolving Credit Facility Lenders whose US dollar equivalent participations in the Revolving Credit Facility Advances then outstanding and undrawn Available Commitments amount in aggregate to more than 50% of the US dollar equivalent of the sum of all Revolving Credit Facility Advances then outstanding and undrawn Available Commitments.

"Margin" means in relation to any Revolving Credit Facility Advance hereunder:

- (a) on and from the Fourth Amendment Effective Date to and including the earlier of (i) the Quarterly Date falling on the last day of the second full Fiscal Quarter of the Company following the Fourth Amendment Effective Date and (ii) the date which is 180 days after the Fourth Amendment Effective Date, 2.50% per annum; and
- (b) thereafter, 1.75% per annum but, if the Leverage Ratio as at the most recent Quarterly Date thereafter is within the range set out below and the Intercreditor Agent has received, in accordance with paragraphs 1 and 2 of Part A of Schedule 5 (*Covenants*) of the Common Terms Agreement, the Company's financial statements for the period ending on such Quarterly Date together with the Compliance Certificate required thereunder then, provided (in the case of any decrease in the Margin) no Default has occurred and is continuing, the Margin will be the percentage per annum specified for that range:

| Leverage Ratio                                 | Margin |
|--|--------|
| Less than 3.0                                  | 1.75%  |
| Greater than or equal to 3.0 but less than 4.0 | 2.00%  |
| Greater than or equal to 4.0 but less than 4.5 | 2.25%  |
| 4.5 or above                                   | 2.50%  |

Any increase or decrease in the Margin shall take effect from the Business Day following the satisfaction of the conditions or expiry of the applicable periods specified above (or, where such Business Day falls less than five Business Days before the end of the then current Interest Period, from the commencement of the next Interest Period).

"Reference Banks" means, in relation to:

- (a) LIBOR, the principal London offices of Deutsche Bank AG, Standard Chartered Bank and The Royal Bank of Scotland plc; and
- (b) HIBOR, the principal Hong Kong offices of Bank of China Limited, Hong Kong Branch, Industrial and Commercial Bank of China (Asia) Limited and Standard Chartered Bank (Hong Kong) Limited,

- 3 -

or such other bank or banks designated from time to time by the Revolving Credit Facility Agent provided that the consent of the Company shall be required if such designation is made prior to an occurrence of an Event of Default which is continuing.

"Revolving Credit Facility" means the Tranche A Facility or the Tranche B Facility and "Revolving Credit Facilities" means each of them.

"**Revolving Credit Facility Advance**" means, as the context may require, a Tranche A Advance or a Tranche B Advance and "**Revolving Credit Facility Advances**" shall mean each Tranche A Advance and Tranche B Advance or any of them.

### "Revolving Credit Facility Finance Documents" means:

- (a) this Agreement;
- (b) the Common Terms Agreement;
- (c) any other Senior Finance Document to which a Revolving Credit Facility Lender is a party in its capacity as a Revolving Credit Facility Lender; and
- (d) any other document designated as such by the Revolving Credit Facility Agent and the Company.

**"Revolving Credit Facility Lender"** means a Tranche A Facility Lender or a Tranche B Facility Lender.

"Revolving Credit Finance Parties" means the Revolving Credit Facility Agent and the Revolving Credit Facility Lenders.

"**Rollover Advance**", in relation to a Revolving Credit Facility, means one or more Revolving Credit Facility Advances under that Revolving Credit Facility:

- (a) made or to be made on the same day that a maturing Revolving Credit Facility Advance under that Revolving Credit Facility is due to be repaid;
- (b) the aggregate amount of which is equal to or less than the maturing Revolving Credit Facility Advance under that Revolving Credit Facility; and
- (c) made or to be made to the Company for the purpose of refinancing a maturing Revolving Credit Facility Advance under that Revolving Credit Facility.

"Screen Rate" means, in relation to:

- (a) LIBOR, the British Bankers' Association Interest Settlement Rate for US dollars for the relevant period, displayed on the appropriate page (being currently "LIBOR01") of the Reuters Monitor Money Rates Service screen; and
- (b) HIBOR, the rate designated as "FIXING@11:00" (or any other designation which may from time to time replace that designation or, if no such designation appears, the arithmetic average (rounded upwards, to four decimal places) of the displayed rates for the relevant period) appearing under the heading "HONG KONG INTERBANK OFFERED RATES (HK DOLLAR)" on the Reuters Screen HIBOR1=R Page.

- 4 -

If the agreed page is replaced or service ceases to be available, the Revolving Credit Facility Agent may specify another page or service displaying the appropriate rate after consultation with the Company and the Revolving Credit Facility Lenders.

"Termination Date" means the fifth anniversary of the Fourth Amendment Signing Date.

"**Tranche A Advance**" means an advance made or to be made by the Tranche A Facility Lenders under the Tranche A Facility or the principal amount outstanding for the time being of that advance including a Rollover Advance.

"Tranche A Facility" means the US dollar revolving credit facility granted to the Company under Clause 3.1.1 (Tranche A Facility).

#### "Tranche A Facility Commitment" means

- (a) in relation to the Original Tranche A Facility Lender, the amount set opposite its name under the column entitled "Tranche A Commitment" in Schedule 1 (*The Revolving Credit Facility Lenders*) and the amount of any other Tranche A Facility Commitment transferred to it under this Agreement; and
- (b) in relation to any other Tranche A Facility Lender, the amount of any Tranche A Facility Commitment transferred to it under this Agreement,

to the extent not cancelled pursuant to Clause 8 (*Repayments, Prepayments and Cancellation*) of the Common Terms Agreement or otherwise reduced pursuant to the terms hereof and/or the Common Terms Agreement, reduced or transferred by it in accordance with the Common Terms Agreement.

"Tranche A Facility Lender" means any commercial bank, financial institution or other entity which:

- (a) is named in Schedule 1 (*The Revolving Credit Facility Lenders*) as a Tranche A Facility Lender (the "**Original Tranche A Facility Lender**"); or
- (b) has become party hereto as a Tranche A Facility Lender in accordance with Clause 13 (*Changes to the Parties*),

and which has not ceased to be a party hereto in accordance with the terms hereof.

"**Tranche B Advance**" means an advance made or to be made by the Tranche B Facility Lenders under the Tranche B Facility or the principal amount outstanding for the time being of that advance including a Rollover Advance.

- 5 -

"Tranche B Facility" means the HK dollar revolving credit facility granted to the Company under Clause 3.1.2 (Tranche B Facility).

#### "Tranche B Facility Commitment" means

- (a) in relation to the Original Tranche B Facility Lender, the amount set opposite its name under the column entitled "Tranche B Commitment" in Schedule 1 (*The Revolving Credit Facility Lenders*) and the amount of any other Tranche B Facility Commitment transferred to it under this Agreement; and
- (b) in relation to any other Tranche B Facility Lender, the amount of any Tranche B Facility Commitment transferred to it under this Agreement,

to the extent not cancelled pursuant to Clause 8 (*Repayments, Prepayments and Cancellation*) of the Common Terms Agreement or otherwise reduced pursuant to the terms hereof and/or the Common Terms Agreement, reduced or transferred by it in accordance with the Common Terms Agreement.

"Tranche B Facility Lender" means any commercial bank, financial institution or other entity which:

- (a) is named in Schedule 1 (*The Revolving Credit Facility Lenders*) as a Tranche B Facility Lender (the "**Original Tranche B Facility Lender**"); or
- (b) has become party hereto as a Tranche B Facility Lender in accordance with Clause 13 (*Changes to the Parties*),

and which has not ceased to be a party hereto in accordance with the terms hereof.

#### 1.2 Interpretation

In this Agreement:

- 1.2.1 the principles of construction contained in Clause 1.2 (*Principles of Construction*) of the Common Terms Agreement and the rules of interpretation contained in Clause 1.3 (*Rules of Interpretation*) of the Common Terms Agreement shall apply to the construction and interpretation of this Agreement;
- 1.2.2 any reference to the "**Revolving Credit Facility Agent**" or any "**Revolving Credit Facility Lender**" shall be construed so as to include its or their (and any subsequent) successors and any permitted transferees in accordance with their respective interests; and
- 1.2.3 references in this Agreement to any Clause or Schedule shall be to a clause or schedule contained in this Agreement.

### 1.3 Third Party Rights

1.3.1 The Contracts (Rights of Third Parties) Act 1999 applies to Clause 1.4 (*Non-recourse liability*) but only for the benefit of the Operatives and subject always to the terms of Clause 17 (*Governing Law*) and Clause 18 (*Jurisdiction*).

- 6 -

- 1.3.2 Except as provided in sub-clause 1.3.1 above, a Person who is not a party to this Agreement has no right under the Contracts (Rights of Third Parties) Act 1999 to enforce or to enjoy the benefit of any term of this Agreement.
- 1.3.3 Save as provided by the Common Terms Agreement, the consent of any Person who is not a party to this Agreement is not required to rescind or vary this Agreement.

### 1.4 Non-recourse Liability

Notwithstanding any provision in the Senior Finance Documents to the contrary, no Operative shall be personally liable for payments due hereunder or under any of the Senior Finance Documents or for the performance of any obligation hereunder or thereunder, save, in relation to any Operative, pursuant to any Senior Finance Document to which such Operative is party. The sole recourse of the Revolving Credit Facility Lenders for satisfaction of any of the obligations of any of the Obligors hereunder and under the other Senior Finance Documents shall be against the Obligors and not against any assets or property of any Operative, save to the extent such Operative is party to a Senior Finance Document and is expressed to be liable for such obligation thereunder. In the case of Mr. Wong Chi Seng, his liability shall be limited to his shares in the Company.

#### 2. COMMON TERMS AGREEMENT

This Agreement and the rights and obligations of the parties hereto shall be subject to the terms and conditions of the Common Terms Agreement which shall be deemed to be incorporated into this Agreement. In the case of any conflict between the terms of this Agreement and the terms of the Common Terms Agreement, the terms of this Agreement shall prevail.

### 3. THE REVOLVING CREDIT FACILITIES

#### 3.1 Grant of the Revolving Credit Facilities

### 3.1.1 Tranche A Facility

The Tranche A Facility Lenders grant to the Company, upon the terms and subject to the conditions hereof, a US dollar revolving credit facility in an aggregate amount of USD433,750,000.00.

#### 3.1.2 Tranche B Facility

The Tranche B Facility Lenders grant to the Company, upon the terms and subject to the conditions hereof, a HK dollar revolving credit facility in an aggregate amount of HKD8,706,750,000.00.

#### 4. **PURPOSE**

The Company shall apply all amounts borrowed by it under the Revolving Credit Facilities to pay for the refinancing of existing indebtedness of the Company, the design, development and construction of the Cotai Project and for the general corporate purposes of the Group (including investment in Excluded Subsidiaries, Excluded Projects or Resort Management Agreements).

- 7 -

### 5. **CONDITIONS OF UTILISATION**

#### 5.1 **Conditions precedent**

The provisions of Clause 2.1 (*Conditions Precedent to the CP Satisfaction Date*) of the Common Terms Agreement are incorporated by reference herein as if the same were set out in full herein.

#### 5.2 Further conditions precedent

The obligation of each Revolving Credit Facility Lender to participate in each Revolving Credit Facility Advance under this Agreement is subject to the Company having satisfied the following conditions:

- (a) in the case of a Rollover Advance, no Event of Default shall have occurred and is continuing and, in the case of any other Revolving Credit Facility Advance, no Default shall have occurred and is continuing;
- (b) the representations and warranties contained in Schedule 4 (*Representations and Warranties*) of the Common Terms Agreement which are repeated by the Company pursuant to Clause 17.2 (*Timing*) of the Common Terms Agreement are true and correct in all material respects with reference to the facts and circumstances existing on the Advance Date;
- (c) receipt by the Intercreditor Agent of each of the reports, financial statements and other information due pursuant to paragraphs 1 and 2 of Part A of Schedule 5 (*Covenants*) of the Common Terms Agreement on or before the proposed Advance Date; and
- (d) the Company shall have paid or arranged for payment out of the requested Revolving Credit Facility Advance of all fees, expenses and other charges then due and payable by it under the Revolving Credit Facility Finance Documents.

### 5.3 Maximum number of Loans

The Company may not deliver an Advance Request for a Revolving Credit Facility Advance if, as a result of the proposed Revolving Credit Facility Advance, more than ten Revolving Credit Facility Advances would be outstanding.

### 6. AVAILABILITY OF THE REVOLVING CREDIT FACILITIES

#### 6.1 Drawdown of Advances

The provisions of Clause 3 (*Drawdown of Advances*) and Clause 4.3 (*Revolving Credit Facility Availability Period*) of the Common Terms Agreement are incorporated by reference herein as if the same were set out in full herein.

### 6.2 Each Revolving Credit Facility Lender's participation

- 6.2.1 If the conditions set out in this Agreement and the Common Terms Agreement have been met, each Tranche A Facility Lender will participate through its Facility Office in each Tranche A Advance made pursuant to Clause 6.1 (*Drawdown of Advances*) in the proportion borne by its Available Tranche A Commitment to the Available Tranche A Facility immediately prior to the making of that Tranche A Advance.
- 6.2.2 If the conditions set out in this Agreement and the Common Terms Agreement have been met, each Tranche B Facility Lender will participate through its Facility Office in each Tranche B Advance made pursuant to Clause 6.1 (*Drawdown of Advances*) in the proportion borne by its Available Tranche B Commitment to the Available Tranche B Facility immediately prior to the making of that Tranche B Advance.

- 8 -

## 7. REPAYMENT

- (a) The Company shall repay each Revolving Credit Facility Advance on the last day of its Interest Period.
- (b) Subject to the other terms of this Agreement and the Common Terms Agreement, any amounts repaid (including with a Rollover Advance) under Clause 7(a) above may be re-borrowed.
- (c) Any amount of any Revolving Credit Facility Advance still outstanding on the Termination Date shall be repaid on the Termination Date.

## 8. CANCELLATION

Any cancellation of Available Commitments shall be made in accordance with this Agreement and the Common Terms Agreement (including Clause 8 (*Repayment, Prepayment and Cancellation*) and Clause 14 (*Illegality*) of the Common Terms Agreement).

### 9. INTEREST

### 9.1 Calculation of interest

The rate of interest on each Revolving Credit Facility Advance for each Interest Period is the percentage rate per annum which is the aggregate of:

- (a) the applicable Margin; and
- (b) LIBOR (in the case of a Tranche A Advance) or HIBOR (in the case of a Tranche B Advance).

## 9.2 **Payment of interest**

The Company shall pay accrued interest on each Revolving Credit Facility Advance on the last day of each Interest Period.

## 9.3 Default Interest

Default interest shall be calculated and paid in accordance with Clause 9.4 (Default Interest) of the Common Terms Agreement.

## 9.4 Notification of rates of interest

The Revolving Credit Facility Agent shall promptly notify the Company and the Revolving Credit Facility Lenders of each determination of LIBOR and HIBOR under this Agreement.

# 9.5 Changes to interest rates

The Revolving Credit Facility Agent shall promptly notify the Company and the Intercreditor Agent of any change to any interest rate occasioned by the operation of Clause 10 (*Changes to the calculation of interest*) of the Common Terms Agreement.

- 9 -

## 10. INTEREST PERIODS

The duration of each Interest Period shall be determined in accordance with Clause 9.3 (Interest Periods) of the Common Terms Agreement.

### 11. NOTIFICATION

Promptly, and in any event, not less than 4 Business Days before the proposed Advance Date for each Revolving Credit Facility Advance, the Revolving Credit Facility Agent shall notify each of the relevant Revolving Credit Facility Lenders of the proposed amount of the relevant Revolving Credit Facility Advance and the aggregate principal amount of the relevant Revolving Credit Facility Advance allocated to such Revolving Credit Facility Lender pursuant to Clause 6.2 (*Each Revolving Credit Facility Lender's participation*) and each Revolving Credit Facility Lender shall, on such Advance Date, subject to the terms and conditions of this Agreement, make available to the Revolving Credit Facility Agent for the account of the Company its said portion of such Revolving Credit Facility Advance.

## 12. **COMMITMENT FEE**

- 12.1 The Company shall pay to each Revolving Credit Facility Lender a fee computed.
  - (a) on and from the Fourth Amendment Effective Date to and including the earlier of (i) the Quarterly Date falling on the last day of the second full Fiscal Quarter of the Company following the Fourth Amendment Effective Date and (ii) the date which is 180 days after the Fourth Amendment Effective Date, at the rate of 0.8750% per annum (being 35% of the Margin for the applicable period); and
  - (b) thereafter, 0.6125% per annum but, if the Leverage Ratio as at the most recent Quarterly Date thereafter is within the range set out below and the Intercreditor Agent has received, in accordance with paragraphs 1 and 2 of Part A of Schedule 5 (*Covenants*) of the Common Terms Agreement, the Company's financial statements for the period ending on such Quarterly Date together with the Compliance Certificate required thereunder then, provided (in the case of any decrease in the rate set out below) no Default has occurred and is continuing, the fee will be computed at the percentage rate per annum specified for that range (being 35% of the Margin for the applicable period):

| Leverage Ratio                                 | Rate (% per annum) |
|--|--------------------|
| Less than 3.0                                  | 0.6125%            |
| Greater than or equal to 3.0 but less than 4.0 | 0.7000%            |
| Greater than or equal to 4.0 but less than 4.5 | 0.7875%            |
| 4.5 or above                                   | 0.8750%            |

in each case, on that Revolving Credit Facility Lender's daily Available Commitment from the Effective Date to the end of the Revolving Credit Facility Availability Period. Any increase or decrease in such fee shall take effect from the Business Day following the satisfaction of the conditions or expiry of the applicable periods specified above.

12.2 The accrued commitment fee is payable (in US dollars in respect of accrued commitment fee payable on a Revolving Credit Facility Lender's Available Tranche A Commitment and in

- 10 -

HK dollars in respect of accrued commitment fee payable on a Revolving Credit Facility Lender's Available Tranche B Commitment) on the last day of each successive period of three months which ends during the Revolving Credit Facility Availability Period, on the last day of the Revolving Credit Facility Availability Period and, if cancelled in full or part, on the cancelled amount of the relevant Revolving Credit Facility Lender's Available Commitment, immediately prior to the time the cancellation is effective.

## 13. CHANGES TO THE PARTIES

### 13.1 Transfers by the Revolving Credit Facility Agent

The Revolving Credit Facility Agent may resign in accordance with the Common Terms Agreement and may assign and transfer all of its rights and obligations under the Revolving Credit Facility Finance Documents to a replacement Revolving Credit Facility Agent appointed in accordance with the terms of the Common Terms Agreement.

### 13.2 Transfers by the Company

The Company may not assign, transfer, novate or dispose of any of its rights or obligations under the Revolving Credit Facility Finance Documents.

## 13.3 Transfers by the Revolving Credit Facility Lenders

A Revolving Credit Facility Lender may assign, transfer or novate any of its rights and/or obligations under the Revolving Credit Facility Finance Documents in accordance with Clause 21.4 (*Assignment and Transfer by Lenders*), Clause 21.5 (*Assignments by Lenders*) and Clause 21.6 (*Transfers by Lenders*) of the Common Terms Agreement.

## 13.4 Assignment and Transfer Fees

On the date upon which an assignment takes effect pursuant to Clause 21.5 (*Assignments by Lenders*) of the Common Terms Agreement or a transfer takes effect pursuant to Clause 21.6 (*Transfers by Lenders*) of the Common Terms Agreement, the relevant assignee or Transferee shall pay to the Intercreditor Agent for its own account a fee in accordance with Clause 21.7 (*Assignment and Transfer Fees*) of the Common Terms Agreement.

### 14. PAYMENTS

### 14.1 **Payments**

- 14.1.1 All payments under this Agreement shall be made in accordance with Clause 26 (*Payment Mechanics*) of the Common Terms Agreement.
- 14.1.2 Subject to Clause 26 (*Payment Mechanics*) of the Common Terms Agreement, on each date on which this Agreement requires an amount to be paid by the Company or a Revolving Credit Facility Lender, the Company or, as the case may be, such Revolving Credit Facility Lender, shall make the same available to the Revolving Credit Facility Agent for value on such due date and at such time and in such funds and to such account with such bank as the Revolving Credit Facility Agent shall specify from time to time.

### 14.2 Partial Payments

- 14.2.1 If the Revolving Credit Facility Agent receives a payment that is insufficient to discharge all the amounts then due and payable by the Company to the Revolving Credit Facility Lenders under the Revolving Credit Facility Finance Documents, the Revolving Credit Facility Agent shall apply that payment towards the obligations of the Company under the Revolving Credit Facility Finance Documents in the following order:
  - (a) **first**, in or towards payment *pro rata* of all amounts paid by the Revolving Credit Facility Lenders under Clause 23.15 (*Indemnity to Intercreditor Agent*) of the Common Terms Agreement but which have not been reimbursed by the Company;

- 11 -

- (b) **secondly**, in or towards payment *pro rata* of all amounts paid by the Revolving Credit Facility Lenders under Clause 15.3 (*Indemnity to Revolving Credit Facility Agent*) but which have not been reimbursed by the Company;
- (c) **thirdly**, in or towards payment *pro rata* of all costs and expenses incurred by the Revolving Credit Facility Lenders which the Company is obliged to reimburse;
- (d) **fourthly**, in or towards payment *pro rata* of all accrued and unpaid interest (including default interest), fees and commissions due to the Revolving Credit Facility Lenders under the Revolving Credit Facility Finance Documents;
- (e) **fifthly**, in or towards payment *pro rata* of any principal due to the Revolving Credit Facility Lenders under the Revolving Credit Facility Finance Documents but unpaid; and
- (f) **sixthly**, in or towards payment *pro rata* of any other sum due to the Revolving Credit Facility Lenders under the Revolving Credit Facility Finance Documents but unpaid.
- 14.2.2 The Revolving Credit Facility Agent shall, if so directed by the Majority Revolving Credit Facility Lenders, vary the order set out in subclause 14.2.1 above.
- 14.2.3 Sub-clause 14.2.1 above will override any appropriation made by the Company.

### 15. DECISION MAKING AMONGST REVOLVING CREDIT FACILITY LENDERS

### 15.1 Decisions

Save as otherwise set out herein and subject to the Common Terms Agreement, the required Senior Secured Creditors for the purposes of any decision within the scope of Clause 34.2 (*Amendment and waiver of Facility Agreements*) of the Common Terms Agreement relating to this Agreement shall be the Revolving Credit Facility Agent acting on the instructions of the Majority Revolving Credit Facility Lenders.

## 15.2 Failure to Give Instructions

If the Revolving Credit Facility Agent gives notice to the Revolving Credit Facility Lenders requesting their specific instructions on any matter referred to in Clause 15.1 (*Decisions*) and it specifies in such notice that the Revolving Credit Facility Lenders are to give such instructions by a certain date and time specified in such notice, any Revolving Credit Facility Lender which fails to respond by the date and time so specified shall have its portion of the Revolving Credit Facility Advances and its Available Commitment disregarded for all purposes of determining whether instructions have been given to the Revolving Credit

- 12 -

Facility Agent by the Majority Revolving Credit Facility Lenders (and, for the purposes of determining the Available Facility or the amount of all Revolving Credit Facility Advances outstanding, the Available Commitments and portion of Revolving Credit Facility Advances of such Revolving Credit Facility Lender shall be deducted).

## 15.3 Indemnity to Revolving Credit Facility Agent

- 15.3.1 Each Revolving Credit Facility Lender shall, rateably in accordance with the proportion that the US dollar equivalent of the sum of its Available Commitments and its participations in any outstanding Revolving Credit Facility Advances bear to the US dollar equivalent of the aggregate of the Available Commitments and such participations of all the Revolving Credit Facility Lenders (or, if all such amounts have been reduced to zero, such proportion determined immediately prior to such reduction) for the time being, indemnify the Revolving Credit Facility Agent, within fifteen days of demand, against any cost, loss or liability incurred by the Revolving Credit Facility Agent (other than by reason of the negligence or wilful misconduct of the Revolving Credit Facility Agent) in acting as Revolving Credit Facility Agent under any of the Senior Finance Documents (unless the Revolving Credit Facility Agent has been reimbursed by the Company pursuant to a Senior Finance Document).
- 15.3.2 Provided that the Company is required to reimburse or indemnify the Revolving Credit Facility Agent for such cost, loss or liability in accordance with the terms of the Senior Finance Documents, the Company shall, within fifteen days of demand in writing by any Revolving Credit Facility Lender, indemnify such Revolving Credit Facility Lender in relation to any payment actually made by such Revolving Credit Facility Lender pursuant to Clause 15.3.1 above.

### 16. COUNTERPARTS

This Agreement may be executed in any number of counterparts, all of which taken together shall constitute one and the same instrument.

### 17. GOVERNING LAW

This Agreement and any non-contractual obligations arising out of or in connection with it are governed by English law.

### 18. JURISDICTION

### 18.1 Jurisdiction of English courts

- 18.1.1 The courts of England have exclusive jurisdiction to settle any dispute arising out of or in connection with this Agreement (including a dispute regarding the existence, validity or termination of this Agreement or the consequences of its nullity) (a "**Dispute**").
- 18.1.2 The parties hereto agree that the courts of England are the most appropriate and convenient courts to settle Disputes and accordingly they will not argue to the contrary.
- 18.1.3 This Clause 18.1 (*Jurisdiction of English courts*) is for the benefit of the Revolving Credit Facility Lenders only. As a result, no Revolving Credit Facility Lender shall be prevented from taking proceedings relating to a Dispute in any other courts with jurisdiction. To the extent allowed by law and the Revolving Credit Facility Finance Documents, the Revolving Credit Facility Lenders may take concurrent proceedings in any number of jurisdictions.

- 13 -

## 18.2 Service of process

Without prejudice to any other mode of service allowed under any relevant law, the Company:

- (a) irrevocably appoints Law Debenture Corporate Services Limited as its agent for service of process in relation to any proceedings before the English courts in connection with this Agreement; and
- (b) agrees that failure by a process agent to notify the Company of the process will not invalidate the proceedings concerned.

This Agreement has been entered into on the date stated at the beginning of this Agreement.

# SCHEDULE 1 THE REVOLVING CREDIT FACILITY LENDERS

-

| Revolving Credit Facility Lender    | Tranche A Commitment<br>(USD) | Tranche B Commitment<br>(HKD) |
|-------------------------------------|-------------------------------|-------------------------------|
| Bank of China Limited, Macau Branch | 433,750,000                   | 8,706,750,000                 |
| Total                               | 433,750,000                   | 8,706,750,000                 |

# SCHEDULE 2

### FORM OF ADVANCE REQUEST FOR A REVOLVING CREDIT FACILITY ADVANCE

| ] as Intercreditor Agent             |
|--------------------------------------|
| ] as Revolving Credit Facility Agent |
| ]                                    |
|                                      |

### Wynn Resorts (Macau) S.A. - Revolving Credit Facility Agreement dated 14 September 2004 (the "Agreement")

## Advance Request No. [ ]

Dear Sirs,

- We refer to the Agreement and the common terms agreement (the "Common Terms Agreement") dated 14 September 2004 as amended and made between, among others, Wynn Resorts (Macau) S.A. (the "Company"), the financial institutions defined therein as Revolving Credit Facility Lenders, the Revolving Credit Facility Agent, the Intercreditor Agent and the Security Agent. Terms defined in the Common Terms Agreement shall have the same meaning herein and the principles of construction and rules of interpretation set out therein shall also apply.
- 2. This is an Advance Request given pursuant to Clause 3 (*Drawdown of Advances*) of the Common Terms Agreement.
- 3. This Advance Request is irrevocable.
- 4. We hereby give you notice that, upon the terms and subject to the conditions contained in the Common Terms Agreement and the Revolving Credit Facility Agreement, we wish to borrow the following Revolving Credit Facility Advance under the following Revolving Credit Facility on the following terms:

| Proposed Advance Date:     | [ ]                            |
|----------------------------|--------------------------------|
| Revolving Credit Facility: | [Tranche A/Tranche B] Facility |
| Amount:                    | [USD][HKD][ ]                  |
| Interest Period:           | [1, 2, 3 or 6] months          |

### 5. We confirm that:

- (a) each condition specified in Clause 5.2 (*Further conditions precedent*) of the Agreement is satisfied on the date of this Advance Request;
- (b) the Revolving Credit Facility Advance complies with the permitted use of the Revolving Credit Facilities under the Revolving Credit Facility Agreement and Clause 5 (*Purpose*) of the Common Terms Agreement; and
- (c) since the Fourth Amendment Effective Date, no Material Adverse Effect has occurred and is continuing nor could reasonably be expected to occur.

- 16 -

- 6. We attach signed but undated receipts for the Revolving Credit Facility Advance requested above and hereby authorise the Revolving Credit Facility Lenders to date such receipts on the date such Revolving Credit Facility Advances are made.<sup>1</sup>
- 7. The proceeds of the above Revolving Credit Facility Advance should be credited to [ ].

Yours faithfully,

Name: <sup>2</sup>Responsible Officer for and on behalf of **Wynn Resorts (Macau) S.A.** 

Attachments: Receipts for the Revolving Credit Facility Advance

Each Advance Request must be signed by an officer of the Company whose specimen signature has been delivered to the Intercreditor Agent pursuant to Clause 3.2.1 of the Common Terms Agreement.

- 17 -

 $<sup>\</sup>frac{1}{2}$  Each receipt must be numbered in series, the number corresponding to the number in the heading of the relevant Advance Request.

# The Company

# WYNN RESORTS (MACAU) S.A.

| By:                       | /s/ Jason M. Schall                             |  |
|---------------------------|---|--|
|                           | Jason M. Schall                                 |  |
| Address:                  | Wynn Macau, Rua Cidade de Sintra, NAPE<br>Macau |  |
| Telephone:<br>Fax:        | +853 2888 9966<br>+853 2832 9966                |  |
| Attention:                | Chief Financial Officer                         |  |
| With a copy to:           |   |  |
| Wynn Resorts (Macau) S.A. |   |  |
| Address:                  | Wynn Macau, Rua Cidade de Sintra, NAPE<br>Macau |  |
| Tel:                      | +853 2888 9966                                  |  |
| Fax:                      | +853 2832 9966                                  |  |
| Attention:                | Legal Department of Wynn Resorts (Macau) S.A.   |  |

- 18 -

# The Revolving Credit Facility Agent

# BANK OF CHINA LIMITED, MACAU BRANCH

## By:

- Address:13/F Bank of China Building<br/>Avenida Doutor Mario Soares, MacauTel:+853 8792 1639 / 8792 1682Fax:+853 8792 1659 / 8792 0308Attention:Mr. James Wong / Ms. Amy CheongCopy to:Address:17/F Bank of China Building<br/>Avenida Doutor Mario Soares, Macau
- Tel: +853 8792 1623 / 8792 1706
- Fax: +853 8792 1677

Attention: Ms. Wendy Sun / Mr. Jerry Chan

- 19 -

# The Revolving Credit Facility Lender

# BANK OF CHINA LIMITED, MACAU BRANCH

| D., |  |
|-----|--|
| DV  |  |
|     |  |

- Fax: +853 8792 1659 / 8792 0308

Attention: Mr. James Wong / Ms. Amy Cheong

- 20 -

# CLIFFORD

CHANCE

高偉紳律師行

**EXECUTION VERSION** 

## DATED July 31, 2012

## WYNN RESORTS (MACAU) S.A. as Company

## BANK OF CHINA LIMITED, MACAU BRANCH as Term Facility Agent

and

**CERTAIN FINANCIAL INSTITUTIONS** as Term Facility Lenders

TERM FACILITY AGREEMENT THIRD AMENDMENT AGREEMENT

| CONTI | ENTS |
|-------|------|
|-------|------|

|                                       | Page  |
|---------------------------------------|---|
| DEFINITIONS AND INTERPRETATION        | 1   |
| AMENDMENT                             | 1   |
| CONTINUITY AND FURTHER ASSURANCE      | 2   |
| MISCELLANEOUS                         | 2   |
| GOVERNING LAW                         | 2   |
| EDULE Amended Term Facility Agreement | 6   |
|                                       | DEFINITIONS AND INTERPRETATION<br>AMENDMENT<br>CONTINUITY AND FURTHER ASSURANCE<br>MISCELLANEOUS<br>GOVERNING LAW |

- 1 -

### THIS AGREEMENT is dated July 31, 2012 and made between:

## (1) WYNN RESORTS (MACAU) S.A. (the "Company");

## (2) **BANK OF CHINA LIMITED, MACAU BRANCH** (the "**Term Facility Agent**"); and

(3) THE FINANCIAL INSTITUTION named on the signing pages as Term Facility Lender.

#### **RECITALS:**

- (A) The Company owns and operates the Projects and proposes to design, develop and construct the Cotai Project.
- (B) The Secured Parties have agreed to amend certain Senior Finance Documents and enter into additional Senior Finance Documents in connection with the Projects and the Cotai Project (including the design, development and construction of the Cotai Project), for the refinancing of existing indebtedness and for the general corporate purposes of the Group (including investment in Excluded Subsidiaries, Excluded Projects or Resort Management Agreements).
- (C) It has been agreed to amend the Term Facility Agreement as set out below.

### IT IS AGREED as follows:

### 1. DEFINITIONS AND INTERPRETATION

### 1.1 Incorporation of defined terms

- 1.1.1 Unless a contrary indication appears, a term defined in or by reference in the Schedule has the same meaning in this Agreement.
- 1.1.2 The principles of construction and rules of interpretation set out or referred to in the Schedule shall have effect as if set out in this Agreement.
- 1.1.3 Any references in the Senior Finance Documents to the "Hotel Facility Agreement" or the "Term Facility Agreement" shall be taken to be a reference to the Term Facility Agreement as set out in the Schedule, as further amended, consolidated, supplemented, confirmed, novated or replaced from time to time.

## 1.2 Clauses

In this Agreement any reference to a "Clause" or a "Schedule" is, unless the context otherwise requires, a reference to a Clause or a Schedule to this Agreement.

### 2. AMENDMENT

With effect from the Fourth Amendment Effective Date (as defined in the Common Terms Agreement Fourth Amendment Agreement), the Term Facility Agreement shall be amended so that it shall be read and construed for all purposes as set out in the Schedule (*Amended Term Facility Agreement*).

- 1 -

## 3. CONTINUITY AND FURTHER ASSURANCE

### 3.1 **Continuing obligations**

The provisions of the Term Facility Agreement shall, save as amended by this Agreement, continue in full force and effect.

## 3.2 Further assurance

The Company shall, upon the written request of the Term Facility Agent and at its own expense, do all such acts and things reasonably necessary to give effect to the amendments effected or to be effected pursuant to this Agreement.

### 4. MISCELLANEOUS

### 4.1 Incorporation of terms

The provisions of clause 1.3 (*Third Party Rights*), clause 1.4 (*Non-recourse Liability*) and clause 18 (*Jurisdiction*) of the Schedule shall be incorporated into this Agreement as if set out in full in this Agreement and as if references in those clauses to "Agreement" are references to this Agreement and cross-references to specified clauses thereof are references to the equivalent clauses set out or incorporated herein.

## 4.2 Counterparts

This Agreement may be executed in any number of counterparts, and this has the same effect as if the signatures on the counterparts were on a single copy of this Agreement.

### 5. GOVERNING LAW

This Agreement and any non-contractual obligations arising out of or in connection with it are governed by English law.

### This Agreement has been entered into on the date stated at the beginning of this Agreement.

- 2 -

# The Company

-

# WYNN RESORTS (MACAU) S.A.

| By:                       | /S/ JASON M. SCHALL                             |  |
|---------------------------|---|--|
| Address:                  | Wynn Macau, Rua Cidade de Sintra, NAPE<br>Macau |  |
| Telephone:                | +853 2888 9966                                  |  |
| Fax:                      | +853 2832 9966                                  |  |
| Attention:                | Chief Financial Officer                         |  |
| With a copy to:           |   |  |
| Wynn Resorts (Macau) S.A. |   |  |
| Address:                  | Wynn Macau, Rua Cidade de Sintra, NAPE<br>Macau |  |
|                           | Macau   |  |
| Tel:                      | +853 2888 9966                                  |  |
| Fax:                      | +853 2832 9966                                  |  |
|                           | +055 2052 9900                                  |  |

- 3 -

## The Term Facility Agent

# BANK OF CHINA LIMITED, MACAU BRANCH

## By:

Address: 13/F Bank of China Building Avenida Doutor Mario Soares, Macau Tel: +853 8792 1639 / 8792 1682 Fax: +853 8792 1659 / 8792 0308 Attention: Mr. James Wong / Ms. Amy Cheong Copy to: 17/F Bank of China Building Address: Avenida Doutor Mario Soares, Macau Tel: +853 8792 1623 / 8792 1706 Fax: +853 8792 1677

Attention: Ms. Wendy Sun / Mr. Jerry Chan

- 4 -

## The Term Facility Lender

# BANK OF CHINA LIMITED, MACAU BRANCH

# By:

- Address:17/F Bank of China Building<br/>Avenida Doutor Mario Soares, MacauTel:+853 8792 1623 / 8792 1706Fax:+853 8792 1677Attention:Ms. Wendy Sun / Mr. Jerry ChanCopy to:13/F Bank of China Building<br/>Avenida Doutor Mario Soares, Macau
- Tel: +853 8792 1639 / 8792 1682
- Fax: +853 8792 1659 / 8792 0308

Attention: Mr. James Wong / Ms. Amy Cheong

- 5 -

## SCHEDULE AMENDED TERM FACILITY AGREEMENT

## DATED 14 SEPTEMBER 2004

# WYNN RESORTS (MACAU) S.A.

as Company

## BANK OF CHINA LIMITED, MACAU BRANCH

as Term Facility Agent

and

### THE TERM FACILITY LENDERS referred to herein

## TERM FACILITY AGREEMENT

(As amended by the Hotel Facility Agreement Amendment Agreement dated 14 September 2005, the Hotel Facility Second Amendment Agreement dated 27 June 2007 and the Term Facility Third Amendment Agreement dated July 31, 2012)

- 6 -

| Clause  | Page |
|---|------|
| 1. Definitions And Interpretation                 | 8    |
| 2. Common Terms Agreement                         | 14   |
| 3. The Term Facility                              | 14   |
| 4. Purpose  | 15   |
| 5. Conditions Precedent                           | 16   |
| 6. Availability Of The Term Facility              | 16   |
| 7. Repayment                                      | 17   |
| 8. Prepayment And Cancellation                    | 17   |
| 9. Interest                                       | 17   |
| 10. Interest Periods                              | 18   |
| 11. Notification                                  | 18   |
| 12. [Not Used]                                    | 18   |
| 13. Changes To The Parties                        | 18   |
| 14. Payments                                      | 19   |
| 15. Decision Making Amongst Term Facility Lenders | 20   |
| 16. Counterparts                                  | 21   |
| 17. Governing Law                                 | 21   |
| 18. Jurisdiction                                  | 21   |
| SCHEDULE 1 THE TERM FACILITY LENDERS              | 23   |
| SCHEDULE 2 REPAYMENT SCHEDULE                     | 24   |

- 7 -

### THIS AGREEMENT is made on 14 September 2004

### BETWEEN:

(1) WYNN RESORTS (MACAU) S.A. (the "Company");

## (2) BANK OF CHINA LIMITED, MACAU BRANCH (the "Term Facility Agent"); and

(3) THE TERM FACILITY LENDER (as defined below).

## WHEREAS:

The Term Facility Lender has agreed to make certain loan facilities available to the Company in connection with the Projects and the Cotai Project (including the design, development and construction of the Cotai Project), the refinancing of the Company's existing indebtedness and for general corporate purposes of the Group (including investment in Excluded Subsidiaries, Excluded Projects or Resort Management Agreements), in each case, upon the terms and subject to the conditions set out in this Agreement and the Common Terms Agreement.

## IT IS AGREED as follows:

## 1. DEFINITIONS AND INTERPRETATION

### 1.1 **Definitions**

In this Agreement, unless otherwise defined herein, all terms defined or referred to in the Common Terms Agreement shall have the same meaning herein and in addition:

"Available Commitment" means, in relation to a Term Facility Lender at any time and save as otherwise provided herein, the aggregate US dollar equivalent amount of Available Tranche A Commitment and Available Tranche B Commitment of such Term Facility Lender.

"Available Facility" means, at any time, the aggregate US dollar equivalent amount of the Available Tranche A Facility and the Available Tranche B Facility.

"Available Tranche A Commitment" means, in relation to a Term Facility Lender at any time, its Tranche A Facility Commitment less:

- (a) the aggregate amount of its participation in any outstanding Advances (including, without limitation, the Existing Advances) under that Facility; and
- (b) in relation to any Advance Request, the aggregate amount of its participation in any Tranche A Advance due to be made on or before the proposed Advance Date.

### - 8 -

"Available Tranche B Commitment" means, in relation to a Term Facility Lender at any time, its Tranche B Facility Commitment less:

- (a) the aggregate amount of its participation in any outstanding Advances (including, without limitation, the Existing Advances) under that Facility; and
- (b) in relation to any Advance Request, the aggregate amount of its participation in any Tranche B Advance due to be made on or before the proposed Advance Date.

"Available Tranche A Facility" means, at any time, the aggregate amount of the Available Tranche A Commitments of all the Term Facility Lenders at such time.

"Available Tranche B Facility" means, at any time, the aggregate amount of the Available Tranche B Commitments of all the Term Facility Lenders at such time.

"Common Terms Agreement" means the common terms agreement dated 14 September 2004 and made between, among others, the Company, the financial institutions defined therein as Term Facility Lenders and Revolving Credit Facility Lenders, the Term Facility Agent, the Revolving Credit Facility Agent, the Intercreditor Agent and the Security Agent, as amended and restated by the Common Terms Agreement Amendment Agreement, the Common Terms Agreement Second Amendment Agreement, the Common Terms Agreement Fourth Amendment Agreement.

"**Development Accounts**" means: (a) one or more Accounts of the Company opened and maintained solely for the disbursement of amounts (in excess of (i) the Permitted Amount and (ii) the amount required to pay any fees due and payable to any Senior Secured Creditor on the Fourth Amendment Effective Date under the Fourth Amendment Effective Date Fee Letters (to the extent such fees are to be funded from the Advances requested in the Completion Request)) borrowed by the Company under the Term Facility to finance or refinance costs incurred or funds expended on the development of the Cotai Project (and which does not, and will not, contain amounts to be applied for any other purpose other than as set forth in paragraph (b) of this definition) and; (b) one or more interest bearing demand or time deposits opened and maintained from time to time solely for the purpose of investing the amounts referred to in paragraph (a) of this definition prior to their disbursement *provided that* any such interest bearing demand or time deposit (i) meets the requirement of a Permitted Investment and (ii) is secured by a first ranking fixed Lien in favour of the Security Agent.

"Existing Advances" means the Tranche A Advance in an amount equal to USD6,833,333.32 and the Tranche B Advance in an amount equal to HKD426,400,000.00 outstanding as at the Fourth Amendment Effective Date (for the avoidance of doubt, this does not include any advances made pursuant to a Completion Request delivered on the Fourth Amendment Effective Date, as provided for in the Common Terms Agreement Fourth Amendment Agreement).

"HIBOR" means, in relation to any Tranche B Advance:

- (a) the applicable Screen Rate; or
- (b) (if no Screen Rate is available for HK dollars or for the Interest Period for that Term Facility Advance) the arithmetic mean of the rates (rounded upwards to four decimal places) as supplied to the Term Facility Agent at its request quoted by the Reference Banks to leading banks in the Hong Kong interbank market,

- 9 -

at or about 11.00 a.m. (Hong Kong time) on the Quotation Day for the offering of deposits in HK dollars for a period comparable to the Interest Period for that Term Facility Advance (or, if such Screen Rate or arithmetic mean of rates, as the case may be, is less than zero per annum, zero per annum).

"LIBOR" means, in relation to any Tranche A Advance:

- (a) the applicable Screen Rate; or
- (b) (if no Screen Rate is available for US dollars or for the Interest Period for that Term Facility Advance) the arithmetic mean of the rates (rounded upwards to four decimal places) as supplied to the Term Facility Agent at its request quoted by the Reference Banks to leading banks in the London interbank market,

at or about 11:00 a.m. (London time) on the Quotation Day for the offering of deposits in US dollars and for a period comparable to the Interest Period for that Term Facility Advance (or, if such Screen Rate or arithmetic mean of rates, as the case may be, is less than zero per annum, zero per annum).

"**Majority Term Facility Lenders**" means a Term Facility Lender or Term Facility Lenders whose US dollar equivalent participations in the Term Facility Advances then outstanding and undrawn Available Commitments amount in aggregate to more than 50% of the US dollar equivalent of the sum of all Term Facility Advances then outstanding and undrawn Available Commitments.

"Margin" means in relation to any Term Facility Advance hereunder:

- (a) on and from the Fourth Amendment Effective Date to and including the earlier of (i) the Quarterly Date falling on the last day of the second full Fiscal Quarter of the Company following the Fourth Amendment Effective Date and (ii) the date which is 180 days after the Fourth Amendment Effective Date, 2.50% per annum; and
- (b) thereafter, 1.75% per annum but, if the Leverage Ratio as at the most recent Quarterly Date thereafter is within the range set out below and the Intercreditor Agent has received, in accordance with paragraphs 1 and 2 of Part A of Schedule 5 (*Covenants*) of the Common Terms Agreement, the Company's financial statements for the period ending on such Quarterly Date together with the Compliance Certificate required thereunder then, provided (in the case of any decrease in the Margin) no Default has occurred and is continuing, the Margin will be the percentage per annum specified for that range:

| Leverage Ratio                                 | Margin |
|--|--------|
| Less than 3.0                                  | 1.75%  |
| Greater than or equal to 3.0 but less than 4.0 | 2.00%  |
| Greater than or equal to 4.0 but less than 4.5 | 2.25%  |
| 4.5 or above                                   | 2.50%  |

- 10 -

Any increase or decrease in the Margin shall take effect from the Business Day following the satisfaction of the conditions or expiry of the applicable periods specified above (or, where such Business Day falls less than five Business Days before the end of the then current Interest Period, from the commencement of the next Interest Period).

"**Party**" means a party to this Agreement.

"Permitted Amount" means the Permitted HKD Amount and the Permitted USD Amount.

"Permitted HKD Amount" means an amount of HKD2,060,500,000.00.

"Permitted USD Amount" means an amount of USD137,916,666.64

"Reference Banks" means, in relation to:

- (a) LIBOR, the principal London offices of Deutsche Bank AG, Standard Chartered Bank and The Royal Bank of Scotland plc; and
- (b) HIBOR, the principal Hong Kong offices of Bank of China Limited, Hong Kong Branch, Industrial and Commercial Bank of China (Asia) Limited and Standard Chartered Bank (Hong Kong) Limited,

or such other bank or banks designated from time to time by the Term Facility Agent provided that the consent of the Company shall be required if such designation is made prior to an occurrence of an Event of Default which is continuing.

"Screen Rate" means, in relation to:

- (a) LIBOR, the British Bankers' Association Interest Settlement Rate for US dollars for the relevant period, displayed on the appropriate page (being currently "LIBOR01") of the Reuters Monitor Money Rates Service screen; and
- (b) HIBOR, the rate designated as "FIXING@11:00" (or any other designation which may from time to time replace that designation or, if no such designation appears, the arithmetic average (rounded upwards, to four decimal places) of the displayed rates for the relevant period) appearing under the heading "HONG KONG INTERBANK OFFERED RATES (HK DOLLAR)" on the Reuters Screen HIBOR1=R Page.

If the agreed page is replaced or service ceases to be available, the Term Facility Agent may specify another page or service displaying the appropriate rate after consultation with the Company and the Term Facility Lenders.

"Term Facility" means the Tranche A Facility and the Tranche B Facility.

"**Term Facility Advance**" means, as the context may require, a Tranche A Advance or a Tranche B Advance and "**Term Facility Advances**" shall mean each Tranche A Advance and Tranche B Advance or any of them.

- 11 -

"Term Facility Lender" means a Tranche A Facility Lender or a Tranche B Facility Lender.

"Term Facility Finance Documents" means:

- (a) this Agreement;
- (b) the Common Terms Agreement;
- (c) any other Senior Finance Document to which a Term Facility Lender is a party in its capacity as a Term Facility Lender; and
- (d) any other document designated as such by the Term Facility Agent and the Company.

"Term Facility Finance Parties" means the Term Facility Agent and the Term Facility Lenders.

"Term Facility Loan" means the aggregate principal amount for the time being outstanding hereunder.

"**Tranche A Advance**" means an advance (as from time to time reduced by repayment in accordance with the terms hereof or the Common Terms Agreement) made or to be made by the Tranche A Facility Lenders under the Tranche A Facility.

"Tranche A Facility" means the US dollar term loan facility granted to the Company under Clause 3.1.1 (Tranche A Facility).

## "Tranche A Facility Commitment" means

- (a) in relation to the Original Tranche A Facility Lender, the amount set opposite its name under the column entitled "Tranche A Commitment" in Schedule 1 (*The Term Facility Lenders*) and the amount of any other Tranche A Facility Commitment transferred to it under this Agreement; and
- (b) in relation to any other Tranche A Facility Lender, the amount of any Tranche A Facility Commitment transferred to it under this Agreement,

to the extent not cancelled pursuant to clause 8 (*Repayments, Prepayments and Cancellation*) of the Common Terms Agreement or otherwise reduced pursuant to the terms hereof and/or the Common Terms Agreement, reduced or transferred by it in accordance with the Common Terms Agreement.

"Tranche A Facility Lender" means any commercial bank, financial institution or other entity which:

- (a) is named in Schedule 1 (*The Term Facility Lenders*) as a Tranche A Facility Lender (the "**Original Tranche A Facility Lender**"); or
- (b) has become party hereto as a Tranche A Facility Lender in accordance with Clause 13 (*Changes to the Parties*),

- 12 -

and which has not ceased to be a party hereto in accordance with the terms hereof.

**"Tranche B Advance**" means an advance (as from time to time reduced by repayment in accordance with the terms hereof or the Common Terms Agreement) made or to be made by the Tranche B Facility Lenders under the Tranche B Facility.

"Tranche B Facility" means the HK dollar term loan facility granted to the Company under Clause 3.1.2 (Tranche B Facility).

### "Tranche B Facility Commitment" means

- (a) in relation to the Original Tranche B Facility Lender, the amount set opposite its name under the column entitled "Tranche B Commitment" in Schedule 1 (*The Term Facility Lenders*) and the amount of any other Tranche B Facility Commitment transferred to it under this Agreement; and
- (b) in relation to any other Tranche B Facility Lender, the amount of any Tranche B Facility Commitment transferred to it under this Agreement,

to the extent not cancelled pursuant to clause 8 (*Repayments, Prepayments and Cancellation*) of the Common Terms Agreement or otherwise reduced pursuant to the terms hereof and/or the Common Terms Agreement, reduced or transferred by it in accordance with the Common Terms Agreement.

"Tranche B Facility Lender" means any commercial bank, financial institution or other entity which:

- (a) is named in Schedule 1 (*The Term Facility Lenders*) as a Tranche B Facility Lender (the "**Original Tranche B Facility Lender**"); or
- (b) has become party hereto as a Tranche B Facility Lender in accordance with Clause 13 (*Changes to the Parties*),

and which has not ceased to be a party hereto in accordance with the terms hereof.

### 1.2 Interpretation

In this Agreement:

- 1.2.1 the principles of construction contained in clause 1.2 (*Principles of Construction*) of the Common Terms Agreement and the rules of interpretation contained in clause 1.3 (*Rules of Interpretation*) of the Common Terms Agreement shall apply to the construction and interpretation of this Agreement;
- 1.2.2 any reference to the "**Term Facility Agent**" or "**Term Facility Lender**" shall be construed so as to include its or their (and any subsequent) successors and any permitted transferees in accordance with their respective interests; and
- 1.2.3 references in this Agreement to any Clause or Schedule shall be to a clause or schedule contained in this Agreement.

### 1.3 Third Party Rights

- 1.3.1 The Contracts (Rights of Third Parties) Act 1999 applies to Clause 1.4 (*Non-recourse Liability*) but only for the benefit of the Operatives and subject always to the terms of Clause 17 (*Governing Law*) and Clause 18 (*Jurisdiction*).
- 1.3.2 Except as provided in sub-clause 1.3.1 above, a Person who is not a party to this Agreement has no right under the Contracts (Rights of Third Parties) Act 1999 to enforce or to enjoy the benefit of any term of this Agreement.
- 1.3.3 Save as provided by the Common Terms Agreement, the consent of any Person who is not a party to this Agreement is not required to rescind or vary this Agreement.

## 1.4 Non-recourse Liability

Notwithstanding any provision in the Senior Finance Documents to the contrary, no Operative shall be personally liable for payments due hereunder or under any of the Senior Finance Documents or for the performance of any obligation hereunder or thereunder, save, in relation to any Operative, pursuant to any Senior Finance Document to which such Operative is party. The sole recourse of the Term Facility Finance Parties for satisfaction of any of the obligations of any of the Obligors hereunder and under the other Senior Finance Documents shall be against the Obligors, and not against any assets or property of any Operative save to the extent such Operative is party to a Senior Finance Document and is expressed to be liable for such obligation thereunder. In the case of Mr Wong Chi Seng, his liability shall be limited to his shares in the Company.

## 2. COMMON TERMS AGREEMENT

This Agreement and the rights and obligations of the parties hereto shall be subject to the terms and conditions of the Common Terms Agreement which shall be deemed to be incorporated into this Agreement. In the case of any conflict between the terms of this Agreement and the terms of the Common Terms Agreement, the terms of this Agreement shall prevail.

### 3. THE TERM FACILITY

# 3.1 Grant of the Term Facilities

### 3.1.1 Tranche A Facility

The Tranche A Facility Lenders grant to the Company, upon the terms and subject to the conditions hereof, a US dollar term loan facility in an aggregate amount of USD243,750,000.00.

## 3.1.2 Tranche B Facility

The Tranche B Facility Lenders grant to the Company, upon the terms and subject to the conditions hereof, a HK dollar term loan facility in an aggregate amount of HKD3,948,750,000.00.

- 14 -

## 4. **PURPOSE**

- 4.1.1 Subject to Clause 4.1.2, the Company shall apply all amounts borrowed by it under the Term Facility to pay for the refinancing of existing indebtedness of the Company, the design, development and construction of the Cotai Project and for the general corporate purposes of the Group (including investment in Excluded Subsidiaries, Excluded Projects or Resort Management Agreements).
- 4.1.2 The Company shall deposit (on the Fourth Amendment Effective Date) into:
  - (a) one or more Development Accounts denominated in HKD, all HKD amounts borrowed by it under the Term Facility in excess of (i) the Permitted HKD Amount and (ii) the amount required to pay any fees required to be paid in HKD and due and payable to any Senior Secured Creditor on the Fourth Amendment Effective Date under the Fourth Amendment Effective Date Fee Letters (to the extent such fees are to be funded from the Advances requested in the Completion Request); and
  - (b) one or more Development Accounts denominated in USD, all USD amounts borrowed by it under the Term Facility in excess of (i) the Permitted USD Amount and (ii) the amount required to pay any fees required to be paid in USD and due and payable to any Senior Secured Creditor on the Fourth Amendment Effective Date under the Fourth Amendment Effective Date Fee Letters (to the extent such fees are to be funded from the Advances requested in the Completion Request), and

shall only apply amounts withdrawn from the Development Accounts from time to time to finance or refinance costs incurred or funds expended on the development of the Cotai Project (including, without limitation, cost and expenses related to design, development, land acquisition, construction, site preparation, equipping, pre-opening expenses and capitalized interest) *provided that*, for the avoidance of doubt, the Company shall be entitled to transfer funds from one Development Account to another Development Account.

4.1.3 It shall be a condition to the Company making the first withdrawal from a Development Account (other than a withdrawal for the transfer of funds to another Development Account) and to making subsequent withdrawals from any Development Account (other than a withdrawal for the transfer of funds to another Development Account), that the Company shall (immediately prior to making the first such withdrawal and subsequently, for so long as there are any amounts standing to the credit of any Development Account, on the first Business Day in each calendar month thereafter) certify in writing to the Intercreditor Agent and the Term Facility Agent that it (and or Palo, as the case may be) has incurred costs or expended funds on the development of the Cotai Project (or will immediately upon the making of such withdrawal, have expended funds on the development of the Cotai Project) in an amount at least equal to the aggregate amount of all withdrawals from the Development Accounts (excluding withdrawals for the transfer of funds to another Development Account). For the purposes of this Clause 4.1.3 costs incurred or

- 15 -

funds expended on the development of the Cotai Project include, but are not limited to, costs and expenses related to design, development, land acquisition, construction, site preparation, equipping, pre-opening expenses and capitalized interest.

4.1.4 The Company shall notify the Intercreditor Agent and the Term Facility Agent of the details of each Development Account (including the bank with which such Development Account is held, in which currency such Development Account is denominated and the balance from time to time) promptly upon the opening of any such Development Account and from time to time upon request of the Intercreditor Agent and the Term Facility Agent.

## 5. CONDITIONS PRECEDENT

The provisions of clause 2 (*Conditions Precedent*) of the Common Terms Agreement are incorporated by reference herein as if the same were set out in full herein.

## 6. AVAILABILITY OF THE TERM FACILITY

## 6.1 Drawdown of Advances

The provisions of clause 3 (*Drawdown of Advances*) and clause 4.1 (*Term Facility Availability Period*) of the Common Terms Agreement are incorporated by reference herein as if the same were set out in full herein.

### 6.2 Each Term Facility Lender's Participation

- 6.2.1 It is acknowledged and agreed that, as at the Fourth Amendment Effective Date, the Existing Advances have been made and are outstanding hereunder.
- 6.2.2 Each Tranche A Facility Lender will participate through its Facility Office in each Tranche A Advance made pursuant to Clause 6.1 (*Drawdown of Advances*) in the proportion borne by its Available Tranche A Commitment to the Available Tranche A Facility immediately prior to the making of that Tranche A Advance.
- 6.2.3 Each Tranche B Facility Lender will participate through its Facility Office in each Tranche B Advance made pursuant to Clause 6.1 (*Drawdown of Advances*) in the proportion borne by its Available Tranche B Commitment to the Available Tranche B Facility immediately prior to the making of that Tranche B Advance.

## 6.3 Reduction of Available Commitment

If a Term Facility Lender's Available Tranche A Commitment or, as the case may be, Available Tranche B Commitment is reduced in accordance with the terms hereof or the Common Terms Agreement after the Intercreditor Agent or the Term Facility Agent has received an Advance Request for a Tranche A Advance or, as the case may be, a Tranche B Advance and such reduction was not taken into account in the Available Tranche A Facility or, as the case may be, the Available Tranche B Facility, then the amount of that Tranche A Advance or, as the case may be, Tranche B Advance shall be reduced accordingly.

- 16 -

# 7. **REPAYMENT**

## 7.1 **Repayment**

Subject to Clause 7.2 (*Final maturity*), the Company shall repay the Term Facility Loans in quarterly instalments by repaying on each Repayment Date amounts equal to the percentage set out next to the relevant Repayment Date in Schedule 2 (*Repayment Schedule*) of the aggregate US dollar denominated Term Facility Advances and the aggregate HK dollar denominated Term Facility Advances outstanding as at the end of the last day of the Term Facility Availability Period.

# 7.2 Final maturity

The Company shall repay on the Final Repayment Date all amounts outstanding or due and payable under the Term Facility on that day.

## 7.3 No re-borrowing

The Company may not re-borrow any part of the Term Facility which is repaid.

## 8. PREPAYMENT AND CANCELLATION

All prepayments of Term Facility Advances and cancellation of Available Commitments shall be made in accordance with clause 8 (*Repayments, Prepayments and Cancellation*) of the Common Terms Agreement.

## 9. INTEREST

## 9.1 Calculation of Interest

The rate of interest on each Term Facility Advance for each Interest Period is the percentage rate per annum which is the aggregate of:

- 9.1.1 the Margin; and
- 9.1.2 LIBOR (in the case of a Tranche A Advance) or HIBOR (in the case of a Tranche B Advance).

# 9.2 **Payment of interest**

Accrued interest on each Term Facility Advance is payable by the Company on the last day of each Interest Period relating to that Term Facility Advance.

## 9.3 Default Interest

Default interest shall be calculated and paid in accordance with clause 9.4 (Default Interest) of the Common Terms Agreement.

- 17 -

### 10. INTEREST PERIODS

The duration of each Interest Period shall be determined in accordance with clause 9.3 (Interest Periods) of the Common Terms Agreement.

### 11. NOTIFICATION

### 11.1 Advances

Promptly, and in any event, not less than 4 Business Days before the proposed Advance Date for each Term Facility Advance, the Term Facility Agent shall notify each Term Facility Lender of the proposed amount of the relevant Term Facility Advance and the aggregate principal amount of the relevant Term Facility Advance allocated to such Term Facility Lender pursuant to Clause 6.2 (*Each Term Facility Lender's Participation*) and each Term Facility Lender shall, on such Advance Date, subject to the terms and conditions of this Agreement, make available to the Term Facility Agent for the account of the Company its said portion of such Term Facility Advance.

## 11.2 Interest rate determination

The Term Facility Agent shall promptly notify the Company and the Term Facility Lenders of each determination of LIBOR and HIBOR under this Agreement.

### 11.3 Changes to interest rates

The Term Facility Agent shall promptly notify the Company and the Term Facility Lenders of any change to any interest rate occasioned by the operation of clause 10 (*Changes to the Calculation of Interest*) of the Common Terms Agreement.

### 11.4 Interest payment and repayment instalments

Without prejudice to the Company's obligation to make any interest payment or to pay any repayment instalment on the due date, the Term Facility Agent shall provide to the Company and each Term Facility Lender (with a copy to the Intercreditor Agent) a notice setting out the relevant scheduled payment of interest and scheduled repayment of principal under this Agreement at least 15 Business Days before such amounts fall due for payment by the Company.

### 12. [NOT USED]

## 13. CHANGES TO THE PARTIES

### 13.1 Transfers by the Term Facility Agent

The Term Facility Agent may resign in accordance with the Common Terms Agreement and may assign and transfer all of its rights and obligations under the Term Facility Finance Documents to a replacement Term Facility Agent appointed in accordance with the terms of the Common Terms Agreement.

### 13.2 Transfers by the Company

The Company may not assign, transfer, novate or dispose of any of its rights or obligations under the Term Facility Finance Documents.

- 18 -

### 13.3 Transfers by the Term Facility Lenders

A Term Facility Lender may assign, transfer or novate any of its rights and/or obligations under the Term Facility Finance Documents in accordance with clause 21.4 (*Assignment and Transfer by Lenders*), clause 21.5 (*Assignments by Lenders*) and clause 21.6 (*Transfers by Lenders*) of the Common Terms Agreement.

## 13.4 Assignment and Transfer Fees

On the date upon which an assignment takes effect pursuant to clause 21.5 (*Assignments by Lenders*) of the Common Terms Agreement or a transfer takes effect pursuant to clause 21.6 (*Transfers by Lenders*) of the Common Terms Agreements, the relevant assignee or Transferee shall pay to the Intercreditor Agent for its own account a fee in accordance with clause 21.7 (*Assignment and Transfer Fees*) of the Common Terms Agreement.

### 14. PAYMENTS

### 14.1 Payments

- 14.1.1 All payments under this Agreement shall be made in accordance with Clause 26 (*Payment Mechanics*) of the Common Terms Agreement.
- 14.1.2 Subject to clause 26 (*Payment Mechanics*) of the Common Terms Agreement, on each date on which this Agreement requires an amount to be paid by the Company or a Term Facility Lender, the Company or, as the case may be, such Term Facility Lender shall make the same available to the Term Facility Agent for value on such due date and at such time and in such funds and to such account with such bank as the Term Facility Agent shall specify from time to time.

### 14.2 **Partial Payments**

- 14.2.1 If the Term Facility Agent receives a payment that is insufficient to discharge all the amounts then due and payable by the Company to the Term Facility Lenders under the Term Facility Finance Documents, the Term Facility Agent shall apply that payment towards the obligations of the Company under the Term Facility Finance Documents in the following order:
  - (a) **first**, in or towards payment *pro rata* of all amounts paid by the Term Facility Lenders under clause 23.15 (*Indemnity to Intercreditor Agent*) of the Common Terms Agreement but which have not been reimbursed by the Company;
  - (b) **secondly**, in or towards payment *pro rata* of all amounts paid by the Term Facility Lenders under Clause 15.3 (*Indemnity to Term Facility Agent*) but which have not been reimbursed by the Company;
  - (c) **thirdly**, in or towards payment *pro rata* of all costs and expenses incurred by the Term Facility Lenders which the Company is obliged to reimburse;

- 19 -

- (d) **fourthly**, in or towards payment *pro rata* of all accrued but unpaid fees and commissions due to the Term Facility Lenders under the Term Facility Finance Documents;
- (e) **fifthly**, in or towards payment *pro rata* of all accrued but unpaid interest (including default interest) due to the Term Facility Lenders under the Term Facility Finance Documents;
- (f) **sixthly**, in or towards payment *pro rata* of any principal due to the Term Facility Lenders under the Term Facility Finance Documents but unpaid; and
- (g) **seventhly**, in or towards payment *pro rata* of any other sum due to the Term Facility Lenders under the Term Facility Finance Documents but unpaid.
- 14.2.2 The Term Facility Agent shall, if so directed by the Majority Term Facility Lenders, vary the order set out in sub-clause 14.2.1 above.
- 14.2.3 Sub-clause 14.2.1 above will override any appropriation made by the Company.

### 15. DECISION MAKING AMONGST TERM FACILITY LENDERS

### 15.1 Decisions

Save as otherwise set out herein and subject to the Common Terms Agreement, the required Senior Secured Creditors for the purpose of any decision within the scope of clause 34.2 (*Amendment and waiver of Facility Agreements*) of the Common Terms Agreement) relating to this Agreement shall be the Term Facility Agent acting on the instructions of the Majority Term Facility Lenders.

### 15.2 Failure to Give Instructions

If the Term Facility Agent gives notice to the Term Facility Lenders requesting their specific instructions on any matter referred to in Clause 15.1 (*Decisions*) and it specifies in such notice that the Term Facility Lenders are to give such instructions by a certain date and time specified in such notice, any Term Facility Lender which fails to respond by the date and time so specified shall have its portion of the Term Facility Advances and its Available Commitment disregarded for all purposes of determining whether instructions have been given to the Term Facility Agent by the Majority Term Facility Lenders (and, for the purposes of determining the Available Facility or the amount of all Term Facility Advances outstanding, the Available Commitments and portion of Term Facility Advances of such Term Facility Lender shall be deducted).

### 15.3 Indemnity to Term Facility Agent

15.3.1 Each Term Facility Lender shall, rateably in accordance with the proportion that the US dollar equivalent of the sum of its Available Commitments and its participations in any outstanding Term Facility Advances bear to the US dollar equivalent of the aggregate of the Available Commitments and such participations of all the Term Facility Lenders (or, if all such amounts have

- 20 -

been reduced to zero, such proportion determined immediately prior to such reduction) for the time being, indemnify the Term Facility Agent, within fifteen days of demand, against any cost, loss or liability incurred by the Term Facility Agent (other than by reason of the negligence or wilful misconduct of the Term Facility Agent) in acting as Term Facility Agent under any of the Senior Finance Documents (unless the Term Facility Agent has been reimbursed by the Company pursuant to a Senior Finance Document).

15.3.2 Provided that the Company is required to reimburse or indemnify the Term Facility Agent for such cost, loss or liability in accordance with the terms of the Senior Finance Documents, the Company shall, within fifteen days of demand in writing by any Term Facility Lender, indemnify such Term Facility Lender in relation to any payment actually made by such Term Facility Lender pursuant to Clause 15.3.1 above.

### 16. COUNTERPARTS

This Agreement may be executed in any number of counterparts, all of which taken together shall constitute one and the same instrument.

## 17. GOVERNING LAW

This Agreement and any non-contractual obligations arising out of or in connection with it are governed by English law.

### 18. JURISDICTION

## 18.1 Jurisdiction of English courts

- 18.1.1 The courts of England have exclusive jurisdiction to settle any dispute arising out of or in connection with this Agreement (including a dispute regarding the existence, validity or termination of this Agreement or the consequences of its nullity) (a "**Dispute**").
- 18.1.2 The parties hereto agree that the courts of England are the most appropriate and convenient courts to settle Disputes and accordingly they will not argue to the contrary.
- 18.1.3 This Clause 18.1 is for the benefit of the Term Facility Finance Parties only. As a result, no Term Facility Finance Party shall be prevented from taking proceedings relating to a Dispute in any other courts with jurisdiction. To the extent allowed by law and the Senior Finance Documents, the Term Facility Finance Parties may take concurrent proceedings in any number of jurisdictions.

- 21 -

## 18.2 Service of process

Without prejudice to any other mode of service allowed under any relevant law, the Company:

- 18.2.1 irrevocably appoints Law Debenture Corporate Services Limited as its agent for service of process in relation to any proceedings before the English courts in connection with this Agreement; and
- 18.2.2 agrees that failure by a process agent to notify the Company of the process will not invalidate the proceedings concerned.

This Agreement has been entered into on the date stated at the beginning of this Agreement.

### SCHEDULE 1

## THE TERM FACILITY LENDERS

| Term Facility Lender                | Tranche A<br>Commitment (USD) | Tranche B<br>Commitment (HKD) |
|-------------------------------------|-------------------------------|-------------------------------|
| Bank of China Limited, Macau Branch | 243,750,000                   | 3,948,750,000                 |
| Total                               | 243,750,000                   | 3,948,750,000                 |

- 23 -

### **SCHEDULE 2**

# **REPAYMENT SCHEDULE**

| Percentage (%) |
|----------------|
| 50%            |
| 50%            |
|                |

- 24 -

## SIXTH AMENDED AND RESTATED

# ART RENTAL

### AND

## LICENSING AGREEMENT

### between

## STEPHEN A. WYNN

## (Lessor)

### and

# WYNN LAS VEGAS, LLC

### (Lessee)

## Dated July 1, 2012

#### SIXTH AMENDED AND RESTATED ART RENTAL AND LICENSING AGREEMENT

This Sixth Amended and Restated Art Rental and Licensing Agreement ("Agreement"), is entered into this 1st day of July, 2012 (the "Effective Date"), by and between STEPHEN A. WYNN ("Lessor") and WYNN LAS VEGAS, LLC ("Lessee").

#### RECITALS

A. Lessor is the owner of the paintings and other art works identified in <u>Exhibit A</u> attached hereto and incorporated herein by this reference (collectively, the "**Works**"), which may be updated from time to time in accordance with this Agreement. Works shall only include Works deemed to be on display by Lessee as provided in Section 7.

B. Lessor wishes to lease to Lessee, and Lessee wishes to lease from Lessor, the Works, in order to publicly display the Works at various public locations within the Wynn Las Vegas/Encore resort at 3131 Las Vegas Boulevard South, Las Vegas, Nevada as well as the aircraft that is owned by affiliates of Lessee (the "**Resort**"), .

C. By publicly displaying the Works, Lessor and Lessee desire to promote the Works and to enhance the cultural and educational opportunities for Nevada residents and visitors.

### AGREEMENT

Based upon the foregoing and the following terms and conditions, the parties hereto agree that the foregoing recitals are true and correct and as follows:

1. <u>Rental</u>. Upon the terms and subject to the conditions of this Agreement, Lessor hereby grants to Lessee a continuing right to publicly display the Works in the Resort, and Lessee hereby accepts from Lessor the rental of the works. The installation layout and plan for the Works shall be subject to Lessor's prior approval.

2. <u>Compliance with Law</u>. Lessee agrees to maintain the Works on public display, make the Resort available for student tours, and take such other actions as may be necessary or appropriate for meeting the requirements of Sections 361.068, 361.186, 374.291 and 374.2911 of the Nevada Revised Statutes ("**NRS**"), and agrees to comply with NRS 597.720, <u>et seq</u>., and all other applicable laws.

3. <u>Exhibition and Promotion</u>. Lessee agrees (a) to exhibit the Works in a manner approved by Lessor, (b) to transport, handle, care for, and display the Works in a manner consistent with the world-class quality of the Works, (c) to maintain the Resort as a first-class facility, and (d) to promote the Works through "Openings," "Receptions," and public events.

4. <u>Merchandising</u>. To the extent, if any, that he possesses the required rights, Lessor hereby authorizes Lessee to develop, manufacture (by subcontract or otherwise), and sell such merchandising and promotional items based upon the Works as Lessee may determine in its best business judgment. To the extent, if any, that he possesses any such rights in any Work, Lessor hereby grants to Lessee a nonexclusive license for such purposes for the period of the rental of such Work hereunder. In the event that any Work is withdrawn or rental terminated, the corresponding license shall automatically terminate; *provided, however*, that following such termination, Lessee shall have the right to sell its remaining inventory of the applicable merchandise. The merchandise and promotional items based on the Works may include, but are not limited to, educational catalogues, educational works (including audiovisual and audio recordings), fine art reproductions, and retail merchandise based upon the Works. Lessee shall be solely responsible for clearing and/or obtaining such rights, for obtaining all required permissions, and for taking all reasonable steps necessary to obtain intellectual property protection for said items based on the Works, all of which shall, with respect to any Work, inure to the benefit of Lessee during the rental of such Work hereunder and to the benefit of Lessor thereafter. Notwithstanding any other

- 3 -

provision of this Agreement (including without limitation this Section 4 and Section 11 below), Lessor does not make (and hereby disclaims) any and all representations and/or warranties to Lessee or otherwise in respect of the Works or any rights in the Works, including, but not limited to, title, quiet enjoyment, authenticity, copyright, or moral rights. Lessor shall not have any liability to Lessee in respect of any, and Lessee hereby expressly and to the full extent permitted by law waives as against Lessor all, claims. damages, expenses, fees, or losses that may be incurred by or threatened against Lessee as a result of the Works being leased to Lessee, in the possession of Lessee during the term hereof, displayed at the Resort and/or reproduced (by, on behalf of, or with the consent of Lessee) in merchandising, promotional, or other items relating to the Works.

5. <u>Rental Fees</u>. Lessee agrees to pay to Lessor a rental fee of One Dollar (\$1.00) on the Effective Date and each anniversary of the Effective Date.

6. <u>Additions, Withdrawals, and Termination</u>. Lessor and Lessee may, by mutual agreement, add other art works from time to time to the Works covered by this Agreement. Lessor shall have the right to withdraw any but not all of the Works from this Agreement and terminate the rental of such Work(s) hereunder on fifteen (15) days' written notice to Lessee. Lessee shall have the right to return any or all Works covered by this Agreement and terminate the rental of such Work(s) hereunder on thirty (30) days' written notice to Lessor. Upon termination of the rental of any Work hereunder, Lessee shall have no further right or license with respect to such Work, except to the extent that, under Section 4 and Section 11 hereof, Lessee is specifically provided with a six-month period to discontinue sales and use of merchandise. The parties shall periodically update <u>Exhibit A</u> hereto to reflect Works added to or withdrawn from this Agreement. The parties shall also maintain separate records indicating which Works are added to or withdrawn from the Agreement. Notwithstanding the foregoing, and without prejudice to any other rights or remedies that Lessor may have hereunder,

- 4 -

Lessor may terminate the rental of all Works hereunder (a) by delivery of notice to Lessee no less than ninety (90) days in advance of the date selected by Lessor for termination, or (b) immediately by delivery of notice to Lessee at any time if any of the following events occurs: (i) Lessor ceases to be the Chairman of the Board and Chief Executive Officer of Wynn Resorts, Limited ("**Wynn Resorts**"), or any successor company; (ii) there is filed any petition in bankruptcy by or against Lessee or any of Wynn Resorts' other subsidiary companies (collectively, the "Lessee Parties"), which petition is not dismissed within ninety (90) days of its filing, or there is appointed a receiver or trustee to take possession of any of the Lessee Parties or of all or substantially all of the assets of any Lessee Parties, or there is a general assignment by any of the Lessee Parties for the benefit of creditors, or any action is taken by or against any of the Lessee Parties under any state or federal insolvency or bankruptcy act, or any similar law now or hereafter in effect, including, without limitation, the filing of execution or attachment against any of the Lessee Parties and such levy continues in effect for a period of sixty (60) calendar days; or (iii) Lessee defaults under any material provision of this Agreement and fails to cure such default within seven (7) days after it receives written notice of such default from Lessor. Upon termination of the rental of all Works hereunder, this Agreement shall automatically terminate; provided, however, that the termination of this Agreement shall not affect the obligations of the parties under Sections 5, 7, 9 or 10 hereof or any other provision that can be fulfilled only after the termination date.

7. <u>Insurance</u>. Lessee shall, at its expense, insure the Works on a "wall-to-wall" basis, for the full rental period hereunder (including, without limitation, terrorism insurance), in an amount equal to the value of the Works as stipulated in writing by Lessor, provided, however that Lessee shall not be responsible for insuring the Works

during any period when the Works are not on display or are in transit to or from the Resort (the "**Non-Displayed Period**"). The parties agree that for purposes of this Section 7 the Non-Display Period shall also include, and the Works shall be deemed to not be on display, any time the Works are on display in the offices of Lessor (including adjacent conference room) or Lessor's executive assistants. The parties also agree that for purposes of this Section 7, the Works shall be deemed to on display any time the Works appear in any public area of the Resort, including within the offices of any executive located in the Resort, other than the offices of Lessor (including adjacent conference room) or Lessor's executive assistants. Upon reasonable prior notice to Lessee, Lessor shall have the right from time to time to reasonably increase the stated value of any one or more of the Works, and require Lessee to increase the amount of insurance required by this Section 7. Lessor shall be named as an additional insured on Lessee's insurance policy. A certificate of insurance and a copy of those portions of the insurance policy covering the Works and setting forth any exclusions to coverage shall be furnished by Lessee to Lessor, and shall be subject to Lessor's reasonable approval as to form and content (including, without limitation, any deductible). The foregoing insurance policy shall include coverage against all risk of physical loss or damage from any external cause while in transit and on location in the Resort during the rental period hereunder. Lessee shall bear sole responsibility and shall be liable to Lessor for all loss, damage, or destruction of the Works, regardless of fault or the degree of care exercised by Lessee, and regardless of the presence or supervision of, or any direction or approval by, Lessor or any Lessor's representative; provided, however, that Lessee's liability in the event of such loss, damage, or destruction shall not exceed the value of

- 6 -

the Works as stipulated in writing by Lessor. Lessee shall be responsible to pay any and all deductibles relating to the insurance coverage required by this Section 7. In the event any Work is lost or stolen, and then recovered after Lessor has obtained insurance proceeds, Lessor shall have the option to exchange those insurance proceeds for such Work. In the event any Work is damaged but not destroyed, Lessee agrees to be responsible for both the cost of repairing and restoring such Work and the loss in value of such Work as determined by an appraiser mutually agreed upon by the parties.

8. Security. Lessee agrees to take all reasonable steps necessary to secure and protect the Works from loss, theft or injury and to treat them in a manner consistent with maintaining its own most valuable assets at all times the Works are in its possession, control or custody, or in transit to or from Lessor. Without limiting the generality of the foregoing, Lessee shall provide for an adequate number of guards to be on duty in and around the Resort at all times while the Works are in the Resort. All Works shall be within direct sight lines of at least one guard and under direct video surveillance at all times during the rental period hereunder; provided however that with respect to Works located on aircraft, Lessee shall ensure that the Aircraft remains in a secure area at any time any of the Works are placed on the Aircraft. Lessee shall comply with further reasonable security restrictions and arrangements as directed in writing by Lessor. Lessee represents and warrants to Lessor that the Resort is and shall be equipped with adequate fire detection/prevention systems and protected by alarm systems that are activated at all times.

9. <u>Indemnification</u>. Lessee shall indemnify, defend, protect, and hold harmless Lessor, his agents, heirs, assigns, and successors (collectively, "**Indemnitees**") from and against any and all claims, damages, liabilities, losses, actions, complaints, or judgments, including, without limitation, attorneys' fees, threatened against, incurred, or suffered by the Indemnitees, arising out of Lessee's

- 7 -

breach of or failure to perform, under this Agreement, the inaccuracy when made of any representation or warranty made by Lessee, or any act or omission by or on behalf of Lessee or its respective officers, agents, employees, contractors, or representatives, relating to the Works or this Agreement.

10. <u>Taxes</u>. Lessee shall pay all of the following Nevada state and local taxes, along with all interest, penalties, and other additions related thereto: (a) sales and use taxes applicable to the rental of the Works pursuant hereto; and (b) except to the extent provided otherwise in the following sentence, personal property taxes applicable to each of the Works for each fiscal year during which Lessee is renting such Work hereunder. In the event that Lessor withdraws any Work from this Agreement and terminates the rental of such Work hereunder pursuant to the second sentence of Section 6 hereof, Lessor and Lessee shall make an equitable allocation of the personal property taxes applicable to such Work for the fiscal year in which such withdrawal occurs. For purposes of explanation and clarification of this Section 10, the personal property tax obligations of Lessee shall include Works that are on display in the offices of any executive located in the Resort, other than offices of Lessor (including adjacent conference room) or Lessor's executive assistants.

11. <u>Intellectual Properties</u>. Lessor consents to the photography, filming, videotaping and recordation of the Works for the purpose of obtaining photographic and other copyrights in the new derivative works, which shall be owned and controlled by Lessor, but which is hereby licensed to Lessee for use in advertising, promotion, and merchandising of, and education relating to, the Works, such license to run concurrently, with respect to any Work, with the rental of such Work hereunder. In the event that any Work is withdrawn or rental terminated, the corresponding license shall automatically terminate; provided, however, that following such termination, Lessee shall have six (6) months to discontinue sales and use of the applicable merchandise.

12. Notice. Any notice to be given pursuant to this Agreement by either party

- 8 -

to the other shall be in writing and may be effected either by personal delivery, by mail, registered or certified, postage prepaid, with return receipt requested, or by facsimile. Notice by mail shall be sent concurrently with any facsimile notice. Notices shall be addressed to the parties at the address specified below, but each party may change such party's address by written notice in accordance with this Section 12. Notices delivered personally shall be deemed communicated as of actual receipt; and mailed notices and facsimile notices (with a concurrent mailing) shall be deemed communicated three (3) days after mailing. Notices shall be given as follows:

Wynn Las Vegas, LLC Legal Department 3131 Las Vegas Boulevard South Las Vegas, Nevada 89109 Telephone: 702-770-7000 Mr. Stephen A. Wynn c/o Wynn Las Vegas, LLC 3131 Las Vegas Boulevard South Las Vegas, Nevada 89109 Telephone: 702-770-7700

13. <u>Entire Agreement; Amendment</u>. This Agreement constitutes the entire agreement between the parties and supersedes any oral or written communications between Lessor and Lessee, with respect to its subject matter, including, without limitation, that certain Fifth Amended and Restated Art Rental Licensing Agreement, as amended, between Lessor and Lessee. This Agreement may be amended only if such amendment is set forth in writing and executed by each of the parties.

14. <u>Governing Law</u>. This Agreement shall in all respects be construed according to the laws of the State of Nevada, regardless of the choice or conflict of laws provisions of Nevada or any other jurisdiction.

15. <u>Assignment; Binding Effect</u>. Lessor may assign any or all of his rights and obligations under this Agreement. Lessee may not assign all or any portion of its rights or obligations under this Agreement without Lessor's prior consent. Subject to the preceding sentence, this Agreement shall be binding upon, inure to the benefit of, and

- 9 -

be enforceable by the parties hereto and their respective heirs, legal representatives, successors, and assigns. This Agreement does not create, and shall not be construed or deemed to create, any rights or benefits enforceable by or for the benefit of any person or entity other than the parties hereto and their respective heirs, legal representatives, successors, and permitted assigns.

16. <u>Headings; Context</u>. All headings herein are inserted only for convenience and ease of reference and shall not be considered in the construction or interpretation of any provision of this Agreement. Whenever used in this Agreement, the singular shall include the plural and the plural shall include the singular, and the neuter gender shall include the male and female as well as an entity, all as the context and meaning of this Agreement may require.

17. <u>Counterparts</u>. This Agreement may be executed in counterparts, each of which shall be deemed an original, but all of which shall constitute one and the same instrument.

#### [Signature page follows.]

- 10 -

IN WITNESS WHEREOF, the parties hereto have executed this Agreement as of the Effective Date.

"Lessor"

/s/ Stephen A. Wynn

Stephen A. Wynn

"Lessee"

Wynn Las Vegas, LLC, a Nevada limited liability company

| By:   | /s/ Marilyn Spiegel |
|-------|---------------------|
| Name: | Marilyn Spiegel     |
| Its:  | President           |

- 11 -

# EXHIBIT A

E.

# WORKS OF ART

| Description  | Artist           | Location   |  |
|--|------------------|--|--|
| Magritte II,1998 - Sculpture                                     | Escobar, Marisol | Wynn Las Vegas Executive Offices - Lobby                       |  |
| Blue Harbor, 1996  | Gilot, Francoise | Outside Kim Sinatra's Office                                   |  |
| Kutani Porcelain Vase With 47 Ronin Design                       | Kutani, Koko     | Mizumi Restaurant Wynn Las Vegas                               |  |
| Nude After Bathus  | Ludwig, Daniel   | Wynn Las Vegas - Executive Offices<br>(John Strzemp' s Office) |  |
| Jerry-Rigged, 2011 Lacquer on Canvas                             | Pittman, Lari    | In BBJ Aircraft  |  |
| Makeshift, 2011 Lacquer on Canvas                                | Pittman, Lari    | In BBJ Aircraft  |  |
| Pavane, 2011 Lacquer on Canvas                                   | Pittman, Lari    | By Laurie Nady's Desk  |  |
| Yoshino (Area in Japan Where Cherry Blossoms Bloom, size 7' x4') | Ryu, Seiho       | Executive Conference Room                                      |  |
| Composition with Glass   | Yasani, Masoud   | Marc Schorr's Office - Wynn Las Vegas                          |  |
| Imari Vase   |                  | Mizumi Restaurant Wynn Las Vegas                               |  |

### Certification of the Chief Executive Officer Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002

I, Stephen A. Wynn, certify that:

- 1. I have reviewed this Quarterly Report on Form 10-Q of Wynn Resorts, Limited;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f) for the registrant and have:
  - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of registrant's board of directors (or persons performing the equivalent functions):
  - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: November 9, 2012

/s/ Stephen A. Wynn

Stephen A. Wynn Chairman of the Board and Chief Executive Officer (Principal Executive Officer)

### Certification of the Chief Financial Officer Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002

I, Matt Maddox, certify that:

- 1. I have reviewed this Quarterly Report on Form 10-Q of Wynn Resorts, Limited;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f) for the registrant and have:
  - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of registrant's board of directors (or persons performing the equivalent functions):
  - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: November 9, 2012

/s/ Matt Maddox

Matt Maddox Chief Financial Officer and Treasurer (Principal Financial Officer)

### Certification of CEO and CFO Pursuant to 18 U.S.C. Section 1350, as Adopted Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002

In connection with the Quarterly Report on Form 10-Q of Wynn Resorts, Limited (the "Company") for the quarter ended September 30, 2012 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), Stephen A. Wynn, as Chief Executive Officer of the Company and Matt Maddox, as Chief Financial Officer of the Company, each hereby certifies, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that, to the best of his knowledge:

- 1. The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- 2. The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

#### /s/ Stephen A. Wynn

Name: Stephen A. Wynn Title: Chairman and Chief Executive Officer (Principal Executive Officer) Date: November 9, 2012

/s/ Matt Maddox

Name: Matt Maddox Title: Chief Financial Officer and Treasurer (Principal Financial Officer) Date: November 9, 2012

A signed original of this written statement required by Section 906, or other document authenticating, acknowledging, or otherwise adopting the signature that appears in typed form within the electronic version of this written statement required by Section 906, has been provided to Wynn Resorts, Limited and will be retained by Wynn Resorts, Limited and furnished to the Securities and Exchange Commission or its staff upon request.